#### 

<u>Sec. Research Graduate School Committee</u> Under the Copyright Act 1968, this thesis must be used only under the normal conditions of scholarly fair dealing for the purposes of research, criticism or review. In particular no results or conclusions should be extracted from it, nor should it be copied or closely paraphrased in whole or in part without the written consent of the author. Proper written acknowledgement should be made for any assistance obtained from this thesis.

# G60/134

#### ERRATA AND ADDENDUM

- p. 24 The following paragraph should be inserted immediately following the statement of the Clifford-McLean theorem for bands (Theorem 1.3.14): For any band, the relations  $\mathcal{L}$  and  $\mathcal{R}$  commute [111, Proposition 2.1.3]. From [111, Corollary 1.5.12] it follows that  $\mathcal{D} = \mathcal{L} \circ \mathcal{R} = \mathcal{R} \circ \mathcal{L} = \mathcal{L} \vee \mathcal{R}$ . We remark that these equalities are fundamental when manipulating  $\mathcal{D}$  on any band.
- p. 28 Paragraph 1.4.3, Line 7. The sentence:
  Leech has shown that the class of quasilattices is a variety [145, Section 1], ... should read:
  Lasio and Leech have shown that the class of quasilattices is a variety [145, Section 1], ...
- p. 28 Paragraph 1.4.3, Line 10. The sentence:

Leech has also shown that quasilattices satisfy a modified form of the Clifford-McLean theorem:  $\dots$ 

should read:

Laslo and Leech have also shown that quasilattices satisfy a modified form of the Clifford-McLean theorem: ...

p. 28 Paragraph 1.4.3, Line 12. The sentence:

Further information about quasilattices may be found in Leech [145, Sections 1, 2, 3, 4 and 6].

should read:

Further information about quasilattices may be found in Laslo and Leech [145, Sections 1, 2, 3, 4 and 6].

p. 28 Paragraph 1.4.4, Line 6. The sentence:

Leech has observed that the class of paralattices is a variety [145, Section 1],  $\ldots$  should read:

Laslo and Leech have observed that the class of paralattices is a variety [145, Section 1],  $\ldots$ 

- p. 31 The following paragraph should be inserted immediately following the statement of the Clifford-McLean theorem for skew lattices (Theorem 1.4.10):
   The Clifford-McLean theorem for skew lattices is also known as the *first decomposition theorem for skew lattices* in the literature.
- p. 60 Identity (1.34) should read:
  - $(x \div y) \div ((x \div (y \div z)) \approx 0 \tag{1.34}$

p. 105 Line 4. The sentence:

This is shown in the following result, which may be understood as a kind of a 'Clifford-McLean theorem' for pre-BCK-algebras.

should read:

This is shown by the following 'maximal image' result, which may be understood as a pre-cursor to a Clifford-McLean theorem for pre-BCK-algebras. In the sequel we shall in fact see that, under appropriate conditions, Theorem 2.1.14 *merges* with the usual assertion of the Clifford-McLean theorem for bands (Theorem 1.3.14).

#### ERRATA AND ADDENDUM

p. 135 The first two sentences of Paragraph 2.2.16 should read:

**2.2.16.** Positive Implicative Pre-BCK-Algebras. A positive implicative pre-BCK-algebra is a pre-BCK-algebra A such that  $A/\Xi \cong B$  for some positive implicative BCK-algebra B.

p. 299 The first sentence of Remark 3.3.9 should read:

One-sided non-commutative lattices were introduced by Laslo and Leech in [145, Section 4] under the name *flat* non-commutative lattices, in conformance with standard non-commutative lattice theory terminology.

p. 305 The first sentence of the paragraph immediately following Theorem 3.3.15 should read:

An upper implicative BCS band is an algebra  $\langle A; \vee, \backslash, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \vee, 0 \rangle$  is an upper band with zero; ...

**p. 306** Identity (3.72) should read:  $x \setminus (x \setminus (x \lor y \lor x)) \approx x$ 

(3.72)

- p. 307 Top line. The statement:
  PQ<sub>C'</sub> denote the variety of upper pre-BCK-bands when C' = {∧, \, 0}; should read:
  PQ<sub>C'</sub> denote the variety of upper pre-BCK-bands when C' = {∨, \, 0};
- **p. 308** The statement of Corollary 3.3.19 should read: Corollary 3.3.19. Let  $\land \in C'$ . For any  $A \in |Q_{C'}|$ , the principal subalgebra  $\{a\}_A$  generated by  $a \in A$  is a Boolean lattice....
- **p. 309** The statement of Corollary 3.3.20 should read: Corollary 3.3.20. Let  $\{\land,\lor\} \subseteq C'$  and let  $A \in IQ_{C'}$ ....
- p. 319 The first sentence of the statement of Proposition 3.3.35 should read: Corollary 3.3.35. For any  $A \in BP_c$  the following assertions hold: ...
- p. 320 The proof of Corollary 3.3.36 should be deleted. Since  $\mathcal{L} \vee \mathcal{R} = \mathcal{D}$  holds for either  $\wedge$  or  $\vee$ , Corollary 3.3.36 is an immediate consequence of Corollary 3.3.35, and as such does not require any proof.
- p. 328 The statement of Theorem 3.3.49 should read: Theorem 3.3.49. An algebra  $\mathbf{A} := \langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  is an implicative  $BCK \leq_0 - BCK$  local paralattice iff the reduct  $\langle A; \land, \lor, 0 \rangle$  is a local paralattice with zero, ...
- p. 375 Reference [6] should read:
  - [6] P. Agliano, Congruence quasi-orederability in subtractive varieties, J. Austral. Math. Soc. 71 (2001), 421-445.
- p. 375 Reference [7] should read:
  - [7] P. Aglianò, Fregean subtractive varieties with definable congruences, J. Austral. Math. Soc. 71 (2001), 353-366.
  - 5. Australi. Math. 500. 11 (2001), 555-5
- p. 387 Reference [145] should read:
  - [145] G. Laslo and J. Leech, Green's equivalences on noncommutative lattices, Acta Sci. Math. (Szeged) 68 (2002), 501-533.

2

# CONTRIBUTIONS TO THE THEORY OF PRE-BCK-ALGEBRAS

A thesis submitted for the degree of Doctor of Philosophy

by

Matthew Spinks B.App.Sc.(Monash) B.Sc.(Hons)(ANU)

Gippsland School of Computing and Information Technology Faculty of Information Technology Monash University

March 2002

For the Bee

1988-1997

In loving memory

For Martin Seymour-Smith

1928-1998

In memoriam

# Contents

|   | Abst | ract  | iv  |
|---|------|---|-----|
|   | Decl | aration   | ix  |
|   | Ack  | nowledgements                                       | x   |
| 1 | Intr | oduction  | 1   |
|   | 1.1  | Introduction  | 1   |
|   | 1.2  | Notation and Terminology                            | 8   |
|   | 1.3  | Bands, Monoids and Semilattices                     | 18  |
|   | 1.4  | Skew Boolean Algebras and Discriminator Varieties   | 27  |
|   | 1.5  | Varieties with EDPC                                 | 49  |
|   | 1.6  | BCK-Algebras  | 58  |
|   | 1.0  | Ideals and Subtractive Varieties                    | 74  |
|   | 1.8  | Algebraisable and Assertional Logics                | 85  |
| 2 | The  | Theory of Pre-BCK-Algebras                          | 94  |
|   | 2.1  | Pre-BCK-Algebras                                    | 95  |
|   | 2.2  | Varieties of Pre-BCK-Algebras                       | 123 |
|   | 2.3  | Implicative BCS-Algebras                            | 151 |
| 3 | Ap   | plications to Universal Algebra and Algebraic Logic | 214 |
|   | 3.1  | Subtractive Varieties with EDPI                     | 215 |
|   | 3.2  | Binary and Dual Binary Discriminator Varieties      | 267 |
|   | 3.3  | Pre-BCK Quasilattices and BCK Paralattices          | 291 |

| 4 | Con | clusion     |       |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   | 361 |
|---|-----|-------------|-------|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|-----|
|   | 4.1 | Summary .   | <br>• | • | • | • |   | • | • | • | • |   | • | • | • | • |   | • |   |   | • | • | • | • | • |   |   | • | • | 361 |
|   | 4.2 | Future Work | <br>• | • | • | • | • | • | • | ۰ | ٠ | • | • | ٠ | • | • | • | • | • | • | • | • | • | ٠ | • | • | • | • | • | 363 |
|   | Ref | erences     |       |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   |   | 375 |

## Abstract

The class of BCK-algebras (hereafter BCK) is a relatively point-regular quasivariety that arises naturally both in algebraic logic and universal algebra. In algebraic logic, BCK arises as the equivalent algebraic semantics of Meredith's BCK logic, an important substructural logic with applications to proof theory. In universal algebra, BCK arises as the class of all residuation subreducts of partially ordered commutative residuated integral monoids (briefly, pocrims); the class of pocrims is a quasivariety whose relative subvarieties include the varieties of hoops and dual Brouwerian semilattices. The class of pre-BCKalgebras (hereafter PBCK) is a subtractive but not point-regular variety, the members of which naturally generalise BCK-algebras. The theory of pre-BCKalgebras and the applications of this theory to universal algebra and algebraic logic are the subject of this thesis.

Chapter 1 provides a structured account of the theory relevant to the study of pre-BCK-algebras, including: Laslo and Leech's theory of quasilattices, paralattices and skew lattices; Blok and Pigozzi's hierarchy of varieties with Equationally Definable Principal Congruences (briefly, EDPC); the theory of BCKalgebras and BCK-lattices due to Iséki, Idziak and others; Agliano and Ursini's theory of ideals and subtractive varieties; and the theory of algebraisable and assertional logics due to Blok, Pigozzi, Raftery and others. The main new results concern distributivity in skew lattices. A counterexample is presented showing that the middle distributive identities for skew lattices are independent; and a theorem asserting the interderivability of the middle distributive identities for symmetric skew lattices is stated. The results obtained answer two questions of Leech.

Chapter 2 is devoted to a study of the theory of pre-BCK-algebras. The elementary theory of pre-BCK-algebras is considered in Section 1 of Chapter 2. Some results relating PBCK to existing classes of algebras generalising BCK to the subtractive but not (relatively) point-regular case are presented. In one of two key results of the section, a 'Clifford-McLean'-type theorem for

pre-BCK-algebras, it is shown that the equivalence  $\Xi$  induced by the natural quasiordering  $\preceq$  on a pre-BCK-algebra **A** is a congruence on **A** such that the quotient algebra  $\mathbf{A}/\Xi$  is the maximal BCK-algebra homomorphic image of **A**. For an appropriate notion of ideal, the ideal theory of pre-BCK-algebras is investigated. The relationship between ideals and congruences on pre-BCK-algebras is also briefly explored. In the other major result of the section, it is proved that the assertional logics of the variety of pre-BCK-algebras and the quasivariety of BCK-algebras coincide, and hence that a quasi-identity of the form  $\bigotimes_{i\leq n} t_i(\vec{x}) \approx \mathbf{0} \supset t(\vec{x}) \approx \mathbf{0}$  is satisfied by PBCK iff it satisfied by BCK. Collectively, the results indicate that much of the first-order theory of BCK-algebras.

Varieties of pre-BCK-algebras are investigated in Section 2 of Chapt 2. For a variety V of BCK-algebras, the natural pre-BCK-algebraic counterpart  $V_{a}$ of V is the class  $\{A \in PBCK : A/\Xi \cong B \text{ for some } B \in V\}$ . In the main result of the section, it is shown that the natural pre-BCK-algebraic counterpart of any variety of BCK-algebras is itself a variety. The varieties of commutative, positive implicative and implicative BCK-algebras are important classes of BCK-algebras; their natural pre-BCK-algebraic counterparts are the varieties of commutative, positive implicative and implicative pre-BCK-algebras respectively. An order-theoretic characterisation of commutative pre-BCK-algebras is provided, and an ideal-theoretic characterisation of positive implicative pre-BCK-algebras is presented. For a suitable notion of prime ideal, it is also shown that an ideal of an implicative pre-BCK-algebra is prime iff it is maximal iff it is irreducible. The results obtained suggest that, for any variety V of BCK-algebras, the first-order theory of V, stands in relation to V as the firstorder theory of pre-BCK-algebras stands in relation to the first-order theory of BCK-algebras.

The variety of implicative BCS-algebras, a class of pointed groupoids, is studied in Section 3 of Chapter 2. It is shown that the variety of implicative BCSalgebras is a subvariety of the variety of implicative pre-BCK-algebras. Some examples are presented showing that implicative BCS-algebras arise naturally in several contexts in universal algebra and algebraic logic, including binary discriminator varieties (in particular, pseudocomplemented semilattices) and fixedpoint discriminator varieties (in particular, certain varieties of n-potent BCK-algebras). It is shown that an implicative pre-BCK-algebra is an implicative BCS-algebra iff it has a certain left normal band with zero polynomial reduct whose underlying partial ordering respects implicative pre-BCK difference in a precise sense. A representation theorem is proved showing that the category of implicative BCS-algebras is isomorphic to the category of left handed locally Boolean bands for suitable choices of objects and morphisms. The subdirectly irreducible implicative BCS-algebras are characterised (with R. J. Bignall): they are the 2-element implicative BCK-algebra and the algebras  $\hat{\mathbf{B}}$  obtained from the non-trivial Boolean algebras  $\mathbf{B}$  upon replacing the unit element of each B with a two element clique. It is shown that the class of implicative BCK-algebras is generated (as a variety) by a certain 3-element pre-BCK-algebra  $\mathbf{B}_2$ , and hence that the lattice of varieties of implicative BCSalgebras is a three-element chain; the only non-trivial subvariety of the variety of implicative BCS-algebras is the variety of implicative BCK-algebras. Collectively, the results attest that implicative BCS-algebras are a 'non-commutative' analogue of implicative BCK-algebras, and as such, more closely resemble implicative BCK-algebras than do implicative pre-BCK-algebras.

In Chapter 3 the theory of pre-BCK-algebras is applied to the study of certain classes of algebras arising naturally in universal algebra and algebraic logic. In Section 1 of Chapter 3 subtractive varieties with Equationally Definable Principal Ideals (briefly, EDPI) are considered. For subtractive varieties, equational definability of principal ideals is the ideal-theoretic analogue of equationally definable principal congruences. It is shown that the variety of positive implicative pre-BCK-algebras is termwise definitionally equivalent to the variety of MINI-algebras introduced recently by Agliano and Ursini. A result is proved showing that a variety V is subtractive with EDPI iff every  $A \in V$  has a MINI-algebra polynomial reduct whose ideals coincide with those of A. A structure theorem for MINI-algebras is also proved: for a suitable notion of weakly compatible operation, it is shown that a variety is termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations iff it is subtractive, weakly congruence orderable with EDPI. Subtractive weak Brouwerian algebras with filter-preserving operations (briefly, subtractive WBSO varieties) are an important class of subtractive varieties with EDPI that arise in the first instance from algebraic logic. A natural example of a subtractive WBSO variety is the variety of Nelson algebras, which arises from the algebraisation of constructive logic with strong negation. It is shown that the variety of Nelson algebras has a commutative (but not regular) Ternary Deductive term and is congruence permutable. An explicit Quaternary Deductive term is also given. The results obtained answer a question of Blok and Pigozzi.

Binary discriminator and dual binary discriminator varieties are studied in Section 2 of Chapter 3. The binary discriminator and dual binary discriminator were recently introduced by Chajda, Halaš and Rosenberg in an attempt to generalise the ternary discriminator and dual ternary discriminator to varieties with 0 exhibiting congruence permutability and congruence distributivity only locally at 0 respectively. It is shown that the variety generated by the class of all algebras  $\langle A; h, 0 \rangle$ , where h is the dual binary discriminator on A and 0 is a nullary operation, is precisely the variety of left normal bands with zero. A semigroup-theoretic characterisation of dual binary discriminator varieties is also provided. It is shown that the variety generated by the class of all algebras  $\langle A; b, 0 \rangle$ , where b is the binary discriminator on A and 0 is a nullary operation, is exactly the variety of implicative BCS-algebras. A natural characterisation of binary discriminator varieties is presented: a pointed variety is a binary discriminator variety iff it is subtractive with EDPI and is generated by a class of ideal simple algebras. Point-regular binary discriminator varieties are an important subclass of binary discriminator varieties; two results are proved that together show a point-regular variety is a binary discriminator variety iff it is a 'pointed' fixedpoint discriminator variety. In the major result of the section, the 'pointed' fixed point discriminator varieties are characterised: they are precisely the varieties that are ideal determined, semisimple with EDPC. Some theorems connecting 'pointed' fixedpoint discriminator varieties with , pointed ternary discriminator varieties are also presented. The results answer in part a question of Blok and Pigozzi.

Pre-BCK-algebras structurally enriched with band or skew lattice operations are studied in Section 3 of Chapter 3. Bands and skew lattices structurally enriched with difference operations arise naturally in pointed discriminator varieties as skew Boolean algebras and skew Boolean intersection algebras (briefly, skew Boolean  $\cap$ -algebras). Skew Boolean algebras arise as distributive skew lattices structurally enriched with a relative complementation operation; skew Boolean  $\cap$ -algebras are skew Boolean algebras for which finite meets exist with respect to the natural skew lattice partial order. A theory of pre-BCK bands and pre-BCK quasilattices that parallels Laslo and Leech's theory of quasilattices is briefly outlined. In one of the two main results of the section, the skew Boolean algebras are characterised among the pre-BCK quasilattices: they are precisely the pre-BCK quasilattices for which the quasilattice with zero reduct is a join symmetric skew lattice with zero and the pre-BCK-algebra reduct is an implicative BCS-algebra. A theory of  $\leq_0$ -BCK bands and  $\leq_0$ -BCK paralattices that parallels Idziak's theory of BCK-semilattices and BCKlattices is briefly outlined. In the other main result of the section, the skew  $\exists$ oolean  $\cap$ -algebras are characterised among the  $\leq_0$ -BCK paralattices: they are precisely the  $\leq_{0}$ -BCK paralattices for which the paralattice with zero reduct is a join symmetric local skew lattice with zero and the BCK-algebra reduct is an implicative BCK-algebra. A theory of double-pointed skew Boolean  $\cap$ algebras akin to that of skew Boolean  $\cap$ -algebras is also presented, together with an axiomatisation of the assertional logic of the variety of (left handed) double-pointed skew Boolean  $\cap$ -algebras. Collectively, the results obtained intimate that pre-BCK algebras structurally enriched with band or skew lattice operations may provide a unifying framework for the study of several classes of 'generalised Boolean structures' arising naturally in universal algebra and algebraic logic.

In Chapter 4 the research undertaken in this dissertation is briefly reviewed, and some potential avenues for future research are presented.

## Declaration

This thesis contains no material accepted for the award of any other degree or diploma at any university or other institution. To the best of my knowledge this thesis contains no material written or previously published by another person except where due reference is made within the text.

Matthew Spinks

## Acknowledgements

First and foremost among those I would like to thank is my supervisor, Professor Robert Bignall. The amount of time and energy he has invested over the past seven years to see this thesis through to completion has been quite unprecedented. He has been, at all times over the long period of my Ph.D. candidature, the best possible supervisor I could have envisaged: his continual support and guidance, enthusiasm and seemingly endless patience have been the mainstay of my Ph.D. candidature. Above all, his tremendous ability to do mathematics and the extraordinary depth and elegance of his mathematical ideas and insights have been absolutely formative in the development of my mathematical and intellectual life as a graduate student. I consider myself very fortunate to have worked under a mathematician of his calibre.

During my time at Monash University, I was in receipt of a Faculty Scholarship from the Gippsland School of Computing and Information Technology; the financial support so provided is gratefully acknowledged. Professor John Crossley, Head of the Department of Computer Science in the Faculty of Information Technology, was kind enough to act as associate supervisor throughout my candidature, in spite of his myriad other duties. Lloyd Humberstone, of the Department of Philosophy in the Faculty of Arts, generously shared his encyclopaedic knowledge of logic over the course of several years, while Dr. John Jeavons, of the Department of Mathematics in the Faculty of Science, assisted with general mathematical issues over the same period.

I owe a huge debt to the academic community at large. Much of the initial research for this thesis was conducted using computer-based tools, most notably Dr. Jiang Zhang and Dr. Hantao Zhang's finite model enumerator SEM and Dr. William McCune's automated theorem-prover OTTER. In particular, the proof of Proposition 2.3.21 is a manual simplification of an automated proof initially obtained using OTTER; it seems unlikely this proof would have been obtained without the use of automated theorem-proving software. For many helpful discussions about automated theorem-proving, universal algebra and algebraic logic I am indebted to Prof. George Grätzer, Prof. Kiyoshi Iséki, Dr. William McCune, Prof. Don Pigozzi, Prof. James Raftery, Dr. Greg Restall and Prof. Aldo Ursini. I would like in particular to thank Dr. McCune: he always answered my innumerable questions about automated theorem-proving, no matter how trivial, with unfailing courtesy. For the provision of papers, preprints, and other manuscripts I am grateful to Dr. Paolo Agliano, Dr. Clint van Alten, Prof. Martin Bunder, Dr. Isabel Ferreirim, Lloyd Humberstone, Dr. Tomasz Kowalski, Prof. Raftery, Prof. Ursini and Prof. Andrzej Wroński. I would especially like to thank Dr. van Alten, who kindly made his doctoral dissertation available for study at a critical juncture in my PhD candidature; the influence of his work on this thesis is quite obvious.

My sincere thanks go also to various of my family and friends. My parents and my brother Jeremy have always afforded me their care and support, never more so than during the difficult last fifteen months of my candidature. Both Konrad Talmont-Kaminski and Anthony Romeo have remained steadfast in their friendship over two decades; here I would like to acknowledge the hospitality shown me by the former during my stay in Poland in 1999, and the emergency accommodation provided me by the latter over the summer of 1999-2000. Over the past seven years the Rev. Judy Redman, her husband Bruce and children Hugh and Ceiridwen have always been ready to furnish me with a home away from home; their generosity of spirit in so doing is gratefully acknowledged.

Others who should specially be mentioned include Rhondda Curtis, who provided much-needed counsel at various points throughout my Ph.D. candidature; to her I am deeply indebted. During my recent serious illness I was attended by Dr. David Harding-Smith, who treated me with great competency and professionalism; it is not really an exaggeration to say I owe my life to his care. Finally, it would be an omission not to acknowledge the almost subliminal influence of Larry Prosser: in effect, this dissertation witnesses the culmination of advice given me by him some twenty years ago.

Carey St 25 March 2002

### Chapter 1

### Introduction

### **1.1** Introduction

About This Thesis. Beginning with the papers of Krull [143] and 1.1.1. Ward and Dilworth [235] residuated algebraic structures have been the subject of study for over half a century. Nonetheless, it is only recently that the residuated algebraic structures associated with logical systems have been seriously investigated [34, p. 597], even though residuation in such structures is typically the algebraic counterpart of implication in the associated logic. Much of the recent work concerning residuated algebraic structures in logic has focussed on two related classes of algebras: the class of partially ordered commutative residuated integral monoids (briefly, pocrims) [109, 176, 39]; and the class of BCK-algebras [126, 70, 38]. Let  $\langle A; \leq \rangle$  be a poset such that: (i)  $\langle A; \leq \rangle$  is integral in the sense that there exists a least element  $0 \in A$ which acts as an identity element for an order compatible commutative associative binary multiplication  $\oplus$  on A; and (ii)  $\langle A; \leq \rangle$  is residuated in the sense that there exists a binary operation  $\div$  on A such that, for any  $a, b \in A$ ,  $a \doteq b = \min\{c \in A' : a \le c \oplus b\}$ . Then  $\langle A; \le \rangle$  is first-order definitionally equivalent to an algebra  $\langle A; \oplus, -, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$ ; such an algebra is a *pocrim*. The class of all pocrims is a quasivariety [121, 123] but is not a variety [109]. A BCK-algebra is a  $\langle -, 0 \rangle$ -subreduct of a pocrim; equivalently, by results of Wroński [242], Ono and Komori [176], Fleischer [90] and Pałasiński [178], an algebra  $\langle A; -, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a BCK-algebra iff the

following identities and quasi-identity are satisfied:

$$\left(\left(x - y\right) - \left(x - z\right)\right) - \left(z - y\right) \approx 0 \tag{1.1}$$

$$\left(x - (x - y)\right) - y \approx \mathbf{0} \tag{1.2}$$

$$x - x \approx 0 \tag{1.3}$$

$$\mathbf{0} - \mathbf{x} \approx \mathbf{0} \tag{1.4}$$

$$x - y \approx 0 \& y - x \approx 0 \supset x \approx y.$$
(1.5)

The class of all BCK-algebras is thus a quasivariety; it is not a variety [240]. By results of Blok and Pigozzi [31], Blok and Jonsson [28] and Raftery and van Alten [192] the class of BCK-algebras [pocrims] is termwise definitionally equivalent to the equivalent algebraic semantics of Meredith's BCK logic [165], [186, p. 316] [BCK logic with 'fusion' [28]]. BCK logic [with fusion] is an important substructural logic which arises in the first instance from proof theory; see [80, Section 4] for a discussion and references.

A significant body of work now exists showing pocrims and BCK-algebras play a central role in the theory of the residuated structures associated with logical systems (for an extended discussion and references see Blok and Pigozzi [34, Section 6]). This is most readily seen from the theory of hoops, which are pocrims that are 'naturally ordered' in the sense that for any  $a, b \in A, a \leq b$ implies there exists  $c \in A$  such that  $a \oplus c = b$ . The study of hoops and their residuation subreducts is due variously to Büchi and Owens [48], Ferreirim [88], Blok and Pigozzi [34] and Blok and Ferreirim [26, 27] (see also Bosbach [44, 45, 46]), while the study of hoops with 'normal multiplicative operators' is due to Blok and Pigozzi [34]. 'Hoop logics' are deductive systems whose equivalent algebraic semantics are termwise definitionally equivalent to quasivarieties of (structurally enriched) hoops. Included among the hoop logics are the classical and intuitionistic propositional logic; the  $\omega$ -valued propositional calculus of Lukasiewicz [155]; and the logics  $L_{BCK}$  and  $LJ^*$  of Ono

and Komori [176]. Respectively, the equivalent algebraic semantics of these deductive systems are termwise definitionally equivalent to: the varieties of Boolean and Heyting algebras; the varieties of Boolean and Heyting algebras with 'normal multiplicative operators'; the variety of Wajsberg algebras (see Font, Rodriguez and Torrens [92], Chang [59] and Mundici [171]); and certain varieties of residuated lattices (see Blount [43]). See Agliano [5, Section 4.6].

Although the theory of hoops provides a unifying framework for many of the logical systems and associated classes of algebras traditionally considered the domain of algebraic logic, there exist important examples of algebraisable deductive systems whose equivalent algebraic semantics are classes of residuated algebraic structures that are not termwise definitionally equivalent to quasivarieties of (structurally enriched) pocrims or BCK-algebras. For example, linear logic [97] is algebraisable [3, Section 2.2]; its equivalent algebraic semantics is termwise definitionally equivalent to the variety of girales, a class of semilattice ordered residuated monoids that is not a variety of pocrims [5, Section 4.1]. Relevance logic [12] is algebraisable [93]; its equivalent algebraic semantics is termwise definitionally equivalent to the variety of De Morgan monoids, also a class of semilattice ordered residuated monoids that is not a variety of pocrims [5, Section 4.4]. Residuated structures that are not pocrims also arise naturally in universal algebra. Ordinals closed under addition provide natural examples of residuated ordered monoids for which the monoid operation is not commutative [229, Example 1.8]. Ideal lattices of rings (considered with ideal multiplication and set inclusion) also form residuated ordered monoids with a non-commutative monoid operation [229, Example 1.7]; notably, lattices of topologising filters on rings with identity [98] may be understood as such monoids [229, Appendix]. Besides being of independent interest in their own right, classes of residuated algebraic structures found in universal algebra are often a f intrinsic interest from the perspective of algebraic logic in the sense that they may arise naturally as 'quasivarieties of logic': that is, they are termwise definitionally equivalent to the equivalent algebraic semantics of some algebraisable deductive system. See Blok and Raftery [40, Section 5] and Barbour and Raftery [16, Section 6].

Recent developments in algebraic logic and universal algebra have thus lead to the study of several classes of algebras generalising pocrims and BCK-algebras. The most prominent of these generalisations is that of pocrims to polrims and BCK-algebras to BCC- or left residuation algebras. A polrim is a partially ordered integral monoid that is residuated on the left; the residuation subreducts of polrims are the BCC- or left residuation algebras. The classes of polrims and left residuation algebras are both quasivarieties [229, Proposition 1.4] that are not varieties [229, Proposition 4.1]. Commutative polrims are precisely the pocrims, while left residuation algebras satisfying a certain principle of 'quasicommutation' are exactly the BCK-algebras [229, Example 1.5]. Polrims and their residuation subreducts have been investigated by several authors, including Komori [138, 139], Ono and Komori [176], Raftery and van Alten [192], van Alten [229], and van Alten and Raftery [231, 230]. Polrims arise naturally both in universal algebra and algebraic logic; in particular, the quasivariety of polrims [left residuation algebras] is termwise definitionally equivalent to the equivalent algebraic semantics of the  $\{\&, \supset\}$ -fragment  $\{\{\supset\}\}$ -fragment of the logic  $\mathbb{H}_{BCC}$  of Ono and Komori [176]. Other classes of residuated monoids generalising pocrims and BCK-algebras to have been considered in the literature include the sircomonoids of Raftery and van Alten [193] and their residuation subreducts, the BCI-algebras of Iséki [122]; and the semilattice ordered residuated monoids of Agliano [5]. The study of all such monoids is part of a much larger theory of 'residuals without residuation' pioneered by several authors including Meyer and Routley [166], Dunn [85, 86], Ursini [223] and Agliano [5]. Of particular relevance to algebraic logic is Dunn's theory of 'partial gaggles' [86], which seeks to provide a uniform semantical approach to the study of substructural propositional logics, including: classical and intuitionistic logic; the various modal and relevance logics; linear logic; BCK logic; and the Lambek calculus [144]. See also Restall [198, Chapter II].

One principle common to pocrims, BCK-algebras and most of their generalisations is the existence of a finite set of binary terms  $\{d_i(x, y) : i = 1, ..., n\}$  and a (definable) constant **0** such that the following identities and quasi-identity

are satisfied:

$$d_i(x,x) \approx 0$$
 and  $\bigotimes_{i \leq n} d_i(x,y) \approx 0 \supset x \approx y.$  (1.6)

For example, the set of terms  $\{x \div y, y \div x\}$  witnesses (1.6) for the quasivariety of BCK-algebras (and thus for pocrims); this follows immediately from (1.3) and (1.5). In algebraic terms, satisfaction of the identities and quasi-identity of (1.6) by a quasivariety K (with a (definable) constant term) is equivalent to relative point (or 0, if the constant term is specified as 0) regularity. Regularity conditions in universal algebra always demand congruences of algebras be determined by certain subsets of their universes; in particular, relative 0regularity asserts that the K-congruences of any  $A \in K$  are determined by their O<sup>A</sup>-classes. While regularity conditions in universal algebra are well understood (see for instance [74, 89, 104, 78]), the metalogical significance of such conditions has been less clear. Recently the status of these conditions in algebraic logic has been clarified by Blok and Raftery [40], who have shown that a quasivariety K is a 'quasivariety of logic' precisely when the K-congruences of members of K are determined by suitably defined subsets of their universes. In logical terms, therefore, relative point regularity is a sufficient condition for a quasivariety to be a 'quasivariety of logic'; further, although not necessary, the condition of relative point regularity is satisfied by most familiar classes of algebras arising as the equivalent algebraic semantics of some algebraisable deductive system. Blok and Raftery's result has very recently engendered some interest in regularity conditions for algebraic logic, both syntactically (Blok and La Falce [25]) and from the perspective of full regularity (Barbour and Raftery [16]).

Perhaps because of the centrality of regularity conditions to algebraic logic, classes of algebras that are not relatively point regular but which naturally generalise pocrims, BCK-algebras or related structures in some sense have not been previously investigated in algebraic logic to any significant degree, bar two exceptions. The first of these exceptions is a certain variety V of algebras considered by Blok and Raftery in [38, Section 4] and by Agliano and Ursini in [10] (to within termwise definitional equivalence); members of V

(V-algebras) naturally generalise BCK-algebras but are not in general point regular. The role played by V-algebras in the theory of BCK-algebras has been briefly considered by Blok and Raftery in [38, Section 4]. In particular, Blok and Raftery have shown [38, Proposition 2, Theorem 8] that a certain 3-element algebra  $\mathbf{B}_2 \in \mathsf{V}$ , though not itself a BCK-algebra, plays an important role in the theory of BCK-algebras. The second exception is the variety of MINIalgebras considered by Agliano and Ursini in [11]. MINI-algebras naturally generalise positive implicative BCK-algebras (the  $\langle \div, 0 \rangle$ -subreducts of dual Brouwerian semilattices, or, equivalently, hoops satisfying  $x \oplus x \approx x$ ) but are not in general point regular. Results due to Agliano and Ursini show MINIalgebras play a fundamental role in the theory of subtractive varieties with equationally definable principal ideals: see in particular [11, Corollary 3.8]. Collectively, the work of Blok and Raftery and Agliano and Ursini suggests that an appropriate generalisation of BCK-algebras that subsumes both Blok and Raftery's variety V and Agliano and Ursini's variety of MINI-algebras (up to termwise definitional equivalence) may be of interest in algebraic logic.

Call an algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  a *pre-BCK-algebra* iff it satisfies the identities (1.1)-(1.4) and the identity:

$$x - 0 \approx x. \tag{1.7}$$

The identity (1.7) in conjunction with (1.3) ensures that the variety of pre-BCK-algebras is *subtractive* in the sense of Agliano and Ursini [222, 10, 9, 11, 225] and hence that the variety of pre-BCK-algebras contains both Blok and Raftery's variety V and Agliano and Ursini's variety of MINI-algebras (up to terr wise definitional equivalence). Since pre-BCK-algebras do not in general satisfy the quasi-identity (1.5), binary terms  $d_i(x, y)$  satisfying (1.6) need not exist, whence the variety of pre-BCK-algebras is not 0-regular. On the other hand, by a result of Iséki [126, Theorem 2] the class of all BCK-algebras satisfies (1.7), and so is a subquasivariety of the variety of pre-BCK-algebras. Thus the variety of pre-BCK-algebras may be understood as a generalisation of BCK-algebras to the non-relatively point regular case that subsumes both the variety V of Blok and Raftery and (to within termwise definitional equivalence)

6

the variety of MINI-algebras of Agliano and Ursini. In this thesis we offer and explore the theory of pre-BCK-algebras as a generalisation of the theory of BCK-algebras to the non-relatively point regular case. In particular, our program is to investigate the elementary theory of the variety of pre-BCKalgebras and some of its subvarieties, and to apply this theory to the study of some varieties arising naturally in universal algebra and algebraic logic.

Remark 1.1.2. Two fundamental restrictions are imposed on the scope of the work presented in this thesis. First, our study of pre-BCK-algebras does not extend to a study of quasivarieties of pre-BCK-algebras (with the obvious exception of the class of BCK-algebras), despite the fact that quasivarieties are the natural algebraic counterparts of algebraisable logics in the sense that the equivalent algebraic semantics of an algebraisable logic can always be taken to be a quasivariety. The rationale behind this (admittedly artificial) restriction is that the theory of subtractive varieties, which plays an important role in our investigation of the variety of pre-BCK-algebras and its subvarieties, does not extend well to quasivarieties: see for instance Blok and Raftery [40, p. 181, Example 7.2] or van Alten [229, pp. 71-72]. Second, our study of pre-BCK-algebras does not extend to a study of algebras naturally generalising pocrims in some sense but which fail to be relatively point-regular. While the role played by residuated ordered monoids in universal algebra and algebraic logic is an important motivation for the work undertaken in this thesis, it is not yet clear what form such a generalisation of these monoids should take, or even if such a generalisation of these monoids is of any intrinsic interest: cf. Higgs [109, p. 72] and the remarks of  $\S4.2.1$ .

1.1.3. Organisation. This dissertation is organised as follows. In an attempt to keep this thesis self-contained, in the remainder of this chapter we introduce some notation and terminology and review those parts of universal algebra and algebraic logic pertinent to the study of pre-BCK-algebras, including: the theory of bands; Leech's theory of non-commutative and skew lattices, the theory of varieties with equationally definable principal congruences due to Blok, Köhler and Pigozzi; the theory of BCK-algebras and BCK-lattices due to Iséki, Tanaka, Idziak and others; Agliano and Ursini's theory of ideals and

#### **1.2.** Notation and Terminology

subtractive varieties in universal algebra; and the theory of assertional and algebraisable logics due to Blok, Pigozzi and Raftery. Our survey is leisurely: we take the opportunity to present the occasional new result and example, and generally to tidy up some loose ends. Chapter 2 is devoted to the study proper of the theory of pre-BCK-algebras. Adopting the approach of Iséki and Tanaka's survey paper of BCK-algebras [126], we show the elementary theory of pre-BCK-algebras closely parallels that of BCK-algebras. In Chapter 3 we apply the theory of pre-BCK-algebras to three important classes of algebras arising naturally in algebraic logic and universal algebra, namely: subtractive varieties with equationally definable principal ideals; binary discriminator varieties; and pointed ternary discriminator varieties. In Chapter 4 we briefly review the work undertaken in this thesis and make some suggestions for future work.

### **1.2** Notation and Terminology

In this section we fix some of the fundamental notation used throughout this thesis and introduce some terminology of universal algebra and algebraic logic. For notation and terminology not explicitly introduced either here or in the sequel we generally follow Burris and Sankappanavar [55] or Blok and Pigozzi [36].

**1.2.1.** Ordered Sets. A quasiorder (also preorder in the literature) is a reflexive transitive relation. A quasiordered set is a pair  $\langle A; \leq \rangle$  where A is a set and  $\leq$  is a quasiorder. Let  $\langle A; \leq \rangle$  be a quasiordered set. For  $B \subseteq A$  and  $a \in A$  we define:

| $(B] := \{b \in A : b \preceq a \text{ for some } a \in B\},\$  | $(a] := \big(\{a\}\big]$ |
|---|--------------------------|
| $[B) := \{ b \in A : a \preceq b \text{ for some } a \in B \},$ | $[a) := [\{a\}).$        |

The set (a] is called the principal (order) ideal generated by a in  $\langle A; \leq \rangle$ . A subset B of A is said to be hereditary in  $\langle A; \leq \rangle$  if B = (B]. An element  $m \in A$  is minimal [maximal] if  $m \leq a$  [ $a \leq m$ ] for any  $a \in A$ . Observe that minimal and maximal elements of a quasiordered set need not be unique in

#### **1.2.** Notation and Terminology

general. A minimal element [maximal element] of a quasiordered set  $\langle A; \leq \rangle$ is called a *least element* [greatest element] if it is unique. Let  $B \subseteq A$ . An element  $c \in A$  is a *lower bound* [upper bound] of B if  $c \leq b$  [ $b \leq c$ ] for all  $b \in B$ . An element  $d \in A$  is a greatest lower bound [least upper bound] of B if d is a lower bound [upper bound] of B and  $c \leq d$  [ $d \leq c$ ] for any lower bound [upper bound] c of B. Observe that greatest lower and least upper bounds of B need not be unique in general. The set of greatest lower [least upper] bounds of B is denoted glb B [lub B]. For  $B := \{a_1, \ldots, a_n\}$  the set glb B [lub B] is alternatively denoted glb  $\{a_1, \ldots, a_n\}$  [lub  $\{a_1, \ldots, a_n\}$ ]. Given elements a, b of a quasiordered set  $\langle A; \leq \rangle$  with  $a \leq b$ , the interval [a, b] is the set  $\{c \in A : a \leq c \leq b\}$ .

9

A partial order is a reflexive, symmetric and transitive relation. A partially ordered set is a pair  $\langle A; \leq \rangle$ , where A is a set and  $\leq$  is a partial order. We abbreviate the term 'partially ordered set' by *poset*. The following lemma is folklore.

**Lemma 1.2.2.** [196, Theorem I§5.2] Let  $\langle A; \leq \rangle$  be a quasiordered set and let  $\Xi$  be the binary relation defined by:

 $a \equiv b$  iff  $a \preceq b$  and  $b \preceq a$ 

for any  $a, b \in A$ . Then  $\Xi$  is an equivalence relation on A. Moreover, the binary relation  $\leq$  defined on  $A/\Xi$  by:

 $[a]_{\Xi} \leq [b]_{\Xi} \quad iff \quad a \preceq b$ 

for any  $[a]_{\Xi}, [b]_{\Xi} \in A/\Xi$  with  $a, b \in A$  is a partial ordering on  $A/\Xi$ .

**Lemma 1.2.3.** Let  $\langle A; \preceq \rangle$  be a quasiordered set and let  $\Xi$  be the equivalence relation on A induced by  $\preceq$  in the sense of Lemma 1.2.2. The following statements hold for any  $a, b \in A$ :

- 1. If  $d_1 \in A$  is a greatest lower bound of  $\{a, b\}$  and  $d_2 \equiv d_1$  for  $d_2 \in A$ , then  $d_2$  is also a greatest lower bound of  $\{a, b\}$ ;
- 2. If  $d_1, d_2 \in A$  are both greatest lower bounds of  $\{a, b\}$ , then  $d_1 \equiv d_2$ .

Proof. For (1), let  $d_1 \in A$  be a greatest lower bound of  $\{a, b\}$  and suppose  $d_2 \equiv d_1$  for  $d_2 \in A$ . Since  $d_2 \preceq d_1$ ,  $d_2$  is a lower bound of  $\{a, b\}$ . If  $c \in A$  is a lower bound of  $\{a, b\}$ , then  $c \preceq d_1 \preceq d_2$  since  $d_1$  is a greatest lower bound of  $\{a, b\}$ . Thus  $d_2$  is also a greatest lower bound of  $\{a, b\}$ . For (2), let  $d_1, d_2 \in A$  be greatest lower bounds of  $\{a, b\}$ . Since  $d_1$  is a lower bound of  $\{a, b\}$  and  $d_2$  is a greatest lower bound of  $\{a, b\}$ . We have  $d_1 \preceq d_2$ ; likewise we have  $d_2 \preceq d_1$ . Thus  $d_1 \equiv d_2$  as required.

Let  $\langle A; \leq \rangle$  be a quasiordered set. The proper part of  $\leq$ , denoted  $\prec$ , is defined as  $\leq$  but not  $\Xi$ . A subset *B* of *A* is called a *clique* of  $\Xi$  if  $B^2 \subseteq \Xi$ . Let  $\langle A; \leq \rangle$  be a poset. A quasiorder  $\leq$  on *A* is said to be *admissible* if, for all  $a, b \in A, a \leq b$  implies  $a \leq b$ . Further details concerning quasiorderings and partial orderings may be found in Wechler [236, pp. 31-35] and Cleave [51, Chapter 5§5-6].

**1.2.4.** Languages.<sup>1</sup> We fix a countably infinite set X of *variables* for use throughout this thesis. In an algebraic [logical] context, we usually write  $x, y, z, \dots$   $[p, q, r, \dots]$  for metavariables ranging over X. In the sequel we confine our attention to algebraic languages unless otherwise stated. Thus a language  $\mathcal{L}$  consists of a set  $\mathcal{L}$  of function symbols together with an ar*ity* function ar that assigns a natural number to each function symbol in  $\mathcal{L}$ . By abuse of notation we often identify a language  $\mathcal L$  with its set of function symbols  $\mathcal{L}$ , while by abuse of language we sometimes describe an (algebraic) language as a type. Given a language  $\mathcal{L}$ , in an algebraic context the members of X and  $\mathcal{L}$  are called *individual variables* and *operation symbols* respectively, while in a logical context the members of X and  $\mathcal{L}$  are respectively referred to as propositional variables and logical connectives. The term or formula algebra of type  $\mathcal{L}$  over X, denoted  $\mathbf{T}_{\mathcal{L}}(\mathbf{X})$  or  $\mathbf{Fm}_{\mathcal{L}}$ , is the absolutely free algebra of type  $\mathcal{L}$  over X. In an algebraic context elements  $s, t, u, v, \ldots$  of  $T_{\mathcal{L}}(X)$ are called  $\mathcal{L}$ -terms, while in a logical context elements  $\varphi, \psi, \chi, \ldots$  of  $\operatorname{Fm}_{\mathcal{L}}$  are called  $\mathcal{L}$ -formulas. An  $\mathcal{L}$ -substitution is an endomorphism of the formula algebra over  $\mathcal{L}$ ; notice that an  $\mathcal{L}$ -substitution  $\sigma$  may be identified with its restriction to X by the universal mapping property [55, Lemma II§10.6]. We often

<sup>&</sup>lt;sup>1</sup>The material in this subsection is adapted from van Alten [229, Chapter 0].

drop the prefix  $\mathcal{L}$  from the phrases ' $\mathcal{L}$ -substitution', ' $\mathcal{L}$ -term' and ' $\mathcal{L}$ -formula' when  $\mathcal{L}$  is understood.

1.2.5. Universal Algebra. Algebras are denoted A, B, C, ... and their respective universes are denoted A, B, C, .... An algebra A is trivial if |A| = 1and non-trivial otherwise. Classes of algebras are denoted K, V, .... Throughout this thesis we make standard use of the algebraic class operators I, H, S, P, P<sub>s</sub> (for subdirect products) and P<sub>u</sub> (for ultraproducts). For a class K of similar algebras, we write V(K) for the variety HSP(K) generated by K [55, Theorem II§9.5] and Q(K) for the quasivariety ISPP<sub>u</sub>(K) generated by K [102]; recall Q(K) = ISP(K) when K is finite. We also write K<sub>FIN</sub> for the subclass of finite algebras of K and K<sub>SI</sub> for the subclass of subdirectly irreducible members of K.

Let A be a set. The set of equivalence relations on A is denoted Eq(A) and the lattice of equivalence relations on A is denoted Eq(A). The set of all partitions of A is denoted  $\Pi(A)$  and the lattice of all partitions of A is denoted  $\Pi(A)$ . For  $\pi \in \Pi(A)$  let  $\theta(\pi) := \{\langle a, b \rangle \in A^2 : \{a, b\} \subseteq B$  for some  $B \in \pi\}$ . By [55, Theorem I§4.11]  $\Pi(A)$  is isomorphic to Eq(A) under the mapping  $\pi \mapsto \theta(\pi)$  $(\pi \in \Pi(A))$ . Let  $\theta \in \text{Eq}(A)$ . We write variously  $a\theta b$ ,  $a \equiv b \pmod{\theta}$  and  $a \equiv_{\theta} b$ for  $\langle a, b \rangle \in \theta$ . For every  $a \in A$ , we denote the equivalence class of a modulo  $\theta$ by  $[a]_{\theta}$ . We denote the quotient set by  $A/\theta$ . The identity relation on A is denoted  $\omega_A$  and the universal relation  $A \times A$  is denoted  $\iota_A$ .

The set of congruences on an algebra A is denoted Con A and the lattice of congruences on A is denoted Con A. For  $B \subseteq A^2$ , the congruence on A generated by B is denoted  $\Theta^{\mathbf{A}}(B)$ . For  $a, b \in A$  we abbreviate  $\Theta^{\mathbf{A}}(\{(a, b)\})$ by  $\Theta^{\mathbf{A}}(a, b)$ . The identity congruence on A is denoted  $\omega_{\mathbf{A}}$  and the universal congruence  $A \times A'$  is denoted  $\iota_{\mathbf{A}}$ .

Let K be a class of similar algebras and let  $A \in K$ . A constant term of A or K is any nullary or constant unary term function, or, less precisely, the element of A (or of each member of K) that constitutes the range of such a function [34, p. 551]. A constant is a nullary fundamental operation. We say A is with 0 if 0 is a constant term of A. We say K is with 0 if 0 is a constant term of K.

We say K is *pointed* if it is with 0 for some constant term 0; observe that a pointed class may have more than one constant term.

Let A be an algebra. Elements  $a, b \in A$  and said to be residually distinct if they have distinct images in every non-trivial homomorphic image of A; in symbols,  $\Theta^{A}(a, b) = \iota_{A}$  [34, p. 551]. We say A is with {0,1} if A has constant terms 0 and 1 such that  $0^{A}$  and  $1^{A}$  are residually distinct. We say a class K of similar algebras is with {0,1} if K has constant terms 0 and 1 such that  $0^{A}$ and  $1^{A}$  are residually distinct in any non-trivial member A of K. We say K is dcuble-pointed if it is with {0,1} for some constant terms 0 and 1; notice that a double-pointed class may have more than two constant terms. Observe that any variety V with 0 may be associated with a class  $V^+$  with {0,1} upon adjoining a new nullary operation symbol 1 to the language of V and defining:

 $\mathsf{V}^+ := \{ \langle \mathbf{A}; \mathbf{1} \rangle : \mathbf{A} \in \mathsf{V} \text{ and } \mathbf{0}^{\langle \mathbf{A}; \mathbf{1} \rangle}, \mathbf{1}^{\langle \mathbf{A}; \mathbf{1} \rangle} \in A \text{ are residually distinct} \}$ 

where  $\langle \mathbf{A}; \mathbf{1} \rangle$  is the algebra obtained from  $\mathbf{A}$  by enriching the signature of  $\mathbf{A}$  with the nullary operation symbol 1 whose interpretation is a fixed element  $1 \in A$ . V<sup>+</sup> is called the *generic double-pointed expansion of* V. In certain circumstances V<sup>+</sup> is always guaranteed to be a variety: see Blok and Pigozzi [34, p. 551] for details.

Let A be an algebra with 0. We say A is 0-regular if  $[0]_{\theta} = [0]_{\phi}$  implies  $\theta = \phi$  for all  $\theta, \phi \in \text{Con A}$ . A variety V with 0 is 0-regular if every  $A \in V$  is 0-regular. V is said to be *point regular* if it is 0-regular for some constant term 0. V is strongly 0-regular if V is 0-regular and, for any  $A \in V$ , every compact congruence of A is principal [181, p. 483]. V is strongly point regular if it is strongly 0-regular for some constant term 0.

**Proposition 1.2.6.** [17, Lemma 1.4.10]; [104, Corollary 1.7] Let V be a variety with 0. V is 0-regular iff there exist binary terms  $d_1, \ldots, d_n$  of V such that the identities:

 $d_i(x,x) \approx 0, \qquad \qquad 1 \le i \le n$ 

and the quasi-identity:

 $d_1(x, y) \approx \mathbf{0} \& \ldots \& d_n(x, y) \approx \mathbf{0} \supset x \approx y$ 

hold in V.

Let K be a quasivariety an 1 bet  $A \in K$ . A congruence  $\theta$  on A is called a K-congruence if  $A/\theta \in K$ . The set of all K-congruences on A is denoted Con<sub>K</sub> A; notice Con<sub>K</sub> A = Con A if K is a variety. When ordered by inclusion Con<sub>K</sub> A gives rise to an algebraic lattice Con<sub>K</sub> A [38, p. 633]. We say A is K-subdirectly irreducible if A has a minimal non-trivial K-congruence. By a result of Mal'cev [156], every algebra  $B \in K$  is isomorphic to a subdirect product of K-subdirectly irreducible members of K (that are homomorphic images of B). We say A is K-congruence distributive if Con<sub>K</sub> A is a distributive lattice; K is called K-congruence distributive if every member of K is K-congruence distributive if  $B \in S(A)$  and any K-congruence  $\theta$  of B, there is a K-congruence  $\phi$  of A such that  $\phi \cap (B \times B) = \theta$ . We say K has the K-congruence extension property if every member of K has the K-congruence extension property if with 0. A is called K-0-regular if  $[0]_{\theta} = [0]_{\phi}$  implies  $\theta = \phi$  for all  $\theta, \phi \in Con_K A$ ; K is said to be K-0-regular if every member of K is K-0-regular.

Let V be a variety, let  $A \in V$  and let  $K \subseteq V$  be a fixed subquasivariety of V. A congruence  $\theta$  on A is called a V/K-congruence if  $A/\theta \in K$ . The set of all V/K-congruences on A is denoted  $\operatorname{Con}_{V/K} A$  and yields an algebraic lattice  $\operatorname{Con}_{V/K} A$  under inclusion [88, Section 4.2.1, p. 80]. We say A is V/K-congruence distributive if  $\operatorname{Con}_{V/K} A$  is a distributive lattice; V is V/Kcongruence distributive if every member of V is V/K-congruence distributive. We say A has the V/K-congruence extension property if for any  $B \in S(A)$ and any V/K-congruence  $\theta$  of B, there is a V/K-congruence  $\phi$  of A such that  $\phi \cap (B \times B) = \theta$ ; V has the V/K-congruence extension property if every member of V has the V/K-congruence extension property if every member of V has the V/K-congruence extension property. Suppose A is with 0. A is called V/K-0-regular if  $[0]_{\theta} = [0]_{\phi}$  implies  $\theta = \phi$  for all  $\theta, \phi \in \operatorname{Con}_{V/K} A$ ; V is said to be V/K-0-regular if every member of V is V/K-0-regular.

#### 1.2. Notation and Terminology

**Remark 1.2.7.** Let K be a class of similar algebras and let  $\mathbf{A} := \langle A; \cdots \rangle$  be an algebra of the same type. In some recent literature in algebraic logic a 'relative congruence' is a congruence  $\theta \in \text{Con } \mathbf{A}$  such that  $\mathbf{A}/\theta \in \mathsf{K}$ : see for instance Blok and Raftery [38] or Ferreirim [88]. Although this notion of 'relative congruence' clearly subsumes both the notion of K-congruence and the notion of V/K-congruence as defined above, we explicitly distinguish between K- and V/K-congruences for the sake of clarity in the sequel.

**Theorem 1.2.8 (Principle of the Maximal V/K-Homomorphic Image).** (cf. [62, Proposition 1.7]) Let V be a variety, let  $A \in V$  and let  $K \subseteq V$  be a fixed subquasivariety of V. Then the intersection  $\rho$  of all V/K-congruences on A exists and is a V/K-congruence. Thus  $A/\rho$  is the maximal V/K-homomorphic image of A in the sense that  $A/\rho \in K$ , and every other homomorphic image B of A such that  $B \in K$  is a homomorphic image of  $A/\rho$ .

Proof. Let V, A and K be as in the statement of the theorem. Since the V/Kcongruences of A are closed under arbitrary intersection the intersection  $\rho$  of all V/K-congruences of A exists and is itself a V/K-congruence. Let B be any homomorphic image of A such that  $\mathbf{B} \in K$ . By the homomorphism theorem [55, Theorem II§6.12],  $\mathbf{B} \cong \mathbf{A}/\theta$  for some  $\theta \in \text{Con } \mathbf{A}$ . By hypothesis,  $\mathbf{A}/\theta \in K$ . Thus  $\theta$  is a V/K-congruence, and so  $\rho \subseteq \theta$  by definition of  $\rho$ . But then  $\mathbf{A}/\rho$ is a homomorphic image of  $\mathbf{A}/\theta$  (by [55, Theorem II§6.8; Exercise II§6.6]), so  $\mathbf{A}/\rho$  is a homomorphic image of B.

Let V be a variety, let  $A \in V$  and let  $K \subseteq V$  be a fixed subquasivariety of V. Remark 1.2.7 notwithstanding, we invariably describe the maximal V/Khomomorphic image of A as the maximal K-homomorphic image of A, in keeping with the spirit of the existing literature: see for instance Clifford and Preston [62, p. 18].

**1.2.9.** Algebraic Logic.<sup>2</sup> Let  $\mathcal{L}$  be a language. A pair  $\langle \Gamma, \varphi \rangle$ , where  $\Gamma$  is a finite set of  $\mathcal{L}$ -formulas and  $\varphi$  is an  $\mathcal{L}$ -formula, is called an *(inference)* rule (over  $\mathcal{L}$ ). An axiom is an inference rule of the form  $\langle \emptyset, \varphi \rangle$ ; we invariably identify an axiom  $\langle \emptyset, \varphi \rangle$  with the  $\mathcal{L}$ -formula  $\varphi$ . Let  $Ax \cup Ir$  be a set of axioms

<sup>&</sup>lt;sup>2</sup>The material in this subsection is adapted from van Alten [229, Chapter  $\theta$ ].

#### **1.2.** Notation and Terminology

and inference rules over  $\mathcal{L}$  and let  $\Delta \cup \{\psi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ . A derivation of  $\psi$  from  $\Delta$ (with respect to  $Ax \cup Ir$ ) is a non-empty finite sequence  $\varphi_1, \ldots, \varphi_n$  of  $\mathcal{L}$ -formulas such that  $\varphi_n = \psi$  and for  $i = 1, \ldots, n$ , one of the following conditions holds: (i)  $\varphi_i \in \Delta$ ; (ii)  $\varphi_i$  is a substitution instance of an axiom; or (iii) there exists an inference rule  $\langle \Gamma, \chi \rangle$  of  $Ax \cup Ir$  and a substitution  $\sigma$  such that  $\varphi_i = \sigma(\chi)$  and  $\sigma(\upsilon) \in \{\varphi_1, \ldots, \varphi_{i-1}\}$  for each  $\upsilon \in \Gamma$ . A deductive or Hilbert system  $\mathbb{S}$  (over  $\mathcal{L}$ ) is a pair  $\langle \mathcal{L}, \vdash_{\mathbb{S}} \rangle$  where the binary relation  $\vdash_{\mathbb{S}}$ :  $\mathbb{P}(\operatorname{Fm}_{\mathcal{L}}) \to \operatorname{Fm}_{\mathcal{L}}$  is defined by  $\Gamma \vdash_{\mathbb{S}} \varphi$  iff  $\varphi$  is derivable from  $\Gamma$  with respect to  $Ax \cup Ir$ . The relation  $\vdash_{\mathbb{S}}$ is called the consequence relation of  $\mathbb{S}$ , and a deductive system is sometimes identified with its consequence relation. Typical examples of deductive systems include CPC, the classical propositional calculus, and IPC, the intuitionistic propositional calculus.

Let S be a deductive system over a language  $\mathcal{L}$  determined by a set  $Ax \cup Ir$ of axioms and inference rules. The set  $Ax \cup Ir$  is called an *axiomatisation* of S and the axioms and inference rules in  $Ax \cup Ir$  are called the *axioms* and *inference rules* of S, respectively. Clearly a Hilbert system may have more than one axiomatisation. A Hilbert system for which there exists a finite axiomatisation is said to be *finitely axiomatisable*. For an inference rule  $\langle \Gamma, \varphi \rangle$ of S, we usually write  $\Gamma \vdash_{\mathbb{S}} \varphi$ ; we also write  $\vdash_{\mathbb{S}} \varphi$  for  $\emptyset \vdash_{\mathbb{S}} \varphi$ . An  $\mathcal{L}$ -formula  $\varphi$ for which  $\vdash_{\mathbb{S}} \varphi$  is called a *theorem* of S. We adopt the following conventions concerning sets of  $\mathcal{L}$ -formulas  $\Gamma, \Delta$  and  $\mathcal{L}$ -formulas  $\varphi_1, \ldots, \varphi_n, \psi$ :

 $\begin{array}{ll} \varphi_{1},\ldots,\varphi_{n}\vdash_{\mathbb{S}}\psi & \text{abbreviates} & \{\varphi_{1},\ldots,\varphi_{n}\}\vdash_{\mathbb{S}}\psi;\\ \Gamma,\varphi\vdash_{\mathbb{S}}\psi & \text{abbreviates} & \Gamma\cup\{\varphi\}\vdash_{\mathbb{S}}\psi;\\ \Gamma\vdash_{\mathbb{S}}\Delta & \text{abbreviates} & \Gamma\vdash_{\mathbb{S}}\varphi & \text{for all }\varphi\in\Delta;\\ \Gamma\vdash_{\mathbb{S}}\Delta & \text{abbreviates} & \Gamma\vdash_{\mathbb{S}}\Delta & \text{and }\Delta\vdash_{\mathbb{S}}\Gamma. \end{array}$ 

For sets of  $\mathcal{L}$ -formulas  $\Gamma_1, \ldots, \Gamma_n, \Delta$  and  $\mathcal{L}$ -formulas  $\varphi_1, \ldots, \varphi_n, \psi$ , we also write:

$$\frac{\Gamma_1 \vdash_{\mathbb{S}} \varphi_1 \quad \cdots \quad \Gamma_n \vdash_{\mathbb{S}} \varphi_n}{\Delta \vdash_{\mathbb{S}} \psi}$$

as shorthand notation for  $(\Gamma_1 \vdash_{\mathbb{S}} \varphi_1 \text{ and } \dots \text{ and } \Gamma_n \vdash_{\mathbb{S}} \varphi_n \text{ implies } \Delta \vdash_{\mathbb{S}} \psi'; \text{ by abuse of language we call such a metalogical implication a$ *rule*of S.

Let S be a deductive system over a language  $\mathcal{L}$ . For any  $\varphi, \psi \in \operatorname{Fm}_{\mathcal{L}}$  and  $\Gamma, \Delta \subseteq \operatorname{Fm}_{\mathcal{L}}$ , the consequence relation  $\vdash_{S}$  is easily seen to satisfy the following three conditions:

- 1.  $\varphi \in \Gamma$  implies  $\Gamma \vdash_{\mathbb{S}} \varphi$ ;
- 2.  $\Gamma \vdash_{\mathbb{S}} \varphi$  and  $\Gamma \subseteq \varDelta$  implies  $\varDelta \vdash_{\mathbb{S}} \varphi$ ;
- 3.  $\Gamma \vdash_{\mathbb{S}} \varphi$  and  $\Delta \vdash_{\mathbb{S}} \psi$  for each  $\psi \in \Gamma$  implies  $\Delta \vdash_{\mathbb{S}} \varphi$ .

Moreover,  $\vdash_{S}$  is *finitary* in the sense that:

4.  $\Gamma \vdash_{\mathbb{S}} \varphi$  implies  $\Delta \vdash_{\mathbb{S}} \varphi$  for some finite  $\Delta \subseteq \Gamma$ ;

and  $\vdash_{\mathbf{s}}$  is also structural in the sense that:

5.  $\Gamma \vdash_{\mathbb{S}} \varphi$  implies  $\sigma[\Gamma] \vdash_{\mathbb{S}} \sigma(\varphi)$  for every substitution  $\sigma \in \Sigma$ .

Conversely, the Loś-Suszko theorem [153] (see also [238, Chapter 3§2]) asserts that every relation between sets of  $\mathcal{L}$ -formulas and  $\mathcal{L}$ -formulas satisfying conditions (1)–(5) above is the consequence relation of some Hilbert system S over  $\mathcal{L}$ . Without loss of generality, therefore, a deductive system  $\langle \mathcal{L}, \vdash_S \rangle$  may be defined as a relation  $\vdash_S : \mathbb{P}(\operatorname{Fm}_{\mathcal{L}}) \to \operatorname{Fm}_{\mathcal{L}}$  satisfying (1)–(5) above; defining axioms and inference rules need not be assumed. The above remarks notwithstanding, for binary  $\mathcal{L}$ -formulas  $\wedge, \vee, \to, \Delta$ , a unary  $\mathcal{L}$ -formula  $\neg$  and an *n*-ary logical connective  $\varpi$  of  $\mathcal{L}$ , we identify and earmark the following rules for use in the sequel [238, Section 2.3.1]:

(AD)
$$p, q \vdash_{\mathbb{S}} p \land q$$
(Adjunction)(SP) $p \land q \vdash_{\mathbb{S}} p \\ p \land q \vdash_{\mathbb{S}} q$ (Simplification)(AT) $p \vdash_{\mathbb{S}} p \lor q \\ p \vdash_{\mathbb{S}} q \lor p$ (Addition)

(S)

(SM) 
$$\frac{\Gamma, \varphi \vdash_{\mathbb{S}} \chi \quad \Gamma, \psi \vdash_{\mathbb{S}} \chi}{\Gamma, \varphi \lor \psi \vdash_{\mathbb{S}} \chi} \qquad (Summation)$$

 $(MP) p, p \to q \vdash_{S} q (Modus Ponens)$ 

(DT) 
$$\frac{\Gamma, \varphi \vdash_{\mathbb{S}} \psi}{\Gamma \vdash_{\mathbb{S}} \varphi \to \psi}$$
(Deduction Rule)

(CN)  $p, \neg p \vdash_{\mathbb{S}} q$  (Contradiction)

(RA<sub>J</sub>)  $\frac{\Gamma, \varphi \vdash_{\mathfrak{S}} \neg \varphi}{\Gamma \vdash_{\mathfrak{S}} \neg \varphi}$ (Weak Reductio ad Absurdum) (R)  $\vdash_{\mathfrak{S}} \varphi \Delta \varphi$ (Reflexivity)

 $\varphi \Delta \psi \vdash_{\mathbf{S}} \psi \Delta \varphi \qquad (\text{Symmetry})$ 

(T)  $\varphi \Delta \psi, \psi \Delta \chi \vdash_{\mathbf{S}} \varphi \Delta \chi$  (Transitivity)

 $(\Delta-MP) \qquad \varphi, \varphi \Delta \psi \vdash_{S} \psi \qquad (\Delta-Detachment)$ 

(CP- $\varpi$ )  $\varphi_1 \Delta \psi_1, \ldots, \varphi_n \Delta \psi_n \vdash_{\mathbb{S}} \varpi(\varphi_1, \ldots, \varphi_n) \Delta \varpi(\psi_1, \ldots, \psi_n)$  $(\varpi$ -Compatibility)

A binary  $\mathcal{L}$ -formula  $\land [\lor; \rightarrow]$  is a conjunction [disjunction; conditional] for S if the entailments (AD) and (SP) [(AT) and (SM); (MP) and (DT)] are satisfied by S. A binary  $\mathcal{L}$ -formula  $\triangle$  is called a *G*-identity for S if the the entailments (R), (S), (T), ( $\triangle$ -MP) and (CP- $\varpi$ ) (for every *n*-ary logical connective  $\varpi \in \mathcal{L}$ ) are satisfied by S. A unary  $\mathcal{L}$ -formula  $\neg$  is a weak negation for S if the entailments (CN) and (RA<sub>J</sub>) are satisfied by S. Given a conjunction  $\land$  and a conditional  $\rightarrow$  for S, a biconditional for S is the derived connective:

 $p \leftrightarrow q := (p \rightarrow q) \land (q \rightarrow p).$ 

Let S be a deductive system over a language  $\mathcal{L}$ . An extension of S is any system  $S' := \langle \mathcal{L}, \vdash_{S'} \rangle$  over the same language such that  $\Gamma \vdash_S \varphi$  implies  $\Gamma \vdash_{S'} \varphi$ 

for all  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ . S' is said to be *axiomatic* if it obtained by adjoining new axioms to S only (that is, if the inference rules of S are left fixed). For a sublanguage  $\mathcal{L}' \subseteq \mathcal{L}$ , let  $\vdash_{S'}$  denote the restriction of  $\vdash_{S}$  to  $\mathcal{L}'$  in the sense that  $\Gamma \vdash_{S'} \varphi$  iff  $\Gamma \vdash_{S} \varphi$  and  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}'}$ . The resulting deductive system  $S' := \langle \mathcal{L}', \vdash_{S'} \rangle$  is called the  $\mathcal{L}'$ -fragment of S.

### **1.3 Bands, Monoids and Semilattices**

Various classes of bands, monoids and semilattices play an important role in this thesis. Here we briefly review and summarise some of the theory of these classes that will be needed in the sequel.

**1.3.1.** Semigroups. A groupoid is an algebra  $\langle A; \cdot \rangle$  of type  $\langle 2 \rangle$ . A groupoid whose operation is associative is a semigroup. The binary operation  $\cdot$  of a semigroup  $\langle A; \cdot \rangle$  is called *multiplication*; given  $a, b \in A$ , the multiplication  $a \cdot b$  is (informally) written ab if the context is clear. Given a semigroup  $\langle A; \cdot \rangle$  and  $a_1, \ldots, a_n \in A$ , the product  $a_1 a_2 \cdots a_n$  is defined inductively by:

$$a_1 a_2 \cdots a_n := egin{cases} a_1 & ext{if } n = 1 \ (a_1 a_2 \cdots a_{n-1}) a_n & ext{otherwise.} \end{cases}$$

An easy proof by induction [103, Proposition I§2.1] shows this definition of product has an unambiguous meaning.

**Example 1.3.2.** [87, Section 2] On any non-empty set A two semigroups can always be constructed, *viz*.:

- 1. The left zero semigroup  $A_L$  on A, with multiplication  $a \cdot A_L b := a$  for any  $a, b \in A$ ; and
- 2. The right zero semigroup  $A_R \subset A$ , with multiplication  $a \cdot A_R b := b$  for any  $a, b \in A$ .

A semigroup with zero  $\langle A; \cdot \rangle$  is a semigroup with an element 0 such that  $a\emptyset = 0 = 0a$  for all  $a \in A$ ; the element 0 is called the zero of  $\langle A; \cdot \rangle$ . A

semigroup with identity is defined dually. By abuse of language and notation we will often confuse a semigroup with zero A with the algebra  $\langle A; \cdot, 0 \rangle$ obtained from A by enriching the type of A with a new nullary operation symbol 0 whose canonical interpretation on  $\langle A; \cdot, 0 \rangle$  is  $0 \in A$ , where 0 is the zero of A. Like remarks apply concerning semigroups with identity. It is always possible to adjoin a zero to a semigroup  $\langle A; \cdot \rangle$ . Let  $0 \notin A$  and define:

$$a \cdot b := \begin{cases} a \cdot b & \text{if } a, b \in A \\ 0 & \text{otherwise.} \end{cases}$$

The resulting algebra  $\langle A \cup \{0\}; \cdot^0 \rangle$  is a semigroup with zero and is called a *semigroup with a zero adjoined*.

**1.3.3.** Polrims and Pocrims. A monoid is a semigroup with identity. Let  $\langle A; \oplus, 0 \rangle$  be a monoid whose identity element 0 is the least element of a partial order  $\leq$  on A, compatible with the binary operation  $\oplus$  in the sense that  $a \leq b$  implies both  $a \oplus c \leq b \oplus c$  and  $c \oplus a \leq c \oplus b$  for any  $a, b, c \in A$ . If for every  $a, b \in A$  there is a least element c (denoted  $a \div b$  and called the (*left*) residual of a and b) of A such that  $a \leq c \oplus b$ , then the algebra  $\mathbf{A} := \langle A; \oplus, \div, 0 \rangle$  is a partially ordered left residuated integral monoid (briefly, polrim) [38, pp. 81–82], [229, p. 16]. Polrims arise naturally in algebraic logic and have been considered by Raftery and van Alten [192] and van Alten [229] among others. A partially ordered commutative residuated integral monoid, or pocrim for short, is a polrim  $\langle A; \oplus, \div, 0 \rangle$  whose monoid operation is commutative [229, Example 1.5]. Pocrims also arise naturally in algebraic logic and have been studied by several authors, including Blok and Raftery [39], Fleischer [90], Higgs [109] and Iséki [123]. See also Bosbach [45], Ono and Komori [176], Raftery and van Alten [192].

1.3.4. Dually Relatively Pseudocomplemented Semilattices. A join or upper semilattice is a poset  $\langle A; \leq \rangle$  for which  $lub\{a, b\}$  exists for all  $a, b \in A$ . Let  $\langle A; \lor, 0 \rangle$  be an arbitrary join semilattice with least element and let  $a, b \in A$ . Recall that the dual relative pseudocomplement a \* b of b with respect to a is (if it exists) the unique element of A satisfying  $a \leq b \lor c$  iff  $a * b \leq c$  for all  $c \in A$  [38, Example II]. If a \* b exists for all  $a, b \in A$ , then  $(A; \vee, 0)$  is said to be dually relatively pseudocomplemented. In the case where the operation of dual relative pseudocomplementation is distinguished, the resulting algebra  $\langle A; \vee, *, 0 \rangle$  is called a *dual Brouwerian semilattice* [129, Definition III§4.1]. Dually relatively pseudocomplemented semilattices were introduced (in dually isomorphic form) by Birkhoff in [23, pp. 147-149], while dual Brouwerian semilattices have been studied by Köhler in [135] and by Nemitz [175] in dually isomorphic form under the name *implicative semilat*tices. A result of Blok and Pigozzi shows that the class of dual Brouwerian semilattices is precisely the class of all pocrims for which the monoid operation is idempotent [34, Corollary 1.23] (see also Cornish [67] and Blok and Raftery [39, p. 294]). It is folklore that the lattice of varieties of dual Brouwerian semilattices has a unique atom [39, p. 295], namely the class of all dual Brouwerian semilattices A such that  $A \models x * (y * x) \approx x$ : in this case the semilattice ordering on A is a lattice ordering and a \* b is the complement of b in the interval  $[0, a \lor b]$  for any  $a, b \in A$ . Thus the unique atom in the lattice of varieties of dual Brouwerian semilattices is termwise definitionally equivalent to the variety GBA of generalised Boolean algebras, namely the class of all relatively complemented distributive lattices with zero in which the operations of zero and relative complementation are distinguished. For notational purposes, from this point forth we will always denote the operation of relative complementation in a generalised Boolean algebra by a/b.

An algebra  $\langle A; \wedge, \vee, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is called a *dually relatively pseudocom*plemented lattice if: (i) the reduct  $\langle A; \wedge, \vee \rangle$  is a lattice; and (ii) the reduct  $\langle A; \vee, 0 \rangle$  is dually relatively pseudocomplemented. A *dual Brouwerian lattice* is an algebra  $\langle A; \wedge, \vee, *, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  such that the reduct  $\langle A; \wedge, \vee, 0 \rangle$ is a lattice with zero and the reduct  $\langle A; \vee, *, 0 \rangle$  is a dual Brouwerian semilattice. Such a lattice is always distributive: see Curry [75, Theorem 4§C.5]. Dual Brouwerian lattices have been studied extensively in the literature: see for instance McKinsey and Tarski [161, 162] where they are studied in dually isomorphic form under the name Brouwerian lattices.

1.3.5. Pseudocomplemented Semilattices. A meet or lower semilattice

#### **1.3.** Bands, Monoids and Semilattices

is a poset  $\langle A; \leq \rangle$  for which glb $\{a, b\}$  exists for all  $a, b \in A$ . Let  $\langle A; \wedge, 0 \rangle$  be a meet semilattice with zero. An element  $a^* \in A$  is called a *pseudocomplement* of  $a \in A$  if  $a \wedge a^* = 0$ , and  $a \wedge b = 0$  implies  $b \leq a^*$  [101, p. 58]. An element can have at most one pseudocomplement. An algebra  $\langle A; \wedge, *, 0 \rangle$  of type  $\langle 2, 1, 0 \rangle$  is called a *pseudocomplemented semilattice* if the reduct  $\langle A; \wedge, 0 \rangle$ is a meet semilattice with zero and for any  $a \in A$ , the pseudocomplement of a exists and is  $a^*$ ; observe this definition implies any pseudocomplemented semilattice has a greatest element  $1 := 0^*$  with respect to the underlying semilattice order. Pseudocomplemented semilattices have been studied by several authors, including Frink [96], Jones [128] and Ribenboim [200], to whom the following theorem is due.

**Theorem 1.3.6.** [200] An algebra  $\langle A; \wedge, *, 0 \rangle$  of type  $\langle 2, 1, 0 \rangle$  is a pseudocomplemented semilattice iff the reduct  $\langle A; \wedge, 0 \rangle$  is a meet semilattice with zero and the following identities are satisfied:

| $x \wedge x^* pprox 0$ | (1.8) |
|------------------------|-------|
|------------------------|-------|

 $x \wedge (x \wedge y)^* \approx x \wedge y^* \tag{1.9}$ 

$$x \wedge \mathbf{0}^* \approx x \tag{1.10}$$

$$\mathbf{0^{**}} \approx \mathbf{0}.\tag{1.11}$$

Thus the class PCSL of pseudocomplemented semilattices is a variety.

Let  $\mathbf{B} := \langle B; \land, \lor, ', 0, 1 \rangle$  be a non-trivial Boolean algebra with least element 0 and unit element 1. Let  $\check{B} := B \cup \{m\}$  where a < m for all  $a \in B$ . Then  $\langle \check{B}; \leq \rangle$  is a meet semilattice with zero. Moreover,  $\check{B}$  is pseudocomplemented: for any  $b \in \check{B}$ ,

$$b^{'*} = \begin{cases} m & \text{if } b = 0 \\ b' & \text{if } b \in B, \ b \neq 0 \\ 0 & \text{if } b = m. \end{cases}$$

We denote the resulting pseudocomplemented semilattice by B.

Theorem 1.3.7. [128, Theorem 7.2] A non-trivial pseudocomplemented semi-
lattice is subdirectly irreducible iff it is isomorphic to  $\dot{\mathbf{B}}$  for some Boolean algebra  $\mathbf{B}$ .

**Theorem 1.3.8.** [128, Theorem 11.1] The class of pseudocomplemented semilattices is generated (as a variety) by the 3-element chain 3 (considered as a pseudocomplemented semilattice).

**Corollary 1.3.9.** [128, Theorem 11.1] The lattice of varieties of pseudocomplemented semilattices is a 3-element chain. The only non-trivial subvariety of the variety of pseudocomplemented semilattices is the class  $\{A \in PCSL : A \models x^{**} \approx x\}$ , and this class is termwise definitionally equivalent to the variety of Boolean algebras.

Let A be a pseudocomplemented semilattice. The skeleton S(A) of A is the set  $\{a^* : a \in A\}$  and the dense set D(A) of A is the set  $\{a \in A : a^* = 0\}$ . The following properties of S(A) and D(A) are essentially well known [14, p. 153], [101, p. 59]: (i)  $\{0, 1\} \subseteq S(A)$ ; (ii)  $1 \in D(A)$ ; (iii)  $a \in S(A)$  iff  $a = a^{**}$  for any  $a \in A$ ; and (iv)  $a, b \in S(A)$  implies  $a \land b \in S(A)$ .

**Theorem 1.3.10 (Glivenko-Frink Theorem).** [101, Theorem I§6.4] Let A be a pseudocomplemented semilattice with skeleton S(A). Then the underlying partial ordering of A partially orders S(A) and makes S(A) into a Boolean lattice. For  $a, b \in S(A)$  we have  $a \wedge b \in S(A)$ , and the join in S(A) is described by:

 $a \vee b = (a^* \wedge b^*)^*.$ 

An algebra  $\langle A; \land, \lor, *, 0 \rangle$  of type  $\langle 2, 2, 1, 0 \rangle$  is called a *distributive lattice with* pseudocomplementation if: (i) the reduct  $\langle A; \land, \lor \rangle$  is a distributive lattice; and (ii) the reduct  $\langle A; \land, *, 0 \rangle$  is a pseudocomplemented semilattice. Clearly the class DLPC of all distributive lattices with pseudocomplementation is a variety. Distributive lattices with pseudocomplementation have been extensively studied by many authors; standard references include Grätzer [101, Chapter 3] and Balbes and Dwinger [14, Chapter VIII].

**1.3.11.** Bands. An element e of a semigroup  $\langle A; \cdot \rangle$  is an *idempotent* if  $e^2 = e; \langle A; \cdot \rangle$  is an *idempotent semigroup*, or a *band*, if all its elements are

idempotent. The study of bands dates back to the 1950s and the papers of Kimura [133, 134], McLean [163], and Yamada and Kimura [244]. A detailed development of the theory of bands may be found in Petrich [180, Chapter II]; see also Howie [111, Sections 4.4-4.6].

**Example 1.3.12.** (cf. [111, Theorem 1.1.3]) A semigroup  $\langle A; \cdot \rangle$  satisfying aba = a for any  $a, b \in A$  is called a *rectangular band*. Rectangular bands are precisely the bands satisfying the identity  $x \cdot y \cdot z \approx x \cdot z$ . Any such semigroup is isomorphic to a semigroup of the form  $\langle B \times C; \cdot \rangle$ , where B and C are non-empty sets and multiplication on  $B \times C$  is defined as:

 $(b_1, c_1)(b_2, c_2) := (b_1, c_2)$ 

for any  $b_1, b_2 \in B$  and  $c_1, c_2 \in C$ . The name 'rectangular' stems from this last property: if  $(b_1, c_1)$  and  $(b_2, c_2)$  are construed as vertices of a rectangle in the Cartesian plane, the products  $(b_1, c_1)(b_2, c_2) = (b_1, c_2)$  and  $(b_2, c_2)(b_1, c_1) =$  $(b_2, c_1)$  comprise the remaining two vertices of the rectangle.

Let  $\langle A; \cdot \rangle$  be a band. The *Green's quasiorders* on  $\langle A; \cdot \rangle$  are the relations  $\leq_{\mathcal{L}}$ ,  $\leq_{\mathcal{R}}$  and  $\leq_{\mathcal{D}}$  defined respectively by [208, Section 0]:

$$a \preceq_{\mathcal{L}} b$$
 iff  $ab = a$ ,  $a \preceq_{\mathcal{R}} b$  iff  $ba = a$ ,  $a \preceq_{\mathcal{D}} b$  iff  $aba = a$ 

for any  $a, b \in A$ ; these relations are all proper quasiorderings. Each of the relations  $\leq_{\mathcal{R}}$  and  $\leq_{\mathcal{L}}$  is contained in  $\leq_{\mathcal{D}}$ , and  $\leq_{\mathcal{D}}$  is called the *natural quasiordering* [208, Proposition 1]. In the sequel we write simply  $\leq$  for  $\leq_{\mathcal{D}}$  if there is no danger of confusion. The relation  $\leq_{\mathcal{H}}$  defined on  $\langle A; \cdot \rangle$  by:

$$a \leq_{\mathcal{H}} b$$
 iff  $a \preceq_{\mathcal{L}} and \preceq_{\mathcal{R}} b$  iff  $ab = a = ba$ 

for any  $a, b \in A$  is the natural partial order [103, Proposition II§1.1]; it is properly a partial ordering and is denoted by  $\leq$  hereafter if the context is clear. The following lemma is folklore.

**Lemma 1.3.13.** For a band  $\mathbf{A} := \langle A; \cdot \rangle$  and a fixed  $a \in A$  the following assertions hold:

- 1.  $(a] = \{c : c \leq a\} = aAa$  (for  $aAa := \{ada : d \in A\}$ );
- 2. (a) is a subuniverse of A;

3. The principal subalgebra  $(a] := \langle (a]; \cdot^{\mathbf{A}|_{(a)}} \rangle$  generated by a is a monoid.

*Proof.* For (1), let  $b \in aAa$ . Then b = ada for some  $d \in A$ , whence a(ada) = ada = (ada)a. Thus  $ada \leq a$ , and  $b = ada \in \{c : c \leq a\}$ . Conversely, let  $b \in \{c : c \leq a\}$ . Then  $b \leq a$ , whence  $b = bb = abba = aba \in aAa$ . Thus  $aAa = \{c : c \leq a\}$ . Items (2) and (3) now follow trivially.

The Green's relations  $\mathcal{L}, \mathcal{R}$  and  $\mathcal{D}$  on a band  $\langle A; \cdot \rangle$  are the symmetric parts of the Green's quasiorders, namely [208, Section 0]:

$$\mathcal{L} := \{(a, b) : a \preceq_{\mathcal{L}} b \text{ and } b \preceq_{\mathcal{L}} a\},\$$
$$\mathcal{R} := \{(a, b) : a \preceq_{\mathcal{R}} b \text{ and } b \preceq_{\mathcal{R}} a\},\$$
$$\mathcal{D} := \{(a, b) : a \preceq b \text{ and } b \preceq a\}.$$

Each of  $\mathcal{L}$ ,  $\mathcal{R}$  and  $\mathcal{D}$  is an equivalence relation. Note that since  $\leq$  is a partial ordering, no non-trivial equivalence is induced by  $\leq$  on a band.

**Theorem 1.3.14 (Clifford-McLean Theorem).** [111, Theorem 4.4.1] Each  $\mathcal{D}$ -class of a band  $\mathbf{A}$  is a maximal rectangular subalgebra of  $\mathbf{A}$ . Moreover, the equivalence  $\mathcal{D}$  is a congruence on  $\mathbf{A}$  which induces the maximal semilattice homomorphic image  $\mathbf{A}/\mathcal{D}$  of  $\mathbf{A}$ .

1.3.15. Left [Right] Normal Bands. A band is called *regular* if it satisfies the identity  $x \cdot y \cdot x \cdot z \cdot x \approx x \cdot y \cdot z \cdot x$ . For a band A, the following are equivalent [180, Proposition II.3.6], [133, Theorem 4, Corollary 4]: (i) A is regular; (ii) the relations  $\mathcal{L}$  and  $\mathcal{R}$  are congruences on A; (iii) A decomposes uniquely to within isomorphism as the fibre product  $A/\mathcal{L} \times_{A/\mathcal{D}} A/\mathcal{R}$ . A band is *normal* if the identity  $x \cdot y \cdot z \cdot x \approx x \cdot z \cdot y \cdot x$  holds, and a normal band is regular [244, Lemma 1]. The following lemma is folklore.

**Lemma 1.3.16.** For any band A and  $a, b, c, \in A$ , the following are equivalent:

1. A is normal;

- 2. For any  $a \in A$ , the principal subalgebra (a) generated by a is a semilattice;
- 3. If  $a \leq b$  then  $ac \leq bc$  and  $ca \leq cb$ .

*Proof.* (1)  $\Rightarrow$  (2) Let  $a \in A$  and let  $b, c \in (a]$ . Since the principal subalgebra (a) is a monoid with identity a (by Lemma 1.3.13(3)), we deduce bc = abca = acba = cb by normality.

(2)  $\Rightarrow$  (3) Let  $a \leq b$  and let  $c \in A$ . Since  $bab \leq b$  and  $bcb \leq b$  we deduce bcac = bc(bab)c = (bcb)(bab)c = (bab)(bcb)c = a(bcb)c = (ab)cbc = acbc by (2). But acbc = (ab)cbc = a(bc)bc = a(bc) = (ab)c = ac and so  $ac \leq bc$ . The inclusion  $ca \leq cb$  is handled similarly.

 $(3) \Rightarrow (1)$  Assume (3) holds and notice this is equivalent to the implication:

$$a \le c$$
 and  $b \le d$  implies  $ab \le cd$  (1.12)

for any  $a, b, c, d \in A$ . Let  $a, b, c \in A$ . From (1.12),  $acba \leq a$  and  $abca \leq abca$ we have  $acba = acba(abca)acba \leq a(abca)a = abca$ , just because  $acba\mathcal{D}abca$ . Similarly,  $abca \leq acba$  so acba = abca follows.

A band with zero A is *locally Boolean* if for every  $a \in A$  the principal subalgebra (a) is a Boolean lattice with respect to the natural partial ordering; notice this implies A is normal by Lemma 1.3.16. The following useful technical result is due to the author and the author's Ph.D. supervisor.

**Lemma 1.3.17.** Let A be a locally Boolean band. Then A/D is the maximal semilattice with zero homomorphic image of A. Moreover, every principal order ideal of A/D is a Boolean sublattice of A/D under the semilattice partial ordering, so A/D is locally Boolean.

*Proof.* By the Clifford-McLean theorem,  $\mathbf{A}/\mathcal{D}$  is the maximal semilattice with zero homomorphic image of  $\mathbf{A}$ . Denote by  $\nu$  the canonical epimorphism mapping  $\mathbf{A}$  onto  $\mathbf{A}/\mathcal{D}$ . If  $\mathbf{B}$  is a subsemilattice of  $\mathbf{A}$  then it is immediate from the definition of  $\mathcal{D}$  that the restriction of  $\nu$  to  $\mathbf{B}$  is one-to-one and is thus an isomorphism from  $\mathbf{B}$  onto  $\nu[\mathbf{B}]$ , the image of  $\mathbf{B}$  under  $\nu$ .

Let now *I* be a principal order ideal of  $\mathbf{A}/\mathcal{D}$ . Then *I* is generated by some element  $[b]_{\mathcal{D}} \in A/\mathcal{D}$  where *b* is some element of *A*. For any such element *b* let (*b*] be the principal subalgebra of **A** that it generates. Since (*b*] is a subsemilattice of **A**, (*b*] is isomorphic to  $\nu[(b]]$ , and hence  $\nu[(b]]$  is a Boolean lattice. But  $[c]_{\mathcal{D}} \in \nu[(b]]$  iff  $[c]_{\mathcal{D}} \in (\nu(b)]_{A/\mathcal{D}}$  iff  $[c]_{\mathcal{D}} \leq ^{A/\mathcal{D}} [b]_{\mathcal{D}}$ , so  $\nu[(b]] = I$ , which completes the proof.

A band is left regular [right regular] if the identity  $x \cdot y \cdot x \approx x \cdot y \ [x \cdot y \cdot x \approx y \cdot x]$  is satisfied. For a band A, the following are equivalent [180, Proposition II.3.12]: (i) A is left regular [right regular]; (ii)  $\mathcal{L} = \mathcal{D} \ [\mathcal{R} = \mathcal{D}]$ . Trivially a left regular [right regular] band is regular. A band is left normal [right normal] if the identity  $x \cdot y \cdot z \approx x \cdot z \cdot y \ [x \cdot y \cdot z \approx y \cdot x \cdot z]$  is satisfied. Clearly a left normal [right normal] band is both normal and left regular [right regular] [244, Lemma 1]. The variety InB of left normal bands has been studied by several authors in the literature, including Vagner [227] and Schein [203], to whom the following theorem is due.

**Theorem 1.3.18.** [203] Up to isomorphism, the only subdirectly irreducible left normal bands are 2,  $2_L$  and  $3_L$ , where 2 is the one element semilattice with a zero adjoined,  $2_L$  is the left normal band on  $\{a, b\}$  and  $3_L$  is the band  $2_L$  with a zero adjoined. In symbols,  $\ln B_{SI} = \{2, 2_L, 3_L\}$ .

In the statement of the following corollary and in the sequel  $\ln B_0$  denotes the variety of left normal bands with zero.

Corollary 1.3.19. Up to isomorphism, the only subdirectly irreducible left normal bands with zero are 2 and  $3_L$ . In symbols,  $InB_{0SI} = \{2, 3_L\}$ .

*Proof.* Let  $\langle A; \cdot, 0 \rangle$  be a subdirectly irreducible left normal band with zero. Its band reduct  $\langle A; \cdot \rangle$  must also be subdirectly irreducible, since  $\langle A; \cdot, 0 \rangle$  and  $\langle A; \cdot \rangle$  have the same congruences. Hence  $\langle A; \cdot \rangle$  must be either 2 or  $\mathbf{3}_L$ .

**Example 1.3.20.** Let  $L_n := \langle L; \cdot, 0 \rangle$  be an algebra of cardinality n + 1 equipped with a distinguished element 0 and a binary operation  $\cdot$  defined by:

 $ab := \begin{cases} 0 & \text{if } b = 0 \\ a & \text{otherwise} \end{cases}$ 

for any  $a, b \in L$ . Then  $\mathbf{L}_n$  is a left normal band with zero whose only  $\mathcal{D}$ equivalence classes are  $\{0\}$  and  $L - \{0\}$ . For every  $0 \neq a \in L$ , therefore, (a) is a two element Boolean lattice. Hence  $\mathbf{L}_n$  is locally Boolean and (by Lemma 1.3.17)  $\mathbf{L}_n/\mathcal{D}$  is isomorphic to 2, the one element semilattice with a zero adjoined.

# 1.4 Skew Boolean Algebras and Discriminator Varieties

Discriminator varieties have been called 'the most successful generalisation of Boolean algebras to date, successful because we obtain Boolean product representations' [55, Chapter IV§9, p. 186]. Skew Boolean  $\cap$ -algebras are a class of structurally enriched non-commutative lattices that enjoy deep connections with (pointed) discriminator varieties. We briefly review some of these connections in this section.

1.4.1. Non-commutative Lattices. Recall from [145] that a double band is an algebra  $\langle A; \land, \lor \rangle$  of type  $\langle 2, 2 \rangle$  such that the reducts  $\langle A; \land \rangle$  and  $\langle A; \lor \rangle$ are bands; given the associativity of the operations  $\land$  and  $\lor$  we omit parentheses in the sequel where no ambiguity can arise. In view of the remarks of §1.3.11, on any double band  $\langle A; \land, \lor \rangle$  there arise eight Green's quasiorderings: the four quasiorderings  $\preceq_{\mathcal{L}}^{\langle A; \land \rangle}$ ,  $\preceq_{\mathcal{R}}^{\langle A; \land \rangle}$  and  $\leq_{\mathcal{H}}^{\langle A; \land \rangle}$  on  $\langle A; \land \rangle$ ; and the four quasiorderings  $\preceq_{\mathcal{L}}^{\langle A; \lor \rangle}$ ,  $\preceq_{\mathcal{R}}^{\langle A; \lor \rangle}$ , and  $\leq_{\mathcal{H}}^{\langle A; \lor \rangle}$  on  $\langle A; \lor \rangle$ . A non-commutative lattice is a double band  $\langle A; \land, \lor \rangle$  for which at least one of the quasiorders induced by the operation  $\lor$  is dual to one of the quasiorders induced by the operation  $\land$  in the sense that  $a \preceq_{\mathcal{G}_1}^{\langle A; \land \rangle}$  b iff  $b \preceq_{\mathcal{G}_2}^{\langle A; \lor \rangle} a$  for  $\mathcal{G}_1, \mathcal{G}_2 \in \{\mathcal{L}, \mathcal{R}, \mathcal{D}, \mathcal{H}\}$ ; this definition is implicit in Leech [145, Section 1].

Remark 1.4.2. Traditionally, a non-commutative lattice has been understood as an algebra  $\langle A; \wedge, \vee \rangle$  of type  $\langle 2, 2 \rangle$  whose idempotent and associative operations  $\wedge$  and  $\vee$  satisfy certain absorption identities. Non-commutative lattices in this sense have a long history and have been studied by many authors; see the survey paper of Leech [150, Section 0] for details. Our motivation for departing from this tradition stems from lattice theory: if  $\langle A; \wedge, \vee \rangle$  is a lattice, then each of the operations induces a partial ordering on A that dualises the other [145, Section 1]. For a further discussion and justification of the study of non-commutative lattices in the sense of this thesis, see [150, Section 0] and [145, Section 1].

In this thesis we shall be exclusively concerned with three particular classes of non-commutative lattices: quasilattices, paralattices and (above all) skew lattices.

**1.4.3.** Quasilattices. A quasilattice is a double band  $\langle A; \wedge, \vee \rangle$  such that the quasiorder  $\preceq_{\mathcal{D}}^{\langle A; \wedge \rangle}$  and the quasiorder  $\preceq_{\mathcal{D}}^{\langle A, \vee \rangle}$  dualise in the sense that  $a \preceq_{\mathcal{D}}^{\langle A; \wedge \rangle} b$  iff  $b \preceq_{\mathcal{D}}^{\langle A; \vee \rangle} a$  for any  $a, b \in A$  [145, Section 1]. In view of this duality we work solely with the quasiorder  $\preceq_{\mathcal{D}}^{\langle A; \wedge \rangle}$  in the sequel; to simplify notation we write simply  $a \preceq b$  for  $a \preceq_{\mathcal{D}}^{\langle A; \wedge \rangle} b$  for any  $a, b \in A$  when no confusion can exist. The relation  $\preceq$  is called the *natural quasilattice quasiorder* (*cf.* [149, Section 3.8]). Leech has shown that the class of quasilattices is a variety [145, Section 1], axiomatised relative to the variety of double bands by the identities  $x \land (y \lor x \lor y) \land x \approx x$  and  $y \lor (x \land y \land x) \lor y \approx y$ . Leech has also shown that quasilattice is a lattice of its maximal rectangular subalgebras [145, Corollary 3]. Further information about quasilattices may be found in Leech [145, Sections 1, 2, 3, 4 and 6].

**1.4.4. Paralattices.** A double band  $\langle A; \land, \lor \rangle$  for which the partial orders  $\leq_{\mathcal{H}}^{\langle A; \land \rangle}$  and  $\leq_{\mathcal{H}}^{\langle A; \lor \rangle}$  dualise in the sense that  $a \leq_{\mathcal{H}}^{\langle A; \land \rangle} b$  iff  $b \leq_{\mathcal{H}}^{\langle A; \lor \rangle} a$  for any  $a, b \in A$  is called a *paralattice* [145, Section 1]. In view of this duality we work solely with the partial order  $\leq_{\mathcal{H}}^{\langle A; \land \rangle}$  in the sequel, simply writing  $a \leq b$  for  $a \leq_{\mathcal{H}}^{\langle A; \land \rangle} b$  when no confusion can arise. The relation  $a \leq is$  called the *natural paralattice partial order*; see [146, Section 1.1]. Leech has observed that the class of paralattices is a variety [145, Section 1], axiomatised relative to the variety of double bands by the identities  $x \land (x \lor y \lor x) \approx x \approx (x \lor y \lor x) \land x$  and  $x \lor (x \land y \land x) \approx x \approx (x \land y \land x) \lor x$ . A further study of paralattices may be found in [145, Sections 1, 5 and 6].

**1.4.5.** Skew Lattices. A skew lattice  $\langle A; \land, \lor \rangle$  is a double band such that

 $\preceq_{\mathcal{L}}^{\langle A; \wedge \rangle}$  dualises  $\preceq_{\mathcal{R}}^{\langle A; \vee \rangle}$  and  $\preceq_{\mathcal{R}}^{\langle A; \wedge \rangle}$  dualises  $\preceq_{\mathcal{L}}^{\langle A; \vee \rangle}$  in the sense that  $a \preceq_{\mathcal{L}}^{\langle A; \wedge \rangle} b$  iff  $b \preceq_{\mathcal{R}}^{\langle A; \vee \rangle} a$  and  $a \preceq_{\mathcal{R}}^{\langle A; \wedge \rangle} b$  iff  $b \preceq_{\mathcal{L}}^{\langle A; \vee \rangle} a$  for any  $a, b \in A$  [145, Section 1]. For any skew lattice these absorption dualities are equivalent to the following absorption identities [146, Section 1.1]:

$$x \wedge (x \vee y) \approx x \tag{1.13}$$

$$x \lor (x \land y) \approx x \tag{1.14}$$

$$(y \lor x) \land x \approx x \tag{1.15}$$

$$(y \wedge x) \lor x \approx x. \tag{1.16}$$

It follows that the class of skew lattices is a variety [146, Theorem 1.2]. From the obvious equational axiomatisation for the variety of skew lattices it is clear that any skew lattice  $\mathbf{A} := \langle A; \land, \lor \rangle$  is *self-dual* in the sense that it is closed under three distinct dualisations, *viz.*:

- 1. The horizontal dual:  $\mathbf{A}^h := \langle A; \wedge^h, \vee^h \rangle$ , where  $a \wedge^h b := b \wedge a$  and  $a \vee^h b := b \vee a$ , for any  $a, b \in A$ ;
- 2. The vertical dual:  $\mathbf{A}^{v} := \langle A; \wedge^{v}, \vee^{v} \rangle$ , where  $a \wedge^{v} b := a \vee b$  and  $a \vee^{v} b := a \wedge b$ , for any  $a, b \in A$ ;
- 3. The double dual:  $\mathbf{A}^d := \mathbf{A}^{hv}$ .

For convenience, when we refer in the sequel to *skew lattice duality*, we will mean any of the three distinct dualisations cited above.

**Example 1.4.6.** [146, Section 1.1] Lattices provide an immediate example of a class of skew lattices, since every lattice is clearly a skew lattice satisfying the additional identities  $x \lor y \approx y \lor x$  and  $x \land y \approx y \land x$ .

**Example 1.4.7.** ([146, Section 1.4]; cf. [87, Section 2]) A class of skew lattices distinct from lattices is the class of rectangular skew lattices. Let L and R be non-empty sets and let  $D := L \times R$  be their Cartesian product. Define the operations  $\wedge$  and  $\vee$  for any  $a, b \in L$  and  $a', b' \in R$  as follows:

A sequence of easy checks confirms that the algebra  $\langle D; \land, \lor \rangle$  is a skew lattice.

**Lemma 1.4.8.** (cf. [145, Section 1]) Any skew lattice  $\langle A; \land, \lor \rangle$  is a paralattice. That is, the partial orders  $\leq_{\mathcal{H}}^{\langle A; \land \rangle}$  and  $\leq_{\mathcal{H}}^{\langle A; \lor \rangle}$  dualise in the sense that  $a \leq_{\mathcal{H}}^{\langle A; \land \rangle} b$  iff  $b \leq_{\mathcal{H}}^{\langle A; \lor \rangle} a$  for any  $a, b \in A$ .

Proof. Let A be a skew lattice and  $a, b \in A$ . Suppose  $a \leq_{\mathcal{H}}^{\langle A; \wedge \rangle} b$ . Then  $b \lor a = b \lor (b \land a) = b$  by (1.14). Moreover,  $a \lor b = (a \land b) \lor b = b$  by (1.16). Thus  $b \leq_{\mathcal{H}}^{\langle A; \vee \rangle} a$ . Similarly  $b \leq_{\mathcal{H}}^{\langle A; \vee \rangle} a$  implies  $a \leq_{\mathcal{H}}^{\langle .1; \wedge \rangle} b$  using (1.13) and (1.15).

**Lemma 1.4.9.** (cf. [145, Section 2]) Any skew lattice  $\langle A; \land, \lor \rangle$  is a quasilattice. That is, the quasiorders  $\preceq_{\mathcal{D}}^{\langle A; \land \rangle}$  and  $\preceq_{\mathcal{D}}^{\langle A; \lor \rangle}$  dualise in the sense that  $a \preceq_{\mathcal{D}}^{\langle A; \land \rangle} b$  iff  $b \preceq_{\mathcal{D}}^{\langle A; \lor \rangle} a$  for any  $a, b \in A$ .

*Proof.* Let A be a skew lattice and  $a, b \in A$ . Suppose  $a \preceq_{\mathcal{D}}^{\langle A; \wedge \rangle} b$ . Observe this implies:

$$b \wedge a = (a \wedge (b \wedge a)) \vee (b \wedge a) \qquad \text{by (1.16)}$$
$$= a \vee (b \wedge a) \qquad \text{as } a \preceq_{\mathcal{D}}^{\langle A; \, \wedge \rangle} b \qquad (1.17)$$

and hence that:

$$b \lor a = (b \lor a) \land ((b \lor a) \lor (b \land a)) \qquad \text{by (1.13)}$$
  
=  $(b \lor a) \land (b \lor (a \lor (b \land a)))$   
=  $(b \lor a) \land (b \lor (b \land a)) \qquad \text{by (1.17)}$   
=  $(b \lor a) \land b \qquad \text{by (1.14).}$  (1.18)

Therefore we conclude:

b

$$\forall a \lor b = (b \lor a) \lor b$$

$$= ((b \lor a) \land b) \lor b$$

$$= b$$
by (1.18)
by (1.16)

and so  $b \preceq_{\mathcal{D}}^{\langle A; \vee \rangle} a$ . Similarly  $b \preceq_{\mathcal{D}}^{\langle A; \vee \rangle} a$  implies  $a \preceq_{\mathcal{D}}^{\langle A; \wedge \rangle} b$  by skew lattice duality.

The Green's relations  $\mathcal{L}$ ,  $\mathcal{R}$  and  $\mathcal{D}$  on a skew lattice  $\langle A; \land, \lor \rangle$  are the symmetric parts of the Green's quasiorders, namely:

$$\mathcal{L} := \mathcal{L}^{\langle A; \wedge \rangle} = \mathcal{R}^{\langle A; \vee \rangle} := \{(a, b) : a \preceq_{\mathcal{L}} b \text{ and } b \preceq_{\mathcal{L}} a\}$$
$$\mathcal{R} := \mathcal{R}^{\langle A; \wedge \rangle} = \mathcal{L}^{\langle A; \vee \rangle} := \{(a, b) : a \preceq_{\mathcal{R}} b \text{ and } b \preceq_{\mathcal{R}} a\}$$
$$\mathcal{D} := \mathcal{D}^{\langle A; \wedge \rangle} = \mathcal{D}^{\langle A; \vee \rangle} := \{(a, b) : a \preceq b \text{ and } b \preceq a\}$$

for any  $a, b \in A$ , where  $\mathcal{L}^{\langle A; \wedge \rangle}, \mathcal{R}^{\langle A; \wedge \rangle}$  and  $\mathcal{D}^{\langle A; \wedge \rangle} [\mathcal{L}^{\langle A; \vee \rangle}, \mathcal{R}^{\langle A; \vee \rangle}$  and  $\mathcal{D}^{\langle A; \vee \rangle}]$ denote the various Green's relations on the reduct  $\langle A; \wedge \rangle [\langle A; \vee \rangle]$  [150, Section 1.3]. In the sequel the Green's relation  $\mathcal{D}$  on a skew lattice is called *equivalence* [146, Section 1.6].

**Theorem 1.4.10 (Clifford-McLean Theorem for Skew Lattices).** [146, Theorem 1.7] Let A be a skew lattice. Then equivalence as defined above is a congruence relation. The  $\mathcal{D}$ -equivalence classes are the maximal rectangular subalgebras of A, while the quotient algebra  $A/\mathcal{D}$  is the maximal lattice homomorphic image of A. For all  $a, b \in A$ , the following conditions are equivalent:

1.  $a\mathcal{D}b$ ;

- 2.  $a \wedge b \wedge a \stackrel{\cdot}{=} a \text{ and } b \wedge a \wedge b = b;$
- 3.  $a \wedge b = b \vee a$ .

A skew lattice  $\langle A; \land, \lor \rangle$  is said to be *left handed* if the reduct  $\langle A; \land \rangle$  is left regular; right handed skew lattices are defined dually.

Theorem 1.4.11 (Second Decomposition Theorem for Skew Lattices). [146, Theorem 1.15] Let A be a skew lattice. Then A is regular. That is, A satisfies the identities:

 $\begin{array}{l} x \wedge y \wedge x \wedge z \wedge x \approx x \wedge y \wedge z \wedge x \\ x \vee y \vee x \vee z \vee x \approx x \vee y \vee z \vee x. \end{array}$ 

Thus  $\mathcal{L}$  is a congruence on  $\mathbf{A}$ , and  $\mathbf{A}_R := \mathbf{A}/\mathcal{L}$  is the maximal right handed image of  $\mathbf{A}$ . Dually,  $\mathcal{R}$  is a congruence and  $\mathbf{A}_L := \mathbf{A}/\mathcal{R}$  is the maximal left handed image of  $\mathbf{A}$ . Finally, the following commuting diagram forms a pullback of skew lattices:

$$\begin{array}{ccc} \mathbf{A} & \xrightarrow{\mathcal{L}} & \mathbf{A}_{R} \\ \\ \mathcal{R} & & & \downarrow \mathcal{R} \\ \mathbf{A}_{L} & \xrightarrow{\mathcal{L}} & \mathbf{A}/\mathcal{D} \end{array}$$

Skew lattices were introduced by Leech in a 1989 study [146] of bands of idempotents in rings. Let  $\mathbf{A} := \langle A; +, \cdot \rangle$  be a ring. For any  $a, b \in A$ , define the derived operations  $\wedge$  and  $\vee$  by  $a \wedge b := a \cdot b$  and  $a \vee b := a + b - (a \cdot b)$ . Let  $\mathbf{E}(\mathbf{A})$  denote the set of idempotents of  $\mathbf{A}$ . If  $B \subseteq \mathbf{E}(\mathbf{A})$  is closed under both  $\wedge$  and  $\vee$  as defined above, then  $\langle B; \wedge, \vee \rangle$  is a skew lattice [146, Theorem 2.6]. Even more is true: every multiplicative band in  $\mathbf{E}(\mathbf{A})$  that is maximal with respect to being right regular is a skew lattice [146, Theorem 3.2]. Since their introduction skew lattices have been studied by a number of authors, including Bignall and Leech [19], Leech [146, 147, 148, 150, 145], and Spinks [210, 212].

1.4.12. Skew Lattices with Zero. Let A be a skew lattice. A maximal element of A is an element m such that  $a \wedge m \wedge a = a$ , or, equivalently,  $a \leq m$  for all  $a \in A$ . When they exist maximal elements form an equivalence class under  $\mathcal{D}$  called the maximal class [147, Section 1.4]. Minimal elements of A and the minimal class are defined dually. A skew lattice with zero is a skew lattice  $\langle A; \wedge, \vee \rangle$  for which there exists  $0 \in A$  (the zero of  $\langle A; \wedge, \vee \rangle$ ) such that 0 is the least element under the natural skew lattice partial order; a skew lattice with identity is defined dually. For any skew lattice A the following are equivalent: (i) A is a skew lattice with zero; (ii) for all  $a \in A$ ,  $a \wedge 0 = 0 = 0 \wedge A$ ; (iii) for all  $a \in A$ ,  $a \vee 0 = a = 0 \vee a$ ; (iv) A has a unique minimal element. By abuse of language and notation we will often identify a skew lattice with zero  $\mathbf{A} := \langle A; \wedge, \vee \rangle$  with the algebra  $\langle A; \wedge, \vee, 0 \rangle$  obtained from A by enriching the language of A with a new nullary operation symbol 0 whose canonical interpretation on  $\langle A; \wedge, \vee, 0 \rangle$  is  $0 \in A$ , where 0 is the zero of

A; like remarks apply for skew lattices with identity. The following lemma is immediate.

**Lemma 1.4.13.** Let  $\langle A; \land, 0 \rangle$  [ $\langle A; \land, \lor, 0 \rangle$ ] be a band with zero [skew lattice with zero]. Then  $0 \leq a$  for all  $a \in A$ , and  $a \mathcal{D} 0$  iff a = 0. Thus for any elements  $a, b, c \in A$ , if  $a \land c = 0$  then  $a \land b \land c = 0$ .

1.4.14. Left Handed [Right Handed] Skew Lattices. Let A be a skew lattice and  $a, b \in A$ . By [146, Section 1.13], [150, Section 1.5] and [210, Section 3.3] the following are equivalent: (i) A is left handed; (ii)  $\mathcal{D} = \mathcal{L}$ ; (iii)  $A \models x \land y \land x \approx x \land y$  and  $A \models x \lor y \lor x \approx y \lor x$ ; (iv) if  $a\mathcal{D}b$  then  $a \land b = a$  and  $a \lor b = b$ ; (v)  $A \models x \land (y \lor x) \approx x$  and  $A \models (x \land y) \lor x \approx x$ . Clearly left handed [right handed] skew lattices form a variety. To within isomorphism every skew lattice uniquely decomposes as the fibred product of a right handed skew lattice with a left handed skew lattice over a common underlying maximal lattice homomorphic image [146, Corollary 1.16]: this is an immediate consequence of Theorem 1.4.11. See also [150, Section 1.5, Theorem 1.6].

1.4.15. Symmetric Skew Lattices. A skew lattice A is called meet symmetric if  $a \lor b = b \lor a$  implies  $a \land b = b \land a$  for any  $a, b \in A$ . For a skew lattice A, the following are equivalent [210, Section 3.5]: (i) A is meet symmetric; (ii)  $A \models x \land y \land (x \lor y \lor x) \approx (x \lor y \lor x) \land y \land x$ ; (iii)  $A \models x \land y \land (x \lor y) \approx (y \lor x) \land (x \lor y \lor x) \land y \land x$ ; (iii)  $A \models x \land y \land (x \lor y) \approx (y \lor x) \land (y \land x)$ . A skew lattice A is called join symmetric if  $a \land b = b \land a$  implies  $a \lor b = b \lor a$  for any  $a, b \in A$ . For a skew lattice A, the following are equivalent [210, Section 3.5]: (i) A is join symmetric; (ii)  $A \models x \lor y \lor (x \land y \land x) \approx (x \land y \land x) \lor y \lor x$ ; (iii)  $A \models x \lor y \lor (x \land y) \approx (y \land x) \approx (x \land y \land x) \lor y \lor x$ ; (iii)  $A \models x \lor y \lor (x \land y) \approx (y \land x) \lor y \lor x$ . Meet symmetry for an arbitrary skew lattice does not imply join symmetric if it is both meet symmetric and join symmetric. For a skew lattice A the following are equivalent [146, Proposition 2.3, Theorem 2.4]: (i) A is symmetric; (ii) A is biconditionally commutative (that is,  $a \lor b = b \lor a$  iff  $a \land b = b \land a$ ) [146, Section 2.3]; (iii) the subalgebra B generated from any non-empty, element-wise  $\land$ -commuting

33

subset B of A is a sublattice. The motivation for the study of symmetric skew lattices comes from skew lattices in rings, which are symmetric—see [146, Section 2] and [150, Section 2] for details.

1.4.16. Local Skew Lattices. A skew lattice  $\langle A; \wedge, \vee \rangle$  is said to be *local* when its reduct  $\langle A; \wedge \rangle$  is normal. For a skew lattice **A**, the following are equivalent: (i) **A** is local; (ii) **A**  $\models x \wedge y \wedge z \wedge x \approx x \wedge z \wedge y \wedge x$ ; (iii) the natural skew lattice partial order is preserved under meets; (iv) for all  $a \in A$ , the principal subalgebra (a] generated by a is a sublattice. Local skew lattices have been studied under the name *normal skew lattices* by Leech [148] in conformance with standard semigroup terminology; see also [147, Section 2]. A version of the following lemma is asserted without proof in [148].

Lemma 1.4.17. (cf. [148, Section 2.1]) A local skew lattice is meet symmetric. ric. Thus a local skew lattice is symmetric iff it is join symmetric.

*Proof.* Let A be a local skew lattice and let  $a, b \in A$ . For the first statement it is sufficient in view of the remarks of §1.4.15 to show  $a \lor b = b \lor a$  implies  $a \land b = b \land a$ . So suppose  $a \lor b = b \lor a$ . We have:

| $a \wedge b = ((b \vee a) \wedge a) \wedge (b \wedge (b \vee a))$ | by (1.15) and (1.13)        |
|---|-----------------------------|
| $= ig((a \lor b) \land aig) \land ig(b \land (a \lor b)ig)$       | since $a \lor b = b \lor a$ |
| $= ig((a \lor b) \land big) \land ig(a \land (a \lor b)ig)$       | by normality                |
| $= b \wedge a$  | by (1.15) and (1.13).       |

Thus A is meet symmetric. The second statement now follows trivially.

As with symmetric skew lattices, the study of normal skew lattices is motivated by the study of skew lattices in rings: every maximal normal band of idempotents in a ring forms a normal skew lattice which is the full set of idempotents in the subring it generates [148, p. 1], [147, Theorem 2.2].

1.4.18. Distributive Skew Lattices. There are several different notions of distributivity for skew lattices in the literature. In this thesis a skew lattice is *distributive* if it satisfies the following *middle distributive* identities [150,

Section 2.5]:

$$x \wedge (y \vee z) \wedge x \approx (x \wedge y \wedge x) \vee (x \wedge z \wedge x)$$
(1.19)

$$x \vee (y \wedge z) \vee x \approx (x \vee y \vee x) \wedge (x \vee z \vee x). \tag{1.20}$$

By Theorem 2.8 of Leech [146] skew lattices arising in rings satisfy the identities (1.19)-(1.20): this provides the motivation for the study of distributive skew lattices. Since their introduction distributive skew lattices have been studied in a number of contexts; see for instance [150, Section 0.5] and [152, pp. 13 ff]. In [148, Section 2] Leech gave a range of conditions under which the middle distributivity identities are equivalent for skew lattices, and in [150, Section 2.5] posed the following problem: Are the identities (1.19) and (1.20) equivalent for skew lattices, as they are for lattices? The following nine element counterexample, found using the model generating program SEM [246], answers this question in the negative.

Example 1.4.19. The clauses:

 $\begin{aligned} x \wedge (y \wedge z) &\approx (x \wedge y) \wedge z \\ x \vee (y \vee z) &\approx (x \vee y) \vee z \\ x \wedge (x \vee y) &\approx x \\ x \vee (x \wedge y) &\approx x \\ (y \vee x) \wedge x &\approx x \\ (y \wedge x) \vee x &\approx x \\ x \wedge (y \vee z) \wedge x &\approx (x \wedge y \wedge x) \vee (x \wedge z \wedge x) \\ A \vee (B \wedge C) \vee A \not\approx (A \vee B \vee A) \wedge (A \vee C \vee A) \end{aligned}$ 

have the following model:



Figure 1.1. (a): The skew lattice of Example 1.4.19; (b) Its maximal lattice homomorphic image.

Model (found by SEM 1.7).

| ٨    | 0                      | a                | b | с  | d | е | f | g | 1 |   | V | 0 | a | b | С | d | e | f | g | 1 |
|------|------------------------|------------------|---|----|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|---|
| 0    | 0                      | 0                | 0 | 0  | 0 | 0 | 0 | 0 | 0 | - | 0 | 0 | a | b | С | d | e | f | g | 1 |
| a    | 0                      | a                | a | 0  | 0 | 0 | a | 0 | a |   | a | a | a | b | 1 | f | b | f | 1 | 1 |
| b    | 0                      | $\boldsymbol{a}$ | b | d  | d | e | f | е | b |   | b | b | b | b | 1 | b | b | b | 1 | 1 |
| C    | 0                      | 0                | е | с  | d | е | d | g | с |   | с | c | 1 | 1 | C | с | С | 1 | с | 1 |
| d    | 0                      | 0                | е | d  | d | е | d | e | d |   | d | d | f | f | С | d | d | f | с | 1 |
| е    | 0                      | 0                | e | d  | d | е | d | e | e |   | e | e | b | b | g | е | е | b | g | 1 |
| f    | 0                      | а                | b | `d | d | е | f | е | f |   | f | f | f | f | 1 | f | f | f | 1 | 1 |
| g    | 0                      | 0                | е | с  | d | е | d | g | g |   | g | g | 1 | 1 | g | g | g | 1 | g | 1 |
| 1    | 0                      | a                | b | с  | d | е | f | g | 1 |   | 1 | 1 | 1 | 1 | 1 | 1 | 1 | 1 | 1 | 1 |
| A := | A := a; B := b; C := c |                  |   |    |   |   |   |   |   |   |   |   |   | 1 |   |   |   |   |   |   |

The skew lattice of Example 1.4.19 is depicted in Figure 1.1(a); its maximal lattice homomorphic image is illustrated in Figure 1.1(b). In the Hasse diagram of Figure 1.1(a) the unbroken lines depict the natural skew lattice partial ordering, while the broken horizontal lines connect elements lying in

the same non-trivial clique (that is, elements lying in the same non-trivial  $\mathcal{D}$ class); we remark that we adopt these diagramming conventions without further comment in the sequel in relation to the depiction of bands, (structurally enriched) skew lattices, and more generally any algebra with an underlying partial order and an underlying admissible quasiorder. The  $\mathcal{D}$ -classes form the cliques in Figure 1.1(b); from Figure 1.1(b) it is evident that there are three non-trivial  $\mathcal{D}$ -classes  $\{b, f\}$ ,  $\{g, c\}$  and  $\{e, d\}$ , and three trivial classes  $\{1\}, \{a\}$  and  $\{0\}$ . Since  $\mathcal{D}$ -equivalent elements i and j satisfy  $i \wedge j = j$ and  $i \vee j = i$ , the example is right handed. It follows that the entire skew lattice is determined from  $\mathcal{D}$ -equivalence and the natural partial ordering. In particular, 0 is the zero element since  $i \wedge 0 = 0 = 0 \wedge i$ . Similarly 1 is the identity element since  $1 \lor i = 1 = i \lor 1$  holds. In general, calculations of meets and joins in a skew lattice is trivial if either: (i) the elements involved are comparable; or (ii) the involved meet or join classes are trivial. A fully general account of how non-trivial meets and joins are determined from the geometric structure of a skew lattice is given in [150]. With respect to the non-trivial cases involving  $\{b, f\}$ ,  $\{g, c\}$  and the meet class  $\{e, d\}$ , the situation is as follows. Since  $e \leq b, g$  and  $d \leq f, c$ , the meets  $i \wedge g = e$ and  $i \wedge c = d$  obtain for  $i \in \{b, f\}$ , and  $i \wedge b = e$  and  $i \wedge f = d$  for  $i \in \{g, c\}$ . Thus it is apparent that (1.19) holds, but  $A \vee (B \wedge C) \vee A = f$  $\text{iff } (A \lor B \lor A) \land (A \lor C \lor A) = b \text{ for } \langle A, B, C \rangle \in \{ \langle a, b, c \rangle, \langle a, e, c \rangle \}$ and  $A \lor (B \land C) \lor A = b$  iff  $(A \lor B \lor A) \land (A \lor C \lor A) = f$  for  $\langle A, B, C \rangle \in \{ \langle a, d, g \rangle, \langle a, f, g \rangle \}.$ 

In addition to the counterexample given above, SEM was subsequently able to exhibit a further three non-isomorphic nine element counterexamples satisfying either (1.19) or (1.20) but not both. These remaining counterexamples arise by skew lattice duality in the following manner. Let T denote the skew lattice of Example 1.4.19. Then  $T^h$  is a left handed skew lattice with the same diagram as that of Figure 1.1(a). Likewise  $T^v$  and  $T^d$  both share a distinct diagram, which is the result of 'flipping' the illustration of Figure 1.1(a) across the *bfgc* axis. This diagram is used to determine a unique left handed structure  $T^v$ and a unique right handed structure  $T^d$ . All four skew lattices have the same maximal lattice image as that given in Figure 1.1(b).

37

**Theorem 1.4.20.** The four skew lattices  $\mathbf{T}$ ,  $\mathbf{T}^{\mathbf{h}}$ ,  $\mathbf{T}^{\mathbf{v}}$  and  $\mathbf{T}^{\mathbf{d}}$  form a complete set of counterexamples of minimal order showing that for the skew lattice identities, neither (1.19) nor (1.20) implies the other. This set of counterexamples has the property that any one member is sufficient to generate the remaining three.

**Proof.** The first statement of the theorem obtains because the search performed by SEM is exhaustive, no models of order 8 or less were found during its search, and no other models of order 9 distinct from  $\mathbf{T}$ ,  $\mathbf{T}^h$ ,  $\mathbf{T}^v$  and  $\mathbf{T}^d$  exist. The second statement follows upon observing that if  $\mathbf{T}'$  is any dual of  $\mathbf{T}$ , then the set of duals of  $\mathbf{T}$  and the set of duals of  $\mathbf{T}'$  coincide.

Neither the skew lattice of Example 1.4.19 nor any of its duals are symmetric. This observation has lead Leech to ask [151]: Are the identities (1.19) and (1.20) equivalent for symmetric skew lattices? The following theorem of the author answers this question. The long proof obtained using the automated theorem prover OTTER [158] is omitted, but may be found in [210].

**Theorem 1.4.21.** [210, Section 3.7]; [212, Theorem 2.3] For a symmetric skew lattice A, the following are equivalent:

- 1.  $\mathbf{A} \models x \lor (y \land z) \lor x \approx (x \lor y \lor x) \land (x \lor z \lor x);$
- 2.  $\mathbf{A} \models x \land (y \lor z) \land x \approx (x \land y \land x) \lor (x \land z \land x).$

For symmetrical local skew lattices the situation in relation to distributivity is even more pleasing, as the following theorem of Leech [150] shows.

**Proposition 1.4.22.** [150, Theorem 3.2] For a local skew lattice A, the following are equivalent:

- 1. For each  $a \in A$ , the sublattice (a) is distributive;
- 2. The maximal lattice homomorphic image  $\mathbf{A}/\mathcal{D}$  is distributive;
- 3.  $\mathbf{A} \models x \lor (y \land z) \lor x \approx (x \lor y \lor x) \land (x \lor z \lor x);$
- 4.  $\mathbf{A} \models x \land (y \lor z) \land x \approx (x \land y \land x) \lor (x \land z \land x).$





Figure 1.2. The free symmetric local skew lattice with zero on two free generators  $\overline{x}, \overline{y}$ .

Moreover, the skew lattice subvariety of local skew lattices that are both symmetric and distributive is characterised by the identities:

$$x \wedge (y \vee z) \approx (x \wedge y) \vee (x \wedge z) \tag{1.21}$$

$$(x \lor y) \land z \approx (x \land z) \lor (y \land z). \tag{1.22}$$

**Example 1.4.23.** Let 2 be the one-element semilattice with a zero adjoined, let **R** denote the right handed rectangular skew lattice on  $\{a, b\}$  and let **L** denote its left handed dual. Let  $\mathbf{S} := \mathbf{L} \times \mathbf{R} \times 2 \times 2$ . Upon adjoining a zero to **S** we obtain (by [149, Theorem 4.10]) a skew lattice isomorphic to  $\mathbf{F}(\bar{x}, \bar{y})$ , the free symmetric local skew lattice with zero on two free generators  $\bar{x}, \bar{y}$ : see Figure 1.2. (For notational reasons, the free generators  $\bar{x}, \bar{y}$  are denoted simply by x, y in the diagram (and like remarks apply to products of  $\bar{x}, \bar{y}$ ); also sum (+) and product (juxtaposition) notation is used in the figure instead of the familiar symbols  $\vee$  and  $\wedge$  for join and meet respectively.) From Proposition 1.4.22 we may conclude that  $\mathbf{F}(\bar{x}, \bar{y})$  is distributive, simply because  $\mathbf{F}(\bar{x}, \bar{y})/\mathcal{D}$  is the free distributive lattice with zero on two free generators. It follows from these remarks that  $\mathbf{F}(\bar{x}, \bar{y})$  coincides with the free distributive symmetric local skew lattice on two free generators. In contrast, the free symmetric skew lattice on two free generators is infinite: see Leech [149, Theorem 4.12].

39

Throughout the remainder of this section, all skew lattices of interest are distributive symmetric local skew lattices with zero.

**1.4.24.** Skew Boolean Algebras. A skew lattice A is called *locally Boolean* if: (i) it is symmetric; (ii) it has a zero; and (iii) each principal subalgebra (a] generated by  $a \in A$  is a Boolean lattice [150, Section 3.1]. From the remarks of §1.4.16 and Proposition 1.4.22 it is clear that any locally Boolean skew lattice is distributive symmetric local. The following result, which is folklore, properly characterises the locally Boolean skew lattices among the distributive symmetric local skew lattices; the proof is due to the author and the author's Ph.D. supervisor.

**Proposition 1.4.25.** (cf. [65, Proposition 2.3]) Let A be a distributive symmetric local skew lattice with zero. The following are equivalent:

- 1. A is locally Boolean;
- 2. A/D is a relatively complemented distributive lattice with zero;
- 3. For all  $a, b \in A$  there is a unique  $c \in A$  such that  $(a \land b \land a) \lor c = a$ and  $c \land (a \land b \land a) = 0$ ;
- 4. A has relative complements: if  $b \leq a$ , then b has a unique complement in (a).

*Proof.* (1)  $\Rightarrow$  (2) Suppose A is locally Boolean. By previous remarks and Lemma 1.3.17, A/D is a distributive lattice with zero in which every principal order ideal is a Boolean lattice under the semilattice partial ordering. This is sufficient to guarantee A/D is relatively complemented: see Cornish and Hickman [72] or Cornish [64].

 $(2) \Rightarrow (3)$  Suppose  $\mathbf{A}/\mathcal{D}$  is a relatively complemented distributive lattice with zero. To simplify notation we write  $\overline{a}$  for the equivalence class  $[a]_{\mathcal{D}}$  in  $A/\mathcal{D}$  containing  $a \in A$ . In  $\mathbf{A}/\mathcal{D}$  we have that  $\overline{a} \wedge \overline{b}$  is complemented in the interval from  $\overline{0}$  to  $\overline{a}$ , so there is a  $\overline{w} \leq \overline{a}$  such that  $\overline{w \wedge b} = \overline{0}$  and  $\overline{a \wedge b} \vee \overline{w} = \overline{a}$ . In view of these remarks we have:

 $\overline{a} = \overline{a} \wedge \overline{a} \wedge \overline{a}$ 

$$= \overline{a} \wedge ((\overline{a \wedge b}) \vee \overline{w}) \wedge \overline{a}$$
$$= (\overline{a} \wedge \overline{a \wedge b} \wedge \overline{a}) \vee (\overline{a} \wedge \overline{w} \wedge \overline{a})$$
$$= \overline{a \wedge b \wedge a} \vee \overline{a \wedge w \wedge a}.$$

by middle distributivity

Thus  $a \equiv (a \land b \land a) \lor c \pmod{\mathcal{D}}$ , where  $c = a \land w \land a$ . By the Clifford Mc-Lean theorem,

$$a = a \land ((a \land b \land a) \lor c) \land a$$
  
=  $(a \land (a \land b \land a) \land a) \lor (a \land c \land a)$  by middle distributivity  
=  $(a \land b \land a) \lor (a \land (a \land w \land a) \land a)$   
=  $(a \land b \land a) \lor (a \land w \land a)$   
=  $(a \land b \land a) \lor (a \land w \land a)$   
=  $(a \land b \land a) \lor c.$ 

On the other hand, from  $\overline{w \wedge b} = \overline{0}$  we have:

$$0 = a \land 0 \land a$$
  
=  $a \land w \land b \land a$  by Lemma 1.4.13  
=  $(a \land w \land a) \land (a \land b \land a)$  by regularity  
=  $c \land (a \land b \land a)$ .

To see c is unique, suppose  $(a \land b \land a) \lor c_1 = a = (a \land b \land a) \lor c_2$  while  $a \land b \land a \land c_1 = 0$  and  $a \land b \land a \land c_2 = 0$ . We have:

$$a \wedge c_1 = ((a \wedge b \wedge a) \vee c_1) \wedge c_1$$
  
=  $(a \wedge b \wedge a \wedge c_1) \vee (c_1 \wedge c_1)$  by (1.22)  
=  $0 \vee c_1$   
=  $c_1$ .

Also,

$$a \wedge c_1 = ((a \wedge b \wedge a) \vee c_2) \wedge c_1$$

$$= (a \land b \land a \land c_1) \lor (c_2 \land c_1)$$
by (1.22)  
$$= 0 \lor (c_2 \land c_1)$$
$$= c_2 \land c_1.$$

42

Similarly,  $c_2 \wedge a = c_2$  and  $c_2 \wedge a = c_2 \wedge c_1$ . Thus  $c_1 = a \wedge c_1 = c_2 \wedge c_1 = c_2 \wedge a = c_2$ , completing the proof.

 $(3) \Rightarrow (4)$  Suppose  $b \le a$ . Then  $b \land a = b = a \land b$ , whence  $a \land b \land a = b$ . Since a is clearly the maximal element of the sublattice (a], by (3) we have that c is the unique complement of b in (a].

(4)  $\Rightarrow$  (1) Let  $a \in A$  and let  $b \in (a]$ . Then  $b \leq a$ , and by (4) there is a unique complement of b in (a].

Let A be a locally Boolean skew lattice. The standard difference of  $a, b \in A$ is  $a \setminus b$ , the complement of  $a \land b \land a$  in (a) [19, Definition 3.1]. A skew Boolean algebra is an algebra  $\langle A; \lor, \land, \lor, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  that satisfies all the identities determining distributive symmetric local skew lattices with zero, together with the identities [19, Definition 3.1]:

$$(x \setminus y) \lor (x \land y \land x) \approx x \tag{1.23}$$

 $(x \wedge y \wedge x) \vee (x \backslash y) \approx x \tag{1.24}$ 

$$(x \setminus y) \land (x \land y \land x) \approx \mathbf{0} \tag{1.25}$$

$$(x \wedge y \wedge x) \wedge (x \setminus y) \approx 0. \tag{1.26}$$

Clearly the class SBA of skew Boolean algebras is a variety [147, Theorem 1.8]; for axiomatisations see Spinks [210, Section 4], Cornish [65, Section 2] and Leech [150, Section 3.3]. An important consequence of the identities (1.23)-(1.26) is Proposition 1.4.27 below, which is part of the folklore of non-commutative lattice theory.

Lemma 1.4.26. The variety of skew Boolean algebras satisfies the following

identities:

$$x \wedge (x \setminus y) \approx x \setminus y \tag{1.27}$$

$$(x \setminus y) \land x \approx x \setminus y. \tag{1.28}$$

Proof. Let A be a skew Boolean algebra and let  $a, b \in A$ . We have  $a \setminus b = ((a \land b \land a) \lor (a \setminus b)) \land (a \setminus b) = a \land (a \setminus b)$  by (1.15) and (1.24), which establishes (1.27). Also  $a \setminus b = (a \setminus b) \land ((a \setminus b) \lor (a \land b \land a)) = (a \setminus b) \land a$  by (1.13) and (1.23), which establishes (1.28).

**Proposition 1.4.27.** (Bignall) Let A be a skew Boolean algebra. If  $\theta \in Con \langle A; \wedge, \vee \rangle$ , then  $\theta \in Con A$ . Thus  $Con \langle A; \wedge, \vee \rangle = Con A$ .

*Proof.* It is sufficient to show that  $\theta \in \text{Con} \langle A; \wedge, \vee \rangle$  has the substitution property for the  $\setminus$  operation. Let  $a \equiv a_1 \pmod{\theta}$  and  $b \equiv b_1 \pmod{\theta}$  and notice  $a \wedge b \wedge a \equiv_{\theta} a_1 \wedge b_1 \wedge a_1$ . We have:

$$\begin{aligned} a \setminus b &= a \wedge (a \setminus b) & \text{by } (1.27) \\ &\equiv_{\theta} a_1 \wedge (a \setminus b) & \\ &= \left( (a_1 \wedge b_1 \wedge a_1) \vee (a_1 \setminus b_1) \right) \wedge (a \setminus b) & \text{by } (1.24) \\ &= \left( (a_1 \wedge b_1 \wedge a_1) \wedge (a \setminus b) \right) \vee \left( (a_1 \setminus b_1) \wedge (a \setminus b) \right) & \\ &\equiv_{\theta} \left( (a \wedge b \wedge a) \wedge (a \setminus b) \right) \vee \left( (a_1 \setminus b_1) \wedge (a \setminus b) \right) & \\ &= 0 \vee \left( (a_1 \setminus b_1) \wedge (a \setminus b) \right) & \\ &= (a_1 \setminus b_1) \wedge (a \setminus b). & \\ \end{aligned}$$

Also,

$$a_{1} \setminus b_{1} = (a_{1} \setminus b_{1}) \wedge a_{1} \qquad by (1.28)$$

$$\equiv_{\theta} (a_{1} \setminus b_{1}) \wedge a$$

$$= (a_{1} \setminus b_{1}) \wedge ((a \wedge b \wedge a) \vee (a \setminus b)) \qquad by (1.24)$$

$$= ((a_{1} \setminus b_{1}) \wedge (a \wedge b \wedge a)) \vee ((a_{1} \setminus b_{1}) \wedge (a \setminus b)) \qquad by (1.21)$$

$$\equiv_{\theta} ((a_{1} \setminus b_{1}) \wedge (a_{1} \wedge b_{1} \wedge a_{1})) \vee ((a_{1} \setminus b_{1}) \wedge (a \setminus b))$$

$$= 0 \vee ((a_{1} \setminus b_{1}) \wedge (a \setminus b)) \qquad by (1.25)$$

$$=(a_1\backslash b_1)\wedge (a\backslash b).$$

We conclude  $a \setminus b \equiv_{\theta} a_1 \setminus b_1$ , and the proposition is proved.

**Example 1.4.28.** ([147, Example 1.7(b)]; cf. [65, p. 287]) Let  $\mathbf{A} := \langle A; \land$ ,  $\lor, 0 \rangle$  be a rectangular skew lattice with a zero adjoined. Then  $\mathbf{A}$  is relatively complemented: upon distinguishing the operation of standard difference we obtain a *primitive* skew Boolean algebra consisting of 0 together with a single non-zero equivalence class A.

Primitive skew Boolean algebras play an important role in the theory of skew Boolean algebras. In the statement of the following theorem,  $2^p$  denotes the two element primitive skew Boolean algebra, while  $3^p_L$  and  $3^p_R$  denote the primitive left and right handed three-element skew Boolean algebras respectively.

**Theorem 1.4.29.** [147, Theorem 1.13] A skew Boolean algebra is directly indecomposable iff it is primitive. Up to isomorphism, the only subdirectly irreducible skew Boolean algebras are the algebras  $2^p$ ,  $3^p_L$  and  $3^p_R$ .

Let A be a skew Boolean algebra. A. is called *left handed* [right handed] if its skew lattice reduct  $\langle A; \wedge, \vee \rangle$  is left handed [right handed]. The subvariety of left handed skew Boolean algebras [right handed skew Boolean algebras] is denoted lhSBA [rhSBA]. Left handed skew Boolean algebras were introduced by Cornish in [65] under the name *Boolean skew algebras*. Skew Boolean algebras were introduced in full generality under the name *skew quasi-Boolean algebras* by Leech in [147]. It is not immediately apparent that Cornish's class of Boolean skew algebras coincides with the class of left handed skew Boolean algebras; see Spinks [210, Section 4.1] for a discussion and proof. The motivation for the study of skew Boolean algebras comes in the first instance from ring theory: every maximal normal band of idempotents in a ring forms a skew Boolean algebra [146, Theorem 2.2].

Let A be a skew Boolean algebra with maximal class M. An algebra  $\langle A; \land , \lor, \lor, 0, 1 \rangle$  of type  $\langle 2, 2, 2, 0, 0 \rangle$  obtained from A by adjoining to the language of A a new nullary operation symbol 1 whose canonical interpretation on  $\langle A; \land, \lor, \lor, 0, 1 \rangle$  is a fixed element  $1 \in M$  is a quasi-bounded skew Boolean

algebra. Somewhat confusingly, quasi-bounded skew Boolean algebras were introduced by Leech in [147] under the name skew Boolean algebras.

**Example 1.4.30.** Let  $\mathbf{A} := \langle A; \land, \lor, *, 0 \rangle$  be a distributive lattice with pseudocomplementation. For any  $a, b \in A$ , define the operations:

$$a \overline{\wedge} b := a \wedge b^{**}$$
  
 $a \overline{\vee} b := (a \wedge b^*) \vee b$   
 $a \backslash b := a \wedge b^*;$ 

also, recall  $1 := 0^*$ . An easy but tedious verification shows the induced algebra  $\langle A; \overline{\wedge}, \overline{\vee}, \backslash, 0, 1 \rangle$  is a left handed quasi-bounded skew Boolean algebra iff A is a Stone algebra, namely a distributive lattice with pseudocomplementation satisfying  $x^* \vee x^{**} \approx 1$ . Further,  $\langle A; \overline{\wedge}, \overline{\vee}, \backslash, 0, 1 \rangle$  is term equivalent to a Boolean algebra iff  $\langle A; \wedge, \vee, *, 0, 1 \rangle$  is a Boolean algebra. These remarks yield a new solution to Birkhoff [23, Problem 70].

**Example 1.4.31.** Let  $\mathbf{A} := \langle A; \land, \lor, \rightarrow, 0, 1 \rangle$  be a Heyting algebra, namely a Brouwerian lattice with distinguished least element. It is well known (see for instance [14, p. 174]) that upon defining  $a^* := a \rightarrow 0$  for any  $a \in A$  the induced algebra  $\mathbf{A}^* := \langle A; \land, \lor, *, 0 \rangle$  is a distributive lattice with pseudocomplementation with greatest element  $1 = 0^*$ . For any  $a, b \in A$ , define the operations  $\overline{\wedge}, \overline{\vee}$  and  $\backslash$  as in the preceding example. An easy verification using the identities and quasi-identities of [75, Chapter 4, Section C.2] shows that if  $\mathbf{A}$  is a linearly ordered Heyting algebra (that is, if  $\mathbf{A} \models (x \rightarrow y) \lor (y \rightarrow x) \approx 1$ ) then  $\mathbf{A} \models (x \rightarrow y) \lor ((x \rightarrow y) \rightarrow y) \approx 1$ . It follows that the polynomial reduct  $\mathbf{A}^*$  is a Stone algebra, and thus that the induced algebra  $\langle A; \overline{\wedge}, \overline{\vee}, \backslash, 0, 1 \rangle$  is a left handed quasi-bounded skew Boolean algebra.

1.4.32. Skew Boolean  $\cap$ -Algebras. Let *B* be a subset of the universe of a skew lattice **A**. The infimum of *B* with respect to the underlying natural partial ordering of **A**, if it exists, is called the *intersection* of *B* in *A*, and is denoted  $\bigcap B$  [19, Definition 2.5]. A skew lattice is said to have [finite] intersections if every non-empty [finite] subset has an intersection. A skew

Boolean algebra with intersections is a skew Boolean algebra for which the skew lattice reduct has finite intersections [19, Section 1.4].

**Example 1.4.33.** Let  $\mathbf{A} := \langle A; \land, \lor, \backslash, 0 \rangle$  be a finite skew Boolean algebra. Then for every  $a \in A$ , the principal subalgebra (a) of  $\mathbf{A}$  is finite. Since the reduct  $\langle A; \land, \lor, 0 \rangle$  is a local skew lattice with zero,  $\mathbf{A}$  has intersections by [19, Proposition 2.10].

In the sequel we view skew lattices having finite intersections as as algebras  $\langle A; \land, \lor, \cap \rangle$  of type  $\langle 2, 2, 2 \rangle$ . That the class of such algebras is a variety is an immediate consequence of the following proposition.

**Proposition 1.4.34.** [19, Proposition 2.6] A skew lattice having finite intersections is an algebra  $\langle A; \land, \lor, \cap \rangle$  of type  $\langle 2, 2, 2 \rangle$  such that  $\langle A; \cap \rangle$  is a meet semilattice,  $\langle A; \land, \lor \rangle$  is a skew lattice, and for which the following identities hold:

 $x \cap (x \wedge y \wedge x) \approx x \wedge y \wedge x$  $x \wedge (x \cap y) \approx x \cap y \approx (x \cap y) \wedge x.$ 

A skew Boolean intersection algebra, or skew Boolean  $\cap$ -algebra for short, is an algebra  $\langle A; \land, \lor, \backslash, \cap, 0 \rangle$  of type  $\langle 2, 2, 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \land, \lor, \backslash, 0 \rangle$  is a skew Boolean algebra; and (ii) the reduct  $\langle A; \land, \lor, \cap \rangle$ is a skew lattice with intersections. Clearly the class SBIA of skew Boolean  $\cap$ -algebras is a variety.

**Example 1.4.35.** [19, Example 3.2b] Let  $\langle A; \lor, \land, \backslash, 0 \rangle$  be a primitive skew Boolean algebra, as per Example 1.4.28. Primitive skew Boolean algebras possess arbitrary intersections, with  $\bigcap a_i$  being the common value when all the  $a_i$  are equal, and 0 otherwise; this property characterises the primitive skew Boolean algebras among all the non-trivial skew Boolean algebras with intersections.' A primitive skew Boolean  $\cap$ -algebra is a primitive skew Boolean algebra with intersections in which the intersection operation is distinguished: see Leech [150, Section 4.4].

**Theorem 1.4.36.** [19, Theorem 3.5] The following assertions hold in the variety of skew Boolean  $\cap$ -algebras:

46

- 1. The primitive skew Boolean ∩-algebras are the non-trivial simple algebras;
- 2. The primitive skew Boolean ∩-algebras are the subdirectly irreducible algebras;
- 3. Every non-trivial skew Boolean ∩-algebra is a subdirect product of primitive algebras.

A skew Boolean  $\cap$ -algebra is said to be *left handed* [*right handed*] if its skew lattice reduct is left handed [right handed]. The subvariety of left handed skew Boolean  $\cap$ -algebras [right handed skew Boolean  $\cap$ -algebras] is denoted lhSBIA [rhSBIA]. Skew Boolean  $\cap$ -algebras were introduced in their left handed form under the name quasi-Boolean skew lattices by Bignall in [17], and in full generality by Bignall and Leech in [19]. See also [150, Section 4].

1.4.37. Discriminator Varieties and Dual Discriminator Varieties. The ternary discriminator and dual ternary discriminator on a set A are the functions  $t: A^3 \to A$  and  $d: A^3 \to A$  defined respectively by [183, 95]:

$$t(a, b, c) := \begin{cases} c & \text{if } a = b \\ a & \text{otherwise} \end{cases}$$
 and  $d(a, b, c) := \begin{cases} a & \text{if } a = b \\ c & \text{otherwise.} \end{cases}$ 

A ternary term t(x, y, z) [d(x, y, z)] that realises the ternary discriminator [dual ternary discriminator] on an algebra A is called a *ternary discriminator* term [dual ternary discriminator term] for A. An algebra A is said to be a ternary discriminator algebra [dual ternary discriminator algebra] if it has a ternary discriminator term [dual ternary discriminator term]. If K is a class of algebras of the same similarity type with a common ternary discriminator term [a common dual ternary discriminator term], then the variety V(K) generated by K is called a ternary discriminator variety [dual ternary discriminator variety]. Observe that any ternary discriminator variety is a dual ternary discriminator variety, since  $d(x, y, z) \approx t(x, t(x, y, z), z)$  [95, Section 1]; conversely, a dual ternary discriminator variety is a discriminator variety iff it is congruence permutable [95]. We will always drop the qualifier 'ternary' in

the sequel if the context is clear. In the statement of the following theorem and in the sequel,  $x \Delta y$  abbreviates (for skew Boolean  $\cap$ -algebras) the term  $(x \setminus (x \cap y)) \lor (y \setminus (y \cap x)).$ 

**Theorem 1.4.38.** [19, Theorem 4.4, Corollary 4.9] The class of skew Boolean  $\cap$ -algebras is a discriminator variety, with discriminator term given by:

 $t(x, y, z) := (z \setminus (x \Delta y)) \vee (x \setminus (x \cap y)).$ 

The pure pointed discriminator variety is the pointed discriminator variety of type (3,0) generated by the class of all pointed discriminator algebras  $\langle A; t, 0 \rangle$  where t is the discriminator function on A and 0 is a nullary operation [19, Definition 4.6]. In the statement of the following theorem and in the sequel the pure pointed discriminator variety is denoted by PD<sub>0</sub>.

**Theorem 1.4.39.** [19, Theorem 4.7] The variety  $PD_0$  is termwise definitionally equivalent to the variety of left handed skew Boolean  $\cap$ -algebras. In particular, given  $\langle A; t, 0 \rangle \in PD_0$ , left handed skew Boolean  $\cap$ -operations  $\lor, \land$ and  $\backslash$  and  $\cap$  are defined on A by:

 $a \wedge b := t(b, t(b, 0, a), a)$   $a \vee b := t(b, 0, a)$   $a \setminus b := t(0, b, a)$  $a \cap b := t(a, t(a, b, 0), 0)$ 

for any  $a, b \in A$ . Conversely, given a left handed skew Boolean  $\cap$ -algebra  $\langle A; \land, \lor, \lor, \cap, 0 \rangle$  and  $a, b, c \in A$ , the operation  $t(a, b, c) := (c \lor b) \lor (c \land a) \lor (a \lor (a \cap b))$  yields an algebra  $\langle A; t, 0 \rangle$  in PD<sub>0</sub>.

Bignall and Leech have noted [19, p. 396] that  $PD_0$  is also termwise definitionally equivalent to the variety of right handed skew Boolean  $\cap$ -algebras.

**Corollary 1.4.40.** [19, Corollary 4.8] Any algebra  $\mathbf{A}$  in a pointed discriminator variety has a left handed skew Boolean  $\cap$ -algebra polynomial reduct whose congruences coincide with those of  $\mathbf{A}$ .

48

Discriminator varieties were introduced by Pixley in [183], while dual discriminator varieties were introduced by Fried and Pixley in [95]. Discriminator varieties have been extensively studied in the literature: see [55, Chapter IV§9] for an introductory discussion, [237] for a comprehensive study, and Jonsson's survey of congruence distributive varieties [129, Chapter IV] for a more recent discourse.

# **1.5** Varieties with EDPC

Let S be a strongly algebraisable deductive system with equivalent variety semantics V. It was early understood in the study of algebraic logic that Ssatisfies some reasonable form of the deduction-detachment theorem iff V has equationally definable principal congruences. In conjunction with the realisation that varieties with equationally definable principal congruences are congruence distributive, this fact has lead to an intensive study of such varieties by Blok, Köhler, Pigozzi and others. We summarise some results of their investigations in this section.

1.5.1. Equationally Definable Principal Congruences. Let K be a class of similar algebras. A first-order formula  $\varphi(x, y, u, v)$  in the language of K is said to define principal congruences in K if, for all  $A \in K$  and  $a, b \in A$ ,

 $\Theta^{\mathbf{A}}(a,b) = \{ \langle c,d \rangle \in A \times A : \mathbf{A} \models \varphi[a,b,c,d] \}.$ 

K is said to have *definable principal congruences* if there exists a first-order formula in the language of K that defines principal congruences in K [129, Definition III§2.1].

The study of varieties with definable principal congruences was initiated by Baldwin and Berman in [15], and in [94] Fried, Grätzer and Quackenbush introduced the notion of equationally definable principal congruences. A variety V has Equationally Definable Principal Congruences (EDPC for short) if there exist finitely many pairs  $\langle p_1, q_1 \rangle, \ldots, \langle p_n, q_n \rangle$ ,  $i = 1, \ldots, n$ , of 4-ary terms of V 1.5. Varieties with EDPC

such that for all  $A \in V$  and all  $a, b, c, d \in A$ ,

 $c \equiv d \pmod{\Theta^{\mathbf{A}}(a, b)}$  iff  $p_i^{\mathbf{A}}(a, b, c, d) = q_i^{\mathbf{A}}(a, b, c, d)$ 

for each i = 1, ..., n. The following result is due variously to Blok, Köhler and Pigozzi [136] and van Alten [229]. In the statement of the theorem and in the sequel Cp A [Cp A] denotes the set [join semilattice] of compact congruences on an algebra A.

**Theorem 1.5.2.** ([136]; [229, Proposition 5.19(i)]) For any variety V, the following are equivalent:

- 1. V has EDPC;
- 2. The join semilattice  $\langle Cp A; \lor, \omega_A \rangle$  of compact congruences of A is dually relatively pseudocomplemented for any  $A \in V$ .

Moreover, if V has EDPC, the following statements hold:

- 3. V is congruence distributive and has the congruence extension property;
- 4. V is semisimple iff V is generated (as a variety) by a class of simple algebras.

Since their introduction varieties with EDPC arising from algebraic logic have been systematically studied by Blok and Pigozzi in a series of papers [29, 30, 34, 35]; for a survey of much of this work beyond that presented below see Jonsson [129, Chapter III].

1.5.3. WBSO Varieties and QD Terms. Recall from [29, Lemma 2.7] that a variety of weak Brouwerian semilattices with filter preserving operations (briefly, a WBSO variety) is a variety V with 1 such that: (i) the join semilattice (Cp A;  $\lor, \omega_A$ ) of compact congruences is dually relatively pseudocomplemented; and (ii) there exist binary terms  $\rightarrow, \cdot$  and  $\Delta$  in the language of V such that for any  $A \in V$  and  $a, b \in A$ ,

 $\Theta^{\mathbf{A}}(a \to b, 1) = \Theta^{\mathbf{A}}(b, 1) * \Theta^{\mathbf{A}}(a, 1)$  $\Theta^{\mathbf{A}}(a \cdot b, 1) = \Theta^{\mathbf{A}}(a, 1) \vee \Theta^{\mathbf{A}}(b, 1)$ 

$$\Theta^{\mathbf{A}}(a\Delta b,1) = \Theta^{\mathbf{A}}(a,b)$$

where \* denotes dual relative pseudocomplementation in  $\langle \text{Cp A}; \vee, \omega_{\mathbf{A}} \rangle$ . An algebra **A** is called a *weak Brouwerian semilattice with filter preserving operations* if it is a member of a WBSO variety, or, equivalently,  $\bigvee = \bigvee(\mathbf{A})$  for some WBSO variety V. The terms  $\rightarrow$ ,  $\cdot$  and  $\Delta$  are called *weak relative pseudocomplementation, weak meet* and *Gödel equivalence* terms respectively. In general, none of these terms need be unique; see [29, p. 357]. A *weak join* for a WBSO variety V is a binary term x + y of V with the property that  $\Theta^{\mathbf{A}}(a+b,1) = \Theta^{\mathbf{A}}(a,1) \cap \Theta^{\mathbf{A}}(b,1)$  for all  $\mathbf{A} \in \mathbb{V}$  [29, p. 370]. In general, a WBSO variety need not have a weak join; nor need a weak join, if it exists, be unique.

Let A be a weak Brouwerian semilattice with filter preserving operations. The binary relation  $\preceq$  defined on A by the condition  $a \preceq b$  iff  $a \rightarrow b = 1$ for any  $a, b \in A$  is a quasiordering. The equivalence  $\sim$  induced by  $\preceq$  is a congruence on  $\langle A; \cdot, \rightarrow, 1 \rangle$  and  $\langle A; \cdot, \rightarrow, 1 \rangle / \sim$  is a Brouwerian semilattice that is dually isomorphic to the dual Brouwerian semilattice  $(Cp \mathbf{A}; \vee, *, \omega_{\mathbf{A}})$ under the map  $a \mapsto \Theta^{\mathbf{A}}(a, 1)$  [29, p. 352], [35, p. 7]. A is a weak Boolean algebra with filter preserving operations if the dual Brouwerian semilattice of compact congruences  $(Cp A; \lor, *, \omega_A)$  is (termwise definitionally equivalent to) a generalised Boolean algebra. A subset  $F \subseteq A$  is a weak filter of A if  $1 \in F$ ,  $a \cdot b \in F$  whenever  $a, b \in F$ , and  $b \in F$  whenever  $a \in F$  and  $a \rightarrow b = 1$  [29, p. 351]; the set of weak filters of A is denoted Wf A. Blok, Köhler and Pigozzi have shown that the weak filters of A are exactly the subsets of A of the form  $\bigcup G$ , where G is a filter of  $\langle A; \cdot, \rightarrow, 1 \rangle / \sim [29, p. 354;$ Theorem 2.6]. Call a subset F of the universe of A an *implicative filter* if  $1 \in F$ , and  $a, a \to b \in F$  implies  $b \in F$ . A version of the following result is mentioned without proof in [35, p. 7]; see also [29, p. 352].

Lemma 1.5.4. (cf. [35, p. 7]; cf. [29, p. 352]) Let  $\vee$  be a WBSO variety. Let  $A \in \vee$  and  $\{1\} \subseteq F \subseteq A$ . The following are equivalent:

1.  $F = [1]_{\theta}$  for some  $\theta \in \text{Con } \mathbf{A}$ ;

2. F is a weak filter of A;

### 1.5. Varieties with EDPC

## 3. F is an implicative filter of A.

*Proof.* The equivalence of (1) and (2) is proved in [29, p. 352]. To complete the proof of the lemma it is sufficient in view of preceding remarks to show the equivalence of (2) and (3) in the context of Brouwerian semilattices; for this see either Meng *et al* [164, Theorem 10] or Rasiowa [195, Theorem IV§2.1].

**Proposition 1.5.5.** [29, Lemma 2.2] Let A be a weak Brouwerian semilattice with filter preserving operations. The maps  $\theta \mapsto [1]_{\theta}$  ( $\theta \in \text{Con A}$ ) and  $F \mapsto$  $\{\langle a, b \rangle : a \Delta b \in F\}$  ( $F \in \text{Wf A}$ ) are mutually inverse isomorphisms between the congruence and weak filter lattices of A.

**Proposition 1.5.6.** [29, Lemma 2.4] Let V be a WBSO variety. For any  $A \in V$  the following assertions hold:

1. Every compact congruence of A is principal. In particular, for any  $a, b \in A$ ,

$$\Theta^{\mathbf{A}}(\langle a_1, b_1 \rangle, \dots, \langle a_n, b_n \rangle) = \\\Theta^{\mathbf{A}}((\cdots (a_1 \Delta^{\mathbf{A}} b_1) \cdot^{\mathbf{A}} \cdots) \cdot^{\mathbf{A}} (a_n \Delta^{\mathbf{A}} b_n), 1);$$

2. For any  $a, b, c, d \in A$ ,

$$c \equiv d \pmod{\Theta^{\mathbf{A}}(a, b)}$$
 iff  $(a \Delta^{\mathbf{A}} b) \rightarrow^{\mathbf{A}} (c \Delta^{\mathbf{A}} d) = 1.$ 

Thus V has EDPC.

Let A be an algebra. A Quaternary Deductive (QD) term on A is a term q(x, y, z, w) such that for any  $a, b, c, d \in A$  [29, p. 359],

$$q^{\mathbf{A}}(a, b, c, d) := \begin{cases} c & \text{if } a = b \\ d & \text{if } c \equiv d \pmod{\Theta^{\mathbf{A}}(a, b)}. \end{cases}$$

A QD term on a variety V is a term q(x, y, z, w) such that q(x, y, z, w) is a QD term on every member of V. The QD term may be regarded as a natural

generalisation of the normal transform to non-semisimple varieties; see Blok, Köhler and Pigozzi [29, Section 3] and Blok and Pigozzi [34, p. 547] for details.

**Theorem 1.5.7.** [34, Theorem 1.25(i)] A variety has a QD term iff it is congruence permutable and has EDPC.

**Theorem 1.5.8.** [34, Theorem 1.26] For a variety  $\vee$  with 1 the following are equivalent:

1. V is congruence permutable, 1-regular, and has EDPC;

2. V has a QD term and a Gödel equivalence term;

3. V is a congruence permutable WBSO variety.

1.5.9. TD Terms and Fixedpoint Discriminator Varieties. A ternary term e(x, y, z) is a ternary deductive term for a class K of similar algebras if  $K \models e(x, x, z) \approx z$  and, moreover, for all  $A \in K$  and  $a, b, c, d \in A$ ,  $e^{A}(a, b, c) = e^{A}(a, b, d)$  if  $c \equiv d \pmod{\Theta^{A}(a, b)}$ . A version of the following lemma is proved in Blok and Pigozzi [34]. See also Bignall [18, Lemma 2.1] and McKenzie [159, Theorem 1.3].

Lemma 1.5.10. (cf. [34, Theorem 2.3]) Let V be a variety with a TD term e(x, y, z). For any algebra  $A \in V$ , Con  $A = \text{Con } \langle A; e^A \rangle$ .

*Proof.* Let V be a variety with a TD term e(x, y, z) and let  $A \in V$ . By [34, Theorem 2.3(iii)], A satisfies the identity:

 $e(x, y, f(z_1, \ldots, z_n)) \approx e(x, y, f(e(x, y, z_1), \ldots, e(x, y, z_n)))$ 

for every *n*-ary operation symbol f in the type of A. This implies the relation  $\Theta(a, b)$  defined on  $A \times A$  by:

 $\Theta(a,b) := \{ \langle c,d \rangle \in A \times A : e^{\mathbf{A}}(a,b,c) = e^{\mathbf{A}}(a,b,d) \}$ 

is a congruence relation on **A**, and therefore on  $\langle A; e^{\mathbf{A}} \rangle$  as well. Now if  $c \equiv d \pmod{\Theta(a, b)}$  and  $\phi$  is any congruence on  $\langle A; e^{\mathbf{A}} \rangle$  such that  $c \equiv d \pmod{\phi}$ ,

then:

$$c = e^{\mathbf{A}}(a, a, c) \equiv_{\phi} e^{\mathbf{A}}(a, b, c) = e^{\mathbf{A}}(a, b, d) \equiv_{\phi} e^{\mathbf{A}}(a, a, b) = b,$$

which implies that  $\Theta(a, b)$  is the principal congruence generated by  $\langle a, b \rangle$  on both  $\langle A; e^{\mathbf{A}} \rangle$  and  $\mathbf{A}$ . This means that  $\langle A; e^{\mathbf{A}} \rangle$  and  $\mathbf{A}$  have the same principal congruences. But by [55, Theorem II§5.7(d)] any congruence  $\theta$  on  $\langle A; e^{\mathbf{A}} \rangle$  is given by:

$$\theta = \bigvee \{ \Theta^{\langle A; e^{\mathbf{A}} \rangle}(a, b) : a \equiv b \pmod{\theta} \},\$$

where the join is taken in the lattice of equivalence relations on A. By previous remarks, such a join must also yield a congruence relation on A, whence Con  $\langle A; e^{A} \rangle \subseteq \text{Con } A$ . Since the converse is clear, the result follows.

Let V be a variety with a TD term e(x, y, z). By the proof of the preceding lemma,  $e^{\mathbf{A}}(a, b, c) = e^{\mathbf{A}}(a, b, d)$  iff  $c \equiv d \pmod{\Theta^{\mathbf{A}}(a, b)}$ , whence V has EDPC [34, Corollary 2.5]. Moreover, V is congruence 3-permutable by [34, Theorem 2.9]; in general, V need not be congruence permutable. The following proposition is implicit in [29].

**Proposition 1.5.11.** (cf. [29, p. 361]) For a variety V, the following assertions hold:

1. Suppose V has a TD term e(x, y, z). If p(x, y, z) is a Mal'cev term for V then the term:

$$q(x, y, z, w) := p(e(x, y, z), e(x, y, w), w)$$

is a QD term for V;

Suppose V has a QD term q(x, y, z, w). If V is 1-regular (for some constant term 1) and d₁(x, y),..., dₙ(x, y) are binary terms witnessing the 1-regularity of V in the sense of Proposition 1.2.6 then V is a WBSO variety with weak meet, weak relative pseudocomplementation and Gödel

equivalence terms defined respectively by:

$$egin{aligned} x o y &:= q(x, \mathbf{1}, y, \mathbf{1}) \ x \cdot y &:= q(x, \mathbf{1}, y, x) \ x riangle y &:= (\cdots (d_1(x, y) \cdot d_2(x, y)) \cdot \cdots) \cdot d_n(x, y). \end{aligned}$$

Moreover,  $\forall \models 1 \rightarrow x \approx x$ .

*Proof.* Let V be a variety. For (1), assume V has a TD term e(x, y, z). Suppose first that p(x, y, z) is a Mal'cev term for V. Let  $\mathbf{A} \in V$  and let  $a, b, c, d \in A$ . Suppose  $c \equiv d \pmod{\Theta^{\mathbf{A}}(a, b)}$ . Then  $e^{\mathbf{A}}(a, b, c) = g = e^{\mathbf{A}}(a, b, d)$  for some  $g \in A$  by previous remarks and thus:

$$q^{\mathbf{A}}(a, b, c, d) = p^{\mathbf{A}} \left( e^{\mathbf{A}}(a, b, c), e^{\mathbf{A}}(a, b, d), d \right)$$
  
=  $p^{\mathbf{A}}(g, g, d)$   
=  $d$  since  $p(x, x, y) \approx y$ .

On the other hand, suppose a = b. Then:

$$q^{\mathbf{A}}(a, b, c, d) = p^{\mathbf{A}} (e^{\mathbf{A}}(a, a, c), e^{\mathbf{A}}(a, a, d), d)$$
  
=  $p^{\mathbf{A}}(c, d, d)$  since  $e(x, x, y) \approx y$   
=  $c$  since  $p(x, y, y) \approx x$ .

Thus q(x, y, z, w) is a QD term for V as claimed.

For (2), assume q(x, y, z, w) is a QD term for V. Assume also that V is 1-regular (for some constant term 1). By Theorem 1.5.7 V is congruence permutable with EDPC, so V is a congruence permutable WBSO variety by Theorem 1.5.8. Let  $x \cdot y$ ,  $x \to y$  and  $x \Delta y$  be as in the assertion of the proposition. By [29, Theorem 3.5(i), Theorem 3.7(iii)] we have that the term  $x \to y$  is a weak relative pseudocomplement for V such that  $V \models 1 \to x \approx x$ , while by [29, Theorem 3.5(ii)] we have that the term  $x \cdot y$  is a weak meet. Let  $d_1(x, y), \ldots, d_n(x, y)$  be terms witnessing the 1-regularity of V in the sense of Proposition 1.2.6. Since  $x \cdot y$  is a weak meet for V,

$$\Theta^{\mathbf{A}}(a\Delta^{\mathbf{A}}b,1) = \Theta^{\mathbf{A}}\left(\left(\cdots \left(d_{1}^{\mathbf{A}}(a,b)\cdot^{\mathbf{A}}d_{2}^{\mathbf{A}}(a,b)\right)\cdot^{\mathbf{A}}\cdots\right)\cdot^{\mathbf{A}}d_{n}^{\mathbf{A}}(a,b),1\right)$$
$$= \Theta^{\mathbf{A}}\left(d_{1}^{\mathbf{A}}(a,b),1\right)\vee\Theta^{\mathbf{A}}\left(d_{2}^{\mathbf{A}}(a,b),1\right)\vee\cdots\vee\Theta^{\mathbf{A}}\left(d_{n}^{\mathbf{A}}(a,b),1\right)$$
$$= \Theta^{\mathbf{A}}(a,b)$$

by [29, Theorem 0.7]. Thus  $x \Delta y$  is a Gödel equivalence term for V as claimed.

A TD term e(x, y, z) on an algebra **A** is commutative if  $e^{\mathbf{A}}(a, b, e^{\mathbf{A}}(a', b', c)) = e^{\mathbf{A}}(a', b', e^{\mathbf{A}}(a, b, c))$  for all  $a, b, a', b', c \in A$ ; a TD term e(x, y, z) on a variety V is commutative if it is commutative on every member of V. Let A be a set. A ternary operation  $f : A^3 \to A$  is called a *fixedpoint discriminator* if there exists an element  $d \in A$  such that [34, Definition 3.3]:

 $f(a, b, c) := \begin{cases} c & \text{if } a = b \\ d & \text{otherwise,} \end{cases}$ 

in which case d is called the discriminating element of f. Note that in general the discriminating element associated with a fixedpoint discriminator in a fixedpoint discriminator variety need not be a constant term [34, p. 580]. An algebra A is called a fixedpoint discriminator algebra if there is a ternary term f of A that realises the fixedpoint discriminator on A. A variety V is a fixedpoint discriminator variety if there is a ternary term f of V and a subclass K of V such that  $f^A$  is a fixedpoint discriminator on each  $A \in K$  and V = V(K). In this case f is called a fixedpoint discriminator term for V.

**Theorem 1.5.12.** [34, Theorem 3.5] For any variety V the following are equivalent:

1. V is a fixedpoint discriminator variety;

2. V is semisimple and has a commutative TD term.

Moreover, if the equivalent conditions (1)-(2) are met, then a ternary term e(x, y, z) is a commutative TD term for V iff it is a fixed point discriminator term for V.

Let V be a variety with 0. By remarks due to Blok and Pigozzi [34, p. 582], if V is a ternary discriminator variety then it is a fixedpoint discriminator variety. Indeed, if t(x, y, z) is a discriminator term for V then f(x, y, z) :=t(t(x, y, z), t(x, y, 0), 0) is a fixedpoint discriminator term for V. More generally, we have:

**Theorem 1.5.13.** [34, Theorem 3.8] For any pointed variety V the following are equivalent:

- 1. V is a ternary discriminator variety;
- 2.  $\vee$  is a congruence permutable fixed point discriminator variety;
- 3. V is congruence permutable, semisimple and has a commutative TD term;
- 4. V is congruence permutable, semisimple and has EDPC;
- 5. V is a congruence permutable variety of weak Boolean algebras with filter preserving operations.

If these conditions hold then any constant term can be taken to be the discriminating element of a fixed point discriminator term of V.

Let A be an algebra with 1. A TD term e(x, y, z) on A such that  $a \equiv b \pmod{\Theta^{\mathbf{A}}(e^{\mathbf{A}}(a, b, 1), 1)}$  for all  $a, b \in A$  is said to be regular (for A) with respect to 1; note e(x, y, 1) witnesses 1-regularity in the sense of Proposition 1.2.6 by the remarks of [34, p. 585] and hence that  $\mathbf{1}^{\mathbf{A}}$  is a regular element of A in the usual sense. Let V be a variety with 1 and let e(x, y, z) be a TD term for V. Call e(x, y, z) regular (for V) with respect to 1 if it is regular with respect to 1 for every member of V.

**Theorem 1.5.14.** Let V be a variety with  $\{0,1\}$ . Suppose moreover that V is a fixed point discriminator variety generated by a class  $K \subseteq V$  of fixed point discriminator algebras, that f(x, y, z) is a fixed point discriminator term for V and that  $0^{\mathbf{A}}$  is the discriminating element on any  $\mathbf{A} \in K$ . Then the following statements hold:

1. f(x, y, z) is a commutative TD term for V that is regular with respect to 1;
### 1.6. BCK-Algebras

2. V is a variety of weak Boolean algebras with filter preserving operations whose weak meet, weak relative pseudocomplement and Gödel equivalence terms are defined respectively by:

$$\begin{aligned} x \cdot y &:= f(x, \mathbf{1}, y) \\ x \to y &:= f\left(f(x, \mathbf{1}, y), f(x, \mathbf{1}, \mathbf{1}), \mathbf{1}\right) \\ x \Delta y &:= f(x, y, \mathbf{1}). \end{aligned}$$

*Proof.* The first assertion is just [34, Corollary 4.8]. In view of the first assertion, Theorem 1.5.12 and [34, Theorem 4.4], V is a semisimple WBSO variety with the stated weak meet, weak relative pseudocomplementation and Gödel equivalence terms. Since any semisimple WBSO variety is a variety of weak Boolean algebras with filter preserving operations (by [30, Corollary 4.3]), the second assertion follows.

The fixedpoint discriminator was introduced by Blok and Pigozzi in their study of varieties with EDPC [34] as a generalisation of the ternary discriminator to varieties for which congruence permutability fails, while the TD term was introduced in the same paper as a generalisation of the fixedpoint discriminator. See [34, Definition 3.3; pp. 580-583; pp. 588-590] and [35] for more details.

# 1.6 BCK-Algebras

BCK-algebras play a central role in this thesis. Here we briefly survey the elementary theory of BCK-algebras and some related classes.

1.6.1. BCK-Algebras. For the sake of convenience we repeat here the definition given in §1.1.1. An algebra  $\langle A; -, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is called a *BCK-algebra* iff it satisfies the following identities and quasi-identity [126, Definition 1]:

$$((x - y) - (x - z)) - (z - y) \approx 0$$

$$(x - (x - y)) - y \approx 0$$

$$(1.1)$$

$$(1.2)$$

| $x - x \approx 0$ | (1.3 |
|-------------------|------|
|                   |      |

$$\mathbf{0} - x \approx \mathbf{0} \tag{1.4}$$

$$x - y \approx \mathbf{0} \& y - x \approx \mathbf{0} \supset x \approx y. \tag{1.5}$$

It is immediate that the class BCK of BCK-algebras is a quasivariety; results due to Wroński [240] and Higgs [109] show BCK is not a variety. Because of (1.3) the class of BCK-algebras may be construed as a quasivariety of groupoids; consequently we (informally) denote BCK difference - in the sequel by juxtaposition when no confusion can arise.

BCK-algebras were introduced by Imai and Iséki in a 1966 paper [119]. Historically, the motivation behind the introduction of BCK-algebras was twofold. First, Imai and Iséki wished to give an abstract characterisation of set difference and its properties; and second, Imai and Iséki were interested in investigating systems of implicational calculi related to combinatory logic, particularly the BCK system of Meredith [186, p. 316]. The connection between the two motivations arises from the close relationship observed between set difference in set theory and implication in propositional calculi. Since their introduction BCK-algebras have been the subject of a vast amount of critical exegesis (see for instance the survey articles [126] and [70] and the more recent paper [38] of Blok and Raftery), and in particular connections with the original motivations of Imai and Iséki have been clarified. Indeed, results due to Palasiński [178], Ono and Komori [176], Fleischer [90] and Wroński [242] show that an algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a BCK-algebra iff it is the (-, 0)-subreduct of a pocrim; see Blok and Raftery [39] and van Alten [229, Chapter 1] for details. On the other hand, Blok and Pigozzi [31, Section 5.2.3] have shown BCK is termwise definitionally equivalent to the equivalent algebraic semantics (in the sense of [31]) of BCK logic, while Bunder [51, Theorem 1] has proved that an algebra  $\langle A; -, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a BCK-algebra iff it satisfies the quasi-identity (1.5) and the algebraic analogues of the B-combinator  $(p \rightarrow q) \rightarrow ((q \rightarrow r) \rightarrow (p \rightarrow r))$ , the C-combinator  $(p \rightarrow (q \rightarrow r)) \rightarrow (q \rightarrow (p \rightarrow r))$  and the K-combinator  $p \rightarrow (q \rightarrow p)$ ,

59

## 1.6. BCK-Algebras

respectively the identity (1.1) and the identities:

$$(x - y) - z \approx (x - z) - y \tag{1.29}$$

$$(x - y) - x \approx 0. \tag{1.30}$$

Another important identity known to be satisfied by the quasivariety of BCKalgebras is [126, Theorem 2]:

$$x - \mathbf{0} \approx x \tag{1.7}$$

and in fact Higgs [109, p. 70] and Blok and Raftery [38, Section 1] have shown independently that an algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is a BCK-algebra iff it satisfies the identities (1.1), (1.4), (1.7) and the quasi-identity (1.5).

**Lemma 1.6.2.** [126, p. 4] Let  $\langle A; \div, 0 \rangle$  be a BCK-algebra and let  $\leq_0$  be the relation defined by  $a \leq_0 b$  iff ab = 0 for any  $a, b \in A$ . Then  $\langle A; \leq_0 \rangle$ is a partially ordered set with 0 as its least element. Moreover, right [left] multiplication by a fixed element of A is isotone [antitone].

On an arbitrary BCK-algebra A there exists a second partial order  $\leq_1$ , coarser than  $\leq_0$ , and defined by  $a \leq_1 b$  iff  $a \in bA$  for any  $a, b \in A$  (where  $bA := \{bc : c \in A\}$ ); this observation is due to Guzmán [105, Proposition 3.2(a)]. Proposition 1.6.4 below, which sharpens Guzmán's result, will be needed in the sequel; we remark that this proposition does not seem to have been reported in the literature previously.

Lemma 1.6.3. (cf. [105, Lcmma 1.2(i), Proposition 2.2(c)]) The quasivariety of BCK-algebras satisfies the following identities:

$$x - (x - (x - y)) \approx x - y \tag{1.31}$$

$$(x - y) - z \approx x - (x - ((x - y) - z))$$

$$(1.32)$$

$$(x - y) - (x - (z - (z - y))) \approx 0$$

$$(1.33)$$

$$(x - y) - ((x - y) - z) \approx 0.$$
(1.34)

Proof. Let  $\langle A; \div, 0 \rangle$  be a BCK-algebra and let  $a, b, c \in A$ . To see (1.31) holds, observe that  $a(ab) \leq_0 b$  by (1.2), whence  $ab \leq_0 a(a(ab))$  by Lemma 1.6.2. On the other hand, note that  $a(a(ab)) \leq_0 ab$  by (1.2). Thus a(a(ab)) = abby (1.5). For (1.32), we have 0 = (a(a((ab)c)))((ab)c) by (1.2), whence  $a(a((ab)c)) \leq_0 (ab)c$ . Conversely, put  $\alpha := a$  and  $\beta := (ab)c$ . We have:

| 0 = ((ab)c)((ab)c)               | by (1.3)   |
|----------------------------------|------------|
| = ((ab)((ab)c))c                 | by (1.29)  |
| = ((a((ab)c))b)c                 | by (1.29)  |
| = ((lphaeta)b)c                  |            |
| = ((lpha(lpha(lphaeta)))b)c      | by (1.31)  |
| = ((a(a(a((ab)c))))b)c           |            |
| $= \big((ab)(a(a((ab)c)))\big)c$ | by (1.29)  |
| = ((ab)c)(a(a((ab)c)))           | by (1.29). |

Thus  $(ab)c \leq_0 a(a((ab)c))$ , which establishes (1.32). For (1.33), we have 0 = ((ab)(a(c(cb))))((c(cb))b) = (ab)(a(c(cb))) by (1.1), (1.2) and (1.7). For (1.34), we have 0 = ((ab)(a(bc)))((bc)b) = (ab)(a(bc)) by (1.1), (1.30) and (1.7).

Proposition 1.6.4. Let A be a BCK-algebra. The following statements hold:

1. The binary relation  $\leq_1$  defined for any  $a, b \in A$  by:

 $a \leq_1 b$  iff  $a \in bA$  iff  $a \in b \cap A$  iff  $b \cap a = a$ 

where  $b \cap a := b(ba)$  and  $b \cap A := \{b \cap c : c \in A\}$ , is a partial order on A. Moreover for any  $a, b \in A$ , the relation  $\leq_1$  enjoys the following properties:

- (i) If  $a \leq_1 b$  then  $a \leq_0 b$ ;
- (ii) If  $a \leq_1 b$  then  $ac \leq_1 bc$ ;

(iii)  $0 \leq_1 a$ .

2. The binary relation  $\leq_2$  defined for any  $a, b \in A$  by:

$$a \leq_2 b$$
 iff  $(b(ba))(ba) = a$  iff  $b \cap a = a$  and  $a(ba) = a$ 

is a partial order on A. Moreover for any  $a, b \in A$ , the relation  $\leq_2$ enjoys the following properties:

- (i) If  $a \leq_2 b$  then  $a \leq_1 b$ ;
- (ii) If  $a \leq_2 b$  then  $ac \leq_2 bc$ ;
- (iii)  $0 \leq_2 a$ .

Proof. To prove  $\leq_1$  is a partial ordering under the stated conditions it is sufficient to verify the equivalences  $a \in bA$  iff  $a \in b \cap A$  iff  $b \cap a = a$ , just because the relation  $\leq_1$  defined by  $a \leq_1 b$  iff  $a \in bA$  is a partial order on A by [105, Proposition 3.2(a)]. So suppose that  $a \in bA$ . Then a = bc for some  $c \in A$ , whence  $a = bc = b(b(bc)) \in b \cap A$  by (1.31). Suppose  $a \in b \cap A$ . Then a = b(bc) for some  $c \in A$ . By (1.31) we have  $a = b(bc) = b(b(bc)) = b \cap a$ . Suppose  $b \cap a = a$ . Then  $a = b(ba) \in bA$ , and so  $a \in bA$  iff  $a \in b \cap A$  iff  $b \cap a = a$  as required. To see Item (1)(i) holds, observe  $a \leq_1 b$  implies 0 = (b(ba))b = ab by (1.30) and hence that  $a \leq_0 b$ . For (1)(ii), suppose  $a \leq_1 b$ . We have:

$$(bc)((bc)(ac)) = (bc)((bc)((b(ba))c)) \qquad \text{since } b \cap a = a$$
$$= (bc)((bc)((bc)(ba))) \qquad \text{by (1.29)}$$
$$= (bc)(ba) \qquad \text{by (1 )}$$
$$= (b(ba))c \qquad \text{by (1.2)}$$
$$= ac \qquad \text{since } b \cap a = a.$$

Thus  $ac \leq_1 bc$ . Item (1)(iii) is clear.

To prove  $\leq_2$  is a partial ordering under the stated conditions we first show (b(ba))(ba) = a iff  $b \cap a = a$  and a(ba) = a for any  $a, b \in A$ . Suppose (b(ba))(ba) = a. We have:

$$b \cap a = (b(ba))0 \qquad \qquad by (1.7)$$

 $= (b(ba))((b(ba))a) \qquad \text{by (1.2)}$ =  $(b(ba))((b(ba))((b(ba))(ba))) \qquad \text{since } (b(ba))(ba) = a$ =  $(b(ba))(ba) \qquad \text{by (1.31)}$ =  $a \qquad \text{since } (b(ba))(ba) = a.$ 

It follows that a = (b(ba))(ba) = a(ba). Conversely, if  $b \cap a = a$  and a(ba) = athen a = a(ba) = (b(ba))(ba). Thus (b(ba))(ba) = a iff  $b \cap a = a$  and a(ba) = a as claimed. To see  $\leq_2$  is a partial order we verify the properties of reflexivity, anti-symmetry and transitivity directly. For reflexivity just note  $a \leq_2 a$  from (a(aa))(aa) = (a0)0 = a. For anti-symmetry, observe  $a \leq_2 b$ implies  $a \leq_1 b$  and likewise  $b \leq_2 a$  implies  $b \leq_1 a$ , whence a = b. For transitivity, suppose  $a \leq_2 b$  and  $b \leq_2 c$ . It is sufficient to show  $c \cap a = a$  and a(ca) = a. To see  $c \cap a = a$ , put  $\alpha := c$ ,  $\beta := cb$ ,  $\gamma := ba$  and observe:

$$a = b(ba) \qquad \text{since } b \cap a = a$$
  

$$= (c(cb))(ba) \qquad \text{since } c \cap b = b$$
  

$$= (\alpha\beta)\gamma$$
  

$$= \alpha(\alpha((\alpha\beta)\gamma)) \qquad \text{by (1.32)}$$
  

$$= c(c((c(cb))(ba)))$$
  

$$= c(c(b(ba))) \qquad \text{since } c \cap b = b$$
  

$$= c(ca) \qquad \text{since } b \cap a = a.$$

To see a = a(ca), observe:

$$a(ca) = (b(ba))(ca)$$

$$= (b(ca))(ba)$$

$$= ((c(cb))(ca))(ba)$$

$$= ((c(ca))(cb))(ba)$$

$$= (a(cb))(ba)$$

$$= ((b(ba))(cb))(ba)$$

$$= ((b(cb))(ba))(ba)$$

$$by (1.29)$$

$$since c \cap a = a$$

$$since b \cap a = a$$

$$= ((b(cb))(ba))(ba)$$

$$by (1.29)$$

| = (b(ba))(ba) | since $b(cb) = b$    |
|---------------|----------------------|
| = a(ba)       | since $b \cap a = a$ |
| = a           | since $a(ba) = a$ .  |

Thus  $\leq_2$  is transitive and hence a partial order. Item (2)(i) is clear from preceding remarks. For (2)(ii), it is sufficient to show  $a \leq_2 b$  implies  $ac \leq_1 bc$ and (ac)((bc)(ac)) = ac. So suppose  $a \leq_2 b$ . Then  $a \leq_1 b$  and so  $ac \leq_1 bc$ . Moreover, ((ac)((bc)(ac)))(ac) = ((ac)(ac))((bc)(ac)) = 0((bc)(ac)) = 0by (1.29) and (1.7) and so  $(ac)((bc)(ac)) \leq_0 ac$ . On the other hand, put  $\alpha := ac, \beta := ba$  and  $\gamma := bc$ . We have:

$$(ac)((ac)((bc)(ac))) = (ac)((ac)((bc)((bc)(ba))c))) \qquad \text{since } b \cap a = a$$
$$= (ac)((ac)((bc)((bc)(ba)))) \qquad \text{by (1.29)}$$
$$= ((a(ba))c)((ac)((bc)((bc)(ba)))) \qquad \text{since } a(ba) = a$$
$$= ((ac)(ba))((ac)((bc)((bc)(ba)))) \qquad \text{by (1.29)}$$
$$= (\alpha\beta)(\alpha(\gamma(\gamma\beta)))$$
$$= 0 \qquad \qquad \text{by (1.33).}$$

Thus  $ac \leq_0 (ac)((bc)(ac))$ . We conclude (ac)((bc)(ac)) = ac and thus  $a \leq_2 b$  implies  $ac \leq_2 bc$  as claimed. Item (2)(iii) is clear.

In the sequel we work primarily with the partial order  $\leq_0$ ; we write simply  $\leq$  for  $\leq_0$  when there is no danger of confusion.

**Example 1.6.5.** The partial orders  $\leq_0, \leq_1$  and  $\leq_2$  are distinct on an arbitrary BCK-algebra. To see this, consider the BCK-algebra  $\mathbf{A} := \langle A; \div, 0 \rangle$  where  $A := \{0, a, b, c, 1\}$  and BCK difference is defined by the following operation

1.6. BCK-Algebras



Figure 1.3. The BCK-algebra of Example 1.6.5: (a) Under the partial order  $\leq_0$ ; (b) Under the partial order  $\leq_1$ ; (c) Under the partial order  $\leq_2$ .

table:

| <u>.</u> A | 0 | a | b | с | 1 |
|------------|---|---|---|---|---|
| 0          | 0 | 0 | 0 | 0 | 0 |
| a          | a | 0 | 0 | 0 | 0 |
| b          | b | b | 0 | 0 | 0 |
| с          | с | с | b | 0 | 0 |
| 1          | 1 | С | b | a | 0 |

The Hasse diagrams corresponding to the underlying partially ordered sets  $\langle A; \leq_0 \rangle$ ,  $\langle A; \leq_1 \rangle$  and  $\langle A; \leq_2 \rangle$  are depicted in Figure 1.3. From these diagrams it is clear that the partial orders  $\leq_0$ ,  $\leq_1$  and  $\leq_2$  are distinct.

**Remark 1.6.6.** In contrast to  $\leq_0$ ,  $\leq_1$  and  $\leq_2$  are not in general antitone in each of their positions. To see this, consider the BCK-algebra of Example 1.6.5. One easily checks that  $b \leq_1 c$ , but  $a = 1c \not\leq_1 1b = b$ . Likewise  $0 \leq_2 c$  but  $a = 1c \not\leq_2 10 = 1$ .

**Remark 1.6.7.** Define the  $\langle -, 0 \rangle$ -terms  $x - y^n$ ,  $n \in \omega$  inductively by:

 $\begin{aligned} x - y^0 &:= x \\ x - y^{k+1} &:= (x - y^k) - y \end{aligned} \quad \text{for } k \ge 0. \end{aligned}$ 

65

Owing to (1.30) these terms form a descending chain in an arbitrary BCKalgebra:  $xy^k \leq \ldots \leq xy^2 \leq xy \leq x$ . Let A be a BCK-algebra and  $a, b \in A$ . In the preceding parlance, the BCK partial orders  $\leq_0, \leq_1$  and  $\leq_2$  can be described by:

 $a \leq_0 b$  iff ab = 0 $a \leq_1 b$  iff  $b(ba)^1 = a$  $a \leq_2 b$  iff  $b(ba)^2 = a$ .

From the description above of  $\leq_0, \leq_1$  and  $\leq_2$  it is natural to anticipate that the family of relations  $\{\leq_n : n \geq 1\}$  defined by  $a \leq_n b$  iff  $b(ba)^n = a$  is an infinitely descending chain of partial orders on A. This is not the case: using (1.31) and (1.32) and the  $\langle \div, 0 \rangle$ -terms  $xy^n$  it is possible to show directly that, for any n > 2,  $a \leq_{n+1} b$  iff  $a \leq_n b$ .

**Problem 1.6.8.** Investigate the behaviour of the partial orders  $\leq_i$ , i = 0, 1, 2, on a BCK-algebra. Does there exist a natural family of partial orders on an arbitrary BCK-algebra generalising the orders  $\leq_i$ , i = 0, 1, 2?

Let A be a BCK-algebra. An *ideal* of A is a subset I of A such that  $0 \in I$  and  $a, ba \in I$  implies  $b \in I$ . It is folklore that the set I(A) of all ideals of A forms an algebraic lattice I(A) under set inclusion. Let  $\emptyset \neq B \subseteq A$ . The *ideal*  $\langle B \rangle_A$  of A generated by B is  $\bigcap \{J \in I(A) : B \subseteq J\}$ , the intersection of all ideals of A containing B. By definition,  $\langle \emptyset \rangle_A := \{0\}$ . The following characterisation of  $\langle B \rangle_A$  for non-empty  $B \subseteq A$  is due to Iséki [120, Theorem 3]:

$$\langle B \rangle_{\mathbf{A}} := \big\{ a \in A : (\exists n \in \omega) (\exists b_1, \dots, b_n \in B) (ab_1 \dots b_n = 0) \big\}.$$

The following technical result, which will be needed in the sequel, is due to Pałasinski [177] and Cornish [70].

**Proposition 1.6.9.** For any BCK-algebra A, the following assertions hold:

- 1. [177, Theorem 1] The ideal lattice of A is distributive;
- 2. [70, Theorem 4.1] A enjoys the ideal extension property: whenever  $B \in S(A)$  and  $I \in I(B)$  there exists  $J \in I(A)$  such that  $J \cap B = I$ . In

#### 1.6, BCK-Algebras

particular,  $\langle I \rangle_{\mathbf{A}}$ , the ideal of  $\mathbf{A}$  generated by I, can always be taken as a suitable J.

In general, an ideal I of A is the 0-class of at least one, and possibly many congruences on A, of which the largest  $\phi_I := \{(a, b) \in A \times A : ab, ba \in I\}$  is actually a BCK-congruence. Conversely, for any  $\theta \in \text{Con } A$ , the 0-class  $\{a \in A : (a, 0) \in \theta\}$  is an ideal of A.

**Theorem 1.6.10.** [38, Proposition 1] For a BCK-algebra A the following assertions hold:

- 1. The maps  $I \stackrel{\xi}{\mapsto} \phi_I$   $(I \in I(\mathbf{A}))$  and  $\theta \stackrel{\xi^{-1}}{\mapsto} [0]_{\theta}$   $(\theta \in \operatorname{Con}_{\mathsf{BCK}} \mathbf{A})$  are mutually inverse lattice isomorphisms between the BCK-congruence lattice of  $\mathbf{A}$  and the ideal lattice of  $\mathbf{A}$ ;
- 2. A is BCK-0-regular, BCK-congruence distributive and enjoys the BCKcongruence extension property;
- 3.  $H(A) \subseteq BCK$  iff A is 0-regular, in which case A is also congruence distributive. If  $HS(A) \subseteq BCK$ , then A has the congruence extension property.

It is known that the congruences (in the absolute sense) of BCK-algebras are not well-behaved in general. In particular, Wroński [241, Theorem 5, Theorem 6] and Nagayama [173, Theorem 1.3] (see also Example 2.3.10 in the sequel) have shown that the congruence lattices of BCK-algebras need satisfy no lattice identities beyond those satisfied by all lattices. An example due to Blok and Raftery [37] shows also that BCK-algebras do not in general enjoy the congruence extension property. In contrast, Proposition 1.6.9 and Theorem 1.6.10 show the situation to be quite different for *BCK-varieties*, namely those varieties of algebras of type  $\langle 2, 0 \rangle$  whose members are BCK-algebras. In the remainder of this section we describe some BCK-varieties of relevance to the sequel, together with several equational classes of BCK-algebras augmented with additional operations.

**1.6.11.** Commutative BCK-Algebras. A BCK-algebra  $\langle A; \div, 0 \rangle$  for which the underlying partially ordered set  $\langle A; \leq \rangle$  is a meet semilattice is called

a commutative BCK-algebra; in this case  $a \cap b := a(ab)$  (the BCK meet) is the greatest lower bound of  $\{a, b\}$  for any  $a, b \in A$ . For any BCK-algebra A, the following are equivalent [126, Theorem 3]: (i) A is a commutative BCKalgebra; (ii)  $A \models x \cap y \approx y \cap x$ . From (ii) it follows easily that the class cBCK of all commutative BCK-algebras is a variety [245]. Problem 1.6.8 notwithstanding, we also have the following characterisation of commutative BCK-algebras, which will be needed in the sequel. See also Cornish [71, Proposition 1.8].

**Lemma 1.6.12.** A BCK-algebra is commutative iff its underlying partial orders  $\leq_0$  and  $\leq_1$  coincide.

*Proof.* Let A be a BCK-algebra and let  $a, b \in A$ .

(⇒) Suppose A is commutative. If  $a \leq_1 b$  then  $a \leq_0 b$  by Proposition 1.6.4(1)(i). For the converse, if  $a \leq_0 b$  then  $a \cap b = a$ , and hence also  $b \cap a = a$  since A is commutative. But this means that  $a \leq_1 b$  by Proposition 1.6.4(1)(i).

( $\Leftarrow$ ) Suppose  $a \leq_0 b$  iff  $a \leq_1 b$ . Because  $a \cap b \leq_0 b$ , by hypothesis we have that  $a \cap b \leq_1 b$ , whence  $b \cap (a \cap b) = a \cap b$  by Proposition 1.6.4(1)(i). Put  $\alpha := b, \beta := a$  and  $\gamma := ab$ . We have:

$$(a \cap b)(b \cap a) = (b \cap (a \cap b))(b \cap a)$$
  
=  $(b(b(a(ab))))(b(ba))$   
=  $(b(b(ba)))(b(a(ab)))$  by (1.29)  
=  $(ba)(b(a(ab)))$  by (1.31)  
=  $(\alpha\beta)(\alpha(\beta\gamma))$   
= 0 by (1.34).

Thus  $a \cap b \leq_0 b \cap a$ , and by symmetry  $b \cap a \leq_0 a \cap b$ . Therefore  $a \cap b = b \cap a$ , and A is commutative.

Commutative BCK-algebras were introduced by Tanaka [214] and have been studied subsequently by several authors, including Traczyck [216], Cornish [64, Section 3], Romanowska and Traczyck [202, 201], Cornish, Sturm and Traczyk [73] and Yutani [245]. In particular, results due to Traczyk [216, Lemma 2.1,

#### 1.6. BCK-Algebras

Theorem 2.4] show that the underlying meet semilattice  $\langle A; \cap \rangle$  of a commutative BCK-algebra **A** is in fact a distributive nearlattice; recall from [70, p. 112] and [64, p. 489] that a [distributive] nearlattice is a lower semilattice  $\langle A; \wedge \rangle$ in which each principal order ideal  $(m] := \{a \in A : a \leq m\}$  is a [distributive] lattice. For  $a \in (m]$ , the map  $N_m : (m] \to (m]$  defined by  $N_m(a) := ma$  is an involution (dual order isomorphism that is its own inverse), whence the supremum of  $a, b \in (m]$  is  $a \cup b := N_m (N_m(a) \cap N_m(b))$ . See Cornish [64, Lemma 3.1] and Cornish and Hickman [72].

**1.6.13.** Positive Implicative BCK-Algebras. Let A be a BCK-algebra with underlying poset  $\langle A; \leq \rangle$ . For any  $a, b, c \in A$ ,  $(ab)c \leq (ac)(bc)$  [126, p. 12]; in general, the opposite inclusion does not hold. A *positive implicative* BCK-algebra is a BCK-algebra for which the inequality  $(ac)(bc) \leq (ab)c$  is identically satisfied. For a BCK-algebra A, the following are equivalent [38, Proposition 13]: (i) A is positive implicative; (ii)  $\mathbf{A} \models (x - y) - y \approx x - y$ ; (iii)  $\mathbf{A} \models (x \div (x \div y)) \div (y \div x) \approx (y \div (y \div x)) \div (x \div y). \text{ Henkin [108] appears}$ to have been among the first to consider positive implicative BCK-algebras: they are precisely his class of *implicative models*. Since Henkin's 1950 paper positive implicative BCK-algebras have been independently investigated by a number of authors, including Diego [79] (in dually isomorphic form under the name Hilbert algebras—see Kondo [140]), Rasiowa [195, Section II§2] (in dually isomorphic form under the name positive implication algebras), Iseki and Tanaka [126] and more recently Blok and Raftery [39]. Results due to Diego [79] show the class pBCK of positive implicative BCK-algebras is a locally finite variety, while results due to Blok and Pigozzi [34, Corollary 1.23], Cornish [67] and Blok and Raftery [39, p. 294] show pBCK is precisely the class of all  $\langle -, 0 \rangle$ -subreducts of dual Brouwerian semilattices. The following lemma is a variant on this last.

**Lemma 1.6.14.** [7, Lemma 3.2] If each two elements a, b from a non-empty subset B of a join semilattice A have a dual relative pseudocomplement a \* b that belongs to B, then  $\langle B; * \rangle$  is a positive implicative BCK-algebra.

Recall from Remark 1.6.7 that the  $\langle \div, 0 \rangle$ -terms  $x \div y^n$ ,  $n \in \omega$ , are defined

inductively by:

$$\begin{aligned} x \dot{-} y^0 &:= x \\ x \dot{-} y^{k+1} &:= (x \dot{-} y^k) \dot{-} y \qquad \text{for } k \ge 0. \end{aligned}$$

In [68] Cornish studied BCK-algebras satisfying:

$$x - y^{n+1} \approx x - y^n \tag{E}_n$$

where  $1 \leq n \in \omega$  as a natural generalisation of positive implicative BCKalgebras; the variety of positive implicative BCK-algebras is just the class of all BCK-algebras satisfying (E<sub>1</sub>). For each  $n \in \omega$ , the class e<sub>n</sub>BCK of BCK-algebras satisfying (E<sub>n</sub>) is a variety [68, Theorem 1.4], the members of which are known as *n*-potent BCK-algebras. Cornish has shown these varieties form a strictly increasing chain [70, Section 3.6]. Since their introduction the varieties e<sub>n</sub>BCK,  $n \in \omega$ , have been studied by a number of authors, including Cornish [70, Section 4], Blok and Raftery [38, 39] and Pałasiński [179], to whom the following theorem is collectively due.

**Theorem 1.6.15.** [70, Corollary 4.2]; [179]; [39, Theorem 4.2] For a variety  $\vee$  of BCK-algebras the following assertions hold:

 V has a commutative TD term iff V is a subvariety of e<sub>n</sub>BCK for some n ∈ ω. If V is a subvariety of e<sub>n</sub>BCK, n ∈ ω, then a commutative (but not regular) TD term for V is:

$$e(x, y, z) := \left(z - (x - y)^n\right) - (y - x)^n.$$

2. V has EDPC iff V is a subvariety of  $e_n BCK$  for some  $n \in \omega$ . For any algebra  $A \in e_n BCK$ ,  $n \in \omega$ , and  $a, b, c, d \in A$ ,

$$c \equiv d \left( \mod \Theta^{\mathbf{A}}(a, b) \right) \quad iff \quad \left( c(ab)^n \right) (ba)^n = \left( d(ab)^n \right) (ba)^n$$
$$iff \quad \left( (cd)(ab)^n \right) (ba^n) = \left( (dc)(ab)^n \right) (ba)^n.$$

**1.6.16.** Implicative BCK-Algebras. Let  $\mathbf{A} := \langle A; /, 0 \rangle$  be a BCK-algebra. If  $\mathbf{A} \models x/(y/x) \approx x$  then  $\mathbf{A}$  is called an *implicative* BCK-algebra. A well known result of Iséki [126, Theorems 9,10] asserts that a BCK-algebra is implicative iff it is both commutative and positive implicative, whence the class of implicative BCK-algebras is a variety. The following equational characterisation is due independently to Kalman [131] and Abbot [2].

**Theorem 1.6.17.** ([131, p. 402]; [2, Section 1]) An algebra  $\langle A; /, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is an implicative BCK-algebra iff it satisfies the following identities:

$$x/(y/x) \approx x \tag{1.35}$$

$$x/(x/y) \approx y/(y/x) \tag{1.36}$$

$$(x/y)/z \approx (x/z)/y \tag{1.37}$$

$$x/x \approx 0.$$

Thus the class iBCK of implicative BCK-algebras is a variety.

**Example 1.6.18.** Let  $C_n := \langle C; /, 0 \rangle$  be an algebra of type  $\langle 2, 0 \rangle$  with cardinality n + 1 and operation / defined by:

 $ab := egin{cases} a & ext{if } a 
eq b \ 0 & ext{otherwise} \end{cases}$ 

for any  $a, b \in C$ . Then  $C_n$  is an implicative BCK-algebra; we call  $C_n$  a flat implicative BCK-algebra on n+1 elements.

Because of (1.3)-(1.4) and (1.7) any two-element implicative BCK-algebra is flat and so may be identified with  $C_1$ . It is easy to see that  $C_1$  embeds into any non-trivial BCK-algebra, and hence that  $Q(C_1)$  is the smallest non-trivial subquasivariety of BCK-algebras. In fact  $Q(C_1)$  is the unique atom in the lattice of varieties of BCK-algebras; this is a consequence of the following result, due to Kalman, which will be needed in the sequel.

Theorem 1.6.19. [131, Lemma 2] Up to isomorphism,  $iBCK_{SI} = \{C_1\}$ .

Combining Theorem 1.6.19 with Birkhoff's subdirect representation theorem yields the following result, which is also due to Kalman [131] (for a proof of

(1.38)

Theorem 1.6.20 that does not assume the axiom of choice, see Cornish [66, Corollary 1.5]). We remark that it is this result that justifies the change in notation for implicative BCK difference from - to / encountered in this section and in the sequel.

**Theorem 1.6.20.** [131] The class iBCK is precisely the class of all  $\{/, 0\}$ -subreducts of Boolean algebras  $\langle A; \land, \lor, ', 0, 1 \rangle$ , where  $a/b := a \land b'$  for any  $a, b \in A$ .

Implicative BCK-algebras were first introduced by Monteiro [168] (in dually isomorphic form under the name *Tarski algebras*) and independently by Kalman in [131] (under the name *flocks*). Implicative BCK-algebras have been studied subsequently by a range of authors, including Abbott [2, 1] (in dually isomorphic form under the name *implication algebras*), Rasiowa [195, Sections II§5-7] (likewise), and Iséki and Tanaka [126]. The following representation theorem is due to Abbott [2, 1]. In the statement of the theorem, a *semi-Boolean algebra* is a locally Boolean meet semilattice: that is, a meet semilattice  $\langle A; \land, 0 \rangle$  in which for each  $a \in A$  the principal subalgebra (a] is a Boolean lattice under the semilattice partial ordering.

**Theorem 1.6.21.** [2, Theorem 6, Theorem 7] Every implicative BCK-algebra  $\langle A; /, 0 \rangle$  induces a semi-Boolean algebra  $\langle A; \cap, 0 \rangle$  upon defining  $a \cap b := a/(a/b)$  for any  $a, b \in A$ . Conversely, every semi-Boolean algebra  $\langle A; \cap, 0 \rangle$  determines an implicative BCK-algebra  $\langle A; /, 0 \rangle$  under the operation  $a/b := (a \cap b)'_{(a)}$  for any  $a, b \in A$ , where  $(a \cap b)'_{(a)}$  denotes the complement of  $a \cap b$  in the principal subalgebra (a] generated by a. Moreover, these correspondences are inverse to each other.

A BCK-algebra A is bounded if there exists  $1 \in A$  such that  $a \leq 1$  for any  $a \in A$ . As usual, by abuse of language and notation we will often confuse a bounded BCK-algebra A with its expansion to  $\langle A; /, 0, 1 \rangle$ , where 1 is a new nullary operation symbol adjoined to the language of A whose canonical interpretation on  $\langle A; /, 0, 1 \rangle$  is  $1 \in A$ .

Corollary 1.6.22. The underlying poset  $\langle A; \leq \rangle$  of a bounded implicative BCK-algebra  $\langle A; /, 0, 1 \rangle$  is a Boolean lattice. For any  $a, b \in A$ ,

 $a \wedge b = a \cap b$  $a \vee b = 1/((1/a) \cap (1/b)).$ 

**1.6.23.** BCK-[Semi]Lattices. An algebra  $\langle A; \land, \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is called a *lower BCK-semilattice* if: (i) the reduct  $\langle A; \div, 0 \rangle$  is a BCK-algebra; and (ii) the following conditions are satisfied with respect to the BCK partial order  $\leq$  for any  $a, b, c \in A$  [116, p. 840]:

$$a \wedge b \le a \tag{1.39}$$

$$a \wedge b \le b \quad . \tag{1.40}$$

 $c \le a \text{ and } c \le b \text{ imply } c \le a \land b.$  (1.41)

An upper BCK-semilattice is defined analogously as an algebra  $\langle A; \vee, \div, 0 \rangle$ of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \div, 0 \rangle$  is a BCK-algebra; and (ii) for any  $a, b \in A$ ,  $a \vee b$  is the least upper bound of the doubleton  $\{a, b\}$  with respect to the BCK partial order. A BCK-lattice is an algebra  $\langle A; \wedge, \vee, \div, 0 \rangle$ of type  $\langle 2, 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \wedge, \div, 0 \rangle$  is a lower BCKsemilattice; and (ii) the reduct  $\langle A; \vee, \div, 0 \rangle$  is an upper BCK-semilattice. The following characterisation of BCK-[semi]lattices is implicit in the proof of [116, Theorem 1].

**Lemma 1.6.24.** (cf. [116, Theorem 1]) An algebra  $\langle A; \land, \div, 0 \rangle$  [ $\langle A; \lor$ ,  $\div, 0 \rangle$ ] of type  $\langle 2, 2, 0 \rangle$  is a lower BCK-semilattice [upper BCK-semilattice] iff the following conditions hold:

- 1. The reduct  $\langle A; \wedge \rangle$  [ $\langle A; \vee \rangle$ ] is a meet semilattice [join semilattice];
- 2. The reduct  $\langle A; \div, 0 \rangle$  is a BCK-algebra;
- 3. The BCK partial order coincides with [dualises] the semilattice partial order.

An algebra  $\langle A; \land, \lor, \div, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  is a BCK-lattice iff the following conditions hold:

- 1'. The reduct  $\langle A; \land, \lor \rangle$  is a lattice;
- 2. The reduct  $\langle A; \div, 0 \rangle$  is a BCK-algebra;
- 3. The BCK partial order coincides with the lattice partial order.

Proof. To prove the lemma it is sufficient to prove the first statement in the context of lower BCK-semilattices. Let A be a lower BCK-semilattice and let  $a, b, c, d \in A$ . Clearly conditions (1) and (2) are satisfied. From (1.39) we have  $(a \div (a \div b)) \land b \leq ^{\langle A_i \div , 0 \rangle} a \div (a \div b)$ , while from  $a \div (a \div b) \leq ^{\langle A_i \div , 0 \rangle} a \div (a \div b)$ , while from  $a \div (a \div b) \leq ^{\langle A_i \div , 0 \rangle} a \div (a \div b)$ ,  $a \div (a \div b) \leq ^{\langle A_i \div , 0 \rangle} b$  and (1.41) we have  $a \div (a \div b) \leq ^{\langle A_i \div , 0 \rangle} (a \div (a \div b)) \land b$ . Thus  $(a \div (a \div b)) \land b = a \div (a \div b)$ . If  $c \leq ^{\langle A_i \div , 0 \rangle} d$  then  $c = c \div 0 = c \div (c \div d) = (c \div (c \div d)) \land d = (c \div 0) \land d = c \land d$ , so  $c \leq ^{\langle A_i \land \rangle} d$ . On the other hand, from (1.39) we have  $a \land b \leq ^{\langle A_i \div , 0 \rangle} b$ , and so  $(a \land b) \div b = 0$ . If  $c \leq ^{\langle A_i \land \rangle} d$  then  $0 = (c \land d) \div d = c \div d$  and so  $c \leq ^{\langle A_i \div , 0 \rangle} d$ . Thus (3) holds, and A satisfies (1)-(3). Since the converse is clear, the lemma is proved.

**Theorem 1.6.25.** [116, Theorem 1] The class IBS [uBS] of lower [upper] BCK-semilattices is a variety. Therefore the class BL of BCK-lattices is also a variety.

BCK-[semi]lattices were introduced by Idziak in [116]. They have since been studied by Idziak [115, 117], Raftery and Sturm [190], Kondo [141] and Ono and Komori [176] among others. Examples of BCK-[semi]lattices abound in the literature, and include dual Brouwerian semilattices (see [164] and [60] for details) and generalised Boolean algebras (see Corollary 3.3.56 in the sequel).

# 1.7 Ideals and Subtractive Varieties

The theory of ideals in universal algebra and the theory of subtractive varieties, as developed by Agliano, Ursini and others, are the major tools we employ in our study of pre-BCK-algebras. We summarise here the parts of these theories that we exploit in the sequel. Unless otherwise stated, throughout this section all classes of algebras considered are pointed; typically we always assume the existence of a constant term 0.

1.7.1. Ideal Terms. Let K be a class of algebras of the same similarity type. A term  $p(\vec{x}, \vec{y})$  in the language of K is a K-ideal term in  $\vec{y}$  (in symbols  $p(\vec{x}, \vec{y}) \in \mathrm{IT}_{\mathsf{K}}(\vec{y})$  if the identity  $p(\vec{x}, 0, \dots, 0) \approx 0$  holds in K. A non-empty subset I of  $\mathbf{A} \in \mathsf{K}$  is a K-ideal of A if for any  $p(\vec{x}, \vec{y}) \in \mathrm{IT}_{\mathsf{K}}(\vec{y})$  we have  $p^{\mathbf{A}}(\vec{a}, \vec{b}) \in I$  for  $\vec{a} \in A$  and  $\vec{b} \in I$  [104, p. 46]. The intersection of K-ideals is itself a K-ideal and one easily sees that the set  $I_{K}(A)$  of all K-ideals of A forms an algebraic lattice  $I_{K}(A)$  under inclusion [104, Lemma 1.2]. For any  $B \subseteq A$ , the ideal  $\langle B \rangle_{\mathbf{A}}^{\mathsf{K}}$  generated by B is the set  $\{p^{\mathbf{A}}(\vec{a}, \vec{b}) : p(\vec{x}, \vec{y}) \in \mathrm{IT}_{\mathsf{K}}(\vec{y}); \vec{a} \in$  $A, \vec{b} \in B$ . A K-ideal is compact when it is finitely generated; for B := $\{a_1,\ldots,a_n\}$ , the compact ideal  $\langle B \rangle_{\mathbf{A}}^{\mathsf{K}}$  is denoted  $\langle a_1,\ldots,a_n \rangle_{\mathbf{A}}^{\mathsf{K}}$ . By [11, p. 360] the set  $CI_{K}(\mathbf{A})$  of compact K-ideals of **A** forms a join subsemilattice  $CI_{K}(\mathbf{A})$ of  $I_{K}(A)$  under inclusion. A K-ideal is said to be *principal* when it is generated by a single element; the principal ideal  $\langle B \rangle_A^K$  generated by  $B := \{a\}$  is denoted  $\langle a \rangle_{\mathbf{A}}^{\mathsf{K}}$ . Clearly  $\langle 0 \rangle_{\mathbf{A}}^{\mathsf{K}} = \{0\}$ . When K is  $\{\mathbf{A}\}$  (or, equivalently, the variety  $\mathsf{V}(\mathbf{A})$ generated by A) then a K-ideal is simply called an *ideal* and all affixes and suffixes in sight are dropped, provided the context is clear. The set  $\{[0]_{\theta}:$  $\theta \in \text{Con } \mathbf{A}$  is denoted by N(A), and any element of N(A) is called a normal set; trivially  $N(A) \subseteq I(A) \subseteq I_{\kappa}(A)$ . Clearly N(A) inherits in a natural way the lattice structure of Con A: see [222, p. 205]. We say K [an algebra A] has normal ideals if  $I_{K}(A) = N(A)$  for all  $A \in K$  [if I(A) = N(A)]. If K has normal ideals, the set of all ideals [lattice of ideals] of any  $A \in K$  may be denoted simply by I(A) [I(A)] without any reference to K; we adopt this convention in the sequel. We say an algebra A [a class K] is *ideal simple* if the only ideals of A [of all  $A \in K$ ] are {0} and A. We say A is hereditarily ideal simple if every subalgebra of A is ideal simple; K is hereditarily ideal simple if every member of K is hereditarily ideal simple. Finally, an ideal I of A is said to be proper if  $I \neq A$ , and improper otherwise.

The study of K-ideals was initiated by Gumm and Ursini in [104], wherein they proposed the syntactic notion of ideal described above as an abstraction of the

75

familiar closure properties of 'ideals' arising in commonly occurring classes of algebras. Since their introduction, K-ideals in universal algebra have been studied by Agliano and Ursini in several contexts; see [8, 218, 219] for details.

Remark 1.7.2. Blok and Raftery [40] have recently proposed a still more general notion of ideal than that of Gumm and Ursini, applicable to general (not necessarily pointed) quasivarieties. Inspired by [31], Blok and Raftery view the closure systems of (K-) congruences of algebras in a quasivariety K as models of a certain deductive system  $S(K, \tau)$  'extracted' from the quasi-equational theory of K by means of a suitable translation  $\tau$  (for details, see [40, Section 5]). For  $A \in K$  a strong ideal of A is simply an  $S(K, \tau)$ -filter of A, namely a subset of A closed under the axioms and inference rules of  $S(K, \tau)$ . In [40, Section 7] Blok and Raftery compare their notion of ideal with that proposed by Gumm and Ursini, and show in [40, Theorem 7.4] that for subtractive varieties the two notions coincide. Because we restrict ourselves to the investigation of subtractive varieties in this thesis, we are free to work just with the syntactic notion of ideal proposed by Gumm and Ursini; this we (mostly) do in the sequel.

Application of Gumm and Ursini's theory of ideals is primarily directed in the literature towards ideal determined varieties. Recall from [104] that a variety V is *ideal determined* if for all  $\mathbf{A} \in V_{i}$  any V-ideal is the 0-class of exactly one congruence, or equivalently, if the map  $\theta \mapsto [0]_{\theta}$  ( $\theta \in \text{Con } \mathbf{A}$ ) is a lattice isomorphism from **Con A** onto  $\mathbf{I}(\mathbf{A})$ .

**Proposition 1.7.3.** [104, Corollary 1.9] A variety  $\vee$  with **0** is ideal determined iff it is **0**-regular and there exists a binary term s(x, y) in the language of  $\vee$  such that:

 $s(x,x) \approx \mathbf{0} \tag{1.42}$ 

$$s(x,0) \approx x. \tag{1.43}$$

1.7.4. Subtractive Varieties. A variety V is subtractive if there exists a binary term s(x, y) of V such that the identities (1.42) and (1.43) are satisfied in V; an algebra A is subtractive if V(A) is subtractive. A variety V is congruence 0-permutable if for any  $A \in V$  and  $\theta, \phi \in \text{Con } A$ ,  $[0]_{\theta \circ \phi} = [0]_{\phi \circ \theta}$ .

For a variety V, the following are equivalent [222, Proposition 1.2]: (i) V is subtractive; (ii) V is congruence 0-permutable; and (iii) for each  $A \in V$ , the map  $\theta \mapsto [0]_{\theta}$  ( $\theta \in \text{Con } A$ ) is a lattice epimorphism from Con A into  $I_V(A)$ .

**Proposition 1.7.5.** [222, Proposition 1.4] Let V be a subtractive variety and let  $A \in V$ . Then every  $I \in I_V(A)$  is a congruence class. That is,  $I_V(A) = N(A)$ , and so V has normal ideals.

Subtractive varieties were introduced by Ursini in [222] and have been systematically investigated by Agliano and Ursini [9, 10, 11, 222, 225] in a program strongly influenced by recent developments in universal algebra, including commutator theory and the theory of varieties with equationally definable principal congruences. Particularly important among subtractive varieties are those satisfying the identity:

$$s(0,x) \approx 0. \tag{1.44}$$

Indeed, recall from [81, Definition 2] that a variety V is congruence 0-distributive if for any  $\mathbf{A} \in \mathbb{V}$  and  $\theta, \phi, \psi \in \text{Con A}$ ,  $[0]_{(\theta \vee \phi) \wedge \psi} = [0]_{(\theta \wedge \psi) \vee (\phi \wedge \psi)}$ . A variety that is both congruence 0-permutable and congruence 0-distributive is said to be arithmetical at 0. A variety V is ideal distributive if  $\mathbf{I}_{\mathbf{V}}(\mathbf{A})$  is a distributive lattice for any  $\mathbf{A} \in \mathbb{V}$ . For a variety V, the following are equivalent [81, Theorem 4], [9, Proposition 4.1]: (i) there exists a binary term s(x, y) of V such that V satisfies (1.42)-(1.44); (ii) V is arithmetical at 0; and (iii) V is subtractive and ideal distributive.

1.7.6. Congruences of Subtractive Varieties. In general, a subtractive variety need not be ideal determined. Let A be a (subtractive) algebra and  $I \in I(A)$ . Let  $CON(I) := \{\theta \in Con A : [0]_{\theta} = I\}$  denote the set of congruences of A whose 0-classes coincide with I. Let  $I^{\delta} := \bigwedge CON(I) \ [I^{\epsilon} := \bigvee CON(I)]$  denote the least [greatest] congruence of A whose 0-class is I. If A is an algebra with normal ideals,  $CON(I) = \{I^{\delta}, I^{\epsilon}\}$  is an interval in Con A [10, Proposition 1.3]. Let  $()^{\delta} : I(A) \to Con A$  and  $()^{\epsilon} : I(A) \to Con A$  be the maps defined by  $I \mapsto I^{\delta}$  and  $I \mapsto I^{\epsilon}$  respectively. For  $\theta \in Con A$  let  $\theta_0$  [ $\theta_1$ ] denote the least [greatest] congruence  $([0]_{\theta})^{\delta}$  [ $([0]_{\theta})^{\epsilon}$ ] on A whose 0-class is

 $[0]_{\theta}$ . Let  $\hat{\theta} := \text{CON}([0]_{\theta})$ ; notice  $\hat{\theta} = [\theta_0, \theta_1]$ . Let ()<sub>0</sub> : Con A  $\rightarrow$  Con A and ()<sub>1</sub> : Con A  $\rightarrow$  Con A be the maps defined respectively by  $\theta \mapsto \theta_0$  and  $\theta \mapsto \theta_1$ .

A system of 0-terms (without parameters) for a class K of similar algebras is a set  $D := \{d_j(x, y) : j \in J\}$  of terms in the language of K such that:

1.  $\mathsf{K} \models d_j(x, x) \approx \mathbf{0}$  for all  $j \in J$ ;

2. For all  $A \in K$  and  $a \in A$ , if  $d_j^A(0, a) = 0$  for all  $j \in J$ , then a = 0.

Let A be an algebra and  $I \in I(A)$ . Define  $I^D \subseteq A \times A$  by  $(a, b) \in I^D$  iff for all  $j \in J$ ,  $d_j^A(a, b) \in I$ . For  $\theta \in Con A$  let  $\theta^D := [0]_{\theta}^D$ . For a class K of similar algebras, a set  $D := \{d_j(x, y) : j \in J\}$  of terms of K is a system of ideal congruence terms (without parameters) for K (in short, an *IC*-system (without parameters) for K) if  $\theta^D \in Con A$  and  $[0]_{\theta^D} = [0]_{\theta}$  for all  $A \in K$ . Observe that if K has normal ideals, D is an IC-system for K iff for all  $A \in K$  and  $I \in I(A)$ ,  $I^D \in CON(I)$ , or, equivalently,  $I^D \in Con A$  and  $[0]_{I^D} = I$ .

**Proposition 1.7.7.** [10, Proposition 3.8, Proposition 3.9] Let  $\vee$  be a variety with normal ideals and let D be a system of 0-terms for  $\vee$ . The following are equivalent:

- 1. For  $\mathbf{A} \in \mathsf{V}$ ,  $\langle 0 \rangle_{\mathbf{A}}^{D} \in \operatorname{Con} \mathbf{A}$ ;
- 2. D is an IC-system for V;
- 3. For  $\mathbf{A} \in \mathsf{V}$  and  $\theta \in \operatorname{Con} \mathbf{A}$ ,  $\theta^D = \theta_1$ ;
- 4. For  $\mathbf{A} \in \mathsf{V}$  and  $I \in I(\mathbf{A})$ ,  $I^D = I^{\epsilon}$ .

A variety is said to be *finitely congruential* if it has a finite IC-system without parameters; by [10, Theorem 3.10] a finitely congruential variety with normal ideals is subtractive. For subtractive varieties, finite congruentiality generalises point regularity: see Agliano and Ursini [10, Remark (4), p. 322].

**1.7.8.** Equationally Definable Principal Ideals. A class K of algebras of a given similarity type has Equationally Definable Principal K-Ideals (briefly,

1.7. Ideals and Subtractive Varieties

EDPI) if there are terms  $p_i(x, y)$ ,  $q_i(x, y)$ , i = 1, ..., n in the language of K such that for any  $A \in K$  and  $a, b \in A$ :

$$a \in \langle b \rangle_{\mathbf{A}}$$
 iff  $p_i^{\mathbf{A}}(a, b) = q_i^{\mathbf{A}}(a, b)$   $i = 1, \dots, n$ .

Equationally definable principal ideals were introduced by Agliano and Ursini in [9] as the ideal-theoretic analogue of equationally definable principal congruences. Subtractive varieties with EDPI have been investigated at length by Agliano and Ursini [9, 11], to whom the following result is due.

**Theorem 1.7.9.** [6, Theorem 4.1] For a subtractive variety V, the following are equivalent:

- 1. V has EDPI;
- 2. There is a binary term p(x, y) of V such that:

$$a \in \langle b \rangle_{\mathbf{A}}$$
 iff  $p^{\mathbf{A}}(a, b) = 0$ 

for any  $A \in V$ ;

3. The join semilattice  $\langle CI(\mathbf{A}); \vee, \langle 0 \rangle_{\mathbf{A}} \rangle$  of compact ideals is dually relatively pseudocomplemented for any  $\mathbf{A} \in V$ .

Moreover, if V has EDPI then there exists a binary term x - y of V witnessing both subtractivity and EDPI in the sense of (2) above. That is, there exists a binary term x - y of V such that for any  $A \in V$  and  $a, b \in A$ ,

$$a \stackrel{\cdot}{\rightarrow} {}^{\mathbf{A}} a = 0$$
  
$$a \stackrel{\cdot}{\rightarrow} {}^{\mathbf{A}} 0 = a$$
  
$$a \in \langle b \rangle_{\mathbf{A}} \quad iff \quad a \stackrel{\cdot}{\rightarrow} {}^{\mathbf{A}} b = 0.$$

**Proposition 1.7.10.** ([11, Theorem 3.1]; [4, Corollary 2]) Let  $\vee$  be a subtractive variety. If  $\vee$  has EDPC then  $\vee$  has EDPI. Conversely, if  $\vee$  is ideal determined and has EDPI, then  $\vee$  has EDPC and the map  $\theta \mapsto [0]_{\theta}$  ( $\theta \in \text{Con } \mathbf{A}$ ) is a dual Brouwerian semilattice isomorphism from Cp A onto CI(A) for any  $\mathbf{A} \in \vee$ . **Example 1.7.11.** [11, Example 3.7] An algebra  $\langle A; \rightarrow, 1 \rangle$  of type  $\langle 2, 0 \rangle$  is called a *MINI-algebra* if the following identities are satisfied:

$$x \to \mathbf{1} \approx \mathbf{1} \tag{1.45}$$

$$\mathbf{1} \to x \approx x \tag{1.46}$$

$$(x \to (y \to z)) \to ((x \to y) \to (x \to z)) \approx 1$$
 (1.47)

$$x \to (y \to x) \approx \mathbf{1}.\tag{1.48}$$

Hilbert algebras are precisely those MINI-algebras for which the quasi-identity:

$$x \to y \approx \mathbf{1} \& y \to x \approx \mathbf{1} \supset x \approx y \tag{1.49}$$

holds. Hence the variety of MINI-algebras is a natural generalisation of the variety of Hilbert algebras to the subtractive but not 1-regular case. In consequence, the class of MINI-algebras provides a natural example of a subtractive variety with EDPI (that does not have EDPC); indeed, the term  $y \to x$  witnesses *both* subtractivity and EDPI in the variety of MINI-algebras in the sense of Theorem 1.7.9 above.

**Proposition 1.7.12.** [11, Corollary 3.6] For a variety  $\forall$  with language  $\langle \div, \mathbf{0} \rangle$  of type  $\langle 2, 0 \rangle$  the following are equivalent:

1. For all  $A \in V$  and  $a, b \in A$ ,

 $a \in \langle b \rangle_{\mathbf{A}}$  iff  $a \stackrel{\cdot}{-} b = 0$ 

 $a \stackrel{\cdot}{-} b = 0$  and  $b \stackrel{\cdot}{-} a = 0$  implies a = b;

2.  $\lor$  is a variety of positive implicative BCK-algebras.

Let V be a variety. A meet generator term for V is a binary term  $\sqcap$  in the language of V such that for any  $A \in V$  and  $a, b \in A$  [11, p. 378],

 $\langle a \rangle_{\mathbf{A}} \wedge^{\mathbf{I}(\mathbf{A})} \langle b \rangle_{\mathbf{A}} = \langle a \sqcap^{\mathbf{A}} b \rangle_{\mathbf{A}}.$ 

A join generator term for V is a binary term  $\sqcup$  of V such that for any  $A \in V$ 

and  $a, b \in A$  [11, p. 379],

 $\langle a \rangle_{\mathbf{A}} \vee^{\mathbf{I}(\mathbf{A})} \langle b \rangle_{\mathbf{A}} = \langle a \sqcup^{\mathbf{A}} b \rangle_{\mathbf{A}}.$ 

Subtractive varieties with a meet [join] generator term have been studied by Agliano and Ursini in [11, Section 4, Section 5]. For a subtractive variety V, the following are equivalent: (i) V is ideal distributive and the meet of two principal ideals is principal; (ii) V has a meet generator term [11, p. 378; Theorem 4.2]. Concerning join generator terms, the following proposition obtains for arbitrary (that is, not necessarily subtractive) varieties.

**Proposition 1.7.13.** [11, Proposition 4.3] For a variety V, the following are equivalent:

- 1. For any  $A \in V$ , the join of two principal ideals is principal;
- 2. For any  $A \in V$ , every compact ideal is principal;
- 3. There is a binary term  $\sqcup$  and two ternary terms r(x, y, z) and t(x, y, z) of  $\vee$  such that the identities:

 $0 \sqcup 0 \approx 0$   $r(x, y, 0) \approx 0$   $t(x, y, 0) \approx 0$   $r(x, y, x \sqcup y) \approx x$  $t(x, y, x \sqcup y) \approx y$ 

hold in V;

4. V has a join generator term.

If any of (1)–(4) hold, the binary term  $\sqcup$  of (3) is a join generator term for  $\vee$ .

Let K be a class of similar algebras. For any  $A \in K$  let  $\approx_A$  be the relation defined on A by [6, p. 2]:

 $a \approx_{\mathbf{A}} b$  iff  $\Theta^{\mathbf{A}}(0, a) = \Theta^{\mathbf{A}}(0, b)$ 

for any  $a, b \in A$ . If K has normal ideals, then  $a \approx_{\mathbf{A}} b$  iff  $N_{\mathbf{A}}(a) = N_{\mathbf{A}}(b)$  (by Agliano [7, p. 5]), where  $N_{\mathbf{A}}(a), N_{\mathbf{A}}(b)$  denote the normal sets generated by  $\{a\}, \{b\}$  respectively. Therefore if K is a subtractive variety then  $a \approx_{\mathbf{A}} b$  iff  $\langle a \rangle_{\mathbf{A}} = \langle b \rangle_{\mathbf{A}}$ .

**Lemma 1.7.14.** Let  $\vee$  be a subtractive variety with EDPI and let  $x \div y$  be a binary term witnessing both subtractivity and EDPI for  $\vee$  in the sense of Theorem 1.7.9. For any  $\mathbf{A} \in \vee$ ,

$$\approx_{\mathbf{A}} = \{(a, b) : a \stackrel{\cdot}{-} {}^{\mathbf{A}}b = 0 = b \stackrel{\cdot}{-} {}^{\mathbf{A}}a\}.$$

*Proof.* Let V and  $\dot{-}$  be as in the statement of the lemma. Let  $A \in V$ . For any  $a, b \in A$ , we have that  $a \approx_{\mathbf{A}} b$  iff  $\langle a \rangle_{\mathbf{A}} = \langle b \rangle_{\mathbf{A}}$  iff  $a \in \langle b \rangle_{\mathbf{A}}, b \in \langle a \rangle_{\mathbf{A}}$  iff  $a \dot{-}^{\mathbf{A}} b = 0 = b \dot{-}^{\mathbf{A}} a$  (by EDPI).

Because of Lemma 1.7.14, the following theorem may be inferred from results due to Agliano and Ursini [11].

**Theorem 1.7.15.** Let V be a subtractive variety with EDPI. Let x - y witness both subtractivity and EDPI for V in the sense of Theorem 1.7.9 and let  $A \in V$ . The following assertions hold:

- 1. [11, Theorem 3.4(2)]  $\langle PI(\mathbf{A}); *, \langle 0 \rangle_{\mathbf{A}} \rangle$  is a positive implicative BCKalgebra isomorphic with  $\langle A; \div, 0 \rangle / \approx_{\mathbf{A}};$
- [11, p. 383] If V has a meet generator term x ⊓ y, the compact ideals of A are closed under intersection. Thus (CI(A); : 1, V, \*, (0)<sub>A</sub>) is a dually Brouwerian lattice and (PI(A); ∩, \*, (0)<sub>A</sub>) is a (∩, \*)-subreduct of (CI(A); ∩, ∨, \*, (0)<sub>A</sub>) isomorphic with (A; ⊓, ∸, 0)/≈<sub>A</sub>;
- 3. [11, Theorem 5.1(2)] If V has a join generator term  $x \sqcup y$ , then  $\langle PI(\mathbf{A}); \lor , *, \langle 0 \rangle_{\mathbf{A}} \rangle$  is a dual Brouwerian semilattice isomorphic with  $\langle A; \sqcup, -, 0 \rangle / \approx_{\mathbf{A}};$
- 4. [11, Theorem 5.6(2)] If V has both a meet and join generator term then ⟨PI(A); ∩, ∨, \*, ⟨0⟩<sub>A</sub>⟩ is a dually Brouwerian lattice isomorphic with ⟨A; ⊓, ⊔, ∸, 0⟩/≈<sub>A</sub>.

Let K be a class of similar algebras. For any  $A \in K$ , the relation  $\approx_A$  is an equivalence relation. If  $\approx_A = \omega_A$  for any  $A \in K$ , K is said to be congruence orderable. A point-regular congruence orderable variety is Fregean. Congruence orderable varieties were introduced by Büchi and Owens in [49] and have been studied by Idziak, Somczyńska and Wroński [118], Pigozzi [181] and Agliano [7], to whom the following proposition is due.

**Proposition 1.7.16.** [7, Theorem 2.1] Let V be a subtractive variety and let s(x, y) witness subtractivity for V. If V is congruence orderable then V is 0-regular and the terms  $d_1(x, y) := s(x, y)$  and  $d_2(x, y) := s(y, x)$  witness 0regularity for V in the sense of Proposition 1.2.6. Thus a congruence orderable subtractive variety is Fregean.

Let K be a class of similar algebras. If  $\approx_A \in \text{Con } A$  for any  $A \in K$  then K is called *weakly congruence orderable* or *congruence quasi-orderable*. Weak congruence orderability was introduced by Agliano in [6] as a weakening of the concept of congruence orderability. For subtractive varieties with EDPI, weak congruence orderability has been studied by Agliano and Ursini [11] and Agliano [6], to whom the following result is due.

Lemma 1.7.17. [6, Corollary 2.5] Let A be a subtractive algebra. Then A is weakly congruence orderable iff for any binary term s(x, y) witnessing sub-tractivity for A one has:

 $\approx_{\mathbf{A}} = \{(a, b) : s^{\mathbf{A}}(a, b) = 0 = s^{\mathbf{A}}(b, a)\} = \langle 0 \rangle_{\mathbf{A}}^{\epsilon}.$ 

**Proposition 1.7.18.** Let  $\vee$  be a subtractive variety and let s(x, y) be a term witnessing the subtractivity of  $\vee$ . Then  $\vee$  is congruence orderable iff  $\vee$  is weakly congruence orderable and the binary terms  $d_1(x, y) := s(x, y), d_2(x, y) :=$ s(y, x) witness the 0-regularity of  $\vee$  in the sense of Proposition 1.2.6.

*Proof.* Let V be a subtractive variety and let s(x, y) be a term witnessing the subtractivity of V. Suppose V is congruence orderable. Trivially V is weakly congruence orderable. Moreover, by Proposition 1.7.16 the terms  $d_1(x, y) := s(x, y), d_2(x, y) := s(y, x)$  witness the 0-regularity of V in the

### 1.7. Ideals and Subtractive Varieties

sense of Proposition 1.2.6. Conversely, suppose V is weakly congruence orderable and moreover that the binary terms  $d_1(x, y) := s(x, y), d_2(x, y) := s(y, x)$ witness the **0**-regularity of V. Let  $\mathbf{A} \in V$  and  $a, b \in A$ . By Lemma 1.7.17,  $a \approx_{\mathbf{A}} b$  implies  $s^{\mathbf{A}}(a, b) = 0 = s^{\mathbf{A}}(b, a)$ , whence  $d_1^{\mathbf{A}}(a, b) = 0 = d_2^{\mathbf{A}}(a, b)$ , whence a = b by Proposition 1.2.6. Thus V is congruence orderable.

**Proposition 1.7.19.** [6, Theorem 4.2] Let V be a weakly congruence orderable subtractive variety with EDPI. Then  $V_{\epsilon}$  is a congruence orderable subtractive variety with EDPC.

1.7.20. Binary and Dual Binary Discriminator Varieties. Let A be a set. For a fixed but arbitrary  $0 \in A$  the binary discriminator and dual binary discriminator on A are the functions  $b : A^2 \to A$  and  $h : A^2 \to A$  defined respectively by [58, Section 2]:

$$b(a,c) := \begin{cases} a & \text{if } c = 0 \\ 0 & \text{otherwise} \end{cases} \quad \text{and} \quad h(a,c) := \begin{cases} 0 & \text{if } c = 0 \\ a & \text{otherwise} \end{cases}$$

for any  $a, c \in A$ ; the element  $0 \in A$  is called the discriminating element. An algebra A with 0 is called a binary discriminator algebra [dual binary discriminator algebra] if there is a binary term b [a binary term h] of A whose canonical interpretation on A is the binary discriminator [dual binary discriminator] with discriminating element  $0^A$ . A variety V with 0 is said to be a binary discriminator variety [dual binary discriminator variety] if there is a binary term b of V [h of V] and a subclass K of V such that  $b^A$  is the binary discriminator [ $h^A$  is the dual binary discriminator] with discriminating element  $0^A$  on each  $A \in K$  and V = V(K); b [h] is called a binary discriminator variety is a dual binary discriminator variety, since  $h(x, y) \approx b(x, b(x, y))$ ; conversely, a dual binary discriminator variety is a binary discriminator variety iff it is subtractive [58, Theorem 2.1(1)].

**Theorem 1.7.21.** [58, Theorem 5.1] For a variety V the following are equivalent:

### 1.8. Algebraisable and Assertional Logics

- 1. V is a binary discriminator variety;
- 2. The following assertions hold for V and a binary term b(x, y) of V:
  - (a)  $\vee$  satisfies the identities:

 $b(x, 0) \approx x,$   $b(0, x) \approx x,$   $b(x, x) \approx 0,$  $b(x, b(y, x)) \approx x;$ 

(b)  $\vee$  satisfies the identity:

$$b(f(x_1,\ldots,x_n),y)\approx b(f(b(x_1,y),\ldots,b(x_n,y)),y)$$

for every n-ary operation symbol f in the type of V;

- (c) V is generated by a class  $K \subseteq V$  whose members are ideal simple;
- 3. There exists a binary term b(x, y) of V satisfying (2)(a)-(2)(b) above and V is generated by a class  $K \subseteq V$  whose algebras have no proper congruence kernels.

The binary discriminator and dual binary discriminator were introduced by Chajda, Halaš and Rosenberg in a 1999 paper [58] in an attempt to generalise the ternary discriminator and dual ternary discriminator to varieties exhibiting congruence permutability and congruence distributivity only locally at **0** respectively. For a brief discussion contrasting the binary discriminator with the ternary discriminator, see [58, p. 242, pp. 247–248, p. 249].

# **1.8** Algebraisable and Assertional Logics

In [31] Blok and Pigozzi introduced an abstract notion of algebraisability based on a generalisation of the classical Lindenbaum-Tarski process in an attempt to formalise the precise connection between EDPC and the deduction theorem [182, p. 125]. Since the publication of the seminal monograph [31] algebraic logic has been intensively developed by Blok, Czelakowski, Herrmann, Pigozzi and others. In this section we briefly review Blok and Pigozzi's theory of algebraisable logics, and also consider some recent developments in algebraic logic concerning the assertional logics of pointed quasivarieties.

1.8.1. Algebraisable Deductive Systems. Let K be a class of algebras of type  $\mathcal{L}$  and let S be a deductive system of the same type. K is called an *algebraic semantics* for S if  $\vdash_{S}$  can be interpreted in  $\models_{K}$  in the following sense: there exist finite families  $\{\delta_{1}, \ldots, \delta_{r}\}$  and  $\{\varepsilon_{1}, \ldots, \varepsilon_{r}\}$  of equations in one variable such that for all  $\Gamma \cup \{\varphi, \psi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$  and  $l = 1, \ldots, r$ ,

$$\Gamma \vdash_{S} \varphi \quad \text{iff} \quad \left\{ \delta_{t}(\chi) \approx \varepsilon_{t}(\chi) : \chi \in \Gamma; \ t = 1, \dots, r \right\} \models_{\mathsf{K}} \delta_{l}(\varphi) \approx \varepsilon_{l}(\varphi).$$
(1.50)

K is said to be *equivalent* to S if  $\models_{\mathsf{K}}$  can be interpreted in S in the following sense: there exists a finite system  $\{\Delta_1, \ldots, \Delta_m\}$  of formulas in two variables such that for all  $\Gamma \cup \{\varphi, \psi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ ,

$$\varphi \approx \psi = \models_{\mathsf{K}} \left\{ \delta_i(\varphi \Delta_i \psi) \approx \varepsilon_i(\varphi \Delta_i \psi) : i = 1, \dots, m; \ t = 1, \dots, r. \right\}$$
(1.51)

The equations  $\delta_t \approx \varepsilon_t$ , t = 1, ..., r, are called the *defining equations* for S and K while the family  $\{\Delta_1, ..., \Delta_m\}$  of composite binary connectives is called a system of *equivalence formulas* for S and K. A deductive system is said to be *algebraisable* if it has an equivalent algebraic semantics. Suppose S is algebraisable with equivalent algebraic semantics K. Because of [31, Corollary 2.11], K may be identified with the quasivariety  $\mathbf{Q}(K)$  it generates; that is, K is an *equivalent quasivariety semantics*. If K is a variety then K is an *equivalent variety semantics*, and in this case S is said to be *strongly algebraisable*. The following intrinsic characterisation of algebraisable deductive systems is due to Blok and Pigozzi [31].

**Theorem 1.8.2.** [31, Theorem 4.7] A deductive system S is algebraisable iff. there exists a finite family  $\{\Delta_1, \ldots, \Delta_m\}$  of formulas in two variables and finite families  $\{\delta_1, \ldots, \delta_r\}$  and  $\{\varepsilon_1, \ldots, \varepsilon_r\}$  of equations in a single variable such that for any connective  $\varpi$  (of rank n) and formulas  $\varphi_1, \ldots, \varphi_n, \psi_1, \ldots, \psi_n, \varphi, \psi$ ,  $\chi$  the following five conditions hold for  $j = 1, \ldots, m$ :

- 1.  $\vdash_{\mathbf{S}} \varphi \Delta_{\mathbf{j}} \varphi;$
- 2.  $\{\varphi \Delta_i \psi : i = 1, \ldots, m\} \vdash_{\mathbb{S}} \psi \Delta_j \varphi;$
- 3.  $\{\varphi \Delta_i \psi : i = 1, \ldots, m\} \cup \{\psi \Delta_i \chi : i = 1, \ldots, m\} \vdash_{\mathbb{S}} \Delta_j(\varphi, \chi);$
- 4.  $\{\varphi_k \Delta_i \psi_k : i = 1, \ldots, m; k = 1, \ldots, n\} \vdash_{\mathbb{S}} \varpi(\varphi_1, \ldots, \varphi_n) \Delta_j \varpi(\psi_1, \ldots, \psi_n);$

5. 
$$\chi \dashv _{\mathbb{S}} \{ \delta_t(\chi) \Delta_i \varepsilon_t(\chi) : i = 1, \ldots, m; t = 1, \ldots, r \}.$$

In this event  $\{\Delta_1, \ldots, \Delta_m\}$  and  $\delta_t \approx \varepsilon_t$ ,  $t = 1, \ldots, r$  are systems of equivalence formulas and defining equations for S.

It is possible for distinct algebraisable deductive systems to have the same equivalent algebraic semantics: see Blok and Pigozzi [31, Chapter 5.2.4]. On the other hand, the equivalent algebraic semantics associated with a given algebraisable deductive system is unique [31, Theorem 2.15] and is determined by the algorithm of the following theorem.

**Theorem 1.8.3.** [31, Theorem 2.17] Let S be a deductive system given by a set of axioms Ax and a set of inference rules Ir. Assume S is algebraisable with equivalence formulas  $\{\Delta_1, \ldots, \Delta_m\}$  and defining equations  $\delta_t \approx \varepsilon_t$ ,  $t = 1, \ldots, r$ . Then the unique equivalent quasivariety semantics for S is axiomatised by the identities:

1.  $\delta_{t}(\varphi) \approx \varepsilon_{t}(\varphi), \quad t = 1, \dots, r,$ for each  $\varphi \in Ax$ ;

2. 
$$\delta_t(x\Delta_i x) \approx \varepsilon(x\Delta_i x), \quad i=1,\ldots,m; \ t=1,\ldots,r;$$

together with the following quasi-identities:

3.  $\&_{u=1}^{n}\&_{t=1}^{r}\delta_{t}(\chi_{u}) \approx \varepsilon_{t}(\chi_{u}) \supset \delta_{l}(\varphi) \approx \varepsilon_{l}(\varphi), \quad l = 1, ..., r$ for each  $\langle \{\chi_{1}, ..., \chi_{n}\}, \varphi \rangle \in Ir;$ 

4. 
$$\&_{i=1}^{m} \&_{t=1}^{r} \delta_{t}(x \Delta_{i} y) \approx \varepsilon_{t}(x \Delta_{i} y) \supset x \approx y.$$

**Example 1.8.4.** Let  $\mathcal{L} := \{\rightarrow\}$  be a language of type  $\langle 2 \rangle$ . *BCK logic* is the deductive system  $\mathbb{BCK} := \langle \mathcal{L}, \vdash_{\mathbb{BCK}} \rangle$  defined by the following axioms and inference rule [28, Lecture 6, Section 2.2]:

$$(p \to q) \to ((q \to r) \to (p \to r))$$
 (B)

$$(p \to (q \to r)) \to (q \to (p \to r))$$
 (C)

$$p \to (q \to p)$$
 (K)

$$p, p \to q \vdash_{\text{BCK}} q.$$
 (BCK-MP)

BCK logic has been extensively studied in the literature: see [80, Section 4] for a survey and references. Results due to Blok and Pigozzi [31, Theorem 5.10, Theorem 5.11] show BCK is algebraisable; its equivalent algebraic semantics is termwise definitionally equivalent to the quasivariety BCK<sup>D</sup> of dual BCKalgebras introduced by Blok and Pigozzi in [32, Example 7.3] and axiomatised by the following set of identities and quasi-identities:

$$(x \to y) \to ((y \to z) \to (x \to z)) \approx 1$$
$$x \to ((x \to y) \to y) \approx 1$$
$$x \to x \approx 1$$
$$x \to 1 \approx 1$$

 $x \to y \approx \mathbf{1} \& y \to x \approx \mathbf{1} \supset x \approx y.$ 

Clearly  $BCK^D$  is itself termwise definitionally equivalent to (in fact, is dually isomorphic to) the quasivariety of BCK-algebras—for details, see [32, Example 7.3].

Following the publication of the seminal monograph [31] algebraisable logics have been the object of intense investigation; for a (partial) survey of the literature, see the tutorials of Blok and Pigozzi [36] and Blok and Jonsson [28].

1.8.5. Matrix Semantics. An  $\mathcal{L}$ -matrix (or simply a matrix when  $\mathcal{L}$  is understood) is a tuple  $\langle \mathbf{A}, F \rangle$ , where  $\mathbf{A}$  is an algebra of type  $\mathcal{L}$  and F is a

subset of A. For any  $\mathcal{L}$ -matrix  $\langle \mathbf{A}; F \rangle$  and  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ , let  $\models_{\langle \mathbf{A}; F \rangle}$ :  $\mathbb{P}(\operatorname{Fm}_{\mathcal{L}}) \to \operatorname{Fm}_{\mathcal{L}}$  be the relation defined by  $\Gamma \models_{\langle \mathbf{A}; F \rangle} \varphi$  if:

 $h(\psi) \in F$  for every  $\psi \in \Gamma$  implies  $h(\varphi) \in F$ , for every  $h : \mathbf{Fm}_{\mathcal{L}} \to \mathbf{A}$ .

If M is a class of  $\mathcal{L}$ -matrices, then  $\Gamma \models_{\mathsf{M}} \varphi$  if  $\Gamma \models_{\langle \mathbf{A}; F \rangle} \varphi$  for every  $\langle \mathbf{A}; F \rangle \in \mathsf{M}$ .

Let S be a deductive system of type  $\mathcal{L}$ , A be an algebra of the same type and let  $F \subseteq A$ . The subset F of A is called an S-filter of A, or simply a filter when S is understood, if  $\Gamma \vdash_{S} \varphi$  implies  $\Gamma \models_{\langle A; F \rangle} \varphi$  for all  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ ; the set of all S-filters of A is denoted Fis A. If F is an S-filter, then the  $\mathcal{L}$ matrix  $\langle A; F \rangle$  is called a matrix model of S; the class of all matrix models of S is denoted MatS. A congruence  $\theta$  on A is said to be compatible with F if  $a \in F$  and  $(a, b) \in \theta$  implies  $b \in F$  [31, Section 1.4]; the largest congruence on A compatible with F is called the Leibniz congruence on A over F and is denoted  $\Omega_A(F)$  [31, Theorem 1.5]. The natural map  $\Omega_A : \operatorname{Fi}_S A \to \operatorname{Con} A$ defined by  $F \mapsto \Omega_A(F)$  is called the Leibniz operator on A. An S-matrix is said to be reduced if  $\Omega_A(F) = \omega_A$ , and the class of all reduced S-matrices is denoted Mat\*S.

Theorem 1.8.6 (Reduced Matrix Completeness Theorem). [36, Theorem 3.5] Let S be a deductive system and let Mat<sup>\*</sup> S be the class of all reduced S-matrices. For all  $\Gamma \cup \{\varphi\}$ ,

 $\Gamma \vdash_{\mathbf{S}} \varphi \quad iff \quad \Gamma \models_{\mathsf{Mat}^* \mathbf{S}} \varphi.$ 

ないではないので、そのできた。

For a deductive system S over a language  $\mathcal{L}$ , a class M of  $\mathcal{L}$ -matrices is said to be a matrix semantics of S if, for all  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}}, \Gamma \vdash_{\mathbb{S}} \varphi$  iff  $\Gamma \models_{\mathsf{M}} \varphi$ . For a deductive system S, a quasivariety K and a system  $\delta_t \approx \varepsilon_t, t = 1, \ldots, r$ of equations in one variable the following are equivalent [31, Theorem 2.4]: (i) the class of matrices  $\{\langle \mathbf{A}, \{F_{\mathbf{A}}^{\delta \approx \varepsilon}\} \rangle : \mathbf{A} \in \mathsf{K}\}$ , where  $F_{\mathbf{A}}^{\delta \approx \varepsilon} = \{a \in A : \delta_t^{\mathbf{A}}(a) = \varepsilon_t^{\mathbf{A}}(a), t = 1, \ldots, r\}$ , is a matrix semantics for S; and (ii) K is an algebraic semantics for S with defining equations  $\delta_t \approx \varepsilon_t, t = 1, \ldots, r$ .

**Theorem 1.8.7.** [31, Corollary 5.3] Let S be an algebraisable deductive system. Let K be the equivalent quasivariety semantics of S and let Mat<sup>\*</sup> S be

the class of all reduced S-matrices. Then K is the class of all algebra reducts of  $Mat^*S$ , viz.:

$$\mathsf{K} = \{ \mathbf{A} : \langle \mathbf{A}, F \rangle \in \mathsf{Mat}^* \, \mathbb{S} \text{ for some } \mathbb{S} \text{-filter } F \text{ of } \mathbf{A} \}.$$

**Lemma 1.8.8.** Let  $S_1$  and  $S_2$  be deductive systems over the same language  $\mathcal{L}$ . If  $S_1$  and  $S_2$  are algebraisable with the same defining equations and the same equivalent algebraic semantics then they coincide.

*Proof.* Let  $S_1$  and  $S_2$  both be algebraisable with equivalent quasivariety semantics K and defining equations  $\delta_t \approx \varepsilon_t$ , t = 1, ..., r. By previous remarks and Theorem 1.8.7,  $Mat^*S_1 = \{ \langle \mathbf{A}, \{a \in A : \delta_t^{\mathbf{A}}(a) = \varepsilon_t^{\mathbf{A}}(a), t = 1, ..., r \} \rangle : \mathbf{A} \in \mathsf{K} \} = Mat^*S_2$ . Hence  $S_1$  and  $S_2$  coincide by Theorem 1.8.6.

1.8.9. Assertional Logics of Pointed Quasivarieties. For a quasivariety K of algebras with 1 over a language  $\mathcal{L}$ , the assertional logic of K, in symbols S(K, 1), is the closure operator  $\vdash_{S(K,1)}$  defined by the class of matrices  $M(K, 1) := \{ \langle \mathbf{A}, \{\mathbf{1}^{\mathbf{A}}\} \rangle : \mathbf{A} \in K \}$  in the sense that:

 $\Gamma \vdash_{\mathbb{S}(\mathsf{K},\mathbf{1})} \varphi \quad \text{iff} \quad \Gamma \models_{\mathsf{M}(\mathsf{K},\mathbf{1})} \varphi$ 

for any  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ . Because K is a quasivariety,  $\vdash_{\mathfrak{S}(\mathsf{K},\mathbf{1})}$  is finitary and structural, and hence is a deductive system in the sense of this thesis. Since  $\mathsf{M}(\mathsf{K},\mathbf{1})$  is a matrix semantics for  $\mathfrak{S}(\mathsf{K},\mathbf{1})$  by definition, the assertional logic  $\mathfrak{S}(\mathsf{K},\mathbf{1})$  may be defined equivalently by specifying that, for any  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ ,

 $\Gamma \vdash_{\mathbb{S}(\mathsf{K},\mathbf{1})} \varphi \quad \text{iff} \quad \{\psi \approx \mathbf{1} : \psi \in \Gamma\} \models_{\mathsf{K}} \varphi \approx \mathbf{1}.$ 

The entailment  $\{\psi \approx \mathbf{1} : \psi \in \Gamma\} \models_{\mathsf{K}} \varphi \approx \mathbf{1}$  is itself equivalent to the existence of some finite  $\Delta \subseteq \Gamma$  such that:

$$\mathsf{K} \models iggledge_{\psi \in arDelta} \psi(ec p) pprox \mathbf{1} \supset arphi(ec p) pprox \mathbf{1}$$

whence it is a harmless notational convenience to assume, for any entailment  $\Delta \vdash_{S(K,1)} \varphi$ , that  $\Delta \cup \{\varphi\}$  is a *finite* set of terms, all of the form  $\chi(\vec{p})$ , where  $\vec{p}$  is

understood as a *finite* sequence of variables, including all that occur in  $\Delta \cup \{\varphi\}$ . See Raftery and van Alten [193, Section 4.1].

Example 1.8.10. Let  $M(BCK^{D}, 1)$  denote the class of matrices of the form:

 $\langle \mathbf{A}, \{\mathbf{1}^{\mathbf{A}}\} : \mathbf{A} \in \mathsf{BCK}^D \rangle$ 

where  $BCK^{D}$  is the class of dual BCK-algebras (recall Example 1.8.4). By an argument analogous to that of [36, p. 24]  $M(BCK^{D}, 1)$  is precisely the class of reduced BCK-matrices of BCK. By the reduced matrix completeness theorem it follows that the deductive system  $S(BCK^{D}, 1)$  coincides with BCK.

Assertional logics of pointed classes were introduced by Pigozzi in [181, Section 2] (but see also Curry [75, pp. 64 ff.]). For recent work concerning assertional logics of pointed classes, see Blok and Raftery [41], [40, Section 6], Czelakowski and Jansana [77, Section 6], Czelakowski and Pigozzi [78, Section 2.1], Raftery and Barbour [16, Section 2.3] and Raftery and van Alten [193, Section 4.1]. (See also Ursini [224, Section 2] for related work.) For varieties with normal ideals, the associated assertional logics have been investigated by Agliano and Ursini in [10, pp. 314 ff.], wherein the following lemma is stated without proof. The proof we give below is implicit in Blok and Raftery [40, Lemma 5.1, Section 7]; see also Blok and Raftery [40, Theorem 7.4, Corollary 7.5].

Lemma 1.8.11. Let V be a variety with normal ideals and let S(V, 1) be the assertional logic of V. For any  $A \in V$ , the V-ideals of A coincide with the S(V, 1)-filters of S(V, 1).

*Proof.* Let V, S(V, 1) and A be as in the statement of the lemma. By [40, Lemma 5.1] the 1-class of any congruence of A is a S(V, 1)-filter of A. Conversely, any S(V, 1)-filter is a V-ideal by the remarks of [40, Section 7, p. 180].

The following lemma is also stated without proof by Agliano and Ursini in [10, p. 314].

1.8. Algebraisable and Assertional Logics

Lemma 1.8.12. Let  $\forall$  be a variety with normal ideals. For any  $A \in \forall$  and  $I \in I(A)$ ,  $\Omega_A(I) = I^{\epsilon}$ .

*Proof.* Let V, A and I be as stated. In view of the preceding lemma, by [40, Proposition 6.1] we have that  $I = [1]_{\Omega_{A}(I)}$  for any  $I \in I(A)$ , which implies  $\Omega_{A}(I) = I^{\epsilon}$  as required.

For a variety V with normal ideals, let  $V_{\epsilon} := \{ \mathbf{A} : \mathbf{A} \cong \mathbf{B}/\langle 1 \rangle_{\mathbf{B}}^{\epsilon}$  for some  $\mathbf{B} \in \mathsf{V} \}$  be the class of reduced algebras of V.

**Remark 1.8.13.** In [10, p. 296, p. 315] Agliano and Ursini define  $V_{\epsilon} := \{A/\langle 0 \rangle_{\mathbf{A}}^{\epsilon} : \mathbf{A} \in \mathsf{V}\}$  for any variety V with normal ideals. This definition is in error [226]:  $V_{\epsilon}$  must be closed under isomorphic copies by [10, Proposition 3.3]. The definition of  $V_{\epsilon}$  used throughout this thesis reflects this correction.

**Theorem 1.8.14.** Let V be a variety with normal ideals. The following are equivalent:

1. V is finitely congruential;

2. S(V, 1) is algebraisable;

3.  $V_{\epsilon}$  is a quasivariety.

Moreover, if S(V, 1) is algebraisable then  $V_{\epsilon}$  is the equivalent algebraic semantics of S(V, 1).

*Proof.* The first assertion is [10, Theorem 3.13]. By [77, Proposition 6.9] and Lemma 1.8.12,  $V_{\epsilon}$  is exactly the class of algebra reducts of the reduced matrices of S(V, 1), which implies  $V_{\epsilon}$  is the equivalent algebraic semantics of S(V, 1) by Theorem 1.8.7.

**Theorem 1.8.15.** [10, Corollary 3.17] Let V be a variety with normal ideals. The following are equivalent:

1.  $\lor$  is ideal determined;

2.  $V = V_{\epsilon}$ ;

#### **1.8.** Algebraisable and Assertional Logics

3. S(V, 1) is strongly algebraisable and its equivalent algebraic semantics is exactly V.

Generalisations of the previous theorem to the effect that a quasivariety K with 1 is K-1-regular iff K is the equivalent algebraic semantics of its algebraicable assertional logic S(K, 1) have very recently been obtained by several authors: see Blok and Raftery [40] and Barbour and Raftery [16, Corollary 50].

1.8.16. The Deduction Theorem in Algebraic Logic. Let S be a deductive system. S is said to have a *Deduction-Detachment Theorem* (DDT) if there exists a finite set  $\Sigma := \Sigma(p,q) := \{\zeta_i(p,q) : i = 1, ..., n\}$  of formulas of S such that for any set  $\Gamma \cup \{\varphi, \psi\}$  of formulas of S [36, Section 4],

 $\Gamma, \varphi \vdash_{\mathbb{S}} \psi \quad \text{iff} \quad \varphi \vdash_{\mathbb{S}} \Sigma(\varphi, \psi).$ 

In this case,  $\Sigma$  is called a *deduction-detachment set* for S. Observe that the existence of a DDT for S does not imply the existence of a conditional for S; of course, the converse does obtain.

**Theorem 1.8.17.** [36, Theorem 7.3] Let S be a strongly algebraisable deductive system and let V be its equivalent variety semantics. Then S has a DDT iff V has EDPC.

The deduction theorem in algebraic logic has been extensively investigated by Blok, Pigozzi, Czelakowski and others: see for instance [33, 36, 76]. For a history of the deduction theorem in logic see Porte [185].
## Chapter 2

# The Theory of Pre-BCK-Algebras

In this chapter we investigate the theory of pre-BCK-algebras. Our study is based on and guided by Iséki and Tanaka's standard survey paper An introduction to the theory of BCK-algebras [126]. Thus the scope of our study is largely limited to the variety of pre-BCK-algebras simpliciter and to some natural pre-BCK-algebraic counterparts of the varieties of commutative, positive implicative and implicative BCK-algebras. Our study of pre-BCK-algebras does not extend to pre-BCK-algebraic analogues of the varieties of n-potent BCK-algebras described in §1.6.13, or to pre-BCK-algebraic analogues of other varieties of BCK-algebras that have been considered in the literature, such as the residuation subreducts of hoops.

Our main goal in this chapter is to show that pre-BCK-algebras admit a coherent elementary theory. In particular, our aim is to demonstrate that much of the first-order theory of BCK-algebras, suitably generalised, extends to pre-BCK-algebras. (By fiat, we highlight differences between the theory of pre-BCK-algebras and the theory of BCK-algebras where these occur.) Nonetheless, our ultimate motivation in studying pre-BCK-algebras is not simply an interest in generalisation for its own sake; rather, the driving force behind our study lies in the development of the applications of the sequel. This focus is most clearly reflected in this chapter in the emphasis given herein to the development of the order theory and the ideal theory of pre-BCK-algebras.

## 2.1 Pre-BCK-Algebras

In this section we study pre-BCK-algebras as a generalisation of pre-BCKalgebras to the subtractive but not point-regular case.

In §2.1.1 pre-BCK-algebras proper are introduced. It is shown that the variety PBCK of pre-BCK-algebras coincides with a certain variety of algebras generalising BCK and considered by Blok and Raftery in [38] and independently (in dually isomorphic form) by Agliano and Ursini in [10]. We also prove that the variety of pre-BCK-algebras is a subvariety of the varietal closure of the quasivariety of left residuation algebras. In one of the two main results of the section, a 'Clifford-McLean'-type theorem for pre-BCK-algebras, we show that for a pre-BCK-algebra A, the equivalence  $\Xi$  induced by the natural pre-BCK quasiordering  $\preceq$  (in the sense of Lemma 1.2.2) is a congruence on A such that the quotient algebra A/ $\Xi$  is the maximal BCK-algebra homomorphic image of A.

For a suitable notion of pre-ideal, the pre-ideal theory of pre-BCK-algebras is studied in §2.1.20. We provide a simple characterisation of pre-ideal generation in pre-BCK-algebras. For a pre-BCK-algebra  $\mathbf{A}$ , it is shown that a pre-ideal of  $\mathbf{A}$  is just the inverse image of an ideal of the maximal BCK-algebra homomorphic image  $\mathbf{A}/\Xi$ . We also establish some other properties of pre-ideals of pre-BCK-algebras.

In §2.1.25 the relationship between pre-ideals and congruences in pre-BCKalgebras is investigated. It is shown that every pre-ideal of a pre-BCK-algebra A is the 0-class of a PBCK/BCK-congruence on A. Hence we deduce that for pre-BCK-algebras, pre-ideals coincide with the (PBCK-) ideals of Gumm and Ursini described in §1.7.1. We prove that the lattice of all ideals of a pre-BCK-algebra is isomorphic to its lattice of PBCK/BCK-congruences. Further, we establish the existence of a commutative square of isomorphisms between the ideal and PBCK/BCK-congruence lattices of a pre-BCK-algebra and the

的第三人称单数的 化合物 化合物 计合物 化合物 化合物 化合物 化合物 化合物 化合物 化合物 化合物 化合物

#### 2.1. Pre-BCK-Algebras

ideal and BCK-congruence lattices of its maximal BCK-algebra homomorphic image.

The assertional logic of the variety of pre-BCK-algebras is studied in §2.1.33. We prove the variety of pre-BCK-algebras is finitely congruential, and hence show the assertional logic of the variety of pre-BCK-algebras is algebraisable with equivalent algebraic semantics (termwise definitionally equivalent to) BCK. In consequence we infer the other main result of the section: a quasi-identity of the form  $\bigotimes_{i=1}^{n} s_i(\vec{x}) \approx 0 \supset t(\vec{x}) \approx 0$  is satisfied by PBCK iff it is satisfied by BCK.

2.1.1. Pre-BCK-Algebras. An algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is called a pre-BCK-algebra (in the sense of Cornish) if the following identities are satisfied [71, Section 1]:

$$\left((x - y) - (x - z)\right) - (z - y) \approx 0 \tag{2.1}$$

$$\left(x - (x - y)\right) - y \approx 0 \tag{2.2}$$

$$x - x \approx \mathbf{0} \tag{2.3}$$

$$\mathbf{0} - x \approx \mathbf{0}. \tag{2.4}$$

Because of (2.3), the class of pre-BCK-algebras may be understood as a variety of pointed groupoids; consequently we (informally) denote pre-BCK difference by juxtaposition in the sequel when no confusion can arise. Clearly a pre-BCK-algebra is a BCK-algebra iff it satisfies the quasi-identity:

$$x - y \approx 0 \& y - x \approx 0 \supset x \approx y.$$
(2.5)

Recall from [121, 123] that a BCK-algebra  $\mathbf{A} := \langle A; -, 0 \rangle$  satisfies *Iséki's con*dition (S) iff there exists a largest element a + b of the subset  $\{c \in A : ca \leq b\}$ for any  $a, b \in A$ , or, equivalently,  $\mathbf{A}$  possesses another binary operation + such that a(b + c) = (ab)c is identically satisfied. Pre-BCK-algebras were introduced by Cornish in [71] as a means of constructing BCK-algebras from BCK-algebras with condition (S). Cornish's construction, reproduced in Example 2.1.2 below, generalises Wroński's example [240] showing that the class of BCK-algebras is not a variety; see [71, Theorem 2.2] and the remarks thereafter.

Example 2.1.2. [71, Section 2] Let  $\langle A; \div, +, 0 \rangle$  be a BCK-algebra with condition (S) and let t be any element of A. Let  $A' := \{a' : a \in A\}$  and  $A'' := \{a'' : a \in A\}$  be two sets equipotent with A and let  $W_t(A)$  be the (disjoint) union of A, A' and A''. For any  $a, b \in W_t(A)$  let the product  $\div^{\mathbf{W}_t(A)}$  be defined as follows:

ab := a - Ab, a'b' = a''b'' := b - Aa, ab' = ab'' := 0, a'b := (a + b)', a'b := (a + b)'',a'b'' = a''b' := (b + t)a.

By [71, Lemma 2.1]  $\mathbf{W}_{\mathbf{t}}(A) := \langle W_{t}(A); - \mathbf{W}_{\mathbf{t}}(A), 0 \rangle$  is a pre-BCK-algebra. Wroński's example [240] showing that the class of BCK-algebras is not a variety is  $\mathbf{W}_{\mathbf{1}}(\omega)$ .

Let  $\langle A; -, 0 \rangle$  be a pointed groupoid with operation - defined by ab = 0for any  $a, b \in A$ . It is clear from (2.1)-(2.4) that  $\langle A; -, 0 \rangle$  is a pre-BCKalgebra. This observation is indicative of the fact that interesting classes of pre-BCK-algebras are those possessing one or more additional properties. Let  $\langle A; -, +, 0 \rangle$  be a BCK-algebra with condition (S) and let t be any element of A. Because a + 0 = a for any  $a \in A$ ,  $W_t(A)$  is a pre-BCK-algebra satisfying the identity  $x - 0 \approx x$ . Thus it is consistent with Cornish's original construction to mean by 'pre-BCK-algebra' a pre-BCK-algebra (in the sense of Cornish) satisfying  $x - 0 \approx x$ . In this thesis, therefore, a pre-BCK-algebra is a pre-BCK-algebra (in the sense of Cornish) satisfying:

 $\dot{x} - 0 \approx x.$  (2.6)

The variety PBCK of pre-BCK-algebras is thus the class of algebras axiomatised by (2.1)-(2.4) and (2.6). Given these definitions, the following result is immediate.

**Theorem 2.1.3.** The variety of pre-BCK-algebras is subtractive witness x - y.

**Example 2.1.4.** The identities (2.3), (2.4) and (2.6) assert that up to isomorphism there is a single two-element pre-BCK-algebra, namely the two-element implicative BCK-algebra  $C_1$ . It is easy to see  $C_1$  embeds into every non-trivial pre-BCK-algebra and hence that  $Q(C_1)$  is the smallest non-trivial subquasivariety of PBCK; *cf.* [231, p. 6]. Even more is true: because  $C_1$  generates the class of implicative BCK-algebras as a variety (recall Theorem 1.6.19), iBCK is the unique atom in the lattice of *varieties* of pre-BCK-algebras.

**Example 2.1.5.** Denote by  $\mathbf{B}_2$  the algebra  $\langle \{0, 1, 2\}; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  with operation  $\div$  defined by:

| <u>.</u> B <sub>2</sub> | 0 | 1 | 2 |  |
|-------------------------|---|---|---|--|
| 0                       | 0 | 0 | 0 |  |
| 1                       | 1 | 0 | 0 |  |
| 2                       | 2 | 0 | 0 |  |

The 3-element pointed groupoid  $\mathbf{B}_2$  is the simplest example of a pre-BCKalgebra that is distinct from a BCK-algebra. In their Siena paper [54] Burris and Berman have recently catalogued *all* 3-element groupoids; the Siena catalogue number of  $\mathbf{B}_2$  is 216. From this remark and [54, p. 390] the following facts are known about the variety  $\mathbf{V}(\mathbf{B}_2)$  generated by  $\mathbf{B}_2$ :

- The set of types (in the sense of Hobby and McKenzie's tame congruence theory [110]) realised in V(B<sub>2</sub>) is {1,3};
- $V(B_2)$  is not congruence distributive, congruence modular or congruence permutable;
- $V(B_2)$  does not have a decidable first-order theory;
- |F(0)| = 1, |F(1)| = 2 and |F(2)| = 7, where |F(n)| denotes the cardinality of the  $V(B_2)$ -free algebra F(n) on n free generators.

Classes of algebras generalising BCK but which fail in any sense to be pointregular have been considered by several authors in the literature: for instance, see Bunder [52], Blok and Raftery [38], Agliano and Ursini [10] and Humberstone [113]. However, such classes have invariably been introduced in the context of a wider field of study and have not been considered extensively in their own right. Apropos this remark, let V denote the variety of algebras with language  $\langle \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  axiomatised by the identities (2.1), (2.4), (2.6) and introduced by Blok and Raftery in [38, Section 4] in the context of their investigation into the quasivariety of BCK-algebras and its subvarieties.

Lemma 2.1.6. An algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a pre-BCK-algebra iff it satisfies (2.1), (2.4) and (2.6). Thus the variety of pre-BCK-algebras coincides with Blok and Raftery's variety V.

*Proof.* Let  $\langle A; \div, 0 \rangle$  be an algebra of type  $\langle 2, 0 \rangle$  satisfying (2.1), (2.4) and (2.6). Let  $a, b \in A$ . Notice that  $0 = ((a \div 0) \div (a \div b)) \div (b \div 0) = (a \div (a \div b)) \div b$ by (2.1) and (2.6). Thus (2.2) holds. Moreover, by (2.6) we obtain  $0 = ((a \div (a \div 0)) \div 0 = a \div a$  and so (2.3) holds also. The converse is clear.

Let X be the variety of algebras with language  $\langle \div, \mathbf{0} \rangle$  of type  $\langle 2, \mathbf{0} \rangle$  axiomatised by the identities (2.1), (2.3), (2.4), (2.6) and the identity:

$$\left((x - y) - (z - y)\right) - (x - z) \approx 0.$$

$$(2.7)$$

The variety X was introduced by Agliano and Ursini (in dually isomorphic form) in a case study [10, Example 4.5, pp. 330-332] concerning the relationship between ideals and congruences in subtractive varieties.

Lemma 2.1.7. The variety of pre-BCK-algebras satisfies the following identities:

$$(x - y) - (x - (z - (z - y))) \approx 0$$

$$(2.8)$$

$$\left(x - (y - z)\right) - \left(x - ((u - z) - (u - y))\right) \approx 0.$$

$$(2.9)$$

*Proof.* Let A be a pre-BCK-algebra and let  $a, b, c, d \in A$ . For (2.8), put  $\alpha := a, \beta := b$  and  $\gamma := c(cb)$ . We have:

$$0 = ((\alpha\beta)(\alpha\gamma))(\gamma\beta) \qquad \text{by } (2.1)$$
$$= ((\alpha\beta)(\alpha\gamma))((c(cb))b)$$
$$= ((\alpha\beta)(\alpha\gamma))0 \qquad \text{by } (2.2)$$
$$= (\alpha\beta)(\alpha\gamma) \qquad \text{by } (2.6)$$
$$= (ab)(a(c(cb))).$$

For (2.9), put  $\alpha := a, \beta := bc$  and  $\gamma := (dc)(db)$ . We have:

$$0 = ((\alpha\beta)(\alpha\gamma))(\gamma\beta) \qquad \text{by } (2.1)$$
$$= ((\alpha\beta)(\alpha\gamma))(((dc)(db))(bc))$$
$$= ((\alpha\beta)(\alpha\gamma))0 \qquad \text{by } (2.1)$$
$$= (\alpha\beta)(\alpha\gamma) \qquad \text{by } (2.6)$$
$$= (a(bc))(a((dc)(db))).$$

**Proposition 2.1.8.** The variety of pre-BCK-algebras satisfies the following identities:

$$\left(\left(x - y\right) - \left(z - y\right)\right) - \left(x - z\right) \approx \mathbf{0}$$

$$(2.7)$$

$$\left((x - y) - z\right) - \left((x - z) - y\right) \approx \mathbf{0}.$$
(2.10)

*Proof.* Let A be a pre-BCK-algebra and let  $a, b, c \in A$ . We first derive (2.10). For (2.10), put  $\alpha := (ab)c, \beta := (ac), \gamma := b$  and  $\delta := a$ . We have:

$$0 = (\alpha(\beta\gamma))(\alpha((\delta\gamma)(\delta\beta)))$$
by (2.9)  
= (((ab)c)((ac)b))(((ab)c)((ab)(a(ac)))).

Put  $\alpha := ab$ ,  $\beta := c$  and  $\gamma := a$ . We have:

$$\begin{split} \left( (ab)c)((ac)b) \right) \left( ((ab)c)((ab)(a(ac))) \right) \\ &= \left( ((ab)c)((ac)b) \right) \left( (\alpha\beta)(\alpha(\gamma(\gamma\beta))) \right) \\ &= \left( ((ab)c)((ac)b) \right) 0 \qquad \qquad \text{by (2.8)} \\ &= \left( (ab)c)((ac)b \right) \qquad \qquad \qquad \text{by (2.6).} \end{split}$$

For (2.7), put  $\alpha := ab, \beta := cb$  and  $\gamma := ac$ . We have:

$$0 = ((\alpha\beta)\gamma)((\alpha\gamma)\beta)$$
by (2.10)  
=  $((\alpha\beta)\gamma)(((ab)(ac))(cb))$   
=  $((\alpha\beta)\gamma)0$ by (2.1)  
=  $(\alpha\beta)\gamma$ by (2.1)  
=  $((ab)(cb))(ac).$ 

Lemma 2.1.9. An algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a pre-BCK-algebra iff it satisfies (2.1), (2.3), (2.4), (2.6), and (2.7). Thus the variety of pre-BCKalgebras coincides with Agliano and Ursini's variety X.

Proof. Let A be a pre-BCK-algebra. Then  $A \models (2.1), (2.3), (2.4), (2.6)$  by definition. Also  $A \models (2.7)$  by Proposition 2.1.8, so  $A \in X$ . Conversely, if  $\langle A; \div, 0 \rangle$  is an algebra of type  $\langle 2, 0 \rangle$  satisfying (2.1), (2.3), (2.4), (2.6) and (2.7) then  $\langle A; \div, 0 \rangle$  is a pre-BCK-algebra by Lemma 2.1.6.

Several K-O-regular quasivarieties K generalising BCK have also been considered in the literature, of which the most important is the class of left residuation algebras. A *left residuation algebra* is a  $\langle -, 0 \rangle$ -subreduct of a polrim [229, Chapter 1, p. 17]; recall the definition of a polrim from §1.3.3. By van Alten [229, Proposition 1.4(i)] the class LR of all left residuation algebras is a quasivariety, axiomatised by the identities (2.4), (2.6), (2.7) and the quasi-identity (2.5); LR is not a variety [139, Theorem 9]. Because of (2.5) and [78, Theorem 2.3], LR is LR-O-regular, while by van Alten and Raftery [231, Lemma 3.1] BCK is exactly the subquasivariety of left residuation algebras axiomatised by either of the identities:

$$(x - y) - z \approx (x - z) - y$$
 or  $(x - (x - y)) - y \approx 0$ .

Left residuation algebras were introduced under the (misleading—see van Alten [229, Proposition 2.10; p. 40]) name *BCC-algebras* by Komori in [139]. They have since been studied by several authors, including Komori [138, 139], Dudek [82, 83, 84], Ono and Komori [176] and Wroński [242]. A recent major study of left residuation algebras (and their associated assertional logics) is van Alten [229]; see also Raftery and van Alten [192] and van Alten and Raftery [230, 231]. Van Alten [229, Chapter 3] and Komori [138] have also studied the varietal closure H(LR) of the variety of left residuation algebras; the following theorem is due to Komori [138].

**Theorem 2.1.10.** [138, Theorem 6] The variety H(LR) generated by the class LR of all left residuation algebras is finitely based and axiomatised by the identities (2.4), (2.6) and (2.7).

**Proposition 2.1.11.** (cf. [231, Lemma 3.1]) An algebra  $\mathbf{A} := \langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a pre-BCK-algebra iff  $\mathbf{A} \in \mathbf{H}(\mathbf{LR})$  and moreover  $\mathbf{A}$  satisfies:

$$(x - (x - y)) - y \approx 0.$$
(2.2)

Thus the variety of pre-BCK-algebras is an equational subclass of H(LR), the variety generated by the class of all left residuation algebras.

*Proof.* ( $\Rightarrow$ ) Let A be a pre-BCK-algebra. By definition  $\mathbf{A} \models (2.2)$ . Moreover  $\mathbf{A} \models (2.7)$  by Proposition 2.1.8; since  $\mathbf{A} \models (2.4)$ , (2.6) by definition, we have that  $\mathbf{A} \in \mathbf{H}(\mathsf{LR})$  by Theorem 2.1.10.

( $\Leftarrow$ ) Let  $A \in H(LR)$  be such that  $A \models (2.2)$  and let  $a, b, c \in A$ . Throughout the proof we denote  $\dot{-}^{A}$  by juxtaposition for ease of notation. Put  $\alpha := a, \beta := b$  and  $\gamma := c(cb)$ . We have:

$$0 = ((\alpha\beta)(\gamma\beta))(\alpha\gamma) \qquad \qquad \text{by } (2.7)$$

| $= \big((\alpha\beta)((c(cb))b)\big)(\alpha\gamma)$ |          |
|---|----------|
| $= ((\alpha\beta)0)(\alpha\gamma)$                  | by (2.2) |
| $= (\alpha\beta)(\alpha\gamma)$                     | by (2.6) |
| = (ab)(a(c(cb))).                                   | (2.11)   |

Put  $\alpha := a, \beta := bc$  and  $\gamma := (bd)(cd)$ . We also have:

$$D = ((\alpha\beta)(\gamma\beta))(\alpha\gamma) \qquad \text{by } (2.7)$$
  
=  $((\alpha\beta)(((bd)(cd))(bc)))(\alpha\gamma)$   
=  $((\alpha\beta)0)(\alpha\gamma) \qquad \text{by } (2.7)$   
=  $(\alpha\beta)(\alpha\gamma) \qquad \text{by } (2.6)$   
=  $(a(bc))(a((bd)(cd))). \qquad (2.12)$ 

For (2.1), put  $\alpha := (ab)(ac), \beta := cb$  and  $\gamma := ab$ . We have:

$$D = (\alpha\beta) (\alpha(\gamma(\gamma\beta)))$$
 by (2.11)  
= (((ab)(ac))(cb)) (((ab)(ac))((ab)((cb)))).

Put  $\alpha := ab$ ,  $\beta := a$ ,  $\gamma := c$  and  $\delta := b$ . We have:

$$\begin{aligned} (((ab)(ac))(cb))(((ab)(ac))((ab)((ab)(cb)))) \\ &= (((ab)(ac))(cb))((\alpha(\beta\gamma))(\alpha((\beta\delta)(\gamma\delta))))) \\ &= (((ab)(ac))(cb))0 \qquad \qquad by (2.12) \\ &= ((ab)(ac))(cb) \qquad \qquad by (2.6). \end{aligned}$$

Thus  $A \models (2.1)$ . By Theorem 2.1.10 we have that  $A \models (2.4), (2.6)$ , which implies A is a pre-BCK-algebra by Lemma 2.1.6.

Because of Proposition 2.1.11, results obtained by van Alten [229], Komori [138, 139] and others about H(LR) are applicable to PBCK. In particular, the following useful technical lemma may be regarded as a specialisation to pre-BCK-algebras of a result due to van Alten [229].

#### 2.1. Pre-BCK-Algebras

Lemma 2.1.12. (cf. [229, Lemma 1.2]) Let A be a pre-BCK-algebra and let  $\leq$  be the binary relation defined on A by  $a \leq b$  iff ab = 0. Then  $\langle A; \leq \rangle$  is a quasiordered set with least element 0. Moreover, the relation  $\leq$  satisfies the following conditions for any  $a, b, c \in A$ :

- 1. If  $a \leq b$  then  $cb \leq ca$ ;
- 2. If  $a \leq b$  then  $ac \leq bc$ ;
- 3.  $ab \preceq a$ .

Proof. Let A be a pre-BCK-algebra and let  $a, b, c \in A$ . By (2.3),  $a \leq a$ . Suppose  $a \leq b$  and  $b \leq c$ . Then ab = 0 and bc = 0, so ac = ((ac)0)0 = ((ac)(ab))(bc) = 0 by (2.6) and (2.1). Thus  $a \leq c$  and  $\leq$  is a quasiorder on A. Also,  $0 \leq a$  for any  $a \in A$  by (2.4), so 0 is a minimal element under  $\leq$ . Suppose  $0 \neq m \in A$  is another element minimal under  $\leq$ . Then for all  $a \in A$  we have  $m \leq a$ , and in particular  $m \leq 0$ . Thus m0 = 0. But m0 = m by (2.6), so 0 = m, a contradiction. Therefore 0 is unique and so is the least element of A under  $\leq$ . To complete the proof of the lemma it remains to show (1)-(3). For (1), suppose  $a \leq b$ . Then ab = 0 and (cb)(ca) = ((cb)(ca))0 = ((cb)(ca))(ab) = 0 by (2.6) and (2.1). Thus  $cb \leq ca$ . For (2), suppose  $a \leq b$ . Then ab = 0 and (ac)(bc) = ((ac)(bc))(ab) = 0 by (2.6) and (2.7). Thus  $ac \leq bc$ . For (3), just note (ab)a = ((ab)(a0))(0b) = 0 by (2.6), (2.4) and (2.1), whence  $ab \leq a$ .

**Remark 2.1.13.** (cf. [70, Example 3.1]) Let  $\langle A; \leq \rangle$  be a quasiordered set with least element  $0 \in A$ . For any  $a, b \in A$ , let:

 $a \stackrel{\cdot}{-} b := \begin{cases} 0 & \text{if } a \preceq b \\ a & \text{otherwise.} \end{cases}$ 

Then the induced algebra  $\langle A; \div, 0 \rangle$  is a pre-BCK-algebra, whose underlying quasiordering is consistent with the original quasiordering on A. Hence the underlying quasiordering on a pre-BCK-algebra has no interesting properties in general.

The preceding remark notwithstanding, the equivalence  $\Xi$  induced by the preorder  $\preceq$  (in the sense of Lemma 1.2.2) plays a special role in the theory of pre-BCK-algebras. This is shown in the following result, which may be understood as a kind of 'Clifford-McLean theorem' for pre-BCK-algebras.

**Theorem 2.1.14.** For any pre-BCK-algebra A the following assertions hold:

- 1. The equivalence  $\Xi$  induced by  $\preceq$  in the sense of Lemma 1.2.2 is a congruence on A, and A/ $\Xi$  is a BCK-algebra;
- 2.  $\Xi$  is the smallest congruence on A whose quotient algebra is a BCKalgebra, and so  $A/\Xi$  is the maxima BCK-algebra homomorphic image of A;
- 3.  $[0]_{\Xi} = \{0\};$
- 4.  $\Xi = \langle 0 \rangle_{\mathbf{A}}^{\epsilon}$ .

*Proof.* Let A be a pre-BCK-algebra.

For (1), suppose  $a_1 \equiv b_1$ ,  $a_2 \equiv b_2$  for  $a_1$ ,  $b_1$ ,  $a_2$ ,  $b_2 \in A$ . From  $a_1 \preceq b_1$  we have  $a_1a_2 \preceq b_1a_2$  by Lemma 2.1.12(2) and from  $b_2 \preceq a_2$  we have  $b_1a_2 \preceq b_1b_2$  by Lemma 2.1.12(1). By transitivity,  $a_1a_2 \preceq b_1b_2$ . Also, from  $a_2 \preceq b_2$  we have  $b_1b_2 \preceq b_1a_2$  by Lemma 2.1.12(1) and from  $b_1 \preceq a_1$  we have  $b_1a_2 \preceq a_1a_2$  by Lemma 2.1.12(2). By transitivity,  $b_1b_2 \preceq a_1a_2$ . Thus  $a_1a_2 \equiv b_1b_2$  and  $\Xi$  is a congruence on A. To see  $A/\Xi$  is a BCK-algebra it is sufficient to show  $A/\Xi \models (2.5)$ . Suppose  $ab \equiv 0$  and  $ba \equiv 0$ . From  $ab \preceq 0$ ,  $ba \preceq 0$  we have ab = 0 = ba by Lemma 2.1.12, so  $a \equiv b$ . Thus  $A/\Xi \models (2.5)$  and  $A/\Xi$  is a BCK-algebra.

For (2), let  $\theta \in \text{Con } \mathbf{A}$  be such that  $\mathbf{A}/\theta$  is a BCK-algebra. Suppose  $a \equiv b$  for  $a, b \in A$ . Then ab = 0 and ba = 0, and thus  $ab \equiv_{\theta} 0$  and  $ba \equiv_{\theta} 0$ . Since  $\mathbf{A}/\theta \models (2.5)$  we have that  $a \equiv_{\theta} b$ . Thus  $\Xi$  is the smallest congruence on  $\mathbf{A}$  whose quotient algebra is a BCK-algebra, and  $\mathbf{A}/\Xi$  is the maximal BCK-algebra homomorphic image of  $\mathbf{A}$ .

For (3), let  $a \in A$  be such that  $a \equiv 0$ . Then  $a \preceq 0$ , so a = 0 by Lemma 2.1.12.

For (4), it is sufficient by definition of  $\langle 0 \rangle_{\mathbf{A}}^{\epsilon}$  to show  $\Xi$  is the largest congruence  $\theta \in \operatorname{Con} \mathbf{A}$  such that  $[0]_{\theta} = \{0\}$ . Now  $\Xi$  is a congruence on  $\mathbf{A}$  such that  $[0]_{\Xi} = \{0\}$  by (1) and (3). Let  $\theta \in \operatorname{Con} \mathbf{A}$  be such that  $[0]_{\theta} = \{0\}$  and suppose  $a \equiv_{\theta} b$  for  $a, b \in A$ . Then  $ab \equiv_{\theta} aa = 0$  and  $ba \equiv_{\theta} bb = 0$ , so  $ab, ba \in [0]_{\theta}$ . But  $[0]_{\theta} = \{0\}$ , so ab = 0 and ba = 0, which implies  $a\Xi b$ .

**Corollary 2.1.15.** For any pre-BCK-algebra A (with underlying quasiorder  $\preceq$ ) the partial orders  $\preceq/\Xi$  and  $\leq^{A/\Xi}$  coincide, where  $\preceq/\Xi$  denotes the partial order on  $A/\Xi$  induced by  $\preceq$  in the sense of Lemma 1.2.2 and  $\leq^{A/\Xi}$  denotes the underlying partial order of the maximal BCK-algebra homomorphic image  $A/\Xi$ of A.

*Proof.* Let A,  $A/\Xi$ ,  $\preceq/\Xi$  and  $\leq^{A/\Xi}$  be as in the statement of the corollary. Throughout the proof to simplify notation we write  $\overline{a}$  for the equivalence class  $[a]_{\Xi}$  in  $A/\Xi$  containing  $a \in A$ . Let  $\overline{a}, \overline{b} \in A/\Xi$  with  $a, b \in A$ . We have:

 $\overline{a} \leq \Xi \overline{b} \quad \text{iff} \quad a \leq b \qquad \qquad \text{by Lemma 1.2.2}$   $\text{iff} \quad a \stackrel{-}{\longrightarrow} a = 0^{\mathbf{A}} \qquad \qquad \text{by Theorem 2.1.14(3)}$   $\text{iff} \quad \overline{a} \stackrel{-}{\longrightarrow} a^{A/\Xi} \overline{b} = 0^{\mathbf{A}/\Xi}$   $\text{iff} \quad \overline{a} \leq^{\mathbf{A}/\Xi} \overline{b}.$ 

Corollary 2.1.16. Let A be a pre-BCK-algebra. An identity of the form  $t(\vec{x}) \approx 0$  is satisfied by A iff it is satisfied by its maximal BCK-algebra homomorphic image  $A/\Xi$ . In symbols,

 $\mathbf{A} \models t(\vec{x}) \approx \mathbf{0}$  iff  $\mathbf{A}/\Xi \models t(\vec{x}) \approx \mathbf{0}$ .

Proof. Let A be a pre-BCK-algebra. Clearly  $\mathbf{A} \models t(\vec{x}) \approx 0$  implies  $\mathbf{A}/\Xi \models t(\vec{x}) \approx 0$ . For the converse, suppose  $\mathbf{A}/\Xi \models t(\vec{x}) \approx 0$ . Let  $\vec{a} \in A$  and let  $\nu : \mathbf{A} \to \mathbf{A}/\Xi$  denote the natural map. By assumption,  $\nu(t^{\mathbf{A}}(\vec{a})) = \nu(0^{\mathbf{A}})$ ; since  $\nu(0^{\mathbf{A}}) = [0^{\mathbf{A}}]_{\Xi} = \{0^{\mathbf{A}}\}$  (by Theorem 2.1.14(3)), we infer  $t^{\mathbf{A}}(\vec{a}) = 0^{\mathbf{A}}$ . Thus  $\mathbf{A} \models t(\vec{x}) \approx 0$  as desired.

#### 2.1. Pre-BCK-Algebras

Corollary 2.1.17. The class  $PBCK_c$  of reduced algebras of PBCK is exactly the class BCK of BCK-algebras.

*Proof.* Just observe  $\mathbf{A} \in \mathsf{PBCK}_{\epsilon}$  iff  $\langle 0 \rangle_{\mathbf{A}}^{\epsilon} = \omega_{\mathbf{A}}$  iff  $\Xi = \omega_{\mathbf{A}}$  (by Theorem 2.1.14(4)) iff  $\mathbf{A} \in \mathsf{BCK}$ .

Corollary 2.1.18. For any  $A \in PBCK$ ,  $Con_{PBCK/BCK} A \subseteq [\Xi, \iota_A]$ .

*Proof.* Let  $\theta \in \text{ConpBCK/BCK} \mathbf{A}$ . By Theorem 2.1.14(2) we have  $\Xi \leq \theta \leq \iota_{\mathbf{A}}$ , and so  $\theta \in [\Xi, \iota_{\mathbf{A}}]$ . Thus  $\text{ConpBCK/BCK} \mathbf{A} \subseteq [\Xi, \iota_{\mathbf{A}}]$ .

Remark 2.1.19. The converse of the preceding corollary does not hold in general. Consider Wroński's example  $W_1(\omega)$  (Example 2.1.2) that shows the class of BCK-algebras is not a variety. Since  $W_1(\omega)$  is a BCK-algebra,  $\Xi = \omega_{W_1(\omega)}$ . But there exists  $\theta \geq \Xi$  such that  $B_2 \cong W_1(\omega)/\theta$  (see Wroński [240]), whence  $\operatorname{Con_{\mathsf{PBCK/BCK}}} W_1(\omega) \not\subseteq [\Xi, \iota_{W_1(\omega)}]$ .

2.1.20. Pre-Ideal Theory of Pre-BCK-Algebras. Let *I* be a non-empty subset of (the universe of) a pre-BCK-algebra **A**. *I* is called a *pre-ideal* if the following conditions are satisfied:

| $0 \in I$ ,                            | (2.13) |
|--|--------|
| $a \in I$ ha $\in I$ implies $b \in I$ | (2.14) |

The set of all pre-ideals of a pre-BCK-algebra  $\mathbf{A}$  is denoted by  $Pre(\mathbf{A})$ . The following easy lemma collects together some useful facts about pre-ideals.

Lemma 2.1.21. (cf. [229, Lemma 4.16]) For a pre-BCK-algebra A the following assertions hold:

- 1. A pre-ideal of A is a hereditary subset of  $\langle A; \preceq \rangle$ ;
- 2. If I is a pre-ideal of A, and there exist  $a \in A$  and  $b_1, \ldots, b_n \in I$  such that  $(\cdots ((ab_1)b_2)\cdots)b_n = 0$ , then  $a \in I$ ;

3. For any congruence  $\theta$  on  $\mathbf{A}$ , the 0-class  $[0]_{\theta} = \{a \in A : (a,0) \in \theta\}$  of  $\theta$  is a pre-ideal of  $\mathbf{A}$ ;

なるとなどのなどである。現在に、「「」」の「「」」というないで、「」」というないで、「」」の「」」という、

#### 2.1. Pre-BCK-Algebras

#### 4. A pre-ideal of $\mathbf{A}$ is a subuniverse of $\mathbf{A}$ .

Proof. Let A be a pre-BCK-algebra. For (1), let I be a pre-ideal. From  $a \in I$ and  $b \leq a$  we have  $ba = 0 \in I$ , whence  $b \in I$  by the definition of I as a pre-ideal. Since  $0 \in I$ , (2) now follows by repeated application of (1). For (3), let  $\theta \in \text{Con } \mathbf{A}$  and let  $a, ba \in [0]_{\theta}$ . Since  $b = b0 \equiv_{\theta} ba \equiv_{\theta} 0$  we have  $b \equiv_{\theta} 0$ ; that is to say  $b \in [0]_{\theta}$ . Since  $0 \in [0]_{\theta}$ , we have that  $[0]_{\theta}$  is a pre-ideal. For (4), let I be a pre-ideal of  $\mathbf{A}$  and let  $a, b \in I$ . By Lemma 2.1.12(3) we have  $ab \leq a$ ; thus  $ab \in I$  by (1) and I is a subuniverse of  $\mathbf{A}$ .

The conditions (2.13)-(2.14) defining pre-ideals, in conjunction with the results of Lemma 2.1.21, suggest intuitively that every pre-ideal of a pre-BCK-algebra is the inverse image of an ideal of its maximal BCK-algebra homomorphic image. This intuition is made precise in the following theorem, the proof of which is due *mutatis mutandis* to the author's Ph.D. supervisor. See also Corollary 2.1.29 in the sequel.

**Theorem 2.1.22.** [17, Lemma 1.1.9] The set  $Pre(\mathbf{A})$  of all pre-ideals of a pre-BCK-algebra  $\mathbf{A}$  forms an algebraic lattice  $Pre(\mathbf{A})$  when ordered by inclusion, which is isomorphic to the lattice of ideals of  $\mathbf{A}/\Xi$  under the map  $I \stackrel{\psi}{\mapsto} I/\Xi$  for any  $I \in Pre(\mathbf{A})$ .

Proof. It is clear that if I is a pre-ideal of A, then  $\psi(I)$  is an ideal of  $A/\Xi$ . Conversely, if J is an ideal of  $A/\Xi$ , then  $I = \{a \in A : a \in [b]_{\Xi} \text{ for some } [b]_{\Xi} \in J\}$  is a pre-ideal of A with the property that  $\psi(I) = J$ . Thus  $\psi$  is a bijection between the pre-ideals of A and the ideals of  $A/\Xi$ . Suppose now that  $I_1$  and  $I_2$  are pre-ideals of A such that  $I_1 \subseteq I_2$ . Then  $[a]_{\Xi} \in \psi(I_1)$  implies  $\{b \in A : b \equiv a\} \subseteq I_1 \subseteq I_2$ , which implies  $[a]_{\Xi} \in \psi(I_2)$ , and so  $\psi(I_1) \subseteq \psi(I_2)$ . Conversely if  $J_1$  and  $J_2$  are ideals of  $A/\Xi$  such that  $J_1 \subseteq J_2$ , then  $\psi^{-1}(J_1) = \{a \in A : a \in [b]_{\Xi} \text{ for some } [b]_{\Xi} \in J_1\} \subseteq \{a \in A : a \in [b]_{\Xi} \text{ for some } [b]_{\Xi} \in J_2\} = \psi^{-1}(J_2)$ . Thus  $\psi$  is an order isomorphism, and hence a lattice isomorphism between the algebraic lattices of pre-ideals and ideals of A and  $A/\Xi$ .

In light of the preceding theorem, it is natural to anticipate that results concerning ideals of BCK-algebras cited in §1.6.1 extend to pre-ideals of pre-BCKalgebras. Certainly this is the case in relation to the following proposition, which characterises the pre-ideal  $\langle B \rangle_{\mathbf{A}} := \bigcap \{ J \in \operatorname{Pre}(\mathbf{A}) : B \subseteq J \}$  generated by non-empty  $B \subseteq A$  of any pre-BCK-algebra  $\mathbf{A}$  (of course,  $\langle B \rangle_{\mathbf{A}}$  exists because pre-ideals of  $\mathbf{A}$  are closed under arbitrary intersections). The result is due to van Alten [229, Proposition 4.16(iv)], who first stated and proved the proposition in the context of pre-ideals of left residuation algebras. Because of Proposition 2.1.11, van Alten's proof generalises to pre-BCK-algebras without modification.

**Proposition 2.1.23.** (cf. [125, Theorem 3]) Let A be a pre-BCK-algebra. For any  $\emptyset \neq B \subseteq A$ ,

$$\langle B \rangle_{\mathbf{A}} = \big\{ a \in A : (\exists n \in \omega) (\exists c_1, \ldots, c_n \in B) \text{ such that } \big( \cdots (ac_1) \cdots \big) c_n = 0 \big\}.$$

In particular, for any  $b \in A$ ,

$$\langle b \rangle_{\mathbf{A}} = \{ a \in A : (\exists n \in \omega) \text{ such that } ab^n = 0 \}.$$

Proof. Let A be a pre-BCK-algebra and let  $\emptyset \neq B \subseteq A$ . Let  $D := \{a \in A : (\exists n \in \omega) (\exists c_1, \ldots, c_n \in B) \text{ such that } (\cdots (ac_1) \cdots) c_n = 0\}$ . By Lemma 2.1.21(2) we have  $D \subseteq \langle B \rangle_{\mathbf{A}}$ . Also,  $B \subseteq D$ , so it remains only to establish that D is a pre-ideal of A. Let  $a \in A$  and  $b, ab \in D$ , say:

$$(\cdots ((ab)c_1)\cdots)c_n=0=(\cdots (bd_1)\cdots)d_m,$$

where the  $c_i$ ,  $d_j$  are in B. Then:

$$(\cdots (((\cdots (ad_1) \cdots) d_m)c_1) \cdots)c_n$$

$$= (\cdots ((((\cdots (ad_1) \cdots) d_m)0)c_1) \cdots)c_n$$

$$= (\cdots (((((\cdots (ad_1) \cdots) d_m)((\cdots (bd_1) \cdots) d_m))c_1) \cdots)c_n$$

$$\preceq (\cdots ((ab)c_1) \cdots c_n)$$
by (2.7) and Lemma 2.1.12(2)
$$= 0$$

so  $a \in D$ , as required.

Some further properties of pre-ideals of pre-BCK-algebras analogous to prop-

**的复数形式,在这次通过这些正式的过去式和过去分词,这些正式这些问题,这些问题的是一些正式的,这些正式的是一个,这些正式的,你们们们们们们,你们们们的,你们们们** 

erties of ideals of BCK-algebras are established in the following proposition. Concerning the statement of the proposition, a pre-BCK-algebra  $\mathbf{A}$  is said to enjoy the *pre-ideal extension property* if for any  $\mathbf{B} \in \mathbf{S}(\mathbf{A})$  and any  $I \in \operatorname{Pre}(\mathbf{B})$ there is a pre-ideal  $J \in \operatorname{Pre}(\mathbf{A})$  such that  $J \cap B = I$ .

**Proposition 2.1.24.** For any pre-BCK-algebra A, the following assertions hold:

1. Pre(A) is a distributive lattice;

#### 2. A enjoys the pre-ideal extension property.

Proof. (1) follows immediately from Theorem 2.1.22 and Proposition 1.6.9(1). For (2), suppose  $\mathbf{A} \in \mathsf{PBCK}$ ,  $\mathbf{B} \in \mathbf{S}(\mathbf{A})$  and  $I \in \operatorname{Pre}(\mathbf{B})$ . Then for  $\mathbf{A}/\Xi \in \mathsf{BCK}$ ,  $\mathbf{B}/\Xi \in \mathbf{S}(\mathbf{A}/\Xi)$  and  $\psi(I) \in \operatorname{Pre}(\mathbf{B}/\Xi)$  there exists an ideal  $J \in \operatorname{Pre}(\mathbf{A}/\Xi)$ such that  $J \cap B/\Xi = \psi(I)$ , where  $\psi(I)$  is the image of I under the map  $\psi$  of Theorem 2.1.22. Indeed, by Proposition 1.6.9(2) we can pick  $J = \langle \psi(I) \rangle_{\mathbf{A}/\Xi}$ . Now  $\psi^{-1}(J)$  is a pre-ideal of  $\mathbf{A}$  with the property that  $\psi^{-1}(J) \cap B = I$ .

2.1.25. Pre-Ideals and Congruences of Pre-BCK-algebras. Recall from  $\S1.6.1$  that for a BCK-algebra A, any ideal of A (in the sense of  $\S1.6.1$ ) is the 0-class of a BCK-congruence on A, and conversely that the 0-class of any (BCK-) congruence on A is an ideal of A. Inasmuch as pre-ideals and PBCK/BCK-congruences are (for pre-BCK-algebras) the pre-BCK-algebraic analogues of ideals and BCK-congruences (of BCK-algebras), respectively, the preceding remarks invite (for pre-BCK-algebras) a study of the relationship between pre-ideals and PBCK/BCK-congruences.

Theorem 2.1.26. For any pre-BCK-algebra A and  $I \in Pre(A)$ , the relation  $\phi_I$  defined on  $A \times A$  by:

 $\phi_I := \{(a, b) \in A \times A : ab, ba \in I\}$ 

is a congruence on A, and the quotient algebra  $A/\phi_I$  is a BCK-algebra.

*Proof.* Let A be a pre-BCK-algebra, let  $I \in Pre(A)$  and let  $\phi_I$  be as in the statement of the theorem. To see  $\phi_I$  is an equivalence relation, we show (for any  $a, b, c \in A$ ):

- (i)  $a \equiv_{\phi_I} a;$
- (ii)  $a \equiv_{\phi_I} b$  implies  $b \equiv_{\phi_I} a$ ;
- (iii)  $a \equiv_{\phi_I} b, b \equiv_{\phi_I} c$  implies  $a \equiv_{\phi_I} c$ .

For (i), we have that  $aa = 0 \in I$  by (2.3), so  $a \equiv_{\phi_I} a$ .

For (ii), we have that  $a \equiv_{\phi_I} b$  implies  $b \equiv_{\phi_I} a$  by definition of  $\phi_I$ .

For (iii), suppose  $a \equiv_{\phi_I} b$  and  $b \equiv_{\phi_I} c$ . By (2.1) applied twice, we have that  $((ac)(ab))(bc) = 0 \in I$  and  $((ca)(cb))(ba) = 0 \in I$ . Since  $ab, bc \in I$  and  $cb, ba \in I$  by assumption, we have that  $ac, ca \in I$  by Lemma 2.1.21(2) applied twice. Hence  $a \equiv_{\phi_I} c$ .

By (i), (ii) and (iii),  $\phi_I$  is an equivalence relation. To see  $\phi_I$  is a congruence on A, it is sufficient to show (for any  $a_1, a_2, b_1, b_2 \in A$ ):

(iv)  $a_1 \equiv_{\phi_1} a_2$  and  $b_1 \equiv_{\phi_1} b_2$  implies  $a_1 b_1 \equiv_{\phi_1} a_2 b_2$ .

To prove (iv), we show:

(iv)(a)  $b_1 \equiv_{\phi_I} b_2$  implies  $a_1 b_1 \equiv_{\phi_I} a_1 b_2$ ; (iv)(b)  $a_1 \equiv_{\phi_I} a_2$  implies  $a_1 b_2 \equiv_{\phi_I} a_2 b_2$ .

For (iv)(a), suppose  $b_1 \equiv_{\phi_I} b_2$ . By (2.1) applied twice,  $((a_1b_1)(a_1b_2))(b_2b_1) = 0 \in I$  and  $((a_1b_2)(a_1b_1))(b_1b_2) = 0 \in I$ . Since  $b_2b_1 \in I$ ,  $b_1b_2 \in I$  by hypothesis,  $(a_1b_1)(a_1b_2) \in I$ ,  $(a_1b_2)(a_1b_1) \in I$  by Lemma 2.1.21(1) applied twice. Hence  $a_1b_1 \equiv_{\phi_I} a_1b_2$ .

For (iv)(b), suppose  $a_1 \equiv_{\phi_I} a_2$ . By (2.7) applied twice,  $((a_1b_2)(a_2b_2))(a_1a_2) = 0 \in I$  and  $((a_2b_2)(a_1b_2))(a_2a_1) = 0 \in I$ . Since  $a_1a_2 \in I$ ,  $a_2a_1 \in I$  by hypothesis,  $(a_1b_2)(a_2b_2) \in I$ ,  $(a_2b_2)(a_1b_2) \in I$  by Lemma 2.1.21(1) applied twice. Hence  $a_1b_2 \equiv_{\phi_I} a_2b_2$ .

By (iv)(a), (iv)(b) and (iii), we infer that  $a_1 \equiv_{\phi_I} a_2$  and  $b_1 \equiv_{\phi_I} b_2$  implies  $a_1b_1 \equiv_{\phi_I} a_2b_2$ , which establishes (iv).

By (i), (ii), (iii) and (iv),  $\phi_I$  is a congruence on **A**. Hence  $\mathbf{A}/\phi_I$  is a pre-BCKalgebra. To complete the proof, it is sufficient in view of the remarks of §2.1.1 to show  $\mathbf{A}/\phi_I \models (2.5)$ . So suppose  $ab \equiv_{\phi_I} 0$  and  $ba \equiv_{\phi_I} 0$  for  $a, b \in A$ . Then  $(ab)_0 \in I$  and  $(ba)_0 \in I$ , so  $ab \in I$  and  $ba \in I$  by (2.6); that is to say  $a \equiv_{\phi_I} b$ . Hence  $\mathbf{A}/\phi_I \models$  (2.5). Therefore  $\mathbf{A}/\phi_I$  is a BCK-algebra, and the proof is complete.

By Proposition 1.7.5, any subtractive variety has normal ideals. This result does not extend to (even K-0-regular) quasivarieties K with a binary term s(x, y) witnessing (1.42), (1.43): for a counterexample see van Alten [229, pp. 71-72]. Nonetheless, Proposition 4.15 of van Alten [229] implies that, for a BCK-algebra A, a non-empty subset  $I \subseteq A$  is a BCK-ideal of A iff  $I \in N(A)$  iff I is an ideal of A (in the sense of §1.6.1). For pre-BCK-algebras, Theorem 2.1.26 in conjunction with the following lemma yields the crucial connection between pre-ideals and PBCK-ideals presented in Proposition 2.1.28 below.

Lemma 2.1.27. Let A be a pre-BCK-algebra, let  $I \in Pre(A)$  and let  $\phi_I$  be the congruence induced by I in the sense of Theorem 2.1.26. Then  $I = [0]_{\phi_I}$ , and  $\phi_I$  is the largest congruence on A with this property.

Proof. Let A be a pre-BCK-algebra, let  $I \in Pre(A)$  and let  $\phi_I$  be the congruence induced by I in the sense of Theorem 2.1.26. Let  $a \in I$ . By (2.6),  $a0 = a \in I$ , while  $0a = 0 \in I$  by (2.4). Hence  $a \equiv_{\phi_I} 0$ . That is to say,  $a \in [0]_{\phi_I}$ , so  $I \subseteq [0]_{\phi_I}$ . Conversely, let  $a \in [0]_{\phi_I}$ . Then  $a \equiv_{\phi_I} 0$ , and so  $a0 \in I$ . Since  $0 \in I$ , we infer that  $a \in I$  by the definition of I as a pre-ideal. Thus  $[0]_{\phi_I} \subseteq I$  and  $I = [0]_{\phi_I}$ .

Suppose now that  $\theta$  is a congruence on A such that  $[0]_{\theta} = I$ . Let  $a \equiv_{\theta} b$  for  $a, b \in A$ . Then  $ab \equiv_{\theta} aa = 0$  and  $ba \equiv_{\theta} bb = 0$ , so  $ab, ba \in [0]_{\theta}$ ; that is to say  $a \equiv_{\phi_{\{0\}_{\theta}}} b$ . Since  $[0]_{\theta} = I$ ,  $a \equiv_{\phi_I} b$ . Thus  $\theta \subseteq \phi_I$ , and the proof is complete.

**Proposition 2.1.28.** A non-empty subset  $I \subseteq A$  of a pre-BCK-algebra A is a pre-ideal of A iff it is a PBCK-ideal of A. Hence  $I_{PBCK}(A) = Pre(A)$ .

*Proof.* Let A be a pre-BCK-algebra with  $\{0\} \subseteq I \subseteq A$ .

(⇒) Suppose  $I \in Pre(\mathbf{A})$ . By Lemma 2.1.27,  $I = [0]_{\phi_I}$ , where  $\phi_I$  is the congruence on  $\mathbf{A}$  of Theorem 2.1.26. Hence  $I \in N(\mathbf{A})$ . Since  $\mathbf{A}$  is subtractive,  $I \in I_{\mathsf{PBCK}}(\mathbf{A})$  by Proposition 1.7.5. Hence  $\operatorname{Pre}(\mathbf{A}) \subseteq I_{\mathsf{PBCK}}(\mathbf{A})$ .

( $\Leftarrow$ ) Suppose  $I \in I_{\mathsf{PBCK}}(\mathbf{A})$ . Since  $\mathbf{A}$  is subtractive,  $I \in N(\mathbf{A})$  by Proposition 1.7.5. That is,  $I = [0]_{\theta}$  for some  $\theta \in \operatorname{Con} \mathbf{A}$ . Hence  $I \in \operatorname{Pre}(\mathbf{A})$  by Lemma 2.1.21(3). Thus  $I_{\mathsf{PBCK}}(\mathbf{A}) \subseteq \operatorname{Pre}(\mathbf{A})$ .

In view of the preceding proposition, from this point forwards we will always employ the notation and terminology of §1.7.1 in connection with (pre-) ideals of pre-BCK-algebras.

Corollary 2.1.29. The set I(A) of all ideals of a pre-BCK-algebra A yields a distributive algebraic lattice I(A) when ordered by inclusion. For any  $I, J \in I(A)$ ,

$$I \wedge J = I \cap J = \{a \sqcap b : a \in I, b \in J\}$$
$$I \vee J = \{b : for some \ a \in I, \ ba \in J\}$$

where  $a \sqcap b := a(ab)$  for any  $a, b \in A$ .

Proof. Let A be a pre-BCK-algebra. By the remarks of §1.7.1, the set I(A) of ideals of A yields an algebraic lattice I(A) under inclusion, which is distributive by Proposition 2.1.28 and Proposition 2.1.24(1). Let  $I, J \in I(A)$ . It is clear that  $I \wedge J = I \cap J$ . To see  $I \cap J = \{a \sqcap b : a \in I, b \in J\}$ , let  $c := a \sqcap b$  where  $a \in I$  and  $b \in J$ . Then  $c \preceq a \in I$  and  $c \preceq b \in J$  by Lemma 2.1.12(3) and (2.2) respectively, so  $c \in I$  and  $c \in J$  by Lemma 2.1.21(1); that is to say  $c \in I \cap J$ . Hence  $\{a \sqcap b : a \in I, b \in J\} \subseteq I \cap J$  and, since the opposite inclusion is trivial, equality holds. That  $I \vee J = \{b : \text{for some } a \in I, ba \in J\}$  follows immediately from the remarks of §1.7.1 and Part (4) of [222, Proposition 1.3].

By Theorem 1.6.10, the maps  $I \stackrel{\xi}{\mapsto} \phi_I$   $(I \in I(\mathbf{A}))$  and  $\theta \stackrel{\xi^{-1}}{\mapsto} [0]_{\theta}$   $(\theta \in \operatorname{Con}_{\mathsf{BCK}} \mathbf{A})$  are, for any BCK-algebra **A**, mutually inverse lattice isomorphisms between the ideal lattice of **A** and BCK-congruence lattice of **A**, which result suggests the following theorem.

Theorem 2.1.30. For any pre-BCK-algebra A, the maps  $I \stackrel{\varphi}{\mapsto} \phi_I$   $(I \in I(A))$ and  $\theta \stackrel{\varphi^{-1}}{\mapsto} [0]_{\theta}$   $(\theta \in Con_{\mathsf{PBCK}/\mathsf{BCK}} A)$  are mutually inverse lattice isomorphisms between the PBCK/BCK-congruence lattice of A and the ideal lattice of A. *Proof.* Let A be a pre-BCK-algebra. To prove the theorem it suffices to show:

- (i) The maps  $\varphi$  and  $\varphi^{-1}$  are mutually inverse bijections;
- (ii) The maps  $\varphi$  and  $\varphi^{-1}$  are order preserving.

For (i), it is sufficient to show  $\varphi^{-1} \circ \varphi = \omega_{I(A)}$  and  $\varphi \circ \varphi^{-1} = \omega_{Con_{\mathsf{PBCK/BCK}}A}$ . In the former case  $\varphi^{-1} \circ \varphi = \omega_{I(A)}$  since  $I = [0]_{\phi_I}$  for every  $I \in I(A)$  by Lemma 2.1.27 In the latter case  $\theta \leq \phi_{[0]_{\theta}}$  (for  $\theta \in Con_{\mathsf{PBCK/BCK}}A$ ) by Lemma 2.1.27, so it remains only to show  $\phi_{[0]_{\theta}} \leq \theta$ . So let  $a \equiv b \pmod{\phi_{[0]_{\theta}}}$  for  $a, b \in A$ . Then  $ab, ba \in [0]_{\theta}$  by definition of  $\phi_{[0]_{\theta}}$ ; that is to say  $ab, ba \equiv_{\theta} 0$ . Since  $A/\theta \in \mathsf{BCK}$  by hypothesis, we infer that  $a \equiv_{\theta} b$  by (2.5). Hence  $\phi_{[0]_{\theta}} \subseteq \theta$ . Thus  $\phi_{[0]_{\theta}} = \theta$ , and so  $\varphi \circ \varphi^{-1} = \omega_{Con_{\mathsf{PBCK/BCK}}A$ .

For (ii), to see  $\varphi$  is order preserving let  $I, J \in I(\mathbf{A})$  with  $I \subseteq J$ . Suppose  $a \equiv_{\phi_I} b$  for  $a, b \in A$ . Then  $ab, ba \in I$ , and since  $I \subseteq J$  we have also that  $ab, ba \in J$ . Thus  $a \equiv_{\phi_J} b$  and  $\phi_I \subseteq \phi_J$ . Hence  $\varphi$  is order preserving. To see  $\varphi^{-1}$  is order preserving, let  $\theta, \psi \in \operatorname{Con_{PBCK/BCK}} \mathbf{A}$  with  $\theta \subseteq \psi$ . Suppose  $a \in [0]_{\theta}$ . Then  $a \equiv_{\theta} 0$ , and since  $\theta \subseteq \psi$  we have also that  $a \equiv_{\psi} 0$ . Thus  $a \in [0]_{\psi}$  and  $[0]_{\theta} \subseteq [0]_{\psi}$ . Hence  $\varphi^{-1}$  is order preserving.

Corollary 2.1.31. For any pre-BCK-algebra A, the following assertions hold:

1. A is PBCK/BCK-0-regular;

2. A is PBCK/BCK-congruence distributive;

3. A has the PBCK/BCK-congruence extension property.

**Proof.** Let A be a pre-BCK-algebra. For (1), let  $\theta$  and  $\psi$  be PBCK/BCKcongruences on A such that  $[0]_{\theta} = [0]_{\psi}$ . By Lemma 2.1.21(3),  $[0]_{\theta}$  and  $[0]_{\psi}$ are both ideals, which coincide by assumption of the equality of  $[0]_{\theta}$  and  $[0]_{\psi}$ . Because of Theorem 2.1.30, we infer that  $\theta = \psi$ , so A is PBCK/BCK-0-regular and (1) holds. The remaining assertions of the corollary now follow trivially from PBCK/BCK-0-regularity and Proposition 2.1.24.

From Theorem 1.6.10, Theorem 2.1.22 and Theorem 2.1.30, we may also infer:

Theorem 2.1.32. For any pre-BCK-algebra A, the map  $\theta \stackrel{v}{\mapsto} \theta/\Xi$  ( $\theta \in Con_{\mathsf{PBCK/BCK}} A$ ) is an isomorphism from the PBCK/BCK-congruence lattice of A onto the BCK-congruence lattice of A/ $\Xi$ . Therefore if  $H(A/\Xi) \subseteq BCK$  then  $Con_{\mathsf{PBCK/BCK}} A$  and  $Con A/\Xi$  are isomorphic.

Proof. For any pre-BCK-algebra A, we have that  $\operatorname{Con_{PBCK/BCK}} A \cong I(A)$  by Theorem 2.1.30; that  $I(A) \cong I(A/\Xi)$  by Theorem 2.1.22; and that  $I(A/\Xi) \cong$  $\operatorname{Con_{BCK}} A/\Xi$  by Theorem 1.6.10. By composition of maps, it follows that  $\operatorname{Con_{PBCK/BCK}} A \cong \operatorname{Con_{BCK}} A/\Xi$  under the mapping  $\theta \stackrel{v}{\mapsto} \phi_{[0]_{(\theta/\Xi)}}$ , which simplifies to  $\theta \stackrel{v}{\mapsto} \theta/\Xi$  by PBCK/BCK-0-regularity as claimed. The remaining assertion of the theorem now follows, because the condition  $H(A/\Xi) \subseteq$  BCK guarantees that every congruence on  $A/\Xi$  is a BCK-congruence.

We conclude this subsection by noting that the proof of Theorem 2.1.32 implicitly establishes the existence of a commutative square of isomorphisms connecting the ideal and PBCK/BCK-congruence lattices of a pre-BCK-algebra to the ideal and BCK-congruence lattices of its maximal BCK-algebra homomorphic image. In more detail: given a pre-BCK-algebra A (with maximal BCK-algebra homomorphic image  $A/\Xi$ ) and the maps:

| $I \stackrel{\xi}{\mapsto} \phi_I,$        | $I \in I(A/\Xi)$                       |
|--|--|
| $I \stackrel{\psi}{\mapsto} I/\Xi,$        | $I \in I(\mathbf{A})$                  |
| $I \stackrel{\varphi}{\mapsto} \phi_I,$    | $I \in \mathrm{I}(\mathbf{A})$         |
| $\theta \stackrel{v}{\mapsto} \theta/\Xi,$ | $	heta\in\operatorname{Con}\mathbf{A}$ |

of Theorem 1.6.10, Theorem 2.1.22, Theorem 2.1.30 and Theorem 2.1.32 respectively, the diagram:



commutes in the category of lattices.

2.1.33. The Assertional Logic of the Variety PBCK. Let the class  $\mathsf{PBCK}^D$ 

of dual pre-BCK-algebras be the variety of algebras with language  $\langle \rightarrow, 1 \rangle$  of type  $\langle 2, 0 \rangle$  axiomatised by the following identities:

| (x  ightarrow y)  ightarrow | $((y \rightarrow z) \rightarrow$ | $(x \rightarrow z)) \approx 1$ |  | (2.15) |
|-----------------------------|----------------------------------|--------------------------------|--|--------|
|-----------------------------|----------------------------------|--------------------------------|--|--------|

$$x \to ((x \to y) \to y) \approx 1$$
 (2.16)

$$x \to x \approx \mathbf{1} \tag{2.17}$$

$$x \to \mathbf{1} \approx \mathbf{1} \tag{2.18}$$

 $1 \to x \approx x; \tag{2.19}$ 

notice that the class  $BCK^D$  of dual BCK-algebras (recall Example 1.8.4) is exactly the subquasivariety of  $PBCK^D$  axiomatised by the quasi-identity:

$$x \to y \approx \mathbf{1} \& y \to x \approx \mathbf{1} \supset x \approx y.$$

Let  $T_{\mathcal{L}}(\mathbf{X})$  denote the term algebra of type  $\mathcal{L}$  over  $\mathbf{X}$ , where  $\mathcal{L}$  is the language of pre-BCK-algebras. Also, let  $T_{\mathcal{L}^D}(\mathbf{X})$  denote the term algebra of type  $\mathcal{L}^D$ over  $\mathbf{X}$ , where  $\mathcal{L}^D$  is the language of dual pre-BCK-algebras. Consider the maps  $\eta: T_{\mathcal{L}}(\mathbf{X}) \to T_{\mathcal{L}^D}(\mathbf{X})$  and  $\xi: T_{\mathcal{L}^D}(\mathbf{X}) \to T_{\mathcal{L}}(\mathbf{X})$  defined respectively by:

| $\eta(0) := 1$                       |   |
|--------------------------------------|---|
| $\eta(x) := x$                       | $x \in \mathbf{X}$                            |
| $\eta(p - q) := \eta(q) \to \eta(p)$ | $p,q\in \mathrm{T}_{\mathcal{L}}(\mathbf{X})$ |

and:

$$\begin{split} \xi(1) &:= 0 \\ \xi(x) &:= x \\ \xi(r \to s) &:= \xi(s) - \xi(r) \end{split} \qquad x \in \mathbf{X} \\ r, s \in \mathbf{T}_{\mathcal{L}^D}(\mathbf{X}). \end{split}$$

Because of the axiomatisation of PBCK by (2.1)-(2.5) and the axiomatisation of PBCK<sup>D</sup> by (2.15)-(2.19), the proof of the following lemma is trivial and hence is omitted.

Lemma 2.1.34. For  $p, q \in T_{\mathcal{L}}(\mathbf{X})$  and  $r, s \in T_{\mathcal{L}^{D}}(\mathbf{X})$  the following assertions

hold:

1. If PBCK 
$$\models p \approx q$$
 then PBCK<sup>D</sup>  $\models \eta(p) \approx \eta(q)$ ;

2. If  $PBCK^{D} \models r \approx s$  then  $PBCK \models \xi(r) \approx \xi(s)$ .

Moreover,  $\xi \circ \eta = \omega_{T_{\mathcal{L}}(\mathbf{X})}$  and  $\eta \circ \xi = \omega_{T_{\mathcal{L}}(\mathbf{X})}$ .

Recall from §1.8.9 that for any quasivariety K with 1, the inherent assertional logic S(K, 1) of K may be defined by specifying that, for all  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ ,  $\Gamma \vdash_{S(K,1)} \varphi$  iff  $\{\psi \approx 1 : \psi \in \Gamma\} \models_{\mathsf{K}} \varphi \approx 1$ . Because the variety of pre-BCK-algebras is termwise definitionally equivalent to (in fact, is dually isomorphic to) the variety of dual pre-BCK-algebras (by Lemma 2.1.34), the preceding observation implies the assertional logics  $S(\mathsf{PBCK}, \mathbf{0})$  and  $S(\mathsf{PBCK}^D, \mathbf{1})$  are definitionally equivalent. In a sense, it is thus a harmless notational convenience to ascribe to the variety of pre-BCK-algebras the intrinsic assertional logic  $S(\mathsf{PBCK}^D, \mathbf{1})$  of the variety  $\mathsf{PBCK}^D$  of dual pre-BCK-algebras; of course, like remarks apply concerning BCK and  $S(\mathsf{BCK}^D, \mathbf{1})$ . In the sequel, therefore, we shall not hesitate to denote by  $S(\mathsf{PBCK}^D, \mathbf{1})$  [ $S(\mathsf{BCK}^D, \mathbf{1})$ ] the inherent assertional logic of the variety of pre-BCK-algebras [the quasivariety of BCK-algebras] when convenient.

**Proposition 2.1.35.** [10, Proposition 4.5]  $D := \{x - y, y - x\}$  is an ICsystem (without parameters) for PBCK. Hence  $I^{\epsilon} = \phi_I$  for any ideal I and  $\theta^1 = \phi_{[0]_{\theta}}$  for any congruence  $\theta$  of a pre-BCK-algebra.

Proof. Let  $D := \{x \div y, y \div x\}$  and observe D 'a system of 0-terms for PBCK. Let  $A \in PBCK$  and recall  $I^D := \{(a, b) \in A \ A : ab, ba \in I\}$  for any ideal  $I \in I(A)$ . Since  $\langle 0 \rangle_A = \{0\}, \langle 0 \rangle_A^D = \{0\}^D = \{(a, b) \in A \times A : ab, ba = 0\} = \Xi \in Con A$  by Theorem 2.1.14(1). By Proposition 1.7.7(2) we have that D is an IC-system without parameters for PBCK. The second statement now follows immediately from Proposition 1.7.7(4).

In [130] Kabziński studied the lattice  $\Lambda^Q(K)$  of all quasivarieties K over the language  $\langle \rightarrow, 1 \rangle$  of BCK<sup>D</sup> algebras for which  $\models_{M(K,1)} = \models_{M(BCK^D,1)}$ , where  $\models_{M(K,1)}$  is the matrix consequence determined by the class of all matrices

#### 2.1. Pre-BCK-Algebras

 $\{\langle \mathbf{A}, \{\mathbf{1}^{\mathbf{A}}\}\rangle : \mathbf{A} \in \mathsf{K}\}\$  and  $\models_{\mathsf{M}(\mathsf{BCK}^D,\mathbf{1})}\$  is the matrix consequence of Example 1.8.10, or equivalently all quasivarieties K for which K is an algebraic semantics for  $\mathbb{BCK}$ . By [130, Fact 1]  $\mathbb{BCK}^D$  is the smallest element of  $\Lambda^Q(\mathsf{K})$ . In [130, Fact 2] Kabziński asserts without proof that  $\mathsf{PBCK}^D$  is a member of  $\Lambda^Q(\mathsf{K})$ . The following theorem, which shows  $\mathsf{PBCK}$  and  $\mathsf{BCK}$  have the same assertional logic (namely  $\mathbb{BCK}$ ), verifies Kabziński's result. For some recent results related to and generalising the theorem, see Blok and Raftery [41, Example 6.2].

**Theorem 2.1.36.** The assertional logic  $S(PBCK^D, 1)$  of the variety of pre-BCK-algebras is algebraisable with equivalent algebraic semantics  $BCK^D$ . Thus the variety of pre-BCK-algebras and the quasivariety of BCK-algebras have the same assertional logic, name' J BCK.

**Proof.** For the first assertion of the theorem, we have that PBCK is finitely congruential by Proposition 2.1.35, so S(PBCK, 0) is algebraisable with equivalent algebraic semantics PBCK<sub>e</sub> by Theorem 1.8.14. Because PBCK<sub>e</sub> coincides with BCK by Corollary 2.1.17, from Lemma 2.1.34 it follows that  $S(PBCK^D, 1)$  is algebraisable with equivalent algebraic semantics BCK<sup>D</sup>. For the second assertion of the theorem, we have that  $S(BCK^D, 1)$  coincides with BCK by Example 1.8.10, so  $S(BCK^D, 1)$  is algebraisable with equivalent algebraic semantics BCK<sup>D</sup> by Example 1.8.4. Hence  $S(PBCK^D, 1)$  and  $S(BCK^D, 1)$  have the same equivalent algebraic semantics, namely BCK<sup>D</sup>. Since  $S(PBCK^D, 1)$  and  $S(BCK^D, 1)$  also have the same defining equation (viz.,  $p \approx 1$ ), they are identical by Lemma 1.8.8. Hence PBCK and BCK have the same assertional logic, namely BCK.

Remark 2.1.37. It should not be supposed that PBCK is the largest quasivariety K with language  $\langle \rightarrow, 1 \rangle$  of type  $\langle 2, 0 \rangle$  such that  $\models_{M(K,1)} = \models_{M(BCK^D,1)}$ . Indeed, by Kabziński [130, Fact 3] the greatest element of  $\Lambda^Q(K)$  is the quasivariety axiomatised by the following identities and quasi-identity:

 $(x \to y) \to ((z \to x) \to (z \to y)) \approx 1$  $(y \to (z \to x)) \to (z \to (y \to x)) \approx 1$  $x \to (y \to x) \approx 1$ 

$$x \to \mathbf{1} \approx \mathbf{1} \& x \to y \approx \mathbf{1} \supset y \approx \mathbf{1}$$

We remark that Kabziński's result is essentially a specialisation to  $S(PBCK^D, 1)$  of an observation due to Blok and Pigozzi [31, p. 16], which characterise the largest algeoraic semantics of any deductive system having an algebraic semantics.

Corollary 2.1.38. A quasi-identity of the form:

$$\bigotimes_{i=1}^{n} s_{i}(\vec{x}) \approx 0 \supset t(\vec{x}) \approx 0$$
(2.20)

is satisfied by PBCK iff it is satisfied by BCK. In particular, an identity of the form:

 $t(\vec{x}) \approx 0$ 

is satisfied by PBCK iff it is satisfied by BCK.

Proof. Let  $\{s_i(\vec{x}) : i = 1, ..., n\} \cup \{t(\vec{x})\}$  be  $\langle -, 0 \rangle$ -terms in the variables  $\vec{x}$ . Identify the  $\langle -, 1 \rangle$ -terms  $\{\eta(s_i(\vec{x})) : i = 1, ..., n\} \cup \{\eta(t(\vec{x}))\}$  (where  $\eta : T_{\mathcal{L}}(X) \to T_{\mathcal{L}^{\mathcal{D}}}(X)$  is the map of Lemma 2.1.34) in the variables  $\vec{x}$  with the  $\langle -, 1 \rangle$ -formulas  $\{\varphi_i(\vec{p}) : i = 1, ..., n\} \cup \{\psi(\vec{p})\}$  in the variables  $\vec{p}$ . Given this notation, from Theorem 2.1.36 and repeated application of Lemma 2.1.34 we have the following string of equivalences:

$$\begin{aligned} \mathsf{BCK} &\models \&_{i=1}^n s_i(\vec{x}) \approx 0 \supset t(\vec{x}) \approx 0 \quad \text{iff} \\ &\{s_i(\vec{x}) \approx \mathbf{0} : i = 1, \dots, n\} \models_{\mathsf{BCK}} t(\vec{x}) \approx \mathbf{0} \quad \text{iff} \\ &\{\eta(s_i(\vec{x})) \approx \mathbf{1} : i = 1, \dots, n\} \models_{\mathsf{BCK}^D} \eta(t(\vec{x})) \approx \mathbf{1} \quad \text{iff} \\ &\{\varphi_i(\vec{p}) : i = 1, \dots, n\} \vdash_{\mathsf{S}(\mathsf{BCK}^D, \mathbf{1})} \psi(\vec{p}) \quad \text{iff} \\ &\{\varphi_i(\vec{p}) : i = 1, \dots, n\} \vdash_{\mathsf{S}(\mathsf{PBCK}^D, \mathbf{1})} \psi(\vec{p}) \quad \text{iff} \\ &\{\eta(s_i(\vec{x})) \approx \mathbf{1} : i = 1, \dots, n\} \models_{\mathsf{PBCK}^D} \eta(t(\vec{x})) \approx \mathbf{1} \quad \text{iff} \\ &\{s_i(\vec{x}) \approx \mathbf{0} : i = 1, \dots, n\} \models_{\mathsf{PBCK}} t(\vec{x}) \approx \mathbf{0} \quad \text{iff} \\ &\mathsf{PBCK} \models \&_{i=1}^n s_i(\vec{x}) \approx \mathbf{0} \supset t(\vec{x}) \approx \mathbf{0}. \end{aligned}$$

Therefore  $BCK \models (2.20)$  iff  $PBCK \models (2.20)$ , which establishes the first assertion of the theorem. The second assertion now follows trivially by specialisation.

The following corollary seems first to have been proved (in a slightly more general form) using algebraic methods by van Alten and Raftery [231]. See also Raftery and van Alten [193, Corollary 10].

Corollary 2.1.39. (cf. [231, Corollary 3.4]) A quasi-identity of the form:

$$\bigotimes_{i=1}^{n} s_{i}(\vec{x}) \approx 0 \supset t(\vec{x}) \approx 0$$
(2.20)

is satisfied by the varietal closure H(BCK) of the quasivariety of BCK-algebras iff it satisfied by BCK. In particular, an identity of the form:

 $t(\vec{x}) pprox \mathbf{0}$ 

is satisfied by H(BCK) iff it is satisfied by BCK.

Proof. Suppose  $H(BCK) \models (2.20)$ . Because  $BCK \subseteq H(BCK)$  we deduce  $BCK \models (2.20)$ . Conversely, suppose  $BCK \models (2.20)$ . From Theorem 2.1.38 we infer  $PBCK \models (2.20)$ ; since  $H(BCK) \subseteq PBCK$  (by Komori [138, Theorem 7]), we deduce  $H(BCK) \models (2.20)$  as required. The second assertion follows trivially by specialisation.

Recall that, for a quasivariety K, a relative subvariety of K is a quasivariety K' such that  $K' = K \cap V(K'')$  for some  $K'' \subseteq K$ . Let S be an algebraisable deductive system with equivalent algebraic semantics K. By Blok and Jónsson [28, Lecture 6, p. 4, Theorem 1.4], the axiomatic extensions of S are in one-toone correspondence with the relative subvarieties of K. In more detail, let S' be an axiomatic extension of S. By Blok and Pigozzi [31, Corollary 4.9], S' is algebraisable, say with equivalent quasivariety semantics K', and by Theorem 1.8.3(1)-(2), (3)-(4), it follows that K' is a relative subvariety of K. Conversely, it follows from (1.51) and Theorem 1.8.3(1)-(2), (3)-(4) that if K'' is any relative subvariety of K, then there exists an axiomatic extension S'' of S whose equivalent quasivariety semantics is K''. See also van Alten [229, Section 4.4; p. 78].

From the preceding discussion and Theorem 2.1.36, it follows that the inherent assertional logic S(V, 0) of any subvariety V of the variety of pre-BCK-algebras coincides with an axiomatic extension of BCK (in fact, with the axiomatic extension of BCK arising as the intrinsic assertional logic of the quasivariety of BCK-algebras axiomatised relative to V by (2.5)—see Proposition 2.2.4 below). Hence the inherent assertional logic of any variety of pre-BCK-algebras has a familiar description (recall Example 1.8.4). Because of this observation, we shall dismiss from further consideration the assertional logics of those varieties of pre-BCK-algebras we encounter in the sequel.

2.1.40. Quasi-Bounded Pre-BCK-Algebras. A maximal element of a pre-BCK-algebra A is an element  $m \in A$  such that  $a \leq m$  for all  $a \in A$ . When they exist maximal elements form an equivalence class under  $\Xi$  called the maximal class; cf. [147, Section 1.4]. Let A be a pre-BCK-algebra with maximal class M. The algebra  $A^1 := \langle A; \div, 0, 1 \rangle$  obtained from A upon enriching the language of A with a new nullary operation symbol 1 whose canonical interpretation on  $A^1$  is a fixed  $1 \in M$  is called a quasi-bounded pre-BCK-algebra. Clearly the class PBCK<sup>1</sup> of quasi-bounded pre-BCK-algebras is a variety, axiomatised relative to the variety of pre-BCK-algebras by the identity  $x \div 1 \approx 0$ .

Remark 2.1.41. In passing from a given pre-BCK-algebra  $\mathbf{A}$  with maximal class M to a quasi-bounded pre-BCK-algebra  $\mathbf{A}^1$  there is in general no natural choice of maximal element  $1 \in M$ . Indeed, it seems plausible that distinct choices of maximal element give rise to non-isomorphic quasi-bounded pre-BCK-algebras, although we have no proof of this.

For a pre-BCK-algebra A, the derived operation  $a \sqcap b := a(ab)$  is called the *pre-BCK meet* (briefly, *meet*) of  $a, b \in A$ .) Given a quasi-bounded pre-BCK-algebra A<sup>1</sup>, the *pre-BCK complement* (briefly, *complement*) of  $a \in A$  is  $a^* := 1a$ , while the *pre-BCK join* of  $a, b \in A$  (briefly, *join*) is  $a \sqcup b := (a^* \sqcap b^*)^*$ . The three derived operations \*,  $\sqcap$  and  $\sqcup$  play an important role in the sequel; the two trivial lemmas below, whose proofs are omitted, summarise some elementary properties of these operations that will be needed subsequently.

Lemma 2.1.42. The variety of pre-BCK-algebras satisfies the following identities:

| $x \sqcap 0 pprox 0$ | (2.21) |
|----------------------|--------|
|                      |        |

| $x \sqcap x \approx x.$ | (2.23) |
|-------------------------|--------|

Moreover, if A is a pre-BCK-algebra the following statements hold for any  $a, b \in A$ :

1. ab = 0 iff  $a \sqcap b = a$ ;

 $\mathbf{0}\sqcap x\approx\mathbf{0}$ 

2.  $a \sqcap b$  is a lower bound of  $\{a, b\}$ .

Lemma 2.1.43. (cf. [126, Proposition 2]; cf. [126, Corollary 1]) The variety of quasi-bounded pre-BCK-algebras satisfies the following identities:

| $x \sqcap 1 \approx x$      |   | (2.24) |
|-----------------------------|---|--------|
| $1 \sqcap x \approx x^{**}$ |   | (2.25) |
| $x \sqcup x \approx x^{**}$ |   | (2.26) |
| $x \sqcup 1 \approx 1$      | - | (2.27) |
| $1 \sqcup x \approx 1.$     | • | (2.28) |

Moreover, if  $A^1$  is a quasi-bounded pre-BCK-algebra the following statements hold for any  $a, b \in A$ :

a\*\* ≤ a;
 a\*b\* ≤ ba;
 a ≤ b implies b\* ≤ a\*;

4.  $a^*b \equiv b^*a;$ 

5.  $a^{***} \equiv a^*$ .

(2.22)

### 2.2 Varieties of Pre-BCK-Algebras

In this section we study varieties of pre-BCK-algebras, with a focus on the natural pre-BCK-algebraic counterparts of the varieties of commutative, positive implicative and implicative BCK-algebras.

Arbitrary varieties of pre-BCK-algebras are briefly investigated in §2.2.1. For any variety V of BCK-algebras, the natural pre-BCK-algebraic counterpart of V is the class  $\{A \in PBCK : A/\Xi \cong B \text{ for some } B \in V\}$ . In the main result of the section, it is shown that the natural pre-BCK-algebraic counterpart of any variety of BCK-algebras is itself always a variety. We also show that any variety of pre-BCK-algebras containing the 3-element pre-BCK-algebra  $B_2$  of Example 2.1.5 fails to enjoy many of the properties typically associated with a 'tractable' class of algebras.

In §2.2.9 the variety of commutative pre-BCK-algebras is studied as the natural pre-BCK-algebraic counterpart of the variety of commutative BCK-algebras. In particular, it is shown that the commutative pre-BCK-algebras are characterised among the pre-BCK-algebras by means of a certain natural condition on the pre-BCK quasiordering.

In §2.2.16 we investigate the variety of positive implicative pre-BCK-algebras, the natural pre-BCK-algebraic counterpart of the variety of positive implicative BCK-algebras. It is proved that a variety of pre-BCK-algebras is a variety of positive implicative pre-BCK-algebras iff it is subtractive, weakly congruence orderable and has EDPI (witness x - y). For any positive implicative pre-BCK-algebra, we also give an internal characterisation of dual relative pseudocomplementation in the join semilattice of compact ideals.

In §2.2.28 the variety of implicative pre-BCK-algebras is considered as the natural pre-BCK-algebraic counterpart of the variety of implicative BCK-algebras. The variety of implicative pre-BCK-algebras is characterised as precisely the intersection of the varieties of commutative and positive implicative pre-BCKalgebras. For a suitable notion of prime ideal, we also show that an ideal I of an implicative pre-BCK-algebra **A** is prime iff it is maximal iff it is irreducible iff  $A/\phi_I$  is isomorphic to the 2-element implicative BCK-algebra  $C_1$ .

2.2.1. Varieties of Pre-BCK-Algebras. With any variety V of BCKalgebras, we may naturally associate a class  $V_s := \{A \in PBCK : A/\Xi \cong B \text{ for} some B \in V\}$  of pre-BCK-algebras. We call V, the *natural pre-BCK-algebraic* counterpart of V. Because of the following theorem of Blok and Raftery [38, Section 4] (see also Raftery and Sturm [191, Corollary 2.8], Idziak [114, Theorem 1], van Alten [229, Proposition 4.4] and Blok and la Falce [25, Theorem 4.3]), which characterises syntactically those subclasses of PBCK that are varieties of BCK-algebras, the class V, is always a variety. Henceforth, for any algebra with language  $\langle \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$ , we denote by  $x \div \prod_{i=1}^{n} u_i(\vec{x})$ the term  $(\cdots (x \div u_1(\vec{x})) \cdots) \div u_n(\vec{x})$ , where  $n \in \omega$  and  $u_1, \ldots, u_n$  are  $\langle \div, 0 \rangle$ terms in the variables  $\vec{x}$ .

**Theorem 2.2.2.** [38, Corollary 10]; [191, Corollary 2.8] Let K be a class of algebras with language  $\langle \div, \mathbf{0} \rangle$  of type  $\langle 2, \mathbf{0} \rangle$ . Then  $\mathbf{V}(K)$  is a BCK-variety iff  $K \subseteq PBCK$  and K satisfies some identity:

$$x - \prod_{i=1}^{n} u_i(x, y) \approx y - \prod_{j=1}^{m} v_j(x, y)$$
 (2.29)

for fixed  $n, m \in \omega$  and  $\langle -, 0 \rangle$ -terms  $u_1, \ldots, u_n, v_1, \ldots, v_m$  such that BCK satisfies:

 $u_i(x,x) \approx 0 \approx v_j(x,x)$   $i = 1, \ldots, n, \quad j = 1, \ldots, m.$ 

In this case, V(K) is congruence 3-permutable.

**Theorem 2.2.3.** Let V be a variety of BCK-algebras, axiomatised relative to PBCK by some identity:

$$x - \prod_{i=1}^{n} u_i(x, y) \approx y - \prod_{j=1}^{m} v_j(x, y)$$
 (2.29)

for fixed  $n, m \in \omega$  and (-, 0)-terms  $u_1, \ldots, u_n, v_1, \ldots, v_m$  such that BCK

satisfies:

$$u_i(x,x) \approx 0 \approx v_j(x,x)$$
  $i = 1, \ldots, n, \quad j = 1, \ldots, m.$ 

Then V, is a variety, axiomatised relative to PBCK by the pair of identities:

$$\left(x \stackrel{\cdot}{-} \prod_{i=1}^{n} u_i(x, y)\right) \stackrel{\cdot}{-} \left(y \stackrel{\cdot}{-} \prod_{j=1}^{m} v_j(x, y)\right) \approx \mathbf{0}$$

$$(2.30)$$

and:

$$\left(y - \prod_{j=1}^{m} v_j(x, y)\right) - \left(x - \prod_{i=1}^{n} u_i(x, y)\right) \approx 0.$$
 (2.31)

Proof. Let V be a variety of BCK-algebras axiomatised relative to PBCK by the identity (2.29) for a given choice of  $n, m \in \omega$  and  $\langle -, 0 \rangle$ -terms  $u_1, \ldots, u_n$ ,  $v_1, \ldots, v_m$  and let W be the variety of pre-BCK-algebras axiomatised relative to PBCK by the pair of identities (2.30)-(2.31) for the same choice of  $n, m \in \omega$ and  $\langle -, 0 \rangle$ -terms  $u_1, \ldots, u_n, v_1, \ldots, v_m$ . To prove the theorem it suffices to show V, coincides with W.

Let  $A \in V_3$ . Then  $A/\Xi \cong B$  for some  $B \in V$ . Now  $B \models (2.29)$ , so  $B \models (2.30), (2.31)$  by (1.3). Therefore  $A/\Xi \models (2.30), (2.31)$ , because identities are preserved by isomorphic copies. But then  $A \models (2.30), (2.31)$  by Corollary 2.1.16, so  $A \in W$ . Hence  $V_3 \subseteq W$ . For the converse, let  $A \in W$ . Then  $A/\Xi \models (2.30), (2.31)$ , so  $A/\Xi \models (2.29)$  by (2.5). Thus  $A \in V_3$  and  $W \subseteq V_3$ . Hence  $V_3 = W$ , and the proof is complete.

Recall from §1.8.9 that, for any variety V with normal ideals, the class  $V_{\epsilon}$  of reduced algebras of V is  $\{\mathbf{B} : \mathbf{B} \cong \mathbf{A}/\langle 0 \rangle_{\mathbf{A}}^{\epsilon}$  for some  $\mathbf{A} \in V\}$ . Because of Theorem 2.1.14(4), if V is a variety of pre-BCK-algebras then  $V_{\epsilon} = \{\mathbf{B} : \mathbf{B} \cong \mathbf{A}/\Xi_{\mathbf{A}} \text{ for some } \mathbf{A} \in V\}$ . Moreover, one easily sees in this case that  $(V_{\epsilon})_{\mathfrak{I}} = V$ . That is,  $V_{\epsilon}$  may be understood as the *natural BCK-algebraic counterpart* of V. Consequently, the following proposition (which is a specialisation of a fragment of a more general theorem for subtractive varieties due to Agliano and Ursini [10]) can be considered as a kind of a converse of Theorem 2.2.3.

**Proposition 2.2.4.** (cf. [10, Theorem 3.13]) Let V be a variety of pre-BCKalgebras. Then  $V_{\epsilon}$  is the quasivariety of BCK-algebras axiomatised relative to V by the quasi-identity:

$$x - y \approx 0 \& y - x \approx 0 \supset x \approx y.$$
(2.5)

*Proof.* Let V be a variety of pre-BCK-algebras. By Proposition 2.1.35, V is finitely congruential witness  $D := \{x \div y, y \div x\}$ . Therefore  $I^D = I^{\epsilon}$  for any  $I \in I(\mathbf{A})$  and  $\mathbf{A} \in V$ . It follows that:

$$\mathbf{A} \in \mathsf{V}_{\epsilon}$$
 iff  $\langle 0 \rangle_{\mathbf{A}}^{\epsilon} = \omega_{\mathbf{A}}$   
iff  $a \stackrel{\cdot}{-}{}^{\mathbf{A}} b = 0 = b \stackrel{\cdot}{-}{}^{\mathbf{A}} a$  implies  $a = b$ 

for any  $a, b \in A$ . Hence  $V_{\epsilon}$  is exactly the quasivariety of BCK-algebras axiomatised relative to V by the quasi-identity (2.5).

Let V be a variety of BCK-algebras with natural pre-BCK-algebraic counterpart V<sub>2</sub>. Because of Theorem 2.2.3 and the definition of V<sub>2</sub>, Corollary 2.1.14 and Corollary 2.1.38 together indicate that the first-order theory of V<sub>2</sub> stands in relation to V as the first-order theory of PBCK stands in relation to BCK; support for this contention is provided by our study of the natural pre-BCKalgebraic counterparts of the varieties of commutative, positive implicative and implicative BCK-algebras in §2.2.9, §2.2.16 and §2.2.28 respectively in the sequel.

The preceding remarks and Corollary 2.1.31 notwithstanding, in general the second-order theory of V, bears little resemblance to the second content theory of V. This is exemplified by the final result (Corollary 2.2.6 below) of this brief subsection, which shows that any variety of pre-BCK-algebras that contains the 3-element pre-BCK-algebra  $\mathbf{B}_2$  of Example 2.1.5 is not 0-regular, congruence distributive or congruence *n*-permutable for any  $n \geq 2$  (contrast this result with Theorem 1.6.10). The corollary obtains as an easy consequence of the following proposition of Blok and Raftery [38, Section 4], which shows BCK is the splitting quasivariety associated with the algebra  $\mathbf{B}_2$  in PBCK. In the statement of the proposition and in the sequel, for any quasivariety K we

denote by  $\Lambda^Q(K)$  the 'lattice of subquasivarieties' of K, namely the dual of the lattice of corresponding implicational theories of K.

**Proposition 2.2.5.** [38, Proposition 2] The pair  $\langle \mathbf{Q}(\mathbf{B}_2), \mathsf{BCK} \rangle$  splits the lattice  $\Lambda^Q(\mathsf{PBCK})$ . That is, for every quasivariety  $\mathsf{K} \subseteq \mathsf{PBCK}$ , either  $\mathsf{K} \subseteq \mathsf{BCK}$  or  $\mathbf{Q}(\mathbf{B}_2) \subseteq \mathsf{K}$ .

*Proof.* Let  $K \in \Lambda^Q(PBCK)$  with  $K \not\subseteq BCK$ . By hypothesis,  $K \not\models (2.5)$ , so there exists an algebra  $A \in K$  with elements  $a, b \in A$  such that ab = 0 = ba, but  $a \neq b$ . Then  $\{0, a, b\}$  is the universe of a subalgebra of A isomorphic to  $B_2$ . Hence  $B_2 \in K$  and  $Q(B_2) \subseteq K$ .

Corollary 2.2.6. (cf. [25, Corollary 3.5]) For any variety V of pre-BCKalgebras the following assertions hold:

- 1. V is 0-regular iff  $V \subseteq BCK$ ;
- 2. V is congruence distributive iff  $V \subseteq BCK$ ;
- V is congruence n-permutable for some n ≥ 2 iff V ⊆ BCK. In particular, if V ⊆ BCK, then V is congruence 3-permutable (and not congruence permutable).

Proof. Suppose  $V \subseteq BCK$ . Then V is both 0-regular and congruence distributive by Theorem 1.6.10(2); also V is congruence *n*-permutable for n = 3 by Theorem 2.2.2. However, V is not congruence permutable, since the smallest non-trivial variety of BCK-algebras (namely, the variety of implicative BCKalgebras) is not congruence permutable (by results due to Mitschke [167, Theorem 2] and Blok and Pigozzi [34, pp. 583-584]). For the converse, suppose  $V \not\subseteq BCK$ . Then  $\mathbf{Q}(\mathbf{B}_2) \subseteq V$  by Proposition 2.2.5 and therefore  $\mathbf{V}(\mathbf{B}_2) \subseteq V$ . But  $\mathbf{V}(\mathbf{B}_2)$  is not 0-regular (since  $\mathbf{B}_2$  itself is not 0-regular) and is neither congruence distributive nor congruence *n*-permutable for any  $n \ge 2$  (by Blok and Raftery [38, Proposition 3]). Hence V is not 0-regular, congruence distributive or congruence *n*-permutable for any  $n \ge 2$ .

Remark 2.2.7. In their paper on the lattice of subquasivarieties of BCKalgebras, Blok and Raftery report that [38, Corollary 4]:

- 1. BCK is the largest subquasivariety of PBCK that is BCK-0-regular; and
- 2. BCK is the largest subquasivariety of PBCK that is BCK-congruence distributive or that satisfies any non-trivial BCK-congruence identity.

Unfortunately, the proofs provided for both these claims appear flawed by an ambiguous usage of the term 'relative congruence' [24]. We have been unable to exhibit an alternative proof that establishes these claims.

Problem 2.2.8. Is BCK the largest subquasivariety of PBCK that is BCK-0-regular, BCK-congruence distributive or that satisfies any non-trivial BCKcongruence identity?

2.2.9. Commutative Pre-BCK-Algebras. By a commutative pre-BCKalgebra we mean a pre-BCK-algebra A such that  $A/\Xi \cong B$  for some commutative BCK-algebra B. By Yutani [245], the class of all commutative BCKalgebras is a variety, axiomatised relative to PBCK by the identity:

$$x - (x - y) \approx y - (y - x) \tag{2.32}$$

so by Theorem 2.2.3, the class cPBCK of all commutative pre-BCK-algebras is also a variety, axiomatised relative to PBCK by the identity:

$$\left(x - (x - y)\right) - \left(y - (y - x)\right) \approx 0 \tag{2.33}$$

since (2.32) is of the form of (2.29) and is symmetric in the individual variables x and y. Thus we have the following result:

**Theorem 2.2.10.** An algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a commutative pre-BCK-algebra iff the following identities are satisfied:

$$\left((x - y) - (x - z)\right) - (z - y) \approx 0 \tag{2.1}$$

$$\mathbf{0} - \mathbf{x} \approx \mathbf{0} \tag{2.3}$$

 $x - 0 \approx x \tag{2.6}$ 

2.2. Varieties of Pre-BCK-Algebras

$$\left(x - (x - y)\right) - \left(y - (y - x)\right) \approx 0. \tag{2.33}$$

Thus the class cPBCK of commutative pre-BCK-algebras is a variety.

Recall that, by definition, a BCK-algebra A is commutative if its underlying poset  $\langle A; \leq \rangle$  is a meet semilattice, or equivalently, if glb $\{a, b\}$  exists for any  $a, b \in A$ .

**Proposition 2.2.11.** Let A be a pre-BCK-algebra. If A is commutative, then for every  $a, b \in A$ , both  $a \sqcap b$  and  $b \sqcap a$  are greatest lower bounds of  $\{a, b\}$ with respect to the underlying pre-BCK quasiorder  $\preceq$ . Conversely, if for every  $a, b \in A$ , both  $a \sqcap b$  and  $b \sqcap a$  are greatest lower bounds of  $\{a, b\}$  with respect to the underlying pre-BCK-quasiorder  $\preceq$ , then A is commutative.

*Proof.* Let A be a pre-BCK-algebra and let  $a, b \in A$ .

Suppose A is commutative. By Lemma 2.1.42(2),  $a \sqcap b$  is a lower bound of  $\{a, b\}$ . Suppose  $c \preceq a$  and  $c \preceq b$  for some  $c \in A$ . We have:

| 0 = (c(ca))(a(ac)) | by (2.33)        |
|--------------------|------------------|
| = (c0)(a(ac))      | as $c \preceq a$ |
| = c(a(ac))         | by $(2.6)$       |

whence  $c \leq a \sqcap c$ . Put  $\alpha := a, \beta := ac$  and  $\gamma := ab$ . Since:

| $0 = \big((ab)(ac)\big)(cb)$ | by $(2.1)$       | •   |        |
|------------------------------|------------------|-----|--------|
| =((ab)(ac))0                 | as $c \preceq b$ |     |        |
| =(ab)(ac)                    | by (2.6)         | . • | (2.34) |

we have:

$$0 = ((\alpha\beta)(\alpha\gamma))(\gamma\beta) \qquad by (2.1)$$
$$= ((\alpha\beta)(\alpha\gamma))((ab)(ac))$$
$$= ((\alpha\beta)(\alpha\gamma))0 \qquad by (2.34)$$
$$= (a(ac))(a(ab)) \qquad by (2.6)$$
whence  $a \sqcap c \preceq a \sqcap b$ . Thus  $c \preceq a \sqcap c \preceq a \sqcap b$ . By transitivity,  $c \preceq a \sqcap b$ , and so  $a \sqcap b$  is a greatest lower bound of  $\{a, b\}$ . By (2.33),  $a \sqcap b \equiv b \sqcap a$ , so  $b \sqcap a$ lies in the same clique as  $a \sqcap b$ . By Lemma 1.2.3(1),  $b \sqcap a$  is a greatest lower bound of  $\{a, b\}$ .

Conversely, suppose that for every  $a, b \in A$ , both  $a \sqcap b$  and  $b \sqcap a$  are greatest lower bounds of  $\{a, b\}$ . By Lemma 1.2.3(2) we have that  $a \sqcap b \equiv b \sqcap a$ , whence (a(ab))(b(ba)) = 0. Thus A is commutative.

**Remark 2.2.12.** Let A be a pre-BCK-algebra. For A to be commutative, the requirement that both  $a \sqcap b$  and  $b \sqcap a$  be greatest lower bounds of  $a, b \in A$  cannot be weakened to the requirement that just one of  $a \sqcap b$  or  $b \sqcap a$  be a greatest lower bound for  $\{a, b\}$ . To see this, consider the pre-BCK-algebra A with operation table:

| <u>.</u> A | 0 | 1 | 2 | 3 |
|------------|---|---|---|---|
| 0          | 0 | 0 | 0 | 0 |
| 1          | 1 | 0 | 0 | 0 |
| 2          | 2 | 2 | 0 | 2 |
| 3          | 3 | 0 | 0 | 0 |

An easy sequence of checks shows  $a \sqcap b$  is a greatest lower bound of any  $a, b \in \{0, 1, 2, 3\}$ . However, A is not commutative, as  $(1 \div (1 \div 2)) \div (2 \div (2 \div 1)) = (1 \div 0) \div (2 \div 2) = 1 \div 0 = 1 \neq 0$ .

Let A be a commutative pre-BCK-algebra and let  $m \in A$  be fixed. For  $a, b \in A$  such that  $a, b \leq m$ , we have  $ab \leq a \leq m$  by Lemma 2.1.12(3). Thus the principal order ideal (m] is a subuniverse of A. We write (m] for the subalgebra of A with subuniverse (m]. By definition,  $a \leq m$  for all  $a \in (m]$ . Thus m is a maximal element of (m], the restriction  $\Xi|_{(m)}$  of  $\Xi$  to (m] has a maximal class M and (m] induces a quasi-bounded pre-BCK-algebra (m]<sup>1</sup>. In general, there may be many possible choices for an element  $c \in M$  such that  $1^{(m)^1} = c$ . However, we will always fix  $1^{(m)^1} := m$ , whence we will not be careful to distinguish between (m] and (m]<sup>1</sup> in the sequel.

Let A be a commutative pre-BCK-algebra and let  $m \in A$  be fixed. Since (m]

is a quasi-bounded pre-BCK-algebra, it has (by the remarks of §2.1.40) both a derived 'complementation' operation \* and a derived 'join' operation  $\sqcup$ . To signify these derived operations are local to (m], we write  $a^{*(m)}$  and  $a \sqcup^{(m)} b$ for the complement of  $a \in (m]$  and the join of  $a, b \in (m]$  respectively. Further, to simplify notation in the sequel we write  $a^{*}_{(m)}$  for  $a^{*(m)}$ ,  $a^{**}_{(m)}$  for  $(a^{*}_{(m)})^{*}_{(m)}$  and  $a^{***}_{(m)}$  for  $((a^{*}_{(m)})^{*}_{(m)})^{*}_{(m)}$ .

**Lemma 2.2.13.** (cf. [126, Proposition 2]; cf. [126, Corollary 1]) Let  $A^1$  be a quasi-bounded commutative pre-BCK-algebra. For any  $a \in A$ ,  $a^{**} \equiv a$ .

*Proof.* Suppose  $A^1$  is a commutative pre-BCK-algebra. Let  $a \in A$ . By (2.2) we have  $a^{**} = 1(1a) \preceq a$ . By (2.33) we have also that (a(a1))(1(1a)) = 0, whence a(1(1a)) = 0 since  $a \preceq 1$ . Thus  $a \preceq 1(1a)$ , so  $a \preceq a^{**}$ . Hence  $a^{**} \equiv a$ .

If A is a commutative BCK-algebra then, by the remarks of §1.6.11, the underlying poset  $\langle A; \leq \rangle$  is not merely a meet semilattice; it is in fact a (distributive) nearlattice. Let  $\langle A; \leq \rangle$  be a meet semilattice. By definition,  $\langle A; \leq \rangle$  is a nearlattice if, for every  $m \in A$ , the principal order ideal (m] is a lattice. Equivalently,  $\langle A; \leq \rangle$  is a nearlattice if it enjoys the upper bound property: that is, if the supremum lub $\{a, b\}$  exists when  $a, b \in A$  share a common upper bound [64, Section 3, p. 487].

Lemma 2.2.14. (cf. [126, p. 9]) Let A be a commutative pre-BCK-algebra and let  $m \in A$  be fixed. If  $a, b \leq m$  then  $a \sqcup^{(m)} b$  is a-least upper bound of a and b in (m].

*Proof.* Assume A, a, b, m are as stated. By Proposition 2.2.11,  $a_{(m)}^* \sqcap b_{(m)}^* \preceq a_{(m)}^*, b_{(m)}^*, \text{ so } a_{(m)}^{**}, b_{(m)}^{**} \preceq (a_{(m)}^* \sqcap b_{(m)}^*)_{(m)}^*$  by Lemma 2.1.43(3). Now  $a \equiv a_{(m)}^{**}$  and  $b \equiv b_{(m)}^{**}$  by Lemma 2.2.13, so  $a \preceq a_{(m)}^{**} \preceq (a_{(m)}^* \sqcap b_{(m)}^*)_{(m)}^*$  and  $b \preceq b_{(m)}^{**} \preceq (a_{(m)}^* \sqcap b_{(m)}^*)_{(m)}^*$ . By transitivity,  $a, b \preceq (a_{(m)}^* \sqcap b_{(m)}^*)_{(m)}^*$ . Thus  $(a_{(m)}^* \sqcap b_{(m)}^*)_{(m)}^*$  is an upper bound of  $\{a, b\}$ .

Let  $c \leq m$  for some  $c \in A$  with  $a, b \leq c$ . By Lemma 2.1.43(3) we have  $c^*_{(m]} \leq a^*_{(m]}, b^*_{(m]}$ . Thus  $c^*_{(m]} \leq a^*_{(m]} \sqcap b^*_{(m]}$  by Proposition 2.2.11 and so  $(a^*_{(m]} \sqcap b^*_{(m)})^*_{(m]} \leq c^{**}_{(m]}$  by Lemma 2.1.43(3). Because  $c^{**}_{(m]} \equiv c$  by Lemma 2.2.13, we

have that  $(a_{(m]}^* \sqcap b_{(m]}^*)_{(m]}^* \preceq c_{(m]}^{**} \preceq c$ . By transitivity,  $(a_{(m]}^* \sqcap b_{(m]}^*)_{(m]}^* \preceq c$ , whence  $(a_{(m]}^* \sqcap b_{(m)}^*)_{(m]}^*$  is a least upper bound of  $\{a, b\}$ .

Because of Proposition 2.2.11 and Lemma 2.2.14, it is natural to ask if each initial segment  $(m], m \in A$ , of a commutative pre-BCK-algebra A supports (in some sense) a 'generalised lattice' structure, and in particular if  $\langle (m]; \Box, \sqcup^{(m)} \rangle$  is a non-commutative lattice (in the sense of §1.4.1). Let  $A^1$  be the 5-element quasi-bounded commutative pre-BCK-algebra (with derived pre-BCK meet  $\Box^{A^1}$ ) defined by the following operation tables:

| . A <sup>1</sup> | 9 | a | b | с | 1 | $\Box^{\mathbf{A^1}}$ | 0 | a | b | C   | 1 |
|------------------|---|---|---|---|---|-----------------------|---|---|---|-----|---|
| 0                | 0 | 0 | 0 | 0 | 0 | 0                     | 0 | 0 | 0 | 0   | 0 |
| a                | a | 0 | 0 | 0 | 0 | a                     | 0 | a | a | a   | a |
| b                | 6 | 0 | 0 | 0 | 0 | b                     | 0 | b | b | b   | b |
| с                | c | 0 | 0 | C | 0 | С                     | 0 | с | с | · C | с |
| 1                | 1 | b | ٤ | a | 0 | 1                     | 0 | с | a | b   | 1 |

Observe  $1 \sqcap (1 \sqcap a) = 1 \sqcap c = b \neq c = 1 \sqcap a = (1 \sqcap 1) \sqcap a$ , whence  $\langle A; \sqcap \rangle$  is not a band. Thus in particular  $\langle A; \sqcap, \sqcup \rangle$  is not a double band and the induced algebra  $\langle A; \sqcap, \sqcup \rangle$  is not a non-commutative lattice.

Remark 2.2.15. Let A be a commutative pre-BCK-algebra and let  $m \in A$  be fixed. The preceding remarks notwithstanding, (m] does support a 'generalised lattice' structure in a sense made precise as follows. Recall from [223, Section 10.1] that a [quasi-bounded] quasi-lattice (in the sense of Ursini) is a structure  $[\langle A; \land, \lor, 0, 1; \preceq \rangle] \langle A; \land, \lor; \preceq \rangle$  of type  $[\langle 2, 2, 0, 0, 2 \rangle] \langle 2, 2, 2 \rangle$  such that the following conditions hold for any  $a, b \in A$ :

QL1.  $\langle A; \preceq \rangle$  is a quasiordered set;

QL2. The quasi-meet  $c \wedge b$  of a and b satisfies:

- (a)  $a \wedge b \preceq a$ ;
- (b)  $a \wedge b \preceq b$ ;
- (c) For all  $c \in A$ ,  $a \preceq c \land b \preceq c$  implies  $a \land b \preceq c$ ;

## 2.2. Varieties of Pre-BCK-Algebras

QL3. The quasi-join  $a \lor b$  of a and b satisfies:

- (a)  $a \preceq a \lor b$ ;
- (b)  $b \preceq a \lor b$ ;
- (c) For all  $c \in A$ ,  $c \preceq a \land c \preceq b$  implies  $c \preceq a \lor b$ ;

QL4. [For all  $c \in A$ ,  $0 \leq c \leq 1$ .]

[Quasi-bounded] quasi-lattices were introduced by Ursini in his study of algebraic semantics for linear logic [223], in order to provide a model for the 'turbo monoids' of Girard [97, p. 24]. By remarks due to Ursini [223, Section 10.1], the equivalence  $\cong$  on  $A \times A$  induced by  $\preceq$  in the sense of Lemma 1.2.2 is a congruence on any [quasi-bounded] quasi-lattice  $[\langle A; \land, \lor, 0, 1; \preceq \rangle] \langle A; \land, \lor; \preceq \rangle$  such that  $[\langle A; \land, \lor, 0, 1; \preceq \rangle/\cong] \langle A; \land, \lor; \preceq \rangle/\cong$  is a [bounded] lattice. Thus every [quasi-bounded] quasi-lattice has the global outline of a [bounded] lattice (compare this statement with the Clifford-McLean theorem for quasilattices). However, a quasi-lattice (in the sense of Ursini) is not in general a quasilattice in the sense of this thesis (that is, in the sense of §1.4.1); the converse does obtain.

By a reduced [quasi-bounded] quasi-lattice we mean a [quasi-bounded] quasilattice  $[\langle A; \land, \lor, 0, 1; \preceq \rangle] \langle A; \land, \lor; \preceq \rangle$  such that  $a \cong b$  iff a = b for any  $a, b \in A$ . Clearly every reduced [quasi-bounded] quasi-lattice is a [bounded] lattice. A distributive [quasi-bounded] quasi-lattice is a [quasi-bounded] quasilattice  $[\langle A; \land, \lor, 0, 1; \preceq \rangle] \langle A; \land, \lor; \preceq \rangle$  whose reduced image  $[\langle A; \land, \lor, 0, 1; \preceq \rangle/\cong] \langle A; \land, \lor; \preceq \rangle/\cong$  is distributive. Now let A be a commutative pre-BCK-algebra and let  $m \in A$  be fixed. Then the induced structure  $\langle (m]; \sqcup^{(m]}, \sqcap, 0, m; \preceq|_{(m]} \rangle$  is a distributive quasi-bounded quasi-lattice, whose reduced image  $\langle (m]; \sqcup^{(m)}, \sqcap, 0, m; \preceq|_{(m]} \rangle/\Xi$  is precisely the bounded distributive sublattice  $(m/\Xi]_{A/\Xi}$  of the commutative BCK-algebra  $A/\Xi$ .

*Proof.* Let  $\langle A; \div, 0 \rangle$  be a commutative pre-BCK-algebra and let  $m \in A$  be fixed. Let  $a, b, c \in (m]$ . Clearly the restriction  $\preceq|_{(m]}$  of  $\preceq$  to (m] is a quasiorder on (m], whence Condition (QL1) above is satisfied.

To see  $\sqcap$  is a quasi-meet, observe first that  $a \sqcap b \preceq a$  and  $a \sqcap b \preceq b$ , just because  $a \sqcap b$  is a lower bound of  $\{a, b\}$ . Suppose  $a \preceq c \sqcap b \preceq c$ . Put  $\alpha := a$ ,  $\beta := c$  and  $\gamma := c(cb)$ . We have:

$$0 = ((\alpha\beta)(\alpha\gamma))(\gamma\beta) \qquad \text{by } (2.1)$$
  
=  $((\alpha\beta)(\alpha\gamma))((c(cb))c)$   
=  $((\alpha\beta)(\alpha\gamma))0 \qquad \text{by Lemma } 2.1.12(3)$   
=  $((\alpha\beta)(\alpha\gamma)) \qquad \text{by } (2.6)$   
=  $(ac)(a(c(cb)))$   
=  $(ac)0 \qquad \text{since } a \leq c \sqcap b$   
=  $ac \qquad \text{by } (2.6).$ 

Thus  $a \leq c$ . Since  $a \sqcap b \leq a$ , we have  $a \sqcap b \leq a \leq c$ , whence  $a \sqcap b \leq c$  by transitivity. Thus  $\sqcap$  is a quasi-meet and Condition (QL2) above is satisfied.

To see  $\sqcup^{(m)}$  is a quasi-join, notice first that both  $a \preceq a \sqcup^{(m)} b$  and  $b \preceq a \sqcup^{(m)} b$ , because  $a \sqcup^{(m)} b$  is an upper bound of  $\{a, b\}$  in (m]. Suppose  $c \preceq a \sqcap c \preceq b$ . Put  $\alpha := c, \beta := a$  and  $\gamma := a(ac)$ . We have:

$$0 = ((\alpha\beta)(\alpha\gamma))(\gamma\beta) \qquad \text{by } (2.1)$$
  
=  $((\alpha\beta)(\alpha\gamma))((a(ac))a)$   
=  $((\alpha\beta)(\alpha\gamma))0 \qquad \text{by Lemma } 2.1.12(3)$   
=  $(\alpha\beta)(\alpha\gamma) \qquad \text{by } (2.6)$   
=  $(ca)(c(a(ac)))$   
=  $(ca)0 \qquad \text{since } c \preceq a \sqcap c$   
=  $ca \qquad \text{by } (2.6).$ 

Thus  $c \leq a$ . Since  $a \leq a \sqcup^{(m)} b$ , we have  $c \leq a \leq a \sqcup^{(m)} b$ , whence  $c \leq a \sqcup^{(m)} b$  by transitivity. Thus  $\sqcup^{(m)}$  is a quasi-join and Condition (QL3) above is satisfied.

For Condition (QL4) above, just note  $0 \leq a \leq m$  for all  $a \in (m]$ . Thus the induced structure  $\langle (m]; \sqcup^{(m]}, \sqcap, 0, m; \leq |_{(m]} \rangle$  is a quasi-bounded quasi-lattice.

To complete the proof, just note the reduced image  $\langle (m]; \sqcap, \sqcup, 0, 1; \leq |_{(m)} \rangle / \Xi$ of  $\langle (m]; \sqcup^{(m)}, \sqcap, 0, m; \leq |_{(m)} \rangle$  may be identified with the bounded commutative BCK-subalgebra  $(m/\Xi]_{A/\Xi}$ , which is a bounded distributive sublattice of  $A/\Xi$ by the remarks of §1.6.11.

2.2.16. Positive Implicative Pre-BCK-Algebras. A positive implicative pre-BCK-algebra is a pre-BCK-algebra A such that  $A/\Xi \cong B$  for some positive implicative pre-BCK-algebra B. Since the class pBCK of all positive implicative BCK-algebras is a variety, axiomatised relative to PBCK by the identity [38, pp. 294-295]:

$$\left(x \stackrel{\cdot}{-} (x \stackrel{\cdot}{-} y)\right) \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x) \approx \left(y \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x)\right) \stackrel{\cdot}{-} (x \stackrel{\cdot}{-} y)$$
(2.35)

the class pPBCK of all positive implicative pre-BCK-algebras is also a variety, by Theorem 2.2.3, axiomatised relative to PBCK by the identity:

$$\left((x - (x - y)) - (y - x)\right) - \left((y - (y - x)) - (x - y)\right) \approx 0$$

since (2.35) is of the form of (2.29) and is symmetric in the variables x and y. This characterisation of pPBCK notwithstanding, the following axiomatisation is often more useful in practice.

**Theorem 2.2.17.** An algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a positive implicative pre-BCK-algebra iff the following identities are satisfied:

$$\left(\left(x - y\right) - \left(x - z\right)\right) - \left(z - y\right) \approx 0 \qquad (2.1)$$

$$\mathbf{0} - \mathbf{x} \approx \mathbf{0} \tag{2.3}$$

$$\dot{x} - \mathbf{0} \approx x$$
 (2.6)

$$(x - y) - ((x - y) - y) \approx 0.$$

$$(2.36)$$

Thus the class pPBCK of positive implicative pre-BCK-algebras is a variety.

*Proof.* Let A be a positive implicative pre-BCK-algebra. By definition we have that  $A \models (2.1), (2.3), (2.6)$ , so it only remains to show  $A \models (2.36)$ .

Since A is positive implicative,  $A/\Xi \cong B$  for some positive implicative BCKalgebra B. Because B is positive implicative,  $B \models (x \div y) \div y \approx x \div y$ , so  $B \models (2.36)$  by (1.3). As identities are preserved by isomorphic copies, we have that  $A/\Xi \models (2.36)$ , so  $A \models (2.36)$  by Corollary 2.1.16.

Conversely, suppose  $\mathbf{A} := \langle A; \div, 0 \rangle$  is an algebra of type  $\langle 2, 0 \rangle$  such that  $\mathbf{A} \models (2.1), (2.3), (2.6), (2.36)$ . By (2.1), (2.3), (2.6) and Lemma 2.1.6,  $\mathbf{A}$  is a pre-BCK-algebra. By (2.36),  $\mathbf{A}/\Xi \models (x \div y) \div ((x \div y) \div y) \approx 0$ ; also  $\mathbf{A}/\Xi \models ((x \div y) \div y) \div (x \div y) \approx 0$  by (1.30). Hence  $\mathbf{A}/\Xi \models (x \div y) \div y \approx x \div y$  by (1.5), so  $\mathbf{A}$  is positive implicative.

By Proposition 1.7.12, the variety pBCK of positive implicative BCK-algebras is precisely the class of all BCK-algebras with EDPI (witness x - y). By Corollary 2.2.6(1), therefore, pBCK is precisely the class of all 0-regular pre-BCKalgebras with EDPI (witness x - y). Since pBCK may be described alternatively as exactly the class of all 0-regular positive implicative pre-BCK-algebras (by Corollary 2.2.6(1) and Theorem 2.2.17), the preceding remarks suggest that pPBCK is a variety of pre-BCK-algebras with EDPI (witness x - y) (but which is not 0-regular). This last motivates the study of the ideal theory of positive implicative pre-BCK-algebras, the key to which is Lemma 2.2.19 below. See also Agliano [6, Section 4, p. 14].

**Lemma 2.2.18.** The variety of positive implicative pre-BCK-algebras satisfies the following identity:

$$\left((x - y) - (z - y)\right) - \left((x - z) - y\right) \approx 0.$$

$$(2.37)$$

*Proof.* Let A be a positive implicative pre-BCK-algebra. Then  $A/\Xi \cong B$  for some positive implicative BCK-algebra **B**. Since **B** is positive implicative,  $B \models ((x \div y) \div (z \div y)) \approx (x \div z) \div y$  by the remarks of §1.6.13, so  $B \models (2.37)$  by (1.3). Since identities are preserved by isomorphic copies,  $A/\Xi \models (2.37)$ , whence  $A \models (2.37)$  by Corollary 2.1.16.

Lemma 2.2.19. (cf. [68, Lemma 2.3]; cf. [195, Theorem II§1.8]) Let A be a positive implicative pre-BCK-algebra. For any ideal  $I \in I(A)$  and fixed

 $a \in A$ , the supremum  $I \vee^{\mathbf{I}(\mathbf{A})} \langle a \rangle_{\mathbf{A}}$  of I and  $\langle a \rangle_{\mathbf{A}}$  in the ideal lattice  $\mathbf{I}(\mathbf{A})$  is  $\{b \in A : ba \in I\}$ . Consequently,

$$ba \in I$$
 iff  $b \in I \vee^{\mathbf{I}(\mathbf{A})} \langle a \rangle_{\mathbf{A}}$ .

*Proof.* Let A be a positive implicative pre-BCK-algebra and let  $I \in I(A)$ . Let  $a \in A$  be fixed and let  $J := \{b \in A : ba \in I\}$ . Because I(A) is directed,  $I \vee^{I(A)} \langle a \rangle_{A} = \langle I \cup \{a\} \rangle_{A}$ , so to prove the lemma it is sufficient to show:

(i) J is an ideal of  $\mathbf{A}$ ;

(ii) J is the smallest ideal of A such that  $a \in J$  and  $I \subseteq J$ .

For (i), we have that  $0 \in J$  since  $0a = 0 \in I$  by (2.4). Suppose  $b, cb \in J$  for  $b, c \in A$ . Then  $ba, (cb)a \in I$  by definition of J. As  $((ca)(ba))((cb)a) = 0 \in I$  (by (2.37)) we have that  $ca \in I$  by Lemma 2.1.21(2). Hence  $c \in J$  and J is an ideal.

For (ii), we have that  $a \in J$  since  $aa = C \in I$  by (2.3). Suppose  $b \in I$ . Because  $ba \leq b$  by Lemma 2.1.12(3) we have that  $ba \in I$  by Lemma 2.1.21(1). Hence  $b \in J$  and  $I \subseteq J$ . Therefore J is an ideal of A with  $a \in J$  and  $I \subseteq J$ . Let  $K \in I(A)$  be such that  $a \in K$  and  $I \subseteq K$ . Suppose  $b \in J$ . Then  $ba \in I \subseteq K$ . Since  $ba, a \in K$ , we have that  $b \in K$  by definition of K as an ideal. Hence  $J \subseteq K$  and J is the smallest ideal of A such that  $a \in J$  and  $a \subseteq J$ .

Let A be a pre-BCK-algebra, let  $I \in I(A)$  and let  $a \in A$  be fixed. For ease of notation in the sequel, we write  $\langle I, a \rangle_A$  for the supremum  $\langle I \cup \{a\} \rangle_A$ of I and  $\langle a \rangle_A$  in the ideal lattice I(A). Also, we continue to write (a] for  $\{b : b \leq a\}$ , and we write  $A \sqcap a$  for  $\{b \sqcap a : b \in A\}$ , the principal left ideal generated by a.

**Theorem 2.2.20.** For a variety  $\vee$  of pre-BCK-algebras, the following are equivalent:

- 1. V is a variety of positive implicative pre-BCK-algebras;
- 2. V is weakly congruence orderable and the binary term  $x \div y$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9.

If V is a variety of positive implicative pre-BCK-algebras, then the following assertions also hold for any  $A \in V$  and  $a, b \in A$ :

- 3.  $\langle a \rangle_{\mathbf{A}} = (a] = A \sqcap a;$
- 4.  $a \in \langle b \rangle_{\mathbf{A}}$  iff ab = 0 iff  $a \sqcap b = a$ ;
- 5.  $\langle a b \rangle_{\mathbf{A}} = \langle a \rangle_{\mathbf{A}} * \langle b \rangle_{\mathbf{A}}$ , where \* denotes dual relative pseudocomplementation in the join semilattice  $\langle \operatorname{CI}(\mathbf{A}); \lor, \langle 0 \rangle_{\mathbf{A}} \rangle$  of compact ideals of  $\mathbf{A}$ .

*Proof.* Let V be a variety of pre-BCK-algebras.

(1)  $\Rightarrow$  (2) Suppose V is a variety of positive implicative pre-BCK-algebras. By Theorem 2.1.3, V is subtractive witness  $x \div y$ . To see  $x \div y$  also witnesses EDPI for V, just note that  $a \in \langle b \rangle_A$  iff  $a \in \langle \{0\}, b \rangle_A$  iff  $ab \in \langle 0 \rangle_A$  (by Lemma 2.2.19) iff ab = 0 for any  $A \in V$  and  $a, b \in A$ . To see V is weakly congruence orderable, it is sufficient to show  $\Theta^A(0, a) \subseteq \Theta^A(0, b)$  iff  $a \preceq b$  for any  $A \in V$  and  $a, b \in A$ , just because of Lemma 1.7.17 and Theorem 2.1.14(4). So let  $A \in V$  and  $a, b \in A$ . Suppose  $a \preceq b$ . Then  $a = a0 \equiv_{\Theta^A(0,b)} ab = 0$ , so  $\Theta^A(a,0) \subseteq \Theta^A(b,0)$ . Conversely, suppose  $\Theta^A(0,a) \subseteq \Theta^A(0,b)$ . Then  $\langle a \rangle_A \subseteq \langle b \rangle_A$  by normality of ideals and so  $a \in \langle b \rangle_A$ . Since V has EDPI witness  $x \div y$  it follows that ab = 0; that is to say  $a \preceq b$ .

(2)  $\Rightarrow$  (1) Suppose V is weakly congruence orderable and that the binary term  $x \div y$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9. Then  $\forall_{\epsilon}$  is a congruence orderable subvariety of V by Proposition 1.7.19, and so also has EDPI witness  $x \div y$ . Moreover,  $\forall_{\epsilon}$  is 0-regular witness  $\{x \div y, y \div x\}$  by Proposition 1.7.18, and so is a variety of positive implicative BCK-algebras by Proposition 1.7.12. Let  $\mathbf{A} \in \mathbf{V}$ . By the preceding characterisation of  $\forall_{\epsilon}$ , we have that  $\mathbf{A}/\langle 0 \rangle_{\mathbf{A}}^{\epsilon} \cong \mathbf{B}$  for some positive implicative BCK-algebra  $\mathbf{B}$ . But this means that  $\mathbf{A}/\Xi \cong \mathbf{B}$ , because  $\langle 0 \rangle_{\mathbf{A}}^{\epsilon} = \Xi_{\mathbf{A}}$  by Theorem 2.7.14(4). It follows that  $\mathbf{A}$  is a positive implicative pre-BCK-algebra. Hence  $\forall$  is a variety of positive implicative pre-BCK-algebras.

Throughout the remainder of the proof, assume that V is a variety of positive implicative pre-BCK-algebras. Let  $A \in V$  and  $a, b, \in A$ .

#### 2.2. Varieties of Pre-BCK-Algebras

For (3), we have that  $(a] = \langle a \rangle_{c}$  by EDPI. Suppose  $b \in (a]$ . Then ba = 0and so  $b = b0 = b(ba) \in A \sqcap a$ . Hence  $(a] \subseteq A \sqcap a$ . Conversely, suppose  $b \in A \sqcap a$ . Then b = c(ca) for some  $c \in A$ . Since  $c(ca) \preceq a$  by (2.2), we have that  $c(ca) \in (a]$ ; that is to say  $b \in (a]$ . Hence  $A \sqcap a \subseteq (a]$  and  $(a] = A \sqcap a$ . For (4), we have  $a \in \langle b \rangle_A$  iff ab = 0 (by EDPI) iff  $a \sqcap b = a$ (by Lemma 2.1.42). For (5), let  $I \in I(A)$ . Becau: of Lemma 2.2.19, we have  $\langle ab \rangle_A \subseteq I$  iff  $\langle a \rangle_A \subseteq \langle b \rangle_A \lor I$ , so  $\langle ab \rangle_A$  is the dual relative pseudocomplement of  $\langle b \rangle_A$  with respect to  $\langle a \rangle_A$  in  $\langle CI(A); \lor, \langle 0 \rangle_A \rangle$ .

Remark 2.2.21. The variety of pre-BCK-algebras does not have EDPI. In particular, the class jBCK of all BCK-algebras satisfying the identity:

$$x \stackrel{\cdot}{\rightarrow} \left(x \stackrel{\cdot}{\rightarrow} (y \stackrel{\cdot}{\rightarrow} (y \stackrel{\cdot}{\rightarrow} x))\right) \approx y \stackrel{\cdot}{\rightarrow} \left(y \stackrel{\cdot}{\rightarrow} (x \stackrel{\cdot}{\rightarrow} (x \stackrel{\cdot}{\rightarrow} y))\right)$$

of Cornish [69] is a BCK-variety that does not enjoy EDP1. To see this, observe that jBCK is a variety (by Theorem 2.2.2 or Cornish [69, Lemma 1]) which is not contained in any of the varieties  $e_nBCK$ ,  $n \in \omega$  (by Blok and Raftery [38, Proposition 16]). Since a variety of BCK-algebras has EDPI iff it is contained in some  $e_nBCK$ ,  $n \in \omega$  (by Theorem 1.6.10, Theorem 1.6.15 and Proposition 1.7.10), jBCK does not have EDPI. (This argument was communicated to the author by Professor James Raftery [188].)

The construction used by Blok and Raftery to show jBCK is not contained in any  $e_nBCK$ ,  $n \in \omega$ , produces, for each k > 1, an algebra  $A_k$  in jBCK that is not in any  $e_nBCK$ . However, the ideal lattice of  $A_k$  is, for each k > 1, isomorphic to the three-element dually relatively pseudocomplemented semilattice. Thus Blok and Raftery's construction cannot be used to exhibit an explicit example of a BCK-algebra A for which the join semilattice  $\langle CI(A); \vee, \langle 0 \rangle_A \rangle$  of compact ideals is not dually relative pseudocomplemented.

Problem 2.2.22. Exhibit an explicit example of a (pre-) BCK-algebra A for which the join semilattice  $\langle CI(A); \lor, \langle 0 \rangle_A \rangle$  of compact ideals is not dually relatively pseudocomplemented.

Because pre-BCK difference witnesses *both* subtractivity and EDPI for pPBCK in the sense of Theorem 1.7.9, Items (1) and (3) of the following corollary result

as an immediate consequence of Theorem 1.7.15(1). Despite this remark, we give as an easy modification of a result of Agliano [7] a direct proof of the corollary, on the grounds that this is conceptually simpler than an appeal to Theorem 1.7.15.

**Corollary 2.2.23.** (cf. [7, Lemma 3.3]) For any positive implicative pre-BCK-algebra **A**, the following assertions hold:

- 1. The set PI(A) of principal ideals of A is closed under dual relative pseudocomplementation. Thus  $\langle PI(A); *, \langle 0 \rangle_A \rangle$  is a positive implicative BCK-algebra;
- 2. The map  $a \mapsto \langle a \rangle_{\mathbf{A}}$  is a homomorphism from  $\mathbf{A}$  onto  $\langle \operatorname{PI}(\mathbf{A}); *, \langle 0 \rangle_{\mathbf{A}} \rangle$ . Moreover, ker  $f = \Xi$ ;
- 3. The map  $[a]_{\Xi} \mapsto \langle a \rangle_{\mathbf{A}}$  is an isomorphism from  $\mathbf{A}/\Xi$  onto  $\langle \operatorname{PI}(\mathbf{A}); *, \langle 0 \rangle_{\mathbf{A}} \rangle$ .

*Proof.* Let A be a positive implicative pre-BCK-algebra. For (1), just note the set PI(A) is closed under dual relative pseudocomplementation by Theorem 2.2.20(5), and hence that  $\langle PI(A); *, \langle 0 \rangle_A \rangle$  is a positive implicative BCKalgebra by Lemma 1.6.14. For (2) the map  $f : A \to PI(A)$  defined by  $a \mapsto \langle a \rangle_A$ is clearly onto. Moreover  $f(a \doteq Ab) = \langle a \doteq Ab \rangle_A = \langle \tilde{a} \rangle_A *^{\langle PI(A); *, \langle 0 \rangle_A \rangle} \langle b \rangle_A =$  $f(a) *^{\langle PI(A); *, \langle 0 \rangle_A \rangle} f(b)$  by Theorem 2.2.20(5), so (ignoring issues of similarity type) f is a homomorphism from A onto  $\langle PI(A); *, \langle 0 \rangle_A \rangle$ . Also  $\langle a, b \rangle \in \text{ker } f$ iff  $\langle a \rangle_A = \langle b \rangle_A$  iff  $[0]_{\Theta^A(0,a)} = [0]_{\Theta^A(0,b)}$  iff  $a \equiv b \pmod{\Xi}$  iff  $\langle a, b \rangle \in \Xi$  by the proof of Theorem 2.2.20, so ker  $f = \Xi$ . For (3), just note that  $A/\Xi$  is isomorphic to  $\langle PI(A); *, \langle 0 \rangle_A \rangle$  under the map  $[a]_{\Xi} \mapsto \langle a \rangle_A$  as an immediate consequence of (2) and the homomorphism theorem [99, Theorem 1§11.1].

Let A be a pre-BCK-algebra. Theorem 2.2.20(5) essentially provides an internal description of the dual relative pseudocomplement I \* J of principal ideals Iand J in the join semilattice  $\langle CI(A); \lor, \langle 0 \rangle_A \rangle$  of compact ideals of A. Theorem 2.2.26 in the sequel, which follows immediately from Proposition 2.2.25 below, extends this characterisation of dual relative pseudocomplementation to arbitrary members of  $\langle CI(A); \lor, \langle 0 \rangle_A \rangle$ . **Lemma 2.2.24.** Let A be a pre-BCK-algebra. For any  $b, c \in A$  and any  $a_1, \ldots, a_n \in A$ ,

$$(\cdots ((bc)a_1)\cdots)a_n \equiv ((\cdots (ba_1)\cdots)a_n)c \pmod{\Xi}.$$

*Proof.* The proof is by induction on n. We show only  $(\cdots((bc)a_1)\cdots)a_n \preceq ((\cdots(ba_1)\cdots)a_n)c$ ; the proof of the opposite inclusion is similar and is omitted. For n = 1, just note  $(bc)a_1 \preceq (ba_1)c$  by (2.10). Suppose now that the claim holds for k < n. By the inductive hypothesis,  $(\cdots((bc)a_1)\cdots)a_k \preceq ((\cdots(ba_1)\cdots)a_k)c$ , whence:

$$\left(\left(\cdots\left((bc)a_{1}\right)\cdots\right)a_{k}\right)a_{k+1} \preceq \left(\left(\left(\cdots\left(ba_{1}\right)\cdots\right)a_{k}\right)c\right)a_{k+1}$$
(2.38)

by Lemma 2.1.12(2). Put  $\alpha := (\cdots (ba_1) \cdots) a_k$ ,  $\beta := c$  and  $\gamma := a_{k+1}$ . From (2.10) we have  $(\alpha\beta)\gamma \preceq (\alpha\gamma)\beta$ , whence:

$$\left(\left((\cdots (ba_1) \cdots )a_k\right)c\right)a_{k+1} \preceq \left(\left((\cdots (ba_1) \cdots )a_k\right)a_{k+1}\right)c.$$

$$(2.39)$$

From (2.38), (2.39) and transitivity we conclude:

$$\left(\left(\cdots\left((bc)a_{1}\right)\cdots\right)a_{k}\right)a_{k+1} \preceq \left(\left(\left(\cdots\left(ba_{1}\right)\cdots\right)a_{k}\right)a_{k+1}\right)c$$

as required.

**Proposition 2.2.25.** Let A be a positive implicative pre-BCK-algebra. For any  $a_1, \ldots, a_n \in A$ ,

$$\langle a_1,\ldots,a_n\rangle_{\mathbf{A}} = \big\{c \in A: (\cdots((ca_1)a_2)\cdots)a_n = 0\big\}.$$

*Proof.* The proof is by induction on n. For n = 1,  $\langle a_1 \rangle_A = \{c \in A : ca_1 = 0\}$  by EDPI, so the basis case holds. Suppose the claim holds for k < n. We are required to show:

$$\langle a_1, \ldots, a_k, a_{k+1} \rangle_{\mathbf{A}} = \{ c \in A : ((\cdots ((ca_1)a_2)\cdots)a_k)a_{k+1} = 0 \}.$$
 (2.40)

So let  $I := \langle a_1, \ldots, a_k \rangle_A$ . By Lemma 2.2.19 and the inductive hypothesis, we

have:

$$\langle a_1, \ldots, a_k, a_{k+1} \rangle_{\mathbf{A}} = \langle I, a_{k+1} \rangle_{\mathbf{A}}$$

$$= I \vee^{\mathbf{I}(\mathbf{A})} \langle a_{k+1} \rangle_{\mathbf{A}}$$

$$= \{ c \in A : ca_{k+1} \in I \}$$

$$= \{ c \in A : (\cdots ((ca_{k+1})a_1) \cdots )a_k = 0 \}.$$

To complete the proof let  $J := \{c \in A : (\cdots ((ca_{k+1})a_1)\cdots)a_k = 0\}$  and let  $K := \{c \in A : ((\cdots (ca_1)\cdots)a_k)a_{k+1} = 0\}$ ; we show J = K. Let  $b \in J$ . Then  $(\cdots ((ba_{k+1})a_1)\cdots)a_k = 0$ , so  $((\cdots (ba_1)\cdots)a_k)a_{k+1} = 0$  by Lemma 2.2.24 and Corollary 2.1.14(3). Thus  $b \in K$  and  $J \subseteq K$ . For the converse, let  $b \in K$ . Then  $((\cdots (ba_1)\cdots)a_k)a_{k+1} = 0$ , so  $(\cdots ((ba_{k+1})a_1)\cdots)a_k = 0$ , also by Lemma 2.2.24 and Corollary 2.1.14(3). Thus  $b \in J$  and  $K \subseteq J$ . Hence J = K. This establishes (2.40), so the proof is complete.

Theorem 2.2.26. (cf. [68, Theorem 2.5]) Let A be a positive implicative pre-BCK-algebra and let  $I := \langle a_1, \ldots, a_t \rangle_A$ ,  $J := \langle b_1, \ldots, b_r \rangle_A$  be two finitely generated ideals of A. For  $i = 1, \ldots, t$ , let  $d_i := (\cdots (a_i b_1) \cdots )b_r$ . Then the dual relative pseudocomplement I \* J of I and J in the join semilattice  $\langle CI(A); \lor, \langle 0 \rangle_A \rangle$  of compact ideals of A is the ideal  $\langle d_1, \ldots, d_t \rangle_A$ .

By a quasi-bounded positive implicative pre-BCK-algebra we mean any quasibounded pre-BCK-algebra induced from a positive implicative pre-BCK-algebra with a maximal class. Because of Theorem 2.2.17 and the remarks of §2.1.40, the class pPBCK<sup>1</sup> of all quasi-bounded positive implicative pre-BCK-algebras is a variety. We conclude this subsection with a technical lemma concerning pPBCK<sup>1</sup> that will be needed in the sequel.

**Lemma 2.2.27.** The variety  $pPBCK^1$  of quasi-bounded positive implicative pre-BCK-algebras coincides with  $pPBCK^+$ , the generic double-pointed expansion of the variety of positive implicative pre-BCK-algebras.

*Proof.* Let  $A^1 \in pPBCK^1$ . For any  $a \in A$ ,  $a = a \sqcap 1 \equiv_{\Theta^{A^1}(0,1)} a \sqcap 0 = 0$ , so  $\Theta^{A^1}(0,1) = \iota_{A^1}$ ; that is to say  $A^1 \in pPBCK^+$ . Hence  $pPBCK^1 \subseteq pPBCK^+$ . For the opposite inclusion, let  $A^+ \in pPBCK^+$ . We separate the proof into two cases:

- (i) 0 = 1;
- (ii)  $0 \neq 1$ .

For (i), suppose 0 = 1. Then  $\Theta^{\mathbf{A}^+}(0,1) = \omega_{\mathbf{A}^+}$ . But  $\Theta^{\mathbf{A}^+}(0,1) = \iota_{\mathbf{A}^+}$  by hypothesis, so  $\mathbf{A}^+$  is trivial and 1 is maximal. Hence  $\mathbf{A}^+ \in \mathsf{pPBCK}^1$ .

For (ii), suppose  $0 \neq 1$ . Then A is non-trivial, so we may assume to the contrary that 1 is not maximal. Since 1 is not maximal, there exists  $0 \neq a \in A$  such that either  $1 \prec a$  or 1 and a are incomparable under the pre-BCK quasiorder. In either case it follows that  $a \notin \langle 1 \rangle_{A^-}$ , where  $A^-$  denotes the positive implicative pre-BCK-algebra reduct of  $A^+$ , because of the description of the principal ideals of  $A^-$  as the hereditary subsets of A. Thus  $\langle 1 \rangle_{A^-}$  is proper. Since  $A^+$  and  $A^-$  have the same congruences,  $\langle 1 \rangle_{A^+} = \langle 1 \rangle_{A^-}$  by normality of ideals; that is to say  $\langle 1 \rangle_{A^+}$  is proper. But this implies  $\langle 1 \rangle_{A^+}^{\delta}$ ,  $\iota_{A^+}$ , where  $\langle 1 \rangle_{A^+}^{\delta}$  is the least congruence on  $A^+$  whose 0-class is  $\langle 1 \rangle_{A^+}$ , so  $\Theta^{A^+}(0,1) < \iota_{A^+}$  since  $\langle 1 \rangle_{A^+}^{\delta} = \Theta^{A^+}(0,1)$ . Hence 0 and 1 are not residually distinct, a contradiction. Thus 1 is maximal, and  $A^+ \in \text{pPBCK}^1$ .

By (i) and (ii),  $A^+ \in pPBCK^1$ . Hence  $pPBCK^+ \subseteq pPBCK^1$ . Therefore  $pPBCK^+ = pPBCK^1$ , and the proof is complete.

2.2.28. Implicative Pre-BCK-Algebras. By an *implicative pre-BCK-algebra* we mean a pre-BCK-algebra A such that  $A/\Xi \cong B$  for some implicative BCK-algebra B. Since the class iBCK of all implicative BCK-algebras is a variety, axiomatised relative to PBCK by the identity (*cf.* [38, pp. 294–295]):

$$\left(x \stackrel{\cdot}{-} (x \stackrel{\cdot}{-} y)\right) \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x) \approx y \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x)$$
(2.41)

the class iPBCK of all implicative pre-BCK-algebras is also a variety, by Theorem 2.2.3. In particular, iPBCK is axiomatised relative to PBCK by the pair of identities:

$$\left( (x \stackrel{\cdot}{-} (x \stackrel{\cdot}{-} y)) \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x) \right) \stackrel{\cdot}{-} \left( y \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x) \right) \approx \mathbf{0}$$
$$\left( y \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x) \right) \stackrel{\cdot}{-} \left( (x \stackrel{\cdot}{-} (x \stackrel{\cdot}{-} y)) \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x) \right) \approx \mathbf{0}$$

as (2.41) is of the form (2.29). The preceding characterisation of iPBCK

### 2.2. Varieties of Pre-BCK-Algebras

notwithstanding, the following axiomatisation often proves more useful in practice.

**Theorem 2.2.29.** An algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is an implicative pre-BCK-algebra iff the following identities are satisfied:

$$\left(\left(x - y\right) - \left(x - z\right)\right) - \left(z - y\right) \approx 0 \tag{2.1}$$

$$\mathbf{0} - \mathbf{x} \approx \mathbf{0} \tag{2.3}$$

$$x - \mathbf{0} \approx x$$
 (2.6)

$$x - (x - (y - x)) \approx \mathbf{0}. \tag{2.42}$$

Thus the class iPBCK of implicative pre-BCK-algebras is a variety.

*Proof.* Let A be an implicative pre-BCK-algebra. By definition we have that  $A \models (2.1), (2.3), (2.6)$ , so it only remains to show  $A \models (2.42)$ . Since A is implicative,  $A/\Xi \cong B$  for some implicative BCK-algebra B. Because B is implicative,  $B \models x - (y - x) \approx x$ , so  $B \models (2.42)$  by (1.3). As identities are preserved by isomorphic copies, we have that  $A/\Xi \models (2.42)$ , so  $A \models (2.42)$  by Corollary 2.1.16.

Conversely, suppose  $\mathbf{A} := \langle A; \div, 0 \rangle$  is an algebra of type  $\langle 2, 0 \rangle$  such that  $\mathbf{A} \models (2.1), (2.3), (2.6), (2.42)$ . By (2.1), (2.3), (2.6) and Lemma 2.1.6,  $\mathbf{A}$  is a pre-BCK-algebra. By (2.42),  $\mathbf{A} / \Xi \models x \div (x \div (y \div x)) \approx 0$ ; also  $\mathbf{A} / \Xi \models (x \div (y \div x)) \div x \approx 0$  by (1.30). Hence  $\mathbf{A} / \Xi \models x \div (y \div x) \approx x$  by (1.5), so  $\mathbf{A}$  is implicative.

By the remarks of §1.6.16, a BCK-algebra is implicative iff it is both commutative and positive implicative. This observation yields the following alternative characterisation of implicative pre-BCK-algebras, which will be needed in the sequel.

**Proposition 2.2.30.** A pre-BCK-algebra is implicative iff it is both commutative and positive implicative. Thus the variety of implicative pre-BCK-algebras is the intersection of the variety of commutative pre-BCK-algebras and the variety of positive implicative pre-BCK-algebras.

*Proof.* Let A be a pre-BCK-algebra. Suppose A is implicative. Then  $A/\Xi \cong B$  for some implicative BCK-algebra B. Since B is both commutative and positive implicative, A is both commutative and positive implicative. Conversely, suppose A is both commutative and positive implicative. Then  $A \cong B$  for some commutative and positive implicative BCK-algebra B. Since B is implicative, A is implicative. The remaining assertion of the proposition now follows.

Let V be a subtractive variety. Recall ([9, Section 2]) a term  $t(\vec{x}, \vec{y}, \vec{z})$  is a commutator term for  $\vee in \vec{y}, \vec{z}$  (where  $\vec{y}, \vec{z}$  are disjoint sets of variables) if it is both an ideal term in  $\vec{y}$  and an ideal term in  $\vec{z}$ ; that is, if  $t \in \mathrm{IT}_{\vee}(\vec{y}) \cap \mathrm{IT}_{\vee}(\vec{z})$ . We denote the set of all commutator terms for  $\vee$  in  $\vec{y}, \vec{z}$  by  $\mathrm{CT}_{\vee}(\vec{y}, \vec{z})$ . For  $\mathbf{A} \in \vee$  and non-empty  $H, K \subseteq A$ , let:

$$[H, K]_{\mathbf{A}} := \{ t^{\mathbf{A}}(\vec{a}, \vec{h}, \vec{k}) : t(\vec{x}, \vec{y}, \vec{z}) \in \mathrm{CT}_{\mathsf{V}}(\vec{y}, \vec{z}); \ \vec{a} \in A, \vec{h} \in H, \vec{k} \in K \}.$$

 $[H, K]_{\mathbf{A}}$  is called the *commutator* of H, K in  $\mathbf{A}$  for  $\mathbf{V}$ . The commutator of ideals was introduced for ideal determined varieties by Ursini in [220] (see also Gumm and Ursini [104, Section 2]) and for subtractive varieties in general by Ursini in [222, Section 2] (see also Agliano and Ursini [9, Section 2] and Ursini [225]). By [10, Proposition 2.1(i)],  $[H, K]_{\mathbf{A}} \subseteq \langle H \rangle_{\mathbf{A}} \cap \langle K \rangle_{\mathbf{A}}$  always; when  $\mathbf{V}$  is ideal distributive the opposite inclusion also holds, in which case  $[H, K]_{\mathbf{A}} = \langle H \rangle_{\mathbf{A}} \cap \langle K \rangle_{\mathbf{A}}$  [9, Proposition 4.1(2)].

Let V be a subtractive variety and let  $A \in V$  and  $a, b \in A$ . A (proper) ideal I of A is said to be *prime* if whenever  $[a, b]_A \subseteq I$  then  $a \in I$  or  $b \in I$ ; this definition originates with Ursini [221] (see also Chajda and Halaš [57]). Therefore if V is ideal distributive, then I is prime if  $\langle a \rangle_A \cap \langle b \rangle_A \subseteq I$  implies  $a \in I$  or  $b \in I$ .

**Proposition 2.2.31.** For any implicative pre-BCK-algebra A and  $a, b \in A$ ,  $\langle a \rangle_{\mathbf{A}} \cap \langle b \rangle_{\mathbf{A}} = \langle a \sqcap b \rangle_{\mathbf{A}}.$ 

### 2.2. Varieties of Pre-BCK-Algebras

*Proof.* Let A be an implicative pre-BCK-algebra and let  $a, b \in A$ . By the proof of Corollary 2.1.29 we have  $\langle a \sqcap b \rangle_{\mathbf{A}} \subseteq \langle a \rangle_{\mathbf{A}} \cap \langle b \rangle_{\mathbf{A}}$  always. For the opposite inclusion,  $c \in \langle a \rangle_{\mathbf{A}} \cap \langle b \rangle_{\mathbf{A}}$  implies  $c \in \langle a \rangle_{\mathbf{A}}$  and  $c \in \langle b \rangle_{\mathbf{A}}$ , which implies  $c \preceq a, b$ , which implies  $c \preceq a \sqcap b$  (since  $a \sqcap b$  is a greatest lower bound of  $\{a, b\}$ ), which implies  $c \in \langle a \sqcap b \rangle_{\mathbf{A}}$ . Thus  $\langle a \rangle_{\mathbf{A}} \cap \langle b \rangle_{\mathbf{A}} \subseteq \langle a \sqcap b \rangle_{\mathbf{A}}$  as required.

Let A be an implicative pre-BCK-algebra. Because PBCK is ideal distributive (by Proposition 2.1.24(1)), from Proposition 2.2.31 we have that a proper ideal I of A is prime if  $\langle a \sqcap b \rangle_{\mathbf{A}} \subseteq I$  implies either  $a \in I$  or  $b \in I$  for all  $a, b \in A$ . Equivalently, I is prime if:

$$a \sqcap b \in I$$
 implies either  $a \in I$  or  $b \in I$  (2.43)

for all  $a, b \in A$ . For implicative BCK-algebras, an elegant theory of prime ideals (which exploits exactly the notion of primality expressed by (2.43)) has been developed by Rasiowa in [195, Chapter II§6]. This last, in conjunction with preceding remarks, motivates the study of prime ideals in implicative pre-BCK-algebras.

**Lemma 2.2.32.** The variety of (positive) implicative pre-BCK-algebras satisfies the following identity:

$$(x - y) - ((x - y) - (y - x)) \approx 0.$$

$$(2.44)$$

*Proof.* We have been unable to find a derivation of (2.44) (for (positive) implicative BCK-algebras) in the literature, so we do not invoke Corollary 2.1.16 to prove the lemma. Instead we provide a derivation of (2.44) (for (positive) implicative pre-BCK-algebras). Let A be a (positive) implicative pre-BCK-algebras). Let A be a (positive) implicative pre-BCK-algebra. Put  $\alpha := ab, \beta := b$  and  $\gamma := ba$ . We have:

| $0 = ((\alpha\beta)(\alpha\gamma))(\gamma\beta)$         | by (2.1)           |
|--|--------------------|
| $= \big((\alpha\beta)(\alpha\gamma)\big)\big((ba)b\big)$ |                    |
| $= ((lphaeta)(lpha\gamma))0$                             | by Lemma 2.1.12(3) |
| $= (\alpha\beta)(\alpha\gamma)$                          | . by (2.6)         |

146

2.2. Varieties of Pre-BCK-Algebras

$$= ((ab)b)((ab)(ba)).$$

$$(2.45)$$

For (2.44), put  $\alpha := ab$ ,  $\beta := (ab)(ba)$  and  $\gamma := (ab)b$ . We have:

$$0 = ((\alpha\beta)(\alpha\gamma))(\gamma\beta)$$
  
=  $((\alpha\beta)(\alpha\gamma))(((ab)b)((ab)(ba)))$   
=  $((\alpha\beta)(\alpha\gamma))0$  by (2.45)  
=  $(\alpha\beta)(\alpha\gamma)$  by (2.6)  
=  $(\alpha\beta)((ab)((ab)b))$   
=  $(\alpha\beta)0$  by (2.36)  
=  $(ab)((ab)(ba))$  by (2.6).

Let A be an implicative pre-BCK-algebra and let I be a proper ideal of A. We say I is *irreducible* provided that:

$$I = J \cap K$$
 implies either  $I = J$  or  $I = K$  (2.46)

for any two proper ideals  $J, K \in I(\mathbf{A})$ . An ideal that is not irreducible is said to be *reducible*. The following proposition is a modification of a result due to Rasiowa [195, Chapter II].

**Proposition 2.2.33.** (cf. [195, Theorem II $\S$ 6.1]) Let A be an implicative pre-BCK-algebra and let I be an ideal of A. Then I is prime iff it is irreducible.

*Proof.* Let A be an implicative pre-BCK-algebra and I be a proper ideal of A.

(⇒) Suppose I is not irreducible. Then there exist proper ideals  $J, K \in I(A)$  such that  $I = J \cap K$  but  $I \neq J$  and  $I \neq K$ . Clearly  $J \not\subseteq K$  and  $K \not\subseteq J$ . Indeed,  $J \subseteq K$  implies  $I = J \cap K = J$  while  $K \subseteq J$  implies  $I = J \cap K = K$ , both of which contradict the reducibility of I. Thus there exist  $a, b \in A$  such that  $a \in J, b \in K$  but  $a \notin K, b \notin J$ . To complete the proof we show:

(i)  $ab, ba \notin I$ ;

(ii)  $ab \sqcap ba \in I$ .

For (i), we show only that  $ab \notin I$ ; the proof that  $ba \notin I$  is analogous and is omitted. So let  $\langle I, b \rangle_{\mathbf{A}}$  be the ideal generated by  $I \cup \{b\}$ . Clearly  $I \subseteq \langle I, b \rangle_{\mathbf{A}} \subseteq K$ . Hence  $a \notin \langle I, b \rangle_{\mathbf{A}}$ , since  $a \in \langle I, b \rangle_{\mathbf{A}}$  implies  $a \in K$ , a contradiction. But this implies  $ab \notin I$ , because  $\langle I, b \rangle_{\mathbf{A}} = \{c \in A : cb \in I\}$  by Lemma 2.2.19.

For (ii), simply note  $ab \sqcap ba = 0 \in I$  by (2.44).

By (i) and (ii),  $ab, ba \notin I$  but  $ab \sqcap ba \in I$ , which proves by (2.43) that I is not prime.

( $\Leftarrow$ ) Suppose I is not prime. Then there exist  $a, b \in A$  such that  $a \neq b$  and  $a \sqcap b \in I$ , but  $a \notin I$  and  $b \notin I$ . Let  $\langle I, a \rangle_{\mathbf{A}}$  and  $\langle I, b \rangle_{\mathbf{A}}$  be the ideals generated by  $I \cup \{a\}$  and  $I \cup \{b\}$  respectively. To complete the proof we show:

- (i)  $I = \langle I, a \rangle_{\mathbf{A}} \cap \langle I, b \rangle_{\mathbf{A}};$
- (ii)  $\langle I, a \rangle_{\mathbf{A}}$  is a proper ideal such that  $I \neq \langle I, a \rangle_{\mathbf{A}}$ ;
- (iii)  $\langle I, b \rangle_{\mathbf{A}}$  is a proper ideal such that  $I \neq \langle I, b \rangle_{\mathbf{A}}$ .

For (i), the inclusion  $I \subseteq \langle I, a \rangle_{\mathbf{A}} \cap \langle I, b \rangle_{\mathbf{A}}$  is obvious. For the opposite inclusion, we have:

$$\langle I, a \rangle_{\mathbf{A}} \cap \langle I, b \rangle_{\mathbf{A}} = (I \vee^{\mathbf{I}(\mathbf{A})} \langle a \rangle_{\mathbf{A}}) \cap (I \vee^{\mathbf{I}(\mathbf{A})} \langle b \rangle_{\mathbf{A}})$$
  
=  $I \vee^{\mathbf{I}(\mathbf{A})} (\langle a \rangle_{\mathbf{A}} \cap \langle b \rangle_{\mathbf{A}})$   
=  $I \vee^{\mathbf{I}(\mathbf{A})} \langle a \sqcap b \rangle_{\mathbf{A}}$ 

by Proposition 2.1.24(1) and Proposition 2.2.31, so  $\langle I, a \rangle_{\mathbf{A}} \cap \langle I, b \rangle_{\mathbf{A}} = \{ d \in A : d(a \cap b) \in I \}$  by Lemma 2.2.19. Let  $c \in \langle I, a \rangle_{\mathbf{A}} \cap \langle I, b \rangle_{\mathbf{A}}$ . Then  $c(a \cap b) \in I$ , and since  $a \cap b \in I$  we have that  $c \in I$  by the definition of I as an ideal. Hence  $\langle I, a \rangle_{\mathbf{A}} \cap \langle I, b \rangle_{\mathbf{A}} \subseteq I$  and  $I = \langle I, a \rangle_{\mathbf{A}} \cap \langle I, b \rangle_{\mathbf{A}}$ .

For (ii),  $I \neq \langle I, a \rangle_{\mathbf{A}}$  since  $a \notin I$ . To see  $\langle I, a \rangle_{\mathbf{A}}$  is proper, assume to the contrary that  $b \in \langle I, a \rangle_{\mathbf{A}}$ . Then  $ba \in I$  by Lemma 2.2.19. Since  $(b \sqcap a)(a \sqcap b) = 0 \in I$  by (2.33) and  $a \sqcap b \in I$  by hypothesis, we infer  $b \in I$  by Lemma 2.1.21(2), which is a contradiction. Hence  $b \notin \langle I, a \rangle_{\mathbf{A}}$  and  $\langle I, a \rangle_{\mathbf{A}}$  is proper.

For (iii),  $I \neq \langle I, b \rangle_{\mathbf{A}}$  since  $b \notin I$ . To see  $\langle I, b \rangle_{\mathbf{A}}$  is proper, assume to the contrary that  $a \in \langle I, b \rangle_{\mathbf{A}}$ . Then  $ab \in I$  by Lemma 2.2.19. Since  $a \sqcap b \in I$ 

by hypothesis, we infer  $a \in I$  by the definition of I as an ideal, which is a contradiction. Hence  $a \notin \langle I, b \rangle_{\mathbf{A}}$  and  $\langle I, b \rangle_{\mathbf{A}}$  is proper.

By (i), (ii) and (iii),  $I = \langle I, a \rangle_{\mathbf{A}} \cap \langle I, b \rangle_{\mathbf{A}}$  where  $\langle I, a \rangle_{\mathbf{A}}$  and  $\langle I, b \rangle_{\mathbf{A}}$  are proper ideals such that  $I \neq \langle I, a \rangle_{\mathbf{A}}$  and  $I \neq \langle I, b \rangle_{\mathbf{A}}$ , which proves by (2.46) that I is not irreducible.

Let A be an implicative pre-BCK-algebra. A proper ideal I of A is said to be *maximal* provided it is not a proper subset of any proper ideal. The following proposition is an easy modification of a result due to Rasiowa [195, Chapter II].

**Proposition 2.2.34.** (cf. [195, Theorem II $\S 6.2$ ]) Let A be an implicative pre-BCK-algebra and let I be an ideal of A. Then I is prime iff it is maximal.

*Proof.* Let A be an implicative pre-BCK-algebra and I be a proper ideal of A.

 $(\Rightarrow)$  Suppose *I* is prime and assume to the contrary that *I* is not maximal. Then *I* is a proper subset of some proper ideal  $J \in I(\mathbf{A})$  and so there exists  $a \in A$  such that  $a \notin I$  but  $a \in J$ . Let  $\langle I, a \rangle_{\mathbf{A}}$  be the ideal generated by  $I \cup \{a\}$ . Clearly  $I \subseteq \langle I, a \rangle_{\mathbf{A}} \subseteq J$ . To complete the proof it is sufficient to show that  $\langle I, a \rangle_{\mathbf{A}}$  (and hence *J*) is improper. Since  $ca \sqcap a = 0 \in I$  (by (2.36)) and  $a \notin I$  we have that  $ca \in I$  for all  $c \in A$  by (2.43). But this implies  $c \in \langle I, a \rangle_{\mathbf{A}}$  for all  $c \in A$  because  $\langle I, a \rangle_{\mathbf{A}} = \{b \in A : ba \in I\}$  by Lemma 2.2.19. Thus  $\langle I, a \rangle_{\mathbf{A}} = A$  and so J = A, a contradiction.

( $\Leftarrow$ ) Suppose *I* is maximal and assume to the contrary that *I* is not irreducible. Then there exist proper ideals  $J, K \in I(A)$  such that  $I = J \cap K$  but  $I \neq J$ and  $I \neq K$ . But this implies  $I = J \cap K \subset J$  and  $I = J \cap K \subset K$ , whence  $I = J \cap K$  is a proper ideal contained in both the proper ideals *J* and *K*. Thus *I* is not maximal, which is a contradiction. Hence *I* is irreducible, which shows by Proposition 2.2.33 that *I* is prime.

The following result is also an easy modification of a theorem of Rasiowa [195, Chapter II].

**Proposition 2.2.35.** (cf. [195, Theorem II§6.4]) Let A be an implicative pre-BCK-algebra and let I be an ideal of A. Then I is prime iff  $A/\phi_I$  is

149

isomorphic to the two-element flat implicative BCK-algebra  $C_1$ , where  $\phi_I$  is the congruence induced by I in the sense of Theorem 2.1.26.

*Proof.* Let A be an implicative pre-BCK-algebra and let I be an ideal of A.

(⇒) Suppose *I* is prime. By Lemma 2.1.27 we have  $a \in I$  iff  $a \equiv 0 \pmod{\phi_I}$ , whence  $[a]_{\phi_I} = [0]_{\phi_I}$  for  $a \in I$ . Suppose now that  $a \notin I$  and  $b \notin I$ . From  $a \notin I$  and  $ba \sqcap a = 0 \in I$  (by (2.36)) we have that  $ba \in I$  by (2.43). Likewise, from  $b \notin I$  and  $ab \sqcap b = 0 \in I$  (by (2.36)) we have that  $ab \in I$  by (2.43). Since  $ab, ba \in I$  we infer  $a \equiv b \pmod{\phi_I}$ , or equivalently,  $[a]_{\phi_I} = [b]_{\phi_I}$ . Hence (the universe of) the quotient algebra  $\mathbf{A}/\phi_I$  has exactly two elements. Since  $\phi_I \geq \Xi$  we have that  $\mathbf{A}/\phi_I$  is a BCK-algebra, and the result is now forced by the remarks of §1.6.16.

( $\Leftarrow$ ) Suppose  $\mathbf{A}/\phi_I$  is isomorphic to the 2-element flat implicative BCK-algebra  $\mathbf{C}_1$ . Then I is proper (because I improper implies  $\mathbf{A}/\phi_I$  is trivial, a contradiction). Let  $a \notin I$  and  $b \notin I$ . Since the equivalence class  $[0]_{\phi_I}$  contains (by Lemma 2.1.27) exactly those elements of A belonging to I we infer that  $[a]_{\phi_I} = [b]_{\phi_I}$ , just because (the universe of) the quotient algebra  $\mathbf{A}/\phi_I$  has only two elements. Hence  $a \equiv b \pmod{\phi_I}$ , so  $ab, ba \in I$ . But then  $a \sqcap b \notin I$ , because  $a \sqcap b \in I$  and  $ab \in I$  implies  $a \in I$  by the definition of I as an ideal, which is a contradiction. We have shown that  $a \notin I$  and  $b \notin I$  implies  $a \sqcap b \notin I$ , which proves by (2.43) that I is prime.

**Theorem 2.2.36.** (cf. [195, Theorem II§6.4]) Let A be an implicative pre-BCK-algebra. For any ideal  $I \in I(A)$  the following are equivalent:

- 1. I is prime;
- 2. I is maximal;
- 3. I is irreducible;
- 4.  $A/\phi_I$  is isomorphic to the two-element flat implicative BCK-algebra  $C_1$ , where  $\phi_I$  is the congruence induced by I in the sense of Theorem 2.1.26.

Moreover, any proper ideal  $I \in I(A)$  is contained in a prime ideal. In particular,

$$I = \bigcap \{J : J \text{ is a prime ideal of } \mathbf{A} \text{ and } I \subseteq J \}.$$

*Proof.* It remains only to establish the final claim. By Proposition 2.2.33, the prime ideals of any implicative pre-BCK-algebra  $\mathbf{A}$  are precisely the meet irreducible elements of  $I(\mathbf{A})$ . The claim now follows, since any element of an algebraic lattice is the infimum of meet irreducible elements.

# 2.3 Implicative BCS-Algebras

In this section we study the variety of implicative BCS-algebras, a class of pointed groupoids that more closely resemble implicative BCK-algebras than do implicative pre-BCK-algebras.

Implicative BCS-algebras proper are introduced in §2.3.1. We show the variety of implicative BCS-algebras is a subvariety of the variety of implicative pre-BCK-algebras, and also prove that iBCK is the only non-trivial subquasivariety of the variety of implicative BCS-algebras that is a non-trivial subquasivariety of BCK. Some examples showing that implicative BCS-algebras arise noturally in universal algebra in binary discriminator varieties (including pseudocomplemented semilattices) and in algebraic logic in 'pointed' fixedpoint discriminator varieties (including certain subvarieties of n-potent BCK-algebras) are presented.

The role of the pre-BCK-meet  $\sqcap$  in the theory of implicative BCS-algebras is considered in §2.3.19. It is shown that the existence of a left normal band with zero polynomial reduct  $\langle \sqcap, 0 \rangle$  whose underlying natural band partial order  $\leq_{\mathcal{H}}^{(\sqcap,0)}$  respects implicative pre-BCK difference in a certain precise sense distinguishes the implicative BCS-algebras among the implicative pre-BCKalgebras. A representation theorem for implicative BCS-algebras is proved: for suitable choices of objects and morphisms, it is shown that the categories of implicative BCS-algebras and left handed (equivalently, left regular) locally Boolean bands are isomorphic.

In §2.3.42 we characterise (to within isomorphism) the subdirectly irreducible implicative BCS-algebras: they are precisely the 2-element implicative BCK-algebra and the algebras  $\hat{\mathbf{B}}$  obtained from the non-trivial Boolean algebras  $\mathbf{B}$  upon replacing the top element of each  $\mathbf{B}$  with a two-element  $\Xi$ -class.

Quasi-bounded implicative BCS-algebras are studied in §2.3.57. For a quasibounded implicative BCS-algebra  $A^1$ , the skeleton  $S(A^1)$  is the set  $\{a^* : a \in A\}$ . An internal description of the maximal bounded implicative BCS-algebra homomorphic image  $A^1/\Xi$  of a quasi-bounded implicative BCS-algebra  $A^1$ is given in terms of the skeleton  $S(A^1)$ . We apply this characterisation to give a new and conceptually simple proof of the Glivenko-Frink theorem for pseudocomplemented semilattices.

In §2.3.70 the role played by the 3-element pre-BCK-algebra  $B_2$  of Example 2.1.5 in the theory of implicative BCS-algebras is investigated. We show  $B_2$  generates the class of implicative BCS-algebras (as a variety) and hence that the lattice of varieties of implicative BCS-algebras is a 3-element chain; the only non-trivial subvariety of the variety of implicative BCS-algebras is the variety of implicative BCK-algebras.

**2.3.1.** Implicative BCS-Algebras. An implicative BCS-algebra is an algebra  $\langle A; \rangle, 0 \rangle$  of type  $\langle 2, 0 \rangle$  such that the following identities hold:

| $x \setminus x \approx 0$   | (2.47) |
|---|--------|
| $(x \setminus y) \setminus z \approx (x \setminus z) \setminus y$               | (2.48) |
| $(x \setminus z) \setminus (y \setminus z) \approx (x \setminus y) \setminus z$ | (2.49) |
| $x \setminus (y \setminus x) \approx x.$  | (2.50) |

Because of (2.47), the class iBCS of implicative BCS-algebras may be construed as a variety of pointed groupoids; consequently in the sequel we (informally) denote implicative BCS difference by juxtaposition when no confusion can arise. Identity (2.49) is an algebraic analogue of the S-combinator  $(p \to (q \to r)) \to ((p \to q) \to (p \to r))$  of combinatory logic; this accounts for the origin

| $\begin{array}{c ccc} A & 0 & a \\ \hline 0 & 0 & 0 \\ a & a & a \end{array}$ | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | $\begin{array}{c c c} \mathbf{A} & 0 & \mathbf{a} \\ \hline 0 & 0 & 0 \\ \mathbf{a} & 0 & 0 \\ \end{array}$ |
|---|---|---|---|
| (a)   | (b)   | (c)   | (d)   |

Figure 2.1. Independence tables for the implicative BCS identities.

of the term 'BCS-algebra'. The qualifier 'implicative' is intended to suggest (as per implicative BCK-algebras) that the binary operation in question is an algebraic analogue of (specifically) *classical* (that is, material) implication; recall (2.50) is an algebraic analogue of Peirce's law  $((p \rightarrow q) \rightarrow p) \rightarrow p$ .

**Remark 2.3.2.** In the variety of implicative BCS-algebras the identities (2.47)–(2.50) are independent, as the operation tables of Figure 2.1 confirm. Let A be the algebra defined by the operation table of Figure 2.1(a). Then  $A \models (2.48)-(2.50)$ ; however  $A \not\models (2.47)$  since  $a \setminus a = a \neq 0$ . Suppose instead that A is the algebra defined by the operation table of Figure 2.1(b). An easy sequence of checks shows  $A \models (2.47), (2.49)-(2.50)$  but that  $A \not\models (2.48)$ , since  $(a \setminus c) \setminus d = a \setminus d = b \neq a = b \setminus c = (a \setminus d) \setminus c$ . Let A be the algebra defined by the operation table of Figure 2.1(c). It is readily verified that  $A \models (2.47)-(2.48), (2.50)$  but that  $A \not\models (2.49)$ , just because  $(b \setminus a) \setminus c = 0 \setminus c = 0 \neq b = b \setminus 0 = (b \setminus c) \setminus (a \setminus c)$ . Suppose now that A is the algebra defined by the operation table of Figure 2.1(d). Clearly  $A \models (2.47)-(2.49)$ ; however  $A \not\models (2.50)$  as  $a \setminus (0 \setminus a) = a \setminus 0 = 0 \neq a$ .

Lemma 2.3.3. The variety of implicative BCS-algebras satisfies the following identities:

$$(x \setminus y) \setminus y \approx x \setminus y \tag{2.51}$$

$$(x \setminus y) \setminus (z \setminus x) \approx x \setminus y \tag{2.52}$$

 $x \setminus (y \setminus (z \setminus x)) \approx x \setminus y \tag{2.53}$ 

153

$$(x \setminus y) \setminus (y \setminus z) \approx x \setminus y \tag{2.54}$$

$$(x \setminus y) \setminus (z \setminus y) \approx (x \setminus z) \setminus (y \setminus z).$$
(2.55)

*Proof.* Let A be an implicative BCS-algebra and let  $a, b, c \in A$ . For (2.51) note (ab)b = (ab)(b(ab)) = ab by (2.50) applied twice. For (2.52) we have (ab)(ca) = (a(ca))b = ab by (2.48) and (2.50). For (2.53) we have a(b(ca)) =(a(ca))(b(ca)) = (ab)(ca) = ab by (2.50), (2.49) and (2.52). For (2.54) we have (ab)(bc) = (ab)((b(ab))c) = (ab)((bc)(ab)) = ab by (2.50), (2.48) and (2.50). For (2.55) we have (ac)(bc) = (ab)c = (ac)b = (ab)(cb) by (2.49), (2.48) and (2.49). ■

**Proposition 2.3.4.** The variety of implicative BCS-algebras satisfies the following identity:

$$x \setminus (x \setminus (x \setminus y)) \approx x \setminus y. \tag{2.56}$$

*Proof.* Let A be an implicative BCS-algebra and let  $a, b \in A$ . We have:

| ab = (ab)(a(ab))       | by (2.50)   |
|------------------------|-------------|
| = (a(a(ab)))(b(a(ab))) | by (2.49)   |
| = (a(a(ab))(ba)        | by (2.53)   |
| = (a(ba))(a(ab))       | by (2.48)   |
| =a(a(ab))              | by $(2.50)$ |

which establishes (2.56) as required.

In the statement and proof of the following two results we ignore issues of type.

**Proposition 2.3.5.** If A is an implicative BCS-algebra then A is an implicative pre-BCK-algebra. Thus the variety of implicative BCS-algebras is a subvariety of the variety of implicative pre-BCK-algebras.

*Proof.* Let A be an implicative BCS-algebra and let  $a, b, c \in A$ . By (2.50) and (2.47) we have a = a(aa) = a0, so  $A \models (2.6)$ . From (2.50) it follows that

0 = 0(a0) = 0a, whence  $\mathbf{A} \models (2.4)$ . Also,

$$((ab)(ac))(cb) = ((ab)(cb))(ac)$$
 by (2.48)  
=  $((ac)b)(ac)$  by (2.49)  
=  $((ac)(ac))b$  by (2.49)  
=  $0b$  by (2.47)  
=  $0$ 

so  $A \models (2.1)$ . By Lemma 2.1.6 we conclude that A is a pre-BCK-algebra. To see A is implicative, just notice 0 = aa = a(a(ba)) by (2.47) and (2.50).

**Remark 2.3.6.** The variety of implicative BCS-algebras is properly contained within the variety of implicative pre-BCK-algebras. To see this, consider the following algebra A:

 $\begin{array}{c|c|c} - \mathbf{A} & \mathbf{\hat{0}} & a & b & c \\ \hline 0 & 0 & 0 & 0 & 0 \\ a & a & 0 & c & 0 \\ b & b & b & 0 & b \\ c & c & 0 & a & 0 \end{array}$ 

An easy sequence of checks shows that A is an implicative pre-BCK-algebra. However, A is not an implicative BCS-algebra, since  $a(ba) = ab = c \neq a$ .

**Proposition 2.3.7.** For the variety of implicative BCS-algebras the following assertions hold:

- 1. iBCK is contained in any non-trivial subquasivariety of iBCS;
- 2. iBCK is the only non-trivial subquasivariety of iBCS that is a non-trivial subquasivariety of BCK. Thus iBCK is axiomatised relative to iBCS by any identity of the form:

$$x \setminus \prod_{i=1}^n u_i(x,y) \approx y \setminus \prod_{j=1}^m v_j(x,y)$$

where  $n, m \in \omega$  and  $u_1, \ldots, u_n, v_1, \ldots, v_m$  are  $\langle \backslash, 0 \rangle$ -terms such that BCK satisfies:

$$u_i(x,x) \approx 0 \approx v_j(x,x)$$
  $i = 1, \ldots, n, \quad j = 1, \ldots, m$ 

In particular, iBCK is axiomatised relative to iBCS by the identity:

$$x \setminus (x \setminus y) \approx y \setminus (y \setminus x).$$

Proof. For (1) let K be a non-trivial subquasivariety of iBCS. Then K is a non-trivial subquasivariety of PBCK by Proposition 2.3.5 and so iBCK  $\subseteq$  K by Example 2.1.4. For (2), let K be a non-trivial subquasivariety of iBCS such that  $K \subseteq$  BCK. From  $K \models x \setminus (y \setminus x) \approx x$  and  $K \subseteq$  BCK we have that  $K \subseteq$  iBCK; since there are no non-trivial subquasivarieties of iBCK (by [131, Theorem 2] and [39, Theorem 4.4]) we infer that K is iBCK. The remaining claims now follow from Theorem 2.2.2.

Although PBCK has been encountered previously in the literature as a generalisation of BCK that fails in any sense to be point regular (recall Lemma 2.1.6 and Lemma 2.1.9), according to Iséki [124] the variety of implicative BCSalgebras has not previously been considered in the literature in any context. Nonetheless, individual members of iBCS have been employed in a number of studies of BCK-algebras and subtractive varieties; such algebras have invariably belonged to the following class of examples.

**Example 2.3.8.** Let  $\mathbf{B}_n := \langle B_n; \rangle, 0 \rangle$  be an algebra of cardinality n + 1 equipped with a distinguished element 0 and a binary operation  $\backslash$  defined by:

$$ab := \begin{cases} a & ext{if } b = 0 \\ 0 & ext{otherwise} \end{cases}$$

for any  $a, b \in B_n$ . Then  $\mathbf{B}_n$  is a *flat* implicative BCS-algebra (on n + 1 elements). In particular, the algebra  $\mathbf{B}_2$  of Example 2.1.5 is a flat implicative BCS-algebra (on 3 elements); of course,  $\mathbf{B}_2$  is the simplest example of an implicative BCS-algebra that is not an (implicative) BCK-algebra.

Remark 2.3.9. Let  $\mathbf{B}_n := \langle B_n; \langle , 0 \rangle$  be a flat implicative BCS-algebra on n+1 elements. Let  $\theta$  be an equivalence relation on  $B_n - \{0\}$  and let  $\phi := \theta \cup \{\langle 0, 0 \rangle\}$ . An easy inspection of  $\mathbf{B}_n$  shows  $\phi$  is a congruence on  $\mathbf{B}_n$ . Suppose now that  $a \in B_n$  and  $b \neq 0$ . By inspection of  $\mathbf{B}_n$  and Lemma 2.1.42 we have  $a = a \Box b \equiv_{\Theta^{\mathbf{B}_n}(0,b)} a \Box 0 = 0$ , whence  $\iota_{\mathbf{B}_n} = \Theta^{\mathbf{B}_n}(b,0)$ . Therefore the lattice  $\mathbf{Con B}_n$  of congruences on  $\mathbf{B}_n$  is isomorphic to the lattice of equivalence relations on  $B_n - \{0\}$  together with a new largest element  $\iota_{\mathbf{B}_n} = \Theta^{\mathbf{B}_n}(b,0)$  for any  $b \neq 0$  adjoined. (In essence this observation has been made previously by both Blok and Raftery [38, p. 74] and Agliano and Ursini [8, Example 6.2].)

In [65, Lemma 4.8(i)] Cornish considered the congruence structure of primitive left handed skew Boolean algebras and proved that for any primitive left handed skew Boolean algebra **A**, the lattice **Con A** of congruences on **A** is isomorphic to the lattice of equivalence relations of the set  $A - \{0\}$ , together with a new largest element  $\iota_{\mathbf{A}} = \Theta^{\mathbf{A}}(a, 0)$  for any  $a \in A - \{0\}$  adjoined. It follows that left handed skew Boolean algebra operations may be imposed on a flat implicative BCS-algebra  $\mathbf{B}_n$ ,  $n \in \omega$ , without disturbing the congruence structure of  $\mathbf{B}_n$ , upon defining:

$$a \wedge b := \begin{cases} a & \text{if } b \neq 0 \\ 0 & \text{otherwise} \end{cases} \quad \text{and} \quad a \vee b := \begin{cases} b & \text{if } b \neq 0 \\ a & \text{otherwise} \end{cases}$$

for any  $a, b \in B_n$ .

Apropos preceding remarks, in [240] Wroński proved that the class BCK is not a variety by showing  $\mathbf{B}_2 \in \mathbf{H}(BCK)$ . Cornish extended Wroński's result in [71, Theorem 2.2], where he proved that any BCK-algebra with condition (S) can be embedded as a BCK-algebra into a BCK-algebra that has  $\mathbf{B}_2$  as a homomorphic image. In [114] Idziak exploited properties of the algebra  $\mathbf{B}_2$  in his study of the congruence *n*-permutability of BCK-varieties; various properties of the algebra  $\mathbf{B}_2$  have also been exploited by Blok and Raftery [38, Theorem 8] and van Alten [229, Proposition 4.4] in obtaining results about BCK-algebras and related structures. Most recently Blok and La Falce [25] have employed a generalisation of the algebra  $\mathbf{B}_2$  in their study of certain 'varietising' identities arising naturally in algebraic logic. Also, in [8, Example 6.2] Agliano and Ursini exhibited the variety  $V(B_5)$  generated by the algebra  $B_5$  as an example of a subtractive variety that is not ideal determined, and in [11, Example 3.7] again used  $B_5$  in proving that the variety of MINI-algebras (recall Example 1.7.11) does not have EDPC. More generally, all the algebras  $B_n$ ,  $n \in \omega$ , have been considered briefly by Blok and Raftery in the context of their study of the lattice of subquasivarieties of BCK-algebras: see [38, Section 4] for details.

In the context of BCK-algebras, the following example, due in essence to Wroński [241], typifies many of the considerations encountered in the preceding applications and studies of (flat) implicative BCS-algebras.

Example 2.3.10. (cf. [241, Applications]) Recall from [125, Example 1] or [38, Example 1] that the set  $\omega$  of all non-negative integers is the universe of a BCK-algebra  $\omega := \langle \omega; -\omega, 0 \rangle$ , where BCK difference is defined naturally by:

$$a \stackrel{\cdot}{-} {}^{\omega}b := \begin{cases} a - b & \text{if } a \ge b \\ 0 & \text{otherwise} \end{cases}$$

for any  $a, b \in \omega$ . In [241] Wroński employed  $\omega$  in constructing a family of BCKalgebras that generalise his example  $W_1(\omega)$  [240] showing that BCK-algebras do not form a variety. Let  $R(\omega) := \{r_m : m \in \omega\}$  and let  $I_n := \{0, \ldots, n-1\}$ for every  $n = 2, 3, \ldots$  Let  $N_n := \langle (I_n \times R(\omega)) \cup \omega; - N_n, 0 \rangle$  be the algebra with distinguished element 0 and binary operation  $- N_n$  defined as follows [241, p. 222]:

$$\begin{aligned} a \stackrel{\cdot}{-} \stackrel{\mathbf{N}_n}{b} &:= a \stackrel{\cdot}{-} \stackrel{\omega}{b} \\ a \stackrel{\cdot}{-} \stackrel{\mathbf{N}_n}{\langle i, r_b \rangle} &:= 0 \\ \langle i, r_b \rangle \stackrel{\cdot}{-} \stackrel{\mathbf{N}_n}{a} &:= \langle i, r_{b+a} \rangle \\ \langle i, r_a \rangle \stackrel{\cdot}{-} \stackrel{\mathbf{N}_n}{\langle j, r_b \rangle} &:= (b + |j - i|) \stackrel{\cdot}{-} \stackrel{\omega}{-} a \end{aligned}$$

for any  $i, j \in I_n$  and  $a, b \in \omega$ . By Theorem 1 and Theorem 3 of [241]  $N_n$  is a BCK-algebra. Consider now the equivalence relation  $\theta$  on the base set  $(I_n \times R(\omega)) \cup \omega$  of  $N_n$  induced by the partition  $\{\{0\} \times R(\omega), \ldots, \{n-1\} \times$ 

 $R(\omega), \omega$ . Clearly  $\theta$  is a congruence relation on  $N_n$ , and it is easy to see that the quotient algebra  $N_n/\theta$  is a flat implicative BCS-algebra on n + 1 elements. By Remark 2.3.9 the congruences on  $N_n/\theta$  other than the universal congruence are in one-to-one correspondence with the partitions of  $N_n/\theta - \{[0]_\theta\}$ , while by Wroński [241, Theorem 4] the congruences of  $N_n$  other than the universal and identity congruences are in one-to-one correspondence with partitions of the set of indices  $I_n$ . It readily follows that the congruences of  $N_n$  other than the identity congruence are in one-to-one correspondence with the congruences of  $N_n/\theta$ , and hence that  $\langle \operatorname{Con} N_n - \{\omega_{N_n}\}; \subseteq \rangle \cong \operatorname{Con} N_n/\theta$ . From this observation and Remark 2.3.9 we immediately obtain Wroński's results [241, Theorem 5, Theorem 6] that the congruence lattice of  $N_\omega$  obeys no special lattice identities at all, and that  $N_\omega$  is not congruence *m*-permutable for any  $m \in \omega$ ; cf. [38, Section 1].

It is natural to ask if implicative BCS-algebras distinct from flat algebras occur readily in universal algebra and/or algebraic logic. The following example shows (non-flat) implicative BCS-algebras arise naturally from flat implicative BCS-algebras in a large class of varieties occurring in universal algebra.

**Example 2.3.11.** Let V be a binary discriminator variety with binary discriminator term b(x, y) and let  $K \subseteq V$  be a class of binary discriminator algebras generating V as a variety. Let  $A \in K$ . By definition of the binary discriminator,

 $b^{\mathbf{A}}(a,c) = \begin{cases} a & \text{if } c = 0\\ 0 & \text{otherwise} \end{cases}$ 

for any  $a, c \in A$ , whence  $\langle A; b^{\mathbf{A}}, 0 \rangle$  is a flat implicative BCS-algebra by Example 2.3.8. Since the identities satisfied by V are precisely those satisfied by K, it follows that any  $\mathbf{B} \in V$  has a canonical implicative BCS-algebra polynomial reduct  $\langle B; b^{\mathbf{B}}, 0 \rangle$ .

By way of illustration we give a practical application of Example 2.3.11.

Example 2.3.12. (cf. [10, Example 4.4]) The variety PCSL of pseudocomplemented semilattices is a binary discriminator variety with binary discriminator term  $x \wedge y^*$ . Thus for any  $A \in PCSL$  the polynomial reduct  $\langle A; \rangle, 0 \rangle$  is an implicative BCS-algebra, where  $a \setminus b := a \wedge b^*$  for any  $a, b \in A$ .

*Proof.* Let 3 be the 3-element chain (considered as a pseudocomplemented semilattice). An easy inspection of the induced algebra  $\langle 3; \rangle, 0 \rangle$  shows it to be a flat implicative BCS-algebra. Since 3 generates PCSL as a variety (by Theorem 1.3.8), we infer that the variety of pseudocomplemented semilattices is a binary discriminator variety with binary discriminator term  $x \wedge y^*$ . By Example 2.3.11 it follows that the polynomial reduct  $\langle A; \rangle, 0 \rangle$  of any  $A \in PCSL$  is an implicative BCS-algebra.

This last may also be observed directly in an easy and instructive proof. Let  $A \in PCSL$  and let  $a, b, c \in A$ . We verify that the defining identities for implicative BCS-algebras are satisfied. For (2.47) we have:

$$a \setminus a = a \wedge a^*$$
  
= 0 by (1.8).

For (2.48) we have:

$$(a \setminus b) \setminus c = (a \land b^*) \land c^*$$
$$= (a \land c^*) \land b^*$$
$$= (a \setminus c) \setminus b.$$

For (2.49) we have:

(

$$\begin{aligned} a \setminus c \rangle \setminus (b \setminus c) &= (a \wedge c^*) \wedge (b \wedge c^*)^* \\ &= a \wedge (c^* \wedge (b \wedge c^*)^*) \\ &= a \wedge (c^* \wedge (c^* \wedge b)^*) \\ &= a \wedge (c^* \wedge b^*) \qquad \text{by (1.9)} \\ &= (a \wedge b^*) \wedge c^* \\ &= (a \setminus b) \setminus c. \end{aligned}$$

For (2.50) we have:

$$a \setminus (b \setminus a) = a \wedge (b \wedge a^*)^*$$
  
=  $a \wedge (a \wedge (b \wedge a^*))^*$  by (1.9)  
=  $a \wedge ((a \wedge a^*) \wedge b)^*$   
=  $a \wedge (0 \wedge b)^*$  by (1.8)  
=  $a \wedge 0^*$   
=  $a$  by (1.10).

Thus  $\langle A; \rangle, 0 \models (2.47)$ -(2.50), and the proof is complete.

Implicative BCS-algebras distinct from flat algebras (but which nonetheless arise from such structures) also occur quite naturally in a fairly wide class of varieties occurring in algebraic logic, as the following example shows.

Example 2.3.13. (cf. [65, p. 290]; cf. [19, Theorem 4.7]) Let V be a fixed point discriminator variety with 0. Let  $K \subseteq V$  be a class of fixed point discriminator algebras generating V as a variety and suppose that  $0^{\mathbf{A}}$  is the discriminating element on any  $\mathbf{A} \in K$ . Let f(x, y, z) be a discriminator term for V and let  $x \setminus y := f(0, y, x)$ . Let  $\mathbf{A} \in K$ . By definition of the fixed point discriminator,

$$a \setminus^{\mathbf{A}} b = f^{\mathbf{A}}(0, b, a)$$
$$= \begin{cases} a & \text{if } b = 0\\ 0 & \text{otherwise} \end{cases}$$

for any  $a, b \in A$ , whence  $\langle A; \setminus^{\mathbf{A}}, 0 \rangle$  is a flat implicative BCS-algebra by Example 2.3.8. Since the identities satisfied by V are precisely those satisfied by K it follows that any  $\mathbf{B} \in V$  has a canonical implicative BCS-algebra polynomial reduct  $\langle B; \setminus^{\mathbf{B}}, 0 \rangle$ .

To further illustrate Example 2.3.13 we give a concrete application.

Example 2.3.14. Let  $ce_n BCK$ ,  $n \in \omega$ , denote the intersection of the varieties cBCK and  $e_n BCK$ . By the remarks of §1.6.13 the classes  $ce_n BCK$  are

varieties that form an infinite strictly increasing chain. For each  $n \in \omega$ , let  $|ce_nBCK|$  denote the subclass of  $ce_nBCK$  satisfying the identity:

 $(x - y) \cap (y - x) \approx 0.$ 

Concerning the classes ice<sub>n</sub>BCK,  $n \in \omega$ , the following assertions hold:

- 1. For each  $n \in \omega$ , lce<sub>n</sub>BCK is a fixed point discriminator variety with fixedpoint discriminator term  $f(x, y, z) := (z \div (x \div y)^n) \div (y \div z)^n$ ;
- 2. For each  $n \in \omega$ , lce<sub>n</sub>BCK is generated by a single fixed point discriminator algebra  $\mathbf{A}_n$ , for which  $\mathbf{0}^{\mathbf{A}_n}$  is the discriminating element.

Thus for any  $A \in \text{lce}_n BCK$ ,  $n \in \omega$ , the polynomial reduct  $\langle A; \rangle, 0 \rangle$  is an implicative BCS-algebra, where  $a \backslash b := f^A(0, b, a) = ab^n$  for any  $a, b \in A$ .

Proof. For (1), by Theorem 1.6.15(1) each variety  $ce_n BCK$ ,  $n \in \omega$ , has a commutative TD term  $f(x, y, z) := (z \div (x \div y)^n) \div (y \div x)^n$ , just because it is a subvariety of  $e_n BCK$ . Since each  $ce_n BCK$  is semisimple (by Cornish [68, Corollary 3.2]), from Theorem 1.5.12 we have that  $ce_n BCK$  is a fixed point discriminator variety with fixed point discriminator term f(x, y, z). By Cornish [70, Theorem 5.7; Section 3.6] the result now follows for  $lce_n BCK$ ,  $n \in \omega$ .

For (2) let  $A_{\omega}$  be a chain of order type  $\omega$ , say  $0 < a_0 < a_1 \dots < a_c < \dots$  Let  $A_{\omega} := \langle A_{\omega}; -A_{\omega}, 0 \rangle$  where  $a_i - A_{\omega} a_j := a_{\max\{i-j,0\}}$  for any  $a_i, a_j \in A_{\omega}$ . Let  $A_k$  denote the subalgebra  $(a_k]$  of  $A_{\omega}$ . By [68, Theorem 3.5], lce<sub>k</sub>BCK =  $V(A_k)$ . Because  $A_k$  is simple (by Cornish [70, Section 3.6]), from Theorem 1.5.12 it follows that  $A_k$  is a fixed point discriminator algebra. Moreover, from the description of  $A_k$  we may infer additionally that  $O^{A_k}$  is the discriminating element. From these remarks it follows that, for each  $n \in \omega$ , the variety lce<sub>n</sub>BCK is generated as a variety by the fixed point discriminator algebra  $A_n$ , for which  $O^{A_n}$  is the discriminating element.

For the final claim, let  $A \in \text{Ice}_n BCK$ ,  $n \in \omega$ , and let  $a, b \in A$ . By Example 2.3.13 the polynomial reduct  $\langle A; \rangle, 0 \rangle$  is an implicative BCS-algebra.

Moreover,

$$a \setminus b = f^{\mathbf{A}}(0, b, a)$$
  
=  $(a(0b)^{n})(b0)^{n}$   
=  $(a0^{n})b^{n}$  by (1.4), (1.7)  
=  $ab^{n}$  by [68, Lemma 1.1(ii)]

and the result follows.

**Remark 2.3.15.** The proof of Example 2.3.14(1) also shows that, for each  $n \in \omega$ , the class  $\operatorname{ce_nBCK}$  is a fixedpoint discriminator variety with fixedpoint discriminator term  $f(x, y, z) := (z \div (x \div y)^n) \div (y \div x)^n$ . However, for no  $\operatorname{ce_nBCK}$ ,  $n \in \omega$ , does any subclass of  $\{\mathbf{A}_k : k \in \omega\}$  (where the  $\mathbf{A}_k$  are as in the proof of Example 2.3.14(2)) generate  $\operatorname{ce_nBCK}$  as a variety: see Cornish [70, Lemma 5.6] (and also Komori [137, Theorem 3.13] and Iséki and Tanaka [125, Example 5]). Therefore we may not conclude from the proof of Example 2.3.14(2) that each  $\operatorname{ce_nBCK}$ ,  $n \in \omega$ , is generated as a variety by a class  $\mathsf{K} \subseteq \operatorname{ce_nBCK}$  of fixedpoint discriminator algebras such that  $\mathbf{0}^A$  is the discriminating element on any  $\mathbf{A} \in \mathsf{K}$ . However, see Example 3.2.30 in the sequel.

The discussion of Example 2.3.14 can potentially be placed in the wider context of the classes  $e_nBCK$ ,  $n \in \omega$ . Let A be a positive implicative BCK-algebra and let  $a, b \in A$ . By Guzmán [105, Proposition 3.2(c)], the underlying BCK partial ordering  $\leq_1$  on A is a meet semilattice ordering with greatest lower bound  $a \cap_1 b := (a(ab))(ba)$  such that every principal  $\leq_1$ -order ideal of  $\langle A; \leq_1 \rangle$ is a Boolean lattice. (In other words,  $\langle A; \cap_1 \rangle$  is semi-Boolean.) Since any BCK-algebra satisfying  $(E_n)$  also satisfies:

$$(x \stackrel{\cdot}{-} (x \stackrel{\cdot}{-} y)^n) \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x)^n \approx (y \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x)^n) \stackrel{\cdot}{-} (x \stackrel{\cdot}{-} y)^n$$

by Lemma 1.3 of Cornish [68], it is natural to pose the following problem.

**Problem 2.3.16.** Let  $A \in e_n BCK$ ,  $n \in \omega$ . For any  $a, b \in A$ , let  $a \cap_n b := (a(ab)^n)(ba)^n$ . Is the derived algebra  $\langle A; \cap_n \rangle$  always a meet semilattice? If so,

let  $\leq_n$  denote the underlying partial order of  $\langle A; \cap_n \rangle$ . Does  $\langle A; \leq_n \rangle$  support in any sense a semi-Boolean or other 'locally Boolean' structure?

Let A be a positive implicative BCK-algebra. Because the underlying  $\leq_1$ ordering of A is semi-Boolean, A has an implicative BCK-algebra polynomial
reduct  $\langle A; /, 0 \rangle$ , where  $a/b := a(a \cap_1 b)^1$  for any  $a, b \in A$ ; for details, see
Theorem 3.3 of Guzmán [105]. This remark prompts the following problem.

**Problem 2.3.17.** Let  $A \in e_n BCK$ ,  $n \in \omega$ . Let  $a \setminus b := a(a \cap_n b)^n$  for any  $a, b \in A$ . Is the induced algebra  $\langle A; \setminus, 0 \rangle$  always an implicative BCS-algebra?

It is known and quite easy to see [68, p. 419] that any finite BCK-algebra satisfies  $(E_n)$  for some  $n \in \omega$ . This observation in conjunction with preceding remarks suggests the problem below.

**Problem 2.3.18.** Does every finite BCK-algebra have an implicative BCSalgebra polynomial reduct?

In view of the preceding discussion, binary and 'pointed' fixedpoint discriminator varieties may not exhaust those classes of algebras in which implicative BCS-algebras arise naturally. We return to this point in Example 3.2.20 in the sequel.

2.3.19. Left Handed Locally Boolean Bands. Theorem 1.6.21 shows that the derived semilattice meet  $\cap$  plays a fundamental role in the theory of implicative BCK-algebras, even to the point of determining some second-order properties (see Cornish [64, Section 3]). This observation, in conjunction with Proposition 2.3.5 and Proposition 2.2.31, prompts the study of the role played by the pre-BCK meet  $\sqcap$  (hereafter, *implicative BCS meet*  $\sqcap$ ) in implicative BCS-algebras.

**Proposition 2.3.20.** The variety of implicative BCS-algebras satisfies the following identities:

$$(x \sqcap y) \setminus (z \sqcap z) \approx (x \setminus z) \setminus (x \setminus y)$$
(2.57)

$$(x \sqcap y) \setminus z \approx (x \setminus z) \setminus (x \setminus y) \tag{2.58}$$

$$(x \setminus z) \sqcap y \approx (x \setminus z) \setminus (x \setminus y) \tag{2.59}$$

$$(x \setminus z) \sqcap (y \setminus z) \approx (x \setminus z) \setminus (x \setminus y)$$
(2.60)

$$x \sqcap (y \setminus z) \approx (x \setminus z) \setminus (x \setminus y) \tag{2.61}$$

$$(x \sqcap y) \setminus (z \sqcap y) \approx (x \setminus z) \setminus (x \setminus y).$$
(2.62)

*Proof.* Let A be an implicative BCS-algebra and let  $a, b, c \in A$ . For (2.57), we have:

$$(a \sqcap b)(a \sqcap c) = (a(ab))(a(ac))$$
  
=  $(a(a(ac)))(ab)$  by (2.48)  
=  $(ac)(ab)$  by (2.56).

For (2.58), we have:

$$(a \sqcap b)c = (a(ab))c$$
  
= (ac)(ab) by (2.48).

For (2.59), it is sufficient by (2.58) to show  $(ac) \sqcap b = (a \sqcap b)c$ . We have:

$$(ac) \sqcap b = (ac)((ac)b)$$
  
=  $(ac)((ab)c)$  by (2.48)  
=  $(a(ab))c$  by (2.49)  
=  $(a \sqcap b)c$ .

For (2.60), it is sufficient by (2.59) to show  $(ac) \sqcap (bc) = (ac) \sqcap b$ . We have:

$$(ac) \sqcap (bc) = (ac)((ac)(bc))$$
  
=  $(ac)((ab)c)$  by (2.49)  
=  $(ac)((ac)b)$  by (2.48)  
=  $(ac) \sqcap b.$ 

165
a

For (2.61), it is sufficient by (2.60) to show  $a \sqcap (bc) = (ac) \sqcap (bc)$ . We have:

For (2.62), it is sufficient by (2.61) to show  $(a \sqcap b)(c \sqcap b) = a \sqcap (bc)$ . So let  $\alpha := a, \beta := b$  and  $\gamma := c \sqcap b$ . We have:

$$(a \sqcap b)(c \sqcap b) = (\alpha \sqcap \beta)\gamma$$
  
=  $(\alpha\gamma)(\alpha\beta)$  by (2.58)  
=  $\alpha \sqcap (\beta\gamma)$  by (2.61)  
=  $a \sqcap (b(c \sqcap b))$   
=  $a \sqcap (b(c(cb)))$   
=  $a \sqcap (bc)$  by (2.53).

**Proposition 2.3.21.** The variety of implicative BCS-algebras satisfies the following identities:

$$x \sqcap (y \sqcap z) \approx (x \sqcap y) \sqcap z \tag{2.63}$$

$$x \sqcap (y \sqcap z) \approx x \sqcap (z \sqcap y). \tag{2.64}$$

*Proof.* Let A be an implicative BCS-algebra and let  $a, b, c \in A$ . For (2.63),

we have:

$$(a \sqcap b) \sqcap c = (a(ab))((a(ab))c)$$
  
=  $(a(ab))((a \sqcap b)c)$   
=  $(a(ab))((ac)(ab))$  by (2.58).

Put  $\alpha := a$ ,  $\beta := ab$  and  $\gamma := (ac)(ab)$ . We have:

$$(a(ab))((ac)(ab)) = (\alpha\beta)\gamma$$
  
=  $(\alpha\gamma)\beta$  by (2.48)  
=  $(a((ac)(ab)))(ab).$ 

Put  $\alpha := a, \beta := b$  and  $\gamma := c$ . We have:

$$(a((ac)(ab)))(ab) = (a((\alpha\gamma)(\alpha\beta)))(ab)$$
  
=  $(a(\alpha \sqcap (\beta\gamma)))(ab)$  by (2.61)  
=  $(a(a \sqcap (bc)))(ab)$   
=  $(a(a(a(bc))))(ab).$ 

Put  $\alpha := a$  and  $\beta := bc$ . We have:

$$(a(a(a(bc))))(ab) = (\alpha(\alpha(\alpha\beta)))(ab)$$
$$= (\alpha\beta)(ab) \qquad by (2.56)$$
$$= (a(bc))(ab).$$

Put  $\alpha := a, \beta := b$  and  $\gamma := bc$ . We have:

$$(a(bc))(ab) = (\alpha\gamma)(\alpha\beta)$$
  
=  $\alpha \sqcap (\beta\gamma)$  by (2.61)  
=  $a \sqcap (b(bc))$   
=  $a \sqcap (b \sqcap c)$ .

For (2.64), put  $\alpha := a$ ,  $\beta := ab$  and  $\gamma := c$ . We have:

$$a \sqcap (b \sqcap c) = (a \sqcap b) \sqcap c$$
  
=  $(a(ab))((a(ab))c)$   
=  $(a(ab))((\alpha\beta)\gamma)$   
=  $(a(ab))((\alpha\gamma)\beta)$  by (2.48)  
=  $(a(ab))((ac)(ab)).$ 

Put  $\alpha := a$ ,  $\beta := ab$  and  $\gamma := ac$ . We have:

$$(a(ab))((ac)(ab)) = (\alpha\beta)(\gamma\beta)$$
  
=  $(\alpha\gamma)(\beta\gamma)$  by (2.55)  
=  $(a(ac))((ab)(ac)).$ 

Put  $\alpha := a$ ,  $\beta := b$  and  $\gamma := ac$ . We have:

$$\begin{aligned} [a(ac))((ab)(ac)) &= (a(ac))((\alpha\beta)\gamma) \\ &= (a(ac))(\alpha\gamma)\beta) & \text{by } (2.48) \\ &= (a(ac))((a(ac))b) \\ &= (a \sqcap c) \sqcap b \\ &= a \sqcap (c \sqcap b). \end{aligned}$$

Recall from [206, p. 295] that a restrictive bisemigroup is an algebra  $\langle A; \Delta_L, \Delta_R \rangle$  of type  $\langle 2, 2 \rangle$  such that: (i) the reduct  $\langle A; \Delta_L \rangle$  is a left normal band; (ii) the reduct  $\langle A; \Delta_R \rangle$  is a right normal band; and (iii) the following associativity condition is satisfied:

$$(x \triangle_R y) \triangle_L z \approx x \triangle_R (y \triangle_L z). \tag{2.65}$$

Restrictive bisemigroups were introduced by Schein in [206] in connection with the theory of binary relations [228]. Let A and B be sets and let  $\rho$  and  $\sigma$  be

binary relations from A into B. The restrictive composition of the first kind of  $\rho$  and  $\sigma$  is the binary relation  $\triangleright \subseteq A \times B$  defined by [207, p. 309]:

$$\rho \rhd \sigma := (\pi_1(\rho) \times B) \cap \sigma$$

where  $\pi_1$  is the first projection map. Analogously the restrictive composition of the second kind of  $\rho$  and  $\sigma$  is the binary relation  $\lhd \subseteq A \times B$  defined by [207, p. 309]:

 $\rho \lhd \sigma := \rho \cap (A \times \pi_2(\sigma))$ 

where  $\pi_2$  is the second projection map. Upon denoting the set of all one-to-one binary relations from A to B by  $\mathbb{R}(A, B)$ , the structure  $\langle \mathbb{R}(A, B); \triangleright, \triangleleft \rangle$  is a restrictive bisemigroup, the restrictive subbisemigroups of which are known as *restrictive bisemigroups of invertible mappings*.

**Corollary 2.3.22.** For any implicative BCS-algebra A, the following assertions hold:

- 1. A has a left normal band with zero polynomial reduct  $\langle A; \sqcap_L, 0 \rangle$ , where  $a \sqcap_L b := a \sqcap b$  for any  $a, b \in A$ ;
- 2. A has a right normal band with zero polynomial reduct  $\langle A; \sqcap_R, 0 \rangle$ , where  $a \sqcap_R b := b \sqcap a$  for any  $a, b \in A$ ;
- 3. A has a restrictive bisemigroup polynomial reduct  $\langle A; \sqcap_L, \sqcap_R \rangle$ ;
- 4. The restrictive bisemigroup polynomial reduct  $\langle A; \sqcap_L, \sqcap_R \rangle$  of (3) is isomorphic to a restrictive bisemigroup of invertible mappings.

Proof. Let A be an implicative BCS-algebra and let  $a \sqcap_L b := a \sqcap b$  for any  $a, b \in A$ . For (1), just note that the identities (2.63)-(2.64) in conjunction with the identities (2.21)-(2.23) of Lemma 2.1.42 assert that the polynomial reduct  $\langle A; \sqcap_L, 0 \rangle$  is a left normal band with zero. For (2), let  $a \sqcap_R b := b \sqcap a$  for any  $a, b \in A$ . An easy sequence of checks shows that the derived algebra  $\langle A; \sqcap_R, 0 \rangle$  is a right normal band with zero. For (3), recall from Schein [207, p. 313] that any left normal band  $\langle A; \bigtriangleup_L \rangle$  induces a restrictive bisemigroup

 $\langle A; \Delta_L, \Delta_R \rangle$  upon defining an inverse operation  $\Delta_R$  by  $a \Delta_R b := b \Delta_L a$  for any  $a, b \in A$ . For (4), just note by Schein [207, Theorem 5; pp. 313-314] that any restrictive bisemigroup induced by a left normal band in the above manner is isomorphic to a restrictive bisemigroup of invertible mappings.

**Remark 2.3.23.** It is entirely arbitrary whether we consider in general the left normal band with zero polynomial reduct or the right normal band with zero polynomial reduct of an implicative BCS-algebra A. In the sequel we will exclusively consider only the left normal band with zero polynomial reduct, denoting it simply  $\langle A; \Box, 0 \rangle$ , bearing in mind that all results obtained extend to the right normal case.

Because of Corollary 2.3.22(1), implicative BCS-algebras enjoy equationally definable properties of left normal bands with zero. Since the Green's quasiorderings  $\leq_{\mathcal{L}}$ ,  $\leq_{\mathcal{R}}$  and  $\leq_{\mathcal{D}}$  and the natural partial ordering  $\leq_{\mathcal{H}}$  are term definable on any band, we have that they are also definable on any implicative BCS-algebra.

Lemma 2.3.24. Let A be an implicative BCS-algebra and let  $\leq$  be the binary relation defined on A by  $a \leq b$  iff ab = 0. For any  $a, b \in A$ ,  $a \leq b$  iff  $a \leq_{\mathcal{D}}^{\langle A; \sqcap, 0 \rangle} b$  iff  $a \leq_{\mathcal{L}}^{\langle A; \sqcap, 0 \rangle} b$  iff  $a \sqcap b = a$ , and so  $\langle A; \leq \rangle$  is a quasiordered set with least element 0. Moreover, the relation  $\leq$  satisfies the following conditions for any  $a, b, c \in A$ :

- 1. If  $a \leq b$  then  $cb \leq ca$ ;
- 2. If  $a \leq b$  then  $ac \leq bc$ ;
- 3. If  $a \preceq b$  then  $a \sqcap c \preceq b \sqcap c$ ;
- 4. If  $a \preceq b$  then  $c \sqcap a \preceq c \sqcap b$ .

*Proof.* For the first assertion,  $\langle A; \leq \rangle$  is a quasiordered set with 0 as least element (by Lemma 2.1.12) for which Lemma 2.1.42(1) ensures ab = 0 iff  $a \sqcap b = a$ , and for which the remarks of §1.3.11 and §1.3.15 ensure  $a \sqcap b = a$  iff  $a \leq_{\mathcal{L}}^{\langle A; \sqcap, 0 \rangle}$  iff  $a \leq_{\mathcal{D}}^{\langle A; \sqcap, 0 \rangle} b$ . It remains to prove Items (1)-(4). Items (1) and (2) have already been established in Lemma 2.1.12(1), (2) respectively.

Because  $\preceq_{\mathcal{D}}^{\langle A; \sqcap, 0 \rangle}$  and  $\preceq$  coincide, Items (3) and (4) follow from a standard result of semigroup theory asserting that the  $\mathcal{D}$ -quasiorder on an arbitrary band is compatible with band multiplication: see Schein [208, Proposition 1].

**Proposition 2.3.25.** Let A be an implicative BCS-algebra and let  $\leq$  be the binary relation defined on A by  $a \leq b$  iff  $a \sqcap b = a = b \sqcap a$ . For any  $a, b \in A$ ,  $a \leq_{\mathcal{H}}^{(A; \sqcap, 0)}$  iff  $a \preceq_{\mathcal{R}}^{(A; \sqcap, 0)} b$  iff  $a \leq b$ ,  $\langle A; \leq \rangle$  is a partially ordered set with least element 0, and  $\preceq$  is admissible with respect to  $\leq$ . Moreover,  $\leq$  satisfies the following conditions for any  $a, b, c \in A$ :

- 1. If  $a \leq b$  then  $cb \leq ca$ ;
- 2. If  $a \leq b$  then  $ac \leq bc$ .
- 3. If  $a \leq b$  then  $a \sqcap c \leq b \sqcap c$ ;
- 4. If  $a \leq b$  then  $c \sqcap a \leq c \sqcap b$ ;

Proof. For the first assertion,  $a \leq b$  iff  $a \sqcap b = a = b \sqcap a$  iff  $a \leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} b$ , which implies  $\langle A; \leq \rangle$  is a partially ordered set by the remarks of §1.3.11. Moreover,  $0 \leq a$  for any  $a \in A$  and  $\preceq$  is admissible with respect to  $\leq$  by Lemma 2.1.42. It remains to establish  $a \leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} b$  iff  $a \preceq_{\mathcal{R}}^{\langle A; \sqcap, 0 \rangle} b$ . It is clear by definition that  $a \leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} b$  implies  $a \preceq_{\mathcal{R}}^{\langle A; \sqcap, 0 \rangle} b$ . For the converse, assume  $a \preceq_{\mathcal{R}}^{\langle A; \sqcap, 0 \rangle} b$ ; it is sufficient to show  $a \preceq_{\mathcal{L}}^{\langle A; \sqcap, 0 \rangle} b$ . By left normality,  $a \sqcap b = a \sqcap b \sqcap a = a \sqcap (b \sqcap a) = a \sqcap a = a$ , so  $a \preceq_{\mathcal{L}}^{\langle A; \sqcap, 0 \rangle} b$ .

It remains to prove Items (1)-(4). Items (3) and (4) follow from left normality and Lemma 1.3.16. For (1), assume  $a \leq b$ . Then  $a \leq b$  and  $cb \leq ca$  by Lemma 2.3.24(1), whence:

$$(cb) \sqcap (ca) = (cb)((cb)(ca))$$
  
= (cb)0  
= cb by (2.6).

For the opposite inclusion, assume  $a \leq b$ . We have:

$$(ca) \sqcap (cb) = ((ca)b)((ca)c) \qquad by (2.61)$$
$$= ((ca)b)0 \qquad by Lemma 2.1.12(3)$$
$$= (ca)b \qquad by (2.6)$$
$$= (cb)a \qquad by (2.48)$$
$$= (cb)(b(ba)) \qquad since b \sqcap a = a$$
$$= cb \qquad by (2.54).$$

Thus  $cb \leq ca$ , and (1) is proved. For (2), assume  $a \leq b$ . Then  $a \leq b$  and  $ac \leq bc$  by Lemma 2.3.24(2), whence:

$$(ac) \sqcap (bc) = (ac)((ac)(bc))$$
$$= (ac)0$$
$$= ac \qquad by (2.6).$$

For the opposite inclusion, assume  $a \leq b$ . We have:

$$(bc) \sqcap (ac) = ((bc)c)((bc)a) \qquad by (2.61)$$
$$= (bc)((bc)a) \qquad by (2.51)$$
$$= (bc) \sqcap a$$
$$= (bc)(ba) \qquad by (2.59)$$
$$= (b(ba))c \qquad by (2.48)$$
$$= ac \qquad nce \ b \sqcap a =$$

Thus  $ac \leq bc$ , and the proof is complete.

By Lemma 2.3.24 and Proposition 2.3.25, small finite implicative BCS-algebras may be depicted graphically using the Hasse diagramming conventions of §1.4.18. We provide a concrete example by way of illustration.

a.

**Example 2.3.26.** Using Lemma 2.3.3 and Proposition 2.3.4 the free implicative BCS-algebra  $\mathbf{F}(\overline{x}, \overline{y})$  on two free generators  $\overline{x}, \overline{y}$  may be determined simply

by computing all products involving  $\overline{x}$  and  $\overline{y}$ . The resulting operation tables for  $F(\overline{x}, \overline{y})$  and its left normal band with zero polynomial reduct are shown below (where for simplicity of notation in the tables, we denote implicative BCS difference by juxtaposition, and also write x and y for  $\overline{x}$  and  $\overline{y}$  respectively (and similarly for products of  $\overline{x}$  and  $\overline{y}$ )).

| $ig ^{\mathbf{F}(ar{x},ar{y})}$      | 0               | x            | y                | $x_i$      | y          | yx           | $x \sqcap g$ | $y \ y \sqcap x$ |
|--------------------------------------|-----------------|--------------|------------------|------------|------------|--------------|--------------|------------------|
| 0                                    | 0               | 0            | 0                | 0          | )          | 0            | 0            | 0                |
| x                                    | x               | 0            | xy               | xГ         | l y        | x            | xy           | xy               |
| y                                    | y               | yx           | 0                | y          | 1          | $y \sqcap c$ | r yx         | yx               |
| xy                                   | xy              | 0            | xy               | 0          | )          | $xy_{.}$     | xy           | xy               |
| yx                                   | yx              | yx           | 0                | y:         | r          | 0            | yx           | yx               |
| $x\sqcap y$                          | $x \sqcap$      | y 0          | 0                | хī         | 1 <i>y</i> | $x \sqcap g$ | y 0          | 0                |
| $y \sqcap x$                         | y⊓              | <i>x</i> 0   | 0                | yГ         | l x        | $y \sqcap z$ | r 0          | 0                |
|                                      |                 |              |                  |            |            |              |              |                  |
| $\Box^{\mathbf{F}(\bar{x},\bar{y})}$ | <sup>7)</sup> 0 | x            |                  | y          | xy         | yx           | $x \sqcap y$ | $y \sqcap x$     |
| 0                                    | 0               | 0            | _                | 0          | 0          | 0            | 0            | 0                |
| x                                    | 0               | x            | x                | $\sqcap y$ | xy         | 0            | $x \sqcap j$ | ा 🗖 y            |
| $\boldsymbol{y}$                     | 0               | $y \sqcap x$ |                  | y          | 0          | yx           | $y\sqcap x$  | $y \sqcap x$     |
| xy                                   | 0               | xy           |                  | 0          | xy         | 0            | 0            | 0                |
| yx                                   | 0               | 0            | 3                | yx         | 0          | yx           | 0            | 0                |
| $x \sqcap y$                         | , 0             | $x\sqcap y$  | x                | Πy         | 0          | 0            | ្លា ហ y      | $x\sqcap y$      |
| $y \sqcap x$                         | ; 0             | $y \sqcap x$ | $\boldsymbol{y}$ | $\sqcap x$ | 0          | 0            | $y \sqcap x$ | y : T $x$        |

The Hasse diagram of  $\mathbf{F}(\overline{x}, \overline{y})$  is shown in Figure 2.2. For notational reasons, implicative BCS difference is denoted by juxtaposition in the figure; also the free generators  $\overline{x}, \overline{y}$  are denoted simply by x, y respectively (and like remarks apply to products of  $\overline{x}, \overline{y}$ ).

Let A be a pre-BCK-algebra with an underlying partial order  $\leq$  (which need not necessarily coincide with  $\leq$ ). If  $ab \leq a$  for any  $a, b \in A$  we say  $\leq$  respects pre-BCK difference. In Theorem 2.3.29 below we show that the existence of a left normal band with zero polynomial reduct  $\langle \Box, 0 \rangle$  whose underlying natural band partial order respects implicative pre-BCK difference distinguishes the

Ç



174

Figure 2.2. The iBCS-free algebra on two free generators  $\overline{x}, \overline{y}$ .

implicative BCS-algebras among the implicative pre-BCK-algebras. But first, two auxilliary lemmas; Theorem 2.3.29 shows the hypotheses of the second of these lemmas are not artificial.

**Lemma 2.3.27.** The variety of (positive) implicative pre-BCK-algebras satisfies the identity:

$$\left((x - y) - ((x - y) - z)\right) - (z - y) \approx \mathbf{0}.$$
(2.66)

Proof. Let A be a (positive) implicative pre-BCK-algebra and let  $a, b, c \in A$ . We have  $ab \leq (ab)b$ , so  $(ab)((ab)c) \leq ((ab)b)((ab)c)$  (by Lemma 2.1.12(2)), so  $((ab)((ab)c))(cb) \leq (((ab)b)((ab)c))(cb)$  (by Lemma 2.1.12(2)) = 0 by (2.1), so ((ab)((ab)c))(cb) = 0 by Lemma 2.1.12.

Lemma 2.3.28. Let A be an implicative pre-BCK-algebra such that the polynomial reduct  $\langle A; \sqcap, 0 \rangle$  is a left normal band with zero. Suppose further that the underlying natural band partial order  $\leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle}$  on the polynomial reduct  $\langle A; \sqcap, 0 \rangle$  respects implicative pre-BCK difference; that is, for any  $a, b \in A$ ,

$$ab \leq_{\mathcal{H}}^{\langle A; \, \bigcap, 0 \rangle} a. \tag{2.67}$$

Then A satisfies the following identities:

 $(x - y) \sqcap (z - y) \approx (x - y) \sqcap z \tag{2.68}$ 

2.3. Implicative BCS-Algebras

 $x \sqcap (y - z) \approx (x - z) \sqcap y \tag{2.69}$ 

 $(x - y) - z \approx (x - z) \sqcap (x - y).$ (2.70)

*Proof.* Let A be an implicative pre-BCK-algebra for which the polynomial reduct  $\langle A; \sqcap, 0 \rangle$  is a left normal band with zero and suppose further that A satisfies (2.67) for any  $a, b \in A$ . Let  $a, b, c \in A$ . For (2.68) we have:

$$(ab) \sqcap (cb) = (ab) \sqcap (c \sqcap (cb)) \qquad by (2.67)$$
$$= ((ab) \sqcap c) \sqcap (cb)$$
$$= ((ab) \sqcap c) (((ab) \sqcap c)(cb))$$
$$= ((ab) \sqcap c) (((ab)((ab)c))(cb))$$
$$= ((ab) \sqcap c) 0 \qquad by (2.66)$$
$$= (ab) \sqcap c \qquad by (2.6).$$

For (2.69), we have:

a

$$\sqcap (bc) = a \sqcap ((bc) \sqcap a)$$
 by left normality  

$$= a \sqcap ((bc) \sqcap (ac))$$
 by (2.68)  

$$= a \sqcap ((ac) \sqcap (bc))$$
 by left normality  

$$= (a \sqcap (ac)) \sqcap (bc)$$
  

$$= (ac) \sqcap (bc)$$
 by (2.67)  

$$= (ac) \sqcap b$$
 by (2.68).

For (2.70), we have:

$$(ab)c = (ab) \sqcap ((ab)c) \qquad by (2.67)$$
$$= (a \sqcap (ab)) \sqcap ((ab)c) \qquad by (2.67)$$
$$= a \sqcap ((ab) \sqcap ((ab)c))$$
$$= a \sqcap ((ab)c) \qquad by (2.67)$$

 $= (ac) \sqcap (ab) \qquad \qquad \text{by (2.69).}$ 

**Theorem 2.3.29.** For any implicative pre-BCK-algebra  $\mathbf{A}$  the following are equivalent:

- 1. A is an implicative BCS-algebra;
- 2. The polynomial reduct  $\langle A; \Box, 0 \rangle$  is a left normal band with zero, and any one of the following conditions is satisfied:
  - (a) Right [left] multiplication by a fixed element of A is isotone [antitone] with respect to the underlying natural band partial order  $\leq_{\mathcal{H}}^{(A; \, \sqcap, 0)}$ . That is, for any  $a, b, c \in A$ ,
    - i. If  $a \leq_{\mathcal{H}}^{\langle A; \ \Pi, 0 \rangle} b$  then  $cb \leq_{\mathcal{H}}^{\langle A; \ \Pi, 0 \rangle} ca;$ ii. If  $a \leq_{\mathcal{H}}^{\langle A; \ \Pi, 0 \rangle} b$  then  $ac \leq_{\mathcal{H}}^{\langle A; \ \Pi, 0 \rangle} bc.$
  - (b) The underlying natural band partial order  $\leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle}$  respects implicative pre-BCK difference. That is, for any  $a, b \in A$ ,

$$ab \leq_{\mathcal{H}}^{\langle A; \, \sqcap, 0 \rangle} a. \tag{2.67}$$

(c) A satisfies the identities:

$$x - (x \sqcap y) \approx x - y \approx x \sqcap (x - y).$$
(2.71)

*Proof.* Let A be an implicative pre-BCK-algebra. To prove the theorem we show  $(1) \Rightarrow (2)(a) \Rightarrow (2)(b) \Leftrightarrow (2)(c)$  and (2)(b),  $(2)(c) \Rightarrow (1)$ .

 $(1) \Rightarrow (2)(a)$  Suppose A is an implicative BCS-algebra. Then the polynomial reduct  $\langle A; \sqcap, 0 \rangle$  of A is a left normal band with zero (by Corollary 2.3.22(1)) such that right [left] multiplication by a fixed element of A is isotone [anti-tone] with respect to the underlying natural band partial order  $\leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle}$  (by Proposition 2.3.25(1),(2)).

Throughout the remainder of the proof, assume that the polynomial reduct  $\langle A; \Box, 0 \rangle$  of A is a left normal band with zero.

 $(2)(a) \Rightarrow (2)(b)$  Let  $a \in A$  and suppose right [left] multiplication by a fixed element of A is isotone [antitone] with respect to the underlying natural band partial order  $\leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle}$ . Since  $\langle A; \sqcap, 0 \rangle$  is a band with zero,  $0 \leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} b$  for any  $b \in A$ , whence  $ab \leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} a0 = a$  for any  $a \in A$ . Thus  $\leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle}$  respects implicative pre-BCK difference.

(2)(b)  $\Leftrightarrow$  (2)(c) Suppose  $\leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle}$  respects implicative pre-BCK difference. Then  $ab \leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} a$  for any  $a, b \in A$ , so  $a(a \sqcap b) = a(a(ab)) = a \sqcap (ab) = ab$ . Thus  $\mathbf{A} \models (2.71)$ . Conversely, suppose  $\mathbf{A} \models (2.71)$ . Let  $a, b \in A$ . By hypothesis,  $a \sqcap (ab) = ab$ ; also  $(ab) \sqcap a = (ab)((ab)a) = (ab)0 = ab$  by Lemma 2.1.12(3) and (2.6). Thus  $ab \leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} a$ .

(2)(b), (2)(c)  $\Rightarrow$  (1) Suppose both  $\mathbf{A} \models (2.71)$  and  $ab \leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} a$  for any  $a, b \in A$ . We verify directly that the defining identities (2.47)–(2.50) for implicative BCS-algebras are satisfied. So let  $a, b, c \in A$ . It is clear that (2.47) is satisfied, because  $\mathbf{A}$  is a pre-BCK-algebra. For (2.48), we have:

| $(ab)c = (ac) \sqcap (ab)$    | by (2.70)         |
|-------------------------------|-------------------|
| $= a \sqcap (ac) \sqcap (ab)$ | by (2.67)         |
| $= a \sqcap (ab) \sqcap (ac)$ | by left normality |
| $=(ab)\sqcap(ac)$             | by (2.67)         |
| = (ac)b                       | by (2.70).        |

For (2.49), put  $\alpha := a$ ,  $\beta := ac$  and  $\gamma := (ac \sqcap b)$ . We have:

$$(ac)(bc) = (a(bc)) \sqcap (ac) \qquad by (2.70)$$
$$= (a \sqcap (a(bc))) \sqcap (ac) \qquad by (2.67)$$
$$= (a(a \sqcap (bc))) \sqcap (ac) \qquad by (2.71)$$
$$= (a(ac \sqcap b)) \sqcap (ac) \qquad by (2.69)$$
$$= (\alpha\gamma) \sqcap \beta$$
$$= \alpha \sqcap (\beta\gamma) \qquad by (2.69)$$
$$= a \sqcap ((ac)((ac) \sqcap b))$$

 $= a \sqcap ((ac) \sqcap ((ac)b)) \qquad by (2.71)$  $= a \sqcap ((ac)b) \qquad by (2.67)$  $= (ab) \sqcap (ac) \qquad by (2.69)$  $= (ac)b \qquad by (2.70)$ 

$$= (ab)c.$$

For (2.50), we have:

| by (2.67) |
|-----------|
| by (2.71) |
|           |
| by (2.42) |
| by (2.6). |
|           |

**Remark 2.3.30.** The hypothesis that the natural band partial order respect implicative pre-BCK difference cannot be omitted in the assertion of Theorem 2.3.29. To see this, consider the 4-element implicative pre-BCK-algebra **A** of Remark 2.3.6, whose polynomial reduct  $\langle A; \sqcap, 0 \rangle$  has the following operation table:

| ⊓▲ | 0 | a | b | С |  |
|----|---|---|---|---|--|
| 0  | 0 | 0 | 0 | 0 |  |
| a  | 0 | a | 0 | a |  |
| b  | 0 | 0 | b | 0 |  |
| с  | 0 | с | 0 | С |  |

An easy sequence of checks shows that  $\langle A; \sqcap, 0 \rangle$  is a left normal band with zero. However, the natural band partial order  $\leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle}$  does not respect implicative pre-BCK difference, since  $ab = c \not\leq_{\mathcal{H}}^{\langle A; \sqcap, 0 \rangle} a$ ; and **A** is not an implicative BCS-algebra by the remarks of Example 2.3.6.

Recall from Theorem 1.6.21 that every implicative BCK-algebra  $\langle A; /, 0 \rangle$  has a

semi-Boolean algebra polynomial reduct  $\langle A; \cap, 0 \rangle$ , where  $a \cap b := a/(a/b)$  for any  $a, b \in A$ , and conversely that every semi-Boolean algebra  $\langle A; \cap, 0 \rangle$  induces an implicative BCK-algebra  $\langle A; /, 0 \rangle$ , where  $a/b := (a \cap b)'_{\{a\}}$ . In a sense, this representation result characterises implicative BCK-algebras entirely in semilattice-theoretic terms. This remark, in conjunction with Theorem 2.3.29, begs the question of whether implicative BCS-algebras may be characterised similarly in semigroup-theoretic terms, and in particular in terms of left normal bands with zero. We devote the remainder of this subsection to the study and solution of this problem. To begin, let A be an implicative BCS-algebra with left normal band with zero polynomial reduct  $\langle A; \cap, 0 \rangle$ . Recall from §1.3.11 that for any  $m \in A$ ,  $m \sqcap A := \{m \sqcap a : a \in A\}$ .

**Proposition 2.3.31.** Let A be an implicative BCS-algebra and let  $m \in A$  be fixed. Then  $m \sqcap A = \{a \in A : a \leq m\} = \{a \in A : m \sqcap a = a\}$ . Moreover,  $m \sqcap A$  is closed under  $\backslash$ , and on  $m \sqcap A$  the partial order  $\leq$  and the quasiorder  $\preceq$  coincide. Thus the principal subalgebra  $(m] := \langle m \sqcap A; \sqcap^{A|_m \sqcap A} \rangle$  generated by m is a Boolean lattice.

Proof. Let A be an implicative BCS-algebra and let  $m \in A$  be fixed. By left normality and Lemma 1.3.13(1),  $m \sqcap A = \{a : a \leq m\}$ . Since  $m \sqcap a = a$  iff  $a \leq m$ , the first statement of the proposition holds. For the second assertion, suppose  $a, b \in m \sqcap A$ . Then  $a \leq m$ , so  $ab \leq mb$  by Proposition 2.3.25(2). But  $mb \leq m$ , so  $ab \leq m$  by transitivity, whence  $m \sqcap A$  is closed under  $\setminus$ . Also;  $a \leq b$  implies  $a \leq b$  since  $\leq$  is admissible with respect to  $\leq$ . Conversely,  $a \leq b$  implies ab = 0, which implies  $a \sqcap b = a$  (by Lemma 2.1.42(1)), which implies  $b \sqcap a = a$  (since (m] is a subsemilattice of  $\langle A; \sqcap, 0 \rangle$  by Lemma 1.3.16). Thus  $a \leq b$ , and the partial order  $\leq$  and the quasiorder  $\leq$  coincide on  $m \sqcap$ A. Therefore for any  $a, b \in m \sqcap A$ , ab = 0 = ba implies a = b, so  $\langle m \sqcap$  $A; \setminus, 0$  is an implicative BCK-algebra such that the implicative BCK partial ordering coincides with the semilattice partial ordering on  $m \sqcap A$ . Since (m] is bounded (by Lemma 1.3.13(1)),  $\langle m \sqcap A; \setminus, 0 \rangle$  is bounded, so  $\langle m \sqcap A; \leq \rangle$  is a Boolean lattice by Corollary 1.6.22; that is to say (m] is a Boolean lattice. This establishes the final assertion of the proposition, and the proof is complete.

Recall from 1.3.15 that a band with zero **A** is locally Boolean if for every

ないので、「ない」のないで、「ない」のないで、「ない」のないで、

 $a \in A$ , the principal subalgebra (a] generated by a is a Boolean lattice. If A is a locally Boolean band and A is also left regular (equivalently, left normal), then by analogy with non-commutative lattice theory we call A a *left handed locally Boolean band*. (Right handed locally Boolean bands may be defined dually, though our concern here is with left handed locally Boolean bands.) As an immediate consequence of Corollary 2.3.22(1) and Proposition 2.3.31 we have the following result.

**Theorem 2.3.32.** Every implicative BCS-algebra  $\langle A; \backslash, 0 \rangle$  has a left handed locally Boolean band polynomial reduct  $\langle A; \sqcap, 0 \rangle$ , where  $a \sqcap b := a \backslash (a \backslash b)$  for any  $a, b \in A$ .

Let A be an implicative BCS-algebra. Call A bounded if there exists  $1 \in A$  such that  $a \leq 1$  for any  $a \in A$ . As usual, by abuse of language and notation we confuse a bounded implicative BCS-algebra A with its expansion to  $\langle A; \langle 0, 1 \rangle$ , where 1 is a new nullary operation symbol adjoined to the language of A whose canonical interpretation on  $\langle A; \langle 0, 1 \rangle$  is  $1 \in A$ . The following corollary to Theorem 2.3.32 may be inferred immediately from Corollary 1.6.22 and Proposition 2.3.31.

**Corollary 2.3.33.** (cf. [126, Theorem 12]; cf. [2, Theorem 8]) The underlying poset  $\langle A; \leq \rangle$  of a bounded implicative BCS-algebra  $\langle A; /, 0, 1 \rangle$  is a Boolean lattice. For any  $a, b \in A$ ,

 $a \wedge b = a \sqcap b$  $a \vee b = 1 \setminus ((1 \setminus a) \sqcap (1 \setminus b)).$ 

Let  $\langle A; \sqcap, 0 \rangle$  be a left handed locally Boolean band. Because  $a \sqcap b \leq_{\mathcal{H}} a$  for any  $a, b \in A$ , locally Boolean bands possess an induced difference operation  $\backslash$ . In more detail: given  $a, b \in A$ , the difference  $a \backslash b$  is defined to be  $(a \sqcap b)^*_{(a]}$ , namely the complement of  $a \sqcap b$  in the principal subalgebra (a) generated by a. In the following two results, we denote this induced difference by juxtaposition.

Lemma 2.3.34. For a left handed locally Boolean band A and  $a, b \in A$  the following assertions hold:

- 1.  $a \sqcap (ab) = ab = (ab) \sqcap a;$
- 2.  $c^*_{\{ab\}} = (ab) \sqcap c^*_{\{a\}}$  for any  $c \in A$  such that  $c \leq_{\mathcal{H}} ab$ .

*Proof.* For (1), just note  $ab = (a \sqcap b)^*_{(a]} \leq_{\mathcal{H}} a$ . For (2), assume  $c \leq_{\mathcal{H}} ab$ . Since  $(ab) \sqcap c^*_{(a]} \leq_{\mathcal{H}} ab$  by left normality,  $(ab) \sqcap c^*_{(a]} \in (ab]$ . It remains to show  $(ab) \sqcap c^*_{(a]}$  is the complement of c in (ab]. For this, observe:

$$((ab) \sqcap c^*_{(a]}) \sqcap c = (ab) \sqcap (c^*_{(a]} \sqcap c)$$
$$= 0$$

and also:

$$\begin{aligned} \left((ab) \sqcap c^*_{(a]}\right) \sqcup^{(ab]} c &= \left((ab) \sqcap c^*_{(a]}\right) \sqcup^{(a]} c & \text{as } (ab] \leq (a] \\ &= \left((ab) \sqcup^{(a]} c\right) \sqcap (c^*_{(a]} \sqcup^{(a]} c) & \text{by distributivity} \\ &= (ab) \sqcap (c^*_{(a]} \sqcup^{(a]} c) & \text{as } c \leq_{\mathcal{H}} ab, (ab] \leq (a] \\ &= (ab) \sqcap a \\ &= ab & \text{by } (1). \end{aligned}$$

Theorem 2.3.35. Every left handed locally Boolean band  $\langle A; \sqcap, 0 \rangle$  induces an implicative BCS-algebra  $\langle A; \backslash, 0 \rangle$  under the operation  $a \backslash b := (a \sqcap b)^*_{(a]}$  for any  $a, b \in A$ , where  $(a \sqcap b)^*_{(a]}$  denotes the complement of  $a \sqcap b$  in the principal subalgebra (a) generated by a.

*Proof.* Let A be a left handed locally Boolean band and let  $a, c \in A$ . To see the derived algebra  $\langle A; \rangle, 0 \rangle$  is an implicative BCS-algebra, we verify  $\langle A; \rangle, 0 \rangle$  satisfies the defining identities (2.47)-(2.50).

For (2.47), simply notice  $aa = (a \sqcap a)_{(a)}^* = a_{(a)}^* = 0$ .

For (2.48), observe  $(ab) \sqcap c \leq_{\mathcal{H}} ab$  by left normality, and hence that:

$$(ab)c = ((ab) \sqcap c)^*_{(ab]}$$
  
=  $(ab) \sqcap ((ab) \sqcap c)^*_{(a]}$  by Lemma 2.3.34(2)

2.3. Implicative BCS-Algebras

$$= (ab) \sqcap ((ab) \sqcap a \sqcap c)_{(a]}^{*}$$
by Lemma 2.3.34(1)  

$$= (ab) \sqcap ((ab)_{(a]}^{*} \sqcup^{(a)} (a \sqcap c)_{(a]}^{*})$$
by De Morgan's laws  

$$= (ab) \sqcap ((ab)_{(a]}^{*} \sqcup^{(a)} (ac))$$
  

$$= ((ab) \sqcap (ab)_{(a]}^{*}) \sqcup^{(a)} ((ab) \sqcap (ac))$$
by distributivity  

$$= 0 \sqcup^{(a)} ((ab) \sqcap (ac))$$
  

$$= (ab) \sqcap (ac).$$

By symmetry in the elements b and c we deduce also that  $(ac)b = (ac) \sqcap (ab)$ . But this implies (ab)c = (ac)b, since:

$$(ab) \sqcap (ac) = a \sqcap (ab) \sqcap (ac)$$
by Lemma 2.3.34(1) $= a \sqcap (ac) \sqcap (ab)$ by left normality $= (ac) \sqcap (ab)$ by Lemma 2.3.34(1).

For (2.49) it is sufficient to show (ab)c = (ab)(cb). By definition  $(ab)c = ((ab) \sqcap c)^*_{(ab]}$  and  $(ab)(cb) = ((ab) \sqcap (cb))^*_{(ab]}$ ; we claim  $(ab) \sqcap c \mathcal{D}(ab) \sqcap (cb)$ . Because A is locally Boolean,  $A/\mathcal{D}$  is semi-Boolean by Lemma 1.3.17. Since  $(x/y)/((x/y)/z) \approx (x/y)/((x/y)/(z/y))$  is an identity of implicative BCKalgebras (by (2.59) and (2.60)), from Theorem 1.6.21 we may infer  $[(ab) \sqcap c]_{\mathcal{D}} = [(ab) \sqcap (cb)]_{\mathcal{D}}$  in  $A/\mathcal{D}$ ; that is to say  $(ab) \sqcap c \mathcal{D}(ab) \sqcap (cb)$ . Now  $(ab) \sqcap c, (ab) \sqcap (cb) \leq_{\mathcal{H}} ab$  by left normality, so  $(ab) \sqcap c, (ab) \sqcap (cb) \in (ab]$ . But this implies (by Proposition 2.3.31) that the equivalence  $(ab) \sqcap c \mathcal{D}(ab) \sqcap (cb)$ collapses in (ab] to the equality  $(ab) \sqcap (cb) = (ab) \sqcap c$ . Because (ab] is Boolean, it is uniquely complemented, and so we deduce  $((ab) \sqcap c)^*_{(ab]} = ((ab) \sqcap (cb))^*_{(ab]}$ ; that is to say (ab)c = (ab)(cb).

For (2.50), observe first that  $a \sqcap (ba) = 0$ . Indeed,

$$0 = (b \sqcap a) \sqcap (b \sqcap a)^*_{[b]}$$
  
=  $b \sqcap a \sqcap (ba)$   
=  $b \sqcap (ba) \sqcap a$  by left normality  
=  $(ba) \sqcap a$  by Lemma 2.3.34(1)

which implies  $a \sqcap (ba) = 0$  by Lemma 1.4.13, just because  $a \sqcap (ba)\mathcal{D}(ba) \sqcap a$ . To complete the proof, simply notice:

$$a(ba) = (a \sqcap (ba))^*_{(a]}$$
$$= 0^*_{(a]}$$
$$= a.$$

**Theorem 2.3.36.** In the variety of implicative BCS-algebras and the class of left handed locally Boolean bands, the following assertions hold:

- 1. Every implicative BCS-algebra  $\langle A; \backslash, 0 \rangle$  induces a left handed locally Boolean band  $\langle A; \sqcap, 0 \rangle$  upon defining  $a \sqcap b := a \backslash (a \backslash b)$  for any  $a, b \in A$ ;
- Every left handed locally Boolean band (A; □, 0) determines an implicative BCS-algebra (A; \, 0) under the operation a\b := (a □ b)<sup>\*</sup><sub>(a]</sub> for any a, b ∈ A, where (a □ b)<sup>\*</sup><sub>(a]</sub> denotes the complement of a □ b in the principal subalgebra (a] generated by a.

Moreover, the correspondences of (1) and (2) are inverse to each other. In more detail, if  $\langle A; \rangle, 0 \rangle$  is an implicative BCS-algebra with left handed locally Boolean band polynomial reduct  $\langle A; \sqcap, 0 \rangle$ , then implicative BCS difference on  $\langle A; \sqcap, 0 \rangle$  as induced by the operation  $\backslash^{\langle A; \sqcap, 0 \rangle}$  of (2) coincides with implicative BCS difference  $\backslash^{\langle A; \rangle, 0 \rangle}$  on  $\langle A; \rangle, 0 \rangle$ ; that is,  $a \backslash^{\langle A; \sqcap, 0 \rangle} b = a \backslash^{\langle A; \rangle, 0 \rangle} b$  for any  $a, b \in A$ . Conversely, if  $\langle A; \sqcap, 0 \rangle$  is a left handed locally Boolean band with induced implicative BCS-algebra  $\langle A; \rangle, 0 \rangle$ , then the implicative BCS meet on  $\langle A; \rangle, 0 \rangle$  as determined by the operation  $\sqcap^{\langle A; \rangle, 0 \rangle}$  of (1) coincides with the band operation  $\sqcap^{\langle A; \sqcap, 0 \rangle}$  on  $\langle A; \sqcap, 0 \rangle$ ; that is,  $a \sqcap^{\langle A; \backslash, 0 \rangle} b = a \sqcap^{\langle A; \sqcap, 0 \rangle} b$  for any  $a, b \in A$ .

*Proof.* It remains only to prove the final assertion. Let  $\langle A; \rangle, 0 \rangle$  be an implicative BCS-algebra with left handed locally Boolean band polynomial reduct  $\langle A; \Box, 0 \rangle$ . For any  $a, b, \in A$ ,

 $a \setminus {}^{\langle A; \, \Pi, 0 \rangle} b = (a \sqcap b)^*_{(a)}$ 

$$= (a \setminus \langle A; \setminus, 0 \rangle (a \setminus \langle A; \setminus, 0 \rangle b))_{(a]}^{*}$$
  
=  $a \setminus \langle A; \setminus, 0 \rangle (a \setminus \langle A; \setminus, 0 \rangle (a \setminus \langle A; \setminus, 0 \rangle b))$  by Corollary 2.3.33  
=  $a \setminus \langle A; \setminus, 0 \rangle b$  by (2.56).

Conversely, let  $\langle A; \Box, 0 \rangle$  be a left handed locally Boolean band with induced implicative BCS-algebra  $\langle A; \backslash, 0 \rangle$ . For any  $a, b \in A$ ,

$$a \sqcap^{\langle A; \backslash, 0 \rangle} b = a \backslash (a \backslash b)$$
  

$$= (a \sqcap^{\langle A; \sqcap, 0 \rangle} (a \sqcap^{\langle A; \sqcap, 0 \rangle} b)^{*}_{\langle a]})^{*}_{\langle a]}$$
  

$$= a^{*}_{\langle a]} \sqcup^{\langle a|} (a \sqcap^{\langle A; \sqcap, 0 \rangle} b)^{**}_{\langle a|}$$
  

$$= (a \sqcap^{\langle A; \sqcap, 0 \rangle} b)^{**}_{\langle a|}$$
  

$$= a \sqcap^{\langle A; \sqcap, 0 \rangle} b.$$
  
by De Morgan's laws  
by De M

Remark 2.3.37. Let  $A := \langle A; \cdot, 0 \rangle$  be a locally Boolean band. For any  $a, b \in A$ ,  $aba \leq_{\mathcal{H}} a$ , and so A possesses an induced difference operation  $\setminus^{\langle A; \cdot, 0 \rangle}$ , where  $a \setminus^{\langle A; \cdot, 0 \rangle} b := (aba)^*_{\{a|_A}$ , the complement of aba in the principal subalgebra  $\langle a]_A$  generated by a. As with left handed locally Boolean bands, the induced algebra  $\langle A; \setminus^{\langle A; \cdot, 0 \rangle}, 0 \rangle$  is an implicative BCS-algebra. (For a justification of this assertion, see the proof below.) In spite of this observation, however, there seems little profit in studying locally Boolean bands as a generalisation of left handed locally Boolean bands, because the correspondences of Theorem 2.3.36 are not preserved: while every (not necessarily left or right handed) locally Boolean band polynomial reduct of any implicative BCS-algebra is always left (or right) handed.

*Proof.* Let  $\mathbf{A} := \langle A; \cdot, 0 \rangle$  be a locally Boolean band. For any  $a, b \in A$ , let  $a \setminus {}^{\langle A; \cdot, 0 \rangle} b := (aba)^*_{\langle a]_{\mathbf{A}}}$ , the complement of aba in the principal subalgebra  $\{a]_{\mathbf{A}}$  generated by a. To see the induced algebra  $\langle A; \setminus {}^{\langle A; \cdot, 0 \rangle}, 0 \rangle$  is an implicative BCS-algebra, let  $a \sqcap b := aba$  for any  $a, b \in A$ . We claim:

- (i) The polynomial reduct  $\langle A; \sqcap, 0 \rangle$  is a left normal band with zero;
- (ii) For any  $a \in A$ ,  $(a]_{(A; \cdot, 0)} = (a]_{(A; (1, 0))}$ ;
- (iii) If  $b, c \in (a]_{\mathbf{A}}$  then  $b \cdot c = b \sqcap c$ .

For (i), easy but tedious computations show the polynomial reduct  $\langle A; \Box, 0 \rangle$  is a left normal band with zero, just because  $\langle A; \cdot, 0 \rangle$  is normal (by Lemma 1.3.16).

For (ii), it sufficient to show  $a \leq \langle A; \cdot, 0 \rangle b$  iff  $a \leq \langle A; \sqcap, 0 \rangle b$  for any  $a, b \in A$ . So let  $a, b \in A$ . Clearly  $a \leq \langle A; \cdot, 0 \rangle b$  implies  $a \leq \langle A; \sqcap, 0 \rangle b$ . Conversely,  $a \leq \langle A; \sqcap, 0 \rangle b$  implies aba = a = bab, so ab = (bab)b = bab = a and ba = b(bab) = bab = a, whence  $a \leq \langle A; \cdot, 0 \rangle b$ .

For (iii), let  $b, c \in (a]_A$ . Since  $(a]_A$  is a subsemilattice of  $\langle A; \cdot, 0 \rangle$  we have  $b \cdot c = b \cdot c \cdot b = b \sqcap c$  as desired.

To complete the proof, notice (ii)-(iii) imply the principal subalgebra  $(a]_{\langle A; \sqcap, 0 \rangle}$ of  $\langle A; \sqcap, 0 \rangle$  coincides with the principal subalgebra  $(a]_A$  of A for any  $a \in A$ , whence  $\langle A; \sqcap, 0 \rangle$  is a left handed locally Boolean band by (i). By Theorem 2.3.35 we have that the induced algebra  $\langle A; \setminus^{\langle A; \sqcap, 0 \rangle}, 0 \rangle$  is an implicative BCS-algebra, where for any  $a, b \in A$ ,  $a \setminus^{\langle A; \sqcap, 0 \rangle} b := (a \sqcap b)^*_{(a]_{\langle A; \sqcap, 0 \rangle}}$ . But  $(a \sqcap b)^*_{\{a]_{\langle A; \sqcap, 0 \rangle}} = (aba)^*_{\{a]_A}$  for any  $a, b \in A$ , so  $a \setminus^{\langle A; \cdot, 0 \rangle} b = a \setminus^{\langle A; \sqcap, 0 \rangle} b$ . Thus  $\langle A; \setminus^{\langle A; \cdot, 0 \rangle}, 0 \rangle$  is an implicative BCS-algebra as asserted.

We cannot hope for a further sharpening (in purely algebraic terms) of the relationship between the class of implicative BCS-algebras and the class of left handed locally Boolean bands on at least two counts. On the one hand, the class of all left handed locally Boolean bands is not even a quasivariety, since it is not closed under the formation of subalgebras. (To see this, just consider the 4-element Boolean lattice 4 (with universe  $\{0, a, b, 1\}$ , least element 0 and greatest element 1) as a semi-Boolean algebra. Clearly  $\langle \{0, a, 1\}; \sqcap, 0 \rangle$  is a subalgebra of 4 such that 0 < a < 1.) On the other hand, the natural morphisms between left handed locally Boolean bands (namely those left normal band with zero homomorphisms that preserve Boolean sublattices) cannot be given a purely algebraic description. This is the subject of Theorem 2.3.39 below. But first, the following lemma, which is an easy modification of a result due to Cornish [64].

Lemma 2.3.38. (cf. [64, Lemma 3.3]) Let A be a left handed locally Boolean band and let  $a \in A$  be fixed. Let  $\theta$  be a congruence on the underlying left normal band with zero polynomial reduct  $\langle A; \sqcap, 0 \rangle$  such that for each  $b \in A$ , the restriction  $\theta|_{\{b\}}$  of  $\theta$  to the principal subalgebra (b) generated by b is a lattice congruence on (b]. Then  $A/\theta$  is a left handed locally Boolean band and the restriction of the canonical map  $\nu : A \to A/\theta$  is a lattice homomorphism of  $(a]_A$  onto  $(\nu(a)]_{A/\theta}$ .

Proof. Let r denote the restriction of  $\nu$ . Notice that for  $\nu(b) \in A/\theta$ ,  $\nu(b) \in (\nu(a)]_{A/\theta}$  implies  $\nu(b) \leq (\nu(a)]_{A/\theta} \nu(a)$ , which implies  $\nu(b) = \nu(a) \sqcap^{(\nu(a)]_{A/\theta}} \nu(b) = \nu(c)$  for some  $c \in A$ , whereby  $c = a \sqcap^A b \in (a]_A$ . Hence r is onto, and we can regard each element of  $(\nu(a)]_{A/\theta}$  to be of the form  $\nu(c)$  for a suitable  $c \leq^A a$ .

Let  $b, c \in (a]$  and let  $d := b \sqcup^{(a)} c$ . Then  $r(b), r(c) \leq^{(\nu\{a\})_{A/\theta}} r(d)$ . Suppose  $r(b), r(c) \leq^{(\nu(a)]_{A/\theta}} r(e)$  also for some  $e \in (a]$ . Then  $b \equiv b \sqcap e \pmod{\theta}$  and  $c \equiv c \sqcap e \pmod{\theta}$ . Therefore  $d = b \sqcup^{(a]} c \equiv (b \sqcap e) \sqcup^{(a]} (c \sqcap e) \pmod{\theta}$ . In other words,  $r(d) = r((b \sqcap e) \sqcup^{(a)} (c \sqcap e)) \leq^{(\nu(a)]_{A/\theta}} r(e)$ . Therefore  $r(d) = r(b) \sqcup^{(\nu(a)]_{A/\theta}} r(c)$ . Thus  $(\nu(a)]_{A/\theta}$  is a lattice and the quotient  $A/\theta$  is a left handed locally Boolean band.

**Theorem 2.3.39.** Let A be an implicative BCS-algebra with left handed locally Boolean band polynomial reduct  $\langle A; \sqcap, 0 \rangle$ . The following are equivalent:

- 1.  $\theta \in \operatorname{Con} \mathbf{A}$ ;
- 2.  $\theta \in \operatorname{Con} \langle A; \sqcap, 0 \rangle$  and  $\theta|_{(a]} \in \operatorname{Con} (a]$  for each  $a \in A$ , where  $\theta|_{(a]}$  denotes the restriction of  $\theta$  to the principal subalgebra (a] generated by a;

3.  $\theta \in \operatorname{Con} \langle A; \Box, 0 \rangle$  and  $\theta|_{(a)}$  is a lattice congruence on (a) for each  $a \in A$ .

*Proof.* (1)  $\Rightarrow$  (2). Clearly  $\theta$  is a congruence on the polynomial reduct  $\langle A; \sqcap, 0 \rangle$ . Moreover, because (a) is a subalgebra of A for each  $a \in A$ , the restriction  $\theta|_{(a)}$  must be a congruence on (a).

 $(2) \Rightarrow (3)$ . Let  $a \in A$  be fixed. As the infimum  $b \sqcap c$  is a derived operation for any  $b, c \in (a]$ , we deduce that  $\theta$  is a semilattice congruence. Moreover, since

(a) is a bounded BCK-subalgebra of A, for any  $b, c \in A$  the lattice supremum  $b \sqcup c$  is also a derived operation, given by the formula of Corollary 2.3.33. Thus the restriction of  $\theta$  to (a) is a lattice congruence.

(3)  $\Rightarrow$  (1). Suppose that for each  $a \in A$ ,  $\theta|_{(a]}$  is a lattice congruence on (a]. Then  $A/\theta$  is left handed locally Boolean by Lemma 2.3.38. Let  $\nu : A \to A/\theta$  be the canonical homomorphism and let  $a, a_1, b, b_1 \in A$  be such that  $a \equiv a_1 \pmod{\theta|_{(a]}}$  and  $b \equiv b_1 \pmod{\theta|_{(a]}}$ . Now in  $A, a \setminus b$  is the complement of  $a \sqcap b$  in (a]. Since  $A/\theta$  is left handed locally Boolean,  $\nu(a \setminus b)$  is the complement of  $\nu(a) \sqcap \nu(b)$  in the Boolean lattice  $(\nu(a)]_{A/\theta}$ . As  $\nu(a) = \nu(a_1)$  and  $\nu(b) = \nu(b_1)$ ,  $\nu(a_1 \setminus b_1)$  is also a complement of  $\nu(a) \sqcap \nu(b)$  in  $(\nu(a)]_{A/\theta}$ . Because complements are unique in  $(\nu(a)]_{A/\theta}$ , we conclude  $\nu(a \setminus b) = \nu(a_1 \setminus b_1)$ , which implies  $\theta$  is a congruence on A as required.

Although the correspondence between implicative BCS-algebras and left handed locally Boolean bands cannot be given a purely algebraic description, Theorem 2.3.39 immediately suggests a category-theoretic formalisation. Let **IBCS** denote the category for which:

- The objects of **IBCS** are the implicative BCS-algebras;
- The morphisms of IBCS are the implicative BCS homomorphisms.

Also, let LLBB denote the category for which:

- The objects of LLBB are the left handed locally Boolean bands;
- The morphisms of LLBB are the Boolean sublattice preserving homomorphisms, namely those homomorphisms h : A → B, where A and B are left handed locally Boolean bands, such that for each a ∈ A, the restriction (ker h)|<sub>(a]</sub> of the relation kernel ker h to the principal subalgebra (a) generated by a is a lattice congruence on (a).

Recall from category theory [160, Section 3.6] that categories C and D are isomorphic iff there exists a one-to-one functor F mapping C onto D. Since a functor  $F: C \to D$  is an isomorphism iff there exists a functor  $G: D \to C$ such that  $F \circ G$  is the identity map on D and  $G \circ F$  is the identity map on C (where C [D] is the underlying class of objects in the category C [D]) [160, Section 3.6], the following result is clear upon combining Theorem 2.3.36 and Theorem 2.3.39.

Theorem 2.3.40. IBCS and LLBB are isomorphic as categories.

In light of the preceding result, it is natural to anticipate that structural results concerning left normal bands with zero transfer to the setting of implicative BCS-algebras. Certainly this is the case in relation to the Clifford-McLean theorem for bands, for which we have the following natural analogue.

Theorem 2.3.41 (Clifford-McLean Theorem for Implicative BCS-Algebras). Let A be an implicative BCS algebra with left handed locally Boolean band polynomial reduct  $\langle A; \sqcap, 0 \rangle$ . Then  $\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$ -equivalence is a congruence relation on both  $\langle A; \sqcap, 0 \rangle$  and A. The quotient algebra  $A/\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$  is the maximal implicative BCK-algebra homomorphic image of A and the quotient algebra  $\langle A; \sqcap, 0 \rangle/\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$  is the maximal semi-Boolean algebra homomorphic image of  $\langle A; \sqcap, 0 \rangle$ , whilst the  $\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$ -congruence classes of both A and  $\langle A; \sqcap, 0 \rangle$  are the maximal left zero semigroups of  $\langle A; \sqcap, 0 \rangle$ . For all  $a, b \in A$ , the following are equivalent:

- 1.  $a \Xi_{\mathbf{A}} b$ ;
- 2.  $a\mathcal{D}_{\langle A; \sqcap, 0 \rangle}b;$
- 3.  $a\mathcal{L}_{\langle A; \, \Box, 0 \rangle} \dot{b}$ .

Proof. Let A be an implicative BCS-algebra with left handed locally Boolean band polynomial reduct  $\langle A; \sqcap, 0 \rangle$ . Observe first that  $\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$  and  $\mathcal{L}_{\langle A; \sqcap, 0 \rangle}$  coincide by left normality and the remarks of §1.3.15, and also that  $\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$  and  $\Xi_{\mathbf{A}}$ coincide by Lemma 2.3.24 (since they are respectively the equivalences induced by the quasiorderings  $\leq_{\mathcal{D}}^{\langle A; \sqcap, 0 \rangle}$  and  $\preceq$  in the sense of Lemma 1.2.2). Thus Items (1)-(3) are equivalent, and  $\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$ -equivalence is a congruence relation on both  $\langle A; \sqcap, 0 \rangle$  and A. By Theorem 2.1.14 the quotient algebra  $\mathbf{A}/\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$ is the maximal implicative BCK-algebra homomorphic image of A, while by Lemma 1.3.17  $\langle A; \sqcap, 0 \rangle/\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$  is the maximal semi-Boolean algebra homomorphic image of  $\langle A; \sqcap, 0 \rangle$ . Finally, left normality and the Clifford-McLean theorem for bands ensure the maximal left zero semigroups of  $\langle A; \sqcap, 0 \rangle$  are the  $\mathcal{D}_{\langle A; \sqcap, 0 \rangle}$ -congruence classes of  $\langle A; \sqcap, 0 \rangle$ , and hence (by previous remarks) of **A**.

Theorem 2.3.41 justifies dubbing the congruence  $\Xi$  on any implicative BCSalgebra the 'Clifford-McLean congruence'. We will (sometimes) employ this vocabulary in the sequel.

2.3.42. Subdirectly Irreducible Implicative BCS-Algebras. The complete description (to within isomorphism) of both the subdirectly irreducible left normal bands with zero and the subdirectly irreducible implicative BCKalgebras (recall Corollary 1.3.19 and Theorem 1.6.19 respectively), in conjunction with the affinity of implicative BCS-algebras to both implicative BCKalgebras and left normal bands with zero, suggests that the possibility of completely characterising the subdirectly irreducible implicative BCS-algebras (to within isomorphism) may be strong. This subsection is devoted to the study and solution of this problem. Our first order of business is to isolate a family of subdirectly irreducible implicative BCS-algebras. This is the subject of the following lemma, which is due to the author's Ph.D. supervisor.

Lemma 2.3.43. (Bignall) Let  $\mathbf{B} := \langle B; \land, \lor, ', 0, 1 \rangle$  be a non-trivial Boolean algebra with bounded implicative BCK-algebra polynomial reduct  $\langle B; /, 0, 1 \rangle$ , where  $a/b = a \land b'$  for any  $a, b \in B$ . Let  $B' := B - \{1\}$  and let  $\{m_1, m_2\}$  be disjoint from B. Let  $\hat{B} := B' \cup \{m_1, m_2\}$  and let the difference  $\backslash$  be defined on  $\hat{B}$  as follows:

 $a \setminus b := \begin{cases} a/b & \text{if } a, b \in B' \\ 0 & \text{if } b \in \{m_1, m_2\} \text{ and } a \in \hat{B} \\ a & \text{if } a \in \{m_1, m_2\} \text{ and } b = 0 \\ b' & \text{if } a \in \{m_1, m_2\} \text{ and } 0 \neq b \in B'. \end{cases}$ 

Then the derived algebra  $\hat{\mathbf{B}} := \langle \hat{B}; \langle 0 \rangle$  is an implicative BCS-algebra.

*Proof.* Let  $\mathbf{B}$  be a non-trivial Boolean algebra with greatest element 1. Let  $\mathbf{B}$  be the subdirectly irreducible pseudocomplemented semilattice with greatest

element  $m \in \mathring{B}$  induced by  $\mathbb{B}$  as per Theorem 1.3.7. Let  $\overline{\mathbb{B}}$  denote the canonical implicative BCS-algebra polynomial reduct of  $\check{\mathbb{B}}$  (arising as per Example 2.3.12). It is trivial to check that the operation  $\backslash^{\overline{\mathbb{B}}}$  defined on  $\check{B}$  is the same as the operation  $\backslash^{\hat{\mathbb{B}}}$  defined on  $\hat{B}$  when the element 1 is renamed as  $m_1$ and the element m is renamed as  $m_2$ . Hence  $\hat{\mathbb{B}}$  is an implicative BCS-algebra.

**Lemma 2.3.44.** Let  $\mathbf{B}$  be a non-trivial Boolean algebra. Then the implicative BCS-algebra  $\hat{\mathbf{B}}$  induced from  $\mathbf{B}$  as per Lemma 2.3.43 is subdirectly irreducible with monolith  $\Xi$ .

*Proof.* Observe first that  $\omega_{\hat{\mathbf{B}}} \cup \{ \langle m_1, m_2 \rangle, \langle m_2, m_1 \rangle \} = \Xi \in \operatorname{Con} \hat{\mathbf{B}}$ . Therefore to see  $\hat{\mathbf{B}}$  is subdirectly irreducible with monolith  $\Xi$  it is sufficient to show  $m_1 \equiv m_2 \pmod{\theta}$  for any  $\theta \in \operatorname{Con} \hat{\mathbf{B}}$  where  $\theta \neq \omega_{\hat{\mathbf{B}}}$ . So let  $\theta \in \operatorname{Con} \hat{\mathbf{B}}$  be such that  $\theta \neq \omega_{\hat{\mathbf{B}}}$ . There are two cases to consider:

- (i)  $[0]_{\theta} = \{0\};$
- (ii)  $[0]_{\theta} \neq \{0\}.$

For Case (i), suppose  $[0]_{\theta} = \{0\}$ . Then  $\theta \subseteq \Xi = \omega_{\hat{\mathbf{B}}} \cup \{\langle m_1, m_2 \rangle, \langle m_2, m_1 \rangle\}$ by Theorem 2.1.14(4). Assume to the contrary that  $m_1 \not\equiv_{\theta} m_2$ . Then  $\theta = \omega_{\hat{\mathbf{B}}}$ which is a contradiction. Thus  $m_1 \equiv_{\theta} m_2$ .

For Case (ii), suppose  $[0]_{\theta} \neq \{0\}$ . Then there exists  $0 \neq a \in \hat{B}$  such that  $0 \equiv a \pmod{\theta}$ . If  $a \notin \{m_1, m_2\}$  then  $m_1 = m_1 0 \equiv_{\theta} m_1 a = m_2 a \equiv_{\theta} m_2 0 = m_2$  by definition of  $\setminus^{\hat{B}}$ . So suppose  $a \in \{m_1, m_2\}$ . Let  $b \in \{m_1, m_2\}$  be such that  $b \neq a$ . Then  $b = b0 \equiv_{\theta} ba = 0 \equiv_{\theta} a$  by definition of  $\setminus^{\hat{B}}$  as required.

Implicative BCS-algebras arising as per Lemma 2.3.43 may be most easily envisaged as Boolean algebras in which the unit element has been replaced by a 2-element clique. To see this more clearly, let  $\hat{\mathbf{B}}_0 := \mathbf{C}_1$ , where  $\mathbf{C}_1$  is the 2element flat implicative BCK-algebra (recall Example 1.6.18). For  $1 \le n < \omega$ , let  $\hat{\mathbf{B}}_n$  denote the implicative BCS-algebra induced as per Lemma 2.3.43 from the non-trivial finite Boolean algebra  $\mathbf{B}$  of cardinality  $2^n$ . For n = 0, 1, 2, 3, the algebras  $\hat{\mathbf{B}}_n$  are depicted in Figure 2.3. 2.3. Implicative BCS-Algebras



Figure 2.3. The subdirectly irreducible implicative BCS-algebras  $\hat{B}_n$  for n = 0, 1, 2, 3.

We devote the remainder of this subsection to proving that the family of algebras  $\{\hat{\mathbf{B}}_0\} \cup \{\hat{\mathbf{B}} : \mathbf{B} \text{ a non-trivial Boolean algebra}\}$  comprises, to within isomorphism, all subdirectly irreducible members of the variety of implicative BCS-algebras. With this aim in mind we identify and briefly study some standard congruences of implicative BCS-algebras in the following two lemmas.

Lemma 2.3.45. For any implicative BCS-algebra A the following assertions hold:

1. For a fixed  $c \in A$ , the maps:

- (a)  $a \mapsto ac;$
- (b)  $a \mapsto a \sqcap c$ ;
- (c)  $a \mapsto c \sqcap a$

are endomorphisms of A. Respectively, the associated congruences are defined by:

- (a)'  $a \equiv b \pmod{\vartheta_c}$  iff ac = bc;
- $(b)' \ a \equiv b \pmod{\varrho_c} \text{ iff } a \sqcap c = b \sqcap c;$
- (c)'  $a \equiv b \pmod{\varsigma_c}$  iff  $c \sqcap a = c \sqcap b$ .
- 2. For a fixed  $c \in A$ ,

191

(a) 
$$\vartheta_c = \langle c \rangle_{\mathbf{A}}^{\delta} = \Theta^{\mathbf{A}}(0, c);$$
  
(b)  $\Xi \leq \varsigma_c;$   
(c)  $\bigcap \{\varsigma_c\} = \Xi.$ 

Proof. Let A be an implicative BCS-algebra. Let c be a fixed element of A and let  $a, b \in A$ . For (1) we have that (ab)c = (ac)(bc) by (2.49), whence the map  $a \mapsto ac$  is an endomorphism. Also  $(ab) \sqcap c = (ab)(ac) = (a \sqcap c)(b \sqcap c)$ by (2.59) and (2.62), so the map  $a \mapsto a \sqcap c$  is also an endomorphism. And,  $c \sqcap (ab) = (cb)(ca) = (c \sqcap a)(c \sqcap b)$  by (2.61) and (2.57), which implies the map  $a \mapsto c \sqcap a$  is an endomorphism as well. The remaining assertions follow immediately from [55, Theorem II§6.8].

For (2)(a) let  $\vartheta_c$  be as stated. Since  $\langle c \rangle_{\mathbf{A}}^{\delta} = ([0]_{\Theta^{\mathbf{A}}(0,c)})^{\delta} = \Theta^{\mathbf{A}}(0,c)$ , it only remains to show  $\vartheta_c = \Theta^{\mathbf{A}}(0,c)$ . From cc = 0 = 0c we have  $0 \equiv c \pmod{\vartheta_c}$ whence  $\Theta^{\mathbf{A}}(0,c) \subseteq \vartheta_c$ . For the opposite inclusion just notice  $a \equiv b \pmod{\vartheta_c}$ implies  $a = a0 \equiv_{\Theta^{\mathbf{A}}(0,c)} ac = bc \equiv_{\Theta^{\mathbf{A}}(0,c)} b0 = b$ . Thus  $\vartheta_c \subseteq \Theta^{\mathbf{A}}(0,c)$  and so  $\vartheta_c = \Theta^{\mathbf{A}}(0,c)$ .

For (2)(b) let  $a \equiv b \pmod{\Xi}$ . Then  $a \sqcap b = a$  and  $b \sqcap a = b$  by Lemma 2.1.42(1), whence  $c \sqcap a = c \sqcap a \sqcap b = c \sqcap b \sqcap a = c \sqcap b$  by left normality. Thus  $a \equiv b \pmod{\varsigma_c}$ and  $\Xi \leq \varsigma_c$ .

For (2)(c) suppose  $a \equiv b \pmod{\bigcap_{c \in A} \{\varsigma_c\}}$ . Then  $c \sqcap a = c \sqcap b$  for any  $c \in A$ , and in particular  $a \sqcap a = a \sqcap b$  and  $b \sqcap a = b \sqcap b$ . Thus  $a = a \sqcap b$  and  $b \sqcap a = b$ , so  $a \preceq b$  and  $b \preceq a$  by Lemma 2.1.42(1); that is to say  $a \equiv b \pmod{\Xi}$ . Conversely,  $\Xi \leq \varsigma_c$  for any  $c \in A$  by (2/(b), so  $\Xi \leq \bigcap_{c \in A} \{\varsigma_c\}$ . Thus  $\bigcap_{c \in A} \{\varsigma_c\} = \Xi$ .

**Lemma 2.3.46.** Let A be an implicative BCS-algebra and let  $m \in A$  be fixed. The following assertions hold:

- ω<sub>A</sub> = ℘<sub>m</sub> iff a ≤ m for any a ∈ A, where ℘<sub>m</sub> denotes the congruence of Lemma 2.3.45(1)(b)';
- 2.  $\Xi = \varsigma_m$  iff  $a \preceq m$  for any  $a \in A$ , where  $\varsigma_m$  denotes the congruence of Lemma 2.3.45(1)(c)';

Proof. Let A be an implicative BCS-algebra and let  $m \in A$  be fixed. For (1), suppose  $\omega_A = \varrho_m$ . Notice  $a \sqcap m = (a \sqcap m) \sqcap m$  for any  $a \in A$  and hence that  $a \equiv a \sqcap m \pmod{\varrho_m}$ . Now  $\varrho_m \leq \Xi$  since  $\varrho_m$  is the identity congruence, so  $a \equiv a \sqcap m \pmod{\varrho_m}$ . Thus  $a \preceq a \sqcap m$ . But (a(am))m = 0 by (2.2) and so  $a \sqcap m \preceq m$ . By transitivity we conclude  $a \preceq m$  as desired. Conversely, suppose  $c \preceq m$  for any  $c \in A$  and note this implies  $c \sqcap m = c$ . Suppose  $a \equiv b \pmod{\varrho_m}$  for  $a, b \in A$ . Then  $a \sqcap m = b \sqcap m$ , so  $a = a \sqcap m = b \sqcap m = b$ . Thus  $a \equiv b \pmod{\omega_A}$  and hence  $\varrho_m = \omega_A$ .

For (2), suppose  $\Xi = \varsigma_m$ . By left normality  $m \sqcap a = m \sqcap (a \sqcap m)$  for any  $a \in A$ , so  $a \equiv a \sqcap m \pmod{\varsigma_m}$ , which implies  $a \equiv a \sqcap m \pmod{\Xi}$ . Thus  $a \preceq a \sqcap m$ . But (a(am))m = 0 by (2.2) and so  $a \sqcap m \preceq m$ . By transitivity we conclude  $a \preceq m$  as desired. Conversely, suppose  $c \preceq m$  for any  $c \in A$  and note this implies  $c \sqcap m = c$ . Suppose  $a \equiv b \pmod{\varsigma_m}$  for  $a, b \in A$ . Then  $m \sqcap a = m \sqcap b$ , whence  $a = a \sqcap m = a \sqcap m \sqcap a = a \sqcap m \sqcap b = a \sqcap b \sqcap m = a \sqcap b$  and  $b = b \sqcap m = b \sqcap m \sqcap b = b \sqcap m \sqcap a = b \sqcap a \sqcap m = b \sqcap a$  by left normality. By Lemma 2.1.42(1) we conclude  $a \preceq b$  and  $b \preceq a$ . Thus  $a \equiv b \pmod{\Xi}$  and  $\varsigma_m \subseteq \Xi$ . The opposite inclusion follows immediately from Lemma 2.3.45(2)(b).

Recall that an element  $m \in A$  of a pre-BCK-algebra A is maximal if  $a \leq m$  for all  $a \in A$ . Recall also that all maximal elements (where they exist) of a pre-BCK-algebra A lie in a unique  $\Xi$ -class (the maximal class).

Lemma 2.3.47. Let A be a subdirectly irreducible implicative BCS-algebra and let  $\{m_1, m_2\}$  be the pair of elements identified under every non-trivial congruence relation of A. Then either  $A \cong \hat{B}_0$  or A has a maximal class M such that  $\{m_1, m_2\} \subseteq M$ .

*Proof.* Let A be a subdirectly irreducible implicative BCS-algebra. We separate t! proof into two cases:

- (i)  $\mathbb{A} \in \operatorname{HCK}_{\mathbb{C}}$
- (ii)  $\mathbf{A} \notin \mathbf{i} \mathcal{P} \mathcal{C} \mathbf{K}$ .

For Case (i), suppose  $A \in iBCK$ . From Theorem 1.6.19 we deduce A is isomorphic to the 2-element implicative BCK-algebra  $C_1$ , which implies by definition

## that **A** is isomorphic to $\mathbf{B}_0$ .

For Case (ii), suppose  $A \notin iBCK$ . Let  $\{m_1, m_2\}$  be the pair of elements identified under every non-trivial congruence. Because  $A \notin iBCK$  there exists a subalgebra of A isomorphic to  $B_2$  by Proposition 2.2.5, so the Clifford-McLean congruence  $\Xi$  on A is non-trivial. Thus  $m_1 \equiv m_2 \pmod{\Xi}$  and in particular  $m_2 \preceq m_1$ , whence  $m_2 \sqcap m_1 = m_2$  by Lemma 2.1.42.1). Let now  $\varrho_{m_1}$  be the congruence relation of Lemma 2.3.45(1)(b)' defined by  $a \equiv b \pmod{\varrho_{m_1}}$  iff  $a \sqcap m_1 = b \sqcap m_1$  for any  $a, b \in A$ . Because A is subdirectly irreducible  $\varrho_{m_1}$ is the identity congruence. For suppose to the contrary that  $\varrho_{m_1}$  is not the identity congruence. Then  $m_1 \equiv m_2 \pmod{\varrho_{m_1}}$ , so  $m_1 \sqcap m_1 = m_2 \sqcap m_1$ ; that is to say  $m_1 = m_2 \sqcap m_1$ . But  $m_2 \sqcap m_1 = m_2$ . Thus  $m_1 = m_2$ , which contradicts the subdirect irreducibility of A. Since  $\varrho_{m_1}$  is the identity congruence, we deduce from Lemma 2.3.46(1) that  $m_1$  is maximal. Since  $m_2$  lies in the same  $\Xi$ -class as  $m_1$  we have also that  $m_2$  is naximal. We have shown A has a maximal class M such that  $\{m_1, m_2\} \subseteq M$  and the proof is complete.

Lemma 2.3.47 demands attention be focussed on congruences of implicative BCS-algebras with a maximal class, and in particular on the role played by the Clifford-McLean congruence in such algebras.

Lemma 2.3.48. Let A be an implicative BCS-algebra with maximal class M. For a fixed  $m \in M$ ,  $m \equiv ma \pmod{\Xi}$  iff a = 0 for any  $a \in A$ . Therefore if  $a \neq 0$  then  $ma \notin M$ .

Proof. Let A be an implicative BCS-algebra with maximal class M. Let  $m \in M$  be fixed and let  $a \in A$ . If a = 0 then ma = m0 = m, so  $ma \equiv m \pmod{\Xi}$ . Conversely, if  $ma \equiv m \pmod{\Xi}$  then  $m \preceq ma$ , so m(ma) = 0. But then a = a0 = a(m(ma)) = am = 0 by (2.53) and the maximality of m. Thus  $m \equiv ma \pmod{\Xi}$  iff a = 0. Suppose now that  $a \neq 0$  and assume to the contrary that  $ma \in M$ . Then  $ma \equiv m \pmod{\Xi}$  and so a = 0, a contradiction.

The remaining results of this subsection, including the following lemma, are due jointly to the author and the author's Ph.D. supervisor.

Lemma 2.3.49. Let A be an implicative BCS-algebra with maximal class M. Let B be the subalgebra of A with universe B := A - M and let  $\Xi_{\mathbf{B}}$  denote the Clifford-McLean congruence on B. Then the relation:

$$\lfloor \Xi \rfloor := \Xi_{\mathbf{i}\mathbf{j}} \cup \big\{ \langle m, m \rangle : m \in M \big\}$$

is a congruence relation on A.

*Proof.* Let A be an implicative BCS-algebra with maximal class M. Let B be the subalgebra of A with universe B := A - M and let  $\Xi_{\mathbf{B}}$  denote the Clifford-McLean congruence on B. Clearly  $[\Xi]$  is an equivalence relation on A such that  $\Xi_{\mathbf{B}} \subseteq [\Xi] \subseteq \Xi_{\mathbf{A}}$ , where  $\Xi_{\mathbf{A}}$  denotes the Clifford-McLean congruence on A. Suppose  $a_1 \equiv b_1 \pmod{\Xi}$  and  $a_2 \equiv b_2 \pmod{\Xi}$  for  $a_1, b_1, a_2, b_2 \in A$ . To see  $[\Xi]$  is a congruence on A we consider four cases:

- (i)  $a_1, b_1 \in B$  and  $a_2, b_2 \in B$ ; (iii)  $a_1 = b_1 \in M$  and  $a_2, b_2 \in B$ ;
- (ii)  $a_1, b_1 \in B$  and  $a_2 = b_2 \in M$ ; (iv)  $a_1 = b_1 \in M$  and  $a_2 = b_2 \in M$ .

Of these cases, Cases (i) and (iii) are non-trivial. For Case (i), suppose  $a_1, b_1 \in B$  and  $a_2, b_2 \in B$ . Then  $a_1 \equiv b_1 \pmod{\Xi_B}$  and  $a_2 \equiv b_2 \pmod{\Xi_B}$ , so  $a_1 a_2 \equiv b_1 b_2 \pmod{\Xi_B}$ , which implies  $a_1 a_2 \equiv b_1 b_2 \pmod{\Xi_B}$ .

For Case (iii), we distinguish two subcases:

(iii)(a)  $a_1 = b_1 \in M$  and  $a_2 = b_2 = 0 \in B$ ; (iii)(b)  $a_1 = b_1 \in M$  and  $0 \neq a_2 \in B$ ,  $0 \neq b_2 \in B$ .

Of these subcases, only Subcase (iii)(b) is non-trivial. So suppose  $a_1 = b_1 \in M$ and  $0 \neq a_2 \in B$ ,  $0 \neq b_2 \in B$ . Then  $a_1 \equiv b_1 \pmod{\Xi_A}$  and  $a_2 \equiv b_2 \pmod{\Xi_A}$ (since  $a_2 \equiv b_2 \pmod{\Xi_B}$ ), so  $a_1 a_2 \equiv b_1 b_2 \pmod{\Xi_A}$ . Because  $a_1 a_2 \notin M$  and  $b_1 b_2 \notin M$  (by Lemma 2.3.48) we have that  $a_1 a_2 \equiv b_1 b_2 \pmod{\Xi_B}$ , which implies  $a_1 a_2 \equiv b_1 b_2 \pmod{\Xi}$ .

Lemma \_ 3.50. Let A be an implicative BCS-algebra with maximal class M and let  $[\Xi]$  be the congruence relation on A of Lemma 2.3.49. Then the relation:

 $\lfloor \Xi \rfloor_n^m := \lfloor \Xi \rfloor \cup \big\{ \langle m, n \rangle, \langle n, m \rangle \big\}$ 

## is a congruence relation on A for any pair of maximal elements $\{m, n\} \subseteq M$ .

Proof. Let A be an implicative BCS-algebra with maximal class M. Let B be the subalgebra of A with universe B := A - M and let  $\Xi_{\mathbf{B}}$  denote the Clifford-McLean congruence on B. Let  $[\Xi]$  be the congruence relation on A of Lemma 2.3.49 and let  $m, n \in M$ . Clearly  $[\Xi]_n^m$  is an equivalence relation on A such that  $\Xi_{\mathbf{B}} \subseteq [\Xi] \subseteq [\Xi]_n^m \subseteq \Xi_{\mathbf{A}}$ , where  $\Xi_{\mathbf{A}}$  denotes the Clifford-McLean congruence on A. Suppose  $a_1 \equiv b_1 \pmod{[\Xi]_n^m}$  and  $a_2 \equiv b_2 \pmod{[\Xi]_n^m}$  for  $a_1, b_1, a_2, b_2 \in A$ . To see  $[\Xi]_n^m$  is a congruence on A we consider nine cases:

- (i)  $a_1, b_1 \in B$  and  $a_2, b_2 \in B$ ; (vi)  $a_1 = b_1 \in M$  and  $a_2, b_2 \in \{m, n\}$ ; (ii)  $a_1, b_1 \in B$  and  $a_2 = b_2 \in M$ ; (vii)  $a_1, b_2 \in \{m, n\}$  and  $a_2, b_2 \in \{m, n\}$ ;
- (ii)  $a_1, b_1 \in B$  and  $a_2 = b_2 \in M$ ; (vii)  $a_1, b_1 \in \{m, n\}$  and  $a_2, b_2 \in B$ ; (iii)  $a_1, b_1 \in B$  and  $a_2, b_2 \in \{m, n\}$ ; (viii)  $a_1, b_1 \in \{m, n\}$  and  $a_2 = b_2 \in M$ ;
- (iv)  $a_1 = b_1 \in M$  and  $a_2, b_2 \in B$ ; (ix)  $a_1, b_1 \in \{m, n\}$  and  $v_2, b_2 \in \{m, n\}$ .
- (v)  $a_1 = b_1 \in M$  and  $a_2 = b_2 \in M$ ;

Of these cases, Cases (ii), (iii), (v), (vi), (viii) and (ix) are trivial, while Cases (i) and (iv) are covered by Lemma 2.3.49 (since  $[\Xi] \subseteq [\Xi]_n^m$ ). For Case (vii), we distinguish two subcases:

- (vii)(a)  $a_1, b_1 \in \{m, n\}$  and  $a_2 = b_2 = 0 \in B$ ;
- (vii)(b)  $a_1, b_1 \in \{m, n\}$  and  $0 \neq a_2 \in B, 0 \neq b_2 \in B$ .

Of these subcases, only Subcase (vii)(b) is non-trivial. So suppose  $a_1, b_1 \in \{m, n\}$  and  $0 \neq a_2 \in B$ ,  $0 \neq b_2 \in B$ . Then  $a_1 \equiv b_1 \pmod{\Xi_A}$  (since  $a_1, b_1 \in M$ ) and  $a_2 \equiv b_2 \pmod{\Xi_A}$  (since  $a_2 \equiv b_2 \pmod{\Xi_B}$ ), so  $a_1a_2 \equiv b_1b_2 \pmod{\Xi_A}$ . Because  $a_1a_2 \notin M$  and  $b_1b_2 \notin M$  (by Lemma 2.3.48) we have that  $a_1a_2 \equiv b_1b_2 \pmod{\Xi_B}$ , which implies  $a_1a_2 \equiv b_1b_2 \pmod{\Xi_A}$ .

We are now in a position to complete the characterisation of the subdirectly irreducible implicative BCS-algebras. The proof is via three lemmas.

Lemma 2.3.51. Let A be a subdirectly irreducible implicative BCS-algebra such that |A| > 2. Then the relation  $[\Xi]$  of Lemma 2.3.49 is a congruence relation on A; moreover  $[\Xi] = \omega_A$ . *Proof.* Let A be a subdirectly irreducible implicative BCS-algebra such that |A| > 2. Then the relation  $[\Xi]$  is a congruence relation on A by Lemma 2.3.47 and Lemma 2.3.49. Let  $\{m_1, m_2\}$  be the pair of maximal elements identified under every non-trivial congruence on A as per Lemma 2.3.47. Since  $[\Xi]$  does not identify  $m_1$  and  $m_2$  it must be trivial; that is to say  $[\Xi] = \omega_A$ .

**Lemma 2.3.52.** Let A be a subdirectly irreducible implicative BCS-algebra such that |A| > 2. Then  $M = \{m_1, m_2\}$ , where M is the maximal class of A and  $\{m_1, m_2\}$  is the pair of maximal elements identified under every non-trivial congruence relation on A.

Proof. Let A be a subdirectly irreducible implicative BCS-algebra such that |A| > 2. By Lemma 2.3.47, A has a maximal class M such that  $\{m_1, m_2\} \subseteq M$ , where  $\{m_1, m_2\}$  is the pair of elements identified under every non-trivial congruence on A. Assume to the contrary that there exists  $n \in A$  such that  $n \neq m_1, m_2$  but  $n \in M$ . Let  $[\Xi]_n^{m_1} := [\Xi] \cup \{\langle m_1, n \rangle, \langle n, m_1 \rangle\}$  and let  $[\Xi]_n^{m_2} := [\Xi] \cup \{\langle m_2, n \rangle, \langle n, m_2 \rangle\}$ . By Lemma 2.3.50 both  $[\Xi]_n^{m_1}$  and  $[\Xi]_n^{m_2} = \omega_A \cup \{\langle m_2, n \rangle, \langle n, m_2 \rangle\}$  by Lemma 2.3.51. Thus  $[\Xi]_n^{m_1}$  and  $[\Xi]_n^{m_2} = \omega_A \cup \{\langle m_2, n \rangle, \langle n, m_2 \rangle\}$  by Lemma 2.3.51. Thus  $[\Xi]_n^{m_1}$  and  $[\Xi]_n^{m_2}$  are non-trivial congruences on A whose intersection  $[\Xi]_n^{m_1} \cap [\Xi]_n^{m_2}$  is the identity congruence  $\omega_A$ . Since this contradicts the subdirect irreducibility of A, the only maximal elements of A are the elements  $m_1, m_2$ . Thus  $M = \{m_1, m_2\}$ .

Lemma 2.3.53. Let A be a subdirectly irreducible implicative BCS-algebra such that |A| > 2. Then  $\mu = \omega_A \cup \{(m_1, m_2), (m_2, m_1)\} = \Xi$ , where  $\mu$  denotes the monolith on A.

Proof. Let A be a subdirectly irreducible implicative BCS-algebra such that |A| > 2. Let  $[\Xi]_{m_2}^{m_1} := [\Xi] \cup \{\langle m_1, m_2 \rangle, \langle m_2, m_1 \rangle\}$ . By Lemma 2.3.47 and Lemma 2.3.50  $[\Xi]_{m_2}^{m_1}$  is a congruence on A and by Lemma 2.3.51  $[\Xi]_{m_2}^{m_1} = \omega_A \cup \{\langle m_1, m_2 \rangle, \langle m_2, m_1 \rangle\}$ . Since  $m_1 \equiv m_2 \pmod{\mu}$  we have  $[\Xi]_{m_2}^{m_1} \subseteq \mu$ ; conversely  $\mu \subseteq [\Xi]_{m_2}^{m_1}$  because the monolith on A is contained in any non-trivial congruence on A. Thus  $\mu = [\Xi]_{m_2}^{m_1} = \omega_A \cup \{\langle m_1, m_2 \rangle, \langle m_2, m_1 \rangle\}$ .

For the remaining equivalence, let **B** be the subalgebra of **A** with universe  $B := A - \{m_1, m_2\}$  and let  $\Xi_{\mathbf{B}}$  be the Clifford-McLean congruence on **B**.

Let  $\Xi_{\mathbf{A}}$  denote the Clifford-McLean congruence on  $\mathbf{A}$  and let  $\Theta := \Xi_{\mathbf{B}} \cup \{\langle m_1, m_1 \rangle, \langle m_1, m_2 \rangle, \langle m_2, m_1 \rangle, \langle m_2, m_2 \rangle\}$ . From Lemma 2.3.52 we infer  $\Theta = \Xi_{\mathbf{A}}$ , because maximal elements always lie in their own distinct  $\Xi_{\mathbf{A}}$ -class. But clearly  $\Theta = [\Xi] \cup \{\langle m_1, m_2 \rangle, \langle m_2, m_1 \rangle\} = [\Xi]_{m_2}^{m_1} = \mu$ . Therefore  $\Xi_{\mathbf{A}} = \mu = \omega_{\mathbf{A}} \cup \{\langle m_1, m_2 \rangle, \langle m_2, m_1 \rangle\}$  and the lemma is proved.

Let A be a subdirectly irreducible implicative BCS-algebra such that |A| > 2. Then the underlying poset  $\langle A/\Xi; \leq^{A/\Xi} \rangle$  of the maximal implicative BCKalgebra homomorphic image  $A/\Xi$  is a Boolean lattice by Lemma 2.3.47 and Corollary 1.6.22. From Proposition 2.3.31 and Lemma 2.3.53 we deduce that the underlying poset  $\langle A; \leq \rangle$  of A is order isomorphic to a Boolean lattice with its unit element replaced by a two-element clique. This forces the following result.

**Theorem 2.3.54.** A non-trivial implicative BCS-algebra  $\mathbf{A}$  is subdirectly irreducible iff  $\mathbf{A}$  is isomorphic to  $\hat{\mathbf{B}}_0$  or  $\mathbf{A}$  is isomorphic to  $\hat{\mathbf{B}}$  for some non-trivial Boolean algebra  $\mathbf{B}$ .

**Corollary 2.3.55.** An implicative BCS-algebra is subdirectly irreducible iff it is isomorphic to the canonical implicative BCS-algebra polynomial reduct of a subdirectly irreducible pseudocomplemented semilattice.

Recall that for a class K of similar algebras  $K_{FIN}$  denotes the subclass of finite members of K.

Corollary 2.3.56. (cf. [70, Corollary 6.2]) The variety of implicative BCSalgebras is locally finite. Thus  $iBCS = V(iBCS_{FIN})$ ; that is, iBCS is generated as a variety by its finite members.

*Proof.* Let A be a finitely generated (say n generated) implicative BCS-algebra. Then each subdirectly irreducible homomorphic image of A is finitely generated. By Theorem 2.3.54 these finitely generated subdirectly irreducible homomorphic images are all finite; moreover to within isomorphism there are only finitely many such images that are generated by n or fewer elements. Let V be the variety generated by this finite set of finite and subdirectly irreducible algebras. Then  $\mathbf{A} \in \mathsf{V}$  and  $\mathsf{V}$  is locally finite, since any variety generated

by a finite set of finite algebras is locally finite (by [55, Theorem II§10.16]). Thus A is locally finite, and in particular finite. Since every finitely generated implicative BCS-algebra is finite, iBCS is locally finite as asserted. The second statement now follows, since any locally finite variety is generated as a variety by its finite members (see van Alten [229, p. 13]).

2.3.57. Quasi-Bounded Implicative BCS-Algebras. A quasi-bounded implicative BCS-algebra is an implicative BCS-algebra that is quasi-bounded. By the remarks of §2.1.40 and §2.3.1 the class iBCS<sup>1</sup> of all quasi-bounded implicative BCS-algebras is a variety, which coincides with the generic double-pointed expansion iBCS<sup>+</sup> of the variety of implicative BCS-algebras by Lemma 2.2.27. The study of quasi-bounded implicative algebras is prompted by Lemma 2.3.47, which asserts that any subdirectly irreducible implicative BCS-algebra  $\hat{B}$  with  $|\hat{B}| > 2$  has a maximal class, and thus gives rise to a quasi-bounded algebra  $\hat{B}^1$ . Our investigation of quasi-bounded implicative BCS-algebras begins with the following three results, which summarise some elementary properties of these algebras.

Lemma 2.3.58. The variety of quasi-bounded implicative BCS-algebras satisfies the following identities:

| $x \sqcap x^* pprox 0$ | (2.72) |
|------------------------|--------|
|------------------------|--------|

$$x^* \approx x^{***} \tag{2.73}$$

$$(x \setminus y)^{**} \approx x^{**} \setminus y^{**} \tag{2.74}$$

 $x^* \sqcap y^* \approx y^* \sqcap x^* \tag{2.75}$ 

$$x \sqcap (x \sqcup y) \approx x. \tag{2.76}$$

Moreover, for any quasi-bounded implicative BCS-algebra  $A^1$  and  $a, b \in A$ ,

 $a \leq b$  implies  $b^* \leq a^*$ .

*Proof.* Let  $A^1$  be a quasi-bounded implicative BCS-algebra and let  $a, b \in A$ . For (2.72),  $a \sqcap a^* = a(a(1a)) = aa = 0$  by (2.50) and (2.47). For (2.73),  $a^* = 1a = 1(1(1a)) = a^{***}$  by (2.56). For (2.74),  $(ab)^{**} = 1 \sqcap (ab) =$ 

 $(1b)(1a) = (1 \sqcap a)(1 \sqcap b) = a^{**}b^{**}$  by (2.25), (2.61) and (2.57). For (2.75),  $a^* \sqcap b^* = (a^*)^{**} \sqcap b^* = 1 \sqcap a^* \sqcap b^* = 1 \sqcap b^* \sqcap a^* = (b^*)^{**} \sqcap a^* = b^* \sqcap a^*$  by (2.73), (2.25), left normality, (2.25) and (2.73). For (2.76), put  $\alpha := a, \beta := 1$  and  $\gamma := a^* \sqcap b^*$ . We have:

$$a \sqcap (a \sqcup b) = a \sqcap (1(a^* \sqcap b^*))$$

$$= \alpha \sqcap (\beta \gamma)$$

$$= (\alpha \gamma)(\alpha \beta) \qquad by (2.61)$$

$$= (\alpha \sqcap \beta)(\alpha \sqcap \gamma) \qquad by (2.57)$$

$$= (a \sqcap 1)(a \sqcap (a^* \sqcap b^*))$$

$$= (a \sqcap 1)((a \sqcap a^*) \sqcap b^*)$$

$$= (a \sqcap 1)(0 \sqcap b^*) \qquad by (2.72)$$

$$= (a \sqcap 1)0 \qquad by (2.22)$$

$$= (a \sqcap 1) \qquad by (2.6)$$

$$= a \qquad by (2.24).$$

For the final assertion of the lemma, suppose  $a \le b$ . We have  $a^* \sqcap b^* = (1a) \sqcap (1b) = (1a)(1(1b)) = (1(1(1b)))a = (1b)a = (1b)(b(ba)) = 1b = b^*$ by (2.59), (2.48), (2.56) and (2.54). Since  $b^* \sqcap a^* = a^* \sqcap b^*$  (by (2.75)) we deduce  $b^* \le a^*$  as required.

Lemma 2.3.59. cf. ([10, Proposition 4.1]) The variety of quasi-bounded implicative BCS-algebras satisfies the following identities:

$$x \sqcup y \approx y \sqcup x \tag{2.77}$$

$$x \sqcup (y \sqcup z) \approx (x \sqcup y) \sqcup z. \tag{2.78}$$

Moreover, for any quasi-bounded implicative BCS-algebra  $A^1$  and  $a, b, c \in A$ ,

 $b \leq c$  implies  $a \sqcup b \leq a \sqcup c$ .

*Proof.* Let  $A^1$  be a quasi-bounded implicative BCS-algebra and let  $a, b, c \in A$ . For (2.77) we have  $a \sqcup b = (a^* \sqcap b^*)^* = (b^* \sqcap a^*)^* = b \sqcup a$  by (2.75). For (2.78) we have:

$$a \sqcup (b \sqcup c) = (a^* \sqcap ((b^* \sqcap c^*)^*)^*)^*$$
  
=  $(a^* \sqcap (b^* \sqcap c^*)^{**})^*$   
=  $(a^* \sqcap (1 \sqcap (b^* \sqcap c^*)))^*$  by (2.25)  
=  $((a^* \sqcap 1) \sqcap (b^* \sqcap c^*))^*$   
=  $(a^* \sqcap b^* \sqcap c^*)^*$  by (2.24).

But,

$$(a \sqcup b) \sqcup c = (((a^* \sqcap b^*)^*)^* \sqcap c^*)^*$$
  
=  $((a^* \sqcap b^*)^{**} \sqcap c^*)^*$   
=  $((1 \sqcap (a^* \sqcap b^*)) \sqcap c^*)^*$  by (2.25)  
=  $((1 \sqcap a^*) \sqcap (b^* \sqcap c^*))^*$   
=  $((a^*)^{**} \sqcap (b^* \sqcap c^*))^*$  by (2.25)  
=  $(a^* \sqcap b^* \sqcap c^*)^*$  by (2.73),

so we conclude  $a \sqcup (b \sqcup c) = (a \sqcup b) \sqcup c$  as required.

For the final assertion of the lemma, suppose  $b \leq c$ . Then  $c^* \leq b^*$  by Lemma 2.3.58, so  $a^* \sqcap c^* \leq a^* \sqcap b^*$  by Proposition 2.3.25(4), so  $(a^* \sqcap b^*)^* \leq (a^* \sqcap c^*)^*$  by Lemma 2.3.58; that is to say  $a \sqcup b \leq a \sqcup c$ .

**Proposition 2.3.60.** For any quasi-bounded implicative BCS-algebra  $A^1$ , the following assertions hold:

- The polynomial reduct (A; □, 1) is a commutative semigroup with identity whose operation □ is isotone with respect to the underlying natural band partial order of A<sup>1</sup>;
- 2. For any  $a, b \in A$ ,  $\langle a \rangle_{\mathbf{A}^1} \vee^{\mathbf{I}(\mathbf{A}^1)} \langle b \rangle_{\mathbf{A}^1} = \langle a \sqcup b \rangle_{\mathbf{A}^1}$ .

*Proof.* Item (1) follows as an immediate consequence of Lemma 2.3.59, (2.27) and (2.28). For Item (2), it is sufficient to show  $x \sqcup y$  is a join generator term for  $iBCS^1$ . Let  $A^1$  be a quasi-bounded implicative BCS-algebra and let  $a, b \in A$ .
Put  $r(x, y, z) := (x \sqcap z) \setminus (y \setminus x)$ . Then  $r^{A^1}(a, b, 0) = (a \sqcap 0)(ba) = 0(ba) = 0$ by (2.21) and (2.4) and  $r^{A^1}(a, b, a \sqcup b) = (a \sqcap (a \sqcup b))(ba) = a(ba) = a$ by (2.76) and (2.50). Similarly  $t^{A^1}(a, b, 0) = 0$  and  $t^{A^1}(a, b, a \sqcup b) = b$  for  $t(x, y, z) := (y \sqcap z) \setminus (x \setminus y)$ . Since  $0 \sqcup 0 = 0^{**} = 1 \sqcap 0 = 0$  by (2.26) and (2.25), we have that  $\sqcup$  is a join generator term for iBCS<sup>1</sup> by Proposition 1.7.13.

**Example 2.3.61.** (cf. [10, Example 4.4]) Let  $\mathbf{A} := \langle A; \wedge, *, 0 \rangle$  be a pseudocomplemented semilattice with greatest element  $1 := 0^*$ . For any  $a, b \in A$ , let  $a \setminus b := a \wedge b^*$  as per Example 2.3.12. The following assertions hold:

- 1. The polynomial reduct  $\langle A; \rangle, 0, 1 \rangle$  is a quasi-bounded implicative BCSalgebra;
- 2. For any  $a, b \in A$ ,

$$a \sqcap b = a \land b^{**},$$
$$(a^*)^{\langle A; \backslash, 0, 1 \rangle} = (a^*)^{\mathbf{A}},$$
$$a \sqcup b = (a^* \land b^*)^*.$$

*Proof.* For (1), by Example 2.3.12 it is sufficient to show  $a \leq 1$  for any  $a \in A$ . Let  $a \in A$ . We have  $a \setminus 1 = a \wedge 1^* = a \wedge 0^{**} = a \wedge 0 = 0$  by (1.11), so  $a \leq 1$  and  $\langle A; \langle 0, 1 \rangle$  is quasi-bounded.

For (2) let  $a, b \in A$ . We have  $a \sqcap b = a \setminus (a \setminus b) = a \wedge (a \wedge b^*)^* = a \wedge b^{**}$ by (1.9). Also  $(a^*)^{\langle A; \setminus, 0, 1 \rangle} = 1 \setminus a = 1 \wedge (a^*)^A = (a^*)^A$ . Hence  $a \sqcup b = (a^* \sqcap b^*)^* = (a^* \wedge b^{***})^* = (a^* \wedge b^*)^*$  by (2.73).

Example 2.3.61 and the theory of pseudocomplemented semilattices motivate the following definitions. For any quasi-bounded implicative BCS-algebra  $A^1$  let the *skeleton* of  $A^1$  be:

 $\mathcal{S}(\mathbf{A^1}) := \{a^* : a \in A\}.$ 

Also, define the *dense set* of  $A^1$  to be:

$$D(A^{1}) := \{a : a^{*} = 0\}.$$

Call  $A^1$  dense if  $D(A^1) = A - \{0\}$ . The following two technical lemmas collect together some useful properties of the skeleton  $S(A^1)$  and dense set  $D(A^1)$ .

**Lemma 2.3.62.** For any quasi-bounded implicative BCS-algebra  $A^1$  and  $a, b \in A$ , the following assertions hold:

- 1.  $0 \in S(A^1)$  and  $1 \in S(A^1)$ ;
- 2.  $a \in S(A^1)$  iff  $a = a^{**}$ ;
- 3.  $a, b \in S(A^1)$  implies  $ab \in S(A^1)$ .

Proof. Let  $\mathbf{A}^1$  be a quasi-bounded implicative BCS-algebra and let  $a, b \in A$ . For (1) just note  $0 = 11 = 1^* \in S(\mathbf{A}^1)$  by (2.3) and  $1 = 10 = 0^* \in S(\mathbf{A}^1)$  by (2.6). For (2), suppose  $a \in S(\mathbf{A}^1)$ . Then  $a = c^*$  for some  $c \in A$ , and  $a^{**} = (c^*)^{**} = c^* = a$  by (2.73). Conversely, suppose  $a = a^{**}$ . Then  $a = c^*$  with  $c := a^*$  so  $a \in S(\mathbf{A}^1)$ . For (3) suppose  $a, b \in S(\mathbf{A}^1)$ . Then  $a = a^{**}$  and  $b = b^{**}$  by (2) and so  $ab = a^{**}b^{**} = (ab)^{**}$  by (2.74); from (2) we conclude  $ab \in S(\mathbf{A}^1)$  as desired.

**Lemma 2.3.63.** For any quasi-bounded implicative BCS-algebra  $A^1$  and  $a, b \in A$ , the following assertions hold:

- *1.*  $1 \in D(A^1);$
- 2.  $a \in D(A^1)$  iff  $c \preceq a$  for any  $c \in A$ . Thus  $A^1$  is dense iff  $\langle A; \langle 0 \rangle$  is flat;
- 3.  $a^* = b^*$  iff  $a \equiv b \pmod{\Xi}$ .

Proof. Let  $A^1$  be a quasi-bounded implicative BCS-algebra and let  $a, b \in A$ . For (1), just note  $1^* = 11 = 0$  by (2.3). For (2), suppose  $a \in D(A^1)$  and  $c \in A$ . Since  $a^* = 0$ , we have  $ca = (ca)0 = (ca)a^* = (ca)(1a) = (c1)a = 0a$ (as  $c \leq 1$ ) = 0 by (2.49). Thus  $c \leq a$ . Conversely,  $c \leq a$  for any  $c \in A$  implies  $1 \leq a$  in particular. Thus 1a = 0 and  $a^* = 0$ . Hence  $a \in D(A^1)$ . The second assertion follows immediately.

For (3), suppose  $a \equiv b \pmod{\Xi}$ . Then ab = 0 = ba, whence  $a^* = 1a = (1a)0 = (1a)(ba) = (1b)(ab) = (1b)0 = 1b = b^*$  by (2.6), (2.55) and (2.6).

Conversely, suppose  $a^* = b^*$ . Then  $ab = (ab)(1a) = (ab)a^* = (ab)b^* = (ab)(1b) = (a1)b = 0b$  (as  $a \leq 1$ ) = 0 by (2.52), (2.49) and (2.6). Likewise  $ba = (ba)(1b) = (ba)b^* = (ba)a^* = (ba)(1a) = (b1)a = 0a$  (as  $b \leq 1$ ) = 0 by (2.52), (2.49) and (2.6). Thus  $a \equiv b \pmod{\Xi}$ .

For any quasi-bounded implicative BCS-algebra  $A^1$ , the following two theorems show the skeleton  $S(A^1)$  gives an internal description of the maximal bounded implicative BCK-algebra homomorphic image  $A^1/\Xi$ , in the sense that  $S(A^1)$  has the structure of a Boolean lattice order isomorphic to the underlying Boolean lattice of  $A^1/\Xi$  (recall Corollary 1.6.22).

**Theorem 2.3.64.** Let  $A^1$  be a quasi-bounded implicative BCS-algebra with skeleton  $S(A^1)$ . Then  $S(A^1)$  is a subuniverse of  $A^1$ . Thus the underlying natural band partial ordering of  $A^1$  partially orders  $S(A^1)$  and makes  $S(A^1)$ into a Boolean lattice. For any  $a, b \in S(A^1)$ , the meet and join in  $S(A^1)$  are respectively given by:

 $a \wedge b = a \sqcap b$  $a \vee b = (a^* \sqcap b^*)^*.$ 

*Proof.* Let  $A^1$  be a quasi-bounded implicative BCS-algebra with skeleton  $S(A^1)$ . By Lemma 2.3.62(1),  $\{0, 1\} \subseteq S(A^1)$ , while  $a, b \in S(A^1)$  implies  $ab \in S(A^1)$ by Lemma 2.3.62(3). Thus  $S(A^1)$  is a subuniverse of  $A^1$  and so inherits the underlying natural band partial ordering of  $A^1$ . Let  $a, b \in S(A^1)$ . By Lemma 2.3.62(2)  $a = a^{**}$  and  $b = b^{**}$ , whence:

 $a \sqcap b = a^{**} \sqcap b^{**}$ =  $(1 \sqcap a) \sqcap (1 \sqcap b)$  by (2.25) =  $(1 \sqcap b) \sqcap (1 \sqcap a)$  by left normality =  $b^{**} \sqcap a^{**}$  by (2.25) =  $b \sqcap a$ ,

so the quasi-bounded subalgebra  $\langle S(A^1); \langle 0, 1 \rangle$  is a bounded implicative BCK-algebra. From Corollary 1.6.22 we have that  $\langle S(A^1); \leq \rangle$  is a Boolean

lattice, where  $a \wedge b = a \sqcap b$  and  $a \lor b = (a^* \sqcap b^*)^*$  for any  $a, b \in S(A^1)$ .

Let  $A^1$  be a quasi-bounded implicative BCS-algebra with skeleton  $S(A^1)$ . In the statement of the following theorem and in the sequel, let  $S(A^1)$  denote the bounded implicative BCK-algebra  $\langle S(A^1); \langle 0, 1 \rangle$ .

**Theorem 2.3.65.** (cf. [10, Proposition 4.4]) Let  $\mathbf{A}^1$  be a quasi-bounded implicative BCS-algebra. Then  $\mathbf{S}(\mathbf{A}^1)$  is isomorphic to the maximal bounded implicative BCK-algebra homomorphic image  $\mathbf{A}^1/\Xi$  of  $\mathbf{A}^1$  under the map  $a \stackrel{h}{\mapsto} [a]_{\Xi}$  ( $a \in \mathbf{S}(\mathbf{A}^1)$ ).

Proof. Let  $A^1$  be a quasi-bounded implicative BCS-algebra. Suppose  $\overline{b} \in A/\Xi$ . Then  $\overline{b} = [a]_{\Xi}$  for some  $a \in A$ . Now  $a^{**} \in S(A^1)$  and  $a^{**} \equiv a \pmod{\Xi}$  by Lemma 2.2.13, so  $[a^{**}]_{\Xi} = [a]_{\Xi} = \overline{b}$ . Thus h is onto. Let  $a, b \in S(A^1)$ and suppose  $a \equiv b \pmod{\Xi}$ . From Lemma 2.3.63(3) we have that  $a^* = b^*$ . Therefore  $a^{**} = b^{**}$  and  $a = a^{**} = b^{**} = b$  since  $a = a^{**}$  and  $b = b^{**}$ by Lemma 2.3.62(2). Thus h is one-to-one, and hence is a bijection between  $S(A^1)$  and  $A/\Xi$ .

Let  $a, b \in S(A)^1$ . By Lemma 2.3.62(2) we have that  $a = a^{**}$  and  $b = b^{**}$ , so  $h(a \setminus S(A^1)b) = h(a \setminus A^1b)$  (by Theorem 2.3.64)  $= h(a^{**} \setminus A^1b^{**}) = [a^{**} \setminus A^1b^{**}]_{\Xi} = [a^{**}]_{\Xi} \setminus A^{1/\Xi}[b^{**}]_{\Xi} = [a]_{\Xi} \setminus A^{1/\Xi}[b]_{\Xi} = h(a) \setminus A^{1/\Xi}h(b)$ . Moreover,  $h(O^{S(A^1)}) = h(O^{A^1}) = [O^{A^1}]_{\Xi} = O^{A^1}/\Xi$  and  $h(1^{S(A^1)}) = h(1^{A^1}) = [1^{A^1}]_{\Xi} = 1^{A^1}/\Xi$ . Thus h is a map from  $S(A^1)$  into  $A^1/\Xi$  preserving  $\setminus S(A^1)$ ,  $O^{S(A^1)}$  and  $1^{S(A^1)}$ ; that is to say h is an isomorphism.

As an application of Theorem 2.3.64 and Theorem 2.3.65 in a very natural setting, we give below a new and conceptually simple proof of the Glivenko-Frink theorem for pseudocomplemented semilattices.

**Theorem 2.3.66 (Glivenko-Frink Theorem).** Let A be a pseudocomplemented semilattice with canonical quasi-bounded implicative BCS-algebra polynomial reduct  $\langle A; \backslash, 0, 1 \rangle$ . Then S(A) is a subuniverse of A and so inherits the underlying semilattice partial ordering of A. Moreover S(A) = S( $\langle A; \backslash, 0, 1 \rangle$ ) and the semilattice partial ordering on S(A) and the natural band partial ordering on S( $\langle A; \backslash, 0, 1 \rangle$ ) coincide on S(A). Thus the semilattice partial ordering on S(A) makes S(A) into a Boolean lattice, and this Boolean lattice is order isomorphic to the underlying poset of the maximal bounded implicative BCKalgebra homomorphic image of  $\langle A; \rangle, 0, 1 \rangle$ . For any  $a, b \in S(A)$  we have  $a \wedge b \in S(A)$ , and the join in S(A) is described by:

 $a \vee b = (a^* \wedge b^*)^*.$ 

*Proof.* Let A and  $\langle A; \langle 0, 1 \rangle$  be as in the statement of the theorem. Because of Theorem 2.3.64 and Theorem 2.3.65, to prove the theorem it is sufficient to show:

- (i) S(A) is a subuniverse of A;
- (ii)  $S(\mathbf{A}) = S(\langle A; \langle 0, 1 \rangle);$
- (iii) The semilattice partial ordering on S(A) and the natural band partial ordering on  $S(\langle A; \langle 0, 1 \rangle)$  coincide on S(A).

For (i), by the remarks of §1.3.5 we have both  $0 \in S(A)$  and  $a, b \in S(A)$ implies  $a \wedge b \in S(A)$ . Since S(A) is closed under \* by definition we have that S(A) is a subuniverse of A (this observation is also implicit in the original statement of the Glivenko-Frink theorem (Theorem 1.3.10)).

For (ii) just note  $(a^*)^{\langle A; \backslash, 0, 1 \rangle} = (a^*)^{\mathbf{A}}$  for any  $a \in A$  by Example 2.3.61(2), and hence that  $S(\mathbf{A}) = \{(a^*)^{\mathbf{A}} : a \in A\} = \{(a^*)^{\langle A; \backslash, 0, 1 \rangle}\} = S(\langle A; \backslash, 0, 1 \rangle).$ 

For (iii), let  $a, b \in S(A)$ . We have  $a \leq^A b$  iff  $a \wedge b = a = b \wedge a$  iff  $a \wedge b^{**} = a = b^{**} \wedge a$  (since  $b^{**} = b$  by Lemma 2.3.62(2)) iff  $a \sqcap b = a = b \sqcap a$  iff  $a \leq^{\langle A; \backslash, 0, 1 \rangle} b$ , so the semilattice partial ordering on S(A) and the natural band partial ordering on S( $\langle A; \backslash, 0, 1 \rangle$ ) coincide on S(A).

Remark 2.3.67. The Glivenko-Frink theorem for pseudocomplemented semilattices has been established by at least three methods different from the above: see the remark prior to [101, Theorem 1§6.4]. The present proof is direct and would seem in principle to be the most elementary, simpler even than the short proof of Katriňák [132].

Let A be an implicative BCS-algebra. A multiplier on A is a function f:  $A \to A$  such that  $f(a \sqcap b) = f(a) \sqcap b$  for all  $a, b \in A$ . The set of all mul-

#### 2.3. Implicative BCS-Algebras

tipliers is denoted M(A). Multiplier extensions of implicative BCK-algebras have been studied by Cornish [66]; what follows here is a version of a construction due to Bignall [17, pp. 49–51]. For any functions  $f, g: A \to A$ , define  $f \setminus g : A \to A$  and  $f \sqcap g : A \to A$  pointwise by  $(f \setminus g)(a) := f(a) \setminus g(a)$  and  $(f \cap g)(a) := f(a) \cap g(a)$  for all  $a \in A$  respectively. Now  $f \setminus g \in M(\mathbf{A})$  since  $(f \setminus g)(a \sqcap b) = f(a \sqcap b) \setminus g(a \sqcap b) = (f(a) \sqcap b) \setminus (g(a) \sqcap b) = (f(a) \setminus g(a)) \sqcap b$  $(by (2.62)) = (f \setminus g)(a) \sqcap b$ . Also  $f \sqcap g \in M(\mathbf{A})$ , because  $(f \sqcap g)(a \sqcap b) =$  $f(a \sqcap b) \sqcap g(a \sqcap b) = f(a) \sqcap b \sqcap g(a) \sqcap b = f(a) \sqcap g(a) \sqcap b$  (by left normality) =  $(f \sqcap g)(a) \sqcap b$ . Define  $0 \in M(A)$  by 0(a) := 0 and  $1 \in M(A)$  by 1(a) := a for all  $a \in A$ . Put  $M(A) := \langle M(A); \backslash, 0, 1 \rangle$ . Because the operations on M(A) are defined pointwise, the reduct  $(M(A); \setminus, 0)$  of M(A) is an implicative BCS-algebra. Moreover, because  $(f \sqcap g)(a) = f(a) \sqcap g(a) =$  $f(a)\setminus (f(a)\setminus g(a)) = f(a)\setminus ((f\setminus g)(a)) = (f\setminus (f\setminus g))(a), f \sqcap g$  is the implicative BCS meet in  $(M(\mathbf{A}); \backslash, 0)$ . Since  $(f \sqcap 1)(a) = f(a) \sqcap 1(a) = f(a) \sqcap a =$  $f(a \sqcap a) = f(a)$ , from Lemma 2.1.42(1) we deduce  $f \preceq 1$  for any  $f \in M(\mathbf{A})$ ; that is to say M(A) is a quasi-bounded implicative BCS-algebra.

Let  $\mu_a : A \to A$  denote the map defined by  $\mu_a(b) := a \sqcap b$ . Since  $\mu_a(b \sqcap c) = a \sqcap (b \sqcap c) = (a \sqcap b) \sqcap c = \mu_a(b) \sqcap c$  for all  $b, c \in A$  we have  $\mu(a) \in M(A)$  for each  $a \in A$ . Define  $\mu : A \to M(A)$  by  $\mu(a) := \mu_a$ . Then  $\mu$  is a homomorphism, since  $\mu(a \setminus b)(c) = \mu_{(a \setminus Ab)}(c) = (a \setminus b) \sqcap c = (a \sqcap c) \setminus A(b \sqcap c)$  (by (2.62))  $= \mu_a(c) \setminus M(A) \mu_b(c)$  and  $\mu(0^A)(c) = \mu_{0^A}(c) = 0^A \sqcap c = 0^A = 0(c) = 0^{M(A)}$ . Moreover,  $\mu$  is one-to-one, since  $\mu(a) = \mu(b)$  implies  $\mu_a(c) = \mu_b(c)$  for all  $c \in A$ , which implies in particular that  $\mu_a(a) = \mu_b(a)$  and  $\mu_a(b) = \mu_b(b)$ . But then  $a = b \sqcap a$  and  $b = a \sqcap b$ , so  $b = a \sqcap b = b \sqcap a \sqcap b = b \sqcap b \sqcap a$  (by left normality)  $= b \sqcap a = a$ . Hence  $\mu$  is an isomorphism from A onto the subalgebra  $\mu[A]$  of M(A), where  $\mu[A]$  denotes the image of A under  $\mu$ .

Lemma 2.3.68. Every implicative BCS-algebra  $\mathbf{A}$  embeds (as an implicative BCS-algebra) into its canonical quasi-bounded implicative BCS-algebra multiplier extension  $\mathbf{M}(\mathbf{A})$ .

By a  $\langle \backslash, 0 \rangle$ -subreduct of a pseudocomplemented semilattice  $\mathbf{A} := \langle A; \land, *, 0 \rangle$ we mean a subalgebra of the canonical implicative BCS-algebra polynomial reduct  $\langle A; \backslash, 0 \rangle$  of  $\mathbf{A}$ . We conclude this subsection with the following problem, suggested by the preceding lemma and Theorem 1.6.20.

**Problem 2.3.69.** Is an algebra  $\langle A; \rangle, 0 \rangle$  of type  $\langle 2, 0 \rangle$  an implicative BCSalgebra iff it is a  $\langle \rangle, 0 \rangle$ -subreduct of a pseudocomplemented semilattice?

2.3.70. The Lattice of Varieties of Implicative BCS-Algebras. Problem 2.3.69 and the results of the preceding subsections suggest that the theory of implicative BCS-algebras may bear the same relationship to the theory of pseudocomplemented semilattices as the theory of implicative BCKalgebras bears to the theory of Boolean algebras. Because the canonical implicative BCS-algebra polynomial reduct of the 3-element chain 3 (considered as a pseudocomplemented semilattice) is flat, Theorem 1.3.8 and the preceding remarks call for a closer examination of the role played by the algebra  $B_2 := \langle \{0, 1, 2\}; \langle , 0 \rangle$  of Example 2.1.5 in the variety of implicative BCSalgebras.

**Lemma 2.3.71.** Let  $\mathbf{B}_{2}^{k}$ ,  $k \in \omega$ , be the k-th direct power of  $\mathbf{B}_{2}$ ; that is, the direct product of k copies of  $\mathbf{B}_{2}$ . Let:

$$M := \{(c_1, \ldots, c_k) : c_i \neq 0; i = 1, \ldots, k\}$$

be the maximal class of  $\mathbf{B}_2^k$  and let:

$$M_1 := \{ c \in M : \pi_1(c) = 1 \} \quad and \quad M_2 := \{ c \in M : \pi_1(c) = 2 \},$$

where  $\pi_1$  denotes the first projection map. Let **B** be the subalgebra of  $\mathbf{B}_2^k$  with universe  $B := B_2^k - M$  and let  $\Xi_{\mathbf{B}}$  denote the Clifford-McLean congruence on **B**. Then the relation:

 $\Theta := \Xi_{\mathbf{B}} \cup (M_1 \times M_1) \cup (M_2 \times M_2)$ 

is a congruence on  $\mathbf{B}_2^k$ .

*Proof.* Let  $\mathbf{B}_2^k$ , M,  $M_1$ ,  $M_2$ ,  $\mathbf{B}$ ,  $\Xi_{\mathbf{B}}$  and  $\Theta$  be as in the statement of the lemma. Since  $\{B, M_1, M_2\}$  partitions  $B_2^k$  we infer  $\Theta$  is an equivalence relation on  $B_2^k$ . Moreover, clearly  $\Xi_{\mathbf{B}} \subseteq \Theta \subseteq \Xi_{\mathbf{B}_2^k}$ , where  $\Xi_{\mathbf{B}_2^k}$  denotes the Clifford-McLean congruence on  $\mathbb{B}_2^k$ . Suppose  $a_1 \equiv b_1 \pmod{\Theta}$  and  $a_2 \equiv b_2 \pmod{\Theta}$  for  $a_1, b_1, a_2, b_2 \in A$ . To see  $\Theta$  is a congruence we consider nine cases:

- (i)  $a_1, b_1 \in B$  and  $a_2, b_2 \in B$ ; (vi)  $a_1, b_1 \in M_1$  and  $a_2, b_2 \in M_2$ ;
- (ii)  $a_1, b_1 \in B$  and  $a_2, b_2 \in M_1$ ; (vii)  $a_1, b_1 \in M_2$  and  $a_2, b_2 \in B$ ;
- (iii)  $a_1, b_1 \in B$  and  $a_2, b_2 \in M_2$ ; (viii)  $a_1, b_1 \in M_2$  and  $a_2, b_2 \in M_1$ ;
- (iv)  $a_1, b_1 \in M_1$  and  $a_2, b_2 \in B$ ; (ix)  $a_1, b_1 \in M_2$  and  $a_2, b_2 \in M_2$ .
- (v)  $a_1, b_1 \in M_1$  and  $a_2, b_2 \in M_1$ ;

Of these cases, Cases (i), (iv) and (vii) are non-trivial. For Case (i), suppose  $a_1, b_1 \in B$  and  $a_2, b_2 \in B$ . Then  $a_1 \equiv b_1 \pmod{\Xi_B}$  and  $a_2 \equiv b_2 \pmod{\Xi_B}$ , so  $a_1a_2 \equiv b_1b_2 \pmod{\Xi_B}$ , which implies  $a_1a_2 \equiv b_1b_2 \pmod{\Theta}$ .

Cases (iv) and (vii) are analogous, so we show only Case (iv). For Case (iv), we distinguish two subcases:

(iv)(a)  $a_1, b_1 \in M_1$  and  $a_2 = b_2 = 0 \in B$ ;

(iv)(b)  $a_1, b_1 \in M_1$  and  $0 \neq a_2 \in B, 0 \neq b_2 \in B$ .

Of these subcases, only Subcase (iv)(b) is non-trivial. So suppose  $a_1, b_1 \in M_1$ and  $0 \neq a_2 \in B$ ,  $0 \neq b_2 \in B$ . Then  $a_1 \equiv b_1 \pmod{\Xi_{\mathbf{B}_2^k}}$  (since  $a_1, b_1 \in M$ ) and  $a_2 \equiv b_2 \pmod{\Xi_{\mathbf{B}_2^k}}$  (since  $a_2 \equiv b_2 \pmod{\Xi_{\mathbf{B}}}$ ), so  $a_1 a_2 \equiv b_1 b_2 \pmod{\Xi_{\mathbf{B}_2^k}}$ . Because  $a_1 a_2 \notin M$  and  $b_1 b_2 \notin M$  (by Lemma 2.3.48) we have that  $a_1 a_2 \equiv b_1 b_2 \pmod{\Xi_{\mathbf{B}}}$ , which implies  $a_1 a_2 \equiv b_1 b_2 \pmod{\Theta}$ .

**Lemma 2.3.72.** Let  $\mathbf{B}_2^k$ ,  $k \in \omega$ , be the k-th direct power of  $\mathbf{B}_2$  and let  $\Theta$  be the congruence relation on  $\mathbf{B}_2^k$  of Lemma 2.3.71. Then the quotient algebra  $\mathbf{B}_2^k/\Theta$  is isomorphic to  $\hat{\mathbf{B}}_k$ , the finite and subdirectly irreducible implicative BCS-algebra of cardinality  $2^k + 1$ .

*Proof.* Let  $\mathbf{B}_2^k$  and  $\Theta$  be an in the statement of the lemma. To prove the lemma it is sufficient to show:

- (i)  $\mathbf{B}_2^k / \Theta$  is subdirectly irreducible;
- (ii)  $|B_2^k / \Theta| = 2^k + 1$ .

For (i), for ease of notation let  $\mathbf{B} := \mathbf{B}_2^k / \Theta$  and let  $\Xi_{\mathbf{B}}$  denote the Clifford-McLean congruence on  $\mathbf{B}$ . Now  $\Xi_{\mathbf{B}} = \Xi_{\mathbf{B}_2^k / \Theta} = \Xi_{\mathbf{B}_2^k} / \Theta$  (where  $\Xi_{\mathbf{B}_2^k / \Theta}$  and  $\Xi_{\mathbf{B}_2^k}$  denote the Clifford-McLean congruence on  $\mathbf{B}_2^k / \Theta$  and  $\mathbf{B}_2^k$  respectively), so

 $\Xi_{\mathbf{B}} = \omega_{\mathbf{B}} \cup \{ \langle M_1, M_2 \rangle, \langle M_2, M_1 \rangle \}, \text{ where } M_1 \text{ and } M_2 \text{ are as in Lemma 2.3.71.}$ Therefore to see **B** is subdirectly irreducible (with monolith  $\Xi_{\mathbf{B}}$ ) it is sufficient to show  $M_1 \equiv_{\theta} M_2$  for any  $\theta \in \text{Con } \mathbf{B}$  such that  $\theta \neq \omega_{\mathbf{B}}$ . So let  $\theta \in \text{Con } \mathbf{B}$  be such that  $\theta \neq \omega_{\mathbf{B}}$ . There are two cases to consider:

- (1)  $[0^{\mathbf{B}}]_{\theta} = \{0^{\mathbf{B}}\};$
- (2)  $[\mathbf{0}^{\mathbf{B}}]_{\theta} \neq \{\mathbf{0}^{\mathbf{B}}\}.$

For Case (1), suppose  $[\mathbf{0}^{\mathbf{B}}]_{\theta} = \{\mathbf{0}^{\mathbf{B}}\}$ . Then  $\theta \subseteq \Xi_{\mathbf{B}} = \omega_{\mathbf{B}} \cup \{\langle M_1, M_2 \rangle, \langle M_2, M_1 \rangle\}$ by Theorem 2.1.14(4). Assume to the contrary that  $M_1 \not\equiv_{\theta} M_2$ . Then  $\theta = \omega_{\mathbf{B}}$ which is a contradiction. Thus  $M_1 \equiv_{\theta} M_2$ .

For Case (2), suppose  $[\mathbf{0}^{\mathbf{B}}]_{\theta} \neq \{\mathbf{0}^{\mathbf{B}}\}$ . Then there exists  $\mathbf{0}^{\mathbf{B}} \neq A \in B$  (where  $B := B_2^k / \Theta$ ) such that  $\mathbf{0}^{\mathbf{B}} \equiv_{\theta} A$ . We consider two subcases:

- (2)(a)  $A \notin \{M_1, M_2\};$
- (2)(b)  $A \in \{M_1, M_2\}.$

For Subcase (2)(a), suppose  $A \notin \{M_1, M_2\}$ . From  $M_1 \equiv M_2 \pmod{\Xi_B}$  and Lemma 2.1.12(1) we have  $M_1 \setminus^B A \equiv M_2 \setminus^B A \pmod{\Xi_B}$ . But  $M_1 \setminus^B A \notin M$  and  $M_2 \setminus^B A \notin M$  by Lemma 2.3.48, where M is the maximal class of B, which implies by the description of the Clifford-McLean congruence  $\Xi_B$  that  $M_1 \setminus^B A =$  $M_2 \setminus^B A$ . But then  $M_1 = M_1 \setminus^B 0^B \equiv_{\theta} M_1 \setminus^B A = M_2 \setminus^B A \equiv_{\theta} M_2 \setminus^B 0^B = M_2$ , whence  $M_1 \equiv_{\theta} M_2$ .

For Subcase (2)(b), suppose  $A \in \{M_1, M_2\}$ . Let  $C \in \{M_1, M_2\}$  be such that  $C \neq A$ . From  $C \equiv A \pmod{\Xi_{\mathbf{B}}}$  we have  $C = C \setminus^{\mathbf{B}} \mathbf{0}^{\mathbf{B}} \equiv_{\theta} C \setminus^{\mathbf{B}} A = \mathbf{0}^{\mathbf{B}} \equiv_{\theta} A$  as required. This completes the proof that **B** is subdirectly irreducible.

For (ii), just note  $|B| = |B_2^k / \Theta| = |B_2^k / \Xi_{\mathbf{B}_2^k}| - 1 + 2$ . But  $\mathbf{B}_2^k / \Xi_{\mathbf{B}_2^k} \cong (\mathbf{B}_2 / \Xi_{\mathbf{B}_2})^k$ , so  $|B_2^k / \Xi_{\mathbf{B}_2^k}| = 2^k$ , which implies  $|B| = |B_2^k / \Theta| = 2^k - 1 + 2 = 2^k + 1$ . Thus **B** has cardinality  $2^k + 1$  as asserted.

**Theorem 2.3.73.** The 3-element flat implicative BCS-algebra  $B_2$  generates the class of implicative BCS-algebras (as a variety). In symbols, iBCS =  $V(B_2)$ .

*Proof.* Since  $\mathbf{B}_2 \in iBCS$  we have  $\mathbf{V}(\mathbf{B}_2) \subseteq iBCS$ . Conversely, by Theorem 2.3.54 we have that any finite and subdirectly irreducible implicative

BCS-algebra is isomorphic either to  $\hat{\mathbf{B}}_0$  or to some  $\hat{\mathbf{B}}_n$ ,  $n \in \omega$ . Now  $\mathbf{B}_2/\Xi$ is isomorphic to  $\hat{\mathbf{B}}_0$ , so  $\hat{\mathbf{B}}_0 \in \mathbf{H}(\mathbf{B}_2)$ . For any  $k \geq 1$ , let  $\mathbf{B}_2^k$  be the k-th direct power of  $\mathbf{B}_2$  and let  $\Theta$  be the relation defined on  $B_2^k$  of Lemma 2.3.71. By Lemma 2.3.71 and Lemma 2.3.72 we have that  $\Theta$  is a congruence relation on  $\mathbf{B}_2^k$  such that  $\mathbf{B}_2^k/\Theta$  is isomorphic to  $\hat{\mathbf{B}}_k$ , so  $\hat{\mathbf{B}}_k \in \mathbf{HP}(\mathbf{B}_2)$ . Suppose now that  $\mathbf{A}$  is a finite implicative BCS-algebra. Then  $\mathbf{A}$  is isomorphic to a subdirect product of finite and subdirectly irreducible implicative BCS-algebras, and so  $\mathbf{A} \in \mathbf{IPsHPH}(\mathbf{B}_2)$ . But  $\mathbf{IPsHPH} \leq \mathbf{IPsHHP}$  (by [160, Lemma 4.92]) =  $\mathbf{IPsHP} \leq \mathbf{HPsHP} \leq \mathbf{HHPsP}$  (by [160, Lemma 4.92]) =  $\mathbf{HPsP} = \mathbf{HPs}$ (by [160, Lemma 4.92])  $\leq \mathbf{HSP}$ , so  $\mathbf{A} \in \mathbf{V}(\mathbf{B}_2)$ . Thus  $\mathbf{V}(\mathbf{B}_2)$  contains every finite implicative BCS-algebra; that is to say  $\mathbf{iBCS}_{FIN} \subseteq \mathbf{V}(\mathbf{B}_2)$ . But this implies  $\mathbf{V}(\mathbf{iBCS}_{FIN}) \subseteq \mathbf{V}(\mathbf{B}_2)$ , which implies  $\mathbf{iBCS} \subseteq \mathbf{V}(\mathbf{B}_2)$  by Corollary 2.3.56.

**Corollary 2.3.74.** The following assertions hold in the variety of implicative BCS-algebras:

- 1. The equational theory of the variety of implicative BCS-algebras is decidable;
- 2. The first-order theory of the variety of implicative BCS-algebras is undecidable.

**Proof.** Item (1) follows from local finiteness (Corollary 2.3.56) and a well known result due to Harrop [107] (see also Blok and Ferreirim [27, Lemma 3.13]) to the effect that a variety V of algebras over a finite language has a decidable equational theory if V is finitely axiomatisable and is generated (as a variety) by its finite members. Item (2) follows immediately from the remarks of Example 2.1.5 and Theorem 2.3.73.

Let V be a variety. Denote by  $\Lambda^{V}(V)$  the 'lattice of varieties' of V, namely the dual of the lattice of corresponding equational theories (see for example Grätzer [99, p. 172]). For pseudocomplemented semilattices, it follows easily from the fact that the 3-element chain 3 (considered as a pseudocomplemented semilattice) generates PCSL as a variety (recall Theorem 1.3.8) that  $\Lambda^{V}(PCSL)$ is a 3-element chain, whose unique atom { $A \in PCSL : A \models x^{**} \approx x$ } is termwise definitionally equivalent to the variety of Boolean algebras (recall Theorem 1.3.9). This remark, in conjunction with Theorem 2.3.73, calls for a study of  $\Lambda^{V}(iBCS)$ , the lattice of varieties of implicative BCS-algebras. But first, concerning the lattice of varieties of pre-BCK-algebras  $\Lambda^{V}(PBCK)$ , recall iBCK is the unique atom of  $\Lambda^{V}(PBCK)$  by the remarks of Example 2.1.4.

**Theorem 2.3.75.**  $\Lambda^{V}(\text{iBCS})$  is a three-element chain. The only non-trivial subvariety of iBCS is iBCK, the unique atom of  $\Lambda^{V}(\text{PBCK})$ . Thus iBCS is a cover of iBCK in  $\Lambda^{V}(\text{PBCK})$  (in fact, is the only cover of iBCK in  $\Lambda^{V}(\text{PBCK})$ ) that is not a variety of BCK-algebras).

Proof. Let  $V \subseteq iBCS$ . If  $V \subseteq BCK$  then V = iBCK by Proposition 2.3.7(2). If  $V \not\subseteq BCK$  then  $\mathbf{Q}(\mathbf{B}_2) \subseteq V$  by Proposition 2.2.5, so  $iBCS = \mathbf{V}(\mathbf{B}_2) \subseteq V$  by Theorem 2.3.73. Thus iBCS is a cover of iBCK in  $\Lambda^V(PBCK)$  (in fact, is the only cover of iBCK in  $\Lambda^V(PBCK)$  that is not a variety of BCK-algebras).

By Example 2.1.4 iBCK is also the unique atom of  $\Lambda^Q(\text{PBCK})$ , the lattice of quasivarieties of pre-BCK-algebras. However, Theorem 2.3.75 cannot be generalised to the assertion that iBCS covers iBCK in  $\Lambda^Q(\text{PBCK})$ , in view of the following result of Blok and Raftery [38]. The proof given here is new.

**Proposition 2.3.76.** [38, Proposition 6] The quasivariety  $\mathbf{Q}(\mathbf{B}_2)$  is not a variety. Thus iBCS is not a cover of iBCK in  $\Lambda^Q(\text{PBCK})$ .

*Proof.* Assume to the contrary that  $Q(B_2)$  is a variety. By Theorem 2.3.73 and hypothesis we have  $iBCS = V(B_2) = Q(B_2) = ISP(B_2)$ , which implies  $iBCS_{SI} \subseteq IS(B_2)$ , a contradiction. The remaining assertion now follows.

Blok and Raftery first proved Proposition 2.3.76 by showing that the algebra  $B_2 \times C_1$  has a homomorphic image isomorphic to  $\hat{B}_2$ , and hence that  $H(B_2 \times C_1) \not\subseteq IPs(B_2, C_1)$ . Thus they do not exhibit a quasi-identity satisfied by  $Q(B_2)$  but not by iBCS. We also have been unable to exhibit such a quasi-identity.

Problem 2.3.77. Exhibit a quasi-identity satisfied by  $Q(B_2)$  but not by iBCS.

For BCK-algebras, the partially ordered 'set'  $P^{V}(BCK)$  of all subvarieties of BCK has been investigated by several authors, including Wroński [239], Wroński and Kabziński [243], Blok and Raftery [38] and Kowalski [142] (see also Van Alten [229, Chapter 6] and van Alten and Raftery [231]), and in particular it is known that  $P^{V}(BCK) := \langle P^{V}(BCK); \subseteq \rangle$  is a (distributive) lattice with unique atom iBCK [38, Theorem 11]. Let  $\mathbf{L}_3 := \langle \{0, 1, 2\}; \div, 0 \rangle$  be the BCK-algebra with  $0 \le 1 \le 2$  and  $2 \div 1 := 1$  and let  $\mathbf{H}_3 := \langle \{0, 1, 2\}; \div, 0 \rangle$  be the BCK-algebra with  $0 \le 1 \le 2$  and  $2 \div 1 := 2$ . The algebras  $\mathbf{L}_3$  and  $\mathbf{H}_3$  are dually isomorphic to the implicational reducts of the 3-element Lukasiewicz algebra and the 3-element linearly ordered Heyting algebra, respectively, and by Jónsson's lemma  $\mathbf{V}(\mathbf{L}_3)$  and  $\mathbf{V}(\mathbf{H}_3)$  are covers of iBCK in the lattice  $\mathbf{P}^{V}(BCK)$ . By results due to Kowalski [142] the converse also holds, and so  $\mathbf{V}(\mathbf{L}_3), \mathbf{V}(\mathbf{H}_3)$ are the only covers of iBCK in  $\mathbf{P}^{V}(BCK)$ .

**Theorem 2.3.78.** The varieties  $V(\mathbf{L}_3)$ ,  $V(\mathbf{H}_3)$  and iBCS are the only covers of iBCK in  $\Lambda^{V}(\text{PBCK})$ .

Proof. Let V be a cover of iBCK in  $\Lambda^{V}(PBCK)$ . By Proposition 2.2.5 either  $V \in P^{V}(BCK)$  or  $\mathbf{Q}(\mathbf{B}_{2}) \subseteq V$ . If  $\mathbf{Q}(\mathbf{B}_{2}) \subseteq V$  then V is iBCS by Theorem 2.3.75. So suppose  $V \in P^{V}(iBCK)$ . Since  $\mathbf{P}^{V}(BCK)$  is a sublattice of  $\Lambda^{V}(PBCK)$  (by [38, Theorem 11]) we must have that V is a cover of iBCK in  $\mathbf{P}^{V}(BCK)$ . Thus V is either  $\mathbf{V}(\mathbf{L}_{3})$  or  $\mathbf{V}(\mathbf{H}_{3})$ .

## Chapter 3

# Applications to Universal Algebra and Algebraic Logic

In this chapter we consider applications of the theory of pre-BCK-algebras to universal algebra and algebraic logic. In particular, we study three classes of algebras arising naturally in both universal algebra and algebraic logic, namely: subtractive varieties with EDPI; binary (and dual binary) discriminator varieties; and (pointed ternary discriminator varieties *qua*) pre-BCK-algebras structurally enriched with band operations. Our motivation for studying subtractive varieties with EDPI stems from Theorem 2.2.20 and the fundamental role played by MINI-algebras in such varieties (recall the remarks of §1.1.1). Our study of binary discriminator varieties is stimulated by Example 2.3.11 and Example 2.3.13; recall these results collectively assert that implicative BCS-algebras arise naturally as polynomial reducts of members of binary and pointed ternary discriminator varieties. Our investigation of pre-BCK-algebras structurally enriched with band operations is prompted by Theorem 1.4.39 and Corollary 1.4.40, which show that, in a sense, the study of pointed ternary discriminator varieties reduces to the study of skew Boolean  $\cap$ -algebras.

In investigating each of the three classes described above, our main aim is to establish the role played (if any), at a structural level, by the theory of pre-BCK-algebras. In particular, our object (with occasional diversions) is to ascertain the extent to which the ideal theory and/or congruence structure of subtractive varieties with EDPI, binary discriminator varieties, and pointed ternary discriminator varieties, may be reduced to a study of the ideal theory and/or congruence structure of the varieties of MINI-algebras and implicative BCS-algebras (possibly structurally enriched with additional operations) respectively.

### 3.1 Subtractive Varieties with EDPI

By the remarks of Example 1.7.11, the variety of MINI-algebras is a natural generalisation of the variety of Hilbert algebras to the subtractive but not point-regular case, and hence is a natural example of a subtractive variety with EDPI. Recently Agliano and Ursini have shown [10, Corollary 3.8] that the variety of MINI-algebras is in fact a paradigm for subtractive varieties with EDPI in the sense that a variety V is subtractive with EDPI iff every member of V has a MINI-algebra polynomial reduct satisfying a certain weak 'ideal compatibility property'. Insofar as the results of §2.2.16 show that the variety of positive implicative pre-BCK-algebras is both a natural generalisation of the variety of positive implicative BCK-algebras to the subtractive but not point regular case and a natural example of a subtractive varieties with EDPI, the central role played by MINI-algebras in the theory of subtractive varieties with EDPI calls for a study of the role played by positive implicative pre-BCK-algebras in the theory of subtractive pre-BCK-algebras in the theory of subtractive varieties with EDPI calls for a study of the role played by positive implicative pre-BCK-algebras in the theory of subtractive varieties with EDPI.

In §3.1.1 positive implicative pre-BCK-algebras, MINI-algebras and subtractive varieties with EDPI are studied. It is shown that the variety of positive implicative pre-BCK-algebras is termwise definitionally equivalent to (in fact, is dually isomorphic to) Agliano and Ursini's variety of MINI-algebras. We show that a variety V is subtractive with EDPI iff every algebra  $\mathbf{A} \in V$  has a MINI-algebra polynomial reduct whose ideals coincide with those of  $\mathbf{A}$ , sharpening the result of Agliano and Ursini alluded to above. A representation theorem for weakly congruence orderable subtractive varieties with EDPI is also proved: for a suitable notion of weakly compatible operation, a variety is weakly congruence orderable and subtractive with EDPI iff it is termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations.

Subtractive WBSO varieties are studied in §3.1.22. Such varieties arise naturally in algebraic logic and are subtractive (by definition) and have EDPI (since they have EDPC). The subtractive WBSO varieties are characterised: they are precisely the subtractive, strongly point regular varieties with EDPC. We also show that any such variety V is distinguished as a WBSO variety by the presence of a weak relative pseudocomplementation  $\rightarrow$  such that the polynomial reduct  $\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$  is a MINI-algebra for any  $\mathbf{A} \in V$ . An interesting example of a subtractive WBSO variety is the variety N of Nelson algebras, which arises in the first instance from the algebraisation of constructive logic with strong negation. It is shown that N has a commutative (but not regular) TD term and is congruence permutable. An explicit QD term for N is also given. In consequence we infer that, for any variety V of Nelson algebras, the class of implicative subreducts of V is a subvariety of the variety of MINI-algebras. The results answer a question of Blok and Pigozzi.

3.1.1. Positive Implicative Pre-BCK Algebras and MINI-Algebras. Recall from Example 1.7.11 that a MINI-algebra is an algebra  $\langle A; \rightarrow, 1 \rangle$  of type  $\langle 2, 0 \rangle$  satisfying the following identities:

 $x \to \mathbf{1} \approx \mathbf{1} \tag{1.45}$ 

 $\mathbf{1} \to x \approx x \tag{1.46}$ 

$$(x \to (y \to z)) \to ((x \to y) \to (x \to z)) \approx \mathbf{1}$$
 (1.47)

$$x \to (y \to x) \approx \mathbf{1} \tag{1.48}$$

while a Hilbert algebra is a MINI-algebra satisfying the quasi-identity:

$$x \to y \approx \mathbf{1} \& y \to x \approx \mathbf{1} \supset x \approx y. \tag{1.49}$$

By definition, the class MINI of all MINI-algebras is a variety. By the remarks of §1.6.13, the quasivariety HI of all Hilbert algebras is also a variety, which is termwise definitionally equivalent to (in fact, is dually isomorphic to) the variety of positive implicative BCK-algebras.

By the preceding remarks, Agliano and Ursini's variety of MINI-algebras is a natural generalisation of the variety of Hilbert algebras to the subtractive but not point regular case, in the sense that HI is precisely the subquasivariety of MINI axiomatised by the quasi-identity (1.49). On the other hand, Theorem 2.2.17 shows the variety of positive implicative pre-BCK-algebras is a natural generalisation of the variety of positive implicative BCK-algebras to the subtractive but not point regular case, in the sense that pPBCK is precisely the subquasivariety of pBCK axiomatised by the quasi-identity (2.5). Hence pPBCK stands in relation to pBCK as MINI stands in relation to HI. Because the varieties HI and pBCK are dually isomorphic, this remark suggests that the varieties MINI and pPBCK may themselves be dually isomorphic.

To clarify the relationship between the variety of positive implicative pre-BCK-algebras and the variety of MINI-algebras, let the class  $MINI^{D}$  of dual MINI-algebras be the variety of algebras with language  $\langle \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$ axiomatised by the following identities:

$$\mathbf{0} - \mathbf{x} \approx \mathbf{0} \tag{3.1}$$

$$x - \mathbf{0} \approx x \tag{3.2}$$

$$\left(\left(x - y\right) - \left(z - y\right)\right) - \left(\left(x - z\right) - y\right) \approx 0 \tag{3.3}$$

$$(x - y) - x \approx 0. \tag{3.4}$$

Let  $T_{\mathcal{L}}(X)$  denote the term algebra of type  $\mathcal{L}$  over X, where  $\mathcal{L}$  is the language of MINI-algebras. Also, let  $T_{\mathcal{L}^D}(X)$  denote the term algebra of type  $\mathcal{L}^D$ over X, where  $\mathcal{L}^D$  is the language of dual MINI-algebras. Consider the maps  $\xi: T_{\mathcal{L}}(X) \to T_{\mathcal{L}^D}(X)$  and  $\eta: T_{\mathcal{L}^D}(X) \to T_{\mathcal{L}}(X)$  defined respectively by:

| $\xi(1) := 0$                     |   |
|-----------------------------------|---|
| $\xi(x) := x$                     | $x\in \mathbf{X}$                           |
| $\xi(p \to q) := \xi(q) - \xi(p)$ | $p,q\in \mathrm{T}_\mathcal{L}(\mathbf{X})$ |

and:

$$\eta(\mathbf{0}) := \mathbf{1}$$
  
 $\eta(x) := x$   $x \in \mathbf{X}$   
 $\eta(r - s) := \eta(s) \rightarrow \eta(r)$   $r, s \in \mathcal{T}_{\mathcal{L}^D}(\mathbf{X}).$ 

(The maps  $\xi$  and  $\eta$  so defined should not be confused with the similar maps of §2.1.33 in the prequel.) Because of the axiomatisation of MINI by (1.45)– (1.48) and the axiomatisation of MINI<sup>D</sup> by (3.1)–(3.4), the proof of the following lemma is trivial and so is omitted.

**Lemma 3.1.2.** For  $p, q \in T_{\mathcal{L}}(X)$  and  $r, s \in T_{\mathcal{L}^{D}}(X)$  the following assertions hold:

- 1. If MINI  $\models p \approx q$  then MINI<sup>D</sup>  $\models \xi(p) \approx \xi(q)$ ;
- 2. If  $MINI^D \models r \approx s$  then  $MINI \models \eta(r) \approx \eta(s)$ .

Moreover,  $\eta \circ \xi = \omega_{T_{\mathcal{L}}(\mathbf{X})}$  and  $\xi \circ \eta = \omega_{T_{\mathcal{L}}(\mathbf{X})}$ .

By the preceding lemma, the variety of MINI-algebras is termwise definitionally equivalent to (in fact, is dually isomorphic to) the variety of dual MINIalgebras. When coupled with the following proposition, this result yields Theorem 3.1.4 below, which confirms that the varieties of positive implicative pre-BCK-algebras and MINI-algebras are indeed dually isomorphic.

**Proposition 3.1.3.** An algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is a dual MINI-algebra iff it is a positive implicative pre-BCK-algebra. Thus the variety of dual MINI-algebras coincides with the variety of positive implicative pre-BCK-algebras.

*Proof.* ( $\Rightarrow$ ) Let A be a dual MINI-algebra and let  $a, b \in A$ . Throughout the proof we denote  $\dot{-}^{A}$  by juxtaposition for ease of notation. By definition we have that  $A \models (2.4), (2.6)$ , so to see A is a pre-BCK-algebra we have only to show (by Proposition 2.1.11) that  $A \models (2.7), (2.2)$ . For (2.7), put  $\alpha := (ab)(cb), \beta := ac$  and  $\gamma := (ac)b$ . We have:

$$0 = ((\alpha\beta)(\gamma\beta))((\alpha\gamma)\beta)$$
 by (3.3)

| 3.1. Subtractive Varieties with EDPI                                       | 219      |
|--|----------|
| $= \big((\alpha\beta)(\gamma\beta)\big)\big((((ab)(cb))((ac)b))\beta\big)$ |          |
| $= ((lphaeta)(\gammaeta))(0eta)$   | by (3.3) |
| $= \left( \left( - \rho \right) \left( - \rho \right) \right) $            | h (2 1)  |

$$= ((\alpha\beta)(\gamma\beta))^{0} \qquad \qquad \text{by (3.1)}$$
$$= (\alpha\beta)(\gamma\beta) \qquad \qquad \text{by (3.2)}$$

$$= \big(((ab)(cb))(ac)\big)\big(((ac)b)(ac)\big).$$

Put  $\alpha := ac$  and  $\beta := b$ . We have:

$$\begin{aligned} (((ab)(cb))(ac))(((ac)b)(ac)) \\ &= (((ab)(cb))(ac))((\alpha\beta)\alpha) \\ &= (((ab)(cb))(ac))0 \\ &= ((ab)(cb))(ac) \end{aligned} \qquad by (3.4) \\ &= ((ab)(cb))(ac) \end{aligned}$$

For (2.2), put  $\alpha := a(ab), \beta := b$  and  $\gamma := b(ab)$ . We have:

$$0 = ((\alpha\beta)(\gamma\beta))(\alpha\gamma) \qquad \text{by } (2.7)$$
$$= ((\alpha\beta)(\gamma\beta))((\alpha(\alphab))(b(\alphab)))$$

$$= ((\alpha\beta)(\gamma\beta))((a(ab))(b(ab))) = ((\alpha\beta)(\gamma\beta))(((a(ab))(b(ab)))0)$$
by (3.2)  
= (((a(ab))b)((b(ab))b))((((a(ab))(b(ab)))0).

Put  $\alpha := ab$  and  $\beta := 0$ . We have:

$$(((a(ab))b)((b(ab))b)) ((((a(ab))(b(ab)))0) = (((a(ab))b)((b(ab))b)) ((((a(ab))(b(ab)))(((\alpha\beta)\alpha)) = ((((a(ab))b)((b(ab))b)) ((((a(ab))(b(ab)))(((ab)0)(ab))) by (3.4) = (((a(ab))b)((b(ab))b)) ((((a(ab))(b(ab)))((ab)(ab))) by (3.2).$$

Put  $\alpha := a, \beta := ab$  and  $\gamma := b$ . We have:

$$\big(((a(ab))b)((b(ab))b)\big)\big(((a(ab))(b(ab)))((ab)(ab))\big)$$

$$= (((a(ab))b)((b(ab))b))(((\alpha\beta)(\gamma\beta))((\alpha\gamma)\beta))$$
$$= (((a(ab))b)((b(ab))b))0 \qquad by (3.3)$$

= ((a(ab))b)((b(ab))b)by (3.2).

Put  $\alpha := b$  and  $\beta := ab$ . We have:

$$((a(ab))b)((b(ab))b)$$

$$= ((a(ab))b)((\alpha\beta)\alpha)$$

$$= ((a(ab))b)0$$

$$= (a(ab))b$$
by (3.4)
by (3.2)

Hence  $\mathbf{A} \models (2.2), (2.7)$  and so is a pre-BCK-algebra. To see  $\mathbf{A}$  is positive implicative it is sufficient to note:

$$0 = ((ab)(bb))((ab)b) by (3.3) = ((ab)((b0)b))((ab)b) by (3.2) = ((ab)0)((ab)b) by (3.4) = (ab)((ab)b) by (3.2).$$

( $\Leftarrow$ ) Let A be a positive implicative pre-BCK-algebra. By definition we have that  $\mathbf{A} \models (3.1), (3.2)$ . Moreover  $\mathbf{A} \models (3.3)$  by Lemma 2.2.18; since  $\mathbf{A} \models (3.4)$  by Lemma 2.1.12(3) we have that A is a dual MINI-algebra.

**Theorem 3.1.4.** The variety of MINI-algebras is termwise definitionally equivalent to the variety of positive implicative pre-BCK algebras. Given a positive implicative pre-BCK-algebra  $\langle A; \div, 0 \rangle$ , MINI-algebra operations are defined on A by:

```
a \rightarrow b := b - a
1 := 0
```

for any  $a, b \in A$ . Conversely, given a MINI-algebra  $\langle A; \rightarrow, 1 \rangle$ , positive im-

plicative pre-BCK-algebra operations are defined on A by:

$$a \stackrel{\cdot}{-} b := b \rightarrow a$$
  
 $0 := 1$ 

#### for any $a, b \in A$ .

Because the variety of positive implicative pre-BCK-algebras has EDPI witness x - y (by Theorem 2.2.20), from Theorem 3.1.4 it follows immediately that the variety of MINI-algebras is subtractive with EDPI witness  $y \rightarrow x$  (compare this remark with Example 1.7.11). For the sake of developments in the sequel we find it convenient to present this result here explicitly as a corollary.

**Corollary 3.1.5.** (cf. Example 1.7.11) The variety of MINI-algebras is subtractive with EDPI. Moreover, the binary term  $y \rightarrow x$  witnesses both subtractivity and EDPI for MINI in the sense of Theorem 1.7.9. That is, for any MINI-algebra A and  $a, b \in A$ ,

 $a \rightarrow^{\mathbf{A}} a = 1$  $1 \rightarrow^{\mathbf{A}} a = a$ 

 $a \in \langle b \rangle_{\mathbf{A}}$  iff  $b \to^{\mathbf{A}} a = 1$ .

The variety of MINI-algebras is more than just a natural example of a subtractive variety with EDPI; by Agliano and Ursini [11, Corollary 3.8] it is a paradigm for such varieties in that a variety V is subtractive with EDPI iff every  $\mathbf{A} \in V$  has a MINI-algebra polynomial reduct for which any ideal term  $t(\vec{x}, y) \in \mathrm{IT}_{V}(y)$  is compatible with V in the sense that for any  $a, \vec{b} \in A$  [11, p. 375],

 $a \rightarrow^{\mathbf{A}} t^{\mathbf{A}}(\vec{b}, a) = 1.$ 

The following theorem, whose proof is included for the sake of completeness, presents a variant on this result.

**Theorem 3.1.6.** (cf. [11, Corollary 3.8]) For a variety  $\vee$  with 1, the following are equivalent for a binary term  $x \rightarrow y$  of  $\vee$ :

- 1. The term  $y \rightarrow x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9;
- 2. For any  $A \in V$ , the polynomial reduct  $\langle A; \rightarrow^{A}, 1 \rangle$  is a MINI-algebra, and either one of the following conditions is satisfied:
  - (a) Any ideal term  $t(\vec{x}, y) \in IT_{V}(y)$  is compatible with V;
  - (b) For any  $a \in A$ ,  $\langle a \rangle_{\mathbf{A}} = \langle a \rangle_{\langle A; \to \mathbf{A}, \mathbf{1} \rangle}$ .

*Proof.* Let V be as stated and let  $x \to y$  be a binary term of V.

(1)  $\Rightarrow$  (2)(a) Suppose  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9. Then for any  $\mathbf{A} \in V$ ,  $\langle A; \to^{\mathbf{A}}, 1 \rangle / \approx_{\mathbf{A}}$  is a Hilbert algebra dually isomorphic with  $\langle \operatorname{PI}(\mathbf{A}); *, \langle 1 \rangle_{\mathbf{A}} \rangle$  by Theorem 1.7.15(1). Throughout the remainder of the proof to simplify notation we write  $\overline{a}$  for the equivalence class  $[a]_{\approx_{\mathbf{A}}}$  in  $A/\approx_{\mathbf{A}}$  containing  $a \in A$ . As (1.45)-(1.47) are identities in the language  $\langle \to, 1 \rangle$  that hold in the variety of Hilbert algebras, for any  $\overline{a}, \overline{b}, \overline{c} \in A/\approx_{\mathbf{A}}$  with  $a, b, c \in A$  we have:

$$\begin{split} \overline{a} \to^{\mathbf{A}/\approx_{\mathbf{A}}} \overline{1} &= \overline{1} \\ \overline{1} \to^{\mathbf{A}/\approx_{\mathbf{A}}} \overline{a} &= \overline{a} \\ \left(\overline{a} \to^{\mathbf{A}/\approx_{\mathbf{A}}} \left(\overline{b} \to^{\mathbf{A}/\approx_{\mathbf{A}}} \overline{c}\right)\right) \to^{\mathbf{A}/\approx_{\mathbf{A}}} \left(\left(\overline{a} \to^{\mathbf{A}/\approx_{\mathbf{A}}} \overline{b}\right) \to^{\mathbf{A}/\approx_{\mathbf{A}}} \left(\overline{a} \to^{\mathbf{A}/\approx_{\mathbf{A}}} \overline{c}\right)\right) = \overline{1} \\ \overline{a} \to^{\mathbf{A}/\approx_{\mathbf{A}}} \left(\overline{b} \to^{\mathbf{A}/\approx_{\mathbf{A}}} \overline{a}\right) = \overline{1}. \end{split}$$

Because  $\overline{1} = \{1\}$ , we infer:

$$a \rightarrow^{\mathbf{A}} 1 = 1$$
  

$$1 \rightarrow^{\mathbf{A}} a = a$$
  

$$(a \rightarrow^{\mathbf{A}} (b \rightarrow^{\mathbf{A}} c)) \rightarrow^{\mathbf{A}} ((a \rightarrow^{\mathbf{A}} b) \rightarrow^{\mathbf{A}} (a \rightarrow^{\mathbf{A}} c)) = 1$$
  

$$a \rightarrow^{\mathbf{A}} (b \rightarrow^{\mathbf{A}} a) = 1$$

in A, whence  $\langle A; \to^{A}, 1 \rangle$  is a MINI-algebra. To see any ideal term  $t(\vec{x}, y) \in IT_{V}(y)$  is compatible with V let  $t(\vec{x}, y) \in IT_{V}(y)$  and suppose  $\vec{b}, a \in A$ . Then

 $t^{\mathbf{A}}(\vec{b}, a) \in \langle a \rangle_{\mathbf{A}}$  by the description of  $\langle a \rangle_{\mathbf{A}}$  given in §1.7.1. Since  $y \to x$  witnesses EDPI for V we have that  $a \to t^{\mathbf{A}}(\vec{b}, a) = 1$  as required.

 $(2)(a) \Rightarrow (2)(b)$  Suppose that for any algebra  $\mathbf{A} \in \mathsf{V}$ , the polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  is a MINI-algebra and that any ideal term  $t(\vec{x}, y) \in \mathrm{IT}_{\mathsf{V}}(y)$  is compatible with  $\mathsf{V}$ . For any  $a \in A$ , trivially  $\langle a \rangle_{\langle A; \to^{\mathbf{A}}, 1 \rangle} \subseteq \langle a \rangle_{\mathbf{A}}$ . For the opposite inclusion, let  $c \in \langle a \rangle_{\mathbf{A}}$ . Then  $c = t^{\mathbf{A}}(\vec{b}, a)$  where  $t(\vec{x}, y)$  is an ideal term in y and  $\vec{b}, a \in A$ , just because of the description of  $\langle a \rangle_{\mathbf{A}}$  given in §1.7.1. Since  $t(\vec{x}, y)$  is compatible with  $\mathsf{V}$  we have  $a \to t^{\mathbf{A}}(\vec{b}, a) = 1$ , whence  $t^{\mathbf{A}}(\vec{b}, a) \in \langle a \rangle_{\langle A; \to^{\mathbf{A}, 1 \rangle}}$  by Corollary 3.1.5. Thus  $c \in \langle a \rangle_{\langle A; \to^{\mathbf{A}, 1 \rangle}}$  and  $\langle a \rangle_{\mathbf{A}} \subseteq \langle a \rangle_{\langle A; \to^{\mathbf{A}, 1 \rangle}}$ .

 $(2)(b) \Rightarrow (1)$  Suppose that for any  $A \in V$  the polynomial reduct  $\langle A; \to^{A}, 1 \rangle$  is a MINI-algebra and  $\langle a \rangle_{A} = \langle a \rangle_{\langle A; \to^{A}, 1 \rangle}$  for any  $a \in A$ . Since  $y \to x$  witnesses both subtractivity and EDPI for  $\langle A; \to^{A}, 1 \rangle$  in the sense of Theorem 1.7.9 (by Corollary 3.1.5),  $y \to x$  witnesses both subtractivity and EDPI for A in this sense also (by hypothesis). Thus  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9, and the proof is complete.

Let V be a subtractive variety let  $A \in V$ . Since any V-ideal of A is a directed union of principal V-ideals, from Theorem 3.1.6 we may infer:

Corollary 3.1.7. For a variety V with 1 and a binary term  $x \to y$  of V, the term  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9 iff every algebra  $A \in V$  has a MINI-algebra polynomial reduct  $\langle A; \to^A, 1 \rangle$  whose MINI-ideals coincide with the V-ideals of A.

For a variety V with 1 and a binary term  $x \to y$  of V, the preceding corollary cannot be strengthened to the assertion that  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9 iff every  $\mathbf{A} \in V$  has a MINI-algebra polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  whose congruences coincide with those of A. If every  $\mathbf{A} \in V$  has a MINI-algebra polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$ whose congruences coincide with those of A, then certainly  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9. However, as an immediate consequence of the following proposition we have that the converse does not hold.

**Proposition 3.1.8.** The binary term  $x \setminus y$  witnesses both subtractivity and EDPI for the variety IhSBA of left handed skew Boolean algebras in the sense of Theorem 1.7.9. Hence the IhSBA-ideals of any left handed skew Boolean algebra A coincide with the MINI-ideals of its canonical MINI-algebra polynomial reduct  $\langle A; \rightarrow^A, \mathbf{1}^A \rangle$ , where  $a \rightarrow^A b := b \setminus a$  for any  $a, b \in A$  and  $\mathbf{1}^A := 0$ . Nonetheless, there exists a left handed skew Boolean algebra S that has no MINI-algebra polynomial reduct whose congruences coincide with those of S.

*Proof.* To establish the first two assertions of the theorem, it is sufficient to show that for any skew Boolean algebra A and  $a \in A$ ,

(i) The reduct  $\langle A; \rangle, 0 \rangle$  is an implicative BCS-algebra;

(ii) 
$$\langle a \rangle_{\mathbf{A}} = \langle a \rangle_{\langle A; \backslash, 0 \rangle}$$
.

For (i), an easy inspection of the subdirectly irreducible skew Boolean algebras  $2^p$ ,  $3^p_L$  and  $3^p_R$  shows that the reduct  $\langle A; \rangle, 0 \rangle$  of any skew Boolean algebra A is an implicative BCS-algebra.

For (ii), let A be a skew Boolean algebra. From remarks due to Leech [150, Section 4.6] it is known that a non-empty subset  $\{0\} \subseteq I \subseteq A$  is an SBA-ideal iff the following conditions are satisfied [19, Definition 3.3]:

$$a, b \in I \text{ implies } a \lor b \in I;$$
 and (3.5)

 $a \in I, b \in A \text{ implies } b \land a \land b \in I.$  (3.6)

Notice (3.6) above is equivalent to:

$$a \in I, b \preceq_{\mathcal{D}} a$$
, implies  $b \in I$ . (3.7)

For assume (3.6) and let  $a \in I$  and  $b \preceq_{\mathcal{D}} a$ . Since  $b \preceq_{\mathcal{D}} a$ ,  $b \wedge a \wedge b = b$ . By (3.6),  $b \in I$ . Conversely, assume (3.7) and let  $a \in I$  and  $b \in A$ . Since  $b \wedge a \wedge b \preceq_{\mathcal{D}} a$ , by (3.7) we have  $b \wedge a \wedge b \in I$ .

Let  $a \in A$ . From (3.5) and (3.7) it follows easily that  $\langle a \rangle_{\mathbf{A}} = \{b : b \preceq_{\mathcal{D}} a\}$ . On the other hand,  $\langle a \rangle_{\langle A; \backslash, 0 \rangle} = \{b : b \preceq^{\langle A; \backslash, 0 \rangle} a\}$  by Theorem 2.2.20(4), Proposition 2.2.30 and Proposition 2.3.5. Therefore to see  $\langle a \rangle_{\mathbf{A}} = \langle a \rangle_{\langle A; \backslash, 0 \rangle}$  it is sufficient to show  $c \preceq_{\mathcal{D}} d$  iff  $c \preceq^{\langle A; \backslash, 0 \rangle} d$  for any  $c, d \in A$ . So let  $c, d \in A$ .

Suppose  $c \leq_{\mathcal{D}} d$ . Then  $c \wedge d \wedge c = c$ , so  $0 = (c \setminus d) \wedge (c \wedge d \wedge c) = (c \setminus d) \wedge c = c \setminus d$  by (1.25) and (1.28). Thus  $c \leq^{\langle A; \setminus, 0 \rangle} d$ . Conversely, suppose  $c \leq^{\langle A; \setminus, 0 \rangle} d$ . Then  $c \setminus d = 0$ , so  $c = (c \setminus d) \vee (c \wedge d \wedge c) = 0 \vee (c \wedge d \wedge c) = c \wedge d \wedge c$  by (1.23). Thus  $c \leq_{\mathcal{D}} d$  and so  $c \leq_{\mathcal{D}} d$  iff  $c \leq^{\langle A; \setminus, 0 \rangle} d$ . Hence  $\langle a \rangle_{\mathbf{A}} = \langle a \rangle_{\langle A; \setminus, 0 \rangle}$ .

To establish the remaining assertion of the theorem, it is sufficient to show that for any left handed skew Boolean algebra A,

(iii) The only polynomial reduct of A term equivalent to a MINI-algebra is  $\langle A; \rangle, 0 \rangle$ ;

and also that:

(iv) There exists a left handed skew Boolean algebra S with Con  $\langle S; \backslash, 0 \rangle \neq$  Con S.

For (iii), let  $\mathbf{F}(\bar{x}, \bar{y})$  denote the lhSBA-free algebra on two free generators  $\bar{x}, \bar{y}$ . From Example 1.4.23, the remarks of §1.4.24, Example 2.3.26 and (i) we see at once that  $\mathbf{F}(\bar{x}, \bar{y})$  has  $\mathcal{D}$ -equivalence classes:

$$\{\overline{\mathbf{0}}\}, \quad \{\overline{x}, (\overline{x} \vee \overline{y}) \wedge \overline{x}\}, \quad \{\overline{y}, (\overline{y} \vee \overline{x}) \wedge \overline{y}\}$$
$$\{\overline{x} \setminus \overline{y}\}, \quad \{\overline{y} \setminus \overline{x}^{\uparrow}, \quad \{\overline{x} \wedge \overline{y}, \overline{y} \wedge \overline{x}\}, \quad \{\overline{x} \vee \overline{y}, \overline{y} \vee \overline{x}\}$$

with:

| $\overline{x} \wedge \overline{y} \leq \overline{x},  (\overline{y} \lor \overline{x}) \land \overline{y}$     | $\overline{y} \wedge \overline{x} \leq \overline{y},  (\overline{x} \vee \overline{y}) \wedge \overline{x}$     |
|--|---|
| $\overline{x},(\overline{y}ee\overline{x})\wedge\overline{y}\leq\overline{x}ee\overline{y}$                    | $\overline{y},(\overline{x}ee\overline{y})\wedge\overline{x}\leq\overline{y}ee\overline{x}$                     |
| $\overline{x} \setminus \overline{y} \leq \overline{x},  (\overline{x} \vee \overline{y}) \wedge \overline{x}$ | $\overline{y} \setminus \overline{x} \leq \overline{y},  (\overline{y} \vee \overline{x}) \wedge \overline{y}.$ |

Routine computations (by inspection of  $2^p$  and  $3^p_L$ ) show that  $\mathbf{F}(\bar{x}, \bar{y})$  has exactly one other  $\mathcal{D}$ -class distinct from those listed above, namely:

 $\{(\overline{x} \lor \overline{y}) \setminus (\overline{y} \land \overline{x})\}$ 

ŀ

with:

$$\overline{x} \setminus \overline{y}, \, \overline{y} \setminus \overline{x} \leq (\overline{x} \lor \overline{y}) \setminus (\overline{x} \land \overline{y}) \leq \overline{x} \lor \overline{y}, \, \overline{y} \lor \overline{x}$$

225



Figure 3.1. The lhSBA-free algebra on two free generators  $\overline{x}, \overline{y}$ .

whence  $\mathbf{F}(\overline{x}, \overline{y})$  has the Hasse diagram of Figure 3.1 (where for notational purposes the free generators  $\overline{x}$ ,  $\overline{y}$  are denoted in the figure simply by x, y respectively; like remarks apply to products of  $\overline{x}$ ,  $\overline{y}$ ). Conversely, the diagram of Figure 3.1 completely determines  $\mathbf{F}(\overline{x}, \overline{y})$ , just because  $\mathbf{F}(\overline{x}, \overline{y})$  is left handed. From this characterisation of  $\mathbf{F}(\overline{x}, \overline{y})$  it is now easy to see (by inspection of  $2^p$ and  $3^p_L$ ) that there exists no term function  $\rightarrow^{\mathbf{A}}$  (other than that induced by the term  $y \setminus x$ ) definable in terms of the fundamental left handed skew Boolean algebra operations  $\wedge, \vee, \setminus$  such that for any left handed skew Boolean algebra A, the polynomial reduct  $\langle A; \rightarrow^{\mathbf{A}}, 0 \rangle$  is a MINI-algebra. Hence the only polynomial reduct of A that is term equivalent to a MINI-algebra is  $\langle A; \setminus, 0 \rangle$ .

For (iv), let S denote the 6-element algebra defined by the following operation

tables:

| ∧ <sup>s</sup> _ | 0 | a | b | С | d | e | $\vee^{\mathbf{s}}$ | 0 | a | b | с | d | e | $\setminus^{\mathbf{s}}$ | 0 | a | b | С | d | e |
|------------------|---|---|---|---|---|---|---------------------|---|---|---|---|---|---|--------------------------|---|---|---|---|---|---|
| 0                | 0 | 0 | 0 | 0 | 0 | 0 | 0                   | 0 | a | b | С | d | e | 0                        | 0 | 0 | 0 | 0 | 0 | 0 |
| a                | 0 | a | 0 | a | 0 | a | a                   | a | a | С | с | е | е | a                        | a | 0 | a | 0 | a | 0 |
| b                | 0 | 0 | b | b | b | b | b                   | b | C | b | с | d | е | b                        | b | b | 0 | 0 | 0 | 0 |
| с                | 0 | a | b | с | b | С | С                   | С | с | с | С | е | е | с                        | c | b | a | 0 | a | 0 |
| d                | 0 | 0 | d | d | d | d | d                   | d | e | b | С | d | е | d                        | d | d | 0 | 0 | 0 | 0 |
| e                | 0 | a | d | e | d | е | е                   | e | e | с | С | е | е | e                        | е | d | a | 0 | a | 0 |

It is readily verified that **S** is a left handed skew Boolean algebra: see Figure 3.2(a). To complete the proof it is sufficient by (iii) to show Con  $\langle S; \backslash, 0 \rangle \neq$  Con  $\langle S; \wedge, \vee, 0 \rangle$ , just because Con **S** = Con  $\langle S; \wedge, \vee, 0 \rangle$  by Proposition 1.4.27. To this end, let  $\theta$  be the equivalence relation on  $S \times S$  induced by the partition  $\{\{0\}, \{a\}, \{b, d\}, \{c\}, \{e\}\}\}$ . It is tedious but straightforward to check that  $\theta$  is a congruence relation on  $\langle S; \backslash, 0 \rangle$  (for a complete description of the congruence structure of  $\langle S; \backslash, 0 \rangle$ , see Figure 3.2(c); notice that for ease of notation, congruences on  $\langle S; \backslash, 0 \rangle$  are represented in the figure by their corresponding partitions (with all parentheses dropped)). However,  $\theta$  is not a congruence on  $\langle S; \wedge, \vee, 0 \rangle$ . Indeed, suppose  $\theta \in \text{Con } \langle S; \wedge, \vee, 0 \rangle$ . Because  $e \equiv_{\theta} e$  and  $b \equiv_{\theta} d$ , we must have that  $e \vee b \equiv_{\theta} e \vee d$ , which implies  $c \equiv_{\theta} e$ , a contradiction. Hence  $\theta \notin \text{Con} \langle S; \wedge, \vee, 0 \rangle$  (for a complete description of the congruence structure of  $\langle S; \wedge, \vee, 0 \rangle$ .

**Corollary 3.1.9.** Let  $\vee$  be a variety with 1 and let  $x \to y$  be a binary term of  $\vee$  such that  $y \to x$  witnesses both subtractivity and EDPI for  $\vee$  in the sense of Theorem 1.7.9. In general, the congruences on the canonical MINI-algebra polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  of  $\mathbf{A} \in \vee$  need not coincide with those on  $\mathbf{A}$ .

Let A be a set and let f be an n-ary operation on A. The slice  $f_i(a)$ ,  $a \in A$ , is the unary operation obtained from f upon defining:

$$f_i(a) := f(b_1, \ldots, b_{i-1}, a, b_{i+1}, \ldots, b_n)$$

227



Figure 3.2. (a) The left handed skew Boolean algebra S of Proposition 3.1.8; (b) Con  $\langle S; \land, \lor, 0 \rangle$ ; (c) Con  $\langle S; \backslash, 0 \rangle$ .

for fixed  $b_1, \ldots, b_{i-1}, b_{i+1}, \ldots, b_n \in A$ . In other words, the slice  $\tilde{f}_i(a)$  is the unary operation obtained from f by fixing all but one of its arguments. Given an algebra  $\mathbf{A}$ , an *n*-ary operation f of  $\mathbf{A}$  is said to be *compatible* with a congruence  $\theta$  on  $\mathbf{A}$  if  $\theta$  has the substitution property with respect to f; that is, if:

$$a \equiv b \pmod{\theta}$$
 implies  $\widetilde{f}_i(a) \equiv \widetilde{f}_i(b) \pmod{\theta}$ 

for all  $a, b \in A$  and slices  $\tilde{f}_i(a), \tilde{f}_i(b), 1 \leq i \leq n$ . An *n*-ary operation f of A is said to be *compatible* if it is compatible with every congruence on A.

A MINI-algebra [Hilbert algebra] with compatible operations is an algebra  $A := \langle A; \rightarrow, 1, f_j \rangle_{j \in J}$  of type  $\langle 2, 0, \ldots \rangle$  such that: (i) the reduct  $\langle A; \rightarrow, 1 \rangle$  is a MINIalgebra [Hilbert algebra]; and (ii) each additional operation  $f_j$  is compatible with  $\langle A; \rightarrow, 1 \rangle$ . Clearly an algebra  $A := \langle A; \rightarrow, 1, f_j \rangle_{j \in J}$  of type  $\langle 2, 0, \ldots \rangle$ is a MINI-algebra [Hilbert algebra] with compatible operations iff Con A =Con  $\langle A; \rightarrow, 1 \rangle$ . A version of the following lemma is presented without proof in Agliano [7, Section 3].

**Lemma 3.1.10.** (cf. [7, Section 3, p. 9]) Let A be an algebra with 1, say  $\langle A; 1, f_j \rangle_{j \in J}$ , and let  $x \to y$  be a binary term of A such that the polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  is a Hilbert algebra. The following are equivalent:

- 1. A is term equivalent to a Hilbert algebra with compatible operations  $\langle A; \rightarrow^{\mathbf{A}}, 1, f_j \rangle_{j \in J};$
- 2. For any n-ary fundamental operation  $f := f_j$  of  $A_j$ ,

$$(a \rightarrow^{\mathbf{A}} b) \rightarrow^{\mathbf{A}} ((b \rightarrow^{\mathbf{A}} a) \rightarrow (\widetilde{f}_{i}(a) \rightarrow^{\mathbf{A}} \widetilde{f}_{i}(b))) = 1$$

for all  $a, b \in A$  and slices  $\tilde{f}_i(a), \tilde{f}_i(b), 1 \leq i \leq n$ .

*Proof.* Let A be a Hilbert algebra and let  $a, b, c, d \in A$ . By Theorem 1.6.15(2),

$$c \equiv d \pmod{\Theta^{\mathbf{A}}(a,b)} \quad \text{iff} \quad (a \to b) \to ((b \to a) \to (c \to d)) = (a \to b) \to ((b \to a) \to (d \to c));$$

the lemma follows directly from this description of the principal congruences.

A variety V is said to be a variety of MINI-algebras [Hilbert algebras] with compatible operations if every member of V is a MINI-algebra [Hilbert algebra] with compatible operations. Given Corollary 3.1.9 and the remarks immediately preceding Proposition 3.1.8, it is natural to ask if varieties of MINI-algebras with compatible operations admit a relevant structure theorem. For the special case of varieties of Hilbert algebras with compatible operations, a positive answer to this question has been obtained by Agliano in [7, Section 3]. For the sake of both completeness and developments in the sequel, we reproduce Agliano's result in Theorem 3.1.11 below.

**Theorem 3.1.11.** [7, Theorem 3.4] For a variety  $\vee$  with 1 and a binary term  $x \rightarrow y$  of  $\vee$ , the following are equivalent:

- V is termwise definitionally equivalent to a variety of Hilbert algebras with compatible operations. In particular, any algebra A := ⟨A; 1, f<sub>j</sub>⟩<sub>j∈J</sub> ∈ V is term equivalent to a Hilbert algebra with compatible operations ⟨A; →<sup>A</sup>, 1, f<sub>j</sub>⟩<sub>j∈J</sub>;
- 2. V is congruence orderable and subtractive with EDPC. In particular, V is congruence orderable and the binary term  $y \rightarrow x$  witnesses both sub-tractivity and EDPI for V in the sense of Theorem 1.7.9.

*Proof.* Let V be a variety with 1 and let  $x \to y$  be a binary term of V.

(1)  $\Rightarrow$  (2) Suppose V is termwise definitionally equivalent to a variety of Hilbert algebras with compatible operations. For the first assertion, if  $\mathbf{A} := \langle A; 1, f_j \rangle_{j \in J} \in \mathsf{V}$  is term equivalent to a Hilbert algebra with compatible operations  $\langle A; \rightarrow^{\mathbf{A}}, 1, f_j \rangle_{j \in J}$ , then Con  $\mathbf{A} = \operatorname{Con} \langle A; \rightarrow^{\mathbf{A}}, 1, f_j \rangle_{j \in J} = \operatorname{Con} \langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ , so  $\mathbf{A}$  is congruence orderable and subtractive with EDPC. It follows that  $\mathsf{V}$  is congruence orderable and subtractive with EDPC. For the second assertion, if  $\mathbf{A} := \langle A; 1, f_j \rangle_{j \in J} \in \mathsf{V}$  is term equivalent to a Hilbert algebra with compatible operations  $\langle A; \rightarrow^{\mathbf{A}}, 1, f_j \rangle_{j \in J}$ , then Con  $\mathbf{A} = \operatorname{Con} \langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ , so in particular  $\mathbf{I}(\mathbf{A}) = \mathbf{I}(\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle)$ , whence  $a \in \langle b \rangle_{\mathbf{A}}$  iff  $a \in \langle b \rangle_{\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle}$  iff  $b \rightarrow^{\mathbf{A}} a = 1$ . It follows that  $\rightarrow$  witnesses *both* subtractivity and EDPI for V in the sense of Theorem 1.7.9, and the proof is complete.

(2)  $\Rightarrow$  (1) Suppose V is congruence orderable and subtractive with EDPC, and in particular that the binary term  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9. Let  $\mathbf{A} := \langle A; 1, f_j \rangle_{j \in J} \in$ V. Clearly A is term equivalent to the algebra  $\langle A; \to^{\mathbf{A}}, 1, f_j \rangle_{j \in J}$  obtained from A by enriching the type of A with a binary operation symbol  $\to$  whose canonical interpretation is the term function  $\to^{\mathbf{A}}$ . We claim  $\langle A; \to^{\mathbf{A}}, 1, f_j \rangle_{j \in J}$ is a Hilbert algebra with compatible operations. To see this, it is sufficient by Lemma 3.1.10 to show:

(i) The polynomial reduct  $\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$  is a Hilbert algebra;

(ii) For any *n*-ary fundamental operation  $f := f_j$  of A,

$$(a \rightarrow^{\mathbf{A}} b) \rightarrow^{\mathbf{A}} ((b \rightarrow^{\mathbf{A}} a) \rightarrow (\widetilde{f}_{i}(a) \rightarrow^{\mathbf{A}} \widetilde{f}_{i}(b))) = 1$$

for all  $a, b \in A$  and slices  $\tilde{f}_i(a), \tilde{f}_i(b), 1 \leq i \leq n$ .

For (i), by Theorem 3.1.6 we have that the polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  is a MINI-algebra. By Theorem 3.1.4 and Corollary 2.2.23(2), the map  $f: a \mapsto \langle a \rangle_{\mathbf{A}}$  is a homomorphism and  $\langle A; \to^{\mathbf{A}}, 1 \rangle / \ker f$  is a Hilbert algebra, where  $\ker f = \{(a, b): a \to^{\mathbf{A}} b = 1 = b \to^{\mathbf{A}} a\}$ . Because V is congruence orderable, it is 1-regular (in the sense of Proposition 1.2.6) witness  $d_1(x, y) := x \to y$ ,  $d_2(x, y) := y \to x$  by Proposition 1.7.16, from which it follows that  $\ker f = \omega_{\mathbf{A}}$ .

Hence  $\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$  is a Hilbert algebra.

For (ii), let f(x) be a fundamental operation of A, which we may take to be unary without loss of generality. Let  $a, b \in A$  and let:

 $\langle a \rightarrow^{\mathbf{A}} b \rangle \vee^{\mathbf{I}(\mathbf{A})} \langle b \rightarrow^{\mathbf{A}} a \rangle = [1]_{\theta}$ 

for some  $\theta \in \text{Con } \mathbf{A}$ . Then of course  $[a]_{\theta} \to^{\mathbf{A}/\theta} [b]_{\theta} = [1]_{\theta} = [b]_{\theta} \to^{\mathbf{A}/\theta} [a]_{\theta}$ , so  $(a, b) \in \theta$ . It follows that  $(f(a), f(b)) \in \theta$  and by properties of  $\to$ ,  $f(a) \to^{\mathbf{A}} f(b) \in [1]_{\theta}$ . But  $[1]_{\theta} = \langle a \to^{\mathbf{A}} b \rangle_{\mathbf{A}} \vee^{\mathbf{I}(\mathbf{A})} \langle b \to^{\mathbf{A}} a \rangle_{\mathbf{A}}$ , so Theorem 3.1.4 and Lemma 2.2.19 applied twice gives:

$$(a \to^{\mathbf{A}} b) \to^{\mathbf{A}} \left( (b \to^{\mathbf{A}} a) \to^{\mathbf{A}} (f(a) \to^{\mathbf{A}} f(b)) \right) = 1$$

as desired.

Let A be an algebra with 1 and let  $\operatorname{Con}_{\epsilon} A := \{\theta \in \operatorname{Con} A : \theta \geq \langle 1 \rangle_{A}^{\epsilon}\}$ . An *n*-ary operation f of A is said to be weakly compatible if it is compatible with every congruence  $\theta \in \operatorname{Con}_{\epsilon} A$ . By analogy with the theory of MINI-algebras with compatible operations, an algebra  $A := \langle A; \to, 1, f_j \rangle_{j \in J}$  of type  $\langle 2, 0, \ldots \rangle$ is a MINI-algebra with weakly compatible operations if: (i) the reduct  $\langle A; \to, 1 \rangle$ is a MINI-algebra; and (ii) each additional operation  $f_j$  is weakly compatible with  $\langle A; \to, 1 \rangle$ .

Lemma 3.1.12. Let  $\mathbf{A} := \langle A; \rightarrow, 1, f_j \rangle_{j \in J}$  be a MINI-algebra with weakly compatible operations. Then  $\langle 1 \rangle_{\mathbf{A}}^{\epsilon} = \langle 1 \rangle_{\langle A; \rightarrow, 1 \rangle}^{\epsilon} = \approx_{\langle A; \rightarrow, 1 \rangle}$ .

Proof. Let  $\mathbf{A} := \langle A; \to, 1, f_j \rangle_{j \in J}$  be a MINI-algebra with weakly compatible operations. By hypothesis,  $\operatorname{Con}_{\epsilon} \mathbf{A} = \operatorname{Con}_{\epsilon} \langle A; \to, 1 \rangle$ , so  $\langle 1 \rangle_{\mathbf{A}}^{\epsilon} \in \operatorname{Con}_{\epsilon} \mathbf{A} \subseteq$  $\operatorname{Con}_{\epsilon} \langle A; \to, 1 \rangle$ . Hence  $\langle 1 \rangle_{\mathbf{A}}^{\epsilon} \in \operatorname{Con}_{\epsilon} \langle A; \to, 1 \rangle$ ; that is to say  $\langle 1 \rangle_{\mathbf{A}}^{\epsilon} \geq \langle 1 \rangle_{\langle A; \to, 1 \rangle}^{\epsilon}$ . The opposite inclusion is handled similarly. Thus  $\langle 1 \rangle_{\mathbf{A}}^{\epsilon} = \langle 1 \rangle_{\langle A; \to, 1 \rangle}^{\epsilon}$ . Since  $\langle A; \to, 1 \rangle$  is weakly congruence orderable, by Lemma 1.7.17 we have that  $\langle 1 \rangle_{\langle A; \to, 1 \rangle}^{\epsilon} = \approx_{\langle A; \to, 1 \rangle}$ . Hence  $\langle 1 \rangle_{\mathbf{A}}^{\epsilon} = \langle 1 \rangle_{\langle A; \to, 1 \rangle}^{\epsilon}$ .

Lemma 3.1.13. Let  $\mathbf{A} := \langle A; \rightarrow, 1, f_j \rangle_{j \in J}$  be a MINI-algebra with weakly compatible operations. Then  $\operatorname{Con} \mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon} = \operatorname{Con} \langle A; \rightarrow, 1 \rangle/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}$ .

Proof. Let  $\mathbf{A} := \langle A; \to, 1, f_j \rangle_{j \in J}$  be a MINI-algebra with weakly compatible operations. To prove the lemma, just observe that  $\theta \in \operatorname{Con} \mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}$  iff  $\theta = \psi/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}$  for some  $\psi \in \operatorname{Con}_{\epsilon} \mathbf{A}$  (since  $\langle 1 \rangle_{\mathbf{A}}^{\epsilon}$  is the least element of  $\operatorname{Con}_{\epsilon} \mathbf{A}$ ) iff  $\theta = \psi/\langle 1 \rangle_{\langle A; \to, 1 \rangle}^{\epsilon}$  with  $\psi \in \operatorname{Con}_{\epsilon}\langle A; \to, 1 \rangle$  (by Lemma 3.1.12 and hypothesis) iff  $\theta \in \operatorname{Con}\langle A; \to, 1 \rangle/\langle 1 \rangle_{\langle A; \to, 1 \rangle}^{\epsilon}$  (since  $\langle 1 \rangle_{\langle A; \to, 1 \rangle}^{\epsilon}$  is the least element of  $\operatorname{Con}_{\epsilon}\langle A; \to , 1 \rangle$ ) iff  $\theta \in \operatorname{Con}\langle A; \to, 1 \rangle/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}$  (by Lemma 3.1.12). Hence  $\operatorname{Con} \mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon} = \operatorname{Con}\langle A; \to, 1 \rangle/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}$  as desired.

**Lemma 3.1.14.** Let A be an algebra with 1, say  $\langle A; 1, f_j \rangle_{j \in J}$ , and let  $x \to y$  be a binary term of A such that the polynomial reduct  $\langle A; \to^A, 1 \rangle$  is a MINI-algebra. The following are equivalent:

- 1. A is term equivalent to a MINI-algebra with weakly compatible operations  $\langle A; \rightarrow^{\mathbf{A}}, 1, f_j \rangle_{j \in J};$
- 2. For any n-ary fundamental operation  $f := f_j$  of A,

$$(a \rightarrow^{\mathbf{A}} b) \rightarrow^{\mathbf{A}} ((b \rightarrow^{\mathbf{A}} a) \rightarrow (\widetilde{f}_{i}(a) \rightarrow^{\mathbf{A}} \widetilde{f}_{i}(b))) = 1$$

for all  $a, b \in A$  and slices  $\tilde{f}_i(a), \tilde{f}_i(b), 1 \leq i \leq n$ .

*Proof.* Let  $\mathbf{A} := \langle A; 1, f_j^{\mathbf{A}} \rangle_{j \in J}$  and  $\rightarrow$  be as in the statement of the lemma.

 $(1) \Rightarrow (2)$  Suppose A is term equivalent to a MINI-algebra with weakly compatible operations  $\langle A; \rightarrow^{A}, 1, f_{j}^{A} \rangle_{j \in J}$ . Because of Lemma 3.1.13,  $A/\langle 1 \rangle_{A}^{\epsilon}$  is term equivalent to a Hilbert algebra with compatible operations  $\langle A; \rightarrow^{A}, 1, f_{j}^{A} \rangle_{j \in J}/\langle 1 \rangle_{A}^{\epsilon}$ . To simplify notation, throughout the remainder of the proof we write  $\overline{a}$  for the equivalence class  $[a]_{\langle 1 \rangle_{A}^{\epsilon}}$  in  $A/\langle 1 \rangle_{A}^{\epsilon}$  containing  $a \in A$ . Let  $f^{A} := f_{j}^{A}$  be an *n*-ary fundamental operation of A. Since the operation  $f^{A}/\langle 1 \rangle_{A}^{\epsilon}$  of  $A/\langle 1 \rangle_{A}^{\epsilon}$  is compatible with the Hilbert algebra polynomial reduct  $\langle A; \rightarrow^{A}, 1 \rangle/\langle 1 \rangle_{A}^{\epsilon}$ , we have that:

$$(\overline{a} \to^{\mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}} \overline{b}) \to^{\mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}} ((\overline{b} \to^{\mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}} \overline{a}) \to^{\mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}} (\overline{b})) = \overline{1}$$

$$(\widetilde{f}_{i}^{\mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}} (\overline{a}) \to^{\mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}} (\overline{b}))) = \overline{1}$$

for all  $\overline{a}, \overline{b} \in A/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}$  and slices  $\widetilde{f}_{i}^{\mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}}(\overline{a}), \widetilde{f}_{i}^{\mathbf{A}/\langle 1 \rangle_{\mathbf{A}}^{\epsilon}}(\overline{b}), 1 \leq i \leq n$ . Since

 $[1]_{(1)'_{A}} = \{1\}$ , we infer:

$$(a \rightarrow^{\mathbf{A}} b) \rightarrow^{\mathbf{A}} ((b \rightarrow^{\mathbf{A}} a) \rightarrow^{\mathbf{A}} (\widetilde{f}_{i}^{\mathbf{A}}(a) \rightarrow^{\mathbf{A}} \widetilde{f}_{i}^{\mathbf{A}}(b))) = 1$$

for all  $a, b \in A$  and slices  $\tilde{f}_i^{\mathbf{A}}(a), \tilde{f}_i^{\mathbf{A}}(b), 1 \leq i \leq n$ , as required.

(2)  $\Rightarrow$  (1) Clearly **A** is term equivalent to the algebra  $\langle A; \rightarrow^{\mathbf{A}}, 1, f_{j}^{\mathbf{A}} \rangle_{j \in J}$ obtained from **A** by enriching the type of **A** with a binary operation symbol  $\rightarrow$  whose canonical interpretation is the term function  $\rightarrow^{\mathbf{A}}$ . We claim  $\langle A; \rightarrow^{\mathbf{A}}, 1, f_{j}^{\mathbf{A}} \rangle_{j \in J}$  is a MINI-algebra with weakly compatible operations. By assumption,  $\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$  is a MINI-algebra. To complete the proof it is sufficient to show  $\operatorname{Con}_{\epsilon} \mathbf{A} = \operatorname{Con}_{\epsilon} \langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ . If  $\theta \in \operatorname{Con}_{\epsilon} \mathbf{A}$  then certainly  $\theta \in \operatorname{Con}_{\epsilon} \langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ , so  $\operatorname{Con}_{\epsilon} \mathbf{A} \subseteq \operatorname{Con}_{\epsilon} \langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ . For the converse, let  $\theta \in \operatorname{Con}_{\epsilon} \langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ . From the remarks of [11, p. 315] we have that  $\theta = I^{\epsilon}$ for some  $I \in I(\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle)$ , whence  $\theta = \phi_I$  by Theorem 3.1.4 and Proposition 2.1.35, where:

$$\phi_I := \{(a, b) \in A \times A : a \to^{\mathbf{A}} b, b \to^{\mathbf{A}} a \in I\}.$$

To complete the proof we show:

- (i)  $\phi_I$  is a congruence on A;
- (ii)  $\phi_I \geq \langle 1 \rangle_{\mathbf{A}}^{\epsilon}$ .

For (i), by the proof of Theorem 2.1.26,  $\phi_i$  is an equivalence relation on  $A \times A$ . Let  $f := f_j^A$  be an *n*-ary fundamental operation of **A** and suppose  $a_i \approx_A b_i$  for i = 1, ..., n. For i = 1, ..., n and any  $c \in A$ , let  $\tilde{f}_i(c)$  denote the slice:

 $f(\dot{b}_1,\ldots,\dot{b}_{i-1},c,a_{i+1},\ldots,a_n).$ 

In what follows we write  $\rightarrow$  for  $\rightarrow^{\mathbf{A}}$  to simplify notation. By hypothesis, we have that:

$$(a_1 \rightarrow b_1) \rightarrow ((b_1 \rightarrow a_1) \rightarrow (\widetilde{f}_1(a_1) \rightarrow \widetilde{f}_1(b_1))) = 1$$

$$(a_i \to b_i) \to ((b_i \to a_i) \to (\widetilde{f}_i(a_i) \to \widetilde{f}_i(b_i))) = 1$$
  
$$\vdots$$
  
$$(a_n \to b_n) \to ((b_n \to a_n) \to (\widetilde{f}_n(a_n) \to \widetilde{f}_n(b_n))) = 1$$

and by symmetry, also:

$$(b_1 \to a_1) \to ((a_1 \to b_1) \to (\widetilde{f}_1(b_1) \to \widetilde{f}_1(a_1))) = 1$$
  
$$\vdots$$
  
$$(b_i \to a_i) \to ((a_i \to b_i) \to (\widetilde{f}_i(b_i) \to \widetilde{f}_i(a_i))) = 1$$
  
$$\vdots$$
  
$$(b_n \to a_n) \to ((a_n \to b_n) \to (\widetilde{f}_n(b_n) \to \widetilde{f}_n(a_n))) = 1.$$

That is to say,

$$(a_{1} \rightarrow b_{1}) \rightarrow ((b_{1} \rightarrow a_{1}) \rightarrow (f(a_{1}, a_{2}, \dots, a_{n}))) = 1$$

$$\Rightarrow f(b_{1}, a_{2}, \dots, a_{n})) = 1$$

$$\vdots$$

$$(a_{i} \rightarrow b_{i}) \rightarrow ((b_{i} \rightarrow a_{i}) \rightarrow (f(b_{1}, \dots, b_{i-1}, a_{i}, a_{i+1}, \dots, a_{n}))) = 1$$

$$\Rightarrow f(b_{1}, \dots, b_{i-1}, b_{i}, a_{i+1}, \dots, a_{n}))) = 1$$

$$\vdots$$

$$(a_{n} \rightarrow b_{n}) \rightarrow ((b_{n} \rightarrow a_{n}) \rightarrow (f(b_{1}, \dots, b_{n-1}, a_{n}))) = 1.$$

and:

$$(b_1 \rightarrow a_1) \rightarrow ((a_1 \rightarrow b_1) \rightarrow (f(b_1, a_2, \dots, a_n))$$
$$\rightarrow f(a_1, a_2, \dots, a_n))) = 1$$
$$\vdots$$
$$(b_i \rightarrow a_i) \rightarrow ((a_i \rightarrow b_i) \rightarrow (f(b_1, \dots, b_{i-1}, b_i, a_{i+1}, \dots, a_n)))$$

234

$$\rightarrow f(b_1, \ldots, b_{i-1}, a_i, a_{i+1}, \ldots, a_n))) = 1$$

$$\vdots$$

$$(b_n \rightarrow a_n) \rightarrow ((a_n \rightarrow b_n) \rightarrow (f(b_1, \ldots, b_{n-1}, b_n)$$

$$\rightarrow f(b_1, \ldots, b_{n-1}, a_n))) = 1.$$

Since  $a_i \approx b_i$  implies  $a_i \rightarrow b_i$ ,  $b_i \rightarrow a_i \in I$  for i = 1, ..., n, from (the dual of) Lemma 2.1.21(2) we infer that:

$$f(a_{1}, a_{2}, ..., a_{n}) \to f(b_{1}, a_{2}, ..., a_{n}) \in I$$
  

$$\vdots$$
  

$$f(b_{1}, ..., b_{i-1}, a_{i}, a_{i+1}, ..., a_{n}) \to f(b_{1}, ..., b_{i-1}, b_{i}, a_{i+1}, ..., a_{n}) \in I$$
  

$$\vdots$$
  

$$f(b_{1}, ..., b_{n-1}, a_{n}) \to f(b_{1}, ..., b_{n-1}, b_{n}) \in I$$

and also that:

$$f(b_1, a_2, ..., a_n) \to f(a_1, a_2, ..., a_n) \in I$$
  

$$\vdots$$
  

$$f(b_1, ..., b_{i-1}, b_i, a_{i+1}, ..., a_n) \to f(b_1, ..., b_{i-1}, a_i, a_{i+1}, ..., a_n) \in I$$
  

$$\vdots$$
  

$$f(b_1, ..., b_{n-1}, b_n) \to f(b_1, ..., b_{n-1}, a_n) \in I.$$

Hence:

$$f(a_1, a_2, \dots, a_n) \equiv f(b_1, a_2, \dots, a_n) \pmod{\phi_I}$$

$$\vdots$$

$$f(b_1, \dots, b_{i-1}, a_i, a_{i+1}, \dots, a_n) \equiv f(b_1, \dots, b_{i-1}, b_i, a_{i+1}, \dots, a_n) \pmod{\phi_I}$$

$$\vdots$$

$$f(b_1, \dots, b_{n-1}, a_n) \equiv f(b_1, \dots, b_{n-1}, b_n) \pmod{\phi_I}$$

which implies  $f(a_1, \ldots, a_n) \equiv f(b_1, \ldots, b_n) \pmod{\phi_I}$  by transitivity. Hence  $\phi_I$  is a congruence on **A**.

For (ii), just note that  $\phi_I \geq \langle 1 \rangle_{\langle A; \to, 1 \rangle}^{\epsilon}$  (because of Proposition 2.1.35)  $\geq \langle 1 \rangle_{\mathbf{A}}^{\epsilon}$  (by Lemma 3.1.12). Hence  $\phi_I \geq \langle 1 \rangle_{\mathbf{A}}^{\epsilon}$ .

By (i) and (ii),  $\phi_I$  is a congruence on A such that  $\phi_I \geq \langle 1 \rangle_{\mathbf{A}}^{\epsilon}$ . Therefore  $\phi_I \in \operatorname{Con}_{\epsilon} \mathbf{A}$  and hence  $\theta \in \operatorname{Con}_{\epsilon} \mathbf{A}$  (since  $\theta = \phi_I$ ). Thus  $\operatorname{Con}_{\epsilon} \langle A; \to^{\mathbf{A}}, 1 \rangle \subseteq \operatorname{Con}_{\epsilon} \mathbf{A}$ , and the proof is complete.

A variety V is called a variety of MINI-algebras with weakly compatible operations if every  $A \in V$  is a MINI-algebra with weakly compatible operations. If V is a variety of MINI-algebras with compatible operations, then certainly V is a variety of MINI-algebras with weakly compatible operations. In general, however, the converse does not hold. This is shown by Corollary 3.1.16 below, which obtains as an immediate consequence of the following proposition.

**Proposition 3.1.15.** The variety of left handed skew Boolean algebras is termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations. However, there exists a left handed skew Boolean algebra S that has no MINI-algebra polynomial reduct whose congruences coincide with those of S. Hence the variety of left handed skew Boolean algebras is not termwise definitionally equivalent to a variety of MINI-algebras with compatible operations.

*Proof.* For the first assertion, by the proof of Proposition 3.1.8 we have that any skew Boolean algebra  $\mathbf{A} := \langle A; \land, \lor, \backslash, 0 \rangle$  has a MINI-algebra polynomial reduct  $\langle A; \rightarrow^{\mathbf{A}}, \mathbf{1}^{\mathbf{A}} \rangle$ , where  $a \rightarrow^{\mathbf{A}} b := b \backslash a$  for any  $a, b \in A$  and  $\mathbf{1}^{\mathbf{A}} :=$ 0. Further, an easy inspection of the subdirectly irreducible skew Boolean algebras  $\mathbf{2}^{p}, \mathbf{3}_{L}^{p}$  and  $\mathbf{3}_{R}^{p}$  shows SBA satisfies the identities:

$$(x \to y) \to ((y \to x) \to ((x \circ z) \to (y \circ z))) \approx 1$$
$$(x \to y) \to ((y \to x) \to ((z \circ x) \to (z \circ y))) \approx 1$$

where  $\circ$  denotes either the term corresponding to the skew lattice meet  $\wedge$  or the term corresponding skew lattice join  $\vee$ . From these remarks it follows that **A** is term equivalent to a MINI-algebra with weakly compatible operations  $\langle A; -, \widehat{-}, 1^{\mathbf{A}}, \wedge, \vee \rangle$ . Hence SBA (and in particular, lhSBA) is termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations. The remaining statements of the proposition may be inferred from Proposition 3.1.8.

Corollary 3.1.16. Let V be a variety termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations. In general, Vis not termwise definitionally equivalent to a variety of MINI-algebras with compatible operations.

Varieties of MINI-algebras with weakly compatible operations were introduced by Agliano in [6, Section 4] (under the name *MINI-algebras with compatible operations*) in the context of his study of weakly congruence orderable subtractive varieties with EDPI. Because of Lemma 3.1.13, the class of all varieties of MINI-algebras with weakly compatible operations may be understood as a natural generalisation of the class of all varieties of Hilbert algebras with compatible operations; that this generalisation is essential follows from Corollary 3.1.16. Inasmuch as varieties of MINI-algebras with weakly compatible operations generalise varieties of Hilbert algebras with compatible operations, Theorem 3.1.11 lends one to ask if varieties of MINI-algebras with weakly compatible operations also admit a relevant structure theorem. The following result answers this question in the affirmative.

**Theorem 3.1.17.** For a variety V with 1 and a binary term  $x \rightarrow y$  of V, the following are equivalent:

- 1. V is weakly congruence orderable and the binary term  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9;
- V is termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations. In particular, any A := ⟨A; 1, f<sub>j</sub>⟩<sub>j∈J</sub> ∈ V is term equivalent to a MINI-algebra with weakly compatible operations ⟨A; →<sup>A</sup>, 1, f<sub>j</sub>⟩<sub>j∈J</sub>.
Remark 3.1.18. The statement but not the proof of Theorem 3.1.17 is due to Agliano [6, Section 4]. In more detail, in [6, Theorem 4.5] Agliano asserted that:

'A pointed variety is weakly congruence orderable, subtractive and has EDPI iff it is termwise definitionally equivalent to a variety of MINI-algebras with [weakly] compatible operations.'

Although Agliano's proof of the sufficiency of the above assertion is valid, his proof of the necessity of the assertion is not. For his proof of necessity Agliano simply asserts that [6, p. 16]:

'In a MINI-algebra with [weakly] compatible operations the congruences (and hence the ideals) depend only on the MINI-algebra operation. Therefore any such variety is weakly congruence orderable, subtractive and has EDPI.'

Because a variety of MINI-algebras with weakly compatible operations need not have compatible operations (by Corollary 3.1.16), this argument is not sufficient to establish the necessary direction of the preceding assertion. To correct this error, we provide new proofs below of both the necessity and the sufficiency of Theorem 3.1.17.

Proof (of Theorem 3.1.17). Let V be a variety with 1 and let  $x \to y$  be a binary term of V.

(1)  $\Rightarrow$  (2) Suppose V is weakly congruence orderable and that the binary term  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9. Let  $\mathbf{A} := \langle A; 1, f_j^{\mathbf{A}} \rangle_{j \in J} \in V$ . Clearly  $\mathbf{A}$  is term equivalent to the algebra  $\langle A; \to^{\mathbf{A}}, 1, f_j^{\mathbf{A}} \rangle_{j \in J}$  obtained from  $\mathbf{A}$  by enriching the type of  $\mathbf{A}$  with a binary operation symbol  $\to$  whose canonical interpretation on  $\langle A; \to^{\mathbf{A}}, 1, f_j^{\mathbf{A}} \rangle_{j \in J}$  is the term function  $\to^{\mathbf{A}}$ . We claim  $\langle A; \to^{\mathbf{A}}, 1, f_j^{\mathbf{A}} \rangle_{j \in J}$  is a MINI-algebra with weakly compatible operations. Since  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9, from Theorem 3.1.6 we have that the polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  is a MINI-algebra. Also, since V is weakly congruence orderable, from Proposition 1.7.19 we have

that  $V_{\epsilon}$  is a subtractive congruence orderable variety with EDPC. From Theorem 3.1.11 it follows that  $\langle A; \rightarrow^{A}, 1, f_{j}^{A} \rangle_{j \in J} / \approx_{A}$  is a Hilbert algebra with compatible operations. That is to say,  $\langle A; \rightarrow^{A}, 1, f_{j}^{A} \rangle_{j \in J} / \langle 1 \rangle_{A}^{\epsilon}$  is a Hilbert algebra with compatible operations, just because  $\approx_{A} = \langle 1 \rangle_{A}^{\epsilon}$  by Lemma 1.7.17. The proof of Lemma 3.1.14 now shows that for any *n*-ary fundamental operation  $f^{A} := f_{j}^{A}$  of  $\langle A; \rightarrow^{A}, 1, f_{j}^{A} \rangle_{j \in J}$ ,

$$(a \to^{\mathbf{A}} b) \to^{\mathbf{A}} ((b \to^{\mathbf{A}} a) \to^{\mathbf{A}} (\widetilde{f_i}^{\mathbf{A}}(a) \to^{\mathbf{A}} \widetilde{f_i}^{\mathbf{A}}(b))) = 1$$

for all  $a, b \in A$  and slices  $\tilde{f}_i^{\mathbf{A}}(a)$ ,  $\tilde{f}_i^{\mathbf{A}}(b)$ ,  $1 \leq i \leq n$ . Thus  $\langle A; \to^{\mathbf{A}}, 1, f_j^{\mathbf{A}} \rangle_{j \in J}$  is a MINI-algebra with weakly compatible operations. Hence V is termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations, and the proof is complete.

(2)  $\Rightarrow$  (1) Let  $\mathbf{A} := \langle A; 1, f_j \rangle_{j \in J} \in V$ . By hypothesis,  $\mathbf{A}$  is term equivalent to a MINI-algebra with weakly compatible operations  $\langle A; \rightarrow^{\mathbf{A}}, 1, f_j^{\mathbf{A}} \rangle_{j \in J}$ . Therefore V is subtractive (witness  $y \rightarrow x$ ). To see V has EDPI (witness  $y \rightarrow x$ ), it is sufficient by Corollary 3.1.7 to show  $I(\mathbf{A}) = I(\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle)$ . As Con  $\mathbf{A} \subseteq \operatorname{Con} \langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ , we have that  $N(\mathbf{A}) \subseteq N(\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle)$ , and hence that  $I(\mathbf{A}) \subseteq I(\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle)$ . For the converse, let  $I \in I(\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle)$  and let  $\phi_I$  be the relation on  $A \times A$  defined by:

 $\phi_I := \{(a, b) \in A \times A : a \to^{\mathbf{A}} b, b \to^{\mathbf{A}} a \in I\}.$ 

By the proof of Lemma 3.1.14,  $\phi_I$  is a congruence on **A**. Moreover,  $I = [0]_{\phi_I}$ , just because of Theorem 3.1.4 and Lemma 2.1.27. Hence  $I \in I_V(\mathbf{A})$  and  $I(\langle A; \to^{\mathbf{A}}, 1 \rangle) \subseteq I(\mathbf{A})$ . Thus  $I(\mathbf{A}) = I(\langle A; \to^{\mathbf{A}}, 1 \rangle)$  and V has EDPI (witness  $y \to x$ ).

It remains to show V is weakly congruence orderable. Since V is subtractive with EDPI (witness  $y \to x$ ), from Lemma 1.7.14 we have that  $a \approx_{\mathbf{A}} b$  iff  $a \to^{\mathbf{A}} b = 1 = b \to^{\mathbf{A}} a$  iff  $a \equiv b \pmod{\phi_{\{1\}}}$ . Since  $\phi_{\{1\}} \in \text{Con } \mathbf{A}$  by the proof of Lemma 3.1.14 we have that  $\approx_{\mathbf{A}}$  is a congruence on  $\mathbf{A}$ , so  $\mathbf{A}$  is weakly congruence orderable. Hence V is weakly congruence orderable, and the proof is complete.

Remark 3.1.19. In [6, Theorem 4.15] Agliano extended the characterisation of weakly congruence orderable subtractive varieties with EDPI of Theorem 3.1.17 to subtractive weakly congruence orderable varieties with EDPI and both a meet generator and a join generator term. Agliano's proof of [6, Theorem 4.15] relies on the assumption that if **A** is an algebra term equivalent to a (certain notion of) L-algebra with compatible operations  $\langle A; \sqcap^{\mathbf{A}}, \sqcup^{\mathbf{A}}, \rightarrow^{\mathbf{A}}$  $, 1, f_j \rangle_{j \in J}$ , then Con  $\mathbf{A} = \operatorname{Con} \langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ . From the definition of an L-algebra given by Agliano in [6, Section 4, p. 20], it is readily verified by inspection of the subdirectly irreducible skew Boolean algebras  $2^p$ ,  $3^p_L$  and  $3^p_R$  that any skew Boolean algebra  $\langle B; \land, \lor, \backslash, 0 \rangle$  is term equivalent to an L-algebra  $\langle B; \sqcap, \sqcup, \rightarrow, 1 \rangle$ . From Proposition 3.1.15 it follows that, in general, Con  $\mathbf{A} \neq$ Con  $\langle A; \rightarrow^{\mathbf{A}}, 1 \rangle$ . Hence the argument employed by Agliano to characterise subtractive weakly congruence orderable varieties with EDPI and both a meet generator and a join generator term does not hold in general, whence the problem of characterising such varieties remains open.

**Example 3.1.20.** By Example 2.3.12, any pseudocomplemented semilattice A has a canonical implicative BCS-algebra polynomial reduct  $\langle A; \rangle, 0 \rangle$ , where  $a \backslash b := a \land b^*$  for any  $a, b \in A$ . For any  $a, b \in A$ , let  $a \rightarrow^{\mathbf{A}} b := b \backslash a$  and let  $\mathbf{1}^{\mathbf{A}} := 0$ . Then for any  $a, b, c \in A$ , we have that:

$$(a \rightarrow^{\mathbf{A}} b) \rightarrow^{\mathbf{A}} ((b \rightarrow^{\mathbf{A}} a) \rightarrow^{\mathbf{A}} (a^{*} \rightarrow^{\mathbf{A}} b^{*}))$$

$$= ((b^{*} \backslash a^{*}) \backslash (a \backslash b)) \backslash (b \backslash a)$$

$$= b^{*} \land a^{**} \land (a \land b^{*})^{*} \land (b \cdot a^{*})^{*}$$

$$= (a^{**} \land (a^{*} \land b)^{*}) \land (b^{*} \land (b^{*} \land c)^{**})$$

$$= (a^{**} \land (a^{*} \land b)^{*}) \land (b^{*} \land a^{*}) \qquad \text{by (1.9)}$$

$$= (a^{**} \land a^{*}) \land ((a^{*} \land b)^{*} \land b^{*})$$

$$= 0$$

$$= 1^{\mathbf{A}}$$

and:

$$\begin{aligned} a \rightarrow^{\mathbf{A}} b \end{pmatrix} \rightarrow^{\mathbf{A}} \left( (b \rightarrow^{\mathbf{A}} a) \rightarrow^{\mathbf{A}} ((a \wedge c) \rightarrow^{\mathbf{A}} (b \wedge c)) \right) \\ &= \left( ((b \wedge c) \setminus (a \wedge c)) \setminus (a \setminus b) \right) \setminus (b \setminus a) \\ &= (b \wedge c) \wedge (a \wedge c)^* \wedge (a \wedge b^*)^* \wedge (b \wedge a^*)^* \\ &= (c \wedge (c \wedge a)^*) \wedge (b \wedge (b \wedge a^*)^*) \wedge (a \wedge b^*)^* \\ &= (c \wedge a^*) \wedge (b \wedge a^{**}) \wedge (a \wedge b^*)^* \\ &= (c \wedge b) \wedge (a^* \wedge a^{**}) \wedge (a \wedge b^*)^* \\ &= (c \wedge b) \wedge 0 \wedge (a \wedge b^*)^* \\ &= 0 \\ &= \mathbf{1}^{\mathbf{A}} \end{aligned}$$

whence **A** is term equivalent to a MINI-algebra with weakly compatible operations  $\langle A; \to^{\mathbf{A}}, \mathbf{1}^{\mathbf{A}}, \wedge, * \rangle$  (by Theorem 3.1.4). Hence PCSL is termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations; from Theorem 3.1.17 we conclude that PCSL is weakly congruence orderable and subtractive with EDPI.

Conversely, from Example 2.3.12 we have that the variety PCSL of pseudocomplemented semilattices is subtractive witness  $x \setminus y$ . Also, by Agliano and Ursini [11, p. 387] it is known that  $\langle a \rangle_{\mathbf{A}} = \{b : b \leq a^{**}\}$  for any pseudocomplemented semilattice  $\mathbf{A}$  and  $a \in A$ . Suppose  $b \leq a^{**}$ . Then  $b \wedge a^{**} = b$ , whence  $b \wedge a^* = (b \wedge a^{**}) \wedge a^* = b \wedge (a^{**} \wedge a^*) = 0$  by (1.8). Hence  $b \leq \langle A; \setminus 0 \rangle$  a. On the other hand, suppose  $b \leq \langle A; \setminus 0 \rangle$  a. Then  $b \wedge a^* = 0$ , whence  $b \wedge a^{**} = b \wedge (b \wedge a^*)^* = b \wedge 0^* = b$  by (1.9). Hence  $b \leq a^{**}$ and so  $b \leq a^{**}$  iff  $b \leq \langle A; \setminus 0 \rangle$  a. Thus  $\langle a \rangle_{\mathbf{A}} = \langle a \rangle_{\langle A; \setminus 0 \rangle}$  and PCSL has EDPI (we remark that this result has been obtained independently by Agliano and Ursini [11, Example 5.9]). Moreover,  $\approx_{\mathbf{A}}$  is an equivalence relation on  $A \times A$ (recall the remarks of §1.7.8) and for any  $a, b \in A$  we have that  $a \approx_{\mathbf{A}} b$  iff  $a \setminus b = 0 = b \setminus a$  (by Lemma 1.7.14) iff  $a \equiv b \pmod{\Xi_{\langle A; \setminus 0 \rangle}}$  iff  $a^* = b^*$  (by Example 2.3.61(2) and Lemma 2.3.63(3)). Suppose  $a_1 \approx_{\mathbf{A}} b_1$  and  $a_2 \approx_{\mathbf{A}} b_2$  for  $a_1, a_2, b_1, b_2 \in A$ . Since  $a_1^{**} = (a_1^*)^* = (b_1^*)^* = b_1^{**}$  we have that  $\approx_{\mathbf{A}}$  has the substitution property for the \* operation. Also,

$$(a_1 \wedge a_2)^* = (a_1^{**} \wedge a_2^{**})^*$$
 by Jones [128, p. 2]  
$$= ((a_1^*)^* \wedge (a_2^*)^*)^*$$
  
$$= ((b_1^*)^* \wedge (b_2^*)^*)^*$$
  
$$= (b_1^{**} \wedge b_2^{**})^*$$
  
$$= (b_1 \wedge b_2)^*$$
 by Jones [128, p. 2]

and so  $\approx_A$  also has the substitution property for the  $\wedge$  operation. Thus  $\approx_A$  is a congruence on A and so PCSL is weakly congruence orderable. Since PCSL is weakly congruence orderable and subtractive with EDPI, from Theorem 3.1.17 we conclude that PCSL is termwise definitionally equivalent to a variety of MINI-algebras with weakly compatible operations.

Theorem 3.1.17 notwithstanding, we do not know if varieties of MINI-algebras with compatible operations admit a coherent structure theory in general. Hence we conclude this subsection with the following problem.

**Problem 3.1.21.** Do varieties of MINI-algebras with compatible operations admit a coherent structure theory?

3.1.22. Subtractive WBSO Varieties and Nelson Algebras. Recall the definition of a WBSO variety from §1.5.3. By a subtractive WBSO variety we mean a WBSO variety that is subtractive. Subtractive WBSO varieties have been characterised in the literature: by [11, Theorem 5.4] a pointed variety is a subtractive WBSO variety iff it is ideal determined, has EDPI and a join generator term. Let V be a subtractive WBSO variety. By hypothesis, V is subtractive; moreover V is strongly point regular with EDPC by Lemma 1.5.4, Proposition 1.5.5 and Proposition 1.5.6. Suppose now that V is a variety with 1 that is subtractive and strongly 1-regular with EDPC. Subtractivity and 1regularity imply V is ideal determined (by Proposition 1.7.3); also subtractivity and EDPC imply V has EDPI (by Proposition 1.7.10). Let  $A \in V$  and let  $a, b \in$ A. Since V is strongly 1-regular there exists  $c \in A$  such that  $\Theta^A(a, 1) \vee^{CpA}$ 

 $\Theta^{\mathbf{A}}(b,1) = \Theta^{\mathbf{A}}(c,1)$ , whence:

$$\langle a \rangle_{\mathbf{A}} \vee^{\mathbf{CI}(\mathbf{A})} \langle b \rangle_{\mathbf{A}} = [1]_{\Theta^{\mathbf{A}}(a,1)} \vee^{\mathbf{CI}(\mathbf{A})} [1]_{\Theta^{\mathbf{A}}(b,1)}$$
$$= [1]_{\Theta^{\mathbf{A}}(a,1)} \vee^{\mathbf{C}_{\mathbf{P}}\mathbf{A}} \otimes^{\mathbf{A}}(b,1)$$
$$= [1]_{\Theta^{\mathbf{A}}(c,1)}$$
$$= \langle c \rangle_{\mathbf{A}}$$

by subtractivity and the dual Brouwerian semilattice isomorphism  $\theta \mapsto [1]_{\theta}$ between Cp A and CI(A) of Proposition 1.7.10. Thus the join on I(A) of two principal ideals of A is always principal; from Proposition 1.7.13 it follows that V has a join generator term. As V is ideal determined with both EDPI and a join generator term, we have that V is a subtractive WBSO variety. That is, we have proved:

**Proposition 3.1.23.** For a variety V with 1, the following are equivalent:

1. V is a subtractive WBSO variety;

2. V is ideal determined, has EDPI and a join generator term;

3. V is subtractive, strongly 1-regular and has EDPC.

Let V be a WBSO variety. Recall from [29, Section 2] that the following identities and quasi-identity are satisfied for any weak relative pseudocomplementation  $\rightarrow$  of V:

$$x \to x \approx 1 \tag{3.8}$$

$$1 \to x \approx x \supset x \approx 1. \tag{3.9}$$

From (3.8) and (3.9) it is clear that for V to be subtractive the quasi-identity (3.9) need only be strengthened to the identity  $1 \rightarrow x \approx x$ . By a WBSO<sup>#</sup> variety we mean a WBSO variety V such that  $V \models 1 \rightarrow x \approx x$  for a weak relative pseudocomplementation  $\rightarrow$ . Clearly any WBSO<sup>#</sup> variety is subtractive; in [4] Agliano investigated subtractive WBSO varieties and proved that the converse also holds. That is, in a subtractive WBSO variety V it is always possible to choose a weak relative pseudocomplementation  $\rightarrow$  in such a way

that  $V \models 1 \rightarrow x \approx x$ ; we call such a term a subtractive weak relative pseudocomplementation in the sequel. The following proposition, which sharpens Agliano's characterisation of subtractive WBSO varieties, is implicit in results due to Agliano [4] and Agliano and Ursini [11].

**Proposition 3.1.24.** For a variety  $\vee$  with 1 the following are equivalent:

- 1. V is a subtractive WBSO variety;
- 2. V is a WBSO<sup>#</sup> variety;
- 3. V is a WBSO variety and there exists a binary term  $x \rightarrow y$  of V such that:
  - (a) For any  $A \in V$ , the polynomial reduct  $\langle A; \rightarrow^{A}, 1 \rangle$  is a MINI-algebra;
  - (b)  $x \rightarrow y$  is a weak relative pseudocomplementation for V.

Proof. The equivalence of (1) and (2) is proved by Agliano in [4, Theorem 4]. (3)  $\Rightarrow$  (2) is trivial. It therefore remains to show (2)  $\Rightarrow$  (3). So let V be a WBSO<sup>#</sup> variety. Then there exists a binary term  $x \rightarrow y$  of V such that  $V \models 1 \rightarrow x \approx x$  and  $x \rightarrow y$  is a weak relative pseudocomplementation, whence Condition (3)(b) is satisfied. Let  $A \in V$  and let  $a, b \in A$ . Since  $x \rightarrow y$ is a weak relative pseudocomplementation for V,

 $a \rightarrow^{\mathbf{A}} b = 1$  iff  $a \preceq b$  where  $\preceq$  is the quasiorder of §1.5.3 iff  $\Theta^{\mathbf{A}}(b,1) \subseteq \Theta^{\mathbf{A}}(a,1)$  by [29, Lemma 2.5(i)] iff  $[1]_{\Theta^{\mathbf{A}}(b,1)} \subseteq [1]_{\Theta^{\mathbf{A}}(a,1)}$ iff  $\langle b \rangle_{\mathbf{A}} \subseteq \langle a \rangle_{\mathbf{A}}$ iff  $b \in \langle a \rangle_{\mathbf{A}}$ .

Hence  $y \to x$  witnesses both subtractivity and EDPI for V in the sense of Theorem 1.7.9. By Theorem 3.1.6 we conclude that the polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  is a MINI-algebra; thus Condition (3)(a) is satisfied also and the proof is complete.

WBSO<sup>#</sup> varieties arise very naturally in algebraic logic as arbitrary congruence permutable point regular varieties with EDPC. Indeed, suppose V is a congruence permutable point regular (say 1-regular) variety with EDPC. Then V has a QD term q(x, y, z, w) by Theorem 1.5.8 and binary terms  $d_1(x, y), \ldots, d_n(x, y)$  witnessing the 1-regularity of V by Proposition 1.2.6. By Proposition 1.5.11(2),

$$egin{aligned} x o y &:= q(x, \mathbf{1}, y, \mathbf{1}) \ x \cdot y &:= q(x, \mathbf{1}, y, x) \ x \Delta y &:= ig( \cdots (d_1(x, y) \cdot d_2(x, y)) \cdots ig) \cdot d_n(x, y) \end{aligned}$$

are respectively subtractive weak relative pseudocomplementation, weak meet and Gödel equivalence terms for V, whence V is a WBSO# variety.

To within clone equivalence, the term operations induced by a weak meet  $x \cdot y$ and weak relative pseudocomplementation  $x \to y$  in a WBSO variety faithfully reflect conjunction and implication on the Brouwerian semilattice of compact congruences of any member of the variety: see Blok and Pigozzi [34, p. 547]. In view of this remark it is easy to see that  $\cdot$  and  $\rightarrow$  are respectively a conjunction and implication for the intrinsic assertional logic S(V, 1) of V. The following result, which is essentially well known, formalises and extends this observation.

**Proposition 3.1.25.** Let V be a WBSO<sup>#</sup> variety with assertional logic S(V, 1). Then S(V, 1) is algebraisable and its equivalent algebraic semantics is exactly V. Moreover, for binary terms  $\cdot, +, \Delta, \rightarrow$  of V the following assertions hold (upon identifying the individual variables x, y with the propositional variables p, qrespectively):

- 1. [181, Theorem 2.1]  $x \cdot y$  is a weak meet for  $\vee$  iff  $p \cdot q$  is a conjunction for  $\mathbb{S}(\vee, 1)$ ;
- [181, Theorem 2.2] x∆y is a Gödel equivalence term for V iff p∆q is a G-identity for S(V, 1);

3. [181, Theorem 2.5]  $x \to y$  is a weak relative pseudocomplementation for  $\vee$  iff  $p \to q$  is a conditional for  $\mathbb{S}(\vee, 1)$ ;

Ý,

- 4. [181, p. 485] (p → q) · (q → p) is a biconditional for S(V, 1) iff x · y and x → y are respectively a weak meet and weak relative pseudocomplementation for V;
- 5. [224, Theorem 5.1] x + y is a weak join for  $\vee$  iff p + q is a disjunction for  $\mathbb{S}(\vee, 1)$ ;
- 6. If V is double-pointed (say with {0,1}) and x → y is a subtractive weak relative pseudocomplementation for V then p → 0 is a weak negation for S(V,1).

Proof. Let V be a WBSO<sup>#</sup> variety with assertional logic S(V, 1). By Theorem 1.8.15 S(V, 1) is algebraicable and its equivalent algebraic semantics is exactly V. Of the remaining statements, only (6) is not explicit in the literature. So suppose V is with  $\{C, 1\}$  and that  $\rightarrow$  is a subtractive weak relative pseudocomplementation for V. Let  $\neg p := p \rightarrow 0$ . To see  $\neg$  is a weak negation we show (CN) and (RA<sub>J</sub>) are respectively rules of S(V, 1), viz.:

Throughout the remainder of the proof we identify the individual variables  $\vec{x}$  with the propositional variables  $\vec{p}$  and the terms  $s(\vec{x})$ ,  $\{s_i(\vec{x}) : i = 1, ..., n\}, t(\vec{x})$  with the formulas  $\varphi(\vec{p}), \{\varphi_i(\vec{p}) : i = 1, ..., n\}, \psi(\vec{p})$  respectively. To ease notation, for a given formula  $\varphi(\vec{p})$  we also write simply  $\varphi$ .

For (i), it is sufficient to show  $\{s(\vec{x}) \approx 1, \neg s(\vec{x}) \approx 1\} \models_{\mathsf{V}} t(\vec{x}) \approx 1$ . So let  $A \in \mathsf{V}$  and  $\vec{a} \in A$ . If  $s^{\mathsf{A}}(\vec{a}) = 1$  and  $\neg s^{\mathsf{A}}(\vec{a}) = 1$  then  $1 \to 0 = 1$ , which implies 0 = 1 as  $\rightarrow$  is a subtractive weak relative pseudocomplementation. But then 0 and 1 are not residually distinct, so A is trivial and  $t^{\mathsf{A}}(\vec{a}) = 1$ .

For (ii), by the remarks of §1.8.9 we may assume without loss of generality that  $\Gamma$  is finite, say  $\{\varphi_1, \ldots, \varphi_n\}$ . Moreover, by (3)  $\rightarrow$  is a conditional for  $S(V, \mathbf{1})$ , so the entailment  $\varphi_1, \ldots, \varphi_n, \psi \vdash_{S(V,\mathbf{1})} \neg \psi$  is equivalent to the entailment  $\vdash_{S(V,\mathbf{1})} \varphi_1 \rightarrow (\varphi_2 \rightarrow (\cdots \rightarrow (\varphi_n \rightarrow (\psi \rightarrow \neg \psi)) \cdots))$ , and likewise the entailment  $\varphi_1, \ldots, \varphi_n \vdash_{S(V,\mathbf{1})} \neg \psi$  is equivalent to the entailment

 $\vdash_{S(V,1)} \varphi_1 \to (\varphi_2 \to (\cdots \to (\varphi_n \to \neg \psi) \cdots))$ . To complete the proof it is therefore sufficient to show:

$$\lor \models s_1(\vec{x}) \to (s_2(\vec{x}) \to (\cdots \to (s_n(\vec{x}) \to (t(\vec{x}) \to \neg t(\vec{x}))) \cdots)) \approx 1$$

implies:

$$\vee \models s_1(\vec{x}) \rightarrow (s_2(\vec{x}) \rightarrow (\cdots \rightarrow (s_n(\vec{x}) \rightarrow \neg t(\vec{x})) \cdots)) \approx 1.$$

Let  $A \in V$  and  $\vec{a} \in A$ . Since the polynomial reduct  $\langle A; \rightarrow, 1 \rangle$  is a MINIalgebra,  $(c \rightarrow (c \rightarrow b)) \rightarrow (c \rightarrow b) = 1$  for any  $b, c \in A$  by Theorem 3.1.4 and (2.36). Thus in particular  $t^{A}(\vec{a}) \rightarrow (t^{A}(\vec{a}) \rightarrow 0) \preceq t^{A}(\vec{a}) \rightarrow 0$  (where  $\preceq$  is the underlying quasiorder on the MINI-algebra polynomial reduct  $\langle A; \rightarrow, 1 \rangle$ ), whence repeated application of (the dual of) Lemma 2.1.12(2) shows:

$$s_1^{\mathbf{A}}(\vec{a}) \to \left(s_2^{\mathbf{A}}(\vec{a}) \to (\dots \to (s_n^{\mathbf{A}}(\vec{a}) \to (t^{\mathbf{A}}(\vec{a}) \to (t^{\mathbf{A}}(\vec{a}) \to 0))) \dots)\right) \preceq s_1^{\mathbf{A}}(\vec{a}) \to \left(s_2^{\mathbf{A}}(\vec{a}) \to (\dots \to (s_n^{\mathbf{A}}(\vec{a}) \to (t^{\mathbf{A}}(\vec{a}) \to 0)) \dots)\right).$$

Since  $s_1^{\mathbf{A}}(\vec{a}) \to (s_2^{\mathbf{A}}(\vec{a}) \to (\dots \to (s_n^{\mathbf{A}}(\vec{a}) \to (t^{\mathbf{A}}(\vec{a}) \to 0)))\dots)) = 1$ by hypothesis, we conclude  $s_1^{\mathbf{A}}(\vec{a}) \to (s_2^{\mathbf{A}}(\vec{a}) \to (\dots \to (s_n^{\mathbf{A}}(\vec{a}) \to (t^{\mathbf{A}}(\vec{a}) \to 0))\dots)) = 1$  (because  $[1]_{\Xi} = \{1\}$  by (the dual of) Theorem 2.1.14(3)), which implies the required implication holds.

By (i) and (ii), both (CN) and  $(RA_J)$  are rules of S(V, 1). Therefore  $\neg$  is a weak negation for S(V, 1), and the proof is complete.

**Remark 3.1.26.** In general, the proofs of Items (1)-(4) of Proposition 3.1.25 do not require the hypothesis of subtractivity. We have been unable to formulate proofs of Items (5)-(6) without this assumption; however, we have no proof that this additional condition is necessary. More generally, it is not known to what extent satisfaction of the identity  $1 \rightarrow x \approx x$  (where  $\rightarrow$  is a weak relative pseudocomplementation) is reflected in special properties of WBSO varieties: see Blok and Pigozzi [29, p. 365; Theorem 3.7].

Problem 3.1.27. Comprehensively characterise those properties of WBSO<sup>#</sup> varieties that do not extend to arbitrary WBSO varieties.

رك

Recall from [14, Definition XI§2.1] that a De Morgan algebra is an algebra  $(A_1, A_2)$  of two (2, 2, 1, 0, 0) such that the reduct  $(A_1, A_2)$  (0, 1) is a

 $\langle A; \land, \lor, \sim, 0, 1 \rangle$  of type  $\langle 2, 2, 1, 0, 0 \rangle$  such that the reduct  $\langle A; \land, \lor, 0, 1 \rangle$  is a bounded distributive lattice and moreover the following identities are satisfied:

$$\sim \sim x \approx x$$
 (3.10)

$$\sim (x \wedge y) \approx \sim x \vee \sim y$$
 (3.11)

$$\sim (x \vee y) \approx \sim x \wedge \sim y. \tag{3.12}$$

A Nelson algebra (or quasi-pseudo Boolean algebra in the terminology of Rasiowa [196, pp. 75 ff.]) is an algebra  $\langle A; \land, \lor, \rightarrow, \sim, 0, 1 \rangle$  of type  $\langle 2, 2, 2, 1, 0, 0 \rangle$ such that the following conditions are satisfied for all  $a, b, c \in A$  [209, Section 0]:

- N1. The reduct  $\langle A; \land, \lor, \sim, 0, 1 \rangle$  is a De Morgan algebra with greatest element 1, least element 0 and lattice ordering  $\leq$ ;
- N2. The relation  $\leq$  defined by  $a \leq b$  iff  $a \rightarrow b = 1$  is a quasiordering on A;

N3.  $a \wedge b \preceq c$  iff  $a \preceq b \rightarrow c$ ;

N4.  $a \leq b$  iff  $a \leq b$  and  $\sim b \leq \sim a$ ;

N5.  $a \leq c$  and  $b \leq c$  implies  $a \vee b \leq c$ ;

N6.  $a \leq b$  and  $a \leq c$  implies  $a \leq b \wedge c$ ;

N7.  $a \wedge \sim b \preceq \sim (a \rightarrow b)$  and  $\sim (a \rightarrow b) \preceq a \wedge \sim b$ ;

N8. 
$$\sim(a \rightarrow 0) \preceq a \text{ and } a \preceq \sim(a \rightarrow 0);$$

N9. 
$$a \wedge \sim a \preceq b$$
.

Nelson algebras were introduced by Rasiowa in [194] (under the name  $\mathcal{N}$ lattices) as the algebraic counterpart of a particular (non-axiomatic) extension of the intuitionistic sentential calculus called *constructive logic with strong negation*, which latter was introduced independently by Nelson in [174] and Markov in [157] in response to certain philosophical objections concerning the non-constructive nature of falsity in IPC (for a discussion see Rasiowa [195, Chapter XII] or Wójcicki [238, Section 5.3.0]; see also Vorob'ev [233, 234] and Thomason [215]). More particularly, by results of Rasiowa [195, Chapter XII] and Blok and Pigozzi [31, Section 5.2.1] constructive logic with strong negation is precisely the inherent assertional logic S(N, 1) of the class N of Nelson algebras. Since their introduction Nelson algebras have been studied by a number of authors, including Brignole [47], Rasiowa [195, Chapter V], Sendlewski [209] and, in a recent major study, by Viglizzo [232]. By Brignole [47] or Rasiowa [195, Theorem V§2.1] the class N is a variety; by Blok and Pigozzi [29, pp. 357-358] N is a WBSO variety with weak meet  $x \wedge y$ , weak relative pseudocomplementation  $x \rightarrow y$  and Gödel equivalence term:

 $x \Leftrightarrow y := (x \Rightarrow y) \land (y \Rightarrow x)$ 

where:

$$x \Rightarrow y := (x \to y) \land (\sim y \to \sim x).$$

Results due to Rasiowa [195, Theorem V§1.3] show also that the variety of Nelson algebras satisfies the identities:

$$x \to x \approx 1 \tag{3.13}$$

$$\mathbf{1} \to x \approx x, \tag{3.14}$$

whence  $\rightarrow$  witnesses both subtractivity and weak relative pseudocomplementation for N in the sense of Theorem 3.1.24. Thus N is a WBSO<sup>#</sup> variety.

There exist a number of open problems and erroneous communications in the literature concerning Nelson algebras qua weak Brouwerian semilattices with filter preserving operations [29, 34, 4]. In particular, in their paper [34] on the structure of varieties with EDPC, Blok and Pigozzi posed the following problem [34, Problem 7.4]: Does the variety of Nelson algebras have a commutative, regular TD term, or even a TD term? Notice that although N has EDPC (by Proposition 1.5.6, since it is a WBSO<sup>#</sup> variety) this problem is non-trivial, because EDPC in and of itself does not imply the existence of a term

e(x, y, z) of N such that both N  $\models e(x, x, z) \approx z$  and  $e^{\mathbf{A}}(a, b, c) = e^{\mathbf{A}}(a, b, d)$ if  $c \equiv d \pmod{\Theta^{\mathbf{A}}(a, b)}$  for any  $\mathbf{A} \in \mathbb{N}$  and  $a, b, c, d \in A$ . See Blok and Pigozzi [34, p. 570; Problem 7.1].

**Lemma 3.1.28.** The variety of Nelson algebras satisfies the following identities:

$$\sim x \to (x \to y) \approx 1$$
 (3.15)

$$x \wedge (x \to y) \approx x \wedge (\sim x \lor y) \tag{3.16}$$

$$x \to (y \to z) \approx y \to (x \to z)$$
 (3.17)

$$(x \wedge y) \to z \approx x \to (y \to z) \tag{3.18}$$

$$x \wedge (y \vee z) \approx (x \wedge y) \vee (x \wedge z) \tag{3.19}$$

$$x \to (y \land z) \approx (x \to y) \land (x \to z)$$
 (3.20)

$$(x \lor y) \to z \approx (x \to z) \land (y \to z) \tag{3.21}$$

$$(x \Rightarrow y) \rightarrow ((y \Rightarrow x) \rightarrow y) \approx (y \Rightarrow x) \rightarrow ((x \Rightarrow y) \rightarrow x).$$
 (3.22)

Proof. Identity (3.15) is proved in Rasiowa [195, Theorem V§1.3], while identity (3.16) is established in Brignole [47, Theorem 3]. Identities (3.17) and (3.18) are proved in Rasiowa [195, Theorem V§1.3]. Identity (3.19) follows trivially from (N1). Identity (3.20) is proved in Brignole [47, Theorem 3]. Identity (3.21) is established by Monteiro in [169, 170] (see also Viglizzo [232, (1.9), pp. 6–7]). Identity (3.22) is Theorem V§1.4 of Rasiowa [195].

**Lemma 3.1.29.** The variety of Nelson algebras satisfies the following identities:

$$x \to (\sim x \to y) \approx \mathbf{1} \tag{3.23}$$

$$x \to (x \to y) \approx x \to y \tag{3.24}$$

$$\sim (x \wedge y) \to z \approx (\sim x \to z) \wedge (\sim y \to z).$$
 (3.25)

*Proof.* Let A be a Nelson algebra and let  $a, b, c \in A$ . For (3.23), we have  $1 = -a \to (-a \to b) = a \to (-a \to b)$  by (3.15) and (3.10). For (3.24), we have  $a \to (a \to b) = (a \land a) \to b = a \to b$  by (3.18). For (3.25), we

\_ک

**Lemma 3.1.30.** The variety of Nelson algebras satisfies the following identities:

$$x \wedge (x \to y) \approx x \wedge \sim (x \wedge \sim y)$$
 (3.26)

$$x \to \sim (x \land \sim y) \approx x \to y \tag{3.27}$$

$$x \to \sim (x \land y) \approx x \to \sim y. \tag{3.28}$$

*Proof.* Let A be a Nelson algebra and let  $a, b \in A$ . By (3.11), (3.10) and (3.16) we have  $a \wedge \sim (a \wedge \sim b) = a \wedge (\sim a \vee \sim \sim b) = a \wedge (\sim a \vee b) = a \wedge (a \rightarrow b)$ , which establishes (3.26). For (3.27), put  $\alpha := a, \beta := a$  and  $\gamma := a \rightarrow b$ . We have:

Put  $\alpha := a$ ,  $\beta := a$  and  $\gamma := \sim a \lor b$ . We have:

a

$$a \to (a \land (\sim a \lor b))$$

$$= \alpha \to (\beta \land \gamma)$$

$$= (\alpha \to \beta) \land (\alpha \to \gamma) \qquad \text{by (3.20)}$$

$$= (a \to a) \land (a \to (\sim a \lor b))$$

$$= 1 \land (a \to (\sim a \lor b)) \qquad \text{by (3.13)}$$

$$= a \to (\sim a \lor a \lor b)$$

$$= a \to (\sim a \lor \sim b) \qquad \text{by (3.10)}$$

$$= a \rightarrow \sim (a \land \sim b)$$
 by (3.11).

For (3.28) just note  $a \to \sim (a \land b) = a \to \sim (a \land \sim \sim b) = a \to \sim b$  by (3.10) and (3.27).

**Lemma 3.1.31.** The variety of Nelson algebras satisfies the following identities:

$$(x \Leftrightarrow y) \to x \approx (x \Leftrightarrow y) \to y \tag{3.29}$$

$$x \to \sim y \approx x \to \sim (x \to y) \tag{3.30}$$

$$x \to (y \to z) \approx (x \to y) \to (x \to z)$$
 (3.31)

$$x \to (y \lor z) \approx x \to ((x \to y) \lor (x \to z)). \tag{3.32}$$

*Proof.* Let A be a Nelson algebra and let  $a, b, c \in A$ . For (3.29) we have:

$$(a \Leftrightarrow b) \to a = ((a \Rightarrow b) \land (b \Rightarrow a)) \to a$$
$$= (i \Rightarrow b) \to ((b \Rightarrow a) \to a) \qquad by (3.18)$$
$$= (b \Rightarrow a) \to ((a \Rightarrow b) \to a) \qquad by (3.17)$$
$$= (a \Rightarrow b) \to ((b \Rightarrow a) \to b) \qquad by (3.22)$$
$$= ((a \Rightarrow b) \land (b \Rightarrow a)) \to b \qquad by (3.18)$$

$$= (a \Leftrightarrow b) \rightarrow b.$$

For (3.30), put  $\alpha := a$  and  $\beta := a \rightarrow b$ . We have:

$$a \to \sim (a \to b) = \alpha \to \sim \beta$$
  
=  $\alpha \to \sim (\alpha \land \beta)$  by (3.28)  
=  $a \to \sim (a \land (a \to b)).$ 

Put  $\alpha := a \wedge \sim a$  and  $\beta := a \wedge b$ . We have:

$$a \to \sim (a \land (a \to b))$$
  
=  $a \to \sim (a \land (\sim a \lor b))$  by (3.16)

3.1. Subtractive Varieties with EDPI 25  

$$= a \rightarrow \sim ((a \land \sim a) \lor (a \land b)) \qquad \text{by (3.19)}$$

$$= a \rightarrow \sim (\alpha \lor \beta)$$

$$= a \to (\sim \alpha \land \sim \beta) \qquad \qquad \text{by (3.12)}$$

$$= a \rightarrow (\sim (a \land \sim a) \land \sim (a \land b)).$$

Put  $\alpha := a, \beta := \sim (a \wedge \sim a)$  and  $\gamma := \sim (a \wedge b)$ . We have:

$$a \rightarrow (\sim (a \land \sim a) \land \sim (a \land b))$$

$$= \alpha \rightarrow (\beta \land \gamma)$$

$$= (\alpha \rightarrow \beta) \land (\alpha \rightarrow \gamma) \qquad \text{by (3.20)}$$

$$= (a \rightarrow \sim (a \land \sim a)) \land (a \rightarrow \sim (a \land b))$$

$$= (a \rightarrow a) \land (a \rightarrow \sim (a \land b)) \qquad \text{by (3.27)}$$

$$= 1 \land (a \rightarrow \sim (a \land b)) \qquad \text{by (3.13)}$$

$$= a \rightarrow \sim (a \land b)$$

$$= a \rightarrow \sim b \qquad \text{by (3.28).}$$

For (3.31), put  $\alpha := a \rightarrow b$ ,  $\beta := a$  and  $\gamma := c$ . We have:

$$(a \to b) \to (a \to c)$$
  
=  $\alpha \to (\beta \to \gamma)$   
=  $\beta \to (\alpha \to \gamma)$  by (3.17)  
=  $(\beta \land \alpha) \to \gamma$  by (3.18)  
=  $(a \land (a \to b)) \to c$   
=  $(a \land (\alpha \land \alpha b)) \to c$  by (3.26)

Put  $\alpha := a, \beta := \sim (a \wedge \sim b)$  and  $\gamma := c$ . We have:

$$\begin{aligned} (a \wedge \sim (a \wedge \sim b)) &\to c \\ &= (\alpha \wedge \beta) \to \gamma \\ &= \alpha \to (\beta \to \gamma) \end{aligned}$$
 by (3.18)

 $= a \to (\sim (a \land \sim b) \to c).$ 

Put  $\alpha := a, \beta := \sim b$  and  $\gamma := c$ . We have:

Put  $\alpha := a, \beta := \sim a \rightarrow c$  and  $\gamma := b \rightarrow c$ . We have:

$$a \rightarrow ((\sim a \rightarrow c) \land (b \rightarrow c))$$
  
=  $\alpha \rightarrow (\beta \land \gamma)$   
=  $(\alpha \rightarrow \beta) \land (\alpha \rightarrow \gamma)$  by (3.20)  
=  $(a \rightarrow (\sim a \rightarrow c)) \land (a \rightarrow (b \rightarrow c))$   
=  $1 \land (a \rightarrow (b \rightarrow c))$  by (3.23)  
=  $a \rightarrow (b \rightarrow c)$ .

For (3.32), let  $\alpha := a$ ,  $\beta := a$  and  $\gamma := (a \rightarrow b) \lor (a \rightarrow c)$ . We have:

$$a \rightarrow ((a \rightarrow b) \lor (a \rightarrow c))$$
  
= 1 \lapha (a \rightarrow ((a \rightarrow c)))  
= (a \rightarrow a) \lapha (a \rightarrow ((a \rightarrow c))) by (3.13)  
= (a \rightarrow \beta) \lapha (a \rightarrow \beta) \lambda (a \rightarrow c))) by (3.20)  
= a \rightarrow (a \lambda ((a \rightarrow b) \lambda (a \rightarrow c))).

Put  $\alpha := a, \beta := a \rightarrow b$  and  $\gamma := a \rightarrow c$ . We have:

$$a \to (a \land ((a \to b) \lor (a \to c)))$$
  
=  $a \to (\alpha \land (\beta \lor \gamma))$   
=  $a \to ((\alpha \land \beta) \lor (\alpha \land \gamma))$  by (3.19)  
=  $a \to ((a \land (a \to b)) \lor (a \land (a \to c))).$ 

Put  $\alpha := a, \beta := \sim a \lor b$  and  $\gamma := \sim a \lor c$ . We have:

$$a \to ((a \land (a \to b)) \lor (a \land (a \to c)))$$
  
=  $a \to ((a \land (\sim a \lor b)) \lor (a \land (\sim a \lor c)))$  by (3.16)  
=  $a \to ((\alpha \land \beta) \lor (\alpha \land \gamma))$   
=  $a \to (\alpha \land (\beta \lor \gamma))$  by (3.19)  
=  $a \to (a \land (\sim a \lor b \lor \sim a \lor c)).$ 

Put  $\alpha := a, \beta := b \lor c$ . We have:

$$a \rightarrow (a \land (\sim a \lor b \lor \sim a \lor c))$$
  
=  $a \rightarrow (a \land (\sim a \lor (b \lor c)))$   
=  $a \rightarrow (\alpha \land (\sim \alpha \lor \beta))$   
=  $a \rightarrow (\alpha \land (\alpha \rightarrow \beta))$  by (3.16)  
=  $a \rightarrow (a \land (a \rightarrow (b \lor c))).$ 

Put  $\alpha := a, \beta := a$  and  $\gamma := a \rightarrow (b \lor c)$ . We have:

$$a \rightarrow (a \land (a \rightarrow (b \lor c)))$$
  
=  $\alpha \rightarrow (\beta \land \gamma)$   
=  $(\alpha \rightarrow \beta) \land (\alpha \rightarrow \gamma)$  by (3.20)  
=  $(a \rightarrow a) \land (a \rightarrow (a \rightarrow (b \lor c)))$   
=  $1 \land (a \rightarrow (a \rightarrow (b \lor c)))$  by (3.13)

$$= a \rightarrow (a \rightarrow (b \lor c))$$
  
=  $a \rightarrow (b \lor c)$  by (3.24)

Remark 3.1.32 (Added in proof). After obtaining the derivation of the identity (3.31) given in the proof of Lemma 3.1.31 we were made aware of the existence of Viglizzo [232, (1.11), p. 8], wherein a simpler derivation of (3.31) may be found.

**Theorem 3.1.33.** For any Nelson algebra A and  $a, b, c, d \in A$ ,

 $c \equiv d \pmod{\Theta^{\mathbf{A}}(a,b)}$  iff  $p^{\mathbf{A}}(a,b,c) = p^{\mathbf{A}}(a,b,d)$ 

where  $p(x, y, z) := (x \Leftrightarrow y) \rightarrow z$ . Thus the variety of Nelson algebras has EDPC. Moreover, p(x, y, z) is a commutative TD term for N; it is not regular.

*Proof.* To see the variety of Nelson algebras has EDPC, let A be a Nelson algebra and let  $\Phi := \{\langle g, g' \rangle : (a \Leftrightarrow b) \to g = (a \Leftrightarrow b) \to g'\}$ . Clearly  $\Phi$  is an equivalence relation. Let  $c, c', d, d' \in A$  and suppose  $c \equiv c' \pmod{\Phi}$  and  $d \equiv d' \pmod{\Phi}$ . To see  $\Phi$  is a congruence relation we show:

(i) 
$$c \wedge d \equiv c' \wedge d' \pmod{\Phi}$$
; (iii)  $c \to d \equiv c' \to d' \pmod{\Phi}$ ;  
(ii)  $c \vee d \equiv c' \vee d' \pmod{\Phi}$ ; (iv)  $\sim c \equiv \sim c' \pmod{\Phi}$ .

For (i), we have:

$$(a \Leftrightarrow b) \to (c \land d) = ((a \Leftrightarrow b) \to c) \land ((a \Leftrightarrow b) \to d) \qquad \text{by (3.20)}$$
$$= ((a \Leftrightarrow b) \to c') \land ((a \Leftrightarrow b) \to d')$$
$$= (a \Leftrightarrow b) \to (c' \land d') \qquad \text{by (3.20)}.$$

So  $c \wedge d \equiv c' \wedge d' \pmod{\Phi}$ . For (ii), we have:

$$(a \Leftrightarrow b) \to (c \lor d) = (a \Leftrightarrow b) \to$$
$$(((a \Leftrightarrow b) \to c) \lor ((a \Leftrightarrow b) \to d)) \quad \text{by (3.32)}$$
$$= (a \Leftrightarrow b) \to$$

$$\begin{pmatrix} ((a \Leftrightarrow b) \to c') \lor ((a \Leftrightarrow b) \to d') \end{pmatrix}$$
  
=  $(a \Leftrightarrow b) \to (c' \lor d')$  by (3.32)

So  $c \lor d \equiv c' \lor d' \pmod{\Phi}$ . For (iii), we have:

$$(a \Leftrightarrow b) \to (c \to d) = ((a \Leftrightarrow b) \to c) \to ((a \Leftrightarrow b) \to d) \qquad \text{by (3.31)}$$
$$= ((a \Leftrightarrow b) \to c') \to ((a \Leftrightarrow b) \to d')$$
$$= (a \Leftrightarrow b) \to (c' \to d') \qquad \text{by (3.31)}.$$

So  $c \to d \equiv c' \to d' \pmod{\Phi}$ . For (iv), we have:

$$(a \Leftrightarrow b) \to \sim c = (a \Leftrightarrow b) \to \sim ((a \Leftrightarrow b) \to c) \qquad \text{by (3.30)}$$
$$= (a \Leftrightarrow b) \to \sim ((a \Leftrightarrow b) \to c')$$
$$= (a \Leftrightarrow b) \to \sim c' \qquad \text{by (3.30)}.$$

Thus  $\sim c \equiv \sim c' \pmod{\Phi}$  and  $\Phi$  is a congruence relation. Moreover  $\langle a, b \rangle \in \Phi$ by (3.29), so  $\Theta^{\mathbf{A}}(a, b) \subseteq \Phi$ . Conversely, if  $c \equiv d \pmod{\Phi}$  then  $c = 1 \rightarrow c =$  $(a \Leftrightarrow a) \rightarrow c \equiv_{\Theta^{\mathbf{A}}(a,b)} (a \Leftrightarrow b) \rightarrow c = (a \Leftrightarrow b) \rightarrow d \equiv_{\Theta^{\mathbf{A}}(a,b)} (a \Leftrightarrow a) \rightarrow$  $d = 1 \rightarrow d = d$  since  $\rightarrow$  and  $\Leftrightarrow$  are respectively subtractive weak relative pseudocomplementation and Gödel equivalence terms for N. So  $\Phi \subseteq \Theta^{\mathbf{A}}(a, b)$ and the terms p(x, y, z), p(x, y, w) witness EDPC for N.

Let  $a, b, c \in A$ . To see p(x, y, z) is a TD term, it is sufficient to note  $p^{A}(a, a, c) = (a \Leftrightarrow a) \rightarrow c = 1 \rightarrow c = c$ , just because  $\Leftrightarrow$  is a Gödel equivalence term for N and  $\rightarrow$  is a subtractive weak relative pseudocomplementation. To see p(x, y, z) is commutative, let  $a, b, a', b', c \in A$ . We have:

$$p^{\mathbf{A}}(a, b, p^{\mathbf{A}}(a', b', c)) = (a \Leftrightarrow b) \rightarrow ((a' \Leftrightarrow b') \rightarrow c)$$
$$= (a' \Leftrightarrow b') \rightarrow ((a \Leftrightarrow b) \rightarrow c) \qquad \text{by (3.17)}$$
$$= p^{\mathbf{A}}(a', b', p^{\mathbf{A}}(a, b, c)).$$

Finally, an easy inspection of the unique (to within isomorphism) 3-element Nelson algebra establishes that p(x, y, z) is not regular.

**Remark 3.1.34.** Because of (3.18),  $p(x, y, z) := (x \Rightarrow y) \rightarrow ((y \Rightarrow x) \rightarrow z)$  is also a commutative (but not regular) TD term for the variety of Nelson algebras; compare Blok and Pigozzi [34, Corollary 5.2(i)].

Recall from [34, Corollary 5.2] that dual Brouwerian semilattices possess-two distinct TD terms: the commutative (but not regular) TD term (z \* (x \* y)) \* (y \* x) and the commutative and regular TD term  $z \lor ((x * y) \lor (y * x))$ . Thus the existence of a commutative (but not regular) TD term for a variety V need not preclude the existence of a commutative and regular TD term for V.

**Problem 3.1.35.** Does the variety of Nelson algebras possess a commutative, regular TD term?

**Remark 3.1.36.** In view of work due to Blok and Pigozzi [34, Section 5] an obvious candidate for a commutative, regular TD term for the variety of Nelson algebras is the term  $p(x, y, z) := (x \Leftrightarrow y) \land z$ . To see p(x, y, z) is not a TD term for N, consider the following 6-element algebra A:

| $\wedge^{\mathbf{A}}$      | 0 | a | b | С | d | 1 | $\vee^{\mathbf{A}}$ | 0 | a          | b      | с | d | 1 |
|----------------------------|---|---|---|---|---|---|---------------------|---|------------|--------|---|---|---|
| 0                          | 0 | 0 | 0 | 0 | 0 | 0 | 0                   | 0 | a          | -<br>b | с | d | 1 |
| a                          | 0 | a | a | a | a | a | а                   | a | a          | b      | с | d | 1 |
| b                          | 0 | a | b | с | d | b | b                   | b | b          | b      | b | b | 1 |
| с                          | 0 | a | с | с | a | с | c                   | c | с          | b      | С | b | 1 |
| d                          | 0 | a | d | a | d | d | d                   | d | d          | b      | b | d | 1 |
| 1                          | 0 | a | b | с | d | 1 | 1                   | 1 | 1          | 1      | 1 | 1 | 1 |
| ·                          |   |   |   |   |   |   |                     | • |            |        |   |   |   |
| $\rightarrow^{\mathbf{A}}$ | 0 | a | b | с | d | 1 |                     |   | $\sim^{A}$ |        |   |   |   |
| 0                          | 1 | 1 | 1 | 1 | 1 | 1 | •                   |   | 0          | ]]     | L |   |   |
| a                          | 1 | 1 | 1 | 1 | 1 | 1 |                     |   | a          | 1      | 6 |   |   |
| b                          | a | a | 1 | с | d | 1 |                     |   | b          | 1      | 1 |   |   |
| с                          | d | d | 1 | 1 | d | 1 |                     |   | с          | 6      | ł |   |   |
|                            |   |   |   |   |   |   |                     |   |            |        |   |   |   |
| d                          | c | С | 1 | С | 1 | 1 |                     |   | d          | (      | : |   |   |

An easy sequence of checks shows A is a subdirectly irreducible Nelson algebra and that the monolith of A is the congruence  $\mu$  induced by the parti-





259



tion  $\{\{0, a\}, \{1, b\}, \{c\}, \{d\}\}\}$ : see Figure 3.3 (note that in Figure 3.3(b), the congruences of **A** are represented by their corresponding partitions (with all parentheses dropped)). Observe now that  $0 \equiv a \pmod{\mu(1, b)}$ , but  $(1 \Leftrightarrow b) \land 0 = b \land 0 = 0 \neq a = b \land a = (1 \Leftrightarrow b) \land a$ . Thus p(x, y, z) does not even witness EDPC for **A**, and so in particular cannot be a commutative, regular TD term for N.

In [29, p. 361] Blok and Pigozzi assert without proper proof that the variety of Nelson algebras is not congruence permutable. This assertion is corrected in [34, p. 606], where Blok and Pigozzi (citing an unpublished result of Idziak) announce that the variety of Nelson algebras is congruence permutable. They do not provide a proof, and in particular do not give a Mal'cev term.

**Lemma 3.1.37.** The variety of Nelson algebras satisfies the following identities:

$$\sim (x \to y) \to x \approx 1$$
 (3.33)

$$\sim (x \to y) \to z \approx x \to (\sim y \to z)$$
 (3.34)

$$1 \Rightarrow x \approx x. \tag{3.35}$$

*Proof.* Let A be a Nelson algebra and let  $a, b, c \in A$ . For (3.33), observe that  $\sim (a \rightarrow b) \preceq a \land \sim b$  by (N7). But  $a \land \sim b \leq a$ , which implies  $a \land \sim b \preceq a$ 

by (N4). Thus  $\sim (a \to b) \preceq a \land \sim b \preceq a$ , whence  $\sim (a \to b) \preceq a$  by transitivity. By (N2) we conclude  $\sim (a \to b) \to a = 1$ . For (3.34), put  $\alpha := \sim (a \to b), \beta := a$  and  $\gamma := c$ . We have:

$$\sim (a \to b) \to c = 1 \to (\sim (a \to b) \to c)$$
  
=  $(\sim (a \to b) \to a) \to (\sim (a \to b) \to c)$  by (3.33)  
=  $(a \to \beta) \to (a \to \gamma)$   
=  $a \to (\beta \to \gamma)$  by (3.31)  
=  $\beta \to (a \to \gamma)$  by (3.17)  
=  $a \to (\sim (a \to b) \to c)$ .

Put  $\alpha := a, \beta := \sim (a \to b)$  and  $\gamma := c$ . We have:

$$a \to (\sim (a \to b) \to c) = \alpha \to (\beta \to \gamma)^{+}$$
$$= (\alpha \to \beta) \to (\alpha \to \gamma) \qquad \text{by (3.31)}$$
$$= (a \to \sim (a \to b)) \to (a \to c)$$
$$= (a \to \sim b) \to (a \to c) \qquad \text{by (3.30)}$$

 $= a \rightarrow (\sim b \rightarrow c)$  by (3.31).

For (3.35), observe  $1 \Rightarrow a = (1 \rightarrow a) \land (\sim a \rightarrow \sim 1) = a \land (\sim a \rightarrow \sim 1)$ by (3.14). Therefore to see  $1 \Rightarrow a = a$  it is sufficient to show  $a \leq \sim a \rightarrow \sim 1$ , or equivalently (by (N4), (N2)) both  $a \rightarrow (\sim a \rightarrow \sim 1) = 1$  and  $\sim (\sim a \rightarrow \sim 1) \rightarrow \sim a = 1$ . Now  $a \rightarrow (\sim a \rightarrow \sim 1) = 1$  by (3.23). Put  $\alpha := \sim a$ ,  $\beta := \sim 1$  and  $\gamma := \sim a$ . Then  $\sim (\sim a \rightarrow \sim 1) \rightarrow \sim a = \sim (\alpha \rightarrow \beta) \rightarrow \gamma = \alpha \rightarrow (\sim \beta \rightarrow \gamma) = \sim a \rightarrow (\sim \sim 1 \rightarrow \sim a) = \sim a \rightarrow (1 \rightarrow \sim a) = \sim a \rightarrow \sim a = 1$ by (3.34),(3.10) and (3.13). Thus  $a \leq \sim a \rightarrow \sim 1$  as required.

**Lemma 3.1.38.** Let A be a Nelson algebra. The following inequalities are identically satisfied for any  $a, b \in A$ :

 $a \Rightarrow b \le a \to b \tag{3.36}$ 

$$a \Rightarrow b \le \sim b \to \sim a. \tag{3.37}$$

Thus the following equations are identically satisfied for any  $a, b \in A$ :

$$(a \Rightarrow b) \rightarrow (a \rightarrow b) = 1 \tag{3.38}$$

$$\sim (a \to b) \to \sim (a \Rightarrow b) = 1$$
 (3.39)

$$(a \Rightarrow b) \rightarrow \sim (b \rightarrow \sim a) = 1$$
 (3.40)

$$\sim (\sim b \rightarrow \sim a) \rightarrow \sim (a \Rightarrow b) = 1.$$
 (3.41)

Proof. Let A be a Nelson algebra and let  $a, b \in A$ . We have  $(a \Rightarrow b) \land (a \rightarrow b) = ((a \rightarrow b) \land (\sim b \rightarrow \sim a)) \land (a \rightarrow b) = (a \rightarrow b) \land (\sim b \rightarrow \sim a) = a \Rightarrow b$ , which is sufficient to establish (3.36). Also  $(a \Rightarrow b) \land (\sim b \rightarrow \sim a) = ((a \rightarrow b) \land (\sim b \rightarrow \sim a)) \land (\sim b \rightarrow \sim a) = (a \rightarrow b) \land (\sim b \rightarrow \sim a) = a \Rightarrow b$ , which is sufficient to establish (3.37). Equations (3.38)-(3.41) now follow immediately by (N4), (N2).

**Lemma 3.1.39.** Let A be a Nelson algebra. The following inequalities are identically satisfied for any  $a, b \in A$ :

$$a \preceq (a \Rightarrow b) \Rightarrow b \tag{3.42}$$

 $\sim ((a \Rightarrow b) \Rightarrow b) \preceq \sim a.$  (3.43)

Thus the following inequality is identically satisfied for any  $a, b \in A$ :

$$a \le (a \Rightarrow b) \Rightarrow b. \tag{3.44}$$

*Proof.* Let A be a Nelson algebra and let  $a, b \in A$ . For (3.42) observe first that:

$$a \rightarrow ((a \Rightarrow b) \rightarrow b)$$
  
=  $(a \Rightarrow b) \rightarrow (a \rightarrow b)$  by (3.17)  
= 1 by (3.38). (3.45)

Also, put  $\alpha := a$ ,  $\beta := b$  and  $\gamma := \sim (a \Rightarrow b)$  and observe that:

$$a \rightarrow (\sim b \rightarrow \sim (a \Rightarrow b))$$
  
=  $\alpha \rightarrow (\sim \beta \rightarrow \gamma)$   
=  $\sim (\alpha \rightarrow \beta) \rightarrow \gamma$  by (3.34)  
=  $1$  by (3.39). (3.46)

Now put  $\alpha := a, \beta := (a \Rightarrow b) \rightarrow b$  and  $\gamma := \sim b \rightarrow \sim (a \Rightarrow b)$ . We have:

$$a \rightarrow ((a \Rightarrow b) \Rightarrow b)$$

$$= a \rightarrow (((a \Rightarrow b) \rightarrow b) \land (\sim b \rightarrow \sim (a \Rightarrow b)))$$

$$= \alpha \rightarrow (\beta \land \gamma)$$

$$= (\alpha \rightarrow \beta) \land (\alpha \rightarrow \gamma) \qquad \text{by (3.20)}$$

$$= (a \rightarrow ((a \Rightarrow b) \rightarrow b)) \land (a \rightarrow (\sim b \rightarrow \sim (a \Rightarrow b)))$$

$$= 1 \land 1 \qquad \text{by (3.45),(3.46)}$$

$$= 1$$

which (in view of (N2)) establishes (3.42). For (3.43), put  $\alpha := a \Rightarrow b, \beta := b$ and  $\gamma := \sim a$  and observe first that:

$$\sim ((a \Rightarrow b) \rightarrow b) \rightarrow \sim a \qquad -$$

$$= \sim (\alpha \rightarrow \beta) \rightarrow \gamma$$

$$= \alpha \rightarrow (\sim \beta \rightarrow \gamma) \qquad \text{by (3.34)}$$

$$= (a \Rightarrow b) \rightarrow (\sim b \rightarrow \sim a)$$

$$= 1 \qquad \text{by (3.40).} \quad (3.47)$$

Also, put  $\alpha := \sim b$ ,  $\beta := \sim (a \Rightarrow b)$  and  $\gamma := \sim a$  and observe that:

$$(\sim (\sim b \rightarrow \sim (a \Rightarrow b)) \rightarrow \sim a)$$

$$= \sim (\alpha \rightarrow \beta) \rightarrow \gamma$$

$$= \alpha \rightarrow (\sim \beta \rightarrow \gamma)$$

$$= \alpha \rightarrow (\sim \beta \rightarrow \gamma)$$

$$= \sim b \rightarrow (\sim \sim (a \Rightarrow b) \rightarrow \sim a)$$

$$= \sim b \rightarrow ((a \Rightarrow b) \rightarrow \sim a)$$

$$= (a \Rightarrow b) \rightarrow (\sim b \rightarrow \sim a)$$

$$= 1$$

$$by (3.40). (3.48)$$

Now put  $\alpha := (a \Rightarrow b) \rightarrow b$ ,  $\beta := \sim b \rightarrow \sim (a \Rightarrow b)$  and  $\gamma := \sim a$ . We have:

$$\begin{array}{l} \sim ((a \Rightarrow b) \Rightarrow b) \rightarrow \sim a \\ = \sim (((a \Rightarrow b) \rightarrow b) \land \\ (\sim b \rightarrow \sim (a \Rightarrow b))) \rightarrow \sim a \\ = \sim (\alpha \land \beta) \rightarrow \gamma \\ = (\sim \alpha \rightarrow \gamma) \land (\sim \beta \rightarrow \gamma) \qquad \qquad \text{by (3.25)} \\ = (\sim ((a \Rightarrow b) \rightarrow b) \rightarrow \sim a) \land \\ (\sim (\sim b \rightarrow \sim (a \Rightarrow b)) \rightarrow \sim a) \\ = 1 \land 1 \qquad \qquad \qquad \text{by (3.47),(3.48)} \\ = 1 \end{array}$$

which (in view of (N2)) establishes (3.43). The remaining assertion of the lemma now follows immediately by (N4).

**Theorem 3.1.40.** (Idziak) The variety of Nelson algebras is congruence permutable. A Mal'cev term witnessing congruence permutability is:

$$p(x, y, z) := ((x \Rightarrow y) \Rightarrow z) \land ((z \Rightarrow y) \Rightarrow x).$$

*Proof.* Let A be a Nelson algebra and  $a, b \in A$ . We have:

$$p^{A}(a, a, b) = ((a \Rightarrow a) \Rightarrow b) \land ((b \Rightarrow a) \Rightarrow a)$$
  
= (1 \Rightarrow b) \langle ((b \Rightarrow a) \Rightarrow a)  
= b \langle ((b \Rightarrow a) \Rightarrow a)  
= b by (3.35)  
by (3.44)

and:

$$p^{\mathbf{A}}(a, b, b) = ((a \Rightarrow b) \Rightarrow b) \land ((b \Rightarrow b) \Rightarrow a)$$
  
=  $((a \Rightarrow b) \Rightarrow b) \land (1 \Rightarrow a)$   
=  $((a \Rightarrow b) \Rightarrow b) \land a$  by (3.35)  
=  $a$  by (3.44).

## Corollary 3.1.41. The variety of Nelson algebras is arithmetical.

*Proof.* Congruence permutability is clear in view of the preceding theorem. Since the variety of Nelson algebras has EDPC (by Theorem 3.1.33), it is also congruence distributive (by Theorem 1.5.2(3)). Thus N is arithmetical as asserted.

In [29, p. 361] Blok and Pigozzi erroneously assert that the variety of Nelson algebras does not have a QD term since it is not congruence permutable. The following theorem corrects this assertion.

**Theorem 3.1.42.** The variety of Nelson algebras is a congruence permutable WBSO variety. A QD term for N is:

$$q(x, y, z, w) := p(e(x, y, z), e(x, y, w), w)$$

where:

$$e(x, y, z) := (x \Leftrightarrow y) \rightarrow z$$

is the commutative TD term of Theorem 3.1.35 and:

$$p(x, y, z) := ((x \Rightarrow y) \Rightarrow z) \land ((z \Rightarrow y) \Rightarrow x)$$

is the Mal'cev term of Theorem 3.1.40.

*Proof.* The first assertion is clear from previous remarks and Theorem 3.1.40. The second assertion follows immediately from Theorem 3.1.33, Theorem 3.1.40 and Proposition 1.5.11(1).

**Corollary 3.1.43.** The class  $N_{SS}$  of all semisimple Nelson algebras, axiomatised relative to N by the identity:

 $((x \to y) \to x) \to x \approx 1,$ 

is a discriminator variety. A discriminator term for N<sub>SS</sub> is given by q(x, y, z, x), where q(x, y, z, w) is the QD term of Theorem 3.1.42.

*Proof.* By Viglizzo [232, Theorem 4.2] the class  $N_{SS}$  of all semisimple Nelson algebras is a variety, axiomatised relative to N by the identity:

 $((x \rightarrow y) \rightarrow x) \rightarrow x \approx 1.$ 

Since N<sub>SS</sub> is a semisimple congruence permutable WBSO variety, it is a discriminator variety with discriminator term q(x, y, z, x), where q(x, y, z, w) is the QD term of Theorem 3.1.42.

By an *implicative subreduct* of a Nelson algebra  $\langle A; \land, \lor, \rightarrow, \sim, 0, 1 \rangle$  we mean a subalgebra of the reduct  $\langle A; \rightarrow, 1 \rangle$ . If V is a variety of Nelson algebras, we denote the class of implicative subreducts of V by  $S(V^{\{\rightarrow,1\}})$ . The following theorem is an easy modification of a result due to Blok and Pigozzi [34].

**Theorem 3.1.44.** (cf. [34, Corollary 5.3]) For any variety  $\vee$  of Nelson algebras, the class  $S(\vee^{\{\rightarrow,1\}})$  of implicative subreducts of  $\vee$  is a variety.

*Proof.* We prove  $S(V^{\{\rightarrow,1\}})$  is a variety by showing it is closed under S, P and H. It is closed under S by definition, and it is easy to see it is closed under P. So we have only to show  $HS(V^{\{\rightarrow,1\}}) \subseteq S(V^{\{\rightarrow,1\}})$ .

Let  $\mathbf{A} := \langle A; \to, 1 \rangle \in \mathbf{S}(\mathsf{V}^{\{\to,1\}})$  and let  $\mathbf{B} := \langle B; \land, \lor, \to, \sim, 0, 1 \rangle \in \mathsf{V}$  with  $\mathbf{A} \in \mathbf{S}(\langle B; \to, 1 \rangle)$ . Let  $\Phi$  be a congruence on  $\mathbf{A}$ , and let  $F := [1]_{\Phi}$ . By Lemma 1.5.4 (see also Rasiowa [195, Theorem V§4.4, Theorem V§4.3]) we have that F is an implicative filter. That is,  $1 \in F$  and F has the detachment property:  $1 \in F$  trivially, and, if  $c \equiv 1 \pmod{\Phi}$  and  $c \to d \equiv 1 \pmod{\Phi}$ , then:

$$d = 1 \rightarrow d \equiv_{\Phi} c \rightarrow d \equiv_{\Phi} 1.$$

Let G be the filter on B generated by F, and let  $\Theta(G)$  be the congruence on B such that  $G = [1]_{\Theta(G)}$ ; of course,  $\Theta(G)$  exists and is unique, just because of Proposition 1.5.5. By Proposition 1.5.5 (see also Rasiowa [195, Theorem V§4.5])  $\Theta(G) = \{\langle a, b \rangle \in B \times B : a \Rightarrow b, b \Rightarrow a \in G\}$ . Moreover, by (3.17) and Rasiowa [195, Theorem V§4.8] we have that  $a \equiv b \pmod{\Theta(G)}$ iff:

$$c_1 \to (c_2 \to (\dots \to (c_k \to (a \Rightarrow b))) \dots) = 1$$
 and  
 $c_1 \to (c_2 \to (\dots \to (c_k \to (b \Rightarrow a))) \dots) = 1$ 

for some  $c_1, \ldots, c_k \in F$ . If  $a, b \in A$ , then  $a \Rightarrow b, b \Rightarrow a \in F$  follows from the fact that F has the detachment property. Thus by Theorem 3.1.33, Remark 3.1.34 and the definition of F,

$$a = 1 \rightarrow (1 \rightarrow a)$$
  

$$\equiv_{\Phi} (a \Rightarrow b) \rightarrow ((b \Rightarrow a) \rightarrow a)$$
  

$$= (a \Rightarrow b) \rightarrow ((b \Rightarrow a) \rightarrow b)$$
  

$$\equiv_{\Phi} 1 \rightarrow (1 \rightarrow b)$$
  

$$= b.$$

We have shown  $\Theta(G) \cap A^2 \subseteq \Phi$ . Conversely, if  $a, b \in A$  and  $a \equiv b \pmod{\Phi}$ , then  $a \Rightarrow b \equiv_{\Phi} 1 \equiv_{\Phi} b \Rightarrow a$  by Rasiowa [195, Theorem V§4.5]. Hence  $a \Rightarrow b, b \Rightarrow a \in F \subseteq G$  and consequently  $a \equiv b \pmod{\Theta(G)}$ . Thus  $\Phi \subseteq \Theta(G) \cap A^2$ . So  $\Theta(G) \cap A^2 = \Phi$ , and thus  $\mathbf{A}/\Phi$  is isomorphic to a subalgebra of the implicative reduct of  $\mathbf{B}/\Theta(G)$ . Hence  $\mathbf{A}/\Phi \in \mathbf{S}(\vee^{\{\to,1\}})$ , which completes

the proof of the corollary.

**Problem 3.1.45.** Axiomatise the variety  $S(N^{\{\rightarrow,1\}})$ . Is  $S(N^{\{\rightarrow,1\}})$  finitely axiomatisable?

Let V be a variety of Nelson algebras. By the previous theorem, the class  $S(V^{\{\rightarrow,1\}})$  of implicative subreducts of V is a variety. Moreover, by Rasiowa [195, Theorem V§1.3] we have that  $N \models (1.45)-(1.48)$ , so  $S(V^{\{\rightarrow,1\}})$  is a variety of MINI-algebras (this observation also follows from Theorem 3.1.44 and Proposition 3.1.24, since  $\rightarrow$  is a subtractive weak relative pseudocomplementation for N). Consider now the following 4-element Nelson algebra A:

| $\wedge^{\mathbf{A}}$ | 0 | a | b | 1 | $\vee^{\mathbf{A}}$ | 0 | a | b | 1 |   | $\rightarrow^{\mathbf{A}}$ | 0 | a | b | .1 | $\sim^{\mathbf{A}}$ |   |
|-----------------------|---|---|---|---|---------------------|---|---|---|---|---|----------------------------|---|---|---|----|---------------------|---|
| 0                     | 0 | 0 | 0 | 0 | 0                   | 0 | a | b | 1 | • | 0                          | 1 | 1 | 1 | 1  | 0                   | 1 |
| a                     | 0 | a | b | a | a                   | a | a | a | 1 |   | a                          | b | 1 | b | 1  | a                   | b |
| b                     | 0 | b | b | b | b                   | b | a | b | 1 |   | b                          | 1 | 1 | 1 | 1  | Ь                   | a |
| 1                     | 0 | a | b | 1 | 1                   | 1 | 1 | 1 | 1 |   | 1                          | 0 | a | b | 1  | 1                   | 0 |

Since  $((a \to 0) \to 0) \to 0 = (b \to 0) \to 0 = 1 \to 0 = 0 \neq 1$ , we conclude that  $\mathbf{A} \not\models ((x \to y) \to x) \to x \approx 1$ . Therefore in general  $\mathbf{S}(\vee^{\{\to,1\}}) \not\models ((x \to y) \to x) \to x \approx 1$ . From these remarks it clearly follows that the varieties of {commutative, positive implicative, implicative}-pre-BCK-algebras and the variety of implicative BCS-algebras do not exhaust the naturally occurring subvarieties of pre-BCK-algebras that may be of interest in universal algebra and algebraic logic.

# 3.2 Binary and Dual Binary Discriminator Varieties

Recall from Example 2.3.11 that on any binary discriminator algebra the binary discriminator is precisely implicative BCS difference, and thus that any member of a binary discriminator variety has a canonical implicative BCSalgebra polynomial reduct. This observation calls for a study of the role played by implicative BCS-algebras in binary discriminator varieties. More generally,

this remark (in conjunction with the remarks of  $\S1.7.20$ ) calls for the study of bands in dual binary discriminator varieties.

Dual binary discriminator varieties are considered in §3.2.1. It is shown that the variety of left normal bands with zero coincides with the pure dual binary discriminator variety, namely the variety generated by the class of all dual binary discriminator algebras  $\langle A; h, 0 \rangle$ , where h is the dual binary discriminator on A and 0 is a nullary operation. We also give a semigroup-theoretic characterisation of dual binary discriminator varieties in terms of left normal bands with zero.

In §3.2.6 binary discriminator varieties proper are studied. It is shown that the variety of implicative BCS-algebras coincides with the pure binary discriminator variety, namely the variety generated by the class of all binary discriminator algebras  $\langle A; b, 0 \rangle$ , where b is the binary discriminator on A and 0 is a nullary operation. We prove that any member A of a binary discriminator variety V has an implicative BCS-algebra polynomial reduct whose iBCS-ideals coincide with the V-ideals of A. We characterise binary discriminator varieties in ideal-theoretic terms: a pointed variety is a binary discriminator variety iff it is subtractive with EDPI and is generated by a class of ideal simple algebras. A characterisation of binary discriminator varieties in the spirit of Agliano and Ursini's characterisation of subtractive varieties with EDPI is also given. The results are illustrated with some examples.

In §3.2.22 attention is focussed on point regular binary discriminator varieties. We prove two results that together show a pointed variety is a point regular binary discriminator variety iff it is a 'pointed' fixedpoint discriminator variety. In the main result of the section, the 'pointed' fixedpoint discriminator varieties are characterised: a pointed variety is a 'pointed' fixedpoint discriminator variety iff it is ideal determined, semisimple and has EDPC. The results give a partial answer to a question of Blok and Pigozzi.

The relationship between binary and pointed ternary discriminator varieties is considered in §3.2.32. It is shown that a pointed variety is a pointed ternary discriminator variety iff it is a congruence permutable point regular binary

discriminator variety. We also give an (almost) trivial syntactic criterion both necessary and sufficient for a point regular binary discriminator variety to be a pointed ternary discriminator variety.

**3.2.1.** Dual Binary Discriminator Varieties. In [58, Example (2), p. 242] Chajda, Halaš and Rosenberg observed that the simplest non-trivial example of a dual binary discriminator algebra is 2, the one-element semilattice with a zero adjoined. Since 2 is, to within isomorphism, the only subdirectly irreducible meet semilattice with zero, it follows from this example that the variety of meet semilattices with zero is a dual binary discriminator variety. This result may be regarded as a specialisation to meet semilattices with zero of the following theorem, in the statement of which the *pure dual binary discriminator variety* (in symbols, PdBD) denotes the variety generated by the class of all dual binary discriminator algebras  $\langle A; h, 0 \rangle$ , where h is the dual binary discriminator function on A and 0 is a nullary operation.

**Theorem 3.2.2.** The variety of left normal bands with zero coincides with the pure dual binary discriminator variety.

Proof. Let K denote the class of all dual binary discriminator algebras  $\langle A; h, 0 \rangle$ where h is the dual binary discriminator function on A and 0 is a nullary operation. By definition of the dual binary discriminator and Example 1.3.20 every member of K is a left normal band with zero, so  $K \subseteq \ln B_0$ , the variety of left normal bands with zero. Hence the variety V(K) generated by K, namely PdBD, is a subvariety of  $\ln B_0$ . Conversely, the three-element left normal band with zero  $3_L$  is a member of K; since an easy consequence of Corollary 1.3.19 shows  $3_L$  generates  $\ln B_0$  as a variety we have that  $\ln B_0 = V(3_L) \subseteq V(K) =$ PdBD.

Given a band with zero A, we say A is  $\mathcal{D}$ -simple or primitive if the only  $\mathcal{D}$ -equivalence classes of A are  $\{0\}$  and  $A - \{0\}$ . Notice that A is  $\mathcal{D}$ -simple iff  $A/\mathcal{D}$  is isomorphic to 2, the one element semilattice with a zero adjoined. By the proof of Theorem 3.2.2, any dual binary discriminator algebra (considered as a band with zero) is  $\mathcal{D}$ -simple, which observation suggests the following semigroup-theoretic characterisation of dual binary discriminator varieties (implicit in [58, Section 5]). In the statement of the theorem and in the sequel

the dual binary discriminator [term] is denoted  $\sqcap$ ; of course, this change in notation is legitimised by Theorem 3.2.2.

**Theorem 3.2.3.** (cf. [58, Theorem 5.2]) For a variety V the following are equivalent:

4

- 1. V is a dual binary discriminator variety;
- 2. There exists a binary term  $x \sqcap y$  of  $\lor$  such that for any  $A \in \lor$ , the polynomial reduct  $\langle A; \sqcap^A, 0 \rangle$  is a left normal band with zero, and any one of the following conditions is satisfied:
  - (a) V is generated by a class of algebras K such that for each A ∈ K, the left normal band with zero polynomial reduct (A; □<sup>A</sup>, 0) is D<sub>(A; □<sup>A</sup>, 0)</sub>- simple;
  - (b) V is generated by a class of algebras K such that for each  $A \in K$ , the maximal semilattice homomorphic image  $\langle A; \sqcap^A, 0 \rangle / \mathcal{D}_{\langle A; \sqcap^A, 0 \rangle}$ of the left normal band with zero polynomial reduct  $\langle A; \sqcap^A, 0 \rangle$  is isomorphic to 2, the one-element semilattice with a zero adjoined;

*Proof.* Let V be a variety with 0. By previous remarks the equivalence  $(2)(a) \Leftrightarrow (2)(b)$  is clear, so it only remains to prove the equivalence  $(1) \Leftrightarrow (2)(a)$ .

 $(1) \Rightarrow (2)(a)$  Suppose V is a dual binary discriminator variety with dual binary discriminator term  $x \sqcap y$  generated by a class K of dual binary discriminator algebras. By definition of the dual binary discriminator and Example 1.3.20, the polynomial reduct  $\langle A; \sqcap^{\mathbf{A}}, 0 \rangle$  of each  $\mathbf{A} \in \mathsf{K}$  is a left normal band with zero whose only  $\mathcal{D}_{\langle A; \sqcap^{\mathbf{A}}, 0 \rangle}$ -equivalence classes are  $\{0\}$  and  $A - \{0\}$ . For each  $\mathbf{A} \in \mathsf{K}$ , therefore,  $\langle A; \sqcap^{\mathbf{A}}, 0 \rangle$  is primitive. Upon recalling that the class of left normal bands with zero is equationally definable, we may also infer the polynomial reduct  $\langle B; \sqcap^{\mathbf{B}}, 0 \rangle$  of any  $\mathbf{B} \in \mathsf{V}$  is a left normal band with zero, just because the identities satisfied by  $\mathsf{V}$  are precisely those satisfied by  $\mathsf{K}$ .

 $(2)(a) \Rightarrow (1)$  Suppose (2)(a) holds for V. Let K be a subclass of V satisfying the conditions of (2)(a) and let  $A \in K$ . Let  $a, b \in A$ . If b = 0, then  $a \sqcap^A b = 0$  since the polynomial reduct  $\langle A; \sqcap^A, 0 \rangle$  is a left normal band with zero. So

assume  $b \neq 0$ . If a = 0, then  $a \sqcap^{\mathbf{A}} b = 0 = a$ , also because the polynomial reduct  $\langle A; \sqcap^{\mathbf{A}}, 0 \rangle$  is a left normal band with zero. So assume  $a \neq 0$ . Then  $a, b \in A - \{0\}$ , whence  $a \equiv b \pmod{\mathcal{D}_{\langle A; \sqcap^{\mathbf{A}}, 0 \rangle}}$  since  $\langle A; \sqcap^{\mathbf{A}}, 0 \rangle$  is primitive. By the Clifford-McLean theorem for bands,  $a \sqcap^{\mathbf{A}} b \sqcap^{\mathbf{A}} a = a$ , whence  $a \sqcap^{\mathbf{A}} b = a$ by left normality. We have shown that, for any  $a, b \in A$ ,  $a \sqcap^{\mathbf{A}} b = 0$  if b = 0and a otherwise, so  $\sqcap^{\mathbf{A}}$  is the dual binary discriminator on A. Thus K is a class of dual binary discriminator algebras; since K generates V as a variety we have that V is a dual binary discriminator variety.

**Example 3.2.4.** (cf. [58, Section 2, p. 241]) Let V be a pointed dual ternary discriminator variety (say with 0) with dual ternary discriminator term d(x, y, z) generated by a class  $K \subseteq V$  of dual ternary discriminator algebras. For any  $A \in K$ , let  $x \sqcap y := d(0, y, x)$ . By definition of the dual ternary discriminator,

$$a \sqcap^{\mathbf{A}} b = d^{\mathbf{A}}(0, b, a)$$
$$= \begin{cases} 0 & \text{if } b = 0 \\ a & \text{otherwise} \end{cases}$$

whence A is a dual binary discriminator algebra. Hence K is a class of dual binary discriminator algebras and V is a dual binary discriminator variety.

Let A be a band with zero. For any  $B \subseteq A$ , let BAB [BA; AB] denote the set { $bab : b \in B, a \in A$ } [{ $ba; b \in B, a \in A$ }; { $ab : a \in A, b \in B$ }]. A non-empty subset  $I \subseteq A$  such that both  $IA \subseteq A$  and  $AI \subseteq A$  is an *ideal* of A (in the usual semigroup-theoretic sense) [111, pp. 4-5]. By remarks of Ursini [222, Remarks (b), pp. 211-212], the semigroup-theoretic ideals of A are precisely the ideals of A in the sense of §1.7.1; from remarks due to Petrich [180, Chapter I.2.3, pp. 4-5], it follows that A is ideal simple iff B = BcB for all  $c \in B$ , where  $B := A - \{0\}$ . Suppose A is primitive. By assumption,  $a \equiv$  $b \pmod{D}$  for any  $a, b \in B$ , whence  $BcB = \{dcd : d \in B\} = \{d : d \in B\} = B$ for all  $c \in B$ . Hence A is ideal simple. Conversely, suppose A is ideal simple. By hypothesis, BaB = B = BbB for all  $a, b \in B$ . But for any  $a, b \in A$  we have that aDb iff AaA = AbA by Howie [111, Section 2.4, p. 55]. It follows that

the only  $\mathcal{D}$ -equivalence classes of A are  $\{0\}$  and B (:=  $A - \{0\}$ ), whence A is primitive. Thus A is ideal simple iff it is primitive. This observation, in conjunction with Theorem 3.2.3, gives rise to the following problem.

Problem 3.2.5. Give an ideal-theoretic characterisation of dual binary discriminator varieties.

**3.2.6.** Binary Discriminator Varieties. To within isomorphism, there is just one two-element binary discriminator algebra, namely the two-element flat implicative BCK-algebra  $C_1$  of Example 1.6.18. Since  $C_1$  is, to within isomorphism, the only subdirectly irreducible implicative BCK-algebra (recall Theorem 1.6.19), the variety of implicative BCK-algebras is a binary discriminator variety. We take this observation, which does not seem to have been made by Chajda, Halaš and Rosenberg in [58], as the starting point for our study of binary discriminator varieties. To begin, denote by the *pure binary discriminator variety* (in symbols, PBD) the variety generated by the class of all binary discriminator algebras  $\langle A; b, 0 \rangle$ , where b is the binary discriminator function on A and 0 is a nullary operation.

**Theorem 3.2.7.** The variety of implicative BCS-algebras coincides with the pure binary discriminator variety.

Proof. Let K denote the class of all binary discriminator algebras  $\langle A; b, 0 \rangle$ where b is the binary discriminator function on A and 0 is a nullary operation. By Example 2.3.8 every member of K is a flat implicative BCS-algebra, whence  $K \subseteq iBCS$ . It follows that the variety V(K) generated by K, namely PBD, is a subvariety of iBCS. Conversely, the three-element flat implicative BCS-algebra  $B_2$  is a member of K; since  $B_2$  generates iBCS as a variety (by Theorem 2.3.73) we have that iBCS =  $V(B_2) \subseteq V(K) = PBD$ .

The following result is due to the author's Ph.D. supervisor. In the statement of the theorem and in the sequel we denote the binary discriminator [term] by  $\$ ; of course, this change in notation is justified by Theorem 3.2.7.

**Theorem 3.2.8.** (Bignall) Any algebra  $\mathbf{A}$  in a binary discriminator variety V has an implicative BCS-algebra polynomial reduct whose iBCS-ideals coincide with the V-ideals of  $\mathbf{A}$ .

Proof. By Theorem 3.2.7, any algebra A in a binary discriminator variety V has a canonical implicative BCS-algebra polynomial reduct  $\langle A; \rangle, 0 \rangle$ . Let  $I \in I_V(A)$ . Since V is subtractive,  $I = [0]_{\theta}$  for some  $\theta \in Con A$  by Proposition 1.7.5. Since  $\theta$  is also a congruence on  $\langle A; \rangle, 0 \rangle$ , we have that I is an iBCS-ideal of  $\langle A; \rangle, 0 \rangle$ , just because the ideals of any implicative BCS-algebra are the 0-classes of its congruences. For the converse, let  $\langle a \rangle_{\langle A; \rangle, 0 \rangle}$  be a principal iBCS-ideal of the canonical implicative BCS-algebra polynomial reduct  $\langle A; \rangle, 0 \rangle$ . Let  $\vartheta_a$  be the relation defined on  $A \times A$  by  $b \equiv c \pmod{\vartheta_a}$  iff  $b \backslash a = c \backslash a$ . By Lemma 2.3.45(1)(a)',(2)(a)  $\vartheta_a$  is a congruence on  $\langle A; \rangle, 0 \rangle$ with the property that  $[0]_{\vartheta_a} = \langle a \rangle_{\langle A; \rangle, 0 \rangle}$ . Suppose f is an n-ary fundamental operation on A and that  $b_i \equiv c_i \pmod{\vartheta_a}$  for  $i = 1, \ldots, n$ . By Theorem 1.7.21,

$$f(b_1, \dots, b_n) \setminus a = f(b_1 \setminus a, \dots, b_n \setminus a) \setminus a$$
$$= f(c_1 \setminus a, \dots, c_n \setminus a) \setminus a$$
$$= f(c_1, \dots, c_n) \setminus a$$

so  $f(b_1, \ldots, b_n) \equiv f(c_1, \ldots, c_n) \pmod{\vartheta_a}$ . Hence  $\vartheta_a$  is a congruence on A and  $\langle a \rangle_{\langle A; \backslash, 0 \rangle} \in I_V(A)$ . Let now J be an arbitrary iBCS-ideal of  $\langle A; \backslash, 0 \rangle$ . Put:

$$\psi := \bigvee \{ \vartheta_a \in \operatorname{Con} \langle A; \backslash, 0 \rangle : a \in J \},\$$

where the join is taken in the lattice of equivalence relations. Then  $\psi$  must be a congruence on both A and  $\langle A; \rangle, 0 \rangle$ , just because each  $\vartheta_a$  is. Since the 0-class of  $\psi$  is clearly J, we have that J is a V-ideal of A as required.

In general, the converse of Theorem 3.2.8 fails to hold: see Remark 3.2.21 in the sequel.

**Corollary 3.2.9.** Let  $\lor$  be a binary discriminator variety with binary discriminator term  $x \mid y$  and dual binary discriminator term  $x \mid y$ . The following statements hold:

- 1. V has EDPI witness  $x \setminus y$ ;
- 2. V is ideal distributive;
3. V has the ideal extension property;

4.  $x \sqcap y$  is a meet generator term for  $\lor$ ;

Further, if V is with  $\{0, 1\}$ , then:

5.  $x \sqcup y := \mathbf{1} \setminus ((\mathbf{1} \setminus x) \sqcap (\mathbf{1} \setminus y))$  is a join generator term for  $\vee$ .

**Proof.** Because of Theorem 3.2.8, Proposition 2.3.5 and Proposition 2.2.30, (1) follows from Theorem 2.2.20, while (2) and (3) follow from Proposition 2.1.24. For (4) and (5), notice that on any binary discriminator algebra, the binary discriminator coincides with implicative BCS difference, while the dual binary discriminator coincides with the implicative BCS meet. Because of these remarks and Theorem 3.2.8, Proposition 2.3.5 and Proposition 2.2.30,  $x \sqcap y$  and  $x \sqcup y$  are meet and join generator terms for V by Proposition 2.2.31 and Proposition 2.3.60(2) respectively.

In introducing the binary discriminator as a generalisation of the ternary discriminator to varieties exhibiting congruence permutability only locally at 0, Chajda, Halaš and Rosenberg were primarily concerned with generalising a well-known result of Pixley [184, Theorem 3.1] to the effect that a (finite) algebra A is a ternary discriminator algebra iff V(A) is arithmetical and A is hereditarily simple. In particular, in [58] they proved:

**Proposition 3.2.10.** [58, Corollary 2.2] If A is a binary discriminator algebra then V(A) is arithmetic at 0 and A is hereditarily ideal simple.

Chadja, Halaš and Rosenberg were unable to establish a converse of Proposition 3.2.10, except in the restricted case of main ideal term algebras. A main ideal term algebra is an algebra  $\mathbf{A}$  with  $\mathbf{0}$  for which there exists a binary term function  $\circ$  and a unary term function ' of  $\mathbf{A}$  such that the following equations are identically satisfied for any  $a, b \in A$ :

 $(a \circ b'') \circ b' = 0$  and  $a \circ 0' = a$ 

and moreover:

$$\langle a \rangle_{\mathbf{A}} = \{ b \circ a'' : b \in A \}$$

for every  $a \in A - \{0\}$ ,

where a'' is shorthand notation for (a')'. By [58, Theorem 4.1] a main ideal term algebra A is a binary discriminator algebra iff A is ideal simple. In contradistinction to this result, the following theorem characterises binary discriminator algebras solely in ideal-theoretic terms.

**Theorem 3.2.11.** (cf. [229, Proposition 6.13]) An algebra  $\mathbf{A}$  is a binary discriminator algebra iff  $\mathbf{A}$  is ideal simple and the variety  $\mathbf{V}(\mathbf{A})$  generated by  $\mathbf{A}$  is subtractive with EDPI.

*Proof.*  $(\Rightarrow)$  Let A be a binary discriminator algebra. By Proposition 3.2.10, A is ideal simple and V(A) is subtractive. Since V(A) is a binary discriminator variety, by Corollary 3.2.9(1) we have that V(A) has EDPI.

( $\Leftarrow$ ) Let A be ideal simple and suppose V(A) is subtractive with EDPI. By Theorem 1.7.9 there exists a term  $x \setminus y$  of V(A) that witnesses both subtractivity and EDPI in the sense that V(A)  $\models x \setminus x \approx 0, x \setminus 0 \approx x$  and  $a \setminus^B b = 0$ iff  $a \in \langle b \rangle_B$  for any  $B \in V(A)$  and  $a, b \in B$ . We will show  $x \setminus y$  induces the binary discriminator on A. Let  $a, b \in A$ . Suppose b = 0. By subtractivity  $a \setminus^A b = a \setminus^A 0 = a$ . Suppose  $b \neq 0$ . From  $b \in \langle b \rangle_A$  we have that  $\langle b \rangle_A \neq \{0\}$ , which implies by ideal simplicity that  $\langle b \rangle_A = A$ . Therefore  $a \in \langle b \rangle_A$ , whence  $a \setminus^A b = 0$  by EDPI. Thus A is a binary discriminator algebra.

Corollary 3.2.12. A variety V is a binary discriminator variety iff it is subtractive with EDPI and is generated by a class of ideal simple algebras.

In the following theorem, we give an alternative characterisation of binary discriminator varieties, in the spirit of Agliano and Ursini's characterisation of subtractive varieties with EDPI (Theorem 3.1.6); note that this characterisation of binary discriminator varieties is implicit in [58, Section 5].

**Theorem 3.2.13.** (cf. Theorem 1.7.21) For a variety V the following are equivalent:

1. V is a binary discriminator variety;

2. There exists a binary term  $x \setminus y$  of  $\vee$  such that:

# 3.2. Binary and Dual Binary Discriminator Varieties

- (a) For any  $A \in V$ , the polynomial reduct  $\langle A; \setminus^A, 0 \rangle$  is an implicative BCS-algebra;
- (b) For any  $\mathbf{A} \in \mathsf{V}$  and  $a \in A$ ,  $\langle a \rangle_{\mathbf{A}} = \langle a \rangle_{\langle A; \backslash^{\mathbf{A}}, 0 \rangle}$ ;
- (c) V is generated by a class  $K \subseteq V$  whose members are ideal simple.

*Proof.* (1)  $\Rightarrow$  (2) Suppose  $\forall$  is a binary discriminator variety with binary discriminator term  $x \setminus y$ . For any  $\mathbf{A} \in \mathsf{V}$  the polynomial reduct  $\langle A; \setminus^{\mathsf{A}}, 0 \rangle$  is an implicative BCS-algebra by Example 2.3.11, which establishes (2)(a). Moreover, for any  $\mathbf{A} \in \mathsf{V}$  and  $a \in A$  we have  $\langle a \rangle_{\mathbf{A}} = \langle a \rangle_{\langle A; \setminus^{\mathsf{A}}, 0 \rangle}$  as a particular case of Proposition 3.2.8; thus (2)(b) holds. And, since  $\mathsf{V}$  is a binary discriminator variety, from Theorem 1.7.21 we have that  $\mathsf{V}$  is generated by a class  $\mathsf{K} \subseteq \mathsf{V}$  of ideal simple algebras, which establishes (2)(c).

 $(2) \Rightarrow (1)$  Suppose V is a variety satisfying (2)(a)-(c). By (2)(a) and Theorem 2.1.3 we have that V is subtractive, while from (2)(b) and Theorem 2.2.20 we have that V has EDPI. Because of (2)(c), it follows that V is a subtractive variety with EDPI generated by a class  $K \subseteq V$  of ideal simple algebras; the result now follows from Corollary 3.2.12.

In the following series of examples, we list some binary discriminator varieties beyond those given by Chajda, Halaš and Rosenberg in [58].

**Example 3.2.14.** Let V be a fixed point discriminator variety with 0 generated by a class  $K \subseteq V$  of fixed point discriminator algebras such that 0 realises the discriminating element on any  $A \in K$ . Let f(x, y, z) be a fixed point discriminator term for V. By Example 2.3.13, V is a binary discriminator variety with binary discriminator term f(0, y, x), while K is a class of binary discriminator algebras generating V.

**Example 3.2.15.** (cf. [58, Section 2, p. 241]) Let V be a pointed ternary discriminator variety (say with 0) with ternary discriminator term t(x, y, z) generated by a class  $K \subseteq V$  of ternary discriminator algebras. For any  $A \in K$ , let  $x \setminus y := t(0, y, x)$ . By definition of the ternary discriminator,

 $a \setminus^{\mathbf{A}} b = t^{\mathbf{A}}(0, b, a)$ 

 $= \begin{cases} a & \text{if } b = 0 \\ 0 & \text{otherwise} \end{cases}$ 

whence the polynomial reduct  $\langle A; \setminus^{\mathbf{A}}, 0 \rangle$  is a flat implicative BCS-algebra. Hence K is a class of binary discriminator algebras and V is a binary discriminator variety.

**Example 3.2.16.** By the proof of Proposition 3.1.8, the reduct  $\langle A; \rangle, 0 \rangle$  of any primitive skew Boolean algebra A is a flat implicative BCS-algebra. Hence A is a binary discriminator algebra, and the binary discriminator on A is standard difference. By Theorem 1.4.29, the class SBA of skew Poolean algebras is generated as a variety by any family of primitive skew Boolean algebras that contains the 3-element left and right handed primitive algebras  $3_L^p$  and  $3_R^p$ . Hence SBA is a binary discriminator variety.

**Example 3.2.17.** By Example 2.3.12 the class of pseudocomplemented semilattices is a binary discriminator variety (with binary discriminator term  $x \setminus y := x \wedge y^*$ ), generated (as a binary discriminator variety) by the 3-element chain 3 (considered as a pseudocomplemented semilattice). More generally, PCSL is generated (as a binary discriminator variety) by any subclass of the family of bounded chains (considered as pseudocomplemented consilattices) that includes 3. Indeed, if A is a bounded chain, then  $b^* = 0$  if  $b \neq 0$  and 0 otherwise, whence  $a \setminus b = a$  if b = 0 and a otherwise for any  $a, b \in A$ . Hence the canonical implicative BCS-algebra polynomial reduct  $\langle A; \setminus, 0 \rangle$  of A is flat, and A is a binary discriminator algebra.

**Example 3.2.18.** Recall from Example 1.4.30 that a Stone algebra is a distributive lattice with pseudocomplementation satisfying  $x^* \vee x^{**} \approx 1$ . By Balbes and Dwinger [14, Example VIII§7.2] any bounded chain **A** (considered as a distributive lattice with pseudocomplementation) is a Stone algebra. By Example 2.3.12 and the remarks of §1.3.5, **A** has a canonical implicative BCS-algebra polynomial reduct  $\langle A; \rangle, 0 \rangle$ , which must be flat by Example 3.2.17. Hence **A** is a binary discriminator algebra. Since the class of Stone algebras is generated (as a variety) by any subclass of the family of bounded chains that includes **3**, the 3-element chain considered as a Stone algebra (see Balbes and

# 3.2. Binary and Dual Binary Discriminator Varieties

Dwinger [14, Theorem VIII§7.1] or Grätzer [100]), the class of Stone algebras is a binary discriminator variety.

Remark 3.2.19 (Added in proof). The author's Ph.D. supervisor has pointed out that the class of Abelian Rickart semirings, studied by Cornish in [63], is also a binary discriminator variety. This observation generalises Example 3.2.18, since Stone algebras are a subvariety of the variety of Abelian Rickart semirings.

The final example of this subsection shows that the variety DLPC of distributive lattices with pseudocomplementation is *not* a binary discriminator variety. This example is of interest since every distributive lattice with pseudocomplementation has a canonical implicative BCS-algebra polynomial reduct. Binary discriminator varieties therefore do *not* exhaust those classes of algebras in which implicative BCS-algebras arise naturally (recall the remarks following Problem 2.3.18 in the prequel).

Example 3.2.20. By Example 2.3.12 and the remarks of §1.3.5 any distributive lattice with pseudocomplementation A has a canonical implicative BCSalgebra polynomial reduct  $\langle A; \rangle, 0 \rangle$ , where  $a \backslash b := a \land b^*$  for any  $a, b \in A$ . However, DLPC is not a binary discriminator variety. To see this it is sufficient to show there is no subclass K of DLPC with a binary term b(x, y) that realises the binary discriminator on each  $A \in K$  and for which DLPC = V(K). Observe first by inspection of both the DLPC-free algebra on two free generators [217] and the subdirectly irreducible members of DLPC [101, Section 16] that the only binary term b(x, y) of DLPC inducing implicative BCS difference is  $x \wedge y^*$ . Because of Example 2.3.61(2) and Lemma 2.3.63(2), this implies A is a binary discriminator algebra iff  $(A; \setminus, 0)$  is flat iff A is dense (that is, any element in  $A - \{0\}$  is dense). By results due to Agliano and Ursini [9, Result, p. 256] and Grätzer [101, Exercise 3§14.3, p. 164] every dense distributive lattice with pseudocomplementation  $\langle B; \wedge, \vee, *, 0 \rangle$  arises from a bounded distributive lattice  $\langle B; \wedge, \vee \rangle$  upon: (i) adjoining a new element 0 to B such that  $0 \leq b$  for all  $b \in B$ ; (ii) defining a pseudocomplementation operation on B by  $b^* := 0$ if  $b \neq 0$  and  $b^* := 1$  otherwise, where 1 is the greatest element of B; and (iii) distinguishing the operation \* and the least element 0. (See also Balbes and

## 3.2. Binary and Dual Binary Discriminator Varieties

Dwinger [14, Example VIII§7.3].) Let A be a dense distributive lattice with pseudocomplementation. Because of the definition of \* on A, it is easy to see that  $A \models x^* \lor x^{**} \approx 1$ , whence A is a Stone algebra. Hence the class of all binary discriminator algebras of DLPC generates the variety of Stone algebras, not the variety of distributive lattices with pseudocomplementation, so DLPC is not a binary discriminator variety.

**Remark 3.2.21.** Example 3.2.20 shows also that Condition (2)(c) of Theorem 3.2.13 is not artificial. Indeed, we have already observed in Example 3.2.20 that any distributive lattice with pseudocomplementation A has a canonical implicative BCS-algebra polynomial reduct  $\langle A; \rangle, 0 \rangle$ , whence DLPC satisfies Condition (2)(a) of Theorem 3.2.13. Also, by Agliano and Ursini [9, Example 6.1, p. 256] the principal DLPC-ideal  $\langle b \rangle_A$  generated by  $b \in A$  is  $(b^{**}]$ , whence  $a \in \langle b \rangle_A$  iff  $a \leq b^{**}$  iff  $a \wedge b^* = 0$  iff  $a \setminus b = 0$  iff  $a \in \langle b \rangle_{\langle A; \backslash, 0 \rangle}$ . Thus the principal DLPC-ideals of A coincide with the principal iBCS-ideals of  $\langle A; \backslash, 0 \rangle$ , and DLPC satisfies Condition (2)(b) of Theorem 3.2.13. Because DLPC satisfies Conditions (2)(a)-(2)(b) of Theorem 3.2.13 but is not a binary discriminator variety, Condition (2)(c) cannot be omitted from the assertion of the theorem.

3.2.22. Point Regular Binary Discriminator Varieties. Corollary 3.2.9 prompts us to investigate 0-regular binary discriminator varieties, since 0-regularity implies ideal determinacy for such varieties (by Proposition 1.7.3) and hence EDPC (by Corollary 3.2.9(1) and Proposition 1.7.10). To begin, recall from §1.5.9 that the fixedpoint discriminator on a set A is the ternary operation  $f: A^3 \rightarrow A$  defined for any  $a, b, c \in A$  by:

$$f(a, b, c) := \begin{cases} c & \text{if } a = b \\ 0 & \text{otherwise,} \end{cases}$$

where  $0 \in A$  is the discriminating element of f. A pointed fixed point discriminator algebra is an algebra A with 0 for which there is a ternary term f of Athat realises the fixed point discriminator on A such that  $0^A$  is the discriminating element. A pointed fixed point discriminator variety is a variety V with 0 for which there is a subclass K of V such that V = V(K) and a ternary term f of V

such that f realises the fixedpoint discriminator with discriminating element  $0^{A}$  on each  $A \in K$ . In this case f is called a *pointed fixedpoint discriminator* term for V.

**Remark 3.2.23.** Pointed fixedpoint discriminator varieties should not be confused with fixedpoint discriminator varieties that happen to be pointed. In particular, while every pointed fixedpoint discriminator variety is a fixedpoint discriminator variety that is pointed, the converse does not hold. To see this, let  $A := \langle A; 0^A \rangle$  be a pointed set. Let  $f : A^3 \to A$  be a fixedpoint discriminator on A with discriminating element  $0^A \neq d \in A$ . Let  $A := \langle A; f, 0^A \rangle$ . Then V(A) is a fixedpoint discriminator variety that is pointed, but it is not a pointed fixedpoint discriminator variety. See Blok and Pigozzi [34, p. 580].

**Theorem 3.2.24.** Let V be a variety with 0. If V is a 0-regular binary discriminator variety then V is a pointed fixed point discriminator variety. In this case a pointed fixed point discriminator term for V is given by:

$$f(x, y, z) := \big( \cdots (z \setminus d_1(x, y)) \setminus \cdots \big) \setminus d_n(x, y)$$

where  $x \setminus y$  is a binary discriminator term for  $\vee$  and  $d_1(x, y), \ldots, d_n(x, y)$  are binary terms witnessing the 0-regularity of  $\vee$  in the sense of Proposition 1.2.6.

*Proof.* Let V be a variety with 0. Suppose V is a 0-regular binary discriminator variety with binary discriminator term  $x \setminus y$  and that  $d_1(x, y), \ldots, d_n(x, y)$  are binary terms witnessing the 0-regularity of V in the sense of Proposition 1.2.6. Let  $K \subseteq V$  be a class of binary discriminator algebras generating V as a variety. We will show f(x, y, z) induces the pointed fixedpoint discriminator on any member of K. So let  $A \in K$  and  $a, b, c \in A$ . Suppose a = b. By Proposition 1.2.6 we have that  $d_i(x, x) \approx 0$  for all  $1 \leq i \leq n$ , whence:

$$\left(\cdots (c \backslash^{\mathbf{A}} d_{1}^{\mathbf{A}}(a, b)) \backslash^{\mathbf{A}} \cdots\right) \backslash^{\mathbf{A}} d_{n}^{\mathbf{A}}(a, b) = \left(\cdots (c \backslash^{\mathbf{A}} 0) \backslash^{\mathbf{A}} \cdots\right) \backslash^{\mathbf{A}} 0$$
$$= c$$

by repeated application of the identity  $x \setminus 0 \approx x$ . Suppose instead that  $a \neq b$ . If  $d_i^{\mathbf{A}}(a, b) = 0$  for all  $1 \leq i \leq n$  then a = b by Proposition 1.2.6, a

contradiction. Thus  $d_i^{\mathbf{A}}(a, b) \neq 0$  for some  $1 \leq i \leq n$ . By definition of the binary discriminator, this implies that the subexpression of  $f^{\mathbf{A}}(a, b, c)$  of the form:

$$(\cdots (c \setminus^{\mathbf{A}} d_1^{\mathbf{A}}(a, b)) \setminus^{\mathbf{A}} \cdots) \setminus^{\mathbf{A}} d_i^{\mathbf{A}}(a, b)$$

must equal 0. We therefore have that:

$$(\cdots (((\cdots (c \setminus {}^{\mathbf{A}} d_{1}^{\mathbf{A}}(a, b)) \setminus {}^{\mathbf{A}} \cdots ) \setminus {}^{\mathbf{A}} d_{i}^{\mathbf{A}}(a, b)) \setminus {}^{\mathbf{A}} d_{i+1}^{\mathbf{A}}(a, b)) \setminus {}^{\mathbf{A}} d_{n}^{\mathbf{A}}(a, b)$$

$$= (\cdots (0 \setminus {}^{\mathbf{A}} d_{i+1}^{\mathbf{A}}(a, b)) \setminus {}^{\mathbf{A}} \cdots ) \setminus {}^{\mathbf{A}} d_{n}^{\mathbf{A}}(a, b)$$

$$= 0$$

by repeated application of the identity  $0 \setminus x \approx 0$ . Hence  $f^{\mathbf{A}}(a, b, c) = c$  if a = b and 0 otherwise, whence f(x, y, z) induces the fixedpoint discriminator on any member of K. Since K generates V as a variety we have that V is a pointed fixedpoint discriminator variety and that f(x, y, z) is a pointed fixedpoint discriminator term for V.

**Corollary 3.2.25.** Let  $\vee$  be a variety with  $\{0, 1\}$ . If  $\vee$  is a 0-regular binary discriminator variety then the following assertions hold:

- 1. The term f(x, y, z) of Theorem 3.2.24 is a commutative TD term for V that is regular with respect to 1;
- 2. V is a variety of subtractive weak Boolean algebras with filter preserving operations. Weak meet, weak relative pseudocomplement and Gödel equivalence terms for V are defined respectively by:

 $\begin{aligned} x \cdot y &:= f(x, \mathbf{1}, y) \\ x \to y &:= f\big(f(x, \mathbf{1}, y), f(x, \mathbf{1}, \mathbf{1}), \mathbf{1}\big) \\ x \Delta y &:= f(x, y, \mathbf{1}). \end{aligned}$ 

*Proof.* Immediate by Theorem 3.2.24 and Theorem 1.5.14.

**Remark 3.2.26.** The hypothesis that V is double-pointed in Corollary 3.2.25 is essential if the conclusions of the corollary are to obtain, since the variety

of implicative BCK-algebras is a 0-regular binary discriminator variety that does not possess either a TD term that is regular or a weak meet. See Blok and Pigozzi [34, p. 589].

Example 3.2.14 shows that a pointed fixedpoint discriminator variety is a binary discriminator variety. Given Theorem 3.2.24, it is therefore natural to ask if the converse of Theorem 3.2.24 also holds. Perhaps surprisingly, the answer to this question is 'yes'.

**Theorem 3.2.27.** Let V be a variety with **0**. If V is a pointed fixed point discriminator variety, then V is a **0**-regular binary discriminator variety. In this case a binary discriminator term for V is given by:

$$x \setminus y := f(0, y, x)$$

where f(x, y, z) is a pointed fixed point discriminator term for V, while the binary terms:

$$d_1(x,y) := x \setminus f(x,y,x)$$
 and  $d_2(x,y) := y \setminus f(y,x,y)$ 

witness 0-regularity for  $\vee$  in the sense of Proposition 1.2.6.

*Proof.* Let V be a variety with **0**. Suppose V is a pointed fixedpoint discriminator variety with pointed fixedpoint discriminator term f(x, y, z). Let  $K \subseteq V$  be a class of fixedpoint discriminator algebras generating V as a variety. Put  $x \setminus y := f(0, y, x)$ . For any  $A \in K$  and  $a, b \in A$ , by Example 2.3.13 we have that:

$$a \setminus^{\mathbf{A}} b = f^{\mathbf{A}}(0, b, a)$$
$$= \begin{cases} a & \text{if } b = 0\\ 0 & \text{otherwise,} \end{cases}$$

whence  $\langle A; \langle 0 \rangle$  is a flat implicative BCS-algebra. Thus  $x \setminus y$  induces the binary discriminator on any member of K. Since K generates V as a variety, it follows that V is a binary discriminator variety. To see V is 0-regular,

283

put  $x/y := x \setminus f(x, y, x)$ . For any  $A \in K$  and  $a, b \in A$ , by definition of the fixed point discriminator we have that:

$$a/b = a \backslash^{\mathbf{A}} f^{\mathbf{A}}(a, b, a)$$
  
=  $f^{\mathbf{A}}(0, f^{\mathbf{A}}(a, b, a), a)$   
= 
$$\begin{cases} a & \text{if } a \neq b \\ 0 & \text{otherwise,} \end{cases}$$

whence  $\langle A; /, 0 \rangle$  is (by Example 1.6.18) a flat implicative BCK-algebra. Because the identities satisfied by V are precisely those satisfied by K, it follows that any algebra  $\mathbf{B} \in V$  has an implicative BCK-algebra polynomial reduct  $\langle B; /^{\mathbf{B}}, 0 \rangle$ . Since any algebra with a point regular polynomial reduct must itself be point regular, we conclude from the 0-regularity of  $\langle B; /^{\mathbf{B}}, 0 \rangle$  (recall Theorem 1.6.17 and Theorem 1.6.10(3)) that **B** is 0-regular. Thus V is 0-regular and the binary terms  $d_1(x, y) := x/y$ ,  $d_2(x, y) := y/x$  witness 0regularity for V in the sense of Proposition 1.2.6.

Remark 3.2.28 (Added in proof). An implicative BCSK-algebra is an algebra  $\langle A; /, \backslash, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra; (ii) the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra; and (iii) the implicative BCK partial order  $\leq^{\langle A; /, 0 \rangle}$  and the implicative BCS partial order  $\leq^{\langle A; \rangle, 0 \rangle}$  coincide. By an unpublished result of the author, an algebra  $\mathbf{A} := \langle A; /, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is an implicative BCSK algebra iff  $\mathbf{A} \models (1.35) - (1.38), \mathbf{A} \models (2.47) - (2.50)$  and  $\mathbf{A}$  satisfies the identities  $(x \backslash y)/x \approx \mathbf{0}$  and  $x \backslash (x \backslash (x/y)) \approx x/y$ , whence the class iBCSK of implicative BCSK-algebras is a variety.

The variety iBCSK arises in the first instance from algebraic logic. Let  $\mathcal{L} := \{ \rightarrow, \Rightarrow \}$  be a language of type  $\langle 2, 2 \rangle$ . *BCSK logic* is the deductive system  $\mathbb{BCSK} := \langle \mathcal{L}, \vdash_{\mathbb{BCSK}} \rangle$  defined by the following axioms and inference rule:

$$p \Rightarrow (q \Rightarrow p)$$
 (B1)

$$(p \Rightarrow (q \Rightarrow r)) \Rightarrow ((p \Rightarrow q) \Rightarrow (p \Rightarrow r))$$
 (B2)

$$((p \Rightarrow q) \Rightarrow p) \Rightarrow p \tag{B3}$$

3.2. Binary and Dual Binary Discriminator Varieties

$$p \Rightarrow (q \rightarrow p)$$
 (B4)

$$(p \to (q \to r)) \Rightarrow ((p \to q) \to (p \to r))$$
 (B5)

 $(p \to (q \to r)) \Rightarrow (q \to (p \to r))$  (B6)

$$((p \to q) \to p) \Rightarrow p$$
 (B7)

$$((p \Rightarrow q) \rightarrow q) \Rightarrow ((q \Rightarrow p) \rightarrow p)$$
(B8)  
(p \Rightarrow q)  $\rightarrow (p \rightarrow q)$  (B9)

$$p, p \to q \vdash_{BCSK} q.$$
 (BCSK-MP)

BCSK logic was introduced by the author in [211] and has been extensively investigated by Humberstone in [112], where connections with modal logic (including the Lewis system S5) are established. Unpublished results of the author show BCSK is algebraisable (with equivalence formulas  $\{p \Rightarrow q, q \Rightarrow p\}$ and defining equation  $p \approx p \Rightarrow p$ ) and that its equivalent algebraic semantics is termwise definitionally equivalent to iBCSK.

Call an implicative BCSK-algebra *flat* if its underlying poset is flat. The proof of Theorem 3.2.27 shows any pointed fixedpoint discriminator algebra  $\langle A; f, 0 \rangle$  has a flat implicative BCSK-algebra polynomial reduct  $\langle A; /, \backslash, 0 \rangle$ , where:

 $a \setminus b := f(0, b, a)$  and  $a/b := a \setminus f(a, b, a)$ 

for any  $a, b \in A$ . Conversely, unpublished results due to the author and the author's Ph.D. supervisor show that an implicative BCSK-algebra is subdirectly irreducible iff it is flat (for some details, see Humberstone [112, Section 1, Appendix B]), whence the class iBCSK is a pointed fixedpoint discriminator variety with pointed fixedpoint discriminator term:

 $f(x, y, z) := (z \setminus (x/y)) \setminus (y/x).$ 

#### 3.2. Binary and Dual Binary Discriminator Varieties

Let  $FPD_0$  denote the *pure pointed fixedpoint discriminator variety*, namely the variety of type  $\langle 3, 0 \rangle$  generated by the class of all pointed fixedpoint discriminator algebras  $\langle A; f, 0 \rangle$ , where 0 is a nullary operation and f is the fixedpoint discriminator on A with discriminating element 0. In view of the preceding discussion, it is easy to see that  $FPD_0$  is termwise definitionally equivalent to iBCSK. Moreover, because the congruence structure of any algebra in a fixed-point discriminator variety is (by Lemma 1.5.10) completely determined by the fixedpoint discriminator term, any algebra A in a pointed fixedpoint discriminator variety must have an implicative BCSK-algebra polynomial reduct whose congruences coincide with those of A. These remarks extend and contrast with Blok and Pigozzi [34, Section 3]: see in particular [34, Corollary 3.6].

In their study [34] of varieties with equationally definable principal congruences, Blok and Pigozzi posed the following problem [34, Problem 7.3]: Does there exist a purely algebraic characterisation of fixedpoint discriminator varieties similar to the one for ternary discriminator varieties given in Theorem 1.5.13(4)? For pointed fixedpoint discriminator varieties, the following theorem provides an affirmative answer to this question.

**Theorem 3.2.29.** For a variety V with 0, the following are equivalent:

1. V is a pointed fixedpoint discriminator variety;

2. V is a 0-regular binary discriminator variety;

3. V is congruence 0-permutable, 0-regular, semisimple with EDPC;

4. V is ideal determined, semisimple with EDPC.

*Proof.* Let V be a variety with 0. The equivalence  $(1) \Leftrightarrow (2)$  follows from Theorem 3.2.24 and Theorem 3.2.27, while the equivalence  $(3) \Leftrightarrow (4)$  is clear from Proposition 1.7.3. Thus it only remains to demonstrate the equivalence  $(2) \Leftrightarrow (3)$ .

 $(2) \Rightarrow (3)$  Suppose V is a 0-regular binary discriminator variety. Then V is 0-regular and subtractive by hypothesis. Also, V is a pointed fixed point discrim-

inator variety by Theorem 3.2.24, and so is semisimple (by Theorem 1.5.12) and has EDPC (by the remarks of  $\S1.5.9$ ).

(3)  $\Rightarrow$  (2) Suppose V is congruence 0-permutable, 0-regular and semisimple with EDPC. Since V is subtractive, by Proposition 1.7.10 we have that V has EDPI. Since V is semisimple with EDPC, from Theorem 1.5.2(4) we have that V is generated as a variety by a class K of simple algebras. By normality of ideals we have that K is ideal simple, and so V is a subtractive variety with EDPI that is generated by a class K of ideal simple algebras. From Corollary 3.2.12 it follows that V is a binary discriminator variety; since V is 0-regular by hypothesis, V is a 0-regular binary discriminator variety.

Although pointed fixedpoint discriminator varieties do not encompass even those fixedpoint discriminator varieties that are pointed (by Remark 3.2.23), the hypotheses of Theorem 3.2.29 are nonetheless satisfied by most fixedpoint discriminator varieties arising naturally as 'quasivarieties of logic'. In particular, varieties of k-potent Wajsberg algebras and their implicational subreducts (including Boolean algebras and implicative BCK-algebras) are pointed fixedpoint discriminator varieties. See [34, Corollary 3.6].

Example 3.2.30. By Example 2.3.14(1) each variety  $ce_n BCK$ ,  $n \in \omega$ , is a fixed point discriminator variety, with fixed point discriminator term  $f(x, y, z) := (z \div (x \div y)^n) \div (y \div x)^n$ . Although each  $ce_n BCK$  is pointed, the proof of Example 2.3.14(1) does not show each  $ce_n BCK$  is a pointed fixed point discriminator variety: recall Remark 2.3.15. In contrast, we may immediately conclude from Theorem 3.2.29 that each  $ce_n BCK$ ,  $n \in \omega$ , is a pointed fixed point discriminator variety, just because each  $ce_n BCK$  is ideal determined (by Theorem 1.6.10(3)) and semisimple with EDPC (by Example 2.3.14(1)).

Via Theorem 3.2.24, the proof of Theorem 3.2.29 yields a fixedpoint discriminator term for each ce<sub>n</sub>BCK,  $n \in \omega$ , namely  $f'(x, y, z) := (z \setminus (x - y)) \setminus (y - x)$ , where  $x \setminus y$  is a binary discriminator term for ce<sub>n</sub>BCK. For each  $n \in \omega$ , let  $x \setminus y := x - y^n$ . By [68, Lemma 1.1(ii),(iii)] ce<sub>n</sub>BCK  $\models x \setminus x \approx 0$  and ce<sub>n</sub>BCK  $\models x \setminus 0 \approx x$ , whence  $\setminus$  witnesses subtractivity for ce<sub>n</sub>BCK. Also,  $\setminus$  witnesses EDPI for ce<sub>n</sub>BCK, since it is implicit in the proof of Blok and Raftery [39,

### 3.2. Binary and Dual Binary Discriminator Varieties

Theorem 4.2] that  $a \in \langle b \rangle_{\mathbf{A}}$  iff  $a \setminus^{\mathbf{A}} b = 0$  for any  $\mathbf{A} \in \mathsf{ce}_{\mathsf{n}}\mathsf{BCK}$ . Hence  $\setminus$  is a binary discriminator term for each  $\mathsf{ce}_{\mathsf{n}}\mathsf{BCK}$ ,  $n \in \omega$ , and  $f'(x, y, z) = (z \setminus (x \div y)) \setminus (y \div x) = (z \div (x \div y)^n) \div (y \div x)^n = f(x, y, z)$ .

Let K be a quasivariety and  $\mathbf{A} \in K$ . Recall that a strong ideal of  $\mathbf{A}$  in the sense of Blok and Raftery [40] is an  $S(K,\tau)$ -filter of  $\mathbf{A}$ , where  $S(K,\tau)$  is a certain deductive system 'extracted' from the quasi-equational theory of K by means of a translation  $\tau$  [40, Section 5]. Under inclusion, the set  $SId_{K,\tau}$  of all strong ideals of  $\mathbf{A}$  forms an algebraic lattice  $SId_{K,\tau}$ . In general, the natural map  $\tau^{\mathbf{A}}/-: \operatorname{Con}_{\mathsf{K}} \mathbf{A} \to \operatorname{SId}_{\mathsf{K},\tau} \mathbf{A}$  sending a K-congruence to its associated strong ideal is neither injective nor surjective. We say K is *strongly ideal determined* if, for any  $\mathbf{A} \in \mathsf{K}$ , the map  $\tau^{\mathbf{A}}/-: \operatorname{Con}_{\mathsf{K}} \mathbf{A} \to \operatorname{SId}_{\mathsf{K},\tau} \mathbf{A}$  is a lattice isomorphism; for details, see [40, Section 5, Theorem 5.2]. Theorem 3.2.29 and preceding remarks invite the following problem:

**Problem 3.2.31.** Let V be a variety. Is V a fixed point discriminator variety iff V is strongly ideal determined, semisimple with EDPC?

3.2.32. Binary Discriminator Varieties and Pointed Ternary Discriminator Varieties. Theorem 3.2.29 shows a variety with 0 is a pointed fixedpoint discriminator variety iff it is a 0-regular binary discriminator variety. Since the theory of the fixedpoint discriminator closely parallels that of the ternary discriminator [34, p. 548], this observation calls for a study of the relationship between binary and pointed ternary discriminator varieties. The following theorem is an obvious consequence of Theorem 3.2.29 and the results of §1.5.9.

**Theorem 3.2.33.** For a variety V with 0 the following are equivalent:

1. V is a ternary discriminator variety;

2. V is a congruence permutable 0-regular binary discriminator variety.

In particular, if  $x \setminus y$  is a binary discriminator term for V, p(x, y, z) is a Mal'cev term for V, and  $d_1(x, y), \ldots, d_n(x, y)$  are binary terms of V witnessing the 0-regularity of V in the sense of Proposition 1.2.6, then:

t(x, y, z) := p(f(x, y, z), f(x, y, x), x)

287

(

is a ternary discriminator term for V, where f(x, y, z) is the pointed fixed point discriminator term of Theorem 3.2.24.

#### *Proof.* Let V be a variety with **0**.

(1)  $\Rightarrow$  (2) Suppose V is a ternary discriminator variety. By Theorem 1.5.13 we have that V is a congruence permutable pointed fixedpoint discriminator variety, so V is a congruence permutable 0-regular binary discriminator variety by Theorem 3.2.27.

(2)  $\Rightarrow$  (1) Suppose V is a congruence permutable 0-regular binary discriminator variety. Then V is a pointed fixedpoint discriminator variety by Theorem 3.2.24. Let f(x, y, z) be the pointed fixedpoint discriminator term for V of Theorem 3.2.24 and let p(x, y, z) be a Mal'cev term for V. Since f(x, y, z) is a TD term for V (by Theorem 1.5.12), from Lemma 1.5.11(1) we have that q(x, y, z, w) := p(f(x, y, z), f(x, y, w), w) is a QD term for V. Since V has permuting congruences q(x, y, z, x) is a ternary discriminator term for V; clearly q(x, y, z, x) = t(x, y, z), completing the proof.

By Corollary 2.2.6 and the remarks of §3.2.6 the variety of implicative BCKalgebras is a **0**-regular binary discriminator variety that is not congruence permutable. Thus the hypothesis of congruence permutability cannot be omitted from the statement of Theorem 3.2.33. The following example shows that the assumption of **0**-regularity also cannot be dropped.

**Example 3.2.34.** Let  $A := \{0, 1, 2\}$  be a set and let  $t : A^3 \to A$  be the ternary discriminator on A. Let  $\mathbf{A} := \langle A; p, 0 \rangle$  be the algebra with distinguished element 0 and ternary operation  $p : A^3 \to A$  defined by:

 $p(a, b, c) := \begin{cases} a & \text{if } a = 1, \ b = 2 \text{ and } c = 0 \\ b & \text{if } a = 2, \ b = 1 \text{ and } c = 0 \\ t(a, b, c) & \text{otherwise} \end{cases}$ 

for any  $a, b, c \in A$ . Let  $a \setminus b := p(0, b, a)$  for any  $a, b \in A$ . Clearly  $a \setminus b$  is the binary discriminator on A, so A is a binary discriminator algebra. Moreover, one easily checks that  $A \models p(x, x, y) \approx y$  and  $A \models p(x, y, y) \approx x$ ; thus

## 3.2. Binary and Dual Binary Discriminator Varieties

V(A), the variety generated by A, is a congruence permutable binary discriminator variety. However, V(A) is not point regular, and in particular is not 0-regular. Indeed, one easily shows (by inspection of the V(A)-free algebra on one free generator) that the only constant term of V(A) is 0. But the partition  $\{\{0\}, \{1,2\}\}$  induces a non-trivial (in fact, the only non-trivial) congruence on A, whence A itself is not 0-regular. Thus V(A) is a congruence permutable binary discriminator variety that is not point-regular (and in particular not 0-regular), and hence is not a ternary discriminator variety.

In general, it is a non-trivial task to construct a Mal'cev term for a congruence permutable variety: witness for example Theorem 3.1.40. A simpler syntactic criterion both necessary and sufficient for a 0-regular binary discriminator variety to be a ternary discriminator variety is therefore desirable. In the following proposition we give just such a syntactic criterion. But first, let Abe a set and let  $0 \in A$  be fixed but arbitrary. Recall from multiple-valued switching theory [172, Chapter 3] that a binary function  $+: A^2 \rightarrow A$  is called a sum-like operation (with respect to 0) if a + 0 = a = 0 + a for any  $a \in A$ . Let K be a class of algebras with 0. A binary term x + y of K is said to be sumlike (with respect to 0) if the canonical interpretation of x + y on any  $A \in K$  is a sum-like operation with respect to  $0^A$ . See also Werner [237, Theorem 1.3].

**Proposition 3.2.35.** (cf. [39, Theorem 6.1(i)]) For a variety  $\vee$  with 0 the following are equivalent:

- 1. V is a ternary discriminator variety;
- 2. V is a 0-regular binary discriminator variety with a binary term x + y that is sum-like (with respect to 0).

In particular, if  $x \setminus y$  and x + y are respectively a binary discriminator term and a sum-like term for V, then:

$$t(x, y, z) := f(x, y, z) + (x \setminus f(x, y, x))$$

is a ternary discriminator term for V, where f(x, y, z) is the pointed fixed point discriminator term of Theorem 3.2.24.

### *Proof.* Let V be a variety with **0**.

 $(1) \Rightarrow (2)$  Suppose V is a ternary discriminator variety with ternary discriminator term t(x, y, z). By Theorem 3.2.33 we have that V is a 0-regular binary discriminator variety. Put x + y := t(y, 0, x). Because of Theorem 1.4.39, we have that x + A y is the left handed skew lattice join on any  $A \in V$ , whence V has a sum-like term.

(2)  $\Rightarrow$  (1) Suppose V is a 0-regular regular binary discriminator variety with binary discriminator term  $x \setminus y$ . By Theorem 3.2.24 we have that V is a pointed fixedpoint discriminator variety. Let  $K \subseteq V$  be a class of fixedpoint discriminator algebras generating V as a variety. Let x + y be a sum-like term for V and put  $t(x, y, z) := f(x, y, z) + (x \setminus f(x, y, x))$ , where f(x, y, z) is the pointed fixedpoint discriminator term of Theorem 3.2.24. By definition of the fixedpoint discriminator, for any  $A \in K$  and  $a, b, c \in A$  we have:

$$f^{\mathbf{A}}(a, b, c) = \begin{cases} c & \text{if } a = b \\ 0 & \text{otherwise} \end{cases} \text{ and } a \setminus^{\mathbf{A}} f^{\mathbf{A}}(a, b, a) = \begin{cases} a & \text{if } a \neq b \\ 0 & \text{otherwise.} \end{cases}$$

Since + is a sum-like term, it follows that  $t^{\mathbf{A}}(a, b, c) = a$  if  $a \neq b$  and c otherwise. Thus t(x, y, z) induces the ternary discriminator on any member of K; since K generates V as a variety we have that V is a ternary discriminator variety with ternary discriminator term t(x, y, z).

**Remark 3.2.36.** Let V be a dual binary discriminator variety with dual binary discriminator term  $x \wedge y$ . In [19, Section 4.10] Bignall and Leech essentially assert that a necessary and sufficient condition for V to be a ternary discriminator variety is the existence of a binary term  $x \oplus y$  that is both a sum-like term and a Gödel equivalence term for V; in this case a ternary discriminator term for V is given by:

$$t(x, y, z) := (x \land (x \oplus y)) \oplus (z \oplus (z \land (x \oplus y))).$$

Collectively, the results of §1.4.32, §1.4.37, §3.2.6 and §3.2.22 intimate that pre-BCK-algebras structurally enriched with (locally Boolean) band operations arise naturally in pointed discriminator, pointed fixedpoint discriminator and pointed discriminator varieties, which remark calls for a generalisation of Idziak's theory of BCK-[semi]lattices to the non-commutative case. Lemma 1.6.24, which shows any BCK-[semi]lattice may be viewed as the conjunction of a BCK-algebra and a [semi]lattice such that the underlying BCKalgebra partial ordering and the [semi]lattice partial ordering either dualise or coincide, suggests that an appropriate generalisation of BCK-[semi]lattices is to algebras consisting of a pre-BCK-algebra reduct and a band or noncommutative lattice reduct such that the underlying pre-BCK-algebra ordering either coincides with or dualises an ordering on the band or non-commutative lattice reduct. Because there exist two fundamental orderings  $\preceq_{\mathcal{D}}$  and  $\leq_{\mathcal{H}}$  on any band, Idziak's theory of BCK-[semi]lattices bifurcates when generalised to the non-commutative case. On the one hand, the theory of BCK-[semi]lattices generalises to a theory of pre-BCK bands and pre-BCK quasilattices  $PQ_c$ ,  $\{ \div, \mathbf{0} \} \subseteq \mathcal{C} \subseteq \{ \land, \lor, \div, \mathbf{0} \}$ ; and on the other, to a theory of BCK bands and BCK paralattices  $\mathsf{BP}_{\mathcal{C}}$ ,  $\{ \div, \mathbf{0} \} \subseteq \mathcal{C} \subseteq \{ \land, \lor, \div, \mathbf{0} \}$ . Within the context of the families  $PQ_{\mathcal{C}}$ ,  $PQ_{\mathcal{C}}$ , it is natural to focus attention (ignoring issues of type) on certain varieties  $IQ_{\mathcal{C}'}$ ,  $\{\wedge, \backslash, \mathbf{0}\} \subseteq \mathcal{C}' \subseteq \{\wedge, \lor, \backslash, \mathbf{0}\}$ , of implicative pre-BCK bands and implicative BCS quasilattices and certain varieties  $P_{C''}$ ,  $\{\wedge, /, 0\} \subseteq \mathcal{C}'' \subseteq \{\wedge, \vee, /, 0\}$ , of implicative BCK bands and implicative BCK paralattices respectively, since these clasces exhibit 'locally Boolean' behaviour. Throughout this section, our study of all the various families  $PQ_{\mathcal{C}}$ ,  $IQ_{\mathcal{C}'}$ ,  $BP_{\mathcal{C}}$ and  $IP_{C''}$  is informed by both Idziak's theory of BCK-[semi]lattices and Laslo and Leech's recent study of paralattices and quasilattices [145].

In §3.3.2 we study the family of classes  $PQ_{\mathcal{C}}$ ,  $\{\div, 0\} \subseteq \mathcal{C} \subseteq \{\land, \lor, \div, 0\}$ . Members of the classes  $PQ_{\mathcal{C}}$  consist of pre-BCK-algebras (possibly) structurally enriched with band operations  $\land$  or  $\lor$  such that the natural band quasiorder  $\preceq_{\mathcal{D}}$  coincides with or dualises the underlying pre-BCK-algebra quasiorder  $\leq^{(-,0)}$ . Let  $\wedge \in C$  or  $\vee \in C$ . For each choice of C, it is shown that the class  $PQ_C$  is a variety. For each  $FQ_C$ , we present analogues of the Clifford-McLean theorem for lands or quasilattices; some sufficient conditions for each  $PQ_C$  to be regular (in a sense suitably modified from §1.3.15) are also presented.

In §3.3.13 we study the family of classes  $|Q_{C'}, \{\backslash, 0\} \subseteq C' \subseteq \{\land, \lor, \backslash, 0\}$ . Members of the classes  $|Q_{C'}$  consist of implicative BCS-algebras (possibly) structurally enriched with band operations  $\land$  and  $\lor$  such that the natural band partial order  $\leq_{\mathcal{H}}$  and quasiorder  $\preceq_{\mathcal{D}}$  respectively coincide with or dualise the underlying implicative BCS-algebra partial order  $\leq_{(\backslash,0)}$  and quasiorder  $\preceq_{(\backslash,0)}$ . For each choice of  $C', \land \in C'$  or  $\lor \in C'$ , it is shown that the class  $|Q_{C'}$  is a variety. Let  $\land \in C'$ . For each C' and any  $\mathbf{A} \in |Q_{C'}$ , it is shown that the band with zero reduct  $\langle A; \land, 0 \rangle$  is locally Boolean (in the sense of §1.3.15). In one of the two main results of the section, the skew Boole in algebras are characterised amongst the members of  $|Q_{C'}, \{\land, \lor\} \subseteq C'$ . Ignoring issues of similarity type, we also show that Idziak's variety of BCK-lattices is the splitting variety associated with the variety of left handed skew Boolean algebras in a certain large subvariety of PQ<sub>C</sub>,  $\{\land, \lor\} \subseteq C$ .

In §3.3.27 we study the family of classes  $BP_C$ ,  $\{-,0\} \subseteq C \subseteq \{\wedge, \vee, +, 0\}$ . Members of the classes  $BP_C$  consist of BCK-algebras (possibly) structurally enriched with band operations  $\wedge$  or  $\vee$  such that the natural band partial order  $\leq_{\mathcal{H}}$  coincides with or dualises the underlying BCK-algebra partial order  $\leq^{(+,0)}$ . Let  $\wedge \in C$  or  $\vee \in C$ . For each choice of C, we show that the class  $BP_C$  is a variety. It is shown that no non-trivial analogue of the Clifford-McLean theorem exists for each  $BP_C$  and hence that each  $BP_C$  is only trivially regular (in a sense suitably modified from §1.3.15). We also prove that each  $BP_C$  is ideal determined, congruence distributive and (when  $\vee \in C$ ) congruence permutable.

In §3.3.43 we study the family of classes  $IP_{\mathcal{C}''}$ ,  $\{/, 0\} \subseteq \mathcal{C}'' \subseteq \{\wedge, \lor, /, 0\}$ . Members of the classes  $IP_{\mathcal{C}''}$  consist of implicative BCK-algebras (possibly) structurally enriched with band operations  $\wedge$  and  $\vee$  such that the natural band partial order  $\leq_{\mathcal{H}}$  coincides with or dualises the underlying implicative

BCK-algebra partial order  $\leq^{\langle/,0\rangle}$ . For each choice of  $\mathcal{C}''$ ,  $\wedge \in \mathcal{C}''$  or  $\vee \in \mathcal{C}''$ , it is shown that the class  $\mathsf{IP}_{\mathcal{C}''}$  is a variety. Let  $\wedge \in \mathcal{C}''$ . For each  $\mathcal{C}''$  and any  $\mathbf{A} \in \mathsf{IP}_{\mathcal{C}''}$ , it is shown that the band with zero reduct  $\langle A; \wedge, 0 \rangle$  is locally Boolean (in the sense of §1.3.15). In the other main result of the section, the skew Boolean  $\cap$ -algebras are characterised (to within termwise definitional equivalence) amongst the members of  $\mathsf{IP}_{\mathcal{C}''}$ ,  $\{\wedge,\vee\} \subseteq \mathcal{C}''$ . We also present a simple equational axiomatisation of the variety of skew Boolean  $\cap$ -algebras.

The final two subsections of this section are devoted to the further exploration of the theory of skew Boolean  $\cap$ -algebras, and hence, by extension, the theory of pointed discriminator varieties. In §3.3.59 the theory of skew Boolean  $\cap$ -algebras as presented in §1.4.32, §1.4.37 is extended to double-pointed skew Boolean  $\cap$ -algebras. It is shown that the class of double-pointed skew Boolean  $\cap$ -algebras is a variety, and the subdirectly irreducible double-pointed skew Boolean  $\cap$ -algebras are characterised to within isomorphism. It is also shown that the variety of double-pointed left handed skew Boolean  $\cap$ -algebras coincides with the pure double-pointed discriminator variety, namely the variety of type  $\langle 3, 0, 0 \rangle$  generated by the class of all double-pointed discriminator algebras. In consequence, we infer that any algebra **A** in a double-pointed discriminator variety has a double-pointed left handed skew Boolean  $\cap$ -algebra

In §3.3.69 we present an axiomatisation of a certain deductive system SBPC. We show SBPC is definitionally equivalent to the assertional logic of the variety of double-pointed left handed skew Boolean  $\cap$ -algebras, and hence infer that SBPC is definitionally equivalent to the assertional logic of the pure doublepointed discriminator variety. It is also show that, in principle, there exists an axiomatisation of SBPC such that (MP) is the only (proper) rule of inference.

**Remark 3.3.1.** We impose two fundamental restrictions on the scope of our study of the varieties  $BP_{\mathcal{C}}$  (and hence, by extension, on our study of the varieties  $PQ_{\mathcal{C}}$ ) in this section. First, our study of the varieties  $BP_{\mathcal{C}}$  does not extend to a study of those members of  $BP_{\mathcal{C}}$  for which the BCK-algebra reduct has condition (S) (recall the definition of condition (S) from §2.1.1). Although BCK-[semi]lattices with condition (S) were considered by Idziak in his origi-

nal paper [116] on BCK-[semi]lattices, a study of those members of the varieties  $BP_C$  for which the BCK-algebra reduct has condition (S) contravenes (at least in spirit) the restrictions of Remark 1.1.2, because BCK-algebras with condition (S) are precisely the  $\langle -, 0 \rangle$ -reducts of pocrims (by results due to Iséki [121, 123]). Second, our study of the varieties  $BP_C$  does not extend to a study of the assertional logics  $S(BP_C, 0)$  (with the obvious exception of the assertional logic of the variety of double-pointed left handed skew Boolean  $\cap$ -algebras), because the deductive systems  $S(BP_C, 0)$  are, by remarks due to Restall [199], not in general amenable to standard logical analysis (for example, in the sense of [198]): cf. Remark 3.3.80 and the remarks of §4.2.27 in the sequel. By extension, the preceding restrictions apply mutatis mutandis to our study of the varieties  $PQ_C$ , to better enable the uniform development of the theory of the varieties  $BP_C$  and  $PQ_C$ .

**3.3.2.** Pre-BCK Bands and Pre-BCK Quasilattices. A lower pre-BCK-band is an algebra  $\langle A; \land, \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \land, 0 \rangle$  is a band with zero; (ii) the reduct  $\langle A; \div, 0 \rangle$  is a pre-BCK-algebra; and (iii) the natural band quasiorder  $\preceq_{\mathcal{D}}$  coincides with the pre-BCK quasiorder  $\preceq^{\langle A; \div, 0 \rangle}$ . Clearly a lower pre-BCK band A is a lower BCK-semilattice iff either  $\mathbf{A} \models (2.5)$  or  $\mathbf{A} \models x \land y \approx y \land x$ .

**Theorem 3.3.3.** An algebra  $\langle A; \land, \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is a lower pre-BCKband iff the following identities are satisfied:

$$x \wedge (y \wedge z) \approx (x \wedge y) \wedge z \tag{3.49}$$

$$x \wedge x \approx x$$
 (3.50)

$$\left((x - y) - (x - z)\right) - (z - y) \approx \mathbf{0} \tag{3.51}$$

 $x - \mathbf{0} \approx x \tag{3.52}$ 

 $\mathbf{0} - \mathbf{x} \approx \mathbf{0} \tag{3.53}$ 

 $(x \wedge y \wedge x) - y \approx 0 \tag{3.54}$ 

$$\left(x - (x - y)\right) \wedge y \wedge \left(x - (x - y)\right) \approx x - (x - y). \tag{3.55}$$

Thus the class of lower pre-BCK bands is a variety.

*Proof.* Let  $\mathbf{A} := \langle A; \land, \div, 0 \rangle$  be an algebra of type  $\langle 2, 2, 0 \rangle$  satisfying (3.49)–(3.53). To prove the theorem it is sufficient to show:

(i)  $\mathbf{A} \models (3.54) - (3.55)$  iff  $\preceq^{\langle A; \div, 0 \rangle}$  and  $\preceq_{\mathcal{P}}$  coincide;

(ii)  $\mathbf{A} \models (3.54)$ -(3.55) implies  $\langle A; \land, 0 \rangle$  is a band with zero.

For (i), let  $a, b \in A$  and suppose  $A \models (3.54)-(3.55)$ . Suppose  $a \preceq^{(A; -, 0)} b$ . Then  $a = a \div 0$  (by (3.52))  $= a \div (a \div b) = (a \div (a \div b)) \land b \land (a \div (a \div b))$ (by (3.55))  $= (a \div 0) \land b \land (a \div 0) = a \land b \land a$  (by (3.52)). Hence  $a \preceq_{\mathcal{D}} b$ . For the opposite implication, assume  $a \preceq_{\mathcal{D}} b$ . Then  $a \land b \land a = a$ , whence  $a \div b = (a \land b \land a) \div b = 0$  by (3.54). Hence  $a \preceq^{(A; \div, 0)} b$  and thus  $a \preceq_{\mathcal{D}} b$  iff  $a \preceq^{(A; \div, 0)} b$ . Conversely, suppose the quasiorders  $\preceq_{\mathcal{D}} a$  and  $\preceq^{(A; \div, 0)}$ coincide. From  $a \div (a \div b) \preceq^{(A; \div, 0)} b$  we have  $a \div (a \div b) \preceq_{\mathcal{D}} b$ , which implies  $(a \div (a \div b)) \land b \land (a \div (a \div b)) = a \div (a \div b)$ . Also, from  $a \land b \land a \preceq_{\mathcal{D}} b$ we have  $a \land b \land a \preceq^{(A; \div, 0)} b$  whence  $(a \land b \land a) \div b = 0$ . Thus  $A \models$ (3.54)-(3.55).

For (ii), suppose  $\mathbf{A} \models (3.54) - (3.55)$ . We have  $b \land 0 \land b = (b \land 0 \land b) \div 0 = 0$ for any  $b \in A$  by (3.52) and (3.54). Thus  $0 = (a \land 0) \land 0 \land (a \land 0) = (a \land 0) \land (a \land 0) = a \land 0$  for any  $a \in A$ . But then  $0 = a \land 0 \land a = 0 \land a$ , so the reduct  $\langle A; \land, 0 \rangle$  is a band with zero.

An upper pre-BCK-band is an algebra  $\langle A; \lor, -, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \lor, 0 \rangle$  is a band with identity; (ii) the reduct  $\langle A; -, 0 \rangle$  is a pre-BCK-algebra; and (iii) the natural band quasiorder  $\preceq_{\mathcal{D}}$  dualises the pre-BCK quasiorder  $\preceq^{\langle A; -, 0 \rangle}$  in the sense that  $a \preceq_{\mathcal{D}} b$  iff  $b \preceq^{\langle A; -, 0 \rangle} a$ . Clearly an upper pre-BCK band **A** is an upper BCK-semilattice iff either  $\mathbf{A} \models (2.5)$  or  $\mathbf{A} \models x \lor y \approx y \lor x$ . The proof of the following theorem is similar to the proof of Theorem 3.3.3 and is omitted.

**Theorem 3.3.4.** An algebra  $\langle A; \land, \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is an upper pre-BCK-band iff the following identities are satisfied:

$$x \lor (y \lor z) \approx (x \lor y) \lor z \tag{3.56}$$

$$x \lor x \approx x \tag{3.57}$$

$$x \vee \mathbf{0} \approx x \tag{3.58}$$

$$\mathbf{0} \lor x \approx x \tag{3.59}$$

$$\left((x - y) - (x - z)\right) - (z - y) \approx 0 \tag{3.60}$$

$$x - \mathbf{0} \approx x \tag{3.61}$$

$$\mathbf{0} - x \approx \mathbf{0} \tag{3.62}$$

$$x \vee (y - (y - x)) \vee x \approx x \tag{3.63}$$

$$x - (y \lor x \lor y) \approx \mathbf{0}. \tag{3.64}$$

Thus the class of upper pre-BCK-bands is a variety.

**Remark 3.3.5.** The identities (3.58)-(3.59) cannot be omitted from the axiomatisation of the variety of upper pre-BCK bands given in Theorem 3.3.4. To see this, let  $A := \{0, 1, 2\}$ , let  $\mathbf{A} := \langle A; \vee^{\mathbf{A}}, \dot{-}^{\mathbf{A}}, \mathbf{0}^{\mathbf{A}} \rangle$  be the algebra with distinguished element 0 and whose binary operations  $\vee^{\mathbf{A}}$  and  $\dot{-}^{\mathbf{A}}$  are determined by the following operation tables:

| $\vee^{\mathbf{A}}$ | 0 | 1 | 2        | <u> </u> | 0 | 1 | 2 |
|---------------------|---|---|----------|----------|---|---|---|
| 0                   | 0 | 1 | 2        | 0        | 0 | 0 | 0 |
| 1                   | 1 | 1 | <b>2</b> | 1        | 1 | 0 | 0 |
| 2                   | 1 | 1 | 2        | 2        | 2 | 0 | 0 |

and let  $\mathbf{A}' := \langle A; \vee^{\mathbf{A}'}, \dot{-}^{\mathbf{A}'}, \mathbf{0}^{\mathbf{A}'} \rangle$  be the algebra of type  $\langle 2, 2, 0 \rangle$  obtained from  $\mathbf{A}$  upon defining  $\mathbf{0}^{\mathbf{A}'} := \mathbf{0}^{\mathbf{A}}, \ a \dot{-}^{\mathbf{A}'}b := a \dot{-}^{\mathbf{A}}b$  and  $a \vee^{\mathbf{A}'}b := b \vee^{\mathbf{A}} a$ for any  $a, b \in A$ . An easy sequence of checks shows  $\mathbf{A} \models (3.56)-(3.57)$ ,

(3.59)-(3.64). However  $\mathbf{A} \not\models$  (3.58) since  $2 \lor 0 = 1 \neq 2$ . From these remarks and the definition of  $\mathbf{A}'$  it follows at once that  $\mathbf{A}' \models$  (3.56)-(3.58), (3.60)-(3.64) but  $\mathbf{A} \not\models$  (3.59).

A quasilattice with zero is a quaritattice  $\langle A; \land, \lor \rangle$  for which there exists  $0 \in A$ (the zero of  $\langle A; \land, \lor \rangle$ ) such that both  $0 \leq_{\mathcal{H}}^{\langle A; \land \rangle} a$  and  $a \leq_{\mathcal{H}}^{\langle A; \lor \rangle} 0$  for all  $a \in A$ ; by abuse of language and notation we often identify a quasilattice with zero  $\mathbf{A} := \langle A; \land, \lor \rangle$  with the algebra  $\langle A; \land, \lor, 0 \rangle$  obtained from  $\mathbf{A}$  upon augmenting the language of  $\mathbf{A}$  with a new nullary operation symbol 0 whose canonical interpretation on  $\langle A; \land, \lor, 0 \rangle$  is the zero element  $0 \in A$ . A pre-BCK quasilattice is an algebra  $\langle A; \land, \lor, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \land, \lor, 0 \rangle$  is a quasilattice with zero; (ii) the reduct  $\langle A; \div, 0 \rangle$  is a pre-BCK-algebra; and (iii) the natural quasilattice quasiorder  $\preceq_{\mathcal{D}}$  coincides with the pre-BCK quasiorder  $\preceq^{\langle A; \div, 0 \rangle}$ . Clearly a pre-BCK quasilattice  $\mathbf{A}$  is a BCK-lattice iff either  $\mathbf{A} \models (2.5), \mathbf{A} \models x \land y \approx y \land x$  or  $\mathbf{A} \models x \lor y \approx y \lor x$ . From Theorem 3.3.3 and Theorem 3.3.4 the following result is clear.

**Theorem 3.3.6.** An algebra  $\mathbf{A} := \langle A; \land, \lor, \div, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  is a pre-BCK quasilattice iff the reduct  $\langle A; \land, \div, 0 \rangle$  is a lower pre-BCK band and the reduct  $\langle A; \lor, \div, 0 \rangle$  is an upper pre-BCK band. Hence  $\mathbf{A}$  is a pre-BCK quasilattice iff  $\mathbf{A} \models (3.49)$ -(3.64). Therefore the class of pre-BCK quasilattices is a variety.

Let C denote an arbitrary subset of the language  $\{\land, \lor, \div, 0\}$  of pre-BCK quasilattices that contains both  $\div$  and 0. Let:

- PQ<sub>c</sub> denote the variety of pre-BCK-algebras when  $C = \{ \div, 0 \}$ ;
- $PQ_{\mathcal{C}}$  denote the variety of lower pre-BCK bands when  $\mathcal{C} = \{\wedge, \div, 0\}$ ;
- PQ<sub>C</sub> denote the variety of upper pre-BCK bands when  $C = \{\vee, \div, 0\}$ ;
- $PQ_{\mathcal{C}}$  denote the variety of pre-BCK quasilattices when  $\mathcal{C} = \{\wedge, \lor, \div, 0\}$ .

Notice that for each C,  $PQ_C$  is the class of algebras with language C axiomatised by those identities among (3.49)–(3.64) that use only operation symbols from C; of course, this observation is dependent upon the axiomatisation of the variety of pre-BCK quasilattices given in Theorem 3.3.6.

**Remark 3.3.7.** Let A be an algebra of type  $\mathcal{L}$ . For  $\mathcal{L}' \subseteq \mathcal{L}$ , by an  $\mathcal{L}'$ -subreduct of A we mean a subalgebra of the reduct  $\mathbf{A}|_{\mathcal{L}'} := \langle A; f^{\mathbf{A}} \rangle_{f \in \mathcal{L}'}$ . For a class K of algebras, let  $\mathcal{L}'$ -K denote the class of all  $\mathcal{L}'$ -subreducts of members of K. For each  $\mathcal{C}$ , PQ<sub>C</sub> should not be confused with  $\mathcal{C}$ -PQ. In particular, in contrast to each PQ<sub>C</sub>, we do not know in general if each  $\mathcal{C}$ -PQ is even a quasivariety.

Throughout the remainder of this subsection we assume  $\Lambda \in C$  or  $\vee \in C$ . Recall from §1.4.3 that quasilattices satisfy the following modified form of the Clifford-McLean theorem: every quasilattice is a lattice of its maximal rectangular subalgebras. Because any member of any PQ<sub>C</sub> possesses a coherent  $\mathcal{D}$ -quasiordering by definition, it is natural to anticipate that the Clifford-McLean theorem for quasilattices extends to the varieties PQ<sub>C</sub>.

**Theorem 3.3.8 (Clifford-McLean Theorem for PQ\_C).** Let  $A \in PQ_C$ . For any  $a, b \in A$  the following are equivalent:

- 1. aDb;
- 2.  $a \equiv b$ .

Thus the following assertions hold:

- If ∧∈ C then D-equivalence is a congruence relation on ⟨A; ∧, ÷0⟩. The D-equivalence classes are the maximal rectangular subalgebras of ⟨A; ∧,0⟩, while the quotient algebra ⟨A; ∧, ÷,0⟩/D is the maximal lower BCK-semilattice homomorphic image of ⟨A; ∧, ÷,0⟩;
- If ∨∈ C then D-equivalence is a congruence relation on ⟨A; ∨, ∸, 0⟩. The D-equivalence classes are the maximal rectangular subalgebras of ⟨A; ∨, 0⟩, while the quotient algebra ⟨A; ∨, ∸, 0⟩/D is the maximal upper BCK-semilattice homomorphic image of ⟨A; ∨, ∸, 0⟩;
- 3. If {∧,∨} ⊆ C then D-equivalence is a congruence relation on A. The D-equivalence classes are the maximal rectangular subalgebras of ⟨A; ∧ ,∨, 0⟩, while the quotient algebra A/D is the maximal BCK-lattice homomorphic image of A.

**Proof.** Let  $A \in PQ_{\mathcal{C}}$ . For the first assertion of the theorem, just note  $a\mathcal{D}b$  iff  $a \preceq_{\mathcal{D}} b, b \preceq_{\mathcal{D}} a$  iff  $a \preceq^{\langle A; \div, 0 \rangle} b, b \preceq^{\langle A; \div, 0 \rangle} a$  iff  $a \equiv b$ . We prove the remaining assertions of the theorem only in the case that  $\{\wedge, \vee\} \subseteq \mathcal{C}$ ; the proofs in the other cases are not essentially different and are omitted. So suppose  $\{\wedge, \vee\} \subseteq \mathcal{C}$ . Since  $\Xi$  is a congruence on  $\langle A; \div, 0 \rangle$  and  $\mathcal{D}$  is a congruence on  $\langle A; \wedge, \vee, 0 \rangle$ , it follows from the first assertion of the theorem that  $\mathcal{D}$  is a congruence on A. By the Clifford-McLean theorem for quasilattices and Theorem 2.1.14, we deduce that  $A/\mathcal{D}$  is the maximal BCK-lattice homomorphic image of A and that the  $\mathcal{D}$ -equivalence classes are the maximal rectangular subalgebras of  $\langle A; \wedge, \vee, 0 \rangle$ . This establishes the theorem in the case that  $\{\wedge, \vee\} \subseteq \mathcal{C}$ , so the proof is complete.

Let A be a non-commutative lattice. In general, the four Green's equivalences  $\mathcal{L}_{(A;\Lambda)}$ ,  $\mathcal{R}_{(A;\Lambda)}$ ,  $\mathcal{L}_{(A;\Lambda)}$ ,  $\mathcal{L}_{(A;\Lambda)}$ ,  $\mathcal{L}_{(A;\Lambda)}$ ,  $\mathcal{L}_{(A;\Lambda)}$ ,  $\mathcal{R}_{(A;\Lambda)}$ ,  $\mathcal{L}_{(A;\Lambda)}$ ,  $\mathcal{R}_{(A;\Lambda)}$ ,  $\mathcal{L}_{(A;\Lambda)}$ ,  $\mathcal{R}_{(A;\Lambda)}$ ,  $\mathcal{R}_{(A;\Lambda)}$ ,  $\mathcal{L}_{(A;\Lambda)}$ ,  $\mathcal{L}_{(A;\Lambda)}$ , and  $\mathcal{R}_{(A;\Lambda)}$  are congruences on A; notice this definition is consistent with both §1.3.11 and Theorem 1.4.11. Necessary and sufficient conditions for the regularity of non-commutative lattices have been studied extensively by Laslo and Leech in [145, Section 4]. In more detail, recall from non-commutative lattice theory that a non-commutative lattice is *one-sided* if any one of the following pairs of identities is satisfied [145, Section 4]:

| $x \wedge y \wedge x \approx x \wedge y$ | and | $x \lor y \lor x \approx x \lor y$ | (1              | , ] |
|--|-----|------------------------------------|-----------------|-----|
| $x \land y \land x \sim x \land y$       | anu | $x \lor y \lor x \approx x \lor y$ | (1 <sub>)</sub> | ,   |

 $x \wedge y \wedge x \approx x \wedge y$  and  $x \vee y \vee x \approx y \vee x$  (l, r)

 $x \wedge y \wedge x \approx y \wedge x$  and  $x \vee y \vee x \approx x \vee y$  (r, l)

$$x \wedge y \wedge x \approx y \wedge x$$
 and  $x \vee y \vee x \approx y \vee x$ . (r, r)

A non-commutative lattice is *two-sided* if it is not one-sided. By Laslo and Leech [145, Theorem 20] one-sided quasilattices are regular. In turn, the class of all regular quasilattices is a variety [145, Theorem 19] that is generated by the class of all one-sided quasilattices [145, Theorem 20].

Remark 3.3.9. One-sided non-commutative lattices were introduced by Laslo

and Leech in [145, Section 4] under the name *flat* non-commutative lattices, in conformance with standard semigroup terminology. From the perspective of algebraic logic, however, the use of the adjective 'flat' to describe one-sided non-commutative lattices is unfortunate, for it conflicts with the established meaning of the term 'flat' as employed in domain theory. In particular, it conflicts with the term 'flat' as used in this thesis to describe algebras that are flat posets (and hence flat domains) with respect to some underlying partial ordering. We adopt alternative terminology here for this reason.

Let  $\mathbf{A} \in PQ_{\mathcal{C}}$ . By analogy with non-commutative lattice theory, for  $\Lambda \in \mathcal{C}$  $[\vee \in \mathcal{C}; \{\Lambda, \vee\} \subseteq \mathcal{C}]$  we say  $\langle A; \Lambda, \div, 0 \rangle$   $[\langle A; \vee, \div, 0 \rangle; \langle A; \Lambda, \vee, \div, 0 \rangle]$  is regular if  $\mathcal{L}_{\langle A; \Lambda \rangle}$  and  $\mathcal{R}_{\langle A; \Lambda \rangle}$   $[\mathcal{L}_{\langle A; \vee \rangle}, \mathcal{R}_{\langle A; \vee \rangle}; \mathcal{L}_{\langle A; \Lambda \rangle}, \mathcal{R}_{\langle A; \Lambda \rangle}, \mathcal{L}_{\langle A; \vee \rangle}, \mathcal{R}_{\langle A; \vee \rangle}]$  are congruences on  $\langle A; \Lambda, \div, 0 \rangle$   $[\langle A; \vee, \div, 0 \rangle; \langle A; \Lambda, \vee, \div, 0 \rangle]$ . For  $\mathcal{C} = \{\Lambda, \div, 0\}$   $[\mathcal{C} = \{\vee, \div, 0\}]$  we say  $\mathbf{A}$  is one-sided if its band reduct  $\langle A; \Lambda \rangle$   $[\langle A; \vee \rangle]$  is either left regular or right regular (recall §1.3.11). We say  $\mathbf{A}$  is two-sided if it is not one-sided. For  $\mathcal{C} = \{\Lambda, \vee, \div, 0\}$  we say  $\mathbf{A}$  is one-sided if its quasilattice reduct  $\langle A; \Lambda, \vee \rangle$  is one-sided;  $\mathbf{A}$  is two-sided if it is not one-sided.

## **Proposition 3.3.10.** Let $A \in PQ_{\mathcal{C}}$ . If A is one-sided then A is regular.

*Proof.* We prove the proposition only in the case that  $\{\wedge, \vee\} \subseteq C$ ; the proofs in the other cases are not essentially different and are omitted. Let  $\mathbf{A} \in \mathsf{PQ}_{\mathcal{C}}$ . By the Clifford-McLean theorem for  $\mathsf{PQ}_{\mathcal{C}}$ , the  $\mathcal{D}_{\mathbf{A}}$ -classes of  $\mathbf{A}$  form maximal rectangular bands in both  $\langle A; \wedge \rangle$  and  $\langle A; \vee \rangle$ . Since the identities  $(\mathbf{I}, \mathbf{I})$ - $(\mathbf{r}, \mathbf{r})$ are respectively equivalent to assorting that:

- $a \wedge b = a \ [a \vee b = a]$  in each  $\wedge$ -rectangular [ $\vee$ -rectangular] class;
- $a \wedge b = b$  [ $a \vee b = a$ ] in each  $\wedge$ -rectangular [ $\vee$ -rectangular] class;
- $a \wedge b = a \ [a \vee b = b]$  in each  $\wedge$ -rectangular [ $\vee$ -rectangular] class;
- $a \wedge b = b$  [ $a \vee b = b$ ] in each  $\wedge$ -rectangular [ $\vee$ -rectangular] class

it follows that each of  $\mathcal{L}_{\langle A; \wedge \rangle}$ ,  $\mathcal{R}_{\langle A; \wedge \rangle}$ ,  $\mathcal{L}_{\langle A; \vee \rangle}$  and  $\mathcal{R}_{\langle A; \vee \rangle}$  must be either  $\mathcal{D}_{\mathbf{A}}$  or  $\omega_{\mathbf{A}}$ . Hence all of  $\mathcal{L}_{\langle A; \wedge \rangle}$ ,  $\mathcal{R}_{\langle A; \wedge \rangle}$ ,  $\mathcal{L}_{\langle A; \vee \rangle}$  and  $\mathcal{R}_{\langle A; \vee \rangle}$  are congruences on  $\mathbf{A}$ , so  $\mathbf{A}$  is regular.

Let  $\mathbf{A} \in \mathsf{PQ}_{\mathcal{C}}$ . In general (that is, when  $\mathbf{A}$  is two-sided), the equivalences  $\mathcal{L}_{\langle A; \wedge \rangle}, \mathcal{R}_{\langle A; \wedge \rangle}, \mathcal{L}_{\langle A; \vee \rangle}$  and  $\mathcal{R}_{\langle A; \vee \rangle}$  (if they exist) need not be full congruences

on A. This is so even if A is a pre-BCK quasilattice whose quasilattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice. To see this, consider the following 5-element algebra A:

| $\wedge^{\mathbf{A}}$ | 0 | a | h | с | d | $\vee^{\mathbf{A}}$ | 0 | a | b | c | d | <u>.</u> A | 0 | $a_{\_}$ | b | С | d |
|-----------------------|---|---|---|---|---|---------------------|---|---|---|---|---|------------|---|----------|---|---|---|
| 0                     | 0 | 0 | 0 | 0 | 0 | 0                   | 0 | a | b | С | d | 0          | 0 | 0        | 0 | 0 | 0 |
| a                     | 0 | a | b | С | d | a                   | a | a | a | a | a | a          | a | 0        | a | d | 0 |
| b                     | 0 | b | b | b | b | b                   | b | a | b | С | d | b          | b | 0        | 0 | 0 | 0 |
| С                     | 0 | с | С | с | С | с                   | с | a | b | с | d | с          | c | 0        | 0 | 0 | 0 |
| d                     | 0 | a | b | с | d | d                   | d | d | d | d | d | d          | d | 0        | a | a | 0 |

An easy sequence of checks establishes that **A** is a pre-BCK quasilattice for which the quasilattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero. Moreover,  $a \equiv a \pmod{\mathcal{L}_{\langle A; \land, \lor}}$  and  $b \equiv c \pmod{\mathcal{L}_{\langle A; \land, \lor}}$  (since  $\mathcal{L}_{\langle A; \land} = \mathcal{R}_{\langle A; \lor \rangle}$  and  $\mathcal{R}_{\langle A; \land \rangle} = \mathcal{L}_{\langle A; \lor \rangle}$ —recall §1.4.5). However,  $(a \doteq b) \land (a \doteq c) = a \land d = d \neq a = a \doteq b$ , so  $\mathcal{L}_{\langle A; \land, \lor \rangle}$  is not a congruence on **A**.

For each C, the following proposition provides a sufficient condition for  $A \in PQ_C$  to be regular. Theorem 3.3.21 below shows the hypotheses of the proposition are not artificial; indeed, among natural syntactic conditions on members of  $PQ_C$  implying regularity, the assumptions of the proposition are the most general known to us.

**Proposition 3.3.11.** Let  $A \in PQ_{\mathcal{C}}$ . The following assertions hold:

1. If  $\wedge \in C$ ,  $\langle A; \wedge \rangle$  is regular and A satisfies:

$$(x - z) \wedge (y - z) \approx (x \wedge y) - z \tag{3.65}$$

$$(z - x) \wedge (z - y) \approx \left( (z - (x \wedge y)) - (x - y) \right) - (y - x) \quad (3.66)$$

then  $\langle A; \wedge, \div, 0 \rangle$  is regular. That is,  $\mathcal{L}_{\langle A; \wedge \rangle}$  and  $\mathcal{R}_{\langle A; \wedge \rangle}$  are congruences on  $\langle A; \wedge, \div, 0 \rangle$ ;

2. If  $\lor \in C$ ,  $\langle A; \lor \rangle$  is regular and A satisfies:

$$(x - z) \lor (y - z) \approx (x \lor y) - z \tag{3.67}$$

$$(z - x) \lor (z - y) \approx z - \left( \left( (x \lor y) - (x - y) \right) - (y - x) \right)$$
(3.68)

then  $\langle A; \vee, \div, 0 \rangle$  is regular. That is,  $\mathcal{L}_{\langle A; \vee \rangle}$  and  $\mathcal{R}_{\langle A; \vee \rangle}$  are congruences on  $\langle A; \vee, \div, 0 \rangle$ ;

If {∧, ∨} ⊆ C, ⟨A; ∧, ∨, 0⟩ is regular (in particular, if ⟨A; ∧, ∨, 0⟩ is a skew lattice with zero) and A satisfies (3.65)–(3.68), then A is regular. That is, L<sub>(A; ∧)</sub>, R<sub>(A; ∧)</sub>, L<sub>(A; ∨)</sub> and R<sub>(A; ∨)</sub> are congruences on A.

Proof. Let  $\mathbf{A} \in \mathsf{PQ}_{\mathcal{C}}$  and let  $a, b, c \in A$ . For (1), suppose  $\wedge \in \mathcal{C}$ ,  $\langle A; \wedge \rangle$  is regular and that  $\mathbf{A} \models (3.65), (3.66)$ . We show only that  $\mathcal{L}_{\langle A; \wedge \rangle}$  is a congruence on  $\langle A; \wedge, \div, 0 \rangle$ ; the proof that  $\mathcal{R}_{\langle A; \wedge \rangle}$  is a congruence on  $\langle A; \wedge, \div, 0 \rangle$  is similar and is omitted. Since  $\langle A; \wedge \rangle$  is regular, to see  $\mathcal{L}_{\langle A; \wedge \rangle}$  is a congruence on  $\langle A; \wedge, \div, 0 \rangle$  it is sufficient to show:

- (i)  $a \equiv b \pmod{\mathcal{L}_{\langle A; \land \rangle}}$  implies  $a \div c \equiv b \div c \pmod{\mathcal{L}_{\langle A; \land \rangle}};$
- (ii)  $a \equiv b \pmod{\mathcal{L}_{\langle A; \wedge \rangle}}$  implies  $c \div a \equiv c \div b \pmod{\mathcal{L}_{\langle A; \wedge \rangle}}$

since (i) and (ii) together guarantee the substitution property for the - operation. So suppose  $a \equiv b \pmod{\mathcal{L}_{\langle A; \wedge \rangle}}$ . For (i), we have  $(a \div c) \wedge (b \div c) = (a \wedge b) \div c = a \div c$  by (3.65); likewise  $(b \div c) \wedge (a \div c) = b \div c$ . For (ii), notice  $a \equiv b \pmod{\mathcal{L}_{\langle A; \wedge \rangle}}$  implies  $a \equiv b \pmod{\mathcal{D}}$  and hence  $a \equiv b \pmod{\Xi}$ . By (3.66),

$$(c \stackrel{\cdot}{-} a) \wedge (c \stackrel{\cdot}{-} b) = ((c \stackrel{\cdot}{-} (a \wedge b)) \stackrel{\cdot}{-} (a \stackrel{\cdot}{-} b)) \stackrel{\cdot}{-} (b \stackrel{\cdot}{-} a)$$
$$= ((c \stackrel{\cdot}{-} (a \wedge b)) \stackrel{\cdot}{-} 0) \stackrel{\cdot}{-} 0$$
$$= c \stackrel{\cdot}{-} (a \wedge b) \qquad \text{by (3.52)}$$
$$= c \stackrel{\cdot}{-} a.$$

Similarly  $(c \div b) \land (c \div a) = c \div b$ . Thus  $\mathcal{L}_{\langle A; \land \rangle}$  has the substitution property for the  $\div$  operation and  $\mathcal{L}_{\langle A; \land \rangle}$  is a congruence on  $\langle A; \land, \div, 0 \rangle$ .

For (2), suppose  $\forall \in C$ ,  $\langle A; \lor \rangle$  is regular and that  $\mathbf{A} \models (3.67), (3.68)$ . We show only that  $\mathcal{L}_{\langle A; \lor \rangle}$  is a congruence on  $\langle A; \lor, \div, 0 \rangle$ ; the proof that  $\mathcal{R}_{\langle A; \lor \rangle}$  is a congruence on  $\langle A; \lor, \div, 0 \rangle$ ; the proof that  $\mathcal{R}_{\langle A; \lor \rangle}$  is a congruence on  $\langle A; \lor, \div, 0 \rangle$  is analogous and is omitted. Since  $\langle A; \lor \rangle$  is regular, to see  $\mathcal{L}_{\langle A; \lor \rangle}$  is a congruence on  $\langle A; \lor, \div, 0 \rangle$  it is sufficient to show:

- (i)  $a \equiv b \pmod{\mathcal{L}_{\langle A; \vee \rangle}}$  implies  $a \div c \equiv b \div c \pmod{\mathcal{L}_{\langle A; \vee \rangle}};$
- (ii)  $a \equiv b \pmod{\mathcal{L}_{\langle A; \vee \rangle}}$  implies  $c \div a \equiv c \div b \pmod{\mathcal{L}_{\langle A; \vee \rangle}}$

since (i) and (ii) together guarantee the substitution property for the  $\div$  operation. So suppose  $a \equiv b \pmod{\mathcal{L}_{\langle A; \vee \rangle}}$ . For (i), we have  $(a \div c) \vee (b \div c) = (a \vee b) \div c = a \div c$  by (3.67);  $(b \div c) \vee (a \div c) = b \div c$  likewise. For (ii), notice  $a \equiv b \pmod{\mathcal{L}_{\langle A; \vee \rangle}}$  implies  $a \equiv b \pmod{\mathcal{D}}$  and hence  $a \equiv b \pmod{\Xi}$ . By (3.68),

$$(c - a) \vee (c - b) = c - (((a \vee b) - (a - b)) - (b - a))$$
$$= c - (((a \vee b) - 0) - 0)$$
$$= c - (a \vee b) \qquad by (3.52)$$
$$= c - a.$$

Similarly  $(c \div b) \lor (c \div a) = c \div b$ . Thus  $\mathcal{L}_{\langle A; \lor \rangle}$  has the substitution property for the  $\div$  operation and  $\mathcal{L}_{\langle A; \lor \rangle}$  is a congruence on  $\langle A; \lor, \div, 0 \rangle$ .

Item (3) now follows as a trivial consequence of (1), (2) and the regularity of  $\langle A; \land, \lor, 0 \rangle$ .

**Problem 3.3.12.** For each C, let  $K_C$  denote the class of all members of  $PQ_C$  that are regular. Is  $K_C$  equationally definable?

3.3.13. Implicative BCS Bands and Implicative BCS Quasilattices. By the results of §1.4.32, §1.4.37, §3.2.6 and §3.2.22, algebras arising in binary discriminator, pointed fixedpoint discriminator and pointed ternary discriminator varieties all support an underlying 'locally Boolean' structure (in the sense of either §1.3.15 or §1.4.24), which observation motivates the study of those members of PQ<sub>c</sub> for which every (appropriately defined) principal subalgebra is a Boolean battice. Because of the results of §2.3.19, these remarks lead naturally to a consideration of those members of PQ<sub>c</sub> for which the pre-BCK-algebra reduct is an implicative BCS-algebra.

**Proposition 3.3.14.** Let  $\mathbf{A} := \langle A; \land, \backslash, 0 \rangle$  be an algebra of type  $\langle 2, 2, 0 \rangle$  such that the reduct  $\langle A; \land, 0 \rangle$  is a band with zero and the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra. The following are equivalent:

1. For any  $a, b \in A$ ,

$$a \leq_{\mathcal{H}} b \text{ iff } a \leq^{\langle A; \backslash, 0 \rangle} b \text{ and } a \preceq_{\mathcal{D}} b \text{ iff } a \preceq^{\langle A; \backslash, 0 \rangle} b;$$

2. A satisfies the identity:

$$x \wedge y \wedge x \approx x \sqcap y. \tag{3.69}$$

*Proof.* Let  $\mathbf{A} := \langle A; \land, \backslash, 0 \rangle$  be an algebra of type  $\langle 2, 2, 0 \rangle$  such that the reduct  $\langle A; \land, 0 \rangle$  is a band with zero and the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra.

(1)  $\Rightarrow$  (2) To prove the implication, we first observe that for any  $a, b \in A$ ,

- (i)  $a \wedge b \wedge a \equiv a \sqcap b \pmod{\Xi}$ ;
- (ii)  $a \setminus (a \setminus (a \land b \land a)) = a \land b \land a$ .

For (i),  $a \sqcap b$  is a greatest lower bound of  $\{a, b\}$  with respect to  $\preceq^{\langle A; \backslash, 0 \rangle}$  by Proposition 2.3.5, Proposition 2.2.30 and Proposition 2.2.11. Also,  $a \land b \land a$ is a greatest lower bound of  $\{a, b\}$  with respect to  $\preceq^{\langle A; \backslash, 0 \rangle}$  since it is a greatest lower bound of  $\{a, b\}$  with respect to  $\preceq_{\mathcal{D}}$ . Therefore  $a \land b \land a \equiv a \sqcap b \pmod{\Xi}$ by Lemma 1.2.3(2).

For (ii), just note that  $a \wedge b \wedge a \leq_{\mathcal{H}} a$  implies  $a \wedge b \wedge a \leq^{\langle A_i \setminus , 0 \rangle} a$ , whence  $a \setminus (a \setminus (a \wedge b \wedge a)) = a \wedge b \wedge a$ .

To complete the proof of the implication, let  $a, b \in A$  and observe that  $a \wedge b \wedge a, a \sqcap b \in (a]_{\langle A; \backslash, 0 \rangle}$ . Because of Proposition 2.3.31, this implies the equivalence  $a \wedge b \wedge a \equiv a \sqcap b \pmod{(mod \Xi)}$  of (i) collapses in  $(a]_{\langle A; \backslash, 0 \rangle}$  to the equality  $a \wedge b \wedge a \equiv a \sqcap b$ . Hence  $a \backslash b = (a \sqcap b)^*_{(a]_{\langle A; \backslash, 0 \rangle}} = (a \wedge b \wedge a)^*_{(a]_{\langle A; \backslash, 0 \rangle}} = (a \cap (a \wedge b \wedge a))^*_{(a]_{\langle A; \backslash, 0 \rangle}}$  (by (ii)) =  $a \backslash (a \wedge b \wedge a)$ . But then  $a \wedge b \wedge a = a \backslash (a \backslash (a \wedge b \wedge a))$  (by (ii)) =  $a \backslash (a \backslash b)$  as required.

(2)  $\Rightarrow$  (1) To see  $a \leq \langle A; \backslash, 0 \rangle$  b iff  $a \leq_{\mathcal{H}} b$  for any  $a, b \in A$ , suppose  $a \leq \langle A; \backslash, 0 \rangle$  b. Then  $b \sqcap a = a$ , so  $a \land b = (b \sqcap a) \land b = (b \land a \land b) \land b = b \land a \land b = b \sqcap a = a$  by (3.69) applied twice. Similarly,  $b \land a = a$ . Therefore  $a \leq_{\mathcal{H}} b$ . For the converse, suppose  $a \leq_{\mathcal{H}} b$ . Then  $a \wedge b \wedge a = a$ , so  $a = a \wedge b \wedge a = a \sqcap b$ by (3.69). Similarly,  $a = b \sqcap a$ . Therefore  $a \leq^{\langle A; \backslash, 0 \rangle} b$ .

To see  $a \preceq^{\langle A; \backslash, 0 \rangle} b$  iff  $a \preceq_{\mathcal{D}} b$  for any  $a, b \in A$ , suppose  $a \preceq^{\langle A; \backslash, 0 \rangle} b$ . Then  $a \backslash b = 0$ , so  $a = a \backslash 0 = a \backslash (a \backslash b) = a \land b \land a$  by (3.69). Hence  $a \preceq_{\mathcal{D}} b$ . For the converse, suppose  $a \preceq_{\mathcal{D}} b$ . Then  $a \land b \land a = a$  and so  $a = a \land b \land a = a \sqcap b$  by (3.69). Therefore  $a \backslash b = 0$  by Lemma 2.1.42(1). Hence  $a \preceq^{\langle A; \backslash, 0 \rangle} b$ .

A lower implicative BCS band is an algebra  $\langle A; \wedge, \backslash, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \wedge, 0 \rangle$  is a band with zero; (ii) the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra; (iii) the natural band quasiorder  $\leq_{\mathcal{D}}$  coincides with the implicative BCS-algebra quasiorder  $\leq^{\langle A; \backslash, 0 \rangle}$ ; and (iv) the natural band partial order  $\leq_{\mathcal{H}}$  coincides with the implicative BCS-algebra partial order  $\leq^{\langle A; \backslash, 0 \rangle}$ . From Proposition 3.3.14 the following result is clear.

**Theorem 3.3.15.** An algebra  $\mathbf{A} := \langle A; \land, \backslash, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is a lower implicative BCS band iff the reduct  $\langle A; \land, 0 \rangle$  is a band with zero, the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra, and  $\mathbf{A} \models (3.69)$ . Thus the class of lower implicative BCS bands is a variety.

An upper implicative BCS band is an algebra  $\langle A; \land, \backslash, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \lor, 0 \rangle$  is a band with zero; (ii) the reduct  $\langle A; \backslash, 0 \rangle$ is an implicative BCS-algebra; (iii) the natural band quasiorder  $\preceq_{\mathcal{D}}$  dualises the implicative BCS-algebra quasiorder  $\preceq^{\langle A; \backslash, 0 \rangle}$  in the sense that  $a \preceq_{\mathcal{D}} b$  iff  $b \preceq^{\langle A; \div, 0 \rangle}$  a for any  $a, b \in A$ ; and (iv) the natural band partial order  $\leq_{\mathcal{H}}$ dualises the implicative BCS-algebra partial order  $\leq^{\langle A; \backslash, 0 \rangle}$  in the sense that  $a \preceq_{\mathcal{H}} b$  iff  $b \leq^{\langle A; \div, 0 \rangle} a$  for any  $a, b \in A$ . Although we know of no elegant characterisation of upper implicative BCS bands analogous to that of Theorem 3.3.15 in general, we nonetheless have the following result, the proof of which is omitted.

**Theorem 3.3.16.** An algebra  $\mathbf{A} := \langle A; \lor, \lor, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is an upper implicative BCS band iff the reduct  $\langle A; \lor, 0 \rangle$  is a band with zero, the reduct  $\langle A; \lor, 0 \rangle$  is an implicative BCS-algebra, and  $\mathbf{A}$  satisfies the following identities:

$$x \vee (y \backslash (y \backslash x)) \vee x \approx x \tag{3.70}$$

| 3.3. Pre-BCK Quasilattices and BCK Paralattices                         | 300    |  |  |
|---|--------|--|--|
| $x \setminus (y \lor x \lor y) \approx 0$                               | (3.71) |  |  |
| $x \setminus (x \setminus x \lor y \lor x)) \approx x$                  | (3.72) |  |  |
| $(x \lor y \lor x) \setminus ((x \lor y \lor x) \setminus x) \approx x$ | (3.73) |  |  |
|   |        |  |  |

$$x \lor (x \setminus y) \approx x \tag{3.74}$$

$$(x \setminus y) \lor x \approx x. \tag{3.75}$$

# Thus the class of upper implicative BCS bands is a variety.

Recall from [145, Section 1] that a *fine quasilattice* is a non-commutative lattice that is simultaneously both a quasilattice and a paralattice. By the remarks of §1.4.3 and §1.4.4, the class of all fine quasilattices is a variety. Fine quasilattices naturally generalise skew lattices, inasmuch as several important structural results for skew lattices extend to fine quasilattices: see Laslo and Leech [145, Section 5, pp. 28-29].

An implicative BCS quasilattice is an algebra of type  $\langle 2, 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \land, \lor, 0 \rangle$  is a fine quasilattice with zero; (ii) the reduct  $\langle A; \land, 0 \rangle$ is an implicative BCS-algebra; (iii) the natural quasilattice quasiorder  $\preceq_{\mathcal{D}}$  coincides with the implicative BCS-algebra quasiorder  $\preceq^{\langle A; \backslash, 0 \rangle}$ ; and (iv) the natural quasilattice partial order  $\leq_{\mathcal{H}}$  coincides with the implicative BCS-algebra partial order  $\leq^{\langle A; \backslash, 0 \rangle}$ . From Proposition 3.3.14 the following result is clear.

**Theorem 3.3.17.** An algebra  $\mathbf{A} := \langle A; \land, \lor, \backslash, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is an implicative BCS quasilattice iff the reduct  $\langle A; \land, \lor, 0 \rangle$  is a fine quasilattice with zero, the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra, and  $\mathbf{A} \models (3.69)$ . Thus the class of implicative BCS quasilattices is a variety.

For consistency with the prequel, let C' denote an arbitrary subset of the language  $\{\wedge, \lor, \backslash, 0\}$  of implicative BCS quasilattices that contains all of  $\wedge, \backslash$  and 0. Ignoring issues of type, let:

- $PQ_{C'}$  denote the variety of pre-BCK-algeleras when  $C' = \{\setminus, 0\}$ ;
- $PQ_{C'}$  denote the variety of lower pre-BCK bands when  $C' = \{\land, \backslash, 0\}$ ;

- $PQ_{C'}$  denote the variety of upper pre-BCK bands when  $C' = \{\land, \backslash, 0\}$ ;
- $PQ_{C'}$  denote the variety of pre-BCK quasilattices when  $C' = \{\land, \lor, \lor, 0\}$ .

Also, let:

- $IQ_{\mathcal{C}'}$  denote the variety of implicative BCS-algebras for  $\mathcal{C}' = \{ \setminus, 0 \}$ ;
- $IQ_{\mathcal{C}'}$  denote the variety of lower implicative BCS bands for  $\mathcal{C}' = \{\land, \backslash, \mathbf{0}\};$
- $IQ_{C'}$  denote the variety of upper implicative BCS bands for  $C' = \{\lor, \lor, \mathbf{0}\};$
- $|Q_{C'}|$  denote the variety of implicative BCS quasilattices for  $C' = \{\land, \lor, \lor, \lor, 0\}$ .

Given the above notation, clearly  $|Q_{\mathcal{C}'} \subseteq PQ_{\mathcal{C}'}$  for any fixed choice of  $\mathcal{C}'$ .

In non-commutative lattice theory, there exists a fundamental connection between principal subalgebras of normal bands in semigroup theory and the study of 'locally Boolean' structures: see Leech [150, Section 0.10]. For the varieties  $IQ_{C'}, \Lambda \in C'$ , however, it is the underlying implicative BCS-algebra principal subalgebra structure that is decisive. To see this, let  $\mathbf{A} \in PQ_{C'}, \Lambda \in C'$ , be such that the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra. Notice that in this case  $\mathbf{A}$  has both a band with zero reduct  $\langle A; \Lambda, 0 \rangle$  and a left normal band with zero polynomial reduct  $\langle A; \sqcap, 0 \rangle$ , where  $\langle A; \sqcap, 0 \rangle$  is determined by  $\langle A; \backslash, 0 \rangle$ as per Corollary 2.3.22(1). From this observation it follows that every  $a \in A$ generates both a principal subalgebra  $(a]_{\langle A; \Lambda, 0 \rangle}$  of the band with zero reduct  $\langle A; \Lambda, 0 \rangle$  (recall Lemma 1.3.13) and a principal subalgebra  $(a]_{\langle A; \backslash, 0 \rangle}$  of the left normal band with zero polynomial reduct  $\langle A; \sqcap, 0 \rangle$  (recall Proposition 2.3.31), whence we have the following proposition.

**Proposition 3.3.18.** Let  $\land \in C$  and let  $\mathbf{A} \in \mathsf{PQ}_{C'}$  be such that the reduct  $\langle A; \backslash, 0 \rangle$  of  $\mathbf{A}$  is an implicative BCS-algebra. Then  $\mathbf{A} \in \mathsf{IQ}_{C'}$  iff the principal subalgebras  $(\mathbf{a}]_{\langle A; \land, 0 \rangle}$  and  $(\mathbf{a}]_{\langle A; \backslash, 0 \rangle}$  coincide for each  $\mathbf{a} \in A$ .

*Proof.* Let  $\wedge \in C$  and let  $\mathbf{A} \in \mathsf{PQ}_{C'}$  be such that the reduct  $\langle A; \backslash, 0 \rangle$  of  $\mathbf{A}$  is an implicative BCS-algebra. Throughout the proof, we denote by  $(a]_{\langle A; \wedge, 0 \rangle}$  and  $(a]_{\langle A; \prec, 0 \rangle}$  the respective universes of the principal subalgebras  $(a]_{\langle A; \wedge, 0 \rangle}$ 

and  $(a]_{\langle A; -, 0 \rangle}$ . To simplify notation, we may also write simply  $(a]_A$  for both  $(a]_{\langle A; \wedge, 0 \rangle}$  and  $(a]_{\langle A; -, 0 \rangle}$  when these sets coincide.

 $(\Rightarrow)$  Suppose  $\mathbf{A} \in |Q_{C'}$ . To see the principal subalgebras  $(a]_{\langle A; \land, 0 \rangle}$  and  $(a]_{\langle A; \backslash, 0 \rangle}$  coincide for every  $a \in A$ , it is sufficient to show:

- (i) For any  $a \in A$ ,  $(a]_{\langle A; \land, 0 \rangle} = (a]_{\langle A; \backslash, 0 \rangle}$ ;
- (ii) If  $b, c \in (a]_{\mathbf{A}}$  then  $b \wedge c = b \sqcap c$ .

For (i), simply observe that for any  $a \in A$ ,  $(a]_{\langle A; \land, 0 \rangle} = \{b : b \leq_{\mathcal{H}} a\} = \{b : b \leq_{\mathcal{H}} a\} = \{b : b \leq_{\mathcal{H}} a\} = \{a\}_{\langle A; \backslash, 0 \rangle}$  because  $\leq_{\mathcal{H}}$  and  $\leq^{\langle A; \backslash, 0 \rangle}$  coincide.

For (ii), by (i) we have that  $(a]_{\langle A; \wedge, 0 \rangle}$  and  $(a]_{\langle A; -, 0 \rangle}$  coincide, so the reference to  $(a]_{\mathbf{A}}$  makes sense. Let  $b, c \in (a]_{\mathbf{A}}$ . From  $b \wedge c \equiv b \wedge c \wedge b \pmod{\mathcal{D}}$  we have that  $b \wedge c \equiv b \wedge c \wedge b \pmod{\Xi}$ . Also,  $b \wedge c \wedge b \equiv b \sqcap c \pmod{\Xi}$  by the proof of Theorem 3.3.14. Hence  $b \wedge c \equiv b \sqcap c \pmod{\Xi}$ . Now for any  $a \in A$  we have that the restriction of  $\Xi$  to  $(a]_{\langle A; \setminus, 0 \rangle}$  is the identity congruence on  $(a]_{\langle A; \setminus, 0 \rangle}$  (by Proposition 2.3.31), whence the equivalence  $b \wedge c \equiv b \sqcap c \pmod{\Xi}$  collapses to the equality  $b \wedge c = b \sqcap c$  as desired.

( $\Leftarrow$ ) Suppose the principal subalgebras  $(a]_{\langle A; \land, 0 \rangle}$  and  $(a]_{\langle A; \backslash, 0 \rangle}$  coincide for each  $a \in A$ . Since  $A \in PQ_{C'}$ , the quasiorders  $\preceq_{\mathcal{D}}$  and  $\preceq^{\langle A; \backslash, 0 \rangle}$  coincide by definition, so to establish the assertion we need only show that the partial orders  $\leq_{\mathcal{H}}$  and  $\leq^{\langle A; \backslash, 0 \rangle}$  coincide also. For this, just note that for any  $b, c \in A$ ,  $b \leq_{\mathcal{H}} c$  iff  $(b]_{\langle A; \land, 0 \rangle} \subseteq (c]_{\langle A; \land, 0 \rangle}$  iff  $(b]_{\langle A; \backslash, 0 \rangle} \subseteq (c]_{\langle A; \backslash, 0 \rangle}$  iff  $b \leq^{\langle A; \backslash, 0 \rangle} c$ .

Because of the proposition, for any  $A \in |Q_{C'}, A \in C$ , we may unambiguously denote by  $(a]_A$  the principal subalgebra generated by  $a \in A$ ; we follow this convention in the sequel.

**Corollary 3.3.19.** For any  $A \in IQ_{C'}$ , the principal subalgebra  $(a]_A$  generated by  $a \in A$  is a Boolean lattice. Consequently the band with zero reduct  $\langle A; \land, 0 \rangle$  is normal.

*Proof.* Let  $\mathbf{A} \in |\mathbf{Q}_{C'}$ . For every  $a \in A$ , the principal subalgebra  $(a]_{\mathbf{A}}$  generated by a is a Boolean lattice, because of Proposition 3.3.18 and Proposition 2.3.31. Hence  $\langle A: \wedge, 0 \rangle$  is normal (by Lemma 1.3.16), and the proof is complete.

Given the preceding corollary, throughout the remainder of this subsection we assume  $\Lambda \in \mathcal{C}'$ .

**Corollary 3.3.20.** Let  $\forall \in C'$  and let  $\mathbf{A} \in \mathbb{IQ}_{C'}$ . If the quasilattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero, then it is distributive local.

*Proof.* Let  $\forall \in C'$  and let  $\mathbf{A} \in |\mathbf{Q}_{C'}$  be such that the quasilattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero. It is clear that  $\langle A; \land, \lor, 0 \rangle$  is local. Since for each  $a \in A$ , the sublattice  $(a]_{\mathbf{A}}$  is distributive, from Proposition 1.4.22 we deduce that  $\langle A; \land, \lor, 0 \rangle$  is also distributive. Thus  $\langle A; \land, \lor, 0 \rangle$  is distributive local.

An implicative BCS skew lattice is an algebra  $\langle A; \land, \lor, \backslash, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$ such that: (i) the reduct  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero; (ii) the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra; (iii) the natural skew lattice quasiorder  $\preceq_{\mathcal{D}}$  coincides with the implicative BCS-algebra quasiorder  $\preceq^{\langle A; \backslash, 0 \rangle}$ ; and (iv) the natural skew lattice partial order  $\leq_{\mathcal{H}}$  coincides with the implicative BCS-algebra partial order  $\leq^{\langle A; \backslash, 0 \rangle}$ . Clearly the class of implicative BCS skew lattices is a subvariety of the variety of implicative BCS quasilattices.

Corollary 3.3.19 and Corollary 3.3.20 direct attention towards those members of  $|Q_{C'}, \forall \in C'$ , that are implicative BCS skew lattices, inasmuch as these algebras preserve several important structural properties of skew Boolean algebras. In particular, if  $\langle A; \land, \lor, \backslash, 0 \rangle$  is an implicative BCS skew lattice, then: (i) the skew lattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is distributive local; (ii) the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra; and (iii) for every  $a \in A$ , the principal subalgebra  $(a]_A$  generated by a is a Boolean sublattice. The precise relationship between implicative BCS skew lattices and skew Boolean algebras is clarified in the following theorem, a first-order proof of (a slightly less general form of) which may be found in [210, Section 4.2].

**Theorem 3.3.21.** An algebra  $\mathbf{A} := \langle A; \land, \lor, \backslash, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  is a skew Boolean algebra iff the following conditions are satisfied:

- 1. The reduct  $\langle A; \land, \lor, 0 \rangle$  is a join symmetric skew lattice with zero;
- 2. The reduct  $\langle A; \rangle, 0 \rangle$  is an implicative BCS algebra;
3. A satisfies the identity:

$$x \wedge y \wedge x \approx x \sqcap y. \tag{3.69}$$

310

Given assertions (1)-(3) above, implicative BCS difference coincides with standard difference.

# *Proof.* Let $\mathbf{A} := \langle A; \land, \lor, \backslash, 0 \rangle$ be an algebra of type $\langle 2, 2, 2, 0 \rangle$ .

(⇒) Suppose A is a skew Boolean algebra. Then the reduct  $\langle A; \land, \lor, 0 \rangle$  is a locally Boolean skew lattice (in the sense of §1.4.24) by definition and so is join symmetric. Also, the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra by the proof of Proposition 3.1.8. Moreover,  $\mathbf{A} \models (3.69)$  since  $2^p, 3^p_L, 3^p_R \models (3.69)$ . Hence A is an implicative BCS skew lattice.

( $\Leftarrow$ ) Suppose A satisfies Conditions (1)-(3) of the theorem. By (3) and Proposition 3.3.14,  $\mathbf{A} \in |Q_{C'}$ . By Corollary 3.3.20, therefore, the skew lattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is distributive local. By (1) and Lemma 1.4.17 it follows that  $\langle A; \land, \lor, 0 \rangle$  is symmetric. Since for every  $a \in A$ , the principal subalgebras  $(a]_{\langle A; \land, 0 \rangle}$  and  $(a]_{\langle A; \land, \lor, 0 \rangle}$  must coincide, by Corollary 3.3.19 we have that  $\langle A; \land, \lor, 0 \rangle$  is locally Boolean. To complete the proof it remains to show  $a \setminus b$  is the standard difference of  $a, b \in A$ . For this, just note that  $a \setminus b$  is the complement of  $a \sqcap b$  in  $(a]_{\langle A; \land, 0 \rangle}$  by (2) and Corollary 2.3.33, and hence that  $a \setminus b$  is the complement of  $a \land b \land a$  in  $(a]_{\langle A; \land, 0 \rangle}$  must coincide,  $a \setminus b$  is the complement of  $a \land b \land a$  in  $(a]_{\langle A; \land, 0 \rangle}$ . Hence  $a \setminus b$  is the standard difference of  $a, b \in A$ , and A is a skew Boolean algebra.

**Corollary 3.3.22.** An algebra  $\mathbf{A} := \langle A; \land, \lor, \backslash, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  is a left handed skew Boolean algebra iff the following conditions are satisfied:

- 1. The reduct  $\langle A; \land, \lor, 0 \rangle$  is a join symmetric skew lattice with zero;
- 2. The reduct  $\langle A; \rangle, 0 \rangle$  is an implicative BCS algebra;
- 3. The skew lattice meet  $\land$  coincides with the implicative BCS meet  $\sqcap$ .

Given assertions (1)–(3) above, implicative BCS difference coincides with standard difference.

**Remark 3.3.23.** The condition of join symmetry cannot be omitted from the assertion of Theorem 3.3.21 and its corollary since the variety of left handed skew Boolean algebras is properly contained within the class (in fact, variety) of all implicative BCS skew lattices for which the skew lattice reduct is left handed. To see this, consider the algebra  $\mathbf{A} := \langle A; \land, \lor, \backslash, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  with universe  $A := \{0, a, b, c, d, e, f\}$  and derived binary operation  $\sqcap$  defined by  $i \sqcap j := i \setminus (i \setminus j)$  for any  $i, j \in A$  determined by the following operation tables:

| $\wedge^{\mathbf{A}}$            | 0                               | a                               | b                               | c                               | d                               | e                               | f                               | $\vee^{\mathbf{A}}$                                      | 0                          | a                               | b                               | С                               | d                               | e                               | f                               |
|----------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|--|----------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|
| 0                                | 0                               | 0                               | 0                               | 0                               | 0                               | 0                               | 0                               | 0  | 0                          | a                               | b                               | С                               | d                               | e                               | f                               |
| a                                | 0                               | a                               | 0                               | a                               | a                               | 0                               | a                               | a  | a                          | a                               | d                               | С                               | d                               | С                               | f                               |
| b                                | 0                               | 0                               | b                               | b                               | b                               | b                               | 0                               | b  | b                          | С                               | b                               | С                               | d                               | e                               | đ                               |
| С                                | 0                               | a                               | е                               | С                               | с                               | е                               | a                               | С  | c                          | с                               | d                               | С                               | d                               | с                               | d                               |
| d                                | 0                               | f                               | b                               | d                               | d                               | b                               | f                               | d  | d                          | с                               | d                               | С                               | d                               | с                               | d                               |
| e                                | 0                               | 0                               | e                               | е                               | е                               | e                               | 0                               | е  | e                          | с                               | b                               | С                               | d                               | e                               | d                               |
| f                                | 0                               | f                               | 0                               | f                               | f                               | 0                               | <u>f</u>                        | f  | f                          | a                               | d                               | с                               | d                               | с                               | f                               |
|                                  |                                 |                                 |                                 |                                 |                                 |                                 |                                 |  |                            |                                 |                                 |                                 |                                 |                                 |                                 |
|                                  |                                 |                                 |                                 |                                 |                                 |                                 |                                 |  |                            |                                 |                                 |                                 |                                 |                                 |                                 |
| \ <b>A</b>                       | 0                               | a                               | b                               | с                               | d                               | е                               | f                               | □▲   | 0                          | a                               | b                               | с                               | d                               | e                               | <u></u>                         |
| \ <b>A</b><br>0                  | 0<br>0                          | a<br>0                          | <i>b</i><br>0                   | <i>с</i><br>0                   | <i>d</i><br>0                   | е<br>0                          | <i>f</i><br>0                   | <b>^</b> ▲   | 0                          | a<br>0                          | <i>b</i><br>0                   | с<br>0                          | <i>d</i><br>0                   | е<br>0                          | <u>f</u><br>0                   |
| A<br>0<br>a                      | 0<br>0<br>a                     | a<br>0<br>0                     | b<br>0<br>a                     | с<br>0<br>0                     | <i>d</i><br>0<br>0              | е<br>0<br>а                     | <i>f</i><br>0<br>0              | <b>A</b> 0   | 0<br>0<br>0                | a<br>0<br>a                     | b<br>0<br>0                     | с<br>0<br>а                     | d<br>0<br>a                     | е<br>0<br>0                     | f<br>0<br>a                     |
| \A<br>0<br>a<br>b                | 0<br>0<br>a<br>b                | a<br>0<br>0<br>b                | b<br>0<br>a<br>0                | <i>c</i><br>0<br>0              | <i>d</i><br>0<br>0<br>0         | e<br>0<br>a<br>0                | f<br>0<br>0<br>b                | $ \begin{array}{c}                                     $ | 0<br>0<br>0<br>0           | a<br>0<br>a<br>0                | b<br>0<br>0<br>b                | с<br>0<br>а<br>b                | d<br>0<br>a<br>b                | e<br>0<br>0<br>b                | f<br>0<br>a<br>0                |
| \A<br>0<br>a<br>b<br>c           | 0<br>0<br>a<br>b<br>c           | a<br>0<br>0<br>b<br>e           | b<br>0<br>a<br>0<br>a           | c<br>0<br>0<br>0<br>0           | <i>d</i><br>0<br>0<br>0         | e<br>0<br>a<br>0<br>a           | f<br>0<br>0<br>b<br>e           | $ \begin{array}{c}                                     $ | 0<br>0<br>0<br>0           | a<br>0<br>a<br>0<br>a           | b<br>0<br>0<br>b<br>e           | с<br>0<br>а<br>b<br>с           | d<br>0<br>a<br>b<br>c           | e<br>0<br>0<br>b<br>e           | f<br>0<br>a<br>0<br>a           |
| \A<br>0<br>a<br>b<br>c<br>d      | 0<br>0<br>a<br>b<br>c<br>d      | a<br>0<br>0<br>b<br>e<br>b      | b<br>0<br>a<br>0<br>a<br>f      | c<br>0<br>0<br>0<br>0<br>0      | d<br>0<br>0<br>0<br>0<br>0      | e<br>0<br>a<br>0<br>a<br>f      | f<br>0<br>0<br>b<br>e<br>b      | $ \begin{array}{c}                                     $ | 0<br>0<br>0<br>0<br>0      | a<br>0<br>a<br>0<br>a<br>f      | b<br>0<br>0<br>b<br>e<br>b      | c<br>0<br>a<br>b<br>c<br>d      | d<br>0<br>a<br>b<br>c<br>d      | e<br>0<br>0<br>b<br>e<br>b      | f<br>0<br>a<br>0<br>a<br>f      |
| \A<br>0<br>a<br>b<br>c<br>d<br>e | 0<br>0<br>a<br>b<br>c<br>d<br>e | a<br>0<br>0<br>b<br>e<br>b<br>e | b<br>0<br>a<br>0<br>a<br>f<br>0 | c<br>0<br>0<br>0<br>0<br>0<br>0 | d<br>0<br>0<br>0<br>0<br>0<br>0 | e<br>0<br>a<br>0<br>a<br>f<br>0 | f<br>0<br>0<br>b<br>e<br>b<br>e | $ \begin{array}{c}                                     $ | 0<br>0<br>0<br>0<br>0<br>0 | a<br>0<br>a<br>0<br>a<br>f<br>0 | b<br>0<br>0<br>b<br>e<br>b<br>e | c<br>0<br>a<br>b<br>c<br>d<br>e | d<br>0<br>a<br>b<br>c<br>d<br>e | e<br>0<br>0<br>b<br>e<br>b<br>e | f<br>0<br>a<br>0<br>a<br>f<br>0 |

An easy sequence of checks shows: (i) the reduct  $\langle A; \wedge, \vee, 0 \rangle$  is a left handed skew lattice with zero; (ii) the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra (with implicative BCS meet  $\sqcap$ ); and (iii)  $i \leq_{\mathcal{H}} j$  iff  $i \leq^{\langle A; \backslash, 0 \rangle} j$  and  $i \leq_{\mathcal{D}} j$  iff  $i \leq^{\langle A; \backslash, 0 \rangle} j$  for any  $i, j \in A$ . Hence **A** is an implicative BCS skew lattice for which the skew lattice reduct is left handed. However, **A** is not a left handed skew Boolean algebra, since the skew lattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is not locally Boolean (in the sense of §1.4.24). In particular,  $\langle A; \land, \lor, 0 \rangle$  is not symmetric, since it is not join symmetric:  $(a \lor b) \lor (a \land b \land a) = d \lor 0 = d$ but  $(a \land b \land a) \lor (b \lor a) = 0 \lor c = c$ .

Denote by IPQ the variety of (l, r)-pre-BCK quasilattices, namely the subvariety of  $PQ_{C'}$ ,  $C' = \{\land, \lor, \backslash, \mathbf{0}\}$ , satisfying the identities (l, r); also, recall the definition of Idziak's variety BL of BCK-lattices from §1.6.23. Perhaps surprisingly, there exists an intimate connection between the variety of left handed skew Boolean algebras and the variety of BCK-lattices (in the context of the variety IrPQ). In more detail, the variety of left handed skew Boolean algebras is the natural conjugate of BL in  $\Lambda^{V}(IrPQ)$  in the same way that BCK is the natural conjugate of the quasivariety  $Q(B_2)$  in  $\Lambda^{V}(PBCK)$  (recall Proposition 2.2.5). This is shown by the following proposition, in the statement of which (and throughout the remainder of this subsection), the type of Idziak's variety BL is understood to be  $\{\land, \lor, \backslash, \mathbf{0}\}$ .

**Proposition 3.3.24.** The pair  $\langle V(3_L^p), BL \rangle$  [equivalently  $\langle IhSBA, BL \rangle$ ] splits the lattice of varieties  $\Lambda^V(IrPQ)$  of (I, r)-pre-BCK quasilattices. Thus for every variety  $K \subseteq IrPQ$ , either  $K \subseteq BL$  or  $V(3_L^p) \subseteq K$  [equivalently  $IhSBA \subseteq K$ ] (and not both).

Proof. Suppose  $K \in \Lambda^{V}(\operatorname{IrPQ})$  and  $K \not\subseteq BL$ . By hypothesis, K does not satisfy (2.5), so there is an algebra  $\mathbf{A} \in K$  and there are elements  $a, b \in A$ such that  $a \setminus b = 0 = b \setminus a$  but  $a \neq b$ . Therefore  $a \equiv b \pmod{\Xi}$  and so  $a \equiv b \pmod{D}$ . Thus  $a \wedge b \wedge a = u$  and  $b \wedge a \wedge b = b$ ; because  $\mathbf{A} \models (\mathbf{l}, \mathbf{r})$ we infer  $a \wedge b = a$  and  $b \wedge a = b$ . Similarly we deduce that  $a \vee b = b$  and  $b \vee a = a$ . Because the reduct  $\langle A; \wedge, \vee, 0 \rangle$  is a quasilattice with zero, we conclude that  $\{0, a, b\}$  is the universe of a subalgebra of  $\mathbf{A}$  isomorphic to  $\mathbf{3}_{L}^{p}$ . Hence  $\mathbf{3}_{L}^{p} \in K$  and, by Theorem 1.4.29,  $|\mathsf{hSBA} = \mathbf{V}(\mathbf{3}_{L}^{p}) \subseteq K$ .

**Remark 3.3.25.** BL is also the splitting variety associated with the 3-element right handed skew Boolean algebra  $3_R^p$  in the variety of (r, l)-pre-BCK quasilattices, namely the subvariety of  $PQ_{C'}$ ,  $C' = \{\land, \lor, \backslash, 0\}$ , satisfying the identities (r, l). **Corollary 3.3.26.** For a variety  $\vee$  of (1, r)-pre-BCK quasilattices the following statements hold:

- 1. V is 0-regular iff  $V \subseteq BL$ ;
- 2. V is congruence distributive iff  $V \subseteq BL$ ;
- 3. V is congruence n-permutable for some  $n \ge 2$  iff  $V \subseteq BL$ . In particular, if  $V \subseteq BL$  then V is congruence permutable.

Proof. Let V be a variety of (l, r)-pre-BCK quasilattices. Suppose V  $\subseteq$  BL. Put  $d_1(x, y) := x \setminus y$  and  $d_2(x, y) := y \setminus x$ . Then  $d_1(x, y)$  and  $d_2(x, y)$  are binary terms of V satisfying the identities and quasi-identities of Proposition 1.2.6, so V is 0-regular. Because V is subtractive (witness  $x \setminus y$ ), from Proposition 1.7.3 we infer that V is ideal determined. Since V  $\models (1.3), (1.4), (1.7), V$  is ideal distributive by the remarks of §1.7.4, so V is congruence distributive by ideal determinacy. Moreover,  $p(x, y, z) := (x \setminus (y \setminus z)) \lor (z \setminus (y \setminus x))$  is a Mal'cev term for V by Idziak [116, Theorem 2], so V is congruence permutable. For the converse, suppose V  $\not\subseteq$  BL. Then IhSEA =  $V(3_L^p) \subseteq V$  by Proposition 3.3.24. Since the variety of left handed skew Boolean algebras is not 0-regular, V is not 0-regular. Moreover, because the variety of left handed skew Boolean algebras satisfies no non-trivial congruence identities (by Lemma 4.8 of Cornish [65]) V does not satisfy any non-trivial congruence identity. In particular, V is not congruence *n*-permutable for any  $n \geq 2$ .

Corollary 3.3.26 in conjunction with the theory of BCK-lattices shows that several important properties of BCK-lattices do not extend to pre-BCK quasilattices. On the other hand, by Theorem 3.3.8 the varieties  $PQ_C$  satisfy a modified form of the Clifford-McLean theorem, while by Proposition 3.3.10, one-sided members of any  $PQ_C$  are regular. Thus fundamental properties of bands and quasilattices are preserved by pre-BCK quasilattices. Collectively, these observations suggest that, for pre-BCK bands and pre-BCK quasilattices, the behaviour of these algebras is more closely aligned to that of bands and quasilattices than to that or BCK-semilattices and BCK-lattices. We return to this point in the following subsection. **3.3.27.** BCK Bands and BCK Paralattices. Recall from Lemma 1.6.2 and Proposition 1.6.4 that for any BCK-algebra A, there exist distinct partial orders  $\leq_n$ , n = 0, 1, 2 on A, where for any  $a, b \in A$ ,

 $a \leq_0 b$  iff ab = 0 $a \leq_1 b$  iff  $b \cap a = a$  $a \leq_2 b$  iff  $(b \cap a)(ba) = a$ .

A lower  $\leq_n$ -BCK band, n = 0, 1, 2, is an algebra  $\langle A; \land, \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$ such that: (i) the reduct  $\langle A; \land, 0 \rangle$  is a band with zero; (ii) the reduct  $\langle A; \div, 0 \rangle$ is a BCK-algebra; and (iii) the natural band partial order  $\leq_{\mathcal{H}}$  coincides with the BCK partial order  $\leq_n^{\langle A; \div, 0 \rangle}$ .

**Theorem 3.3.28.** An algebra  $\langle A; \land, \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is a lower  $\leq_0$ -BCK band iff the following identities are satisfied:

 $x \wedge (y \wedge z) \approx (x \wedge y) \wedge z \tag{3.76}$ 

$$x \wedge x \approx x \tag{3.77}$$

$$\left((x - y) - (x - z)\right) - (z - y) \approx 0 \tag{3.78}$$

$$x - \mathbf{0} \approx x \tag{3.79}$$

$$\mathbf{0} - \mathbf{x} \approx \mathbf{0} \tag{3.80}$$

$$\left(x - (x - y)\right) \wedge y \approx x - (x - y) \tag{3.81}$$

$$y \wedge \left(x - (x - y)\right) \approx x - (x - y) \tag{3.82}$$

$$(x \wedge y \wedge x) - x \approx 0. \tag{3.83}$$

Thus the class  $c_j$  tower  $\leq_0$ -BCK bands is a variety.

*Proof.* Let  $\mathbf{A} = \langle A; \land, \div, 0 \rangle$  be an algebra of type  $\langle 2, 2, 0 \rangle$  satisfying (3.76)-(3.80) and let  $a, b \in A$ . To prove the theorem it is sufficient to show:

(i)  $A \models (3.31)$ -(3.83) implies  $\langle A; \div, 0 \rangle$  is a BCK-algebra;

- (ii)  $\mathbf{A} \models (3.81) (3.83)$  iff  $\leq_0^{\langle A; \div, 0 \rangle}$  and  $\leq_{\mathcal{H}}$  coincide;
- (iii)  $\mathbf{A} \models (3.81)$ -(3.83) implies  $\langle A; \land, 0 \rangle$  is a band with zero.

For (i), let  $a, b \in A$  and suppose  $\mathbf{A} \models (3.81)-(3.83)$ . To see  $\langle A; \div, 0 \rangle$  is a BCK-algebra it is sufficient to show  $a \div b = 0 = b \div a$  implies a = b, just because  $\langle A; \div, 0 \rangle$  is a pre-BCK-algebra by (3.78)-(3.80). So let  $a \div b = 0 = b \div a$ . Then  $a \land b = (a \div 0) \land b$  (by (3.79)) =  $(a \div (a \div b)) \land b = a \div (a \div b)$  (by (3.81)) =  $a \div 0 = a$  (by (3.79)). Also  $a \land b = a \land (b \div 0)$  (by (3.79)) =  $a \land (b \div (b \div a)) = b \div (b \div a)$  (by (3.82)) =  $b \div 0 = b$  (by (3.79)). Hence  $a = a \land b = b$ .

For (ii), suppose  $\mathbf{A} \models (3.81) - (3.83)$ . By (i),  $\langle A; \div, 0 \rangle$  is a BCK-algebra, so the reference to  $\leq_{0}^{\langle A; \div, 0 \rangle}$  makes sense. Assume  $a \leq_{0}^{\langle A; \div, 0 \rangle} b$ . Then  $a = a \div 0$  (by (3.79)) =  $a \div (a \div b) = (a \div (a \div b)) \land b$  (by (3.81)) =  $(a \div 0) \land b$  (by (3.79)) =  $a \land b$ . Also  $b \land a = b \land (a \div 0)$  (by (3.79)) =  $b \land (a \div (a \div b)) = a \div (a \div b)$  (by (3.82)) =  $a \div 0 = a$  (by (3.79)). Hence  $a \leq_{0}^{\langle A; \div, 0 \rangle} b$  implies  $a \leq_{\mathcal{H}} b$ . For the opposite implication assume  $a \leq_{\mathcal{H}} b$ . Then  $a \land b = a = b \land a$ , so  $b \land a \land b = a$  and  $a \div b =$  $(b \land a \land b) \div b = 0$  by (3.83). Thus  $a \leq_{0}^{\langle A; \div, 0 \rangle} b$  and so  $a \leq_{\mathcal{H}}$  iff  $a \leq_{0}^{\langle A; \div, 0 \rangle} b$ . Conversely, suppose the partial orders  $\leq_{\mathcal{H}} and \leq_{0}^{\langle A; \div, 0 \rangle} co$ incide. From  $(a \div (a \div b)) \leq_{0}^{\langle A; \div, 0 \rangle} b$  we have  $(a \div (a \div b)) \leq_{\mathcal{H}} b$ , which implies  $(a \cdots (a \div b)) \land b = a \div (a \div b) = b \land (a \div (a \div b))$ . Moreover, from  $a \land b \land a \leq_{\mathcal{H}} a$  we have  $a \land b \land a \leq_{0}^{\langle A; \div, 0 \rangle} a$  and thus  $(a \land b \land a) \div a = 0$ . Hence  $\mathbf{A} \models (3.81)$ -(3.83).

For (iii), suppose  $\mathbf{A} \models (3.81) - (3.83)$ . Then  $\langle A; \div, 0 \rangle$  is a BCK-algebra by (i), so  $0 \leq_{0}^{\langle A; \div, 0 \rangle} a$  for all  $a \in A$  by Lemma 1.6.2. By (ii),  $0 \leq_{\mathcal{H}} a$  for all  $a \in A$ , so  $\langle A; \land, 0 \rangle$  is a band with zero.

For a lower  $\leq_1$ -BCK band A and  $a, b \in A$ , an argument similar to the proof of Theorem 3.3.28 shows that  $a \leq_1^{\langle A; \div, 0 \rangle} b$  implies  $a \leq_{\mathcal{H}} b$  iff A satisfies the identicies:

 $(x - y) \wedge x \approx x - y \tag{3.84}$ 

 $x \wedge (x - y) \approx x - y. \tag{3.85}$ 

Conversely,  $a \leq_{\mathcal{H}} b$  implies  $a \leq_{1}^{\langle A; \div, 0 \rangle} b$  iff A satisfies the identity:

$$x - (x - (x \wedge y \wedge x)) \approx x \wedge y \wedge x.$$
(3.86)

(For a lower  $\leq_2$ -BCK band, we know of no corresponding equational characterisation of the coincidence of the partial orders  $\leq_2^{\langle A; -, 0 \rangle}$  and  $\leq_{\mathcal{H}}$ .) Nonetheless, the identities (3.76)-(3.80), (3.84)-(3.86) are not sufficient to ensure satisfaction of the quasi-identity (1.5). To see this, consider  $\langle \wedge, \backslash, \mathbf{0} \rangle$ - $\mathbf{3}_L^p$ , the  $\langle \wedge, \backslash, \mathbf{0} \rangle$ -reduct of the 3-element left handed primitive skew Boolean algebra  $\mathbf{3}_L^p$ . An easy sequence of checks (ignoring issues of type) shows  $\langle \wedge, \backslash, \mathbf{0} \rangle$ - $\mathbf{3}_L^p \models (3.76)-(3.80), (3.84)-(3.86)$  but that  $\mathbf{A} \not\models (1.5)$ .

**Problem 3.3.29.** For n = 1, 2, is the class of lower  $\leq_n$ -BCK bands equationally definable?

An upper  $\leq_n$ -BCK band, n = 0, 1, 2, is an algebra  $\langle A; \lor, -, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$ such that: (i) the reduct  $\langle A; \lor, 0 \rangle$  is a band with identity; (ii) the reduct  $\langle A; -, 0 \rangle$  is a BCK-algebra; and (iii) the natural band partial order  $\leq_{\mathcal{H}}$ dualises the BCK partial order  $\leq_n^{\langle A; -, 0 \rangle}$  in the sense that  $a \leq_{\mathcal{H}} b$  iff  $b \leq_n^{\langle A; -, 0 \rangle}$ a for any  $a, b \in A$ . The proof of the following theorem is similar to the proof of Theorem 3.3.28 and is omitted.

**Theorem 3.3.30.** An algebra  $\langle A; \vee, \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is an upper  $\leq_{0^{-}}$ BCK band iff the following identities are satisfied:

$$x \lor (y \lor z) \approx (x \lor y) \lor z \tag{3.87}$$

$$x \lor x \approx x \tag{3.88}$$

$$\left((x - y) - (x - z)\right) - (z - y) \approx \mathbf{0} \tag{3.89}$$

$$x - \mathbf{0} \approx x \tag{3.90}$$

$$\mathbf{0} - \mathbf{x} \approx \mathbf{0} \tag{3.91}$$

 $\left(x - (x - y)\right) \lor y \approx y \tag{3.92}$ 

$$y \vee \left(x - (x - y)\right) \approx y \tag{3.93}$$

$$x - (x \vee y \vee x) \approx 0. \tag{3.94}$$

Thus the class of upper  $\leq_0$ -BCK bands is a variety.

For n = 1, 2, remarks concerning lower  $\leq_n$ -BCK bands apply mutatis mutandis to upper  $\leq_n$ -BCK bands. In particular, for an upper  $\leq_1$ -BCK band A and  $a, b \in A, a \leq_n^{\langle A; -, 0 \rangle} b$  implies  $b \leq_{\mathcal{H}} a$  iff A satisfies the identities:

$$(x - y) \lor x \approx x \tag{3.95}$$

$$x \lor (x - y) \approx x. \tag{3.96}$$

Conversely,  $a \leq_{\mathcal{H}} b$  implies  $b \leq_n^{(A; \div, 0)} a$  iff A satisfies the identity:

$$(x \vee y \vee x) \stackrel{\cdot}{-} ((x \vee y \vee x) \stackrel{\cdot}{-} x) \approx x. \tag{3.97}$$

(For an upper  $\leq_2$ -BCK band, we know of no corresponding equational characterisation of the dualisation of the partial orders  $\leq_2^{\langle A; -, 0 \rangle}$  and  $\leq_{\mathcal{H}}$ .) However, the identities (3.87)-(3.91), (3.95)-(3.97) are not sufficient to ensure satisfaction of the quasi-identity (2.5). To see this, consider  $\langle \vee, \backslash, \mathbf{0} \rangle$ - $\mathbf{3}_R^p$ , the  $\langle \vee, \backslash, \mathbf{0} \rangle$ -reduct of the 3-element right handed primitive skew Boolean algebra  $\mathbf{3}_R^p$ . An easy sequence of checks (ignoring issues of type) shows that  $\langle \vee, \backslash, \mathbf{0} \rangle$ - $\mathbf{3}_R^p \models$  (3.87)-(3.91), (3.95)-(3.97) but that  $\mathbf{A} \not\models$  (1.5).

**Problem 3.3.31.** For n = 1, 2, is the class of upper  $\leq_n$ -BCK bands equationally definable?

A paralattice with zero is a paralattice  $\langle A; \land, \lor \rangle$  for which there exists  $0 \in A$ (the zero of  $\langle A; \land, \lor \rangle$ ) such that 0 is the least element under the natural paralattice partial order. As is usual, by abuse of language and notation we often identify a paralattice with zero  $\mathbf{A} := \langle A; \land, \lor \rangle$  with the algebra  $\langle A; \land$  $,\lor, 0\rangle$  obtained from  $\mathbf{A}$  upon enriching the language of  $\mathbf{A}$  with a new nullary operation symbol 0 whose canonical interpretation on  $\langle A; \land, \lor, 0\rangle$  is the zero element  $0 \in A$ . A  $\leq_0$ -BCK paralattice is an algebra  $\langle A; \land, \lor, \circ, 0\rangle$  of type  $\langle 2, 2, 2, 0\rangle$  such that: (i) the reduct  $\langle A; \land, \lor, 0\rangle$  is a paralattice with zero; (ii)

the reduct  $\langle A; \div, 0 \rangle$  is a BCK-algebra; and (iii) the natural paralattice partial order  $\leq_{\mathcal{H}}$  coincides with the BCK partial order  $\leq_{0}^{\langle A; \div, 0 \rangle}$ . From Theorem 3.3.28 and Theorem 3.3.30 the following result is clear.

**Theorem 3.3.32.** An algebra  $\mathbf{A} := \langle A; \land, \lor, \div, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  is a  $\leq_0$ -BCK paralattice iff the reduct  $\langle A; \land, \div, 0 \rangle$  is a lower  $\leq_0$ -BCK band and the reduct  $\langle A; \lor, \div, 0 \rangle$  is an upper  $\leq_0$ -BCK band. Thus  $\mathbf{A}$  is a  $\leq_0$ -BCK paralattice iff  $\mathbf{A} \models (3.76) - (3.83), (3.87) - (3.94)$ . Therefore the class of  $\leq_0$ -BCK paralattices is a variety.

Let C denote an arbitrary subset of the language  $\{\wedge, \lor, \div, 0\}$  of  $\leq_0$ -BCK paralattices that contains both  $\div$  and 0. Let:

- $BP_{\mathcal{C}}$  denote the quasivariety of BCK-algebras when  $\mathcal{C} = \{ \div, 0 \}$ ;
- BP<sub>C</sub> denote the variety of lower  $\leq_0$ -BCK bands when  $\mathcal{C} = \{\land, \div, 0\};$
- $BP_{\mathcal{C}}$  denote the variety of upper  $\leq_0$ -BCK bands when  $\mathcal{C} = \{\vee, \div, 0\};$
- $BP_{\mathcal{C}}$  denote the variety of  $\leq_0$ -BCK paralattices when  $\mathcal{C} = \{\land, \lor, \div, 0\}$ .

Notice that for each C distinct from  $\{\div, 0\}$ ,  $BP_C$  is the class of algebras with language C axiomatised by those identities among (3.76)–(3.83), (3.87)–(3.94) that use only operation symbols from C; of course, this observation is dependent upon the axiomatisation of the variety of  $\leq_0$ -BCK paralattices given in Theorem 3.3.32.

**Remark 3.3.33.** As per pre-BCK quasilattices,  $BP_c$  should not be confused (for each C) with C-BP, the class of all C-subreducts of the variety of  $\leq_0$ -BCK paralattices. In particular, in contrast to each  $BP_c$ , we do not know in general if each C-BP is even a quasivariety.

Throughout the remainder of this subsection assume  $\Lambda \in C$  or  $\forall \in C$ . By remarks due to Laslo and Leech [145, Section 5, p. 23], the Green's equivalences  $\mathcal{D}_{\langle A; \Lambda \rangle}$  and  $\mathcal{D}_{\langle A; \vee \rangle}$  on a paralattice **A** are not typically congruences, whence paralattices do not in general possess a coherent Clifford-McLean structure. In fact, by Laslo and Leech [145, Theorem 26], a paralattice **A** supports a coherent Clifford-McLean structure iff it is simultaneously a quasilattice, in which case  $\mathcal{D}_{\langle A; \wedge \rangle} = \mathcal{D}_{\langle A; \vee \rangle}$ . Concerning the varieties  $\mathsf{BP}_{\mathcal{C}}$ , the following proposition shows that the presence of BCK difference causes members of  $\mathsf{BP}_{\mathcal{C}}$  to have a coherent Clifford-McLean structure only in the trivial (that is, commutative) case.

**Proposition 3.3.34.** For any  $A \in BP_c$  the following assertions hold:

- 1. If  $\wedge \in C$  then  $\mathcal{D}_{\langle A; \wedge \rangle}$  is a congruence on  $\langle A; \wedge, -, 0 \rangle$  iff  $\mathcal{D}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}}$ ;
- 2. If  $\lor \in \mathcal{C}$  then  $\mathcal{D}_{\langle A; \lor \rangle}$  is a congruence on  $\langle A; \lor, \div, 0 \rangle$  iff  $\mathcal{D}_{\langle A; \lor \rangle} = \omega_{\mathbf{A}}$ ;
- 3. If  $\{\wedge, \vee\} \subseteq C$  then  $\mathcal{D}_{\langle A; \wedge \rangle}$  and  $\mathcal{D}_{\langle A; \vee \rangle}$  are congruences on  $\mathbf{A}$  iff  $\mathcal{D}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}} = \mathcal{D}_{\langle A; \vee \rangle}$ .

*Proof.* We prove the proposition only for the case where  $\{\land,\lor\} \subseteq C$ ; the proofs in the remaining cases do not differ significantly and are omitted. Let  $\{\wedge,\vee\} \subseteq \mathcal{C} \text{ and let } \mathbf{A} \in \mathsf{BP}_{\mathcal{C}}. \text{ Suppose both } \mathcal{D}_{\langle A; \wedge \rangle} \text{ and } \mathcal{D}_{\langle A; \vee \rangle} \text{ are congruences}$ on A. Then in particular  $\mathcal{D}_{\langle A; \wedge \rangle}$  and  $\mathcal{D}_{\langle A; \vee \rangle}$  are congruences on the paralattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$ , so  $\mathcal{D}_{\langle A; \land \rangle} = \mathcal{D}_{\langle A; \lor}$  by previous remarks. In view of this last, to simplify notation we can and will write  $\mathcal{D}$  for  $\mathcal{D}_{\langle A; \wedge \rangle} = \mathcal{D}_{\langle A; \vee \rangle}$ throughout the remainder of the proof. Let  $a, b \in A$ . To complete the proof it is sufficient to show  $a \equiv b \pmod{\mathcal{D}}$  implies a = b. So suppose  $a \equiv b \pmod{\mathcal{D}}$ . From  $a \wedge b \wedge a \equiv a \wedge b \wedge a \pmod{\mathcal{D}}$  and  $a \equiv b \pmod{\mathcal{D}}$  we have that  $(a \land b \land a) \doteq a \equiv (a \land b \land a) \doteq b \pmod{\mathcal{D}}$ , since  $\mathcal{D}$  has the substitution property for the  $\div$  operation. Because  $(a \land b \land a) \div a = 0$  by (3.83) we have that  $(a \land b \land a) \doteq b \equiv 0 \pmod{\mathcal{D}}$ , whence  $(a \land b \land a) \doteq b = 0$  by Lemma 1.4.13. But then a - b = 0 since  $a \equiv b \pmod{D}$ . An analogous argument shows  $b \div a = 0$ ; since  $\langle A; \div, 0 \rangle$  is a BCK-algebra we have that a = b. Thus  $\mathcal{D} = \omega_A$ . Since the converse holds trivially, the proposition is proved.

Let  $A \in BP_{\mathcal{C}}$ . Because  $\mathcal{L}_{\langle A; \wedge \rangle}$ ,  $\mathcal{R}_{\langle A; \wedge \rangle} \subseteq \mathcal{D}_{\langle A; \wedge \rangle}$  and  $\mathcal{L}_{\langle A; \vee \rangle}$ ,  $\mathcal{R}_{\langle A; \vee \rangle} \subseteq \mathcal{D}_{\langle A; \vee \rangle}$ (when these equivalences exist), an easy modification of the proof of Proposition 3.3.34 yields:

**Proposition 3.3.35.** For any  $A \in BP_{\mathcal{C}}$  the following assertions hold:

- 1. If  $\wedge \in C$  then  $\mathcal{L}_{\langle A; \wedge \rangle}$  and  $\mathcal{R}_{\langle A; \wedge \rangle}$  are congruences on  $\langle A; \wedge, -, 0 \rangle$  iff  $\mathcal{L}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}} = \mathcal{R}_{\langle A; \wedge \rangle};$
- 2. If  $\forall \in C$  then  $\mathcal{L}_{\langle A; \forall \rangle}$  and  $\mathcal{R}_{\langle A; \forall \rangle}$  are congruences on  $\langle A; \forall, \div, 0 \rangle$  iff  $\mathcal{L}_{\langle A; \forall \rangle} = \omega_{\mathbf{A}} = \mathcal{R}_{\langle A; \forall \rangle};$
- 3. If  $\{\wedge, \vee\} \subseteq C$  then  $\mathcal{L}_{\langle A; \wedge \rangle}$ ,  $\mathcal{R}_{\langle A; \wedge \rangle}$ ,  $\mathcal{L}_{\langle A; \vee \rangle}$  and  $\mathcal{R}_{\langle A; \vee \rangle}$  are all congruences on A iff  $\mathcal{L}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}} = \mathcal{R}_{\langle A; \wedge \rangle}$  and  $\mathcal{L}_{\langle A; \vee \rangle} = \omega_{\mathbf{A}} = \mathcal{R}_{\langle A; \vee \rangle}$ .

Let  $\mathbf{A} \in \mathsf{BP}_{\mathcal{C}}$ . For  $\wedge \in \mathcal{C}$   $[\vee \in \mathcal{C}; \{\wedge, \vee\} \subseteq \mathcal{C}] \langle A; \wedge, \div, 0 \rangle$   $[\langle A; \vee, \div, 0 \rangle; \langle A; \wedge, \vee, \circ, 0 \rangle]$  is regular if  $\mathcal{L}_{\langle A; \wedge \rangle}$  and  $\mathcal{R}_{\langle A; \wedge \rangle}$   $[\mathcal{L}_{\langle A; \vee \rangle}, \mathcal{R}_{\langle A; \vee \rangle}; \mathcal{L}_{\langle A; \wedge \rangle}, \mathcal{R}_{\langle A; \wedge \rangle}, \mathcal{L}_{\langle A; \wedge \rangle}, \mathcal{L}_{\langle A; \vee \rangle}, \mathcal{R}_{\langle A; \vee \rangle}]$  are congruences on  $\langle A; \wedge, \div, 0 \rangle$   $[\langle A; \vee, \div, 0 \rangle; \langle A; \wedge, \vee, \circ, \circ \rangle]$ . Because of Proposition 3.3.35, an algebra  $\mathbf{A} \in \mathsf{BP}_{\mathcal{C}}$  is regular only in the trivial (commutative) case.

**Corollary 3.3.36.** For  $A \in BP_{\mathcal{C}}$  the following assertions hold:

- 1. If  $\wedge \in C$  then  $\langle A; \wedge, -, 0 \rangle$  is regular iff  $\mathcal{D}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}}$ ;
- 2. If  $\forall \in C$  then  $\langle A; \forall, \dot{-}, 0 \rangle$  is regular iff  $\mathcal{D}_{\langle A; \forall \rangle} = \omega_{\mathbf{A}}$ ;
- 3. If  $\{\wedge, \vee\} \subseteq \mathcal{C}$  then **A** is regular iff  $\mathcal{D}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}} = \mathcal{D}_{\langle A; \vee \rangle}$ .

Proof. We prove the corollary only for the case that  $\{\wedge, \vee\} \subseteq C$ ; the proofs in the other cases are not significantly different and are omitted. So let  $\mathbf{A} \in$  $\mathsf{BP}_C$  where  $\{\wedge, \vee\} \subseteq C$ . Suppose  $\mathbf{A}$  is regular. Then  $\mathcal{L}_{\langle A; \wedge \rangle}$  and  $\mathcal{R}_{\langle A; \wedge \rangle}$  are congruences on  $\mathbf{A}$ , so  $\mathcal{L}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}} = \mathcal{R}_{\langle A; \wedge \rangle}$  by Proposition 3.3.35(3). Since  $\mathcal{D}_{\langle A; \wedge \rangle} = \mathcal{L}_{\langle A; \wedge \rangle} \circ \mathcal{R}_{\langle A; \wedge \rangle}$  (by Howie [111, p. 46]) we have that  $\mathcal{D}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}}$ ; an analogous argument shows  $\mathcal{D}_{\langle A; \vee \rangle} = \omega_{\mathbf{A}}$ . For the converse, suppose  $\mathcal{D}_{\langle A; \wedge \rangle} =$  $\omega_{\mathbf{A}} = \mathcal{D}_{\langle A; \vee \rangle}$ . Since both  $\mathcal{L}_{\langle A; \wedge \rangle} \subseteq \mathcal{D}_{\langle A; \wedge \rangle}$  and  $\mathcal{R}_{\langle A; \wedge \rangle} \subseteq \mathcal{D}_{\langle A; \wedge \rangle}$  we must have both  $\mathcal{L}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}}$  and  $\mathcal{R}_{\langle A; \wedge \rangle} = \omega_{\mathbf{A}}$ . Similar reasoning shows  $\mathcal{L}_{\langle A; \vee \rangle} = \omega_{\mathbf{A}}$  and  $\mathcal{R}_{\langle A; \vee \rangle} = \omega_{\mathbf{A}}$ . Thus all of  $\mathcal{L}_{\langle A; \wedge \rangle}$ ,  $\mathcal{R}_{\langle A; \wedge \rangle}$ ,  $\mathcal{L}_{\langle A; \vee \rangle}$  and  $\mathcal{R}_{\langle A; \vee \rangle}$  are congruences on  $\mathbf{A}$ , so  $\mathbf{A}$  is regular.

Recall from the remarks concluding §3.3.2 that the behaviour of pre-BCK bands and pre-BCK quasilattices more closely resembles that of bands and quasilattices than that of BCK-semilattices and BCK-lattices. In contrast,

Proposition 3.3.34 and Corollary 3.3.36 suggest that the behaviour of  $\leq_{0}$ -BCK bands and  $\leq_{0}$ -BCK paralattices is more closely aligned to that of BCK-semilattices and BCK-lattices than to that of bands and paralattices. Further support for this last contention is provided in the remaining results of this subsection, which show that properties of 0-regularity, congruence distributivity and congruence permutability enjoyed by BCK-semilattices and BCK-lattices are preserved upon passing to  $\leq_{0}$ -BCK bands and  $\leq_{0}$ -BCK paralattices.

**Proposition 3.3.37.** (cf. [116, Theorem 1]) Let  $A \in BP_c$ . For any  $a, b \in A$ , the following are equivalent:

1. a = b;

2.  $a \div b = 0 = b \div a$ .

If  $\forall \in C$ , then either of (1) or (2) is equivalent to:

3.  $(a \div b) \lor (b \div a) = 0.$ 

Thus any variety  $BP_c$  is 0-regular and hence ideal determined. That is, for any  $A \in BP_c$ , the map  $\theta \mapsto [0]_{\theta}$  ( $\theta \in Con A$ ) is a lattice isomorphism from Con A into I(A).

*Proof.* For the first assertion of the proposition, the only non-trivial implication to prove is  $(3) \Rightarrow (2)$ . So let  $\forall \in C$  and let  $\mathbf{A} \in \mathsf{BP}_C$ . Let  $a, b \in A$  and suppose  $(a \div b) \lor (b \div a) = 0$ . We have  $a \div b = (a \div b) \lor 0 = (a \div b) \lor$  $(a \div b) \lor (b \div a) = (a \div b) \lor (b \div a) = 0$ , just because the reduct  $\langle A; \lor, 0 \rangle$ is a band with identity. Similarly,  $b \div a = 0 \lor (b \div a) = (a \div b) \lor (b \div a) \lor$  $(b \div a) = (a \div b) \lor (b \div a) = 0$ . Thus  $a \div b = 0 = b \div a$  as required.

From the first assertion of the proposition and Proposition 1.2.6 it follows that any  $BP_c$  is 0-regular and hence (by Proposition 1.7.3) ideal determined. Thus for any  $A \in BP_c$  the map  $\theta \mapsto [0]_{\theta}$  ( $\theta \in Con A$ ) is a lattice isomorphism from Con A into I(A).

**Remark 3.3.38.** Because any  $BP_c$  is ideal determined, it has a finite basis of ideal terms (in the sense of [57]): see Chajda and Halas [57] or Ursini [221]. Nonetheless, in general we know of no simple description of the  $BP_c$ -ideals

for any variety  $BP_{\mathcal{C}}$ . In particular, for each choice of  $\mathcal{C}$ , there exists  $\mathbf{A} \in BP_{\mathcal{C}}$  such that  $I_{BP_{\mathcal{C}}}(\mathbf{A}) \not\subseteq I_{BCK}(\langle A; \div, 0 \rangle)$  (by an easy modification of a result due to Idziak [116, Lemma 2(ii)], for each choice of  $\mathcal{C}$  the converse  $I_{BCK}(\langle A; \div, 0 \rangle) \subseteq I_{BP_{\mathcal{C}}}(\mathbf{A})$  does obtain for any  $\mathbf{A} \in BP_{\mathcal{C}}$ ). Indeed, examples due to Idziak [116, p. 979] show there exists  $\mathbf{A} \in BP_{\mathcal{C}}$  such that the inclusion  $I_{BP_{\mathcal{C}}}(\mathbf{A}) \subseteq I_{BCK}(\langle A; \div, 0 \rangle)$  is strict when  $\forall \in \mathcal{C}$ . To see the inclusion  $I_{BP_{\mathcal{C}}}(\mathbf{A}) \subseteq I_{BCK}(\langle A; \div, 0 \rangle)$  is strict for  $\mathcal{C} = \{\wedge, \div, 0\}$ , consider the following 4-element lower  $\leq_0$ -BCK band  $\mathbf{A}$ :

| $\wedge^{\mathbf{A}}$ | 0 | a | b | с | <u>.</u> A | 0 | a | b | c |
|-----------------------|---|---|---|---|------------|---|---|---|---|
| 0                     | 0 | 0 | 0 | 0 | 0          | 0 | 0 | 0 | 0 |
| a                     | 0 | a | a | a | a          | a | 0 | 0 | 0 |
| b                     | θ | a | b | b | b          | b | a | 0 | a |
| c                     | 0 | a | с | с | с          | c | С | С | 0 |

An easy sequence of checks shows that **A** is simple but that  $\{0, a, b\}$  is a BCK-ideal of the BCK-algebra reduct  $\langle A; \div, 0 \rangle$ . Hence  $I_{BP_c}(A) \not\subseteq I_{BCK}(\langle A; \div, 0 \rangle)$ .

## **Theorem 3.3.39.** For each C, the variety $BP_C$ is congruence distributive.

*Proof.* Because  $BP_C \models (1.3), (1.4), (1.7)$ , from the remarks of §1.7.4 we have that  $BP_C$  is ideal distributive. Since  $BP_C$  is ideal determined (by Proposition 3.3.37) we conclude that  $BP_C$  is congruence distributive.

**Remark 3.3.40.** It is easily verified that if  $\forall \in C$  then the term  $M(x, y, z) := (x \div (x \div z)) \lor (z \div (z \div y)) \lor (y \div (y \div x))$  is a majority term for  $\mathsf{BP}_C$  (see also Idziak [116, p. 841, Theorem 2]). In contrast, the varieties  $\mathsf{BP}_C$  have no majority term when  $\forall \notin C$ : see Idziak [116, Remark, p. 842].

**Proposition 3.3.41.** Concerning the varieties  $BP_c$ , the following assertions hold:

1. (cf. [187, Theorem]) If  $\land \in C$  then BP<sub>C</sub> is congruence 4-permutable;

2. (cf. [116, p. 841, Theorem 2]) If  $\forall \in C$  then  $BP_C$  is congruence permutable, with Mal'cev term:

$$p(x, y, z) := (x - (y - z)) \vee (z - (y - x)).$$

*Proof.* For (1), suppose  $\land \in \mathcal{C}$ . Let:

$$p_1(x, y, z) := x \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} z)$$

$$p_2(x, y, z) := (x \stackrel{\cdot}{-} (x \stackrel{\cdot}{-} y)) \land (z \stackrel{\cdot}{-} (z \stackrel{\cdot}{-} y))$$

$$p_3(x, y, z) := z \stackrel{\cdot}{-} (y \stackrel{\cdot}{-} x).$$

By (1.7), (3.81), (3.82), (3.76) and (1.3), the variety  $BP_{\wedge C}$  satisfies the identities:

$$p_1(x, y, y) \approx x$$
  
 $p_i(x, x, y) \approx p_{i+1}(x, y, y),$   $i = 1, 2$   
 $p_3(x, x, y) \approx y$ 

and so is congruence 4-permutable by Hagemann and Mitschke [106, Theorem 2].

For (2), suppose  $\forall \in C$  and let p(x, y, z) be as in the statement of the proposition. Because of (3.92) and (3.93), it is easily verified that  $BP_C$  satisfies the identities  $p(x, x, y) \approx y$  and  $p(x, y, y) \approx x$ . Hence p(x, y, z) is a Mal'cev term for V and  $BP_C$  is congruence permutable.

**Remark 3.3.42.** The statement of Proposition 3.3.41(1) cannot be strengthened to the assertion that if  $\land \in C$  then  $BP_C$  is congruence 3-permutable, in view of a result of Raftery [187, Theorem] showing that the variety of lower BCK-semilattices is not congruence 3-permutable.

**3.3.43.** Implicative BCK Bands and Implicative BCK Paralattices. By the results of §1.4.32, §1.4.37, §3.2.6 and §3.2.22, algebras arising in binary discriminator, pointed fixedpoint discriminator and pointed ternary discriminator varieties all support an underlying 'locally Boolean' structure (in the sense of either §1.3.15 or §1.4.24). As in §3.3.13, this observation motivates the study of those members of BP<sub>c</sub> for which every (appropriately defined) principal subalgebra is a Boolean lattice. To begin, recall the definition of the BCK meet  $\cap$  from §1.6.11. For a BCK-algebra A and any  $a \in A$ , let  $(a] := \langle a \cap A; \cap^{A|_{a\cap A}} \rangle$ . By analogy with semigroup theory we call (a] the principal subalgebra generated by a, even though the polynomial reduct  $\langle A; \cap \rangle$ is not a semigroup in general.

Remark 3.3.44. For a BCK-algebra A, the principal subalgebra (a] generated by  $a \in A$  should not be confused with the BCK-subalgebra  $\langle (a]_0; -A|_{(a]}, 0 \rangle$ , where  $(a]_0 := \{b : b \leq_0 a\}$ , and which is also denoted (a]. In the sequel, it will always be clear from context whether (a] denotes the BCK-subalgebra  $\langle (a]_0; -A|_{(a)}, 0 \rangle$  or the principal subalgebra  $\langle a \cap A; \cap^{A|_{a \cap A}} \rangle$  generated by a.

For a BCK-algebra **A**, the principal subalgebra (a] generated by  $a \in A$  may be alternatively defined as  $\langle (a]_1; \cap^{\mathbf{A}|_{a\cap A}} \rangle$ , where  $(a]_1 := \{b : b \leq_1 a\}$ , because  $\{b : b \leq_1 a\} = a \sqcap A$  by Proposition 1.6.4(1). From this observation it follows that if **A** is commutative then  $(a] = \langle (a]_0; \cap^{\mathbf{A}|_{(a)}} \rangle$ , where  $(a]_0 :=$  $\{b : b \leq_0 a\}$ , since in this case  $\leq_0 = \leq_1$  by Lemma 1.6.12. In other words, if **A** is commutative, then (a] is precisely the principal subalgebra generated by a of the semilattice polynomial reduct  $\langle A; \cap \rangle$ . It is this observation that motivates our description of the algebra  $\langle a \cap A; \cap^{\mathbf{A}|_{a\cap A}} \rangle$  above as the principal subalgebra generated by a.

As per §3.3.13, our study of varieties  $BP_C$  supporting a 'locally Boolean' structure centres on the interplay of the underlying band with zero principal subalgebra structure and the underlying BCK-algebra principal subalgebra structure. In more detail, let  $\Lambda \in C$  and let  $\mathbf{A} \in BP_C$ . Then  $\mathbf{A}$  has both a band with zero reduct  $\langle A; \Lambda, 0 \rangle$  and a BCK-algebra reduct  $\langle A; \Pi, 0 \rangle$ . Hence every  $a \in A$ generates both a principal subalgebra  $(\mathbf{a}]_{\langle A; \Lambda, 0 \rangle}$  of the band with zero reduct  $\langle A; \Lambda, 0 \rangle$  (recall Lemma 1.3.13) and a principal subalgebra  $(\mathbf{a}]_{\langle A; \pm, 0 \rangle}$  of the BCK-algebra reduct  $\langle A; \Pi, 0 \rangle$ . Although the principal subalgebra  $(\mathbf{a}]_{\langle A; \pm, 0 \rangle}$ generated by a of the BCK-algebra reduct  $\langle A; -, 0 \rangle$  is not a principal subalgebra in the usual semigroup theoretic sense, the following two results show nonetheless that for the varieties  $BP_C$ ,  $\Lambda \in C$ , the behaviour of the under-

lying BCK-algebra principal subalgebra structure is decisive (cf. Proposition 3.3.18).

**Proposition 3.3.45.** Let  $\wedge \in C$  and let  $\mathbf{A} \in BP_{C}$ . For every  $a \in A$ , the principal subalgebras  $(\mathbf{a}]_{\langle A; \wedge, 0 \rangle}$  and  $(\mathbf{a}]_{\langle A; -, 0 \rangle}$  coincide iff the reduct  $\langle A; \wedge, 0 \rangle$  is a normal band with zero and the reduct  $\langle A; -, 0 \rangle$  is a commutative BCK-algebra.

*Proof.* Let  $\wedge \in C$  and let  $\mathbf{A} \in \mathsf{BP}_{\mathcal{C}}$ . Throughout the proof, we denote by  $(a]_{\langle A; \wedge, 0 \rangle}$  and  $(a]_{\langle A; \pm, 0 \rangle}$  the respective universes of the principal subalgebras  $(a]_{\langle A; \wedge, 0 \rangle}$  and  $(a]_{\langle A; \pm, 0 \rangle}$ . To simplify notation, we may also write simply  $(a]_{\mathbf{A}}$  for both  $(a]_{\langle A; \wedge, 0 \rangle}$  and  $(a]_{\langle A; \pm, 0 \rangle}$  when these sets coincide.

(⇒) Suppose that for every  $a \in A$ , the principal subalgebras  $(a]_{\langle A; \land, 0 \rangle}$  and  $(a]_{\langle A; -, 0 \rangle}$  coincide. To see the reduct  $\langle A; -, 0 \rangle$  is a commutative BCK-algebra, it is sufficient by Lemma 1.6.12 to show the partial orders  $\leq_0^{\langle A; -, 0 \rangle}$  and  $\leq_1^{\langle A; -, 0 \rangle}$  coincide. For this, observe that for any  $a, b \in A$ ,

$$\begin{split} b \leq_{1}^{\langle A; \div, 0 \rangle} a & \text{iff} \quad b \in a \cap A & \text{by Proposition 1.6.4(1)} \\ & \text{iff} \quad b \in (a]_{\langle A; \div, 0 \rangle} & \\ & \text{iff} \quad b \in (a]_{\langle A; \wedge, 0 \rangle} & \text{by hypothesis} \\ & \text{iff} \quad b \leq_{\mathcal{H}} a^{-} & \\ & \text{iff} \quad b \leq_{\mathcal{H}} a^{-} & \text{by Lemma 1.3.13(1)} \\ & \text{iff} \quad b \leq_{0}^{\langle A; \div, 0 \rangle} a & \text{since } A \in \mathsf{BP}_{\mathcal{C}}. \end{split}$$

Hence  $\leq_0^{\langle A; \pm, 0 \rangle}$  and  $\leq_1^{\langle A; \pm, 0 \rangle}$  coincide, and  $\langle A; \pm, 0 \rangle$  is commutative.

It remains to show the band with zero reduct  $\langle A; \land, 0 \rangle$  is normal. To this end, recall from §1.6.11 that since the BCK-algebra reduct  $\langle A; \div, 0 \rangle$  is commutative, it has a distributive nearlattice polynomial reduct  $\langle A; \cap \rangle$ . For any  $a \in A$ , let  $(a]_{\langle A; \cap \rangle}$  denote the principal subalgebra of  $\langle A; \cap \rangle$  generated by a. Because  $\langle A; \cap \rangle$  is a distributive nearlattice, every principle subalgebra  $(a]_{\langle A; \cap \rangle}$ is a distributive sublattice by the remarks of §1.6.11. Because  $(a]_{\langle A; \cap \rangle}$  coincides with  $(a]_{\langle A; \div, 0 \rangle}$  (by commutativity of  $\langle A; \div, 0 \rangle$  and Remark 3.3.44), and  $(a]_{\langle A; \div, 0 \rangle}$  coincides with  $(a]_{\langle A; \land, 0 \rangle}$  (by hypothesis), we have that  $(a]_{\langle A; \land, 0 \rangle}$  is a distributive sublattice. Hence for every  $a \in A$ , the principal subalgebra

 $(a]_{\langle A; \land, 0 \rangle}$  generated by a is a sublattice. By Lemma 1.3.16 the band with zero reduct  $\langle A; \land, 0 \rangle$  is normal, and the proof is complete.

( $\Leftarrow$ ) Suppose the reducts  $\langle A; \land, 0 \rangle$  and  $\langle A; \div, 0 \rangle$  are respectively a normal band with zero and a commutative BCK-algebra. To see the principal subalgebras  $(a]_{\langle A; \land, 0 \rangle}$  and  $(a]_{\langle A; \div, 0 \rangle}$  coincide for every  $a \in A$  it is sufficient to show:

(i) For any  $a \in A$ ,  $(a]_{\langle A; \land, 0 \rangle} = (a]_{\langle A; \div, 0 \rangle}$ ;

(ii) If  $b, c \in (a]_A$  then  $b \wedge c = b \cap c$ .

For (i), just observe that for any  $a \in A$ ,  $(a]_{\langle A; \land, 0 \rangle} = \{b : b \leq_{\mathcal{H}} a\} = \{b : b \leq_{\mathcal{H}} a\} = \{b : b \leq_{0}^{\langle A; \div, 0 \rangle} a\} = \{b : b \leq_{1}^{\langle A; \div, 0 \rangle} a\} = (a]_{\langle A; \div, 0 \rangle}$  by commutativity of  $\langle A; \div, 0 \rangle$  and the coincidence of the partial orders  $\leq_{\mathcal{H}}$  and  $\leq_{0}^{\langle A; \backslash, 0 \rangle}$ .

For (ii), by (i) we have that  $(a]_{\langle A; \land, 0 \rangle}$  and  $(a]_{\langle A; \div, 0 \rangle}$  coincide, so the reference to  $(a]_{\mathbf{A}}$  makes sense. Let  $b, c \in (a]_{\mathbf{A}}$ . Since  $\langle A; \land, 0 \rangle$  is normal,  $b \land c$  is the greatest lower bound of  $\{b, c\}$  with respect to the restriction of  $\leq_{\mathcal{H}}$  to  $(a]_{\langle A; \land, 0 \rangle}$ . Also,  $b \cap c$  is the greatest lower bound of  $\{b, c\}$  with respect to the restriction of  $\leq_{0}^{\langle A; \div, 0 \rangle}$  to  $(a]_{\langle A; \div, 0 \rangle}$ , because  $\langle A; \div, 0 \rangle$  is commutative. Since  $(a]_{\langle A; \land, 0 \rangle} = (a]_{\langle A; \div, 0 \rangle}$  (by (i)) and  $\leq_{\mathcal{H}} = \leq_{0}^{\langle A; \div, 0 \rangle}$  (by our assumptions on  $\mathbf{A}$ ), we have that  $b \land c = b \cap c$  as desired.

**Remark 3.3.46.** Let A be a BCK-algebra and let  $a \in A$ . Clearly, the definition of the principal subalgebra (a] generated by a plays a crucial role in the preceding proof. In particular, if (a] is instead defined as the algebra  $\langle A \cap a; \cap^{\mathbf{A}|_{A\cap a}} \rangle$ , then the argument of the proof is not sufficient to establish Proposition 3.3.45, since in this case  $A \cap a = \{b : b \leq_0 a\}$ .

Let  $A \in BP_{\mathcal{C}}$ ,  $\wedge \in \mathcal{C}$ , be such that the reduct  $\langle A; \wedge, 0 \rangle$  is a normal band with zero and the reduct  $\langle A; \div, 0 \rangle$  is a commutative BCK-algebra. In view of Proposition 3.3.45, we can unambiguously write  $(a]_A$  for the principal subalgebra generated by a; we adopt this practice henceforth.

**Proposition 3.3.47.** Let  $\wedge \in C$  and let  $\mathbf{A} \in \mathsf{BP}_C$  be such that for every  $a \in A$ , the principal subalgebras  $(\mathbf{a}]_{\langle A; \wedge, 0 \rangle}$  and  $(\mathbf{a}]_{\langle A; \div, 0 \rangle}$  coincide. Then for every  $a \in A$ , the principal subalgebra  $(\mathbf{a}]_{\mathbf{A}}$  is a Boolean sublattice iff the reduct  $\langle A; \div, 0 \rangle$  is an implicative BCK-algebra.

326

*Proof.* Let  $\wedge \in C$  and let  $\mathbf{A} \in \mathsf{BP}_C$  be such that for every  $a \in A$ , the principal subalgebras  $(a]_{\langle A; \wedge, 0 \rangle}$  and  $\langle a]_{\langle A; -, 0 \rangle}$  coincide.

(⇒) Suppose that, for every  $a \in A$ , the principal subalgebra  $(a]_A$  is a Boolean sublattice. By our assumptions on **A** and Proposition 3.3.45, the band with zero reduct  $\langle A; \land, 0 \rangle$  is normal and the BCK-algebra reduct  $\langle A; \neg, 0 \rangle$  is commutative. Since  $\langle A; \neg, 0 \rangle$  is commutative, **A** possesses a meet semilattice polynomial reduct  $\langle A; \cap \rangle$ . For any  $a \in A$ , let  $(a]_{\langle A; \cap \rangle}$  denote the principal subalgebra of  $\langle A; \cap \rangle$  generated by a. Because  $b \leq^{\langle A; \cap \rangle} c$  iff  $b \leq^{\langle A; +, 0 \rangle} c$  for any  $b, c \in A$  and the band with zero reduct  $\langle A; \land, 0 \rangle$  is normal, an argument similar to the proof of Proposition 3.3.45 shows that for every  $a \in A$ ,  $(a]_{\langle A; \land, 0 \rangle}$  and  $(a]_{\langle A; \cap \rangle}$  coincide. Hence the principal subalgebra  $(a]_{\langle A; \cap \rangle}$  is a Boolean lattice for every  $a \in A$ , and so  $\langle A; \cap \rangle$  is semi-Boolean. From Theorem 1.6.21 it follows that  $\langle A; \cap \rangle$  has an induced implicative BCK difference operation /, where  $b/c := (b \cap c)'_{\langle b]_{\langle A; \cap \rangle}$  for any  $b, c \in A$ . But for any  $b, c \in A$ ,

$$b/c = (b \cap c)'_{(b]_{(A; \cap)}}$$
  
=  $(b - (b - c))'_{(b]_{(A; + 0)}}$   
=  $b - (b - (b - c))$   
=  $b - c$  by (1.31),

so  $b/c = b \div c$ . Hence the reduct  $\langle A; \div, 0 \rangle$  is an implicative BCK-algebra.

( $\Leftarrow$ ) Suppose the reduct  $\langle A; \div, 0 \rangle$  is an implicative BCK-algebra. Then for every  $a \in A$ , the principal subalgebra  $(a]_A$  is a Boolean lattice by our assumptions on A and Corollary 1.6.22.

Unless otherwise specified, throughout the remainder of this section we assume  $\wedge \in C''$ . Given this convention, Proposition 3.3.47 leads to the study in the sequel of those members of BP<sub>c</sub> for which the band with zero reduct  $\langle A; \wedge, 0 \rangle$  is normal and the BCK-algebra reduct  $\langle A; \div, 0 \rangle$  is implicative.

A lower implicative  $\leq_0$ -BCK normal band is an algebra  $\langle A; \land, /, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \land, 0 \rangle$  is a normal band with zero; (ii)

327

the reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra; and (iii) the natural band partial order  $\leq_{\mathcal{H}}$  coincides with the BCK partial order  $\leq_{0}^{\langle A; /, 0 \rangle}$ .

**Theorem 3.3.48.** An algebra  $\mathbf{A} := \langle A; \land, /, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is a lower implicative  $\leq_0$ -BCK normal band iff the reduct  $\langle A; \land, 0 \rangle$  is a normal band with zero, the reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra, and  $\mathbf{A} \models (3.83), (3.84),$ (3.85). Thus the class of lower implicative  $\leq_0$ -BCK normal bands is a variety.

*Proof.* Let  $A := \langle A; \land, /, 0 \rangle$  be an algebra of type  $\langle 2, 2, 0 \rangle$  such that the reduct  $\langle A; \land, 0 \rangle$  is a normal band with zero and the reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra. Let  $a, b \in A$ . To prove the theorem it is sufficient to show:

(i)  $(a \wedge b \wedge a) \doteq a = 0$  iff  $a \leq_{\mathcal{H}} b$  implies  $a \leq_{0}^{\langle A; /, 0 \rangle} b$ ; (ii)  $(a \doteq b) \wedge a = a \doteq b = a \wedge (a \doteq b)$  iff  $a \leq_{0}^{\langle A; /, 0 \rangle} b$  implies  $a \leq_{\mathcal{H}} b$ .

The proof of (i) is implicit in the proof of Theorem 3.3.28. For (ii), the remarks immediately following Theorem 3.3.28 imply  $(a \div b) \land a = a \div b = a \land (a \div b)$  iff  $a \leq_1^{\langle A; /, 0 \rangle} b$  implies  $a \leq_{\mathcal{H}} b$ . Since the BCK partial orders  $\leq_0^{\langle A; /, 0 \rangle}$  and  $\leq_1^{\langle A; /, 0 \rangle}$  coincide (by the proof of Proposition 3.3.45), the result follows.

By analogy with the theory of skew lattices, call a paralattice  $\langle A; \land, \lor \rangle$  local if its band reduct  $\langle A; \land \rangle$  is normal. An *implicative*  $\leq_0$ -BCK local paralattice is an algebra  $\langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \land, \lor, 0 \rangle$ is a local paralattice with zero; (ii) the reduct  $\langle A; /, 0 \rangle$  is an implicative BCKalgebra; and (iii) the natural band partial order  $\leq_{\mathcal{H}}$  coincides with the BCK partial order  $\leq_0^{\langle A; /, 0 \rangle}$ . The proof of the following result may be established by an argument similar to the proof of Theorem 3.3.48 and hence is omitted.

**Theorem 3.3.49.** An algebra  $\mathbf{A} := \langle A; \land, /, 0 \rangle$  of  $typ^{\varrho} \langle 2, 2, 0 \rangle$  is an implicative  $\leq_0$ -BCK local paralattice iff the reduct  $\langle A; \land, \lor, 0 \rangle$  is a local paralattice with zero, the reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra, and  $\mathbf{A} \models (3.83), (3.95), (3.96)$ . Thus the class of implicative  $\leq_0$ -BCK local paralattices is a variety.

For the sake of notational consistency with the prequel, throughout the remainder of this subsection let C'' denote an arbitrary subset of the language

 $\{\wedge, \vee, /, 0\}$  of implicative  $\leq_0$ -BCK local paralattices that contains either / and 0 or all of  $\wedge$ , / and 0. Ignoring issues of type, let:

- $BP_{\mathcal{C}''}$  denote the quasivariety of BCK-algebras when  $\mathcal{C}'' = \{/, 0\}$ ;
- $BP_{\mathcal{C}''}$  denote the variety of lower  $\leq_0$ -BCK bands when  $\mathcal{C}'' = \{\wedge, /, 0\}$ ;
- $BP_{\mathcal{C}'}$  denote the variety of  $\leq_0$ -BCK paralattices when  $\mathcal{C}'' = \{\land, \lor, /, 0\}$ .

Also, let:

- $P_{\mathcal{C}''}$  denote the variety of implicative BCK-algebras when  $\mathcal{C}'' = \{/, 0\}$ ;
- $IP_{C''}$  denote the variety of normal lower implicative  $\leq_0$ -BCK bands when  $C'' = \{\wedge, /, 0\};$
- $P_{\mathcal{C}''}$  denote the variety of local implicative  $\leq_0$ -BCK paralattices when  $\mathcal{C}'' = \{\wedge, \lor, /, 0\}.$

Given the above notation, clearly  $\mathbb{IP}_{\mathcal{C}''} \subseteq \mathbb{BP}_{\mathcal{C}''}$  for any fixed choice of  $\mathcal{C}''$ .

Remark 3.3.50. An upper implicative  $\leq_0$ -BCK band is an algebra  $\langle A; \vee, /, 0 \rangle$ of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \vee, 0 \rangle$  is a band with identity; (ii) the reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra; and (iii) the natural band partial order  $\leq_{\mathcal{H}}$  dualises the BCK partial order  $\leq_0^{\langle A; /, 0 \rangle}$  in the sense that  $a \leq_{\mathcal{H}} b$  iff  $b \leq_0^{\langle A; /, 0 \rangle} a$  for any  $a, b \in A$ . Clearly the class of upper implicative  $\leq_0$ -BCK bands is a variety.

By normality, it is clear that no non-trivial (that is, non-commutative) normal subvariety of the variety of upper implicative  $\leq_0$ -BCK bands exists. Hence there exists no non-trivial variety of upper implicative  $\leq_0$ -BCK normal bands that stands in relation to the variety of upper implicative BCS bands as the varieties  $|P_{C'}, \{\wedge, \backslash, 0\} \subseteq C'$ , stand in relation to the varieties  $|P_{C''}, \{\wedge, \backslash, 0\} \subseteq C'$ . For consistency and the sake of parity with the theory of the varieties  $PQ_{C'}, IQ_{C'}$ , we let:

- $BP_{\mathcal{C}''}$  denote the variety of upper  $\leq_0$ -BCK bands when  $\mathcal{C}'' = \{\vee, /, 0\}$ ;
- $IP_{C''}$  denote the variety of upper implicative  $\leq_0$ -BCK bands when  $C'' = \{\vee, /, 0\},\$

even though our interest in this thesis lies exclusively in the varieties  $|P_{C''}$ ,  $\wedge \in C''$ . That the preceding definitions are coherent in the context of a unified theory of the varieties  $BP_{C''}$ ,  $\{/, 0\} \subseteq C'' \subseteq \{\wedge, \lor, /, 0\}$ , follows from Theorem 3.3.53 and the remarks of §4.2.27 in the sequel.

**Proposition 3.3.51.** Let  $A \in IP_{C''}$ . Then the polynomial reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra and the polynomial reduct  $\langle A; \land, \backslash, 0 \rangle$  is a lower implicative BCS band, where in both cases  $a \backslash b := a/(a \land b \land a)$  for any  $a, b \in A$ . Consequently, if  $\lor \in C''$  then the polynomial reduct  $\langle A; \land, \lor, \backslash, 0 \rangle$ is an implicative BCS quasilattice. In particular, if  $\lor \in C''$  and the paralattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero, then the polynomial reduct  $\langle A; \land, \lor, \rangle, 0 \rangle$  is an implicative BCS skew lattice.

*Proof.* Let  $A \in IP_{C''}$ . For the first assertion, by Theorem 1.6.21, Proposition 3.3.47 and the definition of  $\setminus$  we have that  $a \setminus b = (a \cap (a \land b \land a))'_{(a|_A)}$  for any  $a, b \in A$ . But,

$$a \cap (a \wedge b \wedge a) = (a \wedge b \wedge a) \cap a \qquad \text{by commutativity of } \cap \\ = (a \wedge b \wedge a)/((a \wedge b \wedge a)/a) \\ = (a \wedge b \wedge a)/0 \qquad \text{by (3.83)} \\ = a \wedge b \wedge a, \qquad (3.98)$$

so  $a \setminus b = (a \land b \land a)'_{(a|_A)}$ . From Remark 2.3.37 it follows that the polynomial reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra. To prove the remaining assertions of the proposition it is sufficient to show the induced implicative BCS-algebra partial order  $\leq^{\langle A; \backslash, 0 \rangle}$  and quasiorder  $\preceq^{\langle A; \backslash, 0 \rangle}$  coincide with the natural band partial order  $\leq_{\mathcal{H}}$  and quasiorder  $\preceq_{\mathcal{D}}$  respectively. Because of Proposition 3.3.14, this reduces to establishing that  $a \setminus (a \setminus b) = a \land b \land a$  for any  $a, b \in A$ . So let  $a, b \in A$ . We have:

$$a \setminus (a \setminus b) = a / (a \wedge (a / (a \wedge b \wedge a)) \wedge a)$$
  
=  $a / (a / (a \wedge b \wedge a))$  by (3.84), (3.85)  
=  $a \cap (a \wedge b \wedge a)$   
=  $a \wedge b \wedge a$  by (3.98).

**Proposition 3.3.52.** Let  $\forall \in C''$  and let  $\mathbf{A} \in \mathsf{IP}_{C''}$ . If the paralattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero, then the following assertions hold:

- 1. The skew lattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is distributive local;
- 2. The polynomial reduct  $\langle A; \land, \lor, \cap \rangle$  is a skew lattice with intersections, where  $a \cap b := a/(a/b)$  for any  $a, b \in A$ .

*Proof.* Let  $\lor \in \mathcal{C}''$  and let  $\mathbf{A} \in \mathsf{IP}_{\mathcal{C}''}$  be such that the paralattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero.

For (1), we have that  $\langle A; \land, \lor, 0 \rangle$  is a local skew lattice by hypothesis. Since for each  $a \in A$ , the sublattice  $(a]_A$  is distributive (by the proof of Proposition 3.3.45), from Proposition 1.4.22 we deduce that  $\langle A; \land, \lor, 0 \rangle$  is distributive.

For (2), the reduct  $\langle A; \land, \lor \rangle$  is a skew lattice, while the reduct  $\langle A; \cap \rangle$  is a meet semilattice. Since  $a \cap b = a$  iff a/b = 0 for any  $a, b \in A$ , the underlying partial order  $\leq^{A; \cap}$  on the semilattice polynomial reduct  $\langle A; \cap \rangle$  coincides with the underlying BCK partial order  $\leq^{\langle A; \div, 0 \rangle}_{0}$  on  $\langle A; \div, 0 \rangle$ . Since  $\leq^{\langle A; \div 0 \rangle}_{0}$  and  $\leq_{\mathcal{H}}$  coincide by hypothesis,  $\leq^{\langle A; \cap \rangle}$  and  $\leq_{\mathcal{H}}$  must coincide also. Therefore the polynomial reduct  $\langle A; \land, \lor, \cap \rangle$  is a skew lattice with intersections.

An implicative  $\leq_0$ -BCK local skew lattice is an algebra  $\langle A; \land, \lor, \backslash, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; \land, \lor, 0 \rangle$  is a local skew lattice with zero; (ii) the reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCK-algebra; (iii) the natural skew lattice partial order  $\leq_{\mathcal{H}}$  coincides with the implicative BCK-algebra partial order  $\leq_0^{(A; /, 0)}$ . Clearly the class of implicative  $\leq_0$ -BCK local skew lattices is a subvariety of the variety of implicative  $\leq_0$ -BCK local paralattices.

Proposition 3.3.52 and Proposition 3.3.51 direct attention towards those members of  $IP_{C''}$ ,  $\forall \in C''$ , that are implicative  $\leq_0$ -BCK local skew lattices, inasmuch as these algebras enjoy several important structural properties of skew Boolean  $\cap$ -algebras. In particular, if  $\langle A; \land, \lor, /, 0 \rangle$  is an implicative  $\leq_0$ -BCK local skew lattice, then: (i) the skew lattice with zero reduct  $\langle \land, \lor, 0 \rangle$  is distributive local; (ii) the polynomial reduct  $\langle A; \backslash, 0 \rangle$  is an implicative BCS-algebra; (iii) the

polynomial reduct  $\langle A; \cap \rangle$  is a meet semilattice; and (iv) for every  $a \in A$ , the principal subalgebra (a) generated by a is a Boolean sublattice. The precise relationship between implicative  $\leq_0$ -BCK local skew lattices and skew Boolean  $\cap$ -algebras is clarified in the following theorem, a version of which is asserted with second-order proof in Bignall and Leech [19, Section 4]. For the sake of completeness, we provide a direct proof here.

**Theorem 3.3.53.** (cf. [19, Theorem 4.2]) A skew Boolean  $\cap$ -algebra is term equivalent to an algebra  $\langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  where:

- 1. The reduct  $\langle A; \land, \lor, 0 \rangle$  is a join symmetric local skew lattice with zero;
- 2. The reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra;
- 3. The natural skew lattice partial order  $\leq_{\mathcal{H}}$  and the BCK partial order  $\leq_{0}^{\langle A; /, 0 \rangle}$  coincide.

In particular, given such an algebra  $\langle A; \land, \lor, /, 0 \rangle$ , standard difference  $\backslash$  and the intersection operation  $\cap$  are respectively defined on A by:

 $a \setminus b := a/(a \wedge b \wedge a)$  and  $a \cap b := a/(a/b)$ 

for any  $a, b \in A$ . Conversely, given a skew Boolean  $\cap$ -algebra  $\langle A; \land, \lor, \backslash, \cap, 0 \rangle$ and  $a, b \in A$ , implicative BCK difference / is defined on A by:

 $a/b := a \backslash (a \cap b).$ 

Thus the variety of skew Boolean  $\cap$ -algebras is termwise definitionally equivalent to the variety of implicative  $\leq_0$ -BCK join symmetric local skew lattices.

Proof. Let  $\mathbf{A} := \langle A; \land, \lor, /, 0 \rangle$  be an algebra of type  $\langle 2, 2, 2, 0 \rangle$  satisfying Conditions (1)-(3) of the theorem. Then  $\mathbf{A} \in \operatorname{IP}_{\mathcal{C}''}$ ,  $\mathcal{C}'' = \{\land, \lor, /, 0\}$ . To see the derived algebra  $\langle A; \land, \lor, \backslash, \cap, 0 \rangle$  (where  $a \backslash b := a/(a \land b \land a)$  and  $a \cap b = a/(a/b)$  for any  $a, b \in A$ ) is a skew Boolean  $\cap$ -algebra it is sufficient to show:

- (i) The polynomial reduct  $\langle A; \land, \lor, \backslash, 0 \rangle$  is a skew Boolean algebra;
- (ii) The polynomial reduct  $\langle A; \cap \rangle$  is a meet semilattice;

(iii) The induced semilattice partial order  $\leq^{\langle A; \cap \rangle}$  coincides with the natural skew lattice partial order  $\leq_{\mathcal{H}}$ .

For (i), because  $\mathbf{A} \in \operatorname{IP}_{C''}$ ,  $\forall \in \mathcal{C}''$ , from Proposition 3.3.51 we have that the polynomial reduct  $\langle A; \land, \lor, \backslash, 0 \rangle$  is an implicative BCS skew lattice. Since the skew lattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is join symmetric, from Theorem 3.3.21 it follows that the polynomial reduct  $\langle A; \land, \lor, \backslash, 0 \rangle$  is a skew Boolean algebra.

For (ii), because  $\mathbf{A} \in \mathsf{IP}_{\mathcal{C}''}$ ,  $\forall \in \mathcal{C}''$ , from Proposition 3.3.52(2) we have that polynomial reduct  $\langle A; \land, \lor, \cap \rangle$  is a skew lattice with intersections.

For (iii), we have  $a \leq \langle A; \cap \rangle b$  iff  $a \leq_0^{\langle A; /, 0 \rangle} b$  for any  $a, b \in A$ . Since  $A \in |\mathbb{P}_{\mathcal{C}}^{\prime\prime}$ , we infer that  $a \leq \langle A; \cap \rangle b$  iff  $a \leq_{\mathcal{H}} b$  for any  $a, b \in A$ . Hence the induced semilattice partial order  $\leq \langle A; \cap \rangle$  coincides with the natural skew lattice partial order  $\leq_{\mathcal{H}}$ .

For the converse, let  $\mathbf{A} := \langle A; \land, \lor, \backslash, \cap, 0 \rangle$  be a skew Boolean  $\cap$ -algebra. We verify Conditions (1)-(3) above are satisfied.

For (1), the reduct  $\langle A; \wedge, \vee, 0 \rangle$  is a distributive symmetric local skew lattice with zero by definition, and so in particular is a join symmetric local skew lattice with zero.

For (2), we have that the reduct  $\langle A; \cap \rangle$  is a meet semilattice by definition. For any  $a \in A$ , let  $(a]_{\langle A; \cap \rangle}$  denote the principal subalgebra of  $\langle A; \cap \rangle$  generated by a. Because of Proposition 1.4.34 and the locality of the skew lattice with zero reduct  $\langle A; \wedge, \vee, 0 \rangle$ , an argument similar to the proof of Proposition 3.3.45 shows that for every  $a \in A$ , the principal subalgebra  $(a]_{\langle A; \wedge, \vee, 0 \rangle}$  coincides with  $(a]_{\langle A; \cap \rangle}$ . Without loss of generality, therefore, throughout the remainder of the proof we can and will write  $(a]_A$  for both  $(a]_{\langle A; \wedge, \vee, 0 \rangle}$  and  $(a]_{\langle A; \cap \rangle}$ . Since  $\langle A; \wedge, \vee, 0 \rangle$  is locally Boolean (in the sense of §1.4.24),  $(a]_A$  is a Boolean lattice for every  $a \in A$ . Hence  $\langle A; \cap \rangle$  is semi-Boolean. From Theorem 1.6.21 it follows that  $\langle A; \cap \rangle$  possesses an induced implicative BCK difference operation /, where  $a/b := (a \cap b)'_{\langle a]_A}$  for any  $a, b \in A$ . To see  $a/b = a \setminus (a \cap b)$ , note:

(i)  $a \setminus (a \cap b)$  is the complement of  $a \wedge (a \cap b) \wedge a$  in  $(a]_A$  by definition of

standard difference;

(ii)  $a \wedge (a \cap b) \wedge a = a \cap b$  by Proposition 1.4.34.

Because of (i) and (ii), we have that  $a \setminus (a \cap b)$  is the complement of  $a \cap b$  in  $(a]_A$ ; that is to say  $a \setminus (a \cap b) = (a \cap b)'_{(a]_A}$ . Hence the induced implicative BCK difference / is term definable, and A has an implicative BCK-algebra polynomial reduct  $\langle A; /, 0 \rangle$ , where  $a/b := a \setminus (a \cap b)$  for any  $a, b \in A$ .

For (3), it is sufficient to show the BCK partial ordering  $\leq_0^{\langle A; /, 0 \rangle}$  coincides with the semilattice partial ordering  $\leq^{\langle A; \cap \rangle}$ . So suppose  $a \leq_0^{\langle A; /, 0 \rangle} b$ . Then  $0 = a/b = a \setminus (a \cap b)$ , so  $0 = (a \wedge (a \cap b) \wedge a)^*_{(a]_A} = (a \cap b)^*_{(a]_A}$ , which implies  $a \cap b = a$ . Thus  $a \leq^{\langle A; \cap \rangle} b$ . On the other hand, from  $a \leq^{\langle A; \cap \rangle} b$  we have  $a \cap b = a$ , whence  $a/b = a \setminus (a \cap b) = (a \wedge (a \cap b) \wedge a)^*_{(a]_A} = (a \cap b)^*_{(a]_A} = a^*_{(a]_A} = 0$ . Thus  $a \leq_0^{\langle A; /, 0 \rangle} b$ .

**Remark 3.3.54.** The condition of join symmetry cannot be omitted from the assertion of Theorem 3.3.53 since the variety of left handed skew Boolean  $\cap$ -algebras is properly contained within the class (in fact, variety) of all implicative  $\leq_0$ -BCK local skew lattices for which the skew lattice reduct is left handed. To see this, consider the algebra  $\mathbf{A} := \langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$ whose reduct  $\langle A; \land, \lor, 0 \rangle$  is the s<sup>1</sup>-ew lattice with zero reduct of the implicative BCS skew lattice of Remark 3.3.23, and whose operation /<sup>A</sup> is determined by the following operation table:

| $/^{\mathbf{A}}$ | 0 | a | b | С | d | е | f |
|------------------|---|---|---|---|---|---|---|
| 0                | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| a                | a | 0 | a | 0 | a | a | a |
| b                | b | b | 0 | b | 0 | b | b |
| С                | c | е | с | 0 | С | a | с |
| d                | d | d | f | d | 0 | d | b |
| е                | е | е | е | 0 | е | 0 | е |
| f                | f | f | f | f | 0 | f | 0 |

An easy sequence of checks shows that: (i) the skew lattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is local; (ii) the reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra;

334

and (iii)  $i \leq_{\mathcal{H}} j$  iff  $i \leq^{\langle A; /, 0 \rangle} j$  for any  $i, j \in A$ . Hence **A** is an implicative  $\leq_0$ -BCK local skew lattice for which the skew lattice reduct is left handed. However, **A** is not a left handed skew Boolean  $\cap$ -algebra, since the skew lattice with zero reduct  $\langle A; \land, \lor, 0 \rangle$  is not locally Boolean (in the sense of §1.4.24). In particular,  $\langle A; \land, \lor, 0 \rangle$  is not symmetric, since it is not join symmetric:  $(a \lor b) \lor (a \land b \land a) = d \lor 0 = d$  but  $(a \land b \land a) \lor (b \lor a) = 0 \lor c = c$ .

Remark 3.3.55 (Added in proof). A lower implicative BCK left normal band is an algebra  $\langle A; /, \Box, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  such that: (i) the reduct  $\langle A; /, 0 \rangle$ is an implicative BCK-algebra; (ii) the reduct  $\langle A; \Box, 0 \rangle$  is a left normal band with zero; and (iii) the implicative BCK partial order  $\leq_0^{\langle A; /, 0 \rangle}$  and the natural band partial order  $\leq_{\mathcal{H}}$  coincide. By an unpublished result of the author, an algebra  $\mathbf{A} := \langle A; /, \Box, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is a lower implicative BCK left normal band iff  $\mathbf{A} \models (1.35)-(1.38)$ ,  $\mathbf{A} \models (x \Box y) \Box z \approx x \Box (y \Box z), x \Box x \approx$  $x, (x \Box y) \Box z \approx (x \Box z) \Box y$ , and  $\mathbf{A} \models (x \Box y) / x \approx 0, x \Box (x/y) \approx x/y$ , whence the class iBCKInB of lower implicative BCK left normal bands is a variety.

Recall the definition of an implicative BCSK-algebra from Remark 3.2.28. Because of Corollary 2.3.22(1), any implicative BCSK-algebra  $\langle A; /, \backslash, 0 \rangle$  has a lower implicative BCK left normal band polynomial reduct  $\langle A; /, \Box, 0 \rangle$ , where  $a \Box b := a \backslash (a \backslash b)$  for any  $a, b \in A$ . Conversely, an unpublished result of the author shows that any lower implicative BCK left normal band  $\langle A; /, \Box, 0 \rangle$  has an implicative BCSK-algebra polynomial reduct  $\langle A; /, \backslash, 0 \rangle$ , where  $a \backslash b := a/(a \Box b)$  for any  $a, b \in A$ . Hence the variety of lower implicative BCK left normal bands is termwise definitionally equivalent to the variety of implicative BCSK-algebras (compare this result to that of Theorem 3.3.53).

Call a lower implicative BCK left normal band *flat* if its underlying poset is flat. The proof of Theorem 3.2.27, in conjunction with preceding remarks, implies that any pointed fixedpoint discriminator algebra  $\langle A; f, 0 \rangle$  has a flat lower implicative BCK left normal band polynomial reduct  $\langle A; /, \Box, 0 \rangle$ , where:

$$a/b := f(0, f(a, b, a), a)$$
 and  $a \sqcap b := f(0, f(0, b, a), a)$ 

for any  $a, b \in A$ . Conversely, the description of the subdirectly irreducible implicative BCSK-algebras given in Remark 3.2.28 implies that a lower implicative BCK left normal band is subdirectly irreducible iff it is flat, whence the class iBCKlnB is a pointed fixed point discriminator variety, with pointed fixed point discriminator term:

$$f(x, y, z) := (z \setminus (z \sqcap (x/y))) \sqcap (z \setminus (z \sqcap (y/x))).$$

Recall from Remark 3.2.28 that  $FPD_0$  denotes the pure pointed fixedpoint discriminator variety. In view of the preceding discussion, it is easy to see that  $FPD_0$  is termwise definitionally equivalent to iBCKlnB. Since the congruence structure of any algebra in a fixedpoint discriminator variety is (by Lemma 1.5.10) completely determined by the fixedpoint discriminator term, any algebra A in a pointed fixedpoint discriminator variety must have a lower implicative BCK left normal band polynomial reduct whose congruences coincide with those of A (compare this result to that of Corollary 1.4.40).

Ignoring issues of similarity type, it is clear from the conditions (i), (ii) and (iii) above defining lower implicative BCK left normal bands that iBCKInB is a subvariety of the variety of lower implicative  $\leq_0$ -BCK normal bands. This observation, in conjunction with the above remarks and the results of §3.2.6 and §1.4.37, implies that the study of the classes BP<sub>C</sub>,  $\{ \div, 0 \} \subseteq C \subseteq \{ \land, \lor , \div, 0 \}$  (and hence, by extension—recall Proposition 3.3.51—the classes PQ<sub>C</sub>,  $\{ \div, 0 \} \subseteq C \subseteq \{ \land, \lor, \div, 0 \}$ ) encompasses, to within termwise definitional equivalence, the study of the pure binary discriminator, pure pointed fixed-point discriminator and pure pointed ternary discriminator varieties. Consequently, the study of pre-BCK quasilattices, BCK paralattices and related structures provides a unifying framework for the study of several important classes of 'generalised Boolean structures' arising naturally in universal algebra and algebraic logic.

**Corollary 3.3.56.** An algebra  $\langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  is a generalised Boolean algebra iff the following conditions hold:

1. The reduct  $\langle A; \land, \lor, 0 \rangle$  is a lattice with zero;

336

- 2. The reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra;
- 3. The BCK meet  $x \cap y$  and the lattice meet  $x \wedge y$  coincide.

Henceforth, by abuse of language and notation we will always understand by the term 'skew Boolean  $\cap$ -algebra' an algebra  $\langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$ satisfying conditions (1)-(3) of Theorem 3.3.53. Conversely, by abuse of language and notation an algebra  $\langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  satisfying the defining conditions of Theorem 3.3.53 will always be called a 'skew Boolean  $\cap$ algebra'. See also Bignall and Leech [19, Section 4]. Given these conventions, we have the following result, a first-order proof of which may be found in [210, Section 5.2].

**Theorem 3.3.57.** An algebra  $\langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  is a skew Boolean  $\cap$ -algebra iff the following identities are satisfied:

$$(x \lor y) \lor z \approx x \lor (y \lor z) \tag{3.99}$$

$$(x \wedge y) \wedge z \approx x \wedge (y \wedge z) \tag{3.100}$$

$$x \wedge (x \vee y) \approx x \tag{3.101}$$

$$(y \wedge x) \lor x \approx x \tag{3.102}$$

$$x \wedge (y \vee z) \approx (x \wedge y) \vee (x \wedge z)$$

$$(x \vee y) \wedge z \approx (x \wedge z) \vee (y \wedge z)$$

$$(3.103)$$

$$(3.104)$$

$$x/(x/y) \approx y/(y/x) \tag{3.105}$$

$$(x/y)/z \approx (x/z)/y \tag{3.106}$$

$$x/(y/x) \approx x \tag{3.107}$$

 $(x \wedge y \wedge x)/x \approx 0 \tag{3.108}$ 

$$x \lor (x/y) \approx x \tag{3.109}$$

 $(x/y) \lor x \approx x. \tag{3.110}$ 

#### Thus the class of skew Boolean $\cap$ -algebras is a variety.

*Proof.* Let  $\mathbf{A} := \langle A; \land, \lor, /, 0 \rangle$  be an algebra of type  $\langle 2, 2, 2, 0 \rangle$ .

(⇒) Suppose A is a skew Boolean ∩-algebra. Then A \models (3.99)-(3.102), (3.105)-(3.107) by the defining conditions of Theorem 3.3.53. Moreover, because of Theorem 3.3.53,  $A \in |P_{C''}, C'' = \{ \land, \lor, \land, 0 \}$ , so  $A \models (3.108)-(3.110)$ by Theorem 3.3.49. Since  $A \in |P_{C''}$ , the skew lattice with zero reduct  $\langle A; \land$  $,\lor, 0 \rangle$  is distributive local by Proposition 3.3.52, and so is symmetric by Lemma 1.4.17 and hypothesis. Therefore  $A \models (3.103)-(3.104)$  by Proposition 1.4.22, and the proof is complete.

( $\Leftarrow$ ) Suppose  $\mathbf{A} \models (3.99)$ -(3.110). To see  $\mathbf{A}$  is a skew Boolean  $\cap$ -algebra it is sufficient to show the defining conditions of Theorem 3.3.53 are satisfied, viz.:

- (i) The reduct  $\langle A; \land, \lor, 0 \rangle$  is a join symmetric local skew lattice with zero;
- (ii) The reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra;
- (iii) The partial orders  $\leq_{\mathcal{H}}$  and  $\leq_0^{\langle A; /, 0 \rangle}$  coincide.

For (i), we first show  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero. So let  $a, b \in A$ . By (3.101) and (3.109)  $a = a \land (a \lor (a/b)) = a \land a$ , whence  $\land$  is idempotent. By (3.101), idempotence of  $\land$  and (3.103) we have  $a = a \land (a \lor b) = (a \land a) \lor (a \land b) = a \lor (a \land b)$ . Similarly  $(b \lor a) \land a = (b \land a) \lor (a \land a) = (b \land a) \lor a = a$  by (3.104), idempotence of  $\land$  and (3.102). By absorption,  $a = a \lor (a \land (a \lor b)) = a \lor a$  and thus  $\langle A; \land, \lor \rangle$  is a skew lattice. Also,  $0 = a/(a \land a \land a) = a/a$  by (3.108) and idempotence of  $\land$ , whence  $a = a \lor (a/a) = a \lor 0$  by (3.109) and  $a = (a/a) \lor a = 0 \lor a$  by (3.110). Thus  $\langle A; \land, \lor, 0 \rangle$  is a skew lattice with zero, and the identities (3.103)-(3.104) in conjunction with Proposition 1.4.22 now ensure  $\langle A; \land, \lor, 0 \rangle$  is distributive symmetric local. Hence  $\langle A; \land, \lor, 0 \rangle$  is join symmetric local.

For (ii), we have already observed a/a = 0 for all  $a \in A$ , so  $\mathbf{A} \models x/x \approx \mathbf{0}$ . From this remark, the identities (3.105)-(3.107), and Theorem 1.6.17 it follows that the reduct  $\langle A; /, 0 \rangle$  is an implicative BCK-algebra.

For (iii), just notice that  $\langle A; \land, \lor, /, 0 \rangle$  is a BCK paralattice by (i), Lemma 1.4.8, (ii), (3.108)-(3.110) and Theorem 3.3.49. Therefore the natural skew lattice

partial order  $\leq_{\mathcal{H}}$  and the BCK partial order  $\leq_0^{\langle A; /, 0 \rangle}$  coincide.

**Corollary 3.3.58.** An algebra  $\langle A; \land, \lor, /, 0 \rangle$  of type  $\langle 2, 2, 2, 0 \rangle$  is a left handed skew Boolean  $\cap$ -algebra iff the identities of Theorem 3.3.57 are satisfied, where (3.108) is replaced by the identity:

$$(x \wedge y)/x \approx 0. \tag{3.108'}$$

Proof. Let  $\mathbf{A} := \langle A; \land, \lor, /, 0 \rangle$  be an algebra of type  $\langle 2, 2, 2, 0 \rangle$  such that  $\mathbf{A} \models (3.99)-(3.107), (3.108'), (3.109)-(3.110)$  and let  $a, b \in A$ . By (3.103), we have that  $a \land (b \lor a) = (a \land b) \lor (a \land a) = (a \land b) \lor a$ . But by (3.108') and (3.92),  $(a \land b) \lor a = ((a \land b)/0) \lor a = ((a \land b)/((a \land b)/a)) \lor a = a$ . Hence  $a \land (b \lor a) = a = (a \land b) \lor a$ , and  $\mathbf{A}$  is left handed by the remarks of §1.4.14. Since the converse is clear, the corollary is proved.

**3.3.59.** Double-Pointed Skew Boolean  $\cap$ -algebras. Let A be a skew Boolean  $\cap$ -algebra with maximal class M. By analogy with the theory of pre-BCK-algebras, an algebra  $\mathbf{A}^1 := \langle A; \land, \lor, /, 0, 1 \rangle$  obtained from A upon adjoining to the language of A a new nullary operation symbol 1 whose canonical interpretation on  $\mathbf{A}^1$  is a fixed  $1 \in M$  is called a *quasi-bounded skew Boolean*  $\cap$ -algebra. Clearly the class of quasi-bounded skew Boolean  $\cap$ -algebras is a variety, axiomatised relative to the variety of skew Boolean  $\cap$ -algebras by either of the identities:

 $x \setminus \mathbf{1} \approx \mathbf{0}$  or  $x \wedge \mathbf{1} \wedge x \approx x$ .

**Remark 3.3.60.** As per pre-BCK-algebras, in passing from a given skew Boolean  $\cap$ -algebra A with maximal class M to a quasi-bounded skew Boolean  $\cap$ -algebra  $A^1$  there is typically no natural choice of maximal element  $1 \in M$ . In general, it seems plausible that distinct choices of maximal element could give rise to non-isomorphic quasi-bounded skew Boolean  $\cap$ -algebras, although this possibility cannot occur in the *completely reducible* case (that is, when A is isomorphic to a direct product of primitive algebras).

A version of the following result occurs in Blok and Pigozzi [34, Section 1].

339

**Proposition 3.3.61.** (cf. [34, Theorem 1.1]) An algebra  $\langle A; \land, \lor, /, 0, 1 \rangle$  of type  $\langle 2, 2, 2, 0, 0 \rangle$  is a double-pointed skew Boolean  $\cap$ -algebra iff it is a quasibounded skew Boolean  $\cap$ -algebra. Thus the class SBIA<sup>+</sup> of double-pointed skew Boolean  $\cap$ -algebras is a variety, axiomatised relative to the variety of skew Boolean  $\cap$ -algebras by either of the identities  $x \backslash 1 \approx 0$  or  $x \land 1 \land x \approx x$ .

**Proof.** Because the ideals of any skew Boolean  $\cap$ -algebra coincide with the ideals of its implicative BCS-algebra polynomial reduct, the first assertion of the proposition follows from Lemma 2.2.27. The second assertion of the proposition now follows from previous remarks.

Let A be a primitive skew Boolean  $\cap$ -algebra (recall Example 1.4.35). Since A is primitive, it has a maximal class M, which consists of all non-zero elements of A. Thus A induces a double-pointed *primitive* skew Boolean  $\cap$ -algebra A<sup>+</sup>. Because of Remark 3.3.60, A<sup>+</sup> is unique to within isomorphism.

**Theorem 3.3.62.** The following assertions hold in the variety of doublepointed skew Boolean  $\cap$ -algebras:

- The double-pointed primitive skew Boolean ∩-algebras are the non-trivial simple algebras;
- 2. The double-pointed primitive skew Boolean ∩-algebras are the subdirectly irreducible algebras;
- 3. Every non-trivial double-pointed skew Boolean ∩-algebra is a subdirect product of primitive algebras.

Proof. For Item (1) [Item (2)] let  $\mathbf{A}^+$  be a non-trivial simple [subdirectly irreducible] double-pointed skew Boolean  $\cap$ -algebra. Its skew Boolean  $\cap$ -algebra reduct  $\mathbf{A} := \langle A; \land, \lor, /, 0 \rangle$  must also be non-trivial and simple [subdirectly irreducible], since  $\mathbf{A}$  and  $\mathbf{A}^+$  have the same congruences. By Theorem 1.4.36 we deduce that  $\mathbf{A}$  is primitive, so  $\mathbf{A}^+$  is primitive. Item (3) now follows from Birkhoff's subdirect representation theorem [55, Theorem II§8.6].

Recall from Theorem 1.4.38 that  $t(x, y, z) := (z \setminus (x \Delta y)) \vee (x/y)$  is a ternary discriminator term for SBIA, where  $x \Delta y \approx (x/y) \vee (y/x)$  (since  $x \setminus (x \cap y) \approx x/y$ ). Because of Theorem 3.3.62, we may infer the following result.

**Corollary 3.3.63.** The class of double-pointed skew Boolean  $\cap$ -algebras is a discriminator variety, with discriminator term given by:

$$t(x, y, z) := (z \setminus (x \Delta y)) \vee (x/y).$$

Let V be a pointed discriminator variety (say with 0) with discriminator term t(x, y, z). By the remarks of §1.5.9, V is a pointed fixedpoint discriminator variety, with pointed fixedpoint discriminator term f(x, y, z) :=t(t(x, y, z), t(x, y, 0), 0). For [double-pointed] skew Boolean  $\cap$ -algebras we have the following simplification.

**Corollary 3.3.64.** The class of [double-pointed] skew Boolean  $\cap$ -algebras is a fixed point discriminator variety, with fixed point discriminator term given by  $f(x, y, z) := z \setminus (x \Delta y).$ 

*Proof.* The result is established by an easy inspection of the subdirectly irreducible [double-pointed] skew Boolean ∩-algebras.

**Remark 3.3.65.** Because of Remark 3.2.28,  $f(x, y, z) := (z \setminus (x/y)) \setminus (y/x)$  is also a pointed fixed point discriminator term for both SBIA and SBIA<sup>+</sup>.

By the pure double-pointed discriminator variety  $PD_0^+$  we mean the doublepointed discriminator variety of type (3,0,0) generated by the class of all double-pointed discriminator algebras (A; t, 0, 1) where t is the discriminator and 0 and 1 are residually distinct nullary operations, while by the variety of double-pointed left handed skew Boolean  $\cap$ -algebras  $hSBIA^+$  we mean the variety of all left handed skew Boolean  $\cap$ -algebras that are double-pointed. From Theorem 1.4.39, Proposition 3.3.61 and [34, Theorem 1.1] we may infer the following result.

**Theorem 3.3.66.** The variety  $PD_0^+$  is termwise definitionally equivalent to the variety of double-pointed left handed skew Boolean  $\cap$ -algebras. In particular, given  $\langle A; t, 0, 1 \rangle \in PD_0^+$ , double-pointed left handed skew Boolean  $\cap$ -operations  $\wedge, \vee$  and / are defined on A by:

 $a \wedge b := t(b, t(b, 0, a), a)$ 

341

 $a \lor b := t(b, 0, a)$ a/b := t(a, b, 0)

for any  $a, b \in A$ . Conversely, given a double-pointed left handed skew Boolean  $\cap$ -algebra  $\langle A; \land, \lor, /, 0, 1 \rangle$  and  $a, b, c \in A$ , the operation  $t(a, b, c) := (c/(c \land b)) \lor (c \land a) \lor (a/b)$  yields an algebra  $\langle A; t, 0, 1 \rangle$  in  $PD_0^+$ .

Clearly  $PD_0^+$  is also termwise definitionally equivalent to the variety of doublepointed right handed skew Boolean  $\cap$ -algebras.

**Corollary 3.3.67.** Any algebra  $\mathbf{A}$  in a double-pointed discriminator variety has a double-pointed left handed skew Boolean  $\cap$ -algebra polynomial reduct whose congruences coincide with those of  $\mathbf{A}$ .

*Proof.* The result follows immediately from Theorem 3.3.66 and Lemma 1.5.10. ■

**Remark 3.3.68.** A pseudo-interior algebra is an algebra  $\langle A; \cdot, \rightarrow, \circ, 1 \rangle$  of type (2, 2, 1, 0) that is essentially a hybrid of an interior algebra and a residuated partially ordered monoid (for a precise definition, see [35, Definition 2.6]). A pseudo-interior algebra with compatible operations is an algebra  $\langle A; \cdot, \rightarrow \rangle$  $\langle \circ, 1, F_i \rangle_{i \in I}$  where  $\langle A; \cdot, \rightarrow, \circ, 1 \rangle$  is a pseudo-interior algebra and the additional operations  $\langle F_i \rangle_{i \in I}$  are such that every congruence on A has the substitution property with respect to each  $F_i$  [34, Corollary 2.17]. Pseudo-interior algebras with compatible operations were introduced by Blok and Pigozzi in [35] as the algebraic counterpart of a certain assertional logic inherent in any variety with a commutative, regular TD term; for details, see [34, Theorem 4.1]. By [35, Corollary 4.3], a double-pointed variety V is a ternary discriminator variety iff it is termwise definitionally equivalent to a congruence permutable, semisimple variety of pseudo-interior algebras with compatible operations; the hypothesis that V is double-pointed is essential. The description of doublepointed discriminator varieties and the associated assertional logics afforded by this result should be compared and contrasted with Corollary 3.3.67 and the developments of §3.3.69 in the sequel.

**3.3.69.** The Assertional Logic of the Variety IhSBIA<sup>+</sup>. Recall from §1.8.9 that for any quasivariety K with 1, the inherent assertional logic S(K, 1) of K may be defined by specifying that, for all  $\Gamma \cup \{\varphi\} \subseteq \operatorname{Fm}_{\mathcal{L}}, \Gamma \vdash_{S(K,1)} \varphi$  iff  $\{\psi \approx 1 : \psi \in \Gamma\} \models_{\mathsf{K}} \varphi \approx 1$ . It follows from this observation and Theorem 3.3.66 that the assertional logic  $S(\mathsf{IhSBIA}^+, 0)$  of the variety of double-pointed left handed skew Boolean  $\cap$ -algebras is definitionally equivalent to the assertional logic  $S(\mathsf{PD}_0^+, 0)$  of the pure double-pointed discriminator variety. This remark calls for a study of  $S(\mathsf{IhSBIA}^+, 0)$ ; in this subsection we provide a framework for such a study by axiomatising this deductive system.

Throughout this subsection, we work with a fixed language  $\mathcal{L} := \langle \wedge, \vee, \Rightarrow, \mathbf{0}, \mathbf{1} \rangle$  of type  $\langle 2, 2, 2, 0, 0 \rangle$ , with fixed abbreviations:

 $p \Leftrightarrow q := (p \Rightarrow q) \land (q \Rightarrow p)$  $p \rightarrow q := (p \lor q) \Rightarrow q$  $p \leftrightarrow q := (p \rightarrow q) \land (q \rightarrow p)$  $\neg p := p \rightarrow 0$ 

for propositional variables p, q, and with a fixed and defining collection  $Ax \cup Ir$  of axioms and inferences rules, *viz.*:

| (  | $(p \wedge q) \wedge r$ | )⇔( | $(p \wedge (q \wedge r))$ | ) (5 | S1) |
|----|-------------------------|-----|---------------------------|------|-----|
| ۰. | · · · ·                 | , , |                           |      |     |

$$((p \lor q) \lor r) \Leftrightarrow (p \lor (q \lor r)) \tag{S2}$$

 $(p \land (p \lor q)) \Leftrightarrow p$  (S3)

$$((q \land p) \lor p) \Leftrightarrow p \tag{S4}$$

$$((p \lor q) \land (p \lor r)) \Leftrightarrow (p \lor (q \land r))$$
(S5)

$$((p \lor r) \land (q \lor r)) \Leftrightarrow ((p \land q) \lor r)$$
(S6)

 $((p \Rightarrow q) \Rightarrow p) \Leftrightarrow p \tag{S7}$ 

$$((q \Rightarrow p) \Rightarrow p) \Leftrightarrow ((p \Rightarrow q) \Rightarrow q)$$
 (S8)

| 3.3. Pre-BCK Quasilattices and BCK Paralattices                                       | 344   |
|---|-------|
| $(p \Rightarrow (q \Rightarrow r)) \Leftrightarrow (q \Rightarrow (p \Rightarrow r))$ | (S9)  |
| $(p \Rightarrow (q \lor p)) \Leftrightarrow 1$  | (S10) |
|   |       |

$$(p \land (q \Rightarrow p)) \Leftrightarrow p$$
 (S11)

$$((q \Rightarrow p) \land p) \Leftrightarrow p \tag{S12}$$

 $(\mathbf{0} \lor p) \Leftrightarrow p \tag{S13}$ 

 $\frac{p \Leftrightarrow q}{p \Leftrightarrow p} (\mathbf{R}) \qquad \frac{p \Leftrightarrow q}{q \Leftrightarrow p} (\mathbf{S}) \qquad \frac{p \Leftrightarrow q, \quad q \Leftrightarrow r}{p \Leftrightarrow r} (\mathbf{T})$   $\frac{p \Leftrightarrow q, \quad r \Leftrightarrow s}{(p \land r) \Leftrightarrow (q \land s)} (\mathbf{CP} \land \wedge) \qquad \frac{p \Leftrightarrow q, \quad r \Leftrightarrow s}{(p \lor r) \Leftrightarrow (q \lor s)} (\mathbf{CP} \cdot \vee)$   $\frac{p \Leftrightarrow q, \quad r \Leftrightarrow s}{(p \Rightarrow r) \Leftrightarrow (q \Rightarrow s)} (\mathbf{CP} \cdot \Rightarrow)$   $\frac{p \Leftrightarrow \mathbf{1}}{p} (\mathbf{1} \cdot \mathbf{I}) \qquad \frac{p}{p \Leftrightarrow \mathbf{1}} (\mathbf{1} \cdot \mathbf{E})$ 

The skew Boolean propositional calculus is the deductive system SRPC over the language  $\mathcal{L}$  determined by the axiomatisation  $Ax \cup Ir$ . We denote by  $\vdash_{\text{SRPC}}$  the consequence relation of SRPC. The skew Boolean propositional calculus was introduced implicitly by the author's Ph.D. supervisor in [18] and explicitly by the author's Ph.D. supervisor in [20]; applications of the skew Boolean propositional calculus to theoretical computer science have been considered by the author and the author's Ph.D. supervisor in [21, 22].

**Theorem 3.3.70.** SBPC is algebraisable with equivalence formula  $p \Leftrightarrow q$  and defining equation  $p \approx 1$ .

*Proof.* Immediate by Theorem 1.8.2 and the description of  $Ax \cup Ir$ .

By Theorem 1.8.3, the equivalent algebraic semantics of SBPC is the quasivariety K over the language  $\mathcal{L}$  axiomatised by the following collection of identities

| 3.3. Pre-BCK Quasilattices and BCK Paralattices  | 345     |  |  |  |
|--|---------|--|--|--|
| nd quasi-identities:   |         |  |  |  |
| $ig((x \wedge y) \wedge zig) \Leftrightarrow ig(x \wedge (y \wedge z)ig) pprox 1$  | (3.111) |  |  |  |
| $ig((x \lor y) \lor zig) \Leftrightarrow ig(x \lor (y \lor z)ig) pprox 1$  | (3.112) |  |  |  |
| $ig(x \wedge (x \lor y)ig) \Leftrightarrow x pprox 1$  | (3.113) |  |  |  |
| $ig((y \wedge x) \lor xig) \Leftrightarrow x pprox 1$  | (3.114) |  |  |  |
| $ig((x \lor y) \land (x \lor z)ig) \Leftrightarrow ig(x \lor (y \land z)ig) pprox 1$   | (3.115) |  |  |  |
| $ig((x \lor z) \land (y \lor z)ig) \Leftrightarrow ig((x \land y) \lor zig) pprox 1$   | (3.116) |  |  |  |
| $((x \Rightarrow y) \Rightarrow x) \Leftrightarrow x \approx 1$  | (3.117) |  |  |  |
| $((y \Rightarrow x) \Rightarrow x) \Leftrightarrow ((x \Rightarrow y) \Rightarrow y) \approx 1$                                    | (3.118) |  |  |  |
| $(x \Rightarrow (y \Rightarrow z)) \Leftrightarrow (y \Rightarrow (x \Rightarrow z)) \approx 1$                                    | (3.119) |  |  |  |
| $(x \Rightarrow (y \lor x)) \Leftrightarrow 1 \approx 1$   | (3.120) |  |  |  |
| $ig(x \wedge (y \Rightarrow x)ig) \Leftrightarrow x pprox 1$   | (3.121) |  |  |  |
| $((y\Rightarrow x)\wedge x)\Leftrightarrow xpprox 1$   | (3.122) |  |  |  |
| $(0 \lor x) \Leftrightarrow x pprox 1$   | (3.123) |  |  |  |
| $x \Leftrightarrow x pprox 1$  | (3.124) |  |  |  |
| $y \Leftrightarrow x pprox 1 \supset x \Leftrightarrow y pprox 1$  | (3.125) |  |  |  |
| $x \Leftrightarrow y \approx 1 \& y \Leftrightarrow z \approx 1 \supset x \Leftrightarrow z \approx 1$                             | (3.126) |  |  |  |
| $x \Leftrightarrow y pprox 1 \ \& \ z \Leftrightarrow w pprox 1 \supset x \land z \Leftrightarrow y \land w pprox 1$               | (3.127) |  |  |  |
| $x \Leftrightarrow y \approx 1 \ \& \ z \Leftrightarrow w \approx 1 \supset x \lor z \Leftrightarrow y \lor w \approx 1$           | (3.128) |  |  |  |
| $x \Leftrightarrow y \approx 1 \& z \Leftrightarrow w \approx 1 \supset x \Rightarrow z \Leftrightarrow y \Rightarrow w \approx 1$ | (3.129) |  |  |  |
|  |         |  |  |  |

ر برای برای جنها
| 3.3. Pre-BCK Quasilattices and BCK Paralattices     | 346     |
|---|---------|
| $x \approx 1 \supset x \Leftrightarrow 1 \approx 1$ | (3.130) |
| $x \Leftrightarrow 1 pprox 1 \supset x pprox 1$     | (3.131) |
| $x \Leftrightarrow y pprox 1 \supset x pprox y$     | (3.132) |

where  $x \Leftrightarrow y$  abbreviates  $(x \Rightarrow y) \land (y \Rightarrow x)$  for individual variables x, y. To see the quasivariety K has a familiar description, let  $hSBIA^{+D}$  denote the variety with language  $\mathcal{L}$  defined by the following set of identities:

$$(x \wedge y) \wedge z \approx x \wedge (y \wedge z) \tag{3.133}$$

$$(x \lor y) \lor z \approx x \lor (y \lor z) \tag{3.134}$$

$$x \wedge (x \vee y) \approx x \tag{3.135}$$

$$(y \wedge x) \lor x \approx x \tag{3.136}$$

$$(x \lor y) \land (x \lor z) \approx x \lor (y \land z) \tag{3.137}$$

$$(x \lor z) \land (y \lor z) \approx (x \land y) \lor z \tag{3.138}$$

$$(x \Rightarrow y) \Rightarrow x \approx x \tag{3.139}$$

$$(y \Rightarrow x) \Rightarrow x \approx (x \Rightarrow y) \Rightarrow y$$
 (3.140)

$$x \Rightarrow (y \Rightarrow z) \approx y \Rightarrow (x \Rightarrow z) \tag{3.141}$$

$$x \Rightarrow (y \lor x) \approx 1 \tag{3.142}$$

$$x \wedge (y \Rightarrow x) \approx x \tag{3.143}$$

$$(y \Rightarrow x) \land x \approx x \tag{3.144}$$

$$\mathbf{0} \lor x \approx x. \tag{3.145}$$

Let  $\mathbf{T}_{\mathcal{L}}(\mathbf{X})$  denote the term algebra of type  $\mathcal{L}$  over  $\mathbf{X}$  and let  $\mathbf{T}_{\mathcal{L}^{D}}(\mathbf{X})$  denote the term algebra of type  $\mathcal{L}^{D}$  over  $\mathbf{X}$ , where  $\mathcal{L}^{D}$  is the language of skew Boolean  $\cap$ -

algebras. Consider the maps  $\xi : T_{\mathcal{L}}(\mathbf{X}) \to T_{\mathcal{L}^{\mathcal{D}}}(\mathbf{X})$  and  $\eta : T_{\mathcal{L}^{\mathcal{D}}}(\mathbf{X}) \to T_{\mathcal{L}}(\mathbf{X})$ defined respectively by:

| $\xi(1) := 0$                             |   |
|---|---|
| $\xi(0) := 1$                             |   |
| $\xi(x):=x$                               | $x\in \mathbf{X}$                               |
| $\xi(p \wedge q) := \xi(q) \vee \xi(p)$   | $p,q\in \mathrm{T}_{\mathcal{L}^D}(\mathbf{X})$ |
| $\xi(p \lor q) := \xi(q) \land \xi(p)$    | $p,q\in \mathrm{T}_{\mathcal{L}^D}(\mathbf{X})$ |
| $\xi(p \Rightarrow q) := \xi(q) / \xi(p)$ | $p,q\in \mathrm{T}_{\mathcal{L}^D}(\mathbf{X})$ |

and:

| $\eta(0) := 1$                             |  |
|--|--|
| $\eta(1):=0$                               |  |
| $\eta(x):=x$                               | $x\in \mathbf{X}$                              |
| $\eta(r \wedge s) := \eta(s) \vee \eta(r)$ | $r,s\in \mathrm{T}_{\mathcal{L}}(\mathbf{X})$  |
| $\eta(r \lor s) := \eta(s) \land \eta(r)$  | $r,s\in \mathrm{T}_\mathcal{L}(\mathbf{X})$    |
| $\eta(r/s) := \eta(s) \Rightarrow \eta(r)$ | $r,s\in \mathrm{T}_{\mathcal{L}}(\mathbf{X}).$ |

(The maps  $\xi$  and  $\eta$  so defined should not be confused with the similar maps of §2.1.33 or §3.1.1 in the prequel.) Because of Corollary 3.3.58 and the definition of lhSBIA<sup>+D</sup>, the proof of the following lemma is trivial and so is omitted.

**Lemma 3.3.71.** For  $p, q \in T_{\mathcal{L}}(X)$  and  $r, s \in T_{\mathcal{L}^{D}}(X)$  the following assertions hold:

1. If  $hSBIA^{+D} \models p \approx q$  then  $hSBIA^{+} \models \xi(p) \approx \xi(q)$ ;

2. If  $hSBIA^+ \models r \approx s$  then  $hSBIA^{+D} \models \eta(r) \approx \eta(s)$ .

Moreover,  $\eta \circ \xi = \omega_{T_{\mathcal{L}}(\mathbf{X})}$  and  $\xi \circ \eta = \omega_{T_{\mathcal{L}}\mathcal{D}}(\mathbf{X})$ .

By Lemma 3.3.71, the variety  $hSBIA^{+D}$  is termwise definitionally equivalent to (in fact, is dually isomorphic to) the variety of double-pointed left handed

skew Boolean  $\cap$ -algebras. In other words,  $hSBIA^{+D}$  is precisely the variety of dual double-pointed left handed skew Boolean  $\cap$ -algebras.

**Proposition 3.3.72.** The quasivariety K coincides with the variety  $hSBIA^{+D}$  of dual double-pointed left handed skew Boolean  $\cap$ -algebras. Hence SBPC is strongly algebraisable.

*Proof.* (Sketch) To prove the proposition, it is sufficient to show the quasivariety K and the variety  $IhSBIA^{+D}$  coincide. To establish the inclusion  $K \subseteq IhSBIA^{+D}$ , note first that:

$$\mathsf{IhSBIA}^{+D} \models x \approx y \quad \text{iff} \quad \mathsf{IhSBIA}^{+D} \models x \Leftrightarrow y \approx 1 \tag{3.146}$$

by Lemma 3.3.71 and Proposition 3.3.37(3). To complete the proof we show:

- (i)  $lhSBIA^{+D} \models (3.111) (3.123);$
- (ii)  $hSBIA^{+D} \models (3.124) (3.132)$ .

For (i), consider a defining identity of K, say (3.111). Then  $K \models (3.111)$ , viz.:

$$\mathsf{K} \models ((x \land y) \land z) \Leftrightarrow (x \land (y \land z)) \approx \mathbf{1}.$$

Now by (3.133), we have that:

$$\mathsf{IhSBIA}^{+D} \models (x \land y) \land z \approx x \land (y \land z),$$

so by (3.146), we have that:

$$\mathsf{lhSBIA}^{+D} \models ((x \land y) \land z) \Leftrightarrow (x \land (y \land z)) \approx \mathbf{1},$$

whence  $hSBIA^{+D} \models (3.111)$ . Because of the axiomatisation of  $hSBIA^{+D}$  by (3.133)-(3.145), a suitable modification of the preceding argument now shows that  $hSBIA^{+D} \models (3.112)-(3.123)$  for each of the remaining identities (3.112)-(3.123) defining K.

For (ii), consider a defining quasi-identity of K, say (3.129). Then  $K \models (3.129)$ ;

that is to say:

 $\mathsf{K} \models x \Leftrightarrow y \approx \mathbf{1} \& z \Leftrightarrow w \approx \mathbf{1} \supset x \Rightarrow z \Leftrightarrow y \Rightarrow w \approx \mathbf{1}.$ 

By properties of  $\approx$ , we have that:

$$\mathsf{IhSBIA}^{+D} \models x \approx y \& z \approx w \supset x \Rightarrow z \approx y \Rightarrow w$$

vacuously, so by (3.146) we have that:

$$|\mathsf{hSBIA}^{+D} \models x \Leftrightarrow y \approx \mathbf{1} \& z \Leftrightarrow w \approx \mathbf{1} \supset x \Rightarrow z \Leftrightarrow y \Rightarrow w \approx \mathbf{1}.$$

Hence  $hSBIA^{+D} \models (3.129)$ . Because of properties of  $\approx$ , a suitable modification of the preceding argument now shows  $hSBIA^{+D} \models (3.125)-(3.128)$ , (3.130)-(3.132) for each of the remaining quasi-identities (3.125)-(3.128) and (3.130)-(3.132) axiomatising K.

By (i) and (ii),  $hSBIA^{+D} \models (3.111)-(3.132)$ , so  $K \subseteq hSBIA^{+D}$  as desired.

To establish the inclusion  $IhSBIA^{+D} \subseteq K$ , note first that:

$$\mathsf{K} \models x \approx y \quad \text{iff} \quad \mathsf{K} \models x \Leftrightarrow y \approx 1, \tag{3.147}$$

because of (3.124) and (3.132). To complete the proof, consider a defining identity of  $hSBIA^{+D}$ , say (3.133). Then  $hSBIA^{+D} \models (3.133)$ ; that is to say:

$$\mathsf{hSBIA}^{+D} \models (x \land y) \land z \approx x \land (y \land z).$$

Now by (3.111), we have that:

$$\mathsf{K} \models \big( (x \land y) \land z \big) \Leftrightarrow \big( x \land (y \land z) \big) \approx \mathbf{1},$$

so by (3.147) we have that:

$$\mathsf{K} \models (x \land y) \land z \approx x \land (y \land z).$$

Thus  $K \models (3.133)$ . Because of the axiomatisation of K by (3.111)-(3.132), a suitable modification of the preceding argument now shows  $K \models (3.134)$ -(3.145) for each of the remaining identities (3.134)-(3.145) defining lhSBIA<sup>+D</sup>. Hence lhSBIA<sup>+D</sup>  $\subseteq K$ .

Remark 3.3.73 (Added in proof). It is implicit in Blok and Pigozzi [36, Sections 3.4.5-3.4.7] and the constructive proof of Andréka, Kurucz, Németi and Sain [13, Theorem 3.2.3] that any axiomatisation of a K-1-regular quasivariety K with 1 can be translated into an axiomatisation of the inherent assertional logic S(K, 1) of K. In more detail, let K be a K-1-regular quasivariety (for some constant term 1) and let  $\Delta := \{\Delta_1, \ldots, \Delta_m\}$  be a set of binary terms witnessing the K-1-regularity of K in the sense of Czelakowski and Pigozzi [78, Theorem 2.3]. Let  $S_K$  be the 2-dimensional deductive system associated with K in the sense of Blok and Pigozzi [36, Section 3.4.7] and let:

 $au(p) := \{\langle p, \mathbf{1} \rangle\} \text{ and } 
ho(\langle p, q \rangle) := \Delta(p, q)$ 

be (1, 2)-translations and (2, 1)-translations in the sense of Blok and Pigozzi [36, Section 6.1]. Then the image  $\rho[Id \cup QId]$  of any axiomatisation  $Id \cup QId$  of K under  $\rho$  yields a deductive system S. Adjoin to S the further rule:

$$\Delta_1(p,1),\ldots,\Delta_m(p,1)\vdash_{\mathbf{S}} p \tag{3.148}$$

and denote the resulting ded ctive system by S'. Because of (3.148) and the description of  $\rho[Id \cup QId]$  is an interpretation of S' in S<sub>K</sub> and  $\rho$  is an interpretation of S<sub>K</sub> in S', so S' and S<sub>K</sub> are equivalent: see for instance Blok and Pigozzi [36, Section 6.1]. Therefore S' is algebraisable with equivalence formulas  $\Delta_1, \ldots, \Delta_m$  and defining equation  $p \approx 1$ . Let K' be the equivalent quasivariety semantics of S'. Then S' is S(K', 1). Moreover, because both K and K' satisfy the quasi-identities:

$$x \approx y \supset \Delta_1(x, y) \approx \mathbf{1} \& \ldots \& \Delta_m(x, y) \approx \mathbf{1}$$

350

and:

$$\Delta_1(x,y) \approx \mathbf{1} \& \ldots \& \Delta_m(x,y) \approx \mathbf{1} \supset x \approx y$$

one sees easily that K and K' coincide. Hence S' = S(K', 1) = S(K, 1). Theorem 3.3.70 and its counterpart Proposition 3.3.72 may be seen as a particular instance of this general argument, due in the form above to Professor James Raftery and the author [189].

**Theorem 3.3.74.** SBPC coincides with  $S(IhSBIA^{+D}, 1)$ , the assertional logic of the variety of double-pointed left handed skew Boolean  $\cap$ -algebras.

*Proof.* By Theorem 3.3.70 and Proposition 3.3.72, SBPC is algebraisable with equivalent algebraic semantics  $|hSB|A^{+D}$  and defining equation  $p \approx 1$ . But by Theorem 1.8.15,  $S(|hSB|A^{+D}, 1)$  is algebraisable with the same equivalent algebraic semantics and defining equation, so by Lemma 1.8.8, SBPC and  $S(|hSB|A^{+D}, 1)$  coincide.

In the sequel we continue to denote  $S(lhSBIA^{+D}, 1)$  by SBPC. Upon recalling from §3.3.59 that  $PD_0^+$  denotes the pure double-pointed discriminator variety, the following theorem may now be inferred from the preceding result and Theorem 3.3.66.

**Corollary 3.3.75.** SBPC is definitionally equivalent to  $S(PD_0^+, 0)$ , the assertional logic of the pure double-pointed discriminator variety.

Although SBPC is a deductive system in the sense of §1.2.9, its presentation  $Ax \cup Ir$  is too closely related to the underlying axiomatisation of its equivalent algebraic semantics to be a Hilbert-style axiomatisation in any familiar sense (for example, in the sense of Sundholm [213, Section 1]). In particular,  $Ax \cup Ir$  is not an axiomatisation of SBPC for which (MP) is the only (proper) rule of proof. We claim that there exists just such an axiomatisation of SBPC (in principle). Our observation to this effect results as an easy consequence of Theorem 3.3.77 below, which characterises the logical connectives of SBPC. But first, the following useful technical lemma, a version of which occurs in Bulman-Fleming and Werner [50]. See also Burris [53].

**Lemma 3.3.76.** (cf. [50, Lemma 1.3]; cf. [53, Proposition 2.3.10]) For any (double-pointed) skew Boolean  $\cap$ -algebra A and any  $a, b \in A$ , the following assertions hold:

1. 
$$\Theta^{\mathbf{A}}(a \wedge b \wedge a, 0) = \Theta^{\mathbf{A}}(a, 0) \wedge^{\mathbf{Con}\,\mathbf{A}} \Theta^{\mathbf{A}}(b, 0);$$

2. 
$$\Theta^{\mathbf{A}}(a \lor b \lor a, 0) = \Theta^{\mathbf{A}}(a, 0) \lor^{\mathbf{Con}\mathbf{A}} \Theta^{\mathbf{A}}(b, 0);$$

3. 
$$\Theta^{\mathbf{A}}(a \setminus b, 0) = \Theta^{\mathbf{A}}(a, 0) *^{\mathbf{Cp}\mathbf{A}} \Theta^{\mathbf{A}}(b, 0);$$

$$4. \ \Theta^{\mathbf{A}}(a\Delta b, 0) = \Theta^{\mathbf{A}}(a, b).$$

*Proof.* Let A be a (double-pointed) skew Boolean  $\cap$ -algebra and  $a, b \in A$ .

For (1), from Proposition 2.2.31 we have that  $\langle a \sqcap b \rangle_{\mathbf{A}} = \langle a \rangle_{\mathbf{A}} \cap \langle b \rangle_{\mathbf{A}}$ , just because the ideals of  $\mathbf{A}$  coincide with the ideals of its canonical implicative BCS-algebra polynomial reduct  $\langle A; \backslash, 0 \rangle$ . Since  $a \land b \land a = a \sqcap b$ (by (3.69)), we have that  $\langle a \land b \land a \rangle_{\mathbf{A}} = \langle a \rangle_{\mathbf{A}} \cap \langle b \rangle_{\mathbf{A}}$ . By normality of ideals,  $[0]_{\Theta^{\mathbf{A}}(a \land b \land a, 0)} = [0]_{\Theta^{\mathbf{A}}(a, 0)} \cap [0]_{\Theta^{\mathbf{A}}(b, 0)}$ , so by ideal determinacy,  $\Theta^{\mathbf{A}}(a \land b \land a)$  $a, 0) = \Theta^{\mathbf{A}}(a, 0) \cap \Theta^{\mathbf{A}}(b, 0)$ .

For (2), we first observe that  $x \lor y \lor x$  is a join generator term for SBA. Indeed, let:

$$\begin{aligned} x \sqcup y &:= x \lor y \lor x \\ r(x, y, z) &:= z \backslash (y \backslash x) \\ t(x, y, z) &:= y \land (z \backslash (x \backslash y)) \land y. \end{aligned}$$

By an easy inspection of the subdirectly irreducible skew Boolean algebras, we see that SBA satisfies the identities:

 $r(x, y, 0) \approx 0 \qquad t(x, y, 0) \approx 0$  $r(x, y, x \sqcup y) \approx x \qquad t(x, y, x \sqcup y) \approx y$  $0 \sqcup 0 \approx 0$ 

so  $x \lor y \lor x$  is a join generator term for SBA by Proposition 1.7.13. Because the ideals of any skew Boolean  $\cap$ -algebra coincide with the ideals of its

canonical skew Boolean algebra polynomial reduct,  $x \vee y \vee x$  is also a join generator term for SBIA. Hence  $\langle a \vee b \vee a \rangle_{\mathbf{A}} = \langle a \rangle_{\mathbf{A}} \vee^{\mathbf{I}(\mathbf{A})} \langle b \rangle_{\mathbf{A}}$ . By normality of ideals,  $[0]_{\Theta^{\mathbf{A}}(a \vee b \vee a, 0)} = [0]_{\Theta^{\mathbf{A}}(a, 0)} \vee^{\mathbf{I}(\mathbf{A})} [0]_{\Theta^{\mathbf{A}}(b, 0)}$ , so by ideal determinacy,  $\Theta^{\mathbf{A}}(a \vee b \vee a, 0) = \Theta^{\mathbf{A}}(a, 0) \vee^{\operatorname{Con} \mathbf{A}} \Theta^{\mathbf{A}}(b, 0)$ .

For (3), from Theorem 2.2.20(5) we have that  $\langle a \setminus b \rangle_{\mathbf{A}} = \langle a \rangle_{\mathbf{A}} * \langle b \rangle_{\mathbf{A}}$  (where \* denotes dual relative pseudocomplementation in the join semilattice  $\langle \operatorname{CI}(\mathbf{A}); \vee$ ,  $\langle 0 \rangle_{\mathbf{A}} \rangle$  of compact ideals of  $\mathbf{A}$ ), just because the ideals of  $\mathbf{A}$  coincide with the ideals of its canonical implicative BCS-algebra polynomial reduct  $\langle A; \setminus, 0 \rangle$ . By normality of ideals,  $[0]_{\Theta^{\mathbf{A}}(a \setminus b, 0)} = [0]_{\Theta^{\mathbf{A}}(a, 0)} *^{\operatorname{CI}(\mathbf{A})} [0]_{\Theta^{\mathbf{A}}(b, 0)}$ , so by subtractivity and Proposition 1.7.10,  $\Theta^{\mathbf{A}}(a \setminus b, 0) = \Theta^{\mathbf{A}}(a, 0) *^{\operatorname{Cp}\mathbf{A}} \Theta^{\mathbf{A}}(b, 0)$ .

For (4), recall  $a\Delta b = (a/b) \vee (b/a)$  for any skew Boolean  $\cap$ -algebra A and  $a, b \in A$ . Let  $\theta \in \text{Con } A$ . From Proposition 3.3.37, we have that  $a\Delta b = 0$  iff a = b. Applying this to the quotient algebra  $A/\theta$ , we infer that  $a\Delta b \equiv 0 \pmod{\theta}$  iff  $a \equiv b \pmod{\theta}$ , which implies that  $\Theta^{\mathbf{A}}(a\Delta b, 0) = \Theta^{\mathbf{A}}(a, b)$ .

**Theorem 3.3.77.** IhSBIA<sup>+</sup> is a WBSO<sup>#</sup> variety with:

- 1. Weak join  $x \wedge y \wedge x$ ;
- 2. Weak meet  $x \lor y \lor x$ ;
- 3. Subtractive weak relative pseudocomplementation  $x \setminus y$ ;
- 4. Gödel equivalence term  $x \Delta y$ .

Hence the following assertions hold concerning the logical connectives of SBPC:

- 1'.  $\land$  is a conjunction;
- $\mathcal{Z}$ .  $\lor$  is a disjunction;
- $\mathcal{G}. \rightarrow is a conditional;$
- 4'.  $\Leftrightarrow$  is a G-identity;
- 5'.  $\leftrightarrow$  is a biconditional;
- 6'.  $\neg$  is a weak negation.

*Proof.* The theorem follows easily from Lemma 3.3.76, Theorem 3.3.74 and Proposition 3.1.25.

**Corollary 3.3.78.** For any  $\mathcal{L}$ -formulas  $\varphi, \psi, \chi$ , the following formulas are theorems of SEPC:

$$\begin{split} \varphi \to (\psi \to \varphi) \\ (\varphi \to (\psi \to \chi)) \to ((\varphi \to \psi) \to (\varphi \to \chi)) \\ (\varphi \land \psi) \to \varphi \\ (\varphi \land \psi) \to \psi \\ (\chi \to \varphi) \to ((\chi \to \psi) \to (\chi \to (\varphi \land \psi))) \\ \varphi \to (\varphi \lor \psi) \\ \psi \to (\varphi \lor \psi) \\ (\varphi \to \chi) \to ((\psi \to \chi) \to ((\varphi \lor \psi) \to \chi)). \end{split}$$

Proof. The result follows from Theorem 3.3.77 and Wójcicki [238, Theorem 2.4.7].

Corollary 3.3.79 (Deduction-Detachment Theorem for SEPC). For all  $\Gamma \cup \{\varphi, \psi\} \subseteq \operatorname{Fm}_{\mathcal{L}}$ ,

 $\Gamma, \varphi \vdash_{\mathsf{SHPC}} \psi \quad iff \quad \Gamma \vdash_{\mathsf{SHPC}} \varphi \to \psi.$ 

**Remark 3.3.80.** By Corollary 3.3.79,  $\varphi(\vec{p}) \vdash_{\mathbb{SHPC}} \psi(\vec{p})$  iff  $\vdash_{\mathbb{SHPC}} \varphi(\vec{p}) \to \psi(\vec{p})$ iff  $|\mathsf{hSB}|\mathsf{A}^{+D} \models s(\vec{x}) \to t(\vec{x}) \approx 1$  iff  $s^{\mathbf{A}}(\vec{a}) \preceq_{\mathcal{D}} t^{\mathbf{A}}(\vec{a})$  for all  $\mathbf{A} \in |\mathsf{hSB}|\mathsf{A}^{+D}$ and  $\vec{a} \in A$ , where  $s(\vec{x}), t(\vec{x})$  are  $\mathcal{L}$ -terms in the individual variables  $\vec{x}$  identified respectively with  $\mathcal{L}$ -formulas  $\varphi(\vec{p}), \psi(\vec{p})$  in the propositional variables  $\vec{p}$ . Hence the consequence relation  $\vdash_{\mathbb{SHPC}}$  of  $\mathbb{SHPC}$  induces only a quasiordering on its underlying algebraic models. In general, therefore, it is not the case that  $\varphi(\vec{p}) \dashv_{\mathbb{SHPC}} \psi(\vec{p})$  iff  $|\mathsf{hSB}|\mathsf{A}^{+D} \models s(\vec{x}) \approx t(\vec{x})$ . Because of Font and Jansana [91, Proposition 2.43], this shows that  $\mathbb{SHPC}$  is not congruential in the sense of Rautenberg [197].

Apropos the claim prefixing Lemma 3.3.76, because SBPC has the DDT (witness  $x \to y$ ), any axiomatisation of SBPC (including  $Ax \cup Ir$ ) may be converted

354

into one in which (MP) (for  $\rightarrow$ ), viz.:

 $p,p \to q \vdash_{\mathtt{SRPC}} q$ 

is the only (proper) rule of inference. To see this, apply the DDT from left to right repeatedly on all inference rules of a given axiomatisation of SBPC until they are in the form of axioms. Then the axiomatisation obtained by adjoining (MP) (for  $\rightarrow$ ) as a (proper) rule of inference to the resulting collection of axioms is clearly an axiomatisation of SBPC. (More generally, we remark that if K is a quasivariety with 1 such that the assertional logic S(K, 1) has a DDT with deduction-detachment set  $\Sigma := \{\zeta_i(p,q) : i = 1, ..., n\}$ , then any axiomatisation of S(K, 1) may be converted into one in which:

 $\varphi, \{\Sigma_i(\varphi, \psi) : i = 1, \ldots, n\} \vdash_{\mathbb{S}} \psi$ 

is the only (proper) rule of inference [189].) Hence there exists (in principle) an axiomatisation of SBPC for which (MP) (for  $\rightarrow$ ) is the only (proper) rule of inference.

By the preceding observation, a direct demonstration that (MP) (for  $\rightarrow$ ) is a derived rule of proof for SBPC is of some independent interest. In the final result of this subsection, we provide just such a direct proof of the detachment property (for  $\rightarrow$ ).

**Lemma 3.3.81.** For any *L*-formula  $\varphi$ , the following formula is a theorem of SBPC:

$$(1 \Rightarrow \varphi) \Leftrightarrow \varphi \tag{3.149}$$

Proof.

- (S10),  $p := \varphi, q := \varphi$ (1)  $(\varphi \Rightarrow (\varphi \lor \varphi)) \Leftrightarrow 1$
- (2)  $\varphi \Leftrightarrow \varphi$
- (3)  $((\varphi \Rightarrow (\varphi \lor \varphi)) \Rightarrow \varphi) \Leftrightarrow (\mathbf{1} \Rightarrow \varphi)$
- (4)  $(\mathbf{1} \Rightarrow \varphi) \Leftrightarrow ((\varphi \Rightarrow (\varphi \lor \varphi)) \Rightarrow \varphi)$

- $(\mathbf{R})$  $(1), (2), (CP \rightarrow)$

(3), (S)

| 3.3. | Pre-BCK | Quasilattices | and BCK | Paralattices |  |
|------|---------|---------------|---------|--------------|--|
|      |         |               |         |              |  |

(5)  $((\varphi \Rightarrow (\varphi \lor \varphi)) \Rightarrow \varphi) \Leftrightarrow \varphi$  (S7),  $p := \varphi, q := \varphi \lor \varphi$ (6)  $(1 \Rightarrow \varphi) \Leftrightarrow \varphi$  (4), (5), (T)

**Proposition 3.3.82** (Modus Ponens for  $\Rightarrow$ ). For all *L*-formulas  $\varphi, \psi$ ,

 $\varphi,\varphi \Rightarrow \psi \vdash_{\mathbb{S}\mathbb{H}^{\mathbb{C}}} \psi.$ 

Proof.

| (1) | arphi   | Assumption                     |
|-----|---|--------------------------------|
| (2) | $\varphi \Leftrightarrow 1$                                       | (1), ( <b>1</b> -I)            |
| (3) | $\psi \Leftrightarrow \psi$                                       | (R)                            |
| (4) | $(\varphi \Rightarrow \psi) \Leftrightarrow (1 \Rightarrow \psi)$ | (2), (3), (CP- $\Rightarrow$ ) |
| (5) | $(1\Rightarrow\psi)\Leftrightarrow\psi$                           | by (3.149)                     |
| (6) | $(\varphi \Rightarrow \psi) \Leftrightarrow \psi$                 | (4), (5), (T)                  |
| (7) | $\psi \Leftrightarrow (arphi \Rightarrow \psi)$                   | (6), (S)                       |
| (8) | $arphi \Rightarrow \psi$  | Assumption                     |
| (9) | $(arphi \Rightarrow \psi) \Leftrightarrow 1$                      | (8), ( <b>1</b> -I)            |
| 10) | $\psi \Leftrightarrow 1$  | (7), (9), (T)                  |
| 11) | $\psi$  | (10), ( <b>1</b> -E).          |
|     |   |                                |

**Lemma 3.3.83.** For any  $\mathcal{L}$ -formula  $\varphi$ , the following formulas are theorems of SBPC:

| $(\varphi \land \varphi) \Leftrightarrow \varphi$ | (3.150) |
|---|---------|
| $(\varphi \lor \varphi) \Leftrightarrow \varphi$  | (3.151) |
| $\varphi \Rightarrow \varphi.$                    | (3.152) |

356

*Proof.* Let  $\varphi, \psi$  be  $\mathcal{L}$ -formulas. For (3.150), we have the following derivation:

(1)  $((\varphi \Rightarrow \psi) \Rightarrow \varphi) \Leftrightarrow \varphi$ (2)  $\varphi \Leftrightarrow \varphi$ (3)  $(((\varphi \Rightarrow \psi) \Rightarrow \varphi) \land \varphi) \Leftrightarrow (\varphi \land \varphi)$ (4)  $(\varphi \land \varphi) \Leftrightarrow (((\varphi \Rightarrow \psi) \Rightarrow \varphi) \land \varphi)$ (5)  $(((\varphi \Rightarrow \psi) \Rightarrow \varphi) \land \varphi) \Leftrightarrow \varphi$ (6)  $(\varphi \land \varphi) \Leftrightarrow \varphi$ 

For (3.151), we have the following derivation:

| (1) | $(\varphi \land \varphi) \Leftrightarrow \varphi$   | by $(3.150)$                          |
|-----|---|---------------------------------------|
| (2) | $\varphi \Leftrightarrow \varphi$   | (R)                                   |
| (3) | $((\varphi \land \varphi) \lor \varphi) \Leftrightarrow (\varphi \lor \varphi)$           | (1), (2), (CP- $\lor$ )               |
| (4) | $(\varphi \lor \varphi) \Leftrightarrow \bigl((\varphi \land \varphi) \lor \varphi\bigr)$ | (3), (S)                              |
| (5) | $((\varphi \land \varphi) \lor \varphi) \Leftrightarrow \varphi$                          | (S4), $q := \varphi$ , $p := \varphi$ |
| (6) | $(\omega \lor \omega) \Leftrightarrow \omega$   | (4), (5), (T).                        |

For (3.152), we have the following derivation:

| (1) | $\varphi \Leftrightarrow \varphi$  | (R)                                 |
|-----|--|-------------------------------------|
| (2) | $(\varphi \lor \varphi) \Leftrightarrow \varphi$   | by (3.151)                          |
| (3) | $(\varphi \Rightarrow (\varphi \lor \varphi)) \Leftrightarrow (\varphi \Rightarrow \varphi)$ | (1), (2), (CP- $\Rightarrow$ )      |
| (4) | $(\varphi \Rightarrow \varphi) \Leftrightarrow (\varphi \Rightarrow (\varphi \lor \varphi))$ | (3), (S)                            |
| (5) | $ig(arphi \Rightarrow (arphi \lor arphi)ig) \Leftrightarrow 1$                               | (S10), $p := \varphi, q := \varphi$ |
| (6) | $(arphi \Rightarrow arphi) \Leftrightarrow 1$  | (4), (5), (T)                       |
| (7) | $\varphi \Rightarrow \varphi$  | (6), ( <b>1</b> -E).                |

**Lemma 3.3.84.** For any *L*-formulas  $\varphi, \psi$ , the following formulas are theorems

(S7),  $p := \varphi, q := \psi$ 

(S12),  $q := \varphi \Rightarrow \psi, p := \varphi$ 

(1), (2), (CP-A)

(4), (5), (T).

(R)

(3), (S)

of SBPC:

$$(\mathbf{1} \land \varphi) \Leftrightarrow \varphi$$
 (3.153)  
$$(\varphi \lor (\varphi \land \psi)) \Leftrightarrow \varphi$$
 (3.154)

$$1 \vee (a$$

$$(3.155)$$

$$\mathbf{1} \lor \varphi. \tag{3.155}$$

*Proof.* Let  $\varphi, \psi, \chi$  be  $\mathcal{L}$ -formulas. For (3.153) we have the following derivation:

| (1) | $\varphi \Rightarrow \varphi$   | by (3.152)                             |
|-----|---|--|
| (2) | $(arphi \Rightarrow arphi) \Leftrightarrow 1$   | (1), ( <b>1</b> -I)                    |
| (3) | $\varphi \Leftrightarrow \varphi$   | (R)                                    |
| (4) | $((\varphi \Rightarrow \varphi) \land \varphi) \Leftrightarrow (1 \land \varphi)$         | (2), (3), (CP-∧)                       |
| (5) | $(1 \land \varphi) \Leftrightarrow \big((\varphi \Rightarrow \varphi) \land \varphi\big)$ | (4), (S)                               |
| (6) | $((\varphi \Rightarrow \varphi) \land \varphi) \Leftrightarrow \varphi$                   | (S12), $q := \varphi$ , $p := \varphi$ |
| (7) | $(1\wedge\varphi)\Leftrightarrow\varphi$  | (5), (6), (T).                         |

For (3.154) we have the following derivation:

For (3.155) we have the following derivation:

(1)  $\mathbf{1} \Leftrightarrow \mathbf{1}$  (R)

| (2) | $(1\wedgearphi) \Leftrightarrow arphi$                        | by (3.153)       |
|-----|---|------------------|
| (3) | $(1 \lor (1 \land \varphi)) \Leftrightarrow (1 \lor \varphi)$ | (2), (3), (CP-V) |
| (4) | $(1 \lor arphi) \Leftrightarrow ig(1 \lor (1 \land arphi)ig)$ | (3), (S)         |
| (5) | $ig(1ee(1\wedgearphi)ig)\Leftrightarrow1$                     | by (3.154)       |
| (6) | $(1 \lor \varphi) \Leftrightarrow 1$                          | (4), (5), (T)    |
| (7) | $1 \lor \varphi$  | (6), (1-E).      |

**Proposition 3.3.85** (Modus Ponens for  $\rightarrow$ ). For all  $\mathcal{L}$ -formulas  $\varphi, \psi$ ,

 $\varphi, \varphi \to \psi \vdash_{\text{SEPC}} \psi.$ 

*Proof.* By the definition of  $\rightarrow$  and Proposition 3.3.82, to prove the proposition we need only show  $\varphi \vdash_{SEPC} \varphi \lor \psi$  for all  $\mathcal{L}$ -formulas  $\varphi, \psi$ . For this we have:

| (1) | arphi   | Assumption              |
|-----|---|-------------------------|
| (2) | $\varphi \Leftrightarrow 1$                       | (1), ( <b>1</b> -I)     |
| (3) | $\psi \Leftrightarrow \psi$                       | (R)                     |
| (4) | $(arphi \lor \psi) \Leftrightarrow (1 \lor \psi)$ | (2), (3), (CP- $\vee$ ) |
| (5) | $1 \lor \psi$                                     | by (3.155)              |
| (6) | $(1 ee \psi) \Leftrightarrow 1$                   | (5), ( <b>1</b> -I)     |
| (7) | $(arphi \lor \psi) \Leftrightarrow 1$             | (4), (6), (T)           |
| (8) | $\varphi \lor \psi$                               | (7), ( <b>1-E</b> ).    |

The results of this subsection notwithstanding, we have been unable to obtain a Hilbert-style axiomatisation of SBPC in which (MP) (for  $\rightarrow$ ) is the only (proper) rule of inference that is 'aesthetically pleasing' in the sense that its axioms have a familiar description. In particular, we have been unable to provide such an axiomatisation of SBPC whose axioms are based on the theorems (or some variants of the theorems) of Corollary 3.3.78. Hence we conclude this subsection with the following problem.

**Problem 3.3.86.** Give an 'aesthetically pleasing' Hilbert-style axiomatisation of SEPC for which the only rule of inference is (MP) (for  $\rightarrow$ ):  $p, p \rightarrow q \vdash_{\text{SEPC}} q$ .

# Chapter 4

# Conclusion

# 4.1 Summary

Motivated by work of Blok and Raftery [38, Section 4] and Agliano and Ursini [11, Example 3.7, Corollary 3.8], this thesis witnessed the introduction of pre-BCK-algebras as a generalisation of BCK-algebras to the subtractive but not BCK-0-regular case. In particular, this dissertation oversaw the investigation of the elementary theory of the variety of pre-BCK-algebras and some of its subvarieties, and the application of this theory to the study of some varieties arising naturally in universal algebra and algebraic logic.

Chapter 1 provided a structured account of the theory relevant to the study of pre-BCK-algebras, including: Laslo and Leech's theory of quasilattices, paralattices and skew lattices; Blok and Pigozzi's hierarchy of varieties with EDPC; the theory of BCK-algebras and BCK-lattices due to Iséki, Idziak and others; Agliano and Ursini's theory of ideals and subtractive varieties; and the theory of algebraisable and assertional logics due to Blok, Pigozzi, Raftery and others. The main new results concerned distributivity in skew lattices. By the results of  $\S1.4.18$ , the equivalence of the middle distributive identities (1.19)-(1.20) for lattices extends to symmetric skew lattices, but not to skew lattices. The results provide support for Leech's contention (initially prompted by the remarks of  $\S1.4.15$ ) that '...symmetric skew lattices are the really *nice* skew lattices' [152, p. 17].

# 4.1. Summary

In Chapter 2 the elementary theory of the variety of pre-BCK-algebras and some of its subvarieties was investigated. In Section 2.1 pre-BCK-algebras proper were considered. The results obtained show that a significant fragment of the first-order theory of BCK-algebras extends to pre-BCK-algebras; in particular, the ideal theory of BCK-algebras carries over to pre-BCK-algebras almost in its entirety. In Section 2.2 varieties of pre-BCK-algebras were studied. It was shown that the correspondence between the theory of BCK-algebras and the theory of pre-BCK-algebras exhibited in Section 2.1 extends to subvarieties of BCK-algebras, inasmuch as with any variety V of BCK-algebras there may be associated a variety  $V_{3}$  of pre-BCK-algebras ('the natural pre-BCK-algebraic counterpart of V') such that  $V_3$  enjoys many of the same (firstorder) properties as V. In Section 2.3 the variety of implicative BCS-algebras was investigated. Although the variety of implicative BCS-algebras fails to enjoy many of the properties typically associated with a 'tractable' class of algebras, the results of Section 2.3 nonetheless provide for these algebras a fairly complete elementary theory closely resembling that of implicative BCKalgebras. Collectively, the results of Chapter 2 indicate that pre-BCK-algebras enjoy a coherent elementary theory that largely parallels the theory of BCKalgebras. This suggests that pre-BCK-algebras are an appropriate generalisation of BCK-algebras to the subtractive but not BCK-0-regular case.

In Chapter 3 the theory of pre-BCK-algebras was applied to the study of some varieties arising naturally in universal algebra and algebraic logic. In Section 3.1 positive implicative pre-BCK-algebras in subtractive varieties with EDPI were considered. The results obtained show in particular that the study of ideals in subtractive varieties with EDPI reduces to the study of ideals in positive implicative pre-BCK-algebras. In Section 3.2 connections between implicative BCS-algebras and binary discriminator varieties were investigated. It was shown that the variety of implicative BCS-algebras provides a convenient framework for the study of binary discriminator varieties, and this observation was exploited in clarifying relationships between binary discriminator, pointed fixedpoint discriminator and pointed ternary discriminator varieties. In Section 3.3 varieties  $PQ_c$ ,  $BP_c$  of pre-BCK/BCK-algebras structurally enriched with band [double band] operations were studied as a generalisation of Idziak's

362

varieties of lower/upper BCK-semilattices [BCK-lattices]. It was shown that the theory of the varieties  $PQ_C$  and  $BP_C$  encompasses the theory of skew Boolean algebras/skew Boolean  $\cap$ -algebras, and hence (to within termwise definitional equivalence) subsumes the theory of pointed discriminator varieties. Further, the results hint that the theory of the varieties  $BP_C$  accommodates the theory of pointed fixedpoint discriminator varieties (to within termwise definitional equivalence). Collectively, the results of Chapter 3 indicate that (structurally enriched) pre-BCK-algebras (distinct from BCK-algebras) arise naturally in and simplify the study of several classes of varieties found in universal algebra and algebraic logic. This suggests that classes of (structurally enriched) pre-BCK-algebras may provide a unifying framework simplifying the study of several hitherto unrelated areas of universal algebraic logic.

As a generalisation of BCK-algebras to the subtractive but not BCK-0-regular case, the work of this thesis thus attests that pre-BCK-algebras are of interest both in their own right and in their application to the study of varieties arising naturally in universal algebra and algebraic logic. Pre-BCK-algebras are of interest in their own right, inasmuch as they are a natural and coherent generalisation of BCK-algebras to the subtractive but not BCK-0-regular case; and their applications to universal algebraic logic are of interest, insofar as such algebras provide a unifying framework simplifying the study of several important classes of varieties occurring naturally in universal algebra and algebraic logic.

# 4.2 Future Work

The remarks of Section 4.1 clearly call for a further study of pre-BCK-algebras and cf their application to universal algebra and algebraic logic. In what follows we present a brief selection of problems outlining some possibilities for future research.

**4.2.1. Pre-BCK-Algebras.** By the remarks of §1.1.1, residuated structures play a central role in the algebraic study of logical systems. The residuated structures most commonly encountered in algebraic logic are (left) residuated structures.

uated partially ordered groupoids, or structures of the form  $\langle A; \oplus; \leq \rangle$  where: (i)  $\leq$  is a partial order on A; (ii)  $\oplus$  is a binary operation on A isotone in each of its positions; and (iii) the equivalence  $a \leq c \oplus b$  iff  $a \div b \leq c$  is satisfied for any  $a, b, c \in A$ . Pocrims and polrims provide natural examples of (left) residuated partially ordered groupoids. On the other hand, members of pointed discriminator varieties, WBSO<sup>#</sup> varieties and double-pointed binary discriminator varieties provide important examples of algebras that do not in general enjoy the underlying structure of a residuated partially ordered groupoid. Instead, such algebras have the underlying structure of a (left) residuated quasiordered groupoid, where a (left) residuated quasiordered groupoid is a structure of the form  $\langle A; \oplus; \preceq \rangle$  such that: (i)  $\preceq$  is a quasiorder on A; (ii)  $\oplus$  is a binary operation on A that is isotone in each of its positions; and (iii) the equivalence  $a \preceq c \oplus b$  iff  $a \div b \preceq c$  is satisfied for any  $a, b, c \in A$ . Indeed, let V be a pointed discriminator variety, WBSO<sup>#</sup> variety or double-pointed binary discriminator variety and let  $\sqcup$  be a join generator term for V. For any algebra A in V, the structure  $\langle A; \sqcup^{\mathbf{A}}; \preceq \rangle$  is a (left) residuated quasiordered groupoid, where  $\preceq$ denotes the underlying quasiorder of the canonical MINI-algebra polynomial reduct of A (which must exist as V has EDPI). Inasmuch as the results of Section 2.1 lend to the conjecture that the class of residuated quasiordered groupoids may admit a coherent elementary theory closely resembling that of the class of residuated ordered groupoids, the preceding remarks call attention to the following problem.

**Problem 4.2.2.** Investigate the class of residuated quasiordered groupoids.

Considered as residuated partially ordered groupoids, pocrims (or more generally polrims) are of particular interest in algebraic logic because they are amenable to purely algebraic investigation, since any such structure  $\langle A; \oplus, 0; \leq \rangle$  satisfies  $a \leq b$  iff a - b = 0 for any  $a, b \in A$ , whence the partial order  $\leq$  may be completely recovered from the residuation operation  $\div$ . These remarks suggest that attention be focussed on residuated quasiordered groupoids that (in some sense) naturally generalise pocrims, and in fact at least one such class of residuated quasiordered groupoids has already been considered in the

literature. Call an algebra  $\mathbf{A} := \langle A; \oplus, \div, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  a pre-pocrim if: (i) the reduct  $\langle A; \oplus, 0 \rangle$  is a commutative monoid; (ii) the reduct  $\langle A; \div, 0 \rangle$  is a pre-BCK-algebra; and (iii) **A** satisfies the following identities:

$$((x - y) - z) - (x - (y \oplus z)) \approx \mathbf{0} (x - (y \oplus z)) - ((x - y) - z) \approx \mathbf{0}.$$

Clearly any pre-pocrim  $\langle A; \oplus, \div, 0 \rangle$  has the underlying structure of a residuated quasiordered groupoid  $\langle A; \oplus; \preceq \rangle$ , where  $\preceq$  denotes the underlying quasiordering of the pre-BCK-algebra reduct  $\langle A; -, 0 \rangle$ . Pre-pocrims were introduced by Higgs in [109] in connection with his example showing that the class of all pocrims is not a variety. By [109, pp. 72–73] pre-pocrims are known to preserve several important properties of pre-BCK-algebras: in particular, for any pre-pocrim A, the relation  $\Xi$  of Theorem 2.1.14 is a congruence on A. such that  $\mathbf{A}/\boldsymbol{\Xi}$  is the maximal pocrim homomorphic image of  $\mathbf{A}$ . Nonetheless, it is unclear if pre-pocrims (considered as residuated quasiordered groupoids) are the most appropriate generalisation of pocrims (understood as residuated partially ordered groupoids). For let  $\mathbf{A}$  be a pseudocomplemented semilattice. By previous remarks, A has the underlying structure of a residuated quasiordered groupoid  $\langle A; \sqcup^{\mathbf{A}}; \preceq \rangle$ , where  $\sqcup$  is the join generator term of Proposition 2.3.60(2) and  $\leq$  denotes the underlying quasiorder of the canonical implicative BCS-algebra polynomial reduct  $\langle A; \langle 0 \rangle$  of A. However, the polynomial reduct  $\langle A; \sqcup^{\mathbf{A}}, \backslash, 0 \rangle$  is not a pre-pocrim, since  $\mathbf{A} \not\models x \sqcup \mathbf{0} \approx x$ .

**Problem 4.2.3.** Identify an appropriate generalisation of pocrims (considered as residuated partially ordered groupoids) to residuated quasiordered groupoids. Does the quasivariety of pocrims admit a generalisation to a class of residuated quasiordered groupoids analogous to that of BCK-algebras to pre-BCK-algebras?

Recall from §1.1.1 that the quasivariety of BCK-algebras is precisely the class of all  $\langle \div, 0 \rangle$ -subreducts of pocrims.

**Problem 4.2.4.** Is an algebra  $\langle A; \div, 0 \rangle$  of type  $\langle 2, 0 \rangle$  a pre-BCK-algebra iff it is a  $\langle \div, 0 \rangle$ -subreduct of a residuated quasiordered groupoid in the sense of

# Problem 4.2.3?

Residuated quasiordered groupoids may not be the only generalisation of pre-BCK-algebras of interest in algebraic logic. Indeed, the results of Section 2.1, in conjunction with work due to van Alten [229], suggest that the varietal closure H(LR) of the quasivariety LR of left residuation algebras is of interest in algebraic logic as a naturally occurring generalisation of the variety of pre-BCK-algebras (recall Proposition 2.1.11). In the statement of the following problem, an algebra  $A := \langle A; -, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is called an H(LR)-algebra if  $A \in H(LR)$ .

**Problem 4.2.5.** Investigate the variety H(LR). Do H(LR)-algebras stand in relation to pre-BCK-algebras as left residuation algebras stand in relation to BCK-algebras?

4.2.6. Varieties of Pre-BCK-Algebras. In Section 2.2 the varieties of commutative, positive implicative and implicative pre-BCK-algebras were studied as the pre-BCK-algebraic counterparts of the varieties of commutative, positive implicative and implicative BCK-algebras respectively. In light of the results of Section 2.3, Section 3.1 and Section 3.2, it is natural to anticipate that the pre-BCK-algebraic counterparts of other naturally occurring varieties of BCK-algebras may themselves be of interest in universal algebra and algebraic logic. Hence we posit:

**Problem 4.2.7.** Investigate those varieties V<sub>2</sub> of pre-BCK-algebras arising as the pre-BCK-algebraic counterpart of a naturally occurring variety V of BCK-algebras. In particular, investigate the natural pre-BCK-algebraic counterparts of the varieties  $e_nBCK$ ,  $n \in \omega$ , and the natural pre-BCK-algebraic counterpart of the variety HBCK [38, Theorem 3.15] of all residuation subreducts of hoops.

Apropos the preceding problem, the remarks of §1.5.1 and the results of Section 3.1 collectively indicate that variaties of positive implicative pre-BCKalgebras generalising the variety of positive implicative BCK-algebras may be of particular interest in universal algebraic logic. Let K denote the class of all positive implicative pre-BCK-algebras A with a left normal band with zero

polynomial reduct  $\langle A; \sqcap_1, 0 \rangle$ , where  $a \sqcap_1 b := (a(ab))(ba)$  for any  $a, b \in A$ , such that the underlying natural band partial order  $\leq_{\mathcal{H}}^{\langle A; \sqcap_1, 0 \rangle}$  respects positive implicative pre-BCK difference. Members of the class K more closely resemble positive implicative BCK-algebras than do positive implicative pre-BCK-algebras, inasmuch as any positive implicative BCK-algebra A has a semi-Boolean algebra polynomial reduct  $\langle A; \cap_1 \rangle$ , where  $a \cap_1 b := (a(ab))(ba)$ for any  $a, b \in A$  (recall the remarks prior to Problem 2.3.16). Thus the following problem would seem relevant:

**Problem 4.2.8.** Characterise the class K. Is it a variety? What role, if any, does K play in the theory of pre-BCK-algebras? In algebraic logic?

**4.2.9.** Implicative BCS-Algebras. Recall from Theorem 2.3.29 that implicative BCS-algebras are precisely those implicative pre-BCK-algebras **A** for which the polynomial reduct  $\langle A; \sqcap, 0 \rangle$  is a left normal band with zero whose underlying natural band partial order respects implicative pre-BCK difference. This result warrants the study of bands in pre-BCK-algebras, and in particular the study of those pre-BCK-algebras **A** for which the polynomial reduct  $\langle A; \sqcap, 0 \rangle$  is a (normal) band with zero. As we know of no such studies in the existing literature concerning (normal) bands in BCK-algebras, the preceding remarks call particularly for a study of bands arising in BCK-algebras.

**Problem 4.2.10.** Investigate (normal) bands in pre-BCK-algebras. In particular, investigate (normal) bands in BCK-algebras. Is the class of all BCKalgebras  $\langle A; -, 0 \rangle$  such that the polynomial reduct  $\langle A; \cap \rangle$  is a (normal) band equationally definable?

By Theorem 2.3.73, the 3-element flat implicative BCS-algebra  $\mathbf{B}_2$  generates the class iBCS of implicative BCS-algebras as a variety. However, by Proposition 2.3.76, the quasivariety  $\mathbf{Q}(\mathbf{B}_2)$  generated by  $\mathbf{B}_2$  is not a variety, whence the inclusion  $\mathbf{Q}(\mathbf{B}_2) \subseteq \mathbf{V}(\mathbf{B}_2)$  is strict. In consequence, it is natural to focus attention on the quasivariety  $\mathbf{Q}(\mathbf{B}_2)$  as a specialisation of iBCS. In particular, it is natural to ask to what extent membership of the quasivariety  $\mathbf{Q}(\mathbf{B}_2)$  is reflected in special properties of implicative BCS-algebras. In the statement of the following problem and in the sequel, an algebra  $\mathbf{A} := \langle A; -, 0 \rangle$  of type  $\langle 2, 0 \rangle$  is called a  $\mathbf{Q}(\mathbf{B}_2)$ -algebra if  $\mathbf{A} \in \mathbf{Q}(\mathbf{B}_2)$ . **Problem 4.2.11.** Characterise the  $Q(B_2)$ -algebras amongst the implicative BCS-algebras.

The quasivariety  $\mathbf{Q}(\mathbf{B}_2)$  also seems of intrinsic interest in its own right, in view of the pivotal role it plays in the theory of BCK-algebras (recall Proposition 2.2.5). Two of the most pressing problems concerning  $\mathbf{Q}(\mathbf{B}_2)$  are the following:

**Problem 4.2.12.** Give an implicational characterisation of the quasivariety  $Q(B_2)$ . Is  $Q(B_2)$  finitely axiomatisable?

**Problem 4.2.13.** Characterise the  $Q(B_2)$ -subdirectly irreducible  $Q(B_2)$ -algebras. To within isomorphism, are  $C_1, B_2$  the only  $Q(B_2)$ -subdirectly irreducible  $Q(B_2)$ -algebras?

In Problem 2.3.69 we asked if every implicative BCS-algebra arises as a 'residuation subreduct' of a pseudocomplemented semilattice. It would also be of interest to possess an embedding theorem for  $Q(B_2)$ -algebras. Upon recalling that the class of  $\langle \rangle, 0 \rangle$ -reducts of members of SBA<sub>SI</sub> is, to within isomorphism, precisely {C<sub>1</sub>, B<sub>2</sub>}, the following problem is suggested by Problem 4.2.13 and Kalman's proof of Theorem 1.6.20.

**Problem 4.2.14.** Is an algebra  $\langle A; \rangle, 0 \rangle$  of type  $\langle 2, 0 \rangle$  a  $\mathbf{Q}(\mathbf{B}_2)$ -algebra iff it is a  $\langle \rangle, 0 \rangle$ -subreduct of a skew Boolean algebra?

By Theorem 2.3.75, iBCS is a cover of iBCK in  $\Lambda^{V}(\text{PBCK})$ , the lattice of varieties of pre-BCK-algebras. However, by Proposition 2.3.76 this result does not extend to  $\Lambda^{Q}(\text{PBCK})$ , the lattice of quasivarieties of pre-BCK-algebras. Therefore we pose the following problem:

**Problem 4.2.15.** Is  $Q(B_2)$  a (unique) cover of the atom iBCK in  $\Lambda^Q(PBCK)$ ?

4.2.16. Subtractive Varieties with EDPI. Corollary 3.1.7 shows that the study of the ideal theory of subtractive varieties with EDPI may be reduced to the study of the ideal theory of MINI-algebras (or equivalently, positive implicative pre-BCK-algebras) inasmuch as a variety V with 1 is subtractive

with EDPI (witness  $y \to x$ ) iff every  $\mathbf{A} \in V$  has a MINI-algebra polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  whose MINI-ideals coincide with the V-ideals of  $\mathbf{A}$ . Because any dual Brouwerian semilattice  $\mathbf{A} := \langle A; \vee, *, 0 \rangle$  is termwise definitionally equivalent to a generalised Boolean algebra iff  $\mathbf{A} \models x * (y * x) \approx x$  (recall the remarks of §1.3.4), the preceding observation calls for a study of the role played by MINI-algebras satisfying  $((x \to y) \to x) \to x \approx 1$  in the theory of subtractive varieties with EDPI; see also Blok and Pigozzi [30, Corollary 4.3]. In the statement of the problem below, a MINI-algebra  $\mathbf{A}$  is said to be *classical* if  $\mathbf{A} \models ((x \to y) \to x) \to x \approx 1$ ; clearly any classical MINI-algebra is term equivalent to (in fact, is dually isomorphic to) an implicative pre-BCK-algebra by Theorem 3.1.4.

**Problem 4.2.17.** Investigate the role played by classical MINI-algebras in the theory of subtractive varieties with EDPI. If V is a subtractive variety with EDPI (witness  $y \to x$ ), does it follow that the polynomial reduct  $\langle A; \to^{\mathbf{A}}, 1 \rangle$  is a classical MINI-algebra iff for any  $\mathbf{A} \in V$ , the join semilattice  $\langle \operatorname{CI}(\mathbf{A}); \vee, \langle 0 \rangle_{\mathbf{A}} \rangle$  of compact ideals of  $\mathbf{A}$  is dually relatively complemented?

By the pure [subtractive] WBSO variety we mean the variety generated by the class of all algebras  $\mathbf{A} := \langle A; \cdot, \rightarrow, \Delta, 1 \rangle$  of type  $\langle 2, 2, 2, 1 \rangle$  where the join semilattice  $\langle \operatorname{Cp} \mathbf{A}; \vee, \omega_{\mathbf{A}} \rangle$  of compact congruences is dually relatively pseudocomplemented and  $\cdot, \rightarrow$  and  $\Delta$  are weak meet, [subtractive] weak relative pseudocomplemention and Gödel equivalence terms respectively. The pure [subtractive] WBSO variety may be of interest in algebraic logic inasmuch as it may provide a convenient framework for the study of [subtractive] WBSO varieties, similar to the manner in which skew Boolean  $\cap$ -algebras provide a point of reference for the study of pointed ternary discriminator varieties (recall Theorem 1.4.39) and implicative BCS-algebras provide a point of reference for the study of binary discriminator varieties (recall Theorem 3.2.7).

**Problem 4.2.18.** Investigate the pure [subtractive] WBSO variety.

In [34, p. 549] Blok and Pigozzi note that with some important exceptions, the varieties of traditional algebraic logic all have (commutative) ' $\Gamma$ D terms.

Included among these classes are the varieties of dual Brouwerian semilattices and positive implicative BCK-algebras, which have the (commutative) TD term  $z \div ((x \div y) \div (y \div x))$  (by [34, Corollary 5.2(ii)]), and the variety of Nelson algebras, which has the (commutative) TD term  $((x \Rightarrow y) \rightarrow (y \Rightarrow x)) \rightarrow z$  (by Theorem 3.1.33). Upon observing that the (commutative) TD terms for these varieties are all of the form  $((\cdots (d_1(x, y) \rightarrow d_2(x, y)) \rightarrow \cdots) \rightarrow d_n(x, y)) \rightarrow z$ , where  $d_1(x, y), \ldots, d_n(x, y)$  are binary terms witnessing point regularity in the sense of Proposition 1.2.6 and  $\rightarrow$  is a subtractive weak relative pseudocomplementation, the following problem suggests itself.

**Problem 4.2.19.** Investigate the class of all varieties with a (commutative) TD of the form  $((\cdots(d_1(x, y) \rightarrow d_2(x, y)) \rightarrow \cdots) \rightarrow d_n(x, y)) \rightarrow z$ , where  $d_1(x, y), \ldots, d_n(x, y)$  are binary terms witnessing point regularity in the sense of Proposition 1.2.6 and  $\rightarrow$  is a (subtractive) weak relative pseudocomplementation. In particular, investigate the class of all varieties with a (commutative) TD term of the form  $(x\Delta y) \rightarrow z$ , where  $\Delta$  and  $\rightarrow$  are a Gödel equivalence term and a (subtractive) weak relative pseudocomplementation respectively.

4.2.20. Binary Discriminator Varieties. By Theorem 3.2.29, a variety with 0 is a pointed fixedpoint discriminator variety iff it is a 0-regular binary discriminator variety. From this result it follows that the binary discriminator may be legitimately considered a generalisation of the pointed fixedpoint discriminator to the subtractive (but not ideal determined) case. Inasmuch as the theory of the fixedpoint discriminator closely parallels that of the ternary discriminator [34, p. 548; Section 3], the preceding observation gives rise to the following problem:

**Problem 4.2.21.** To what extent does the theory of the binary discriminator parallel the theory of the (pointed) fixedpoint discriminator? Of the ternary discriminator?

Apropos the preceding problem, a suitable representation theorem would help clarify connections between binary discriminator, pointed fixedpoint discriminator and ternary discriminator varieties. In particular, the following problem

suggests itself, just because binary discriminator varieties have a well developed theory of prime ideals (by Theorem 3.2.8 and the results of §2.2.28).

**Problem 4.2.22.** Is there a sheaf-theoretic representation (in the sense of Burris and Werner [56]) of binary discriminator varieties?

In Remark 3.2.19, it was noted that the class of Abelian Rickart semirings is a binary discriminator variety. This observation seems to be of some interest, since it implies several classes of rings arising in real and complex analysis are binary discriminator varieties.

**Problem 4.2.23.** Investigate the class of Abelian Rickart semirings (considered as a binary discriminator variety).

By Example 2.3.12, another important example of a naturally occurring binary discriminator variety is the class of pseudocomplemented semilattices. Call an algebra  $\langle A; \wedge, \rangle, 0 \rangle$  of type  $\langle 2, 2, 0 \rangle$  a locally pseudocomplemented semilattice if: (i) the reduct  $\langle A; \wedge, 0 \rangle$  is a meet semilattice with zero; and (ii) for all  $a, b \in A$ , the difference  $a \setminus b$  is the pseudocomplement of b in the principal subalgebra (a] generated by a. Locally pseudocomplemented semilattices are a natural generalisation of pseudocomplemented semilattices that stand in relation to such algebras as generalised Boolean algebras stand in relation to Boolean algebras. By an unpublished theorem of the author, the class of locally pseudocomplemented semilattices is a finitely based variety. This result, in conjunction with preceding remarks, invites the study of the following problem.

**Problem 4.2.24.** Is the class of locally pseudocomplemented semilattices a binary discriminator variety?

A further unpublished result of the author shows that, for any locally pseudocomplemented semilattice  $\mathbf{A}$ , the reduct  $\langle A; \rangle, 0 \rangle$  of  $\mathbf{A}$  is an implicative BCS-algebra. This result prompts the following question, which may be regarded as a generalisation of Problem 2.3.69.

**Problem 4.2.25.** Is an algebra  $\langle A; \rangle, 0 \rangle$  of type  $\langle 2, 0 \rangle$  an implicative BCSalgebra iff it is a  $\langle \rangle, 0 \rangle$ -reduct of a locally pseudocomplemented semilattice?

Locally pseudocomplemented semilattices were introduced by Grätzer in [101] in connection with the following problem, which we include for the sake of completeness.

**Problem 4.2.26.** [101, Problem 1§6.22, p. 67] Let A be a meet semilattice with 0 for which (a) is pseudocomplemented for each  $a \in A$ ; let S(a) denote the pseudocomplements in (a]. Characterise the family of Boolean algebras  $\{S(a) : a \in A\}$ .

4.2.27. Pre-BCK Quasilattices and BCK Paralattices. Recall that, when generalised to bands, Idziak's theory of BCK-[semi]lattices bifurcates (owing to the absence of commutativity). In one direction, Idziak's theory of BCK-[semi]lattices generalises to a theory of pre-BCK bands and pre-BCK quasilattices  $PQ_C$ ; while in another, it generalises to a theory of BCK bands and BCK paralattices  $BP_C$ . As it stands, the study of these complementary theories is largely orthogonal, since for each choice of C the varieties  $PQ_C$  and  $BP_C$  are incomparable. On the other hand, the results of §3.3.2 and §3.3.27 show that the study of both families  $PQ_C$ ,  $BP_C$  is relevant to the study of pointed fixedpoint discriminator and pointed ternary discriminator varieties. Prompted by the desire to provide a unified framework for the study of binary discriminator, pointed fixedpoint discriminator and pointed ternary discriminator varieties, these remarks engender the following question:

**Problem 4.2.28.** Find a common generalisation of pre-BCK bands [pre-BCK quasilattices] and BCK bands [BCK paralattices].

In §3.3.2 and §3.3.27, interest was naturally centred on those varieties  $PQ_c$ and  $BP_c$  for which  $\{\wedge, \div, 0\} \subseteq C \subseteq \{\wedge, \lor, \div, 0\}$ . However, the varieties  $PQ_c$ and  $BP_c$  may also be of interest for other choices of C. In particular, those varieties  $PQ_c$  and  $BP_c$  for which  $\{\lor, \div, 0\} \subseteq C$  would seem to merit attention, in view of the following two observations:

- 1. The variety SBA of skew Boolean algebras is termwise definitionally equivalent to a subvariety of  $PQ_{C'}$ ,  $C' = \{\vee, \backslash, 0\}$ ;
- 2. The variety SBIA of skew Boolean  $\cap$ -algebras is termwise definitionally equivalent to a subvariety of  $BP_{\mathcal{C}''}$ ,  $\mathcal{C}'' = \{\lor, /, 0\}$ .

A DESCRIPTION OF THE OWNER OWNER OF THE OWNER OWNER

To justify (1), it is sufficient to note SBA  $\models x \land y \approx (y \lor x) \setminus ((x \setminus y) \lor (y \setminus x))$ , whence any skew Boolean algebra  $\langle A; \land, \lor, \setminus, 0 \rangle$  is term equivalent to its own reduct  $\langle A; \lor, \setminus, 0 \rangle$ . Similarly, to justify (2) it is sufficient to note SBIA  $\models$  $x \land y \approx (y \lor x)/((x/y) \lor (y/x))$ , whence any skew Boolean  $\cap$ -algebra  $\langle A; \land, \lor, /, 0 \rangle$  is term equivalent to its own reduct  $\langle A; \lor, /, 0 \rangle$ .

**Problem 4.2.29.** Investigate those varieties  $PQ_c$ ,  $BP_c$  for which  $\{\lor, \div, 0\} \subseteq C \subseteq \{\land, \lor, \div, 0\}$ .

For any variety  $V \subseteq PQ_c$ , an easy modification of the proof of Theorem 2.1.36 shows the assertional logic S(V, 0) coincides with the assertional logic  $S(V_{\epsilon}, 0)$ of the variety of BCK-[semi]lattices  $V_{\epsilon}$  axiomatised relative to V by the quasiidentity (2.5). Thus the assertional logic S(V, 0) has a familiar description. In contrast, for any variety  $V \subseteq BP_c$  we have that V is 0-regular witness  $\{x \doteq y, y \doteq x\}$  (by Proposition 3.3.37), whence the equivalent algebraic semantics of the algebraisable assertional logic S(V, 0) is exactly V (by Theorem 1.8.15). By Remark 3.3.1, therefore, the following problem is apposite:

**Problem 4.2.30.** For varieties  $V \subseteq BP_c$ , investigate the assertional logics S(V, 0). In particular, investigate the assertional logics  $S(BP_c, 0)$  of the varieties  $BP_c$ .

In terms of gaining insight into the unfamiliar properties and behaviours of the deductive systems  $S(BP_c, 0)$  (recall Remark 3.3.1), the problem of obtaining a Gentzen-style axiomatisation for each  $S(BP_c, 0)$  would seem particularly relevant.

**Problem 4.2.31.** For each  $BP_c$ , give a Gentzen-style axiomatisation (if one exists) of the assertional logic  $S(BP_c, 0)$ .

Let K be a K-0-regular quasivariety. In traditional algebraic logic, a standard approach for constructing a Gentzen-style axiomatisation of S(K, 0) lies in conservatively extending S [193, Definition 9.1] with a multiplicative conjunction or *fusion* [199]. In algebraic logic, fusion of premisses is closely related to the calculation of residuals [86, Section 2]. Inasmuch as BCK-algebras satisfying Iséki's condition (S) are precisely the residuation *reducts* of poerims, the preceding remarks call (for each BP<sub>c</sub>) for the study of the class of all members of BP<sub>c</sub> for which the BCK-algebra reduct has condition (S).

**Problem 4.2.32.** For each  $BP_c$ , investigate the class of all members of  $BP_c$  for which the BCK-algebra reduct enjoys condition (S). In particular, study the class of all BCK skew lattices for which the BCK-algebra reduct has condition (S).

By the remarks of §1.1.1, the study of residuated structures in universal algebra was initiated by the papers of Kru'' [143] and Ward and Dilworth [235] on residuated lattices. Recently, the algebraic logic community has shown renewed interest in residuated lattices and their associated logics [43, 42, 127]. In view of these remarks and the initial motivations of this thesis, the following problem (which is related to Problem 4.2.2) would seem pertinent.

**Problem 4.2.33.** Can the theory of residuated lattices be usefully extended to skew lattices?

- J. C. Abbott, *Implicational algebras*, Bull. Math. Soc. Sci. Math. R. S. Roumanie **11** (1967), 3–23.
- [2] \_\_\_\_\_, Semi-Boolean algebras, Mat. Vesnik 4 (1967), 177–198.
- [3] P. Agliano, An algebraic investigation of linear logic, Rapporto Matematico 297, Università di Siena, 1996.
- [4] \_\_\_\_\_, On subtractive weak Brouwerian semilattices, Algebra Universalis 38 (1997), 214-220.
- [5] \_\_\_\_\_, Ternary deduction terms in residuated structures, Acta Sci. Math. (Szeged) 64 (1998), 397-429.
- [6] \_\_\_\_\_, Congruence quasi-orderability in subtractive varieties, Preprint, Jûne 2000.
- [7] \_\_\_\_\_, Fregean subtractive varieties with definable congruences, Preprint, June 2000.
- [8] P. Agliano and A. Ursini, Ideals and other generalisations of congruence classes, J. Austral. Math. Soc. Ser. A 53 (1992), 103-115.
- [9] \_\_\_\_\_, On subtractive varieties II: General properties, Algebra Universalis 36 (1996), 222-259.
- [10] \_\_\_\_\_, On subtractive varieties III: From ideals to congruences, Algebra Universalis 37 (1997), 296–333.

- [11] \_\_\_\_\_, On subtractive varieties IV: Definability of principal ideals, Algebra Universalis 38 (1997), 355–389.
- [12] A. R. Anderson and N. D. Belnap Jr., Entailment: The Logic of Relevance and Necessity, vol. 1, Princeton University Press, Princeton, 1975.
- [13] H. Andréka, A. Kurucz, I. Németi, and I. Sain, Applying algebraic logic; A general methodology, Manuscript (available via http://ci.cle.math-inst.hu/pub/algebraic-logic/meth.dvi), September 1994.
- [14] R. Balbes and P. Dwinger, *Distributive Lattices*, University of Missouri Press, Columbia, 1974.
- [15] J. T. Baldwin and J. Berman, The number of subdirectly irreducible algebras in a variety, Algebra Universalis 5 (1975), 379-389.
- [16] G. D. Barbour and J. G. Raftery, Quasivarieties of logic, regularity conditions and parameterised algebraisation, Internal Report 1/2001, University of Natal, Durban, 2001.
- [17] R. J. Bignall, Quasiprimal Varieties, and Components of Universal Algebras, Ph.D. thesis, The Flinders University of South Australia, 1976.
- [18] \_\_\_\_\_, A non-commutative multiple-valued logic, Proceedings of the Twenty-First International Symposium on Multiple-Valued Logic (D. M. Miller, ed.), IEEE Computer Society Press, Los Alamitos, 1991, pp. 49– 54.
- [19] R. J. Bignall and J. Leech, Skew Boolean algebras and discriminator varieties, Algebra Universalis 33 (1995), 387–398.
- [20] R. J. Bignall and M. Spinks, Propositional skew Boolean logic, Proceedings of the Twenty-Sixth International Symposium on Multiple-Valued Logic (R. Sipple, ed.), IEEE Computer Society Press, Los Alamitos, 1996, pp. 43-48.

- [21] \_\_\_\_\_, Multiple-valued logic as a programming language, Proceedings of the Twenty-Seventh International Symposium on Multiple-Valued Logic (R. Sipple and P. Storms, eds.), IEEE Computer Society Press, 1997, pp. 227-232.
- [22] \_\_\_\_\_, Multiple-valued logics for theorem-proving in first-order logic with equality, Proceedings of the Twenty-Eighth International Symposium on Multiple-Valued Logic (B. Werner, ed.), IEEE Computer Society Press, 1998, pp. 102-107.
- [23] G. Birkhoff, Lattice Theory, 2nd ed., Colloquium Publ., no. 25, American Mathematical Society, Providence, 1948.
- [24] W. J. Blok, Personal communication, January 2001.
- [25] W. J. Blok and S. B. La Falce, Komori identities in algebraic logic, Preprint, January 2001.
- [26] W. J. Blok and I. M. A. Ferreirim, Hoops and their implicational reducts (abstract), Algebraic Methods in Logic and Computer Science (C. Rauszer, ed.), Banach Centre Publications, vol. 28, Institute of Mathematics, Polish Academy of Sciences, Warszawa, 1993, pp. 219-230.
- [27] \_\_\_\_\_, On the structure of hoops, Algebra Universalis 43 (2000), 233– 257.
- [28] W. J. Blok and B. Jónsson, Algebraic structures for logic, Twenty-third Holiday Mathematics Symposium, New Mexico State University, Lecture Notes (available via http://emmy.NMSU.Edu/~holsymp/), January 1999.
- [29] W. J. Blok, P. Köhler, and D. Pigozzi, On the structure of varieties with equationally definable principal congruences II, Algebra Universalis 18 (1984), 334-379.
- [30] W. J. Blok and D. Pigozzi, On the structure of varieties with equationally definable principal congruences I, Algebra Universalis 15 (1982), 195– 227.

- [31] \_\_\_\_\_, Algebraisable logics, Mem. Amer. Math. Soc. 77 (1989), no. 396.
- [32] \_\_\_\_\_, The deduction theorem in algebraic logic (preliminary draft), Manuscript, July 1989.
- [33] \_\_\_\_\_, Local deduction theorems in algebraic logic, Algebraic logic (H. Andréka, J. D. Monk, and I. Németi, eds.), Colloquia Mathematica Societatis János Bolyai Budapest (Hungary), no. 54, North-Holland Publ. Co., New York, 1991, pp. 75–109.
- [34] \_\_\_\_\_, On the structure of varieties with equationally definable principal congruences III, Algebra Universalis 32 (1994), 545-608.
- [35] \_\_\_\_\_, On the structure of varieties with equationally definable principal congruences IV, Algebra Universalis 31 (1994), 1–35.
- [36] \_\_\_\_\_, Abstract algebraic logic and the deduction theorem, Preprint, June 2001.
- [37] W. J. Blok and J. G. Raftery, Failure of the congruence extension property in BCK-algebras and related structures, Math. Japon. 38 (1993), 633-638.
- [38] \_\_\_\_\_, On the quasivariety of BCK-algebras and its subvarieties, Algebra Universalis 33 (1995), 68-90.
- [39] \_\_\_\_\_, Varieties of commutative residuated integral pomonoids and their residuation subreducts, J. Algebra 190 (1997), 280-328.
- [40] \_\_\_\_\_, Ideals in quasivarieties of algebras, Models, Algebras and Proofs: Selected papers of the X Latin American Symposium on Mathematical Logic held in Bogotá (X. Caicedo and C. H. Montenegro, eds.), Marcel Dekker, New York, 1999, pp. 167–186.
- [41] \_\_\_\_\_, On assertional logic, Manuscript, February 2002.
- [42] W. J. Blok and C. J. van Alten, *Biresiduated algebras*, Manuscript, February 2001.

- [43] K. Blount, On the Structure of Residuated Lattices, Ph.D. thesis, Vanderbilt University, 1999.
- [44] B. Bosbach, Komplementäre Halbgruppen. Ein Beitrag zur instruktiven Idealtheorie kommutativer Halbgruppen, Math. Ann. 161 (1965), 279– 295.
- [45] \_\_\_\_\_, Komplementäre Halbgruppen. Axiomatik und Arithmetik, Fund. Math. 64 (1969), 257–287.
- [46] \_\_\_\_\_, Komplementäre Halbgruppen. Kongruenzen und Quotienten, Fund. Math. 69 (1970), 1–14.
- [47] D. Brignole, Equational characterisation of Nelson algebra, Notre Dame J. Formal Logic 10 (1969), 285–297.
- [48] J. R. Büchi and T. M. Owens, Complemented monoids and hoops, Manuscript, 1975.
- [49] \_\_\_\_\_, Skolem rings and their varieties, The Collected Works of J. Richard Büchi (S. MacLane and D. Siefkes, eds.), Springer-Verlag, New York, 1990, pp. 161-221.
- [50] S. Bulman-Fleming and H. Werner, Equational compactness in quasiprimal varieties, Algebra Universalis 7 (1977), 33-46.
- [51] M. W. Bunder, Simpler axioms for BCK-algebras and the connection between the axioms and the combinators B, C and K, Math. Japon. 26 (1981), 415-418.
- [52] \_\_\_\_\_, Some weak subsystems of BCK-algebra, Math. Sem. Notes Kobe Univ. 11 (1983), 171–176.
- [53] S. Burris, Existentially closed structures and Boolean products, Seminar Notes (available via http://www.thoralf.uwaterloo.ca/), May 1988.
- [54] S. Burris and J. Berman, A computer study of 3-element groupoids, Logic and Algebra: Proceedings of the Magari Conference (A. Ursini

and P. Agliano, eds.), Lecture Notes in Pure and Applied Mathematics, no. 180, Marcel Dekker, New York, 1996, pp. 379–430.

- [55] S. Burris and H. P. Sankappanavar, A Course in Universal Algebra, Graduate Texts in Mathematics, no. 78, Springer-Verlag, New York, 1981.
- [56] S. Burris and H. Werner, Sheaf constructions and their elementary properties, Trans. Amer. Math. Soc. 248 (1979), 269-309.
- [57] I. Chajda and R. Halaš, Finite basis of ideal terms in ideal determined varieties, Algebra Universalis 37 (1997), 243-252.
- [58] I. Chajda, R. Halaš, and I. G. Rosenberg, Ideals and the binary discriminator in universal algebra, Algebra Universalis 42 (1999), 239-251.
- [59] C. C. Chang, Algebraic analysis of many-valued logic, Trans. Amer. Math. Soc. 88 (1958), 467-490.
- [60] J. Cīrulis, Positive implicative BCK-algebras with condition (S) and implicative semilattices, Bull. Sec. Logic Polon. Acad. Sci. 28 (1999), 131–133.
- [61] J. P. Cleave, A Study of Logics, Oxford Logic Guides, no. 18, Clarendon Press, Oxford, 1991.
- [62] A. H. Clifford and G. B. Preston, The Algebraic Theory of Semigroups, vol. 1, Mathematical surveys, no. 7, American Mathematical Society, Providence, 1961.
- [63] W. H. Cornish, *Abelian Rickart Semirings*, Ph.D. thesis, The Flinders University of South Australia, 1970.
- [64] \_\_\_\_\_, 3-permutability and quasicommutative BCK-algebras, Math. Japon. 25 (1980), 477-496.
- [65] \_\_\_\_\_, Boolean skew algebras, Acta Math. Acad. Sci. Hungar. 36 (1980), 281-291.

and an and a second s

- [66] \_\_\_\_\_, A multiplier approach to implicative BCK-algebras, Math. Sem. Notes Kobe Univ. 8 (1980), 157-169.
- [67] \_\_\_\_, On positive implicative BCK-algebras, Math. Sem. Notes Kobe Univ. 8 (1980), 455-468.
- [68] \_\_\_\_\_, Varieties generated by finite BCK-algebras, Bull. Austral. Math. Soc. 22 (1980), 411-430.
- [69] \_\_\_\_\_, A large variety of BCK-algebras, Math. Japon. 26 (1981), 339– 344.
- [70] \_\_\_\_\_, On Iséki's BCK-algebras, Algebraic Structures and Applications: Proceedings of the First Western Australian Conference on Algebra (P. Schultz, C. E. Praeger, and R. P. Sullivan, eds.), Lecture Notes in Pure and Applied Mathematics, no. 74, Marcel Dekker, New York, 1982, pp. 101-122.
- [71] \_\_\_\_\_, Constructions for BCK-algebras, Math. Sem. Notes Kobe Univ.
   11 (1983), 1-7.
- [72] W. H. Cornish and R. C. Hickman, Weakly distributive semilattices, Acta Math. Acad. Sci. Hungar. 32 (1978), 5–16.
- [73] W. H. Cornish, T. Sturm, and T. Traczyk, Embedding of commutative BCK-algebras into distributive lattice BCK-algebras, Math. Japon. 29 (1984), 309-320.
- [74] B. Csákány, Characterisations of regular varieties, Acta Sci. Math.
   (Szeged) 31 (1970), 187-189.
- [75] H. B. Curry, Foundations of Mathematical Logic, Dover Publications Inc., New York, 1977.
- [76] J. Czelakowski, Algebraic aspects of deduction theorems, Studia Logica 44 (1985), 369-387.
- [77] J. Czelakowski and R. Jansana, Weakly algebraisable logics, J. Symbolic Logic 65 (2000), 641-668.
- [78] J. Czelakowski and D. Pigozzi, Fregean logics with the multiterm deduction theorem and their algebraisation, Mathematics Report M99-15, Iowa State University, 1999.
- [79] A. Diego, Sur les Algèbres de Hilbert, Collection de Logique Mathematique, Series A, no. 21, Gauthier-Villars, Paris, 1966.
- [80] K. Došen, A historical introduction to substructural logics, Substructural Logics (P. Schroeder-Heister and K. Došen, eds.), Studies in Logic and Computation, no. 2, Clarendon Press, Oxford, 1993, pp. 1–30.
- [81] J. Duda, Arithmeticity at 0, Czechoslovak Math. J. 37 (1987), 197–206.
- [82] W. A. Dudek, On BCC-algebras, Logique et Analyse 129-130 (1990), 103-111.
- [83] \_\_\_\_\_, On constructions of BCC-algebras, Selected Papers on BCKand BCI-Algebras 1 (1992), 93-96.
- [84] W. A. Dudek and X. Zhang, On ideals and congruences in BCC-algebras, Czechoslovak Math. J. 48 (1998), 21–29.
- [85] J. M. Dunn, Gaggle theory: An abstraction of Galois connections and residuation with applications to negation, implication, and various logical operations, Logics in AI: Proceedings of the European Workshop JELIA.
  '90 (J. van Eijck, ed.), Lecture Notes in Computer Science, no. 478, Springer, Berlin, 1991, pp. 31-51.
- [86] \_\_\_\_\_, Partial gaggles applied to logics with restricted structural rules, Substructural Logics (P. Schroeder-Heister and K. Došen, eds.), Studies in Logic and Computation, no. 2, Clarendon Press, Oxford, 1993, pp. 63– 108.
- [87] T. Evans, Products of points—some simple algebras and their identities, Amer. Math. Monthly 74 (1967), 362-372.
- [88] I. M. A. Ferreirim, On Varieties and Quasivarieties of Hoops and their Reducts, Ph.D. thesis, University of Illinois at Chicago, 1992.

- [89] K. Fichtner, Eine Bemerkung über Mannigfaltigkeiten universeller Algebren mit Idealen, Monatsch. deutsch. Akad. Wiss. (Berlin) 12 (1970), 21-25.
- [90] I. Fleischer, Every BCK-algebra is a set of residuables in an integral pomonoid, J. Algebra 119 (1988), 360-365.
- [91] J. M. Font and R. Jansana, A General Algebraic Semantics for Sentential Logics, Lecture Notes in Logic, no. 7, Springer-Verlag, Berlin, 1996.
- [92] J. M. Font, A. J. Rodríguez, and A. Torrens, Wajsberg algebras, Stochastica 8 (1984), 5-31.
- [93] J. M. Font and G. Rodríguez, Note on algebraic models for relevance logic, Z. Math. Logik Grundlagen Math. 36 (1990), 535-540.
- [94] E. Fried, G. Grätzer, and R. Quackenbush, Uniform congruence schemes, Algebra Universalis 10 (1980), 176–188.
- [95] E. Fried and A. F. Pixley, The dual discriminator function in universal algebra, Acta Sci. Math. (Szeged) 41 (1979), 83-100.
- [96] O. Frink, Pseudo-complements in semi-lattices, Duke Math. J. 29 (1962), 505-514.
- [97] J.-Y. Girard, *Linear logic*, Theoret. Comput. Sci. 50 (1987), 1–102.
- [98] J. S Golan, Linear Topologies on a Ring: An Overview, Pitman Research Notes in Mathematics, Series No. 159, Longman Scientific and Technical, Harlow, 1987.
- [99] G. Grätzer, Universal Algebra, The University Series in Higher Mathematics, D. Van Nostrand Co. Inc., Princeton, 1968.
- [100] \_\_\_\_\_, Stone algebras form an equational class (Remarks on lattice theory III), J. Austral. Math. Soc. 9 (1969), 308–309.
- [101] \_\_\_\_\_, Lattice Theory: First Concepts and Distributive Lattices, W. H. Freeman and Co., San Fransisco, 1971.

## References

- [102] G. Grätzer and H. Lakser, A note on the implicational class generated by a class of structures, Canad. Math. Bull. 16 (1973), 603-605.
- [103] P. A. Grillet, Semigroups: An Introduction to the Structure Theory, Monographs and Textbooks in Pure and Applied Mathematics, no. 193, Marcel Dekker, New York, 1995.
- [104] H. P. Gumm and A. Ursini, *Ideals in universal algebras*, Algebra Universalis 19 (1984), 45-54.
- [105] F. Guzmán, The poset structure of positive implicative BCK-algebras, Algebra Universalis 32 (1994), 398-406.
- [106] J. Hagemann and A. Mitschke, On n-permutable congruences, Algebra Universalis 3 (1973), 8-12.
- [107] R. Harrop, On the existence of finite models and decision procedures for propositional calculi, Proc. Cambridge Philos. Soc. 54 (1958), 1-13.
- [108] L. Heikin, An algebraic characterisation of quantifiers, Fund. Math. 37 (1950), 63-74.
- [109] D. Higgs, Dually residuated commutative monoids with identity element as least element do not form an equational class, Math. Japon. 29 (1984), 69-75.
- [110] D. Hobby and R. McKenzie, The Structure of Finite Algebras (Tame Congruence Theory), Contemporary Mathematics, no. 76, American Mathematical Society, Providence, 1988.
- [111] J. M. Howie, Fundamentals of Semigroup Theory, London Mathematical Society Monographs (New Series), no. 12, Clarendon Press, Oxford, 1995.
- [112] L. Humberstone, An intriguing logic with two implicational connectives, Notre Dame J. Formal Logic 41 (2000), 1-41.
- [113] \_\_\_\_\_, Note on a lemma of Komori, Sci. Math. Japon. 53 (2001), 347-352.

- [114] P. M. Idziak, On varieties of BCK-algebras, Math. Japon. 28 (1983), 157-162.
- [115] \_\_\_\_\_, Filters and congruence relations in BCK-semilattices, Math.
   Japon. 29 (1984), 975–980.
- [116] \_\_\_\_\_, Lattice operations in BCK-algebras, Math. Japon. 29 (1984), 839-846.
- [117] \_\_\_\_\_, Some theorems about BCK-semilattices, Math. Japon. 29 (1984), 919-921.
- [118] P. M. Idziak, K. Somczyńska, and A. Wroński, Equivalential algebras: A study of Fregean varieties, Preprint, September 1999.
- [119] Y. Imai and K. Iséki, On axiom systems of propositional calculi. XIV, Proc. Japan Acad. 42 (1966), 19-22.
- [120] K. Iséki, Cn ideals in BCK-algebras, Math. Sem. Notes Kobe Univ. 3 (1975), 1-12.
- [121] \_\_\_\_\_, BCK-algebras with condition (S), Math. Japon. 24 (1979), 107– 119.
- [122] \_\_\_\_\_, On BCI-algebras, Math. Sem. Notes Kobe Univ. 8 (1980), 125– 130.
- [123] \_\_\_\_\_, On BCK-algebras with condition (S), Math. Japon. (1980), 625– 626.
- [124] \_\_\_\_\_, Personal communication, September 1997.
- [125] K. Iséki and S. Tanaka, Ideal theory of BCK-algebras, Math. Japon. 21 (1976), 351-366.
- [126] \_\_\_\_\_, An introduction to the theory of BCK-algebras, Math. Japon.
   23 (1978), 1-26.
- [127] P. Jipsen and C. Tsinakis, A survey of residuated lattices, Preprint, October 2001.

## References

- [128] G. T. Jones, Pseudo Complemented Semi-Lattices, Ph.D. thesis, University of California at Los Angeles, 1972.
- [129] B. Jónsson, Congruence distributive varieties, Math. Japon. 42 (1995), 353-401.
- [130] J. Kabziński, Quasivarieties for BCK-logic, Bull. Sec. Logic Polon. Acad. Sci. 12 (1983), 130-133.
- [131] J. A. Kalman, Equational completeness and families of sets closed under subtraction, Nederl. Akad. Wetensch. Proc. Ser. A 63 (1960), 402-405.
- [132] T. Katriňák, A new proof of the Glivenko-Frink theorem, Bull. Soc. Roy.
   Sci. Liège 50 (1981), 171.
- [133] N. Kimura, Note on idempotent semigroups. I, Proc. Japan Acad. 33 (1957), 642-645.
- [134] \_\_\_\_\_, The structure of idempotent semigroups (I), Pacific J. Math. 8 (1958), 257-275.
- [135] P. Köhler, Brouwerian semilattices, Trans. Amer. Math. Soc. 268 (1981), 103-126.
- [136] P. Köhler and D. Pigozzi, Varieties with equationally definable principal congruences, Algebra Universalis 11 (1980), 213-219.
- [137] Y. Komori, Super-lukasiewicz implicational logics, Nagoya Math. J. 72 (1978), 127-133.
- [138] \_\_\_\_\_, The variety generated by BCC-algebras is finitely based, Rep. Fac. Sci. Shizuoka Univ. 17 (1983), 13-16.
- [139] \_\_\_\_\_, The class of BCC-algebras is not a variety, Math. Japon. 29 (1984), 391-394.
- [140] M. Kondo, Hilbert algebras are dual isomorphic to positive implicative BCK-algebras, Math. Japon. 49 (1999), 265-268.

- [141] \_\_\_\_\_, Remarks on BCK-semilattices, Math. Japon. 49 (1999), 463– 466.
- [142] T. Kowalski, The bottom of the lattice of BCK-varieties, Rep. Math. Logic 29 (1995), 87-93.
- [143] W. Krull, Axiomatische Begründung der allgemeinen Idealtheorie, Sitzungsber. physik. medizin. Soc. Erlangen 56 (1924), 47-63.
- [144] J. Lambek, The mathematics of sentence structure, Amer. Math. Monthly 65 (1958), 154-169.
- [145] G. Laslo and J. Leech, Green's equivalences on noncommutative lattices, Preprint, July 2001.
- [146] J. Leech, Skew lattices in rings, Algebra Universalis 26 (1989), 48-72.
- [147] \_\_\_\_\_, Skew Boolean algebras, Algebra Universalis 27 (1990), 497–506.
- [148] \_\_\_\_\_, Normal skew lattices, Semigroup Forum 44 (1992), 1-8.
- [149] \_\_\_\_\_, The geometric structure of skew lattices, Trans. Amer. Math. Soc. 335 (1993), 823-842.
- [150] \_\_\_\_\_, Recent developments in the theory of skew lattices, Semigroup Forum 52 (1996), 7-24.
- [151] \_\_\_\_\_, Personal communication, February 1998.
- [152] \_\_\_\_\_, Noncommutative lattices: Foundational issues and recent results, Presented at the Symposium on Universal Algebra and Multiple Valued Logic, Canad. Math. Soc. Winter Meeting, Kingston, December 1998.
- [153] J. Loś and R. Suszko, Remarks on sentential logics, Nederl. Akad. Wetensch. Proc. Ser. A 61 (1958), 177-183.
- [154] J. Lukasiewicz, Untersuchungen über den Aussagenkalkül, Comptes Rendus séances Soc. Sci. Lettres Varsovie (cl. III) 23 (1930), 30-50, Translated into English as [155].

## References

- [155] \_\_\_\_\_, Investigations into the sentential calculus, Selected Works of Jan Lukasiewicz (L. Borowski, ed.), Studies in Logic and the Foundations of Mathematics, North-Holland Publ. Co., Warsaw, 1970, pp. 131-152, Translation of [154] from the German.
- [156] A. I. Mal'cev, Subdirect products of models (Russian), Dokl. Akad. Nauk SSSR 109 (1956), 264–266.
- [157] A. A. Markov, Constructive logic (Russian), Uspehi Mat. Nauk 5 (1950), 187–188.
- [158] W. McCune, Otter 3.0 reference manual and guide, Tech. Report ANL-94/6 (available via ftp://info.mcs.anl.gov/pub/Otter), Argonne National Laboratory, Argonne, January 1994.
- [159] R. McKenzie, On spectra, and the negative solution of the decision problem for identities having a finite nontrivial model, J. Symbolic Logic 40 (1975), 186-196.
- [160] R. McKenzie, G. F. McNulty, and W. F. Taylor, Algebras, Lattices, Varieties, vol. 1, Wadsworth & Brooks/Cole, Monterey, 1987.
- [161] J. C. C. McKinsey and A. Tarski, The algebra of topology, Ann. of Math.
  45 (1944), 141–191.
- [162] \_\_\_\_\_, On closed elements in closure algebras, Ann. of Math. 47 (1946), 122–162.
- [163] D. McLean, *Idempotent semigroups*, Amer. Math. Monthly **61** (1954), 110-113.
- [164] J. Meng, Y. B. Jun, and S. M. Hong, Implicative semilattices are equivalent to positive implicative BCK-algebras with condition (S), Math. Japon. 48 (1998), 251-255.
- [165] C. A. Meredith and A. N. Prior, Notes on the axiomatics of the propositional calculus, Notre Dame J. Formal Logic 4 (1963), 171-187.

- [166] R. K. Meyer and R. Routley, Algebraic analysis of entailment, I, Logique et Anal. (N.S.) 15 (1972), 407-428.
- [167] A. Mitschke, Implication algebras are 3-permutable and 3-distributive, Algebra Universalis 1 (1971), 182–186.
- [168] A. Monteiro, Cours sur les algèbres de Hilbert et de Tarski, Tech. Report, Instituto Mat. Univ. del Sur, Bahía Blanca, 1960.
- [169] \_\_\_\_\_, Construction des algèbres de Nelson finies, Bull. Acad. Polon. Sci. Sér. Sci. Math. Astronom. Phys. 11 (1963), 359-362.
- [170] \_\_\_\_\_, Construction des Algèbres de Lukasiewicz trivalentes dans les algèbres de Boole Monadiques I, Math. Japon. 12 (1967), 1-23.
- [171] D. Mundici, MV-algebras are categorically equivalent to bounded commutative BCK-algebras, Math. Japon. 31 (1986), 889-894.
- [172] J. C. Muzio and T. C. Wesselkamper, Multiple-Valued Switching Theory, Adam Hilger, Bristol, 1986.
- [173] M. Nagayama, On a property of BCK-identities, Studia Logica 53 (1994), 227-234.
- [174] D. Nelson, Constructible falsity, J. Symbolic Logic 14 (1949), 16-28.
- [175] W. C. Nemitz, Implicative semi-lattices, Trans. Amer. Math. Soc. 117 (1965), 128-142.
- [176] H. Ono and Y. Komori, Logics without the contraction rule, J. Symbolic Logic 50 (1985), 169-201.
- [177] M. Pałasiński, On ideal and congruence lattices of BCK-algebras, Math. Japon. 26 (1981), 543-544.
- [178] \_\_\_\_\_, An embedding theorem for BCK-algebras, Math. Sem. Notes Kobe Univ. 10 (1982), 749-751.
- [179] \_\_\_\_\_, BCK-algebras and TD-terms, Manuscript, 1990.

- [180] M. Petrich, Lectures in Semigroups, John Wiley & Sons, London, 1977.
- [181] D. Pigozzi, Fregean algebraic logic, Algebraic logic (H. Andréka, J. D. Monk, and I. Németi, eds.), Colloquia Mathematica Societatis János Bolyai Budapest (Hungary), no. 54, North-Holland Publ. Co., New York, 1991, pp. 473-502.
- [182] \_\_\_\_\_, Abstract algebraic logic: Past, present and future. A personal view, Workshop on Abstract Algebraic Logic (J. M. Font, R. Jansana, and D. Pigozzi, eds.), Quaderns, no. 10, Centre de Recerca Matemàtica, Bellaterra, 1998.
- [183] A. F. Pixley, Functionally complete algebras generating distributive and permutable classes, Math. Z. 114 (1970), 361–372.
- [184] \_\_\_\_\_, The ternary discriminator function in universal algebra, Math. Ann. 191 (1971), 167–180.
- [185] J. Porte, Fifty years of deduction theorems, Proceedings of the Herbrand Symposium: Logic Colloquium '81 (J. Stern, ed.), North-Holland Publ. Co., Amsterdam, 1982, pp. 243-250.
- [186] A. N. Prior, Formal Logic, 2nd ed., Clarendon Press, Oxford, 1962.
- [187] J. G. Raftery, Ideal determined varieties need not be congruence 3permutable, Algebra Universalis 31 (1994), 293-297.
- [188] \_\_\_\_\_, Personal communication, March 2000.
- [189] \_\_\_\_\_, Personal communication, January 2002.
- [190] J. G. Raftery and T. Sturm, On ideal and congruence lattices of BCKsemilattices, Math. Japon. 32 (1987), 465-474.
- [191] \_\_\_\_\_, Tolerance numbers, congruence n-permutability and BCKalgebras, Czechoslovak Math. J. 42 (1992), 727-740.

- [192] J. G. Raftery and C. J. van Alten, On the algebra of noncommutative residuation: Polrims and left residuation algebras, Math. Japon. 46 (1997), 29-46.
- [193] \_\_\_\_\_, Residuation in commutative ordered monoids with minimal zero, Internal Report 6/2000, University of Natal, Durban, 2000.
- [194] H. Rasiowa, N-lattices and constructive logic with strong negation, Fund.
   Math. 46 (1958), 61-80.
- [195] \_\_\_\_\_, An Algebraic Approach to Non-Classical Logics, Studies in Logic and the Foundations of Mathematics, no. 78, North-Holland Publ. Co., Amsterdam, 1974.
- [196] H. Rasiowa and R. Sikorski, The Mathematics of Metamathematics, Polish Academy of Sciences Monographs in Mathematics, no. 41, Państwowe Wydawnictwo Naukowe, Warszawa, 1963.
- [197] W. Rautenberg, 2-element matrices, Studia Logica 40 (1981), 315-353.
- [198] G. Restall, An Introduction to Substructural Logics, Routledge, London, 2000.
- [199] \_\_\_\_\_, Personal communication, March 2001.
- [200] P. Ribenboim, Characterisation of the sup-complement in a distributive lattice with last element, Summa Brasil. Math. 2 (1949), 43-49.
- [201] A. Romanowska and T. Traczyk, On commutative BCK-algebras, Math. Japon. 25 (1980), 567–583.
- [202] \_\_\_\_\_, Commutative BCK-algebras. Subdirectly irreducible algebras and varieties, Math. Japon. 27 (1982), 35–48.
- [203] B. M. Schein, On the theory of restrictive semigroups (Russian), Izv.
   Vysš. Učebn. Zaved. Matematika 33 (1963), 152-154.
- [204] \_\_\_\_\_, Restrictive bisemigroups (Russian), Izv. Vysš. Učebn. Zaved. Matematika 44 (1965), 168–179, Translated into English as [206].

- [205] \_\_\_\_\_, Restrictive bisemigroups of mappings (Russian), Izv. Vysš. Učebn. Zaved. Matematika 56 (1967), 115–121, Translated into English as [207].
- [206] \_\_\_\_\_, Restrictive bisemigroups, Amer. Math. Soc. Transl. 100 (1972), 293-307, Translation of [204] from the Russian.
- [207] \_\_\_\_\_, Restrictive bisemigroups of mappings, Amer. Math. Soc. Transl.
   100 (1972), 308-316, Translation of [205] from the Russian.
- [208] \_\_\_\_\_, Bands of semigroups: Variations on a Clifford theme, Semigroup Theory and its Applications: Proceedings of the 1994 Conference Commemorating the Work of Alfred H. Clifford (K. H. Hofmann and M. W. Mislove, eds.), Cambridge University Press, New York, 1996, pp. 53-80.
- [209] A. Sendlewski, Some investigations of varieties of N-lattices, Studia Logica 43 (1984), 257-280.
- [210] M. Spinks, Automated Deduction in Non-Commutative Lattice Theory, Tech. Report 3/98, Monash University, Gippsland, June 1998.
- [211] \_\_\_\_\_, A non-classical extension of classical implicative propositional logic, Bull. Symbolic Logic 6 (2000), 255, (Presented at the Austral. Assoc. Logic Annual Conference, Melbourne, July 1999).
- [212] \_\_\_\_\_, On middle distributivity for skew lattices, Semigroup Forum 61 (2000), 341-345.
- [213] G. Sundholm, Systems of deduction, Handbook of Philosophical Logic (Volume I) (D. Gabbay and F. Guenthner, eds.), Synthese Library, vol. 164, D. Reidel Publ. Co., Dordrecht, 1983, pp. 133-188.
- [214] S. Tanaka, On ∧-commutative algebras, Math. Sem. Notes Kobe Univ.
   3 (1975), 59-64.
- [215] R. H. Thomason, A semantical study of constructible falsity, Z. Math. Logik Grundlagen Math. 15 (1969), 247-257.

医马克勒氏结核 医马克德氏 网络拉马尔马斯斯 人名英格兰尔英英格兰

- [216] T. Traczyk, On the variety of bounded commutative BCK-algebras, Math. Japon. 24 (1979), 283–292.
- [217] A. Urquhart, Free distributive pseudocomplemented lattices, Algebra Universalis 3 (1973), 13-15.
- [218] A. Ursini, Sulle varietá di algebre con una buona teoria degli ideali, Boll. Un. Mat. Ital. 6 (1972), 90-95.
- [219] \_\_\_\_, Osservazioni sulla varietá BIT, Boll. Un. Mat. Ital. 8 (1973), 205-211.
- [220] \_\_\_\_\_, Ideals and their calculus I, Rapporto Matematico 41, Università di Siena, 1981.
- [221] \_\_\_\_, Prime ideals in universal algebra, Acta Univ. Carolinae-Math. et Phys. 25 (1984), 75-87.
- [222] \_\_\_\_\_, On subtractive varieties I, Algebra Universalis 31 (1994), 204– 222.
- [223] \_\_\_\_\_, Semantical investigations of linear logic, Rapporto Matematico 291, Università di Siena, 1995.
- [224] \_\_\_\_\_, Algebraising natural deduction, I, Preprint, November 2000.
- [225] \_\_\_\_\_, On subtractive varieties V: Congruence modularity and the commutators, Algebra Universalis 43 (2000), 51-78.
- [226] \_\_\_\_\_, Personal communication, June 2001.
- [227] V. V. Vagner, Restrictive semigroups (Russian), Izv. Vysš. Učebn. Zaved.
   Mc ematika 31 (1962), 19-27.
- [228] \_\_\_\_\_, The theory of relations and the algebras of partial mappings (Russian), Theory of Semigroups and Applications I, Izdat. Saratov. Univ., Saratov, 1965, pp. 3-178.

- [229] C. J. van Alten, An algebraic study of residuated ordered monoids and logics without exchange and contraction, Ph.D. thesis, University of Natal, Durban, 1998.
- [230] C. J. van Alten and J. G. Raftery, On quasivariety semantics of fragments of intuitionistic propositional logic without exchange and contraction rules, Internal Report 5/99, University of Natal, Durban, 1999.
- [231] \_\_\_\_\_, On the lattice of varieties of residuation algebras, Internal Report 4/99, University of Natal, Durban, 1999.
- [232] I. D. Viglizzo, Algebras de Nelson, Master's thesis, Universidad Nacional del Sur, Bahía Blanca, 1999.
- [233] N. N. Vorob'ev, Constructive propositional calculus with strong negation (Russian), Dokl. Akad. Nauk SSSR 85 (1952), 456-468.
- [234] \_\_\_\_\_, The problem of provability in constructive propositional calculus with strong negation (Russian), Dokl. Akad. Nauk SSSR 85 (1952), 689– 692.
- [235] M. Ward and R. P. Dilworth, Residuated lattices, Trans. Amer. Math. Soc. 45 (1939), 335-354.
- [236] W. Wechler, Universal Algebra for Computer Scientists, Springer-Verlag, Berlin, 1992.
- [237] H. Werner, Discriminator-Algebras, Studien zur Algebra und ihre Anwendungen, no. 6, Akademie-Verlag, Berlin, 1978.
- [238] R. Wójcicki, Theory of Logical Calculi, Synthese Library, no. 199, Kluwer Academic Publ., Dordrecht, 1988.
- [239] A. Wroński, The degree of completeness of some fragments of intuitionistic propositional logic, Rep. Math. Logic 2 (1974), 55-62.
- [240] \_\_\_\_\_, BCK-algebras do not form a variety, Math. Japon. 28 (1983), 211-213.

- [241] \_\_\_\_\_, Reflections and distensions of BCK-algebras, Math. Japon. 28 (1983), 215-225.
- [242] \_\_\_\_\_, An algebraic motivation for BCK-algebras, Math. Japon. 30 (1985), 187-193.
- [243] A. Wroński and J. Kabziński, There is no largest variety of BCKalgebras, Math. Japon. 29 (1984), 545-549.
- [244] M. Yamada and N. Kimura, Note on idempotent semigroups. II, Proc. Japan Acad. 34 (1958), 110-112.
- [245] H. Yutani, On a system of axioms of a commutative BCK-algebra, Math. Sem. Notes Kobe Univ. 5 (1977), 255-256.
- [246] J. Zhang and H. Zhang, SEM: a System for Enumerating Models, Proceedings of the Fourteenth International Joint Conference on Artificial Intelligence (C. S. Mellish, ed.), Morgan Kaufmann, San Mateo, 1995, pp. 298-303.